MONOGRAPHS IN INEQUALITIES 3

Recent Advances in Hilbert-type Inequalities

A unified treatment of Hilbert-type inequalities Mario Krnić, Josip Pečarić, Ivan Perić and Predrag Vuković

REMEN

Recent Advances in Hilbert-type Inequalities

A unified treatment of Hilbert-type inequalities

Mario Krnić Faculty of Electrical Engineering and Computing University of Zagreb Zagreb, Croatia

Josip Pečarić

Faculty of Textile Technology University of Zagreb Zagreb, Croatia

Ivan Perić

Faculty of Food Technology and Biotechnology University of Zagreb Zagreb, Croatia

Predrag Vuković

Faculty of Teacher Education University of Zagreb Zagreb, Croatia



MONOGRAPHS IN INEQUALITIES 3

Recent Advances in Hilbert-type Inequalities

A unified treatment of Hilbert-type inequalities Mario Krnić, Josip Pečarić, Ivan Perić and Predrag Vuković

Consulting Editors

Ilko Brnetić Faculty of Electrical Engineering and Computing University of Zagreb Zagreb, Croatia

Sanja Varošanec Department of Mathematics University of Zagreb Zagreb, Croatia

1st edition

Copyright© by Element, Zagreb, 2012. www.element.hr/english/

Printed in Croatia. All rights reserved.

A CIP catalogue record for this book is available from the National and University Library in Zagreb under 830844.

ISBN 978-953-197-574-2

No part of this publication may be reproduced or distributed in any form or by any means, or stored in a data base or retrieval system, without the prior written permission of the publisher.

Preface

At the beginning of the 20th century, the following inequalities in discrete and integral forms have been established:

$$\sum_{m=1}^{\infty}\sum_{n=1}^{\infty}\frac{a_mb_n}{m+n} < \frac{\pi}{\sin\frac{\pi}{p}}\left[\sum_{m=1}^{\infty}a_m^p\right]^{\frac{1}{p}}\left[\sum_{n=1}^{\infty}b_n^q\right]^{\frac{1}{q}}$$

and

$$\int_0^\infty \int_0^\infty \frac{f(x)g(y)}{x+y} dx dy < \frac{\pi}{\sin\frac{\pi}{p}} \left[\int_0^\infty f^p(x) dx \right]^{\frac{1}{p}} \left[\int_0^\infty g^q(y) dy \right]^{\frac{1}{q}}.$$

The first inequality refers to non-negative sequences $(a_m)_{m \in \mathbb{N}} \in l^p$ and $(b_n)_{n \in \mathbb{N}} \in l^q$ which are not zero-sequences, while the second inequality holds for non-negative functions $f \in L^p(\mathbb{R}_+)$ and $g \in L^q(\mathbb{R}_+)$ which are not zero-functions. The common parameters, that is, the exponents p and q appearing in both inequalities are mutually conjugate, that is, they fulfill the condition $\frac{1}{p} + \frac{1}{q} = 1$, where p > 1. The above inequalities were first studied by David Hilbert at the end of the nineteenth century, hence, in his honor, they are referred to as the discrete and the integral Hilbert inequalities.

The Hilbert inequality is one of the most important inequalities in mathematical analysis. Applications of this inequality in diverse fields of mathematics have certainly contributed to its importance.

After its discovery, the Hilbert inequality was studied by numerous authors, who either reproved it using various techniques, or applied and generalized it in many different ways. Such generalizations included inequalities with more general kernels, weight functions and integration sets, extension to a multidimensional case, and so forth. The resulting relations are usually referred to as the Hilbert-type inequalities. On the other hand, Hardy, Littlewood and Pólya [33], noted that to every Hilbert-type inequality one can assign its equivalent form, in the sense that one implies another and vice versa. Such forms are usually called Hardy-Hilbert-type inequalities, since they are closely connected with another famous classical inequality, that is, the Hardy inequality. For a comprehensive inspection of the initial development of the Hilbert inequality, the reader is referred to a classical monograph [33].

Although classical, the Hilbert inequality is still of interest to numerous mathematicians. Nowadays, more than a century after its discovery, this problem area offers diverse possibilities for generalizations and extensions. The present book is a crown of decennial research of several authors in this area. More precisely, this book is based on some thirty significant papers dealing with Hilbert-type inequalities, published in the course of the last ten years.

We tried to provide a unified approach to Hilbert-type inequalities. More precisely, the original Hilbert inequality can be regarded in a more general setting, with integrals taken over a σ -finite measure space, and with a general kernel and weight functions. In addition, Hilbert-type inequalities can also be considered with exponents which are not mutually conjugate. On the other hand, in recent time a special emphasis has been dedicated to establishing methods for improving original Hilbert-type inequalities. These are the main topics we deal with in this monograph. The book is divided into ten chapters.

In **Chapter 1** a unified treatment of Hilbert-type inequalities with conjugate exponents is established. The most general form of the Hilbert inequality involves integrals taken over a σ -finite measure space, a general kernel and the weight functions. A special emphasis is given to Hilbert-type inequalities with homogeneous kernels. In addition, considerable attention is dedicated to finding the best possible constant factors appearing in some classes of inequalities. Observe here that a majority of results in this and other chapters will be given in two equivalent forms.

In **Chapter 2** we extend Hilbert-type inequalities derived in the previous chapter to the case of non-conjugate exponents. It should be observed here that the problem of finding the best possible constant factors is not considered in Chapter 2 since it seems to be very hard and remains still open. Additionally, we study some operators between certain weighted Lebesgue spaces, arising from the Hardy-Hilbert form of the corresponding Hilbert-type inequality.

In **Chapter 3** we consider Hilbert-type inequalities involving real valued kernel and weight functions defined on \mathbb{R}^n . Such results will be derived by virtue of the so-called Selberg integral formula.

In **Chapter 4** we derive two types of refined discrete Hilbert-type inequalities by means of the Euler-Maclaurin summation formula, depending on whether the corresponding kernel is of class C^2 or C^4 . In addition, some particular refinements are also established, due to the above summation formula.

In **Chapter 5** a different approach for improving discrete Hilbert-type inequalities is presented. Such improvements are derived by virtue of the Hermite-Hadamard inequality.

In **Chapter 6** we deal with refinements of some particular Hilbert-type inequalities involving the Laplace transform.

In **Chapter 7** a particular class of the so-called Hilbert-Pachpatte-type inequalities is studied. These inequalities are closely connected with Hilbert-type inequalities.

In **Chapter 8** another famous classical inequality, closely connected to the Hilbert inequality, is studied. That is the Hardy inequality. A unified treatment of Hardy-type inequalities with non-conjugate exponents is established.

In **Chapter 9** Hilbert-type inequalities are considered in a more general function space. Namely, all results in previous chapters were related to the weighted Lebesgue spaces. In this chapter Hilbert-type inequalities are established in the weighted Orlicz spaces.

In **Chapter 10** we list another set of recent results of numerous authors, interesting on its own right, which are closely connected with the theory exposed in this book. Namely, some related inequalities and refinements are given without the proof.

Throughout the whole book, presented results are compared with previously known from the literature. Moreover, at the end of a section or a chapter we cite the corresponding references for the results presented there. In addition, we also quote references which are closely connected with presented topics.

Since this monograph is based on numerous papers written by different authors, the terminology in the book is not quite unified. On the other hand, we suppose that the reader is very familiar with mathematical analysis and we mostly use the standard notation. However, to avoid misunderstandings, some extra notation and definitions are presented when it is necessary. Further, in statements of some theorems, conditions concerning convergence of series and integrals are omitted. If nothing else is explicitly stated, they are assumed to be convergent.

Authors

Contents

Preface

1	Hilb	ert-type inequalities with conjugate exponents	1
1.1 Historical overview			2
	1.2	A unified treatment of Hilbert-type inequalities	
		with conjugate exponents	5
	1.3	Applications to homogeneous kernels	9
	1.4	Examples. The best possible constants	14
		1.4.1 Integral case	14
		1.4.2 Discrete case	18
		1.4.3 Some further estimates	25
	1.5	Refined Hilbert-type inequalities	
		via the refined Hölder inequality	29
	1.6	Multidimensional Hilbert-type inequalities	32
		1.6.1 General form	32
		1.6.2 Application to homogeneous kernels	34
		1.6.3 Examples	38
		1.6.4 Inequalities with product-type homogeneous kernels	
		and Schur polynomials	45
2	Hilb	ert-type inequalities with non-conjugate exponents	53
	2.1	General form	55
	2.2	The case of a homogeneous kernel	59
	2.3	Godunova-type inequalities	64
	2.4	Multidimensional case	66
	2.5	Examples with hypergeometric functions	73
		2.5.1 Hilbert-type inequalities and Gaussian hypergeometric function .	74
		2.5.2 Hilbert-type inequalities and generalized hypergeometric functions	
		$_mF_n$	76
	2.6	Hilbert-type inequalities and related operators	84
		2.6.1 Hilbert-type inequalities involving a homogeneous	
		function of zero-degree	84
		-	

V

		2.6.2	On some related Hilbert-type operators	88		
		2.6.3	On some related Hardy-type operators	90		
		2.6.4	Applications	92		
3	Hilbert-type inequalities with vector variables					
	3.1	Explic	cit upper bounds for the doubly weighted Hardy-Littlewood-Sobolev			
		inequa	ality	96		
	3.2	Triline	ear version of standard Beta integral	101		
	3.3	Multip	ple Hilbert-type inequalities via the Selberg integral formula	105		
	3.4	The be	est constants	111		
	3.5	Some	related inequalities with norms	117		
4	Арр	lying tl	he Euler-Maclaurin summation formula	125		
	4.1	Inequa	alities for kernels of class C^2	127		
		4.1.1	Auxiliary results	127		
		4.1.2	Refined discrete Hilbert-type inequalities	131		
	4.2	Inequa	alities for kernels of class C^4	135		
		4.2.1	Auxiliary results	135		
		4.2.2	Refined discrete Hilbert-type inequalities	137		
	4.3	Some	particular refinements	142		
5	Applying the Hermite-Hadamard inequality					
	5.1	Basic	theorem and some remarks	150		
	5.2	A unit	fied approach to inequalities			
		with a	homogeneous kernel	153		
	5.3	Exam	ples with homogeneous kernels	155		
	5.4	A non	-conjugate example	160		
6	Hilb	ert-typ	e inequalities and the Laplace transform	163		
	6.1	The ca	ase of conjugate parameters	164		
	6.2	The ca	ase of non-conjugate parameters	168		
7	A cl	ass of F	Hilbert-Pachpatte-type inequalities	173		
•	7.1	Integr	al case	174		
	7.2	Discre	ete case	179		
	7.3	Non-c	conjugate exponents	181		
8	Gen	oral He	ardy-type inequalities with non-conjugate exponents	185		
U	81	Gener	al inequalities of the Hardy-type	187		
	8.2	Gener	al inequalities with dual Hardy-type kernel	191		
	83	Some	special Hardy-type kernels	171		
	0.5	and w	eight functions	193		
	84	Furthe	er analysis of parameters	199		
	8.5	Unifo	rm bounds of constant factors	202		
	8.6	Applia	cations	205		
	2.0	6 6				

_		
Х		

9	Hilbert-type inequalities in the weighted Orlicz spaces					
	9.1	Weighted Orlicz spaces and a two-dimensional Hilbert-type inequality	210			
	9.2	Multidimensional Hilbert-type inequality	211			
	9.3	A version of Hardy-Hilbert-type inequality	216			
	9.4	Some examples in the weighted Lebesgue spaces	218			
10 Some particular inequalities						
Bi	Bibliography					

Chapter 1

Hilbert-type inequalities with conjugate exponents

One of the most important inequalities in modern mathematics is the well-known Hilbert inequality. Applications of this inequality in various branches of mathematics have certainly contributed to its importance. David Hilbert was the first mathematician who started to deal with the Hilbert inequality, by considering its discrete form. He did not even think that he had opened the space for numerous researches whose results will be far-reaching and fruitful.

Shortly after discovering the discrete form, the integral form of the Hilbert inequality was also established, as well as the generalization for the case of conjugate exponents. During subsequent decades, the Hilbert inequality was also generalized in many different ways by some famous authors. Nowadays, more than a century after Hilbert's discovery, this problem area is still of interest and provides some possibilities for further generalizations.

In this chapter we present some basic generalizations of the Hilbert inequality. After the short historical overview, we expose a recent important generalization, which provides a unified treatment to this inequality with conjugate exponents. In particular, in that result the integrals are taken with σ -finite measures, which includes both integral and discrete case.

The above mentioned main result is then applied to homogeneous functions, which yields numerous interesting examples. Also, the consideration of such examples in particular settings yields numerous results, previously known from the literature. Moreover, all results presented in two-dimensional case can naturally be extended to a multidimensional case.

Finally, numerous inequalities in this chapter include the corresponding constant factor on their right-hand sides. By the classical Hilbert inequality such constant factor was the best possible in the sense that it cannot be replaced with the smaller constant so that the resulting inequality still remains valid. We shall also present here some recent results which include such best possible constant factors.

1.1 Historical overview

We begin this overview with a bilinear form

$$\sum_{m=1}^{\infty}\sum_{n=1}^{\infty}\frac{a_mb_n}{m+n},$$

associated to sequences of real numbers $(a_m)_{m \in \mathbb{N}}$ and $(b_n)_{n \in \mathbb{N}}$, which was first studied by D. Hilbert at the end of the nineteenth century. Hilbert discovered a natural upper bound of this double series and laid the foundations for the theory that will follow. Thus, we present here some basic theorems which arose immediately from Hilbert's considerations.

Theorem 1.1 Let p and q be mutually conjugate exponents, that is, $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $(a_m)_{m \in \mathbb{N}}$ and $(b_n)_{n \in \mathbb{N}}$ be any two sequences of non-negative real numbers such that $0 < \sum_{m=1}^{\infty} a_m^p < \infty$ and $0 < \sum_{n=1}^{\infty} b_n^n < \infty$. Then

$$\sum_{n=1}^{\infty}\sum_{n=1}^{\infty}\frac{a_m b_n}{m+n} < \frac{\pi}{\sin\frac{\pi}{p}} \left[\sum_{m=1}^{\infty}a_m^p\right]^{\frac{1}{p}} \left[\sum_{n=1}^{\infty}b_n^q\right]^{\frac{1}{q}}.$$
(1.1)

The integral form of the previous theorem reads as follows:

n

Theorem 1.2 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $f,g : \mathbb{R}_+ \to \mathbb{R}$ be any two non-negative Lebesgue measurable functions such that $0 < \int_0^{\infty} f^p(x) dx < \infty$ and $0 < \int_0^{\infty} g^q(y) dy < \infty$. Then

$$\int_0^\infty \int_0^\infty \frac{f(x)g(y)}{x+y} dx dy < \frac{\pi}{\sin\frac{\pi}{p}} \left[\int_0^\infty f^p(x) dx \right]^{\frac{1}{p}} \left[\int_0^\infty g^q(y) dy \right]^{\frac{1}{q}}.$$
 (1.2)

Remark 1.1 Suppose that *p* and *q* are mutually conjugate parameters, i.e. $\frac{1}{p} + \frac{1}{q} = 1$, and let p > 1. Then it follows that q > 1. On the other hand, if 0 , then <math>q < 0, and analogously, if 0 < q < 1, then p < 0.

As we see, in the above theorems the same constant factor appears on the right-hand sides of both inequalities. It was proved by Hardy and Riesz that this constant factor was the best possible.

Theorem 1.3 The constant $\pi/\sin(\pi/p)$ appearing in (1.1) and (1.2) is the best possible.

The previous three results are taken from the classical monograph [33], in a slightly altered form. The case of p = q = 2 in Theorem 1.1 was first proved by Hilbert in his lectures about integral equations. The lack of that old proof consisted in the fact that Hilbert didn't know to determine the optimal constant factor π . That drawback was removed by Shur in 1911, who also proved the integral version of the inequality. The extensions to arbitrary pair of positive mutually conjugate exponents are due to G.H. Hardy and M. Riesz.

Some other proofs, as well as various generalizations are due to the following mathematicians: L. Fejér, E. Francis, G. H. Hardy, J. Littlewood, H. Mulholland, P. Owen, G. Pólya, F. Riesz, M. Riesz, I. Schur, G. Szegö. Nevertheless, the inequalities (1.1) and (1.2) remained known as the discrete and the integral Hilbert inequalities. For more details about the initial development of the Hilbert inequality the reader is referred to [33, Chapter 9]. It should be noticed here that generalizations of inequalities (1.1) and (1.2) will be referred to as the Hilbert-type inequalities.

However, we provide two more results from the 1920s, which will also play an important role in further investigations. Namely, Hardy, Littlewood and Pólya noted that to every Hilbert-type inequality one can assign its equivalent form, in the sense that one implies another and vice versa. For example, the equivalent form assigned to inequality (1.1) is contained in the following theorem.

Theorem 1.4 Let p > 1 and let $(a_m)_{m \in \mathbb{N}}$ be the sequence of non-negative real numbers such that $0 < \sum_{m=1}^{\infty} a_m^p < \infty$. Then

$$\sum_{n=1}^{\infty} \left(\sum_{m=1}^{\infty} \frac{a_m}{m+n}\right)^p < \left(\frac{\pi}{\sin\frac{\pi}{p}}\right)^p \sum_{m=1}^{\infty} a_m^p.$$
(1.3)

Obviously, the integral analogue of inequality (1.3) is analogous, with the sum replaced with the integral, and a sequence with a non-negative real function. Such inequalities, derived from the Hilbert-type inequalities will be referred to as the Hardy-Hilbert-type inequalities. Moreover, the Hilbert-type and the Hardy-Hilbert-type inequalities will sometimes simply be referred to as the Hilbert-type inequalities.

Already at that time, the sharper version of inequality (1.1) was also known. That result is presented in the following theorem.

Theorem 1.5 Under the same assumptions as in Theorem 1.1, we have

$$\sum_{m=0}^{\infty} \sum_{n=0}^{\infty} \frac{a_m b_n}{m+n+1} < \frac{\pi}{\sin \frac{\pi}{p}} \left[\sum_{m=0}^{\infty} a_m^p \right]^{\frac{1}{p}} \left[\sum_{n=0}^{\infty} b_n^q \right]^{\frac{1}{q}}.$$
 (1.4)

Inequalities (1.1) and (1.4) are known in the literature as "the Hilbert double series theorems".

These theorems were inspiration to numerous mathematicians. During the 20th century numerous proofs, generalizations and applications of the Hilbert inequality were discovered and it would be impossible to count them here.

Nowadays, more than a century after the discovery of the Hilbert inequality, this research area is still interesting to numerous authors. As an illustration, we indicate here some generalizations obtained in the last ten years. One of the possible extensions arises from studying various kernels. Namely, in presented results such kernel was the function $K(x,y) = (x+y)^{-1}$. In 1998, considering the kernel $K(x,y) = (x+y)^{-s}$, s > 0, B. Yang was the first one to have included the well-known Beta function into the study of Hilbert-type inequalities (see [138]). Recall that the Beta function is an integral function defined by

$$B(a,b) = \int_0^1 t^{a-1} (1-t)^{b-1} dt, \qquad a,b > 0.$$
(1.5)

For example, in [152] one can find the following result in the integral form.

Theorem 1.6 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $s > 2 - \min\{p,q\}$. If $f, g : \mathbb{R}_+ \to \mathbb{R}$ are non-negative measurable functions such that $0 < \int_0^\infty x^{1-s} f^p(x) dx < \infty$ and $0 < \int_0^\infty y^{1-s} g^q(y) dy < \infty$, then

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{f(x)g(y)}{(x+y)^{s}} dx dy < B\left(\frac{p+s-2}{p}, \frac{q+s-2}{q}\right) \left[\int_{0}^{\infty} x^{1-s} f^{p}(x) dx\right]^{\frac{1}{p}} \left[\int_{0}^{\infty} y^{1-s} g^{q}(y) dy\right]^{\frac{1}{q}}$$
(1.6)

and

$$\int_{0}^{\infty} y^{(s-1)(p-1)} \left[\int_{0}^{\infty} \frac{f(x)}{(x+y)^{s}} dx \right]^{p} dy$$

< $B^{p} \left(\frac{p+s-2}{p}, \frac{q+s-2}{q} \right) \int_{0}^{\infty} x^{1-s} f^{p}(x) dx.$ (1.7)

Moreover, these two inequalities are equivalent and include the best possible constant factors on their right-hand sides.

The multidimensional extension of inequality (1.6), involving the usual Gamma function, has also been derived in the above mentioned paper [152]. Recall that the Gamma function is an integral

$$\Gamma(a) = \int_0^\infty t^{a-1} e^{-t} dt, \qquad a > 0.$$
(1.8)

Theorem 1.7 Suppose that p_i are mutually conjugate exponents, i.e. $\sum_{i=1}^{n} \frac{1}{p_i} = 1$, $p_i > 1$, i = 1, 2, ..., n, and let $s > n - \min_{1 \le i \le n} \{p_i\}$. If $f_i : \mathbb{R}_+ \to \mathbb{R}$, i = 1, 2, ..., n, are non-negative measurable functions satisfying $0 < \int_0^\infty x_i^{n-1-s} f_i^{p_i}(x_i) dx_i < \infty$, then

$$\int_{0}^{\infty} \cdots \int_{0}^{\infty} \frac{\prod_{i=1}^{n} f_{i}(x_{i})}{(\sum_{i=1}^{n} x_{i})^{s}} dx_{1} dx_{2} \dots dx_{n}$$

$$< \frac{1}{\Gamma(s)} \prod_{i=1}^{n} \Gamma\left(\frac{p_{i}+s-n}{p_{i}}\right) \left[\int_{0}^{\infty} x_{i}^{n-1-s} f_{i}^{p_{i}}(x_{i}) dx_{i}\right]^{\frac{1}{p_{i}}}.$$
(1.9)

4

Remark 1.2 The above notation for the Beta and the Gamma function will be used throughout the book. The basic relationship between the Beta and the Gamma functions is given by

$$B(a,b) = \frac{\Gamma(a)\Gamma(b)}{\Gamma(a+b)}, \ a,b > 0, \tag{1.10}$$

and this formula will often be exploited. For more details about the Beta and the Gama functions, as well as about their meromorphic extensions to the set of complex numbers, the reader is referred to [1].

It is interesting that the *n*-dimensional inequality (1.9) also posses its equivalent form, which will be discussed in this chapter.

On the other hand, another possible generalization of the presented results is the investigation of the inequalities of the same type, but where the integrals are taken over a bounded interval in \mathbb{R}_+ . Guided by that idea, K. Jichang and T. Rassias [42], obtained the following result.

Theorem 1.8 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a non-negative homogeneous symmetric function of degree -s, where $\max{\{\frac{1}{p}, \frac{1}{q}\}} < s$. If K(1, y) is strictly decreasing in y and $f, g : \mathbb{R}_+ \to \mathbb{R}$ are non-negative measurable functions, then

$$\int_{a}^{b} \int_{a}^{b} K(x,y)f(x)g(y)dxdy$$

$$\leq \left[\int_{a}^{b} \left(I(q) - \varphi(q,x)\right)x^{1-s}f^{p}(x)dx\right]^{\frac{1}{p}} \left[\int_{a}^{b} \left(I(p) - \varphi(p,y)\right)y^{1-s}g^{q}(y)dy\right]^{\frac{1}{q}},$$
(1.11)

where

$$\varphi(r,x) = \left(\frac{a}{x}\right)^{1-\frac{1}{r}} \int_0^1 K(1,u) u^{-\frac{1}{r}} du + \left(\frac{x}{b}\right)^{s+\frac{1}{r}-1} \int_0^1 K(1,u) u^{s+\frac{1}{r}-2} du,$$

 $I(r) = \int_0^\infty K(1, u) u^{-\frac{1}{r}} du, r \in \{p, q\}, and 0 \le a < b \le \infty.$

In the next section, the integrals will be taken over more general sets.

1.2 A unified treatment of Hilbert-type inequalities with conjugate exponents

In the previous historical overview we have seen the classical Hilbert inequality in both discrete and integral case. Moreover, throughout years numerous extensions of these inequalities were derived. However, all these results were given in either integral form, with respect to the Lebesgue measure, or in the discrete form.

The main objective of this section is to present a general result which unifies the integral and discrete cases. This can be done by observing a more general integral. Namely, the classical Hilbert inequality is a consequence of the Hölder inequality and the Fubini theorem. In general, the Fubini theorem holds for the integrals with σ -finite measures, therefore, such measures will be considered.

The most important examples of σ -finite measures are the Lebesgue measure and the counting measure. The Lebesgue measure yields the classical integral case, while the counting measure provides the discrete case.

Further, it is well-known that if one of the mutually conjugate exponents in the Hölder inequality is negative, then the sign of the inequality is reversed (see [103]). Hence, we shall also be concerned with the Hilbert-type inequalities with the reversed sign of inequality. Such inequalities will be referred to as the reverse inequalities.

Now we present the most general form of the Hilbert inequality in the setting described above. It should be noticed here that we suppose that all integrals converge, and such types of conditions will often be omitted. Moreover, integrals will be taken over a general measure space. Results that follow are provided in two equivalent forms: the Hilbert and the Hardy-Hilbert forms.

Theorem 1.9 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let Ω be a measure space with positive σ -finite measures μ_1 and μ_2 . Let $K : \Omega \times \Omega \to \mathbb{R}$ and $\varphi, \psi : \Omega \to \mathbb{R}$ be non-negative measurable functions. If the functions F and G are defined by $F(x) = \int_{\Omega} K(x,y)\psi^{-p}(y)d\mu_2(y)$ and $G(y) = \int_{\Omega} K(x,y)\varphi^{-q}(x)d\mu_1(x)$, then for all non-negative measurable functions f and g on Ω the inequalities

$$\int_{\Omega} \int_{\Omega} K(x,y) f(x)g(y)d\mu_1(x)d\mu_2(y)$$

$$\leq \left[\int_{\Omega} \varphi^p(x)F(x)f^p(x)d\mu_1(x)\right]^{\frac{1}{p}} \left[\int_{\Omega} \psi^q(y)G(y)g^q(y)d\mu_2(y)\right]^{\frac{1}{q}}$$
(1.12)

and

$$\int_{\Omega} G^{1-p}(y) \psi^{-p}(y) \left[\int_{\Omega} K(x,y) f(x) d\mu_1(x) \right]^p d\mu_2(y)$$

$$\leq \int_{\Omega} \varphi^p(x) F(x) f^p(x) d\mu_1(x)$$
(1.13)

hold and are equivalent.

If 0 , then the reverse inequalities in (1.12) and (1.13) are valid, as well as the inequality

$$\int_{\Omega} F^{1-q}(x) \varphi^{-q}(x) \left[\int_{\Omega} K(x,y) g(y) d\mu_2(y) \right]^q d\mu_1(x)$$

$$\leq \int_{\Omega} \psi^q(y) G(y) g^q(y) d\mu_2(y).$$
(1.14)

Proof. The left-hand side of inequality (1.12) can be rewritten in the following form:

$$\int_{\Omega} \int_{\Omega} K(x,y) f(x) g(y) d\mu_1(x) d\mu_2(y) = \int_{\Omega} \int_{\Omega} K(x,y) f(x) \frac{\varphi(x)}{\psi(y)} g(y) \frac{\psi(y)}{\varphi(x)} d\mu_1(x) d\mu_2(y)$$

6

Now, applying the Hölder inequality to the above relation yields

$$\begin{split} &\int_{\Omega} \int_{\Omega} K(x,y) f(x) g(y) d\mu_1(x) d\mu_2(y) \\ &\leq \left[\int_{\Omega} \int_{\Omega} K(x,y) f^p(x) \frac{\varphi^p(x)}{\psi^p(y)} d\mu_1(x) d\mu_2(y) \right]^{\frac{1}{p}} \\ &\times \left[\int_{\Omega} \int_{\Omega} K(x,y) g^q(y) \frac{\psi^q(y)}{\varphi^q(x)} d\mu_1(x) d\mu_2(y) \right]^{\frac{1}{q}}. \end{split}$$

Finally, using the Fubini theorem and definitions of functions F and G we obtain (1.12).

Now, we are going to show the equivalence of inequalities (1.12) and (1.13). For that sake, suppose that inequality (1.12) holds. Defining the function g by

$$g(y) = G^{1-p}(y)\psi^{-p}(y) \left[\int_{\Omega} K(x,y)f(x)d\mu_1(x) \right]^{p-1},$$

taking into account that $\frac{1}{p} + \frac{1}{q} = 1$, and using (1.12), we have

$$\begin{split} &\int_{\Omega} G^{1-p}(y) \psi^{-p}(y) \left[\int_{\Omega} K(x,y) f(x) d\mu_1(x) \right]^p d\mu_2(y) \\ &= \int_{\Omega} \int_{\Omega} K(x,y) f(x) g(y) d\mu_1(x) d\mu_2(y) \\ &\leq \left[\int_{\Omega} \varphi^p(x) F(x) f^p(x) d\mu_1(x) \right]^{\frac{1}{p}} \left[\int_{\Omega} \psi^q(y) G(y) g^q(y) d\mu_2(y) \right]^{\frac{1}{q}} \\ &= \left[\int_{\Omega} \varphi^p(x) F(x) f^p(x) d\mu_1(x) \right]^{\frac{1}{p}} \\ &\times \left[\int_{\Omega} G^{1-p}(y) \psi^{-p}(y) \left[\int_{\Omega} K(x,y) f(x) d\mu_1(x) \right]^p d\mu_2(y) \right]^{\frac{1}{q}}, \end{split}$$

that is, we get (1.13).

On the other hand, suppose that inequality (1.13) holds. In that case, another use of the Hölder inequality yields

$$\begin{split} &\int_{\Omega} \int_{\Omega} K(x,y) f(x) g(y) d\mu_1(x) d\mu_2(y) \\ &= \int_{\Omega} \left[\psi^{-1}(y) G^{-\frac{1}{q}}(y) \int_{\Omega} K(x,y) f(x) d\mu_1(x) \right] \psi(y) G^{\frac{1}{q}}(y) g(y) d\mu_2(y) \\ &\leq \left[\int_{\Omega} G^{1-p}(y) \psi^{-p}(y) \left(\int_{\Omega} K(x,y) f(x) d\mu_1(x) \right)^p d\mu_2(y) \right]^{\frac{1}{p}} \\ &\times \left[\int_{\Omega} \psi^q(y) G(y) g^q(y) d\mu_2(y) \right]^{\frac{1}{q}} \\ &\leq \left[\int_{\Omega} \varphi^p(x) F(x) f^p(x) d\mu_1(x) \right]^{\frac{1}{p}} \left[\int_{\Omega} \psi^q(y) G(y) g^q(y) d\mu_2(y) \right]^{\frac{1}{q}}, \end{split}$$

which implies (1.12). Therefore, inequalities (1.12) and (1.13) are equivalent.

The reverse inequalities, as well as their equivalence, are derived in the same way by virtue of the reverse Hölder inequality. $\hfill \Box$

Remark 1.3 The equality in the previous theorem is possible if and only if it holds in the Hölder inequality, that is, if

$$\left[f(x)\frac{\varphi(x)}{\psi(y)}\right]^p = C\left[g(y)\frac{\psi(y)}{\varphi(x)}\right]^q, \quad \text{a. e. on } \Omega,$$

where *C* is a positive constant. In that case we have

$$f(x) = C_1 \varphi^{-q}(x)$$
 and $g(y) = C_2 \psi^{-p}(y)$ a.e. on Ω , (1.15)

for some constants C_1 and C_2 , which is possible if and only if

$$\int_{\Omega} F(x)\varphi^{-q}(x)d\mu_1(x) < \infty \quad \text{and} \quad \int_{\Omega} G(y)\psi^{-p}(y)d\mu_2(y) < \infty.$$
(1.16)

Otherwise, the inequalities in Theorem 1.9 are strict.

In some applications of the previous theorem it will be more convenient to bound the functions F(x) and G(y). Of course, such result follows immediately from Theorem 1.9.

Theorem 1.10 Suppose that the assumptions as in Theorem 1.9 are fulfilled and let $F_1, G_1 : \Omega \to \mathbb{R}$ be non-negative measurable functions such that $F(x) \leq F_1(x)$ and $G(y) \leq G_1(y)$, a. e. on Ω . Then the inequalities

$$\int_{\Omega} \int_{\Omega} K(x,y) f(x)g(y)d\mu_1(x)d\mu_2(y)$$

$$\leq \left[\int_{\Omega} \varphi^p(x)F_1(x)f^p(x)d\mu_1(x)\right]^{\frac{1}{p}} \left[\int_{\Omega} \psi^q(y)G_1(y)g^q(y)d\mu_2(y)\right]^{\frac{1}{q}}$$
(1.17)

and

$$\int_{\Omega} G_1^{1-p}(y) \psi^{-p}(y) \left[\int_{\Omega} K(x,y) f(x) d\mu_1(x) \right]^p d\mu_2(y)$$

$$\leq \int_{\Omega} \varphi^p(x) F_1(x) f^p(x) d\mu_1(x)$$
(1.18)

hold and are equivalent.

If $0 , <math>F(x) \ge F_1(x)$, and $G(y) \le G_1(y)$, then the reverse inequalities in (1.17) and (1.18) hold, as well as the inequality

$$\int_{\Omega} F_1^{1-q}(x) \varphi^{-q}(x) \left[\int_{\Omega} K(x,y) g(y) d\mu_2(y) \right]^q d\mu_1(x)$$

$$\leq \int_{\Omega} \psi^q(y) G_1(y) g^q(y) d\mu_2(y).$$
(1.19)

The reverse inequalities are also equivalent.

Remark 1.4 The general Hilbert-type inequalities presented in this section have been obtained by M. Krnić and J. Pečarić in [53].

1.3 Applications to homogeneous kernels

Theorem 1.9 from the previous section has unified the classical integral and discrete cases of the Hilbert inequality. In order to approach to some well-known results from the literature, we study here some particular choices of kernels and weight functions.

In this section we consider homogeneous kernels of negative degree of homogeneity, equipped with some additional properties. Recall that a function $K : \Omega \times \Omega \to \mathbb{R}$ is said to be homogeneous of degree -s, s > 0, if $K(tx, ty) = t^{-s}K(x, y)$ for every $x, y \in \Omega$ and $t \in \mathbb{R}$ such that $tx, ty \in \Omega$. In addition, for such homogeneous function we define $k(\alpha)$ as

$$k(\alpha) = \int_0^\infty K(1, u) u^{-\alpha} du, \qquad (1.20)$$

provided that the above integral converges for $1 - s < \alpha < 1$.

We study here the integral case, that is, the Lebesgue integral. The integrals are taken over an arbitrary interval of non-negative real numbers, i.e. $(a,b) \subseteq \mathbb{R}_+$, $0 \le a < b \le \infty$, and the weight functions are chosen to be power functions.

Theorem 1.11 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $K : (a,b) \times (a,b) \to \mathbb{R}$ be a non-negative homogeneous function of degree -s, s > 0, strictly decreasing in both variables. If A_1 and A_2 are real parameters such that $A_1 \in (\frac{1-s}{q}, \frac{1}{q})$, $A_2 \in (\frac{1-s}{p}, \frac{1}{p})$, then for all non-negative measurable functions $f, g : (a,b) \to \mathbb{R}$ the inequalities

$$\int_{a}^{b} \int_{a}^{b} K(x,y)f(x)g(y)dxdy$$

$$\leq \left[\int_{a}^{b} \left(k(pA_{2}) - \varphi_{1}(pA_{2},x)\right)x^{1-s+p(A_{1}-A_{2})}f^{p}(x)dx\right]^{\frac{1}{p}}$$

$$\times \left[\int_{a}^{b} \left(k(2-s-qA_{1}) - \varphi_{2}(2-s-qA_{1},y)\right)y^{1-s+q(A_{2}-A_{1})}g^{q}(y)dy\right]^{\frac{1}{q}} \quad (1.21)$$

and

$$\int_{a}^{b} \left(k(2-s-qA_{1})-\varphi_{2}(2-s-qA_{1},y)\right)^{1-p} y^{(p-1)(s-1)+p(A_{1}-A_{2})} \\ \times \left[\int_{a}^{b} K(x,y)f(x)dx\right]^{p} dy \\ \leq \int_{a}^{b} \left(k(pA_{2})-\varphi_{1}(pA_{2},x)\right) x^{1-s+p(A_{1}-A_{2})} f^{p}(x)dx$$
(1.22)

hold and are equivalent, where

$$\varphi_1(\alpha, x) = \left(\frac{a}{x}\right)^{1-\alpha} \int_0^1 K(1, u) u^{-\alpha} du + \left(\frac{x}{b}\right)^{s+\alpha-1} \int_0^1 K(u, 1) u^{s+\alpha-2} du$$

$$\varphi_2(\alpha, y) = \left(\frac{a}{y}\right)^{s+\alpha-1} \int_0^1 K(u, 1) u^{s+\alpha-2} du + \left(\frac{y}{b}\right)^{1-\alpha} \int_0^1 K(1, u) u^{-\alpha} du$$

If $0 , <math>b = \infty$, and K(x,y) is strictly decreasing in x and strictly increasing in y, then the reverse inequalities in (1.21) and (1.22) are valid for every $A_1 \in (\frac{1}{q}, \frac{1-s}{q})$ and $A_2 \in (\frac{1-s}{p}, \frac{1}{p})$, as well as the inequality

$$\int_{a}^{\infty} \left(k(pA_{2}) - \varphi_{1}(pA_{2}, x) \right)^{1-q} x^{(q-1)(s-1)+q(A_{2}-A_{1})} \left[\int_{a}^{\infty} K(x, y)g(y)dy \right]^{q} dx$$

$$\leq \int_{a}^{\infty} \left(k(2-s-qA_{1}) - \varphi_{2}(2-s-qA_{1}, y) \right) y^{1-s+q(A_{2}-A_{1})} g(y)^{q} dy.$$

Moreover, if 0 , <math>a = 0, and K(x, y) is strictly increasing in x and strictly decreasing in y, then the reverse inequalities in (1.21) and (1.22) hold for every $A_1 \in (\frac{1}{q}, \frac{1-s}{q})$ and $A_2 \in (\frac{1-s}{p}, \frac{1}{p})$, as well as the inequality

$$\begin{split} &\int_0^b \left(k(pA_2) - \varphi_1(pA_2, x)\right)^{1-q} x^{(q-1)(s-1)+q(A_2-A_1)} \left[\int_0^b K(x, y)g(y)dy\right]^q dx \\ &\leq \int_0^b \left(k(2-s-qA_1) - \varphi_2(2-s-qA_1, y)\right) y^{1-s+q(A_2-A_1)}g(y)^q dy. \end{split}$$

Proof. We only prove inequality (1.21). After substituting the power functions $\varphi(x) = x^{A_1}$ and $\psi(y) = y^{A_2}$ in (1.12), the homogeneity of the kernel *K* and the substitution $u = \frac{y}{x}$ yield the following relation:

$$\begin{split} &\int_{a}^{b} \int_{a}^{b} K(x,y) f(x) g(y) dx dy \\ &\leq \left[\int_{a}^{b} x^{1-s+p(A_{1}-A_{2})} \left(\int_{\frac{a}{x}}^{\frac{b}{x}} K(1,u) u^{-pA_{2}} du \right) f^{p}(x) dx \right]^{\frac{1}{p}} \\ &\times \left[\int_{a}^{b} y^{1-s+q(A_{2}-A_{1})} \left(\int_{\frac{y}{b}}^{\frac{y}{a}} K(1,u) u^{qA_{1}+s-2} du \right) g^{q}(y) dy \right]^{\frac{1}{q}}. \end{split}$$

In addition, considering the function $l(y) = y^{\alpha-1} \int_0^y K(1,u) u^{-\alpha} du$, $\alpha < 1$, the integration by parts yields equality

$$l'(y) = y^{\alpha - 2} \int_0^y u^{1 - \alpha} \frac{\partial K(1, u)}{\partial u} du$$

Since the kernel *K* is strictly decreasing in both variables, it follows that $l'(y) < 0, y \in \mathbb{R}_+$, that is, *l* is strictly decreasing on \mathbb{R}_+ .

On the other hand, since

$$\int_{\frac{a}{x}}^{\frac{b}{x}} K(1,u)u^{-pA_2} du = \int_{0}^{\infty} K(1,u)u^{-pA_2} du - \int_{0}^{\frac{a}{x}} K(1,u)u^{-pA_2} du - \int_{0}^{\frac{x}{b}} K(u,1)u^{pA_2+s-2} du,$$

10

and due to the fact that l is strictly decreasing on \mathbb{R}_+ , we obtain the estimate

$$\int_{\frac{a}{x}}^{\frac{b}{x}} K(1,u) u^{-pA_2} du \le k(pA_2) - \varphi_1(pA_2,x)$$

and similarly,

$$\int_{\frac{y}{b}}^{\frac{2}{a}} K(1,u) u^{qA_1+s-2} du \le k(2-s-qA_1) - \varphi_2(2-s-qA_1,y),$$

so the result follows from Theorem 1.9. Note also that the intervals defining the parameters A_1 and A_2 arise from the assumption on the convergence of integral (1.20).

Remark 1.5 If the kernel *K* in the previous theorem is a symmetric function, then $k(2 - s - qA_1) = k(qA_1)$. Then, setting $A_1 = A_2 = \frac{1}{pq}$ in Theorem 1.11, provided that max $\{\frac{1}{p}, \frac{1}{q}\} < s$, we get Theorem 1.8 (see also [42]).

Remark 1.6 In order to justify the convergence interval (1 - s, 1) for the integral $k(\alpha)$ defined by (1.20), observe that the homogeneity of the kernel *K* implies the following sequence of identities:

$$k(\alpha) = \int_0^\infty K\left(\frac{1}{u}, 1\right) u^{-s-\alpha} du = \int_0^\infty K(u, 1) u^{s+\alpha-2} du$$

On the other hand, assuming that *K* is strictly decreasing in each argument, *K* is strictly positive on $\mathbb{R}_+ \times \mathbb{R}_+$. In particular, for $\alpha \ge 1$, monotonicity of *K* in the second argument and the fact that K(1,1) > 0 yield

$$k(\alpha) = \int_0^\infty K(1, u) u^{-\alpha} du \ge \int_0^1 K(1, u) u^{-\alpha} du \ge K(1, 1) \int_0^1 u^{-\alpha} du = \infty.$$

Analogous result holds also for $\alpha \leq 1 - s$, since

$$k(\alpha) = \int_0^\infty K(u,1)u^{s+\alpha-2}du \ge \int_0^1 K(u,1)u^{s+\alpha-2}du$$
$$\ge K(1,1)\int_0^1 u^{s+\alpha-2}du = \infty.$$

Therefore, the interval (1 - s, 1), considered in definition (1.20), covers all arguments α for which $k(\alpha)$ may converge. The same conclusion on convergence of $k(\alpha)$ can be drawn if we consider a function *K* increasing in each argument and such that K(1,1) > 0.

It is interesting to consider a particular case of the previous theorem, that is, when the integrals are taken over the whole set \mathbb{R}_+ . Then, a = 0, $b = \infty$, and we obtain the corresponding inequalities for an arbitrary non-negative homogeneous function of degree -s.

Corollary 1.1 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a non-negative homogeneous function of degree -s, s > 0. If A_1 and A_2 are real parameters such that $A_1 \in (\frac{1-s}{q}, \frac{1}{q}), A_2 \in (\frac{1-s}{p}, \frac{1}{p})$, then for all non-negative measurable functions $f, g : \mathbb{R}_+ \to \mathbb{R}$ the inequalities

$$\int_{0}^{\infty} \int_{0}^{\infty} K(x,y) f(x) g(y) dx dy$$

$$\leq L \left[\int_{0}^{\infty} x^{1-s+p(A_{1}-A_{2})} f^{p}(x) dx \right]^{\frac{1}{p}} \left[\int_{0}^{\infty} y^{1-s+q(A_{2}-A_{1})} g^{q}(y) dy \right]^{\frac{1}{q}}$$
(1.23)

and

$$\int_{0}^{\infty} y^{(p-1)(s-1)+p(A_{1}-A_{2})} \left[\int_{0}^{\infty} K(x,y)f(x)dx \right]^{p} dy$$

$$\leq L^{p} \int_{0}^{\infty} x^{1-s+p(A_{1}-A_{2})} f^{p}(x)dx \qquad (1.24)$$

hold and are equivalent, where $L = k^{\frac{1}{p}} (pA_2) k^{\frac{1}{q}} (2 - s - qA_1)$.

If $0 , then the reverse inequalities in (1.23) and (1.24) are valid for every <math>A_1 \in (\frac{1}{q}, \frac{1-s}{q})$ and $A_2 \in (\frac{1-s}{p}, \frac{1}{p})$, as well as the inequality

$$\int_{0}^{\infty} x^{(q-1)(s-1)+q(A_{2}-A_{1})} \left[\int_{0}^{\infty} K(x,y)g(y)dy \right]^{q} dx$$

$$\leq L^{q} \int_{0}^{\infty} y^{1-s+q(A_{2}-A_{1})}g^{q}(y)dy.$$
(1.25)

Inequalities (1.23) and (1.24), as well as their reverse inequalities are equivalent. Moreover, equality in the above relations holds if and only if f = 0 or g = 0 a.e. on \mathbb{R}_+ .

Proof. The proof follows immediately from Theorem 1.11 by substituting a = 0 and $b = \infty$. Moreover, condition (1.15) gives the nontrivial case of equality in (1.23), while condition (1.16) leads to the divergent integrals. Hence, the observed inequalities are strict, unless f = 0 or g = 0 a. e. on \mathbb{R}_+ .

In the sequel we consider some generalizations of Theorem 1.11. For example, utilizing the substitution $u = x + \mu$ and $v = y + \mu$, $\mu \ge 0$, we have:

Theorem 1.12 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $K : (a + \mu, b + \mu) \times (a + \mu, b + \mu) \rightarrow \mathbb{R}$ be a non-negative homogeneous function of degree -s, s > 0, strictly decreasing in both variables. If A_1 and A_2 are real parameters such that $A_1 \in (\frac{1-s}{q}, \frac{1}{q})$, $A_2 \in (\frac{1-s}{p}, \frac{1}{p})$, then for all non-negative measurable functions $f, g: (a, b) \rightarrow \mathbb{R}$ the inequalities

$$\begin{aligned} &\int_{a}^{b} \int_{a}^{b} K(x+\mu,y+\mu)f(x)g(y)dxdy \\ &\leq \left[\int_{a}^{b} \left(k(pA_{2})-\psi_{1}(pA_{2},x,\mu)\right)(x+\mu)^{1-s+p(A_{1}-A_{2})}f^{p}(x)dx\right]^{\frac{1}{p}} \\ &\times \left[\int_{a}^{b} \left(k(2-s-qA_{1})-\psi_{2}(2-s-qA_{1},y,\mu)\right)(y+\mu)^{1-s+q(A_{2}-A_{1})}g^{q}(y)dy\right]^{\frac{1}{q}} \end{aligned}$$
(1.26)

and

$$\int_{a}^{b} (k(2-s-qA_{1})-\psi_{2}(2-s-qA_{1},y,\mu))^{1-p}(y+\mu)^{(p-1)(s-1)+p(A_{1}-A_{2})} \\ \times \left[\int_{a}^{b} K(x+\mu,y+\mu)f(x)dx\right]^{p}dy \\ \leq \int_{a}^{b} (k(pA_{2})-\psi_{1}(pA_{2},x,\mu))(x+\mu)^{1-s+p(A_{1}-A_{2})}f^{p}(x)dx$$
(1.27)

hold and are equivalent, where

$$\begin{split} \psi_1(\alpha, x, \lambda) &= \left(\frac{a+\lambda}{x+\lambda}\right)^{1-\alpha} \int_0^1 K(1, u) u^{-\alpha} du + \left(\frac{x+\lambda}{b+\lambda}\right)^{s+\alpha-1} \int_0^1 K(u, 1) u^{s+\alpha-2} du, \\ \psi_2(\alpha, y, \lambda) &= \left(\frac{a+\lambda}{y+\lambda}\right)^{s+\alpha-1} \int_0^1 K(u, 1) u^{s+\alpha-2} du + \left(\frac{y+\lambda}{b+\lambda}\right)^{1-\alpha} \int_0^1 K(1, u) u^{-\alpha} du. \end{split}$$

If $0 , <math>b = \infty$, and K(x,y) is strictly decreasing in x and strictly increasing in y, then the reverse inequalities in (1.26) and (1.27) hold for all $A_1 \in (\frac{1}{q}, \frac{1-s}{q})$ and $A_2 \in (\frac{1-s}{p}, \frac{1}{p})$, as well as the inequality

$$\int_{a}^{\infty} (k(pA_{2}) - \psi_{1}(pA_{2}, x, \mu))^{1-q} (x + \mu)^{(q-1)(s-1)+q(A_{2}-A_{1})} \\ \times \left[\int_{a}^{\infty} K(x + \mu, y + \mu)g(y)dy \right]^{q} dx \\ \leq \int_{a}^{\infty} (k(2 - s - qA_{1}) - \psi_{2}(2 - s - qA_{1}, y, \mu)) (y + \mu)^{1-s+q(A_{2}-A_{1})}g^{q}(y)dy.$$
(1.28)

Moreover, inequalities (1.26) and (1.27), as well as their reverses, are equivalent.

Remark 1.7 Considering Theorem 1.12 with a symmetric kernel and parameters $A_1 = A_2 = \frac{2\lambda}{pq}$, provided that $0 < 1 - \frac{2\lambda}{p} < s$, $0 < 1 - \frac{2\lambda}{q} < s$, we obtain the corresponding result from [42].

Remark 1.8 Some other ways of generalizing Theorem 1.11 arise from various substitutions. For example, in [53] the authors also use the substitution $u = Ax^{\alpha}$ and $v = By^{\beta}$, where $A, B, \alpha, \beta > 0$. Such results are here omitted. It should be noticed here that the results in this section are taken from the above mentioned paper [53]. In addition, for some more specific Hilbert-type inequalities with a homogeneous kernel the reader is referred to [164] and [181].

1.4 Examples. The best possible constants

This section is dedicated to Hilbert-type inequalities with some particular homogeneous kernels and weight functions. Numerous interesting examples will be given here. Moreover, the best possible constant factors will be derived in some particular settings.

1.4.1 Integral case

We start with the classical integral case. We are concerned here with Corollary 1.1 from the previous section. It is not hard to see that this corollary covers Theorems 1.2 and 1.6, presented in the historical overview at the beginning of this chapter.

Namely, if $K(x,y) = (x+y)^{-s}$, s > 0, then the integral (1.20) is expressed in terms of the Beta function, that is, $k(\alpha) = B(1 - \alpha, s + \alpha - 1)$. Hence, in this setting the constant factor *L* on the right-hand sides of inequalities (1.23) and (1.24) takes the form

$$L = B^{\frac{1}{p}} (1 - pA_2, s + pA_2 - 1) B^{\frac{1}{q}} (1 - qA_1, s + qA_1 - 1).$$

Moreover, if $A_1 = A_2 = \frac{2-s}{pq}$, then the above constant coincides with the constant factor on the right-hand side of inequality (1.6). Hence, Corollary 1.1 can be regarded as an extension of both Theorems 1.2 and 1.6.

Further, if $K(x,y) = \max\{x,y\}^{-s}$, s > 0, then the above constant *L*, included in (1.23) and (1.24), reads

$$L = \frac{s}{(1 - pA_2)^{\frac{1}{p}} (1 - qA_1)^{\frac{1}{q}} (s + pA_2 - 1)^{\frac{1}{p}} (s + qA_1 - 1)^{\frac{1}{q}}}$$

Similarly, for $A_1 = A_2 = \frac{2-s}{pq}$ the above constant factor reduces to

$$L = \frac{pqs}{(p+s-2)(q+s-2)},$$

and the resulting Hilbert-type inequality coincides with the one from [42].

On the other hand, Hilbert-type inequalities in Theorems 1.2 and 1.6, as well as the above mentioned result from [42], include the best possible constant factor.

Our main task is to determine conditions under which the constant factor $L = k^{\frac{1}{p}} (pA_2)k^{\frac{1}{q}}$ (2-s-qA₁) is the best possible in inequalities (1.23) and (1.24). Observe that inequalities (1.2) and (1.6) include the best possible constant factors without any exponent. Guided by that fact we are going to simplify the constant factor *L* from Corollary 1.1. More precisely, if we set the condition

$$pA_2 + qA_1 = 2 - s, \tag{1.29}$$

then the constant factor *L* in Corollary 1.1 reduces to $L = k(pA_2)$.

In the sequel, we are going to show that, under the above condition (1.29) and assuming some weak conditions on the kernel, inequalities in Corollary 1.1 include the best possible constant factors. In order to prove that result we need the following lemma.

Lemma 1.1 Let p and q be conjugate parameters with p > 1. If $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ is a non-negative measurable function such that K(1,t) is bounded on (0,1), then

$$\int_{1}^{\infty} x^{-\varepsilon - 1} \left(\int_{0}^{1/x} K(1, t) t^{-pA_2 - \frac{\varepsilon}{q}} dt \right) dx = O(1), \ \varepsilon \to 0^+, \tag{1.30}$$

where $A_2 < \frac{1}{p}$.

Proof. Using the assumptions, we have $K(1,t) \le C$ for some C > 0 and every $t \in (0,1)$. Let $\varepsilon > 0$ be such that $\varepsilon < pq(\frac{1}{p} - A_2)$. We have

$$\int_{1}^{\infty} x^{-1-\varepsilon} \left(\int_{0}^{1/x} K(1,t) t^{-pA_2 - \frac{\varepsilon}{q}} dt \right) dx \le C \int_{1}^{\infty} x^{-1-\varepsilon} \left(\int_{0}^{1/x} t^{-pA_2 - \frac{\varepsilon}{q}} dt \right) dx$$
$$= \frac{C}{1-pA_2 - \frac{\varepsilon}{q}} \int_{1}^{\infty} x^{pA_2 + \frac{\varepsilon}{q} - \varepsilon - 2} dx = \frac{C}{\left(1 - pA_2 - \frac{\varepsilon}{q}\right) \left(1 - pA_2 + \frac{\varepsilon}{p}\right)},$$

wherefrom (1.30) follows.

Theorem 1.13 Suppose that the assumptions of Corollary 1.1 are fulfilled. Additionally, if the kernel $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ is such that K(1,t) is bounded on (0,1), and if the parameters A_1 and A_2 fulfill condition (1.29), then the constants $L = k(pA_2)$ and $L^p = k^p(pA_2)$ are the best possible in both inequalities (1.23) and (1.24).

Proof. For this purpose, with $0 < \varepsilon < pq(\frac{1}{p} - A_2)$, set $f(x) = x^{-qA_1 - \frac{\varepsilon}{p}} \chi_{[1,\infty)}(x)$ and $g(y) = y^{-pA_2 - \frac{\varepsilon}{q}} \chi_{[1,\infty)}(y)$, where χ_A is the characteristic function of a set *A*. Now, suppose that there exists a smaller constant 0 < M < L such that inequality (1.23) holds. Let *I* denote the right-hand side of (1.23). Then,

$$I = M \left(\int_{1}^{\infty} x^{-\varepsilon - 1} dx \right)^{\frac{1}{p}} \left(\int_{1}^{\infty} y^{-\varepsilon - 1} dy \right)^{\frac{1}{q}} = \frac{M}{\varepsilon}.$$
 (1.31)

Applying respectively the Fubini theorem, substitution $t = \frac{y}{x}$, and Lemma 1.1, we have

$$\int_{0}^{\infty} \int_{0}^{\infty} K(x,y) f(x)g(y)dxdy$$

$$= \int_{1}^{\infty} x^{-qA_{1}-\frac{\varepsilon}{p}} \left(\int_{1}^{\infty} K(x,y)y^{-pA_{2}-\frac{\varepsilon}{q}}dy \right) dx$$

$$= \int_{1}^{\infty} x^{-\varepsilon-1} \left(\int_{0}^{\infty} K(1,t)t^{-pA_{2}-\frac{\varepsilon}{q}}dt - \int_{0}^{x^{-1}} K(1,t)t^{-pA_{2}-\frac{\varepsilon}{q}}dt \right) dx$$

$$= \frac{1}{\varepsilon} \left[k \left(pA_{2} + \frac{\varepsilon}{q} \right) + o(1) \right].$$
(1.32)

From (1.23), (1.31), and (1.32) we get

$$k\left(pA_2 + \frac{\varepsilon}{q}\right) + o(1) < M. \tag{1.33}$$

Now, letting $\varepsilon \to 0^+$, relation (1.33) yields a contradiction with the assumption $M < L = k(pA_2)$.

Finally, equivalence of inequalities (1.23) and (1.24) means that the constant $L^p = k^p(pA_2)$ is also the best possible in (1.24). The proof is now completed.

As we see, the previous theorem covers the problem of finding the best possible constant factors for a quite weak condition on homogeneous kernel and parameters A_1 , A_2 satisfying (1.29). We have already considered the kernel $K(x,y) = (x+y)^{-s}$, s > 0. This kernel fulfills the above mentioned condition from Theorem 1.13, hence, the best possible constant in that case takes the form $B(1 - pA_2, 1 - qA_1)$. Hilbert-type inequalities with this kernel and parameters A_1 , A_2 were extensively studied in recent papers [10], [11], [52], [53], and [54].

Some other examples of the best possible constant factors arise from various choices of kernels. For example, considering the kernel $K(x,y) = (x+y+\max\{x,y\})^{-s}$, s > 0, the best possible constant factor $k(pA_2)$ from Theorem 1.13 becomes

$$\frac{2^{-s}}{pA_2+s-1}F(s,s+pA_2-1;s+pA_2;-1/2)+\frac{2^{-s}}{1-pA_2}F(s,1-pA_2;2-pA_2;-1/2),$$

where $F(\alpha, \beta; \gamma; z)$ denotes the Gaussian hypergeometric function, that is,

$$F(\alpha,\beta;\gamma;z) = \frac{\Gamma(\gamma)}{\Gamma(\beta)\Gamma(\gamma-\beta)} \int_0^1 \frac{t^{\beta-1}(1-t)^{\gamma-\beta-1}}{(1-zt)^{\alpha}} dt, \ \gamma > \beta > 0, z < 1.$$
(1.34)

The above kernel with degree of homogeneity equal to -1 was also discussed in [81].

We conclude this subsection with some particular Hilbert-type inequalities equipped with homogeneous kernels of degree -1, involving the best possible constant factors.

16

Remark 1.9 Setting $s = 1, A_1 = A_2 = \frac{1}{pq}$ in Corollary 1.1, inequalities (1.23) and (1.24) become respectively

$$\int_0^\infty \int_0^\infty K(x,y)f(x)g(y)dxdy \le k\left(1/q\right) \left[\int_0^\infty f^p(x)dx\right]^{\frac{1}{p}} \left[\int_0^\infty g^q(y)dy\right]^{\frac{1}{q}}$$
(1.35)

and

$$\int_0^\infty \left[\int_0^\infty K(x,y) f(x) dx \right]^p dy \le k^p \left(1/q \right) \int_0^\infty f^p(x) dx.$$
(1.36)

The following kernels *K* are homogenous with bounded K(1,t) on (0,1). For each of these functions we compute constants L = k(1/q) and $L_2 = k(1/2)$, that is, when p = q = 2:

$$\begin{split} K(x,y) &= \frac{1}{x + y + \max\{x,y\}}, \\ L &= \frac{1}{2}qF\left(1, \frac{1}{q}; 1 + \frac{1}{q}; -\frac{1}{2}\right) + \frac{1}{2}pF\left(1, \frac{1}{p}; 1 + \frac{1}{p}; -\frac{1}{2}\right), \\ L_2 &= \sqrt{2}(\pi - 2\arctan\sqrt{2}); \\ K(x,y) &= \frac{1}{x + y + \min\{x,y\}}, \\ L &= qF\left(1, \frac{1}{q}; 1 + \frac{1}{q}; -2\right) + pF\left(1, \frac{1}{p}; 1 + \frac{1}{p}; -2\right), L_2 = 2\sqrt{2}\arctan\sqrt{2}; \\ K(x,y) &= \frac{1}{|x - y| + \max\{x,y\}}, \\ L &= \frac{1}{2}qF\left(1, \frac{1}{q}; 1 + \frac{1}{q}; \frac{1}{2}\right) + \frac{1}{2}pF\left(1, \frac{1}{p}; 1 + \frac{1}{p}; \frac{1}{2}\right), L_2 = 2\arctan\frac{1}{\sqrt{2}}; \\ K(x,y) &= \frac{1}{x + y - \min\{x,y\}} = \frac{1}{\max\{x,y\}}, L = pq, L_2 = 4; \\ K(x,y) &= \frac{1}{x + y - \frac{2}{x + \frac{1}{y}}}, L_2 = \sqrt{\frac{2}{3}}\pi; \\ K(x,y) &= \frac{1}{x + y - \frac{2}{x + \frac{1}{y}}}, L = \frac{\pi}{2}\left(\frac{1}{\cos\frac{\pi}{2p}} + \frac{1}{\cos\frac{\pi}{2q}}\right), L_2 = \pi\sqrt{2}; \\ K(x,y) &= \frac{1}{x + y - \frac{\sqrt{2}}{x + \frac{1}{y}}}, L_2 = \frac{4\pi}{3\sqrt{3}}; \\ K(x,y) &= \frac{1}{x + y - \sqrt{xy}}, L_2 = \frac{8\pi}{3\sqrt{3}}; \\ K(x,y) &= \frac{x^{\lambda - 1} + y^{\lambda - 1}}{x^{\lambda} + y^{\lambda}}, L = \frac{\pi}{\lambda}\left(\cot\frac{\pi}{\lambda p} + \cot\frac{\pi}{\lambda q}\right), \lambda > 1; \end{split}$$

$$K(x,y) = \frac{\log y - \log x}{y - x}, L = \frac{\pi^2}{\left(\sin \frac{\pi}{p}\right)^2}, L_2 = \pi^2.$$

Since parameters s = 1, $A_1 = A_2 = \frac{1}{pq}$ fulfill condition $pA_2 + qA_1 = 2 - s$, all these constant factors are the best possible in both inequalities (1.35) and (1.36).

1.4.2 Discrete case

Discrete case of the Hilbert inequality is more complicated than the integral one. Namely, in order to obtain discrete forms of the corresponding integral inequalities, it is necessary to do some further estimates, which requires some additional conditions.

In Section 1.1 we encountered the Hilbert double series theorems, those were inequalities (1.1) and (1.4). Moreover, the corresponding equivalent form assigned to (1.1) is inequality (1.3), while the equivalent form assigned to (1.4) was derived in [142].

Recently, M. Krnić and J. Pečarić (see [52]), obtained the following discrete version of the Hilbert inequality with conjugate parameters p, q > 1 and real parameters $A, B, \alpha, \beta, s > 0$,

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \frac{a_m b_n}{(Am^{\alpha} + Bn^{\beta})^s} \le M \left[\sum_{m=1}^{\infty} m^{\alpha(1-s) + \alpha p(A_1 - A_2) + (p-1)(1-\alpha)} a_m^p \right]^{\frac{1}{p}} \times \left[\sum_{n=1}^{\infty} n^{\beta(1-s) + \beta q(A_2 - A_1) + (q-1)(1-\beta)} b_n^q \right]^{\frac{1}{q}}, (1.37)$$

where $(a_m)_{m \in \mathbb{N}}$, $(b_n)_{n \in \mathbb{N}}$ are non-negative real sequences, $A_1 \in \left(\max\{\frac{1-s}{q}, \frac{\alpha-1}{\alpha q}\}, \frac{1}{q}\right), A_2 \in \left(\max\{\frac{1-s}{p}, \frac{\beta-1}{\beta p}\}, \frac{1}{p}\right)$ and

$$M = \alpha^{-\frac{1}{q}} \beta^{-\frac{1}{p}} A^{\frac{2-s}{p} + A_1 - A_2 - 1} B^{\frac{2-s}{q} + A_2 - A_1 - 1}$$

$$\times B^{\frac{1}{p}} (1 - pA_2, s - 1 + pA_2) B^{\frac{1}{q}} (1 - qA_1, s - 1 + qA_1)$$

The equivalent form that corresponds to (1.37) is also derived in [52].

Similarly, considering parameters ϕ , ψ , and λ , such that $0 < \phi$, $\psi \le 1$ and $\phi + \psi = \lambda$, B. Yang [162], obtained the following pair of equivalent inequalities

$$\sum_{n=1}^{\infty} \sum_{m=1}^{\infty} \frac{\log\left(\frac{m}{n}\right) a_m b_n}{m^{\lambda} - n^{\lambda}} < \left[\frac{\pi}{\lambda \sin\left(\frac{\pi\phi}{\lambda}\right)}\right]^2 \left[\sum_{n=1}^{\infty} n^{p(1-\phi)-1} a_n^p\right]^{\frac{1}{p}} \left[\sum_{n=1}^{\infty} n^{q(1-\psi)-1} b_n^q\right]^{\frac{1}{q}}$$
(1.38)

and

$$\sum_{n=1}^{\infty} n^{p\psi-1} \left[\sum_{m=1}^{\infty} \frac{\log\left(\frac{m}{n}\right) a_m b_n}{m^{\lambda} - n^{\lambda}} \right]^p < \left[\frac{\pi}{\lambda \sin\left(\frac{\pi\phi}{\lambda}\right)} \right]^{2p} \sum_{n=1}^{\infty} n^{p(1-\phi)-1} a_n^p, \quad (1.39)$$

18

which hold for all non-negative conjugate exponents and non-negative sequences fulfilling $0 < \sum_{n=1}^{\infty} n^{p(1-\phi)-1} a_n^p < \infty$ and $0 < \sum_{n=1}^{\infty} n^{q(1-\psi)-1} b_n^q < \infty$. Moreover, the constant factors included in the right-hand sides of inequalities are the best possible.

On the other hand, B. Yang and T. M. Rassias [152] (see also [149]), studied the kernel expressed in terms of the logarithm function. They obtained the following pair of equivalent inequalities,

$$\sum_{m=2}^{\infty} \sum_{n=2}^{\infty} \frac{a_m b_n}{\log mn} < \frac{\pi}{\sin \pi/p} \left[\sum_{n=2}^{\infty} n^{p-1} a_n^p \right]^{\frac{1}{p}} \left[\sum_{n=2}^{\infty} n^{q-1} b_n^q \right]^{\frac{1}{q}}$$
(1.40)

and

$$\sum_{n=2}^{\infty} \frac{1}{n} \left(\frac{a_m}{\log mn} \right)^p < \left[\frac{\pi}{\sin \pi/p} \right]^p \sum_{n=2}^{\infty} n^{p-1} a_n^p, \tag{1.41}$$

which hold for non-negative conjugate exponents and non-negative sequences such that $0 < \sum_{n=2}^{\infty} n^{p-1} a_n^p < \infty$ and $0 < \sum_{n=2}^{\infty} n^{q-1} b_n^q < \infty$. Moreover, the constant factors $\pi / \sin(\pi/p)$ and $[\pi / \sin(\pi/p)]^p$, on the right-hand sides of inequalities (1.40) and (1.41), are the best possible. Observe that the above inequality (1.40) for p = q = 2 is also known as the Mulholland inequality.

Clearly, the kernel involved in the previous two inequalities, as well as in (1.37) is non-homogeneous, while the kernel in (1.38) and (1.39) is homogeneous.

However, utilizing suitable substitutions, these non-homogeneous kernels can also be interpreted as the homogeneous ones. Thus, in the sequel we provide discrete forms of Hilbert-type inequalities with a general homogeneous kernel. The same conditions as in the integral case are assumed on the convergence of the integral $k(\alpha)$, defined by (1.20).

The following result contains discrete Hilbert-type inequalities for a homogeneous kernel in both equivalent forms. Discrete weight functions involve here differentiable real functions. In addition, for the reader's convenience, we introduce here the following notation: H(r), r > 0, denotes the set of all non-negative differentiable functions $u : \mathbb{R}_+ \to \mathbb{R}$ satisfying the following conditions:

(i) *u* is strictly increasing on \mathbb{R}_+ and there exists $x_0 \in \mathbb{R}_+$ such that $u(x_0) = 1$,

(ii) $\lim_{x\to\infty} u(x) = \infty$, $\frac{u'(x)}{|u(x)|^r}$ is decreasing on \mathbb{R}_+ .

Theorem 1.14 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let s > 0. Further, suppose that $A_1 \in \left(\max\{\frac{1-s}{q},0\},\frac{1}{q}\right), A_2 \in \left(\max\{\frac{1-s}{p},0\},\frac{1}{p}\right), u \in H(qA_1) \text{ and } v \in H(pA_2).$ If $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ is a non-negative homogeneous function of degree -s, strictly decreasing

in each argument, then the inequalities

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} K(u(m), v(n)) a_m b_n$$

$$\leq L \left[\sum_{m=1}^{\infty} [u(m)]^{1-s+p(A_1-A_2)} [u'(m)]^{1-p} a_m^p \right]^{\frac{1}{p}}$$

$$\times \left[\sum_{n=1}^{\infty} [v(n)]^{1-s+q(A_2-A_1)} [v'(n)]^{1-q} b_n^q \right]^{\frac{1}{q}}$$
(1.42)

and

$$\sum_{n=1}^{\infty} [v(n)]^{(s-1)(p-1)+p(A_1-A_2)}v'(n) \left[\sum_{m=1}^{\infty} K(u(m),v(n))a_m\right]^p$$

$$\leq L^p \sum_{m=1}^{\infty} [u(m)]^{(1-s)+p(A_1-A_2)}[u'(m)]^{1-p}a_m^p$$
(1.43)

hold for all non-negative sequences $(a_m)_{m \in \mathbb{N}}$, $(b_n)_{n \in \mathbb{N}}$, where

$$L = k^{\frac{1}{p}} (pA_2) k^{\frac{1}{q}} (2 - s - qA_1).$$
(1.44)

Moreover, inequalities (1.42) *and* (1.43) *are equivalent.*

Proof. Rewrite inequality (1.12) for the counting measure on \mathbb{N} , $(\varphi \circ u)(m) = [u(m)]^{A_1}$ $[u'(m)]^{-\frac{1}{q}}$, $(\psi \circ v)(n) = [v(n)]^{A_2}[v'(n)]^{-\frac{1}{p}}$, and the sequences $(a_m)_{m \in \mathbb{N}}$ and $(b_n)_{n \in \mathbb{N}}$. Clearly, these substitutions are well-defined, since u and v are injective functions. Thus, in this setting we have

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} K(u(m), v(n)) a_m b_n$$

$$\leq \left[\sum_{m=1}^{\infty} [u(m)]^{pA_1} [u'(m)]^{1-p} (F \circ u)(m) a_m^p \right]^{\frac{1}{p}}$$

$$\times \left[\sum_{n=1}^{\infty} [v(n)]^{qA_2} [v'(n)]^{1-q} (G \circ v)(n) b_n^q \right]^{\frac{1}{q}}, \qquad (1.45)$$

where

$$(F\circ u)(m) = \sum_{n=1}^{\infty} \frac{K(u(m),v(n))v'(n)}{[v(n)]^{pA_2}}$$

and

$$(G \circ v)(n) = \sum_{m=1}^{\infty} \frac{K(u(m), v(n))u'(m)}{[u(m)]^{qA_1}}.$$

Now, since the kernel K is strictly decreasing in each argument and $u \in H(qA_1)$, $v \in H(pA_2)$, it follows that $F \circ u$ and $G \circ v$ are strictly decreasing. Hence, we have

$$(F \circ u)(m) < \int_0^\infty \frac{K(u(m), v(y))}{[v(y)]^{pA_2}} v'(y) dy,$$
(1.46)

since the left-hand side of this inequality is obviously the lower Darboux sum for the integral on the right-hand side of inequality. Further, utilizing substitution v(y) = tu(m) and homogeneity of the kernel *K*, we have

$$\int_0^\infty \frac{K(u(m), v(y))}{[v(y)]^{pA_2}} v'(y) dy = [u(m)]^{1-s-pA_2} \int_0^\infty K(1, t) t^{-pA_2} dt,$$

so by virtue of (1.20) and (1.46) we get

$$(F \circ u)(m) < [u(m)]^{1-s-pA_2}k(pA_2).$$
(1.47)

By the similar arguments as for the function $F \circ u$, we obtain

$$(G \circ v)(m) < \int_{0}^{\infty} \frac{K(u(x), v(n))}{[u(x)]^{qA_{1}}} u'(x) dx$$

= $[v(n)]^{1-s-qA_{1}} \int_{0}^{\infty} K(t, 1)t^{-qA_{1}} dt$
= $[v(n)]^{1-s-qA_{1}} \int_{0}^{\infty} K(1, t)t^{-2+s+qA_{1}} dt$
= $[v(n)]^{1-s-qA_{1}} k(2-s-qA_{1}).$ (1.48)

Finally, relations (1.45), (1.47), and (1.48) imply inequality (1.42).

On the other hand, if we rewrite inequality (1.13) with the counting measure on \mathbb{N} and the same functions as in the proof of inequality (1.42), after using estimates (1.47) and (1.48), we also obtain (1.43).

Clearly, Theorem 1.14 covers discrete Hilbert and Hardy-Hilbert-type inequalities with homogeneous kernels, decreasing in both arguments.

Remark 1.10 Suppose $(a_m)_{m \in \mathbb{N}}$ and $(b_n)_{n \in \mathbb{N}}$ are non-negative real sequences, not identically equal to trivial zero sequence. Then, according to estimates (1.47) and (1.48), it follows that inequalities (1.42) and (1.43) are sharp. In other words, equalities in (1.42) and (1.43) hold if and only if $a_m \equiv 0$ or $b_n \equiv 0$.

Remark 1.11 If the homogeneous kernel $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ is a symmetric function, that is, K(x,y) = K(y,x), for all $x, y \in \mathbb{R}_+$, then the constant *L*, defined by (1.44), simplifies to $L = k^{\frac{1}{p}} (pA_2)k^{\frac{1}{q}} (qA_1)$.

As emphasized above, inequalities (1.42) and (1.43) are sharp if the sequences $(a_m)_{m \in \mathbb{N}}$ and $(b_n)_{n \in \mathbb{N}}$ are not identically equal to the zero sequence. Therefore, it is interesting to consider the problem of finding the best possible constant factors for inequalities (1.42) and (1.43).

The main idea in obtaining the best possible constant factor is a reduction of constant defined by (1.44) to the form without exponents, which was already considered in the integral case. Thus, the parameters A_1 and A_2 fulfill (1.29), that is, $pA_2 + qA_1 = 2 - s$, which implies that $k(pA_2) = k(2 - s - qA_1)$. In such a way, the constant factor *L* from Theorem 1.14 becomes

$$L^* = k(pA_2). (1.49)$$

Moreover, under assumption (1.29), inequalities (1.42) and (1.43) respectively read

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} K(u(m), v(n)) a_m b_n \le L^* \left[\sum_{m=1}^{\infty} [u(m)]^{-1 + pqA_1} [u'(m)]^{1 - p} a_m^p \right]^{\frac{1}{p}} \times \left[\sum_{n=1}^{\infty} [v(n)]^{-1 + pqA_2} [v'(n)]^{1 - q} b_n^q \right]^{\frac{1}{q}} (1.50)$$

and

$$\sum_{n=1}^{\infty} [v(n)]^{(p-1)(1-pqA_2)} v'(n) \left[\sum_{m=1}^{\infty} K(u(m), v(n)) a_m \right]^p \\ \leq (L^*)^p \sum_{m=1}^{\infty} [u(m)]^{-1+pqA_1} [u'(m)]^{1-p} a_m^p.$$
(1.51)

The following theorem shows that the constants on the right-hand sides of inequalities (1.50) and (1.51) are the best possible.

Theorem 1.15 Suppose that parameters p, q, s, A_1 , A_2 , and the functions $u, v : \mathbb{R}_+ \to \mathbb{R}$, $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ are defined as in the statement of Theorem 1.14. If parameters A_1 and A_2 fulfill condition $pA_2 + qA_1 = 2 - s$, then the constant factors L^* and $(L^*)^p$ are the best possible in inequalities (1.50) and (1.51).

Proof. It is enough to show that L^* is the best possible constant factor in inequality (1.50), since (1.50) and (1.51) are equivalent. For this purpose, we consider sequences $\widetilde{a}_m = [u(m)]^{-qA_1 - \frac{\varepsilon}{p}} u'(m)$ and $\widetilde{b}_n = [v(n)]^{-pA_2 - \frac{\varepsilon}{q}} v'(n)$, where $\varepsilon > 0$ is sufficiently small number. Since $u \in H(qA_1)$ we may assume that u is strictly increasing in \mathbb{R}_+ , and there exists $x_0 \in \mathbb{R}_+$ such that $u(x_0) = 1$. Therefore, considering integral sums, we have

$$\begin{split} \frac{1}{\varepsilon} &= \int_1^\infty [u(x)]^{-1-\varepsilon} d[u(x)] < \sum_{m=1}^\infty [u(m)]^{-1-\varepsilon} u'(m) \\ &= \sum_{m=1}^\infty [u(m)]^{-1+pqA_1} [u'(m)]^{1-p} \widetilde{a}_m^p \\ &< \zeta(1) + \int_1^\infty [u(x)]^{-1-\varepsilon} d[u(x)] = \zeta(1) + \frac{1}{\varepsilon}, \end{split}$$

where the function ζ is defined by $\zeta(x) = [u(x)]^{-1-\varepsilon}u'(x)$. Hence, we have

$$\sum_{m=1}^{\infty} [u(m)]^{-1+pqA_1} [u'(m)]^{1-p} \widetilde{a}_m^p = \frac{1}{\varepsilon} + O(1),$$
(1.52)

and similarly,

$$\sum_{n=1}^{\infty} [v(n)]^{-1+pqA_2} [v'(n)]^{1-q} \widetilde{b}_n^q = \frac{1}{\varepsilon} + O(1).$$
(1.53)

Now, suppose that there exists a positive constant M, smaller than L^* , such that (1.50) holds after replacing L^* with M. Then, combining relations (1.52) and (1.53) with inequality (1.50), we get

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} K(u(m), v(n)) \widetilde{a}_m \widetilde{b}_n < \frac{1}{\varepsilon} (M + o(1)).$$
(1.54)

On the other hand, we can also estimate the left-hand side of inequality (1.50). Namely, inserting the above defined sequences $(\tilde{a}_m)_{m\in\mathbb{N}}$ and $(\tilde{b}_n)_{n\in\mathbb{N}}$ in the left-hand side of (1.50), we easily obtain

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} K(u(m), v(n)) \widetilde{a}_m \widetilde{b}_n$$

> $\int_1^{\infty} [u(x)]^{-qA_1 - \frac{\varepsilon}{p}} \left[\int_1^{\infty} K(u(x), v(y)) [v(y)]^{-pA_2 - \frac{\varepsilon}{q}} d(v(y)) \right] d(u(x))$
= $\int_1^{\infty} [u(x)]^{-1-\varepsilon} \left[\int_{\frac{1}{u(x)}}^{\infty} K(1, t) t^{-pA_2 - \frac{\varepsilon}{q}} dt \right] d(u(x)).$ (1.55)

Moreover, since the kernel K is strictly decreasing in both arguments, it follows that K(1,0) > K(1,t), for t > 0, so we have

$$\int_{\frac{1}{u(x)}}^{\infty} K(1,t)t^{-pA_2-\frac{\varepsilon}{q}}dt > \int_{0}^{\infty} K(1,t)t^{-pA_2-\frac{\varepsilon}{q}}dt - K(1,0)\int_{0}^{\frac{1}{u(x)}} t^{-pA_2-\frac{\varepsilon}{q}}dt$$
$$= k\left(pA_2+\frac{\varepsilon}{q}\right) - \frac{K(1,0)}{1-pA_2-\frac{\varepsilon}{q}}[u(x)]^{-1+pA_2+\frac{\varepsilon}{q}},$$

and consequently

$$\int_{1}^{\infty} [u(x)]^{-1-\varepsilon} \left[\int_{\frac{1}{u(x)}}^{\infty} K(1,t) t^{-pA_2 - \frac{\varepsilon}{q}} dt \right] d(u(x))$$

$$\geq \frac{1}{\varepsilon} k \left(pA_2 + \frac{\varepsilon}{q} \right) - \frac{K(1,0)}{\left(1 - pA_2 - \frac{\varepsilon}{q} \right) \left(1 - pA_2 + \frac{\varepsilon}{p} \right)}.$$
(1.56)

In other words, relations (1.55) and (1.56) yield the lower bound for the left-hand side of inequality (1.50):

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} K(u(m), v(n)) \widetilde{a}_m \widetilde{b}_n > \frac{1}{\varepsilon} (L^* + o(1)).$$
(1.57)

Finally, comparing relations (1.54) and (1.57), and letting $\varepsilon \to 0^+$, it follows that $L^* \leq M$, which contradicts the assumption that the constant *M* is smaller than L^* . This means that L^* is the best possible constant in inequality (1.50).

We conclude this discussion with a few remarks which connect Theorems 1.14 and 1.15 with particular results presented at the beginning of this subsection.

Remark 1.12 Observe that Theorem 1.14 is a generalization of inequality (1.37) (see also [52]). Moreover, Theorem 1.15 yields the form of inequality (1.37) with the best possible constant factor. Namely, putting the kernel $K(x,y) = (x+y)^{-s}$, s > 0, and power functions $u(x) = Ax^{\alpha}$ and $v(y) = By^{\beta}$, $A, B, \alpha, \beta > 0$, in (1.50), we obtain the corresponding form of inequality (1.37), with the best possible constant factor

$$\alpha^{-\frac{1}{q}}\beta^{-\frac{1}{p}}A^{-1+qA_1}B^{-1+pA_2}B(1-pA_2,1-qA_1)$$

Remark 1.13 If s = 1, then parameters $A_1 = A_2 = \frac{1}{pq}$ fulfill condition (1.29). Thus, putting these parameters in (1.50) and (1.51), together with kernel $K(x,y) = (x+y)^{-1}$ and differentiable functions $u(x) = v(x) = \log(x+1)$, we obtain inequalities (1.40) and (1.41) with the best possible constants (see also [152]).

Remark 1.14 Since parameters $A_1 = A_2 = \frac{2-s}{pq}$, where $2 - \min\{p,q\} < s < 2$, fulfill condition (1.29), they can be substituted in inequalities (1.50) and (1.51). In addition, considering the kernel $K(x,y) = (x+y)^{-s}$, s > 0, and differentiable functions u(x) = v(x) = x + v, $0 \le v < 1$, inequalities (1.50) and (1.51) reduce to

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \frac{a_m b_n}{(m+n+2\nu)^s} \le S_1 \left[\sum_{m=1}^{\infty} (m+\nu)^{1-s} a_m^p \right]^{\frac{1}{p}} \left[\sum_{n=1}^{\infty} (n+\nu)^{1-s} b_n^q \right]^{\frac{1}{q}}$$
(1.58)

and

$$\sum_{n=1}^{\infty} (n+\nu)^{(p-1)(s-1)} \left[\sum_{m=1}^{\infty} \frac{a_m}{(m+n+2\nu)^s} \right]^p \le S_1^p \sum_{m=1}^{\infty} (m+\nu)^{1-s} a_m^p, \tag{1.59}$$

where the constant factors $S_1 = B(\frac{1}{p} + \frac{s-1}{q}, \frac{1}{q} + \frac{s-1}{p})$ and S_1^p are the best possible. If s = 1, then S_1 becomes $\pi/\sin(\pi/p)$. Thus, setting $v = \frac{1}{2}$ and s = 1 in (1.58) and (1.59), we obtain the sharper version of the Hilbert double series theorem, as well as its equivalent form (see also [142]).

Remark 1.15 Some particular discrete Hilbert-type inequalities regarding homogeneous kernels are also obtained in [54]. They can be derived as the consequences of Theorems 1.14 and 1.15. For example, setting $K(x,y) = (x+y)^{-s}$, s > 0, $u(x) = xa^x$, $v(y) = ya^y$, a > 1, inequalities (1.50) and (1.51) respectively read

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \frac{a_m b_n}{(ma^m + na^n)^s} \le L^* \left[\sum_{m=1}^{\infty} (ma^m)^{-1 + pqA_1} (a^m + ma^m \log a)^{1-p} a_m^p \right]^{\frac{1}{p}} \\ \times \left[\sum_{n=1}^{\infty} (na^n)^{-1 + pqA_2} (a^n + na^n \log a)^{1-p} b_n^q \right]^{\frac{1}{q}}$$
and

$$\sum_{n=1}^{\infty} (na^n)^{(p-1)(1-pqA_2)} (a^n + na^n \log a) \left[\sum_{m=1}^{\infty} \frac{a_m}{(ma^m + na^n)^s} \right]^p$$

$$\leq (L^*)^p \sum_{m=1}^{\infty} (ma^m)^{-1+pqA_1} (a^m + ma^m \log a)^{1-p} a_m^p,$$

where $L^* = B(1 - pA_2, 1 - qA_1)$.

Similarly, if $K(x,y) = (x+y)^{-s}$, s > 0, $u(x) = x \arctan x$, $v(y) = y \arctan y$, then inequalities (1.50) and (1.51) become

$$\sum_{m=1}^{\infty}\sum_{n=1}^{\infty}\frac{a_mb_n}{(m\arctan m+n\arctan n)^s} \le L^*\left[\sum_{m=1}^{\infty}\omega_p(m)a_m^p\right]^{\frac{1}{p}}\left[\sum_{n=1}^{\infty}\omega_q(n)b_n^q\right]^{\frac{1}{q}}$$

and

$$\sum_{n=1}^{\infty} (n \arctan n)^{(p-1)(1-pqA_2)} \left(\arctan n + \frac{n}{1+n^2}\right) \\ \times \left[\sum_{m=1}^{\infty} \frac{a_m}{(m \arctan m + n \arctan n)^s}\right]^p \le (L^*)^p \sum_{m=1}^{\infty} \omega_p(m) a_m^p,$$

where $L^* = B(1 - pA_2, 1 - qA_1)$ and

$$\omega_r(x) = (x \arctan x)^{1-s-r(A_2-A_1)} \left(\arctan x + \frac{x}{1+x^2}\right)^{1-r}, \ r \in \{p,q\}.$$

Of course, the above inequalities include the best possible constants. For some other examples arising from various choices of weight functions, the reader is referred to [54].

1.4.3 Some further estimates

In this subsection we study a few particular Hilbert-type inequalities involving the homogeneous kernel $K(x,y) = (x+y)^{-s}$, s > 0. In addition to the Hilbert inequality, the following results will be derived with a help of some additional estimates that arise from this particular setting. More precisely, we shall use Theorems 1.9 and 1.10, as well as various methods for estimating the integral of type

$$\int_{\frac{a}{x}}^{\frac{b}{x}} K(1,u) u^{-\alpha} du.$$

The first in the series of results is the following pair of inequalities.

Corollary 1.2 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let s > 0. If $(a, b) \subseteq \mathbb{R}_+$, then the inequalities

$$\int_{a}^{b} \int_{a}^{b} \frac{f(x)g(y)}{(x+y)^{s}} dx dy \\
\leq B\left(\frac{s}{2}, \frac{s}{2}\right) \left\{ \int_{a}^{b} \left[1 - \frac{1}{2} \left(\frac{a}{x}\right)^{\frac{s}{2}} - \frac{1}{2} \left(\frac{x}{b}\right)^{\frac{s}{2}} \right] x^{-\frac{sp}{2} + p - 1} f^{p}(x) dx \right\}^{\frac{1}{p}} \\
\times \left\{ \int_{a}^{b} \left[1 - \frac{1}{2} \left(\frac{a}{y}\right)^{\frac{s}{2}} - \frac{1}{2} \left(\frac{y}{b}\right)^{\frac{s}{2}} \right] y^{-\frac{sq}{2} + q - 1} g^{q}(y) dy \right\}^{\frac{1}{q}} \tag{1.60}$$

and

$$\int_{a}^{b} \left[1 - \frac{1}{2} \left(\frac{a}{y} \right)^{\frac{s}{2}} - \frac{1}{2} \left(\frac{y}{b} \right)^{\frac{s}{2}} \right]^{1-p} y^{\frac{sp}{2}-1} \left[\int_{a}^{b} \frac{f(x)}{(x+y)^{s}} dx \right]^{p} dy$$

$$\leq B^{p} \left(\frac{s}{2}, \frac{s}{2} \right) \int_{a}^{b} \left[1 - \frac{1}{2} \left(\frac{a}{x} \right)^{\frac{s}{2}} - \frac{1}{2} \left(\frac{x}{b} \right)^{\frac{s}{2}} \right] x^{-\frac{sp}{2}+p-1} f^{p}(x) dx \qquad (1.61)$$

hold for all non-negative measurable functions $f, g : (a, b) \to \mathbb{R}$. In addition, inequalities (1.60) and (1.61) are equivalent.

Proof. Considering Theorem 1.9 with the kernel $K(x,y) = (x+y)^{-s}$ and weight functions $\varphi(x) = x^{\frac{2-s}{2q}}, \ \psi(y) = y^{\frac{2-s}{2p}}$, we have

$$\begin{split} &\int_{a}^{b} \int_{a}^{b} \frac{f(x)g(y)}{(x+y)^{s}} dx dy \\ &\leq \left[\int_{a}^{b} x^{-\frac{sp}{2}+p-1} \left(\int_{\frac{a}{x}}^{\frac{b}{x}} \frac{u^{\frac{s}{2}-1}}{(1+u)^{s}} du \right) f^{p}(x) dx \right]^{\frac{1}{p}} \\ &\times \left[\int_{a}^{b} y^{-\frac{sq}{2}+q-1} \left(\int_{\frac{y}{b}}^{\frac{y}{a}} \frac{u^{\frac{s}{2}-1}}{(1+u)^{s}} du \right) g^{q}(y) dy \right]^{\frac{1}{q}}. \end{split}$$

Now, we are going to estimate integrals in the above inequality, dependent on variables x and y. Taking into account an obvious relation

$$B\left(\frac{s}{2},\frac{s}{2}\right) = 2\int_{1}^{\infty} \frac{u^{\frac{s}{2}-1}}{(1+u)^{s}} du = 2\alpha^{\frac{s}{2}}\int_{\alpha}^{\infty} \frac{u^{\frac{s}{2}-1}}{(\alpha+u)^{s}} du,$$

and inequality

$$\int_{\alpha}^{\infty} \frac{u^{\frac{s}{2}-1}}{(\alpha+u)^s} du < \int_{\alpha}^{\infty} \frac{u^{\frac{s}{2}-1}}{(1+u)^s} du,$$

where $\alpha > 1$, we have

$$\int_{\alpha}^{\infty} \frac{u^{\frac{s}{2}-1}}{(1+u)^{s}} du > \frac{1}{2} \alpha^{-\frac{s}{2}} B\left(\frac{s}{2}, \frac{s}{2}\right), \qquad \alpha > 1.$$
(1.62)

Finally, considering the relation

$$\int_{\frac{a}{x}}^{\frac{b}{x}} \frac{u^{\frac{s}{2}-1}}{(1+u)^{s}} du = B\left(\frac{s}{2}, \frac{s}{2}\right) - \int_{\frac{b}{x}}^{\infty} \frac{u^{\frac{s}{2}-1}}{(1+u)^{s}} du - \int_{\frac{x}{a}}^{\infty} \frac{u^{\frac{s}{2}-1}}{(1+u)^{s}} du,$$

and (1.62), we obtain the estimate

$$\int_{\frac{a}{x}}^{\frac{b}{x}} \frac{u^{\frac{s}{2}-1}}{(1+u)^{s}} du < B\left(\frac{s}{2}, \frac{s}{2}\right) \left[1 - \frac{1}{2}\left(\frac{a}{x}\right)^{\frac{s}{2}} - \frac{1}{2}\left(\frac{x}{b}\right)^{\frac{s}{2}}\right],$$

which yields inequality (1.60). Equivalent form (1.61) follows in a similar way.

Remark 1.16 Combining the well-known arithmetic-geometric mean inequality

$$\frac{1}{2}\left(\frac{a}{x}\right)^{\frac{s}{2}} + \frac{1}{2}\left(\frac{x}{b}\right)^{\frac{s}{2}} \ge \left(\frac{a}{b}\right)^{\frac{s}{4}},$$

with inequalities (1.60) and (1.61), we have

$$\int_{a}^{b} \int_{a}^{b} \frac{f(x)g(y)}{(x+y)^{s}} dx dy$$

$$\leq B\left(\frac{s}{2}, \frac{s}{2}\right) \left[1 - \left(\frac{a}{b}\right)^{\frac{s}{4}}\right] \left[\int_{a}^{b} x^{-\frac{sp}{2} + p - 1} f(x)^{p} dx\right]^{\frac{1}{p}} \left[\int_{a}^{b} y^{-\frac{sq}{2} + q - 1} g(y)^{q} dy\right]^{\frac{1}{q}}$$

and

$$\int_{a}^{b} y^{\frac{sp}{2}-1} \left[\int_{a}^{b} \frac{f(x)}{(x+y)^{s}} dx \right]^{p} dy \leq \left[B\left(\frac{s}{2}, \frac{s}{2}\right) \left(1 - \left(\frac{a}{b}\right)^{\frac{s}{4}} \right) \right]^{p} \int_{a}^{b} x^{-\frac{sp}{2}+p-1} f(x)^{p} dx.$$

Putting p = q = 2 in these inequalities, we obtain a pair of inequalities derived in [152]. Moreover, if a = 0 and $b = \infty$, the above inequalities reduce to corresponding relations obtained in [11].

We finish this section with another specific Hilbert-type inequality referring to kernel $K(x,y) = (x+y)^{-s}$, s > 0.

Corollary 1.3 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, s > 0, and let A_1 and A_2 be real parameters such that $A_1 \in (\frac{1-s}{q}, \frac{1}{q})$ and $A_2 \in (\frac{1-s}{p}, \frac{1}{p})$. If $Q = k_{l_1}^{\frac{1}{p}}(pA_2)k_{l_2}^{\frac{1}{q}}(qA_1)$, $l_1 = \frac{1-pA_2}{s}$, $l_2 = \frac{1-qA_1}{s}$, and $k_l(\alpha) = \int_{\frac{a^l-b^l}{b(b^{l-1}-a^{l-1})}}^{\frac{a^l-b^l}{q}} \frac{u^{-\alpha}}{(1+u)^s} du$,

then the inequalities

$$\int_{a}^{b} \int_{a}^{b} \frac{f(x)g(y)}{(x+y)^{s}} dx dy$$

$$\leq Q \left[\int_{a}^{b} x^{1-s+p(A_{1}-A_{2})} f^{p}(x) dx \right]^{\frac{1}{p}} \left[\int_{a}^{b} y^{1-s+q(A_{2}-A_{1})} g^{q}(y) dy \right]^{\frac{1}{q}}$$
(1.63)

and

$$\int_{a}^{b} y^{(p-1)(s-1)+p(A_{1}-A_{2})} \left[\int_{a}^{b} \frac{f(x)}{(x+y)^{s}} dx \right]^{p} dy \le Q^{p} \int_{a}^{b} x^{1-s+p(A_{1}-A_{2})} f^{p}(x) dx \quad (1.64)$$

hold for all non-negative measurable functions $f, g: (a, b) \to \mathbb{R}$ and are equivalent.

Proof. We start as in the proof of Corollary 1.2, but for the estimate of the integral

$$\int_{\frac{a}{x}}^{\frac{b}{x}} \frac{u^{-\alpha}}{(1+u)^s} du,$$

we use the fact that the function $f(x) = \int_{a/x}^{b/x} u^{-\alpha} (1+u)^{-s} du, x \in \mathbb{R}_+$, attains its maximum value at $x = \frac{ab^l - ba^l}{a^l - b^l}, l = \frac{1-\alpha}{s}$.

Remark 1.17 Setting $A_1 = A_2 = \frac{2-s}{pq}$, provided that $s > 2 - \min\{p, q\}$, inequalities (1.63) and (1.64) reduce respectively to

$$\int_{a}^{b} \int_{a}^{b} \frac{f(x)g(y)}{(x+y)^{s}} dx dy \le Q_{1} \left[\int_{a}^{b} x^{1-s} f^{p}(x) dx \right]^{\frac{1}{p}} \left[\int_{a}^{b} y^{1-s} g^{q}(y) dy \right]^{\frac{1}{q}}$$

and

$$\int_{a}^{b} y^{(s-1)(p-1)} \left[\int_{a}^{b} \frac{f(x)}{(x+y)^{s}} dx \right]^{p} dy \le Q_{1}^{p} \int_{a}^{b} x^{1-s} f^{p}(x) dx,$$

where

$$Q_{1} = k_{\frac{q}{qs}}^{\frac{1}{p}} \left(\frac{2-s}{q}\right) k_{\frac{p+s-2}{ps}}^{\frac{1}{q}} \left(\frac{2-s}{p}\right).$$

Similarly, if $A_1 = \frac{2-s}{2q}$ and $A_2 = \frac{2-s}{2p}$, inequalities (1.63) and (1.64) respectively read

$$\int_{a}^{b} \int_{a}^{b} \frac{f(x)g(y)}{(x+y)^{s}} dx dy$$

$$\leq k_{\frac{1}{2}} \left(\frac{2-s}{2}\right) \left[\int_{a}^{b} x^{-\frac{sp}{2}+p-1} f(x)^{p} dx \right]^{\frac{1}{p}} \left[\int_{a}^{b} y^{-\frac{sq}{2}+q-1} g(y)^{q} dy \right]^{\frac{1}{q}}$$

and

$$\int_{a}^{b} y^{\frac{sp}{2}-1} \left[\int_{a}^{b} \frac{f(x)}{(x+y)^{s}} dx \right]^{p} dy \le k_{\frac{1}{2}}^{p} \left(\frac{2-s}{2} \right) \int_{a}^{b} x^{-\frac{sp}{2}+p-1} f(x)^{p} dx.$$

Remark 1.18 General results from this section, covering the best possible constant factors for Hilbert-type inequalities with a homogeneous kernel in integral and discrete case are established in papers [63] and [111]. Particular inequalities in subsection 1.4.3 are taken from paper [53]. For the similar problem area, the reader is referred to the following references: [13], [36], [38], [80], [85], [101], [122], [126], [132], [135], [146], [149], [155], [162], [164], [176], [181], [182], and [183].

1.5 Refined Hilbert-type inequalities via the refined Hölder inequality

In this section we provide an interesting improvement of the general Hilbert-type inequality based on the improvement of the Hölder inequality, obtained by H. Leping et. al. in [75].

For the reader's convenience, we first introduce some definitions. Let f and g be elements of the inner product space of measurable functions, where the inner product is denoted by $\langle f, g \rangle$. Further, let $S_r(f, u)$ be defined as

$$S_r(f,u) = \langle f^{\frac{r}{2}}, u \rangle ||f||_r^{-\frac{r}{2}}$$

where *u* is the unit vector and $||f||_r = \sqrt[r]{\langle f^{\frac{r}{2}}, f^{\frac{r}{2}} \rangle}$. Clearly, $S_r(f, u) = 0$ when the vector *u* selected is orthogonal to $f^{\frac{r}{2}}$.

By virtue of the positive definiteness of the Gramm matrix, G. Mingzhe and L. Debnath [98], derived an important inequality

$$\langle f,g \rangle^2 \le ||f||^2 ||g||^2 - (||f||x - ||g||y)^2 = ||f||^2 ||g||^2 [1 - r(h)],$$
 (1.65)

where $r(h) = \left(\frac{y}{||f||} - \frac{x}{||g||}\right)^2$, $x = \langle g, h \rangle$, $y = \langle f, h \rangle$, ||h|| = 1, and $xy \ge 0$. Here, $|| \cdot ||$ denotes the usual 2-norm in L^2 space. In addition, equality in (1.65) holds if and only if vectors f and g are linearly dependent or vector h is a linear combination of f and g, provided $xy = 0, x \ne y$. It should be noticed here that inequality (1.65) is a consequence of an earlier result of Mitrović (see paper [104]).

Now, regarding the previous definitions, the above mentioned refinement of the Hölder inequality from [75] asserts that

$$\langle f,g \rangle \le ||f||_p ||g||_q [1-R(h)]^m,$$
(1.66)

where $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, $R(h) = [S_p(f,h) - S_q(g,h)]^2 \neq 0$, ||h|| = 1, $m = \min\{\frac{1}{p}, \frac{1}{q}\}$, provided $f^{\frac{p}{2}}, g^{\frac{q}{2}}$, *h* are linearly independent.

In order to explain the idea from paper [75], we derive here improvement of the reverse Hölder inequality. Moreover, the integrals will be taken with σ -finite measures, as in Section 1.2, and the corresponding inner product will be defined as $\langle f,g \rangle = \int_{\Omega} K(x) f(x) g(x) d\mu(x).$

Lemma 1.2 Let $\frac{1}{p} + \frac{1}{q} = 1$, $0 , and let <math>K : \Omega \to \mathbb{R}$, $f : \Omega \to \mathbb{R}$, $g : \Omega \to \mathbb{R}$, $h : \Omega \to \mathbb{R}$ be non-negative measurable functions such that $f^{\frac{p}{2}}$, $g^{\frac{q}{2}}$ and h are linearly independent. Then,

$$\langle f,g \rangle \ge ||f||_p ||g||_q [1-R(h)]^{\frac{1}{q}},$$
(1.67)

where $R(h) = [S_p(f,h) - S_q(g,h)]^2 \neq 0$ and ||h|| = 1.

Proof. The inner product $\langle f, g \rangle$ can be rewritten in the following form:

$$\langle f,g\rangle = \int_{\Omega} K(x) \left(f^{\frac{p}{q}}(x)g(x) \right) f^{1-\frac{p}{q}}(x) d\mu(x).$$

Now, since parameters $A = \frac{q}{2}$ and $B = \frac{q}{q-2}$ are conjugate, that is, $\frac{1}{A} + \frac{1}{B} = 1$, applying the reverse Hölder inequality to the above expression, we have

$$\langle f,g \rangle \geq \left[\int_{\Omega} K(x) \left(f^{\frac{p}{q}}(x)g(x) \right)^{A} d\mu(x) \right]^{\frac{1}{A}} \left[\int_{\Omega} K(x) \left(f^{1-\frac{p}{q}}(x) \right)^{B} d\mu(x) \right]^{\frac{1}{B}}$$

$$= \langle f^{\frac{p}{2}}, g^{\frac{q}{2}} \rangle ||f||_{p}^{p(1-\frac{2}{q})}.$$

$$(1.68)$$

On the other hand, replacing f and g with $f^{\frac{p}{2}}$ and $g^{\frac{q}{2}}$ in (1.65), we obtain

$$\langle f^{\frac{p}{2}}, g^{\frac{q}{2}} \rangle^2 \le ||f||_p{}^p||g||_q{}^q[1-R(h)],$$
 (1.69)

that is, (1.67).

In the sequel we provide an extension of Theorem 1.9 via the above described improvement of the Hölder inequality. In the following two theorems, the exponent *m* indicates $m = \min\{\frac{1}{p}, \frac{1}{q}\}$, where *p* and *q* are conjugate exponents. Moreover, regarding the above definitions, we denote $R(\overline{f}, \overline{g}, \overline{h}) = (S_p(\overline{f}, \overline{h}) - S_q(\overline{g}, \overline{h}))^2$, where $S_p(\overline{f}, \overline{h}) = \langle \overline{f}^{\frac{p}{2}}, \overline{h} \rangle ||\overline{f}||_p^{-\frac{p}{2}}$. Obviously, $S_p(\overline{f}, \overline{h})$ depends on the inner product. In order to obtain improved Hilbert-type inequalities, the inner product will be defined as

$$\langle \overline{f}, \overline{g} \rangle = \int_{\Omega} \int_{\Omega} K(x, y) \overline{f}(x, y) \overline{g}(x, y) d\mu_1(x) d\mu_2(y).$$
(1.70)

Theorem 1.16 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let Ω be a measure space with positive σ -finite measures μ_1 and μ_2 . Let $K : \Omega \times \Omega \to \mathbb{R}$ and $\varphi, \psi : \Omega \to \mathbb{R}$ be non-negative measurable functions. If the functions F and G are defined by $F(x) = \int_{\Omega} K(x,y)\psi^{-p}(y)d\mu_2(y)$ and $G(y) = \int_{\Omega} K(x,y)\varphi^{-q}(x)d\mu_1(x)$, then for all non-negative measurable functions f and g on Ω the inequality

$$\int_{\Omega} \int_{\Omega} K(x,y) f(x) g(y) d\mu_{1}(x) d\mu_{2}(y) \\
\leq (1 - R(\overline{f}, \overline{g}, \overline{h}))^{m} \left[\int_{\Omega} \varphi^{p}(x) F(x) f^{p}(x) d\mu_{1}(x) \right]^{\frac{1}{p}} \\
\times \left[\int_{\Omega} \psi^{q}(y) G(y) g^{q}(y) d\mu_{2}(y) \right]^{\frac{1}{q}}$$
(1.71)

holds, where $\overline{f}(x,y) = f(x) \frac{\varphi(x)}{\psi(y)}$, $\overline{g}(x,y) = g(y) \frac{\psi(y)}{\varphi(x)}$, and $\overline{h}(x,y)$ is such that

$$\int_{\Omega} \int_{\Omega} K(x, y) \overline{h}^2(x, y) d\mu_1(x) d\mu_2(y) = 1.$$

If 0 , then the reverse inequality in (1.71) holds.

Proof. Inequality (1.71) is an immediate consequence of the relation

$$\int_{\Omega} \int_{\Omega} K(x,y) f(x) g(y) d\mu_1(x) d\mu_2(y) = \int_{\Omega} \int_{\Omega} K(x,y) \overline{f}(x,y) \overline{g}(x,y) d\mu_1(x) d\mu_2(y)$$

and inequality (1.66). On the other hand, the reverse inequality is a consequence of Lemma 1.2 and the above relation. $\hfill \Box$

In addition, replacing g in (1.71) with the function

$$\widetilde{g}(y) = G^{1-p}(y)\psi^{-p}(y)\left(\int_{\Omega} K(x,y)f(x)d\mu_1(x)\right)^{p-1},$$

we also obtain an improvement of the Hardy-Hilbert-type inequality (1.13).

Theorem 1.17 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let functions K, φ , ψ , F, G be defined as in the statement of Theorem 1.16. Then the inequality

$$\int_{\Omega} G^{1-p}(y) \psi^{-p}(y) \left[\int_{\Omega} K(x,y) f(x) d\mu_1(x) \right]^p d\mu_2(y) \\ \leq (1 - R(\overline{f}, \overline{\widetilde{g}}, \overline{h}))^{mp} \int_{\Omega} \varphi^p(x) F(x) f^p(x) d\mu_1(x)$$
(1.72)

holds for all non-negative measurable functions $f: \Omega \to \mathbb{R}$, provided the functions $\overline{f}, \overline{\tilde{g}}, \overline{h}$ are defined as in Theorem 1.16. If 0 , then the reverse inequality in (1.72) is valid.

Proof. Since $\frac{1}{p} + \frac{1}{q} = 1$, utilizing (1.71) we have

$$\begin{split} &\int_{\Omega} G^{1-p}(y) \psi^{-p}(y) \bigg(\int_{\Omega} K(x,y) f(x) d\mu_1(x) \bigg)^p d\mu_2(y) \\ &= \int_{\Omega} \int_{\Omega} K(x,y) f(x) \widetilde{g}(y) d\mu_1(x) d\mu_2(y) \\ &\leq (1 - R(\overline{f}, \overline{\widetilde{g}}, \overline{h}))^m \left[\int_{\Omega} \varphi^p(x) F(x) f^p(x) d\mu_1(x) \right]^{\frac{1}{p}} \left[\int_{\Omega} \psi^q(y) G(y) \widetilde{g}^q(y) d\mu_2(y) \right]^{\frac{1}{q}} \\ &= \left[\int_{\Omega} \varphi^p(x) F(x) f^p(x) d\mu_1(x) \right]^{\frac{1}{p}} \\ &\times \left[\int_{\Omega} G^{1-p}(y) \psi^{-p}(y) \bigg(\int_{\Omega} K(x,y) f(x) d\mu_1(x) \bigg)^p d\mu_2(y) \right]^{\frac{1}{q}}, \end{split}$$

that is, we get (1.72). The reverse inequality is derived in the same way, by virtue of Lemma 1.2. $\hfill \Box$

Remark 1.19 Note that inequalities (1.71) and (1.72) present refinements of inequalities (1.12) and (1.13) from Theorem 1.9.

Remark 1.20 Clearly, the method developed in this section can be applied to results from Sections 1.3 and 1.4. For more details about this problem area the reader is referred to [58]. Some other methods of improving Hilbert-type inequalities will be studied in Chapters 4, 5, and 6.

1.6 Multidimensional Hilbert-type inequalities

1.6.1 General form

Regarding Theorem 1.9, in this section we derive multidimensional forms of the general Hilbert-type and Hardy-Hilbert-type inequality. In such setting we deal with conjugate parameters $p_1, p_2, \ldots, p_n, n \ge 2$. Recall that these parameters fulfill condition $\sum_{i=1}^{n} \frac{1}{p_i} = 1$. If all parameters are non-negative, then $p_i > 1, i = 1, 2, \ldots, n$.

The following result includes integrals taken over general subsets of \mathbb{R}_+ , with respect to σ -finite measures.

Theorem 1.18 Let $\sum_{i=1}^{n} \frac{1}{p_i} = 1$, $p_i > 1$, i = 1, 2, ..., n, and let Ω be measure space with σ -finite measures μ_i , i = 1, 2, ..., n. Further, suppose that $K : \Omega^n \to \mathbb{R}$ and $\phi_{ij} : \Omega \to \mathbb{R}$, i, j = 1, ..., n, are non-negative measurable functions such that $\prod_{i,j=1}^{n} \phi_{ij}(x_j) = 1$. If $\frac{1}{q} = \sum_{i=1}^{n-1} \frac{1}{p_i}$ and functions F_i , h are defined by

$$F_{i}(x_{i}) = \int_{\Omega^{n-1}} K(x_{1}, \dots, x_{n}) \prod_{j=1, j \neq i}^{n} \phi_{ij}^{p_{i}}(x_{j}) \\ \times d\mu_{1}(x_{1}) \dots d\mu_{i-1}(x_{i-1}) d\mu_{i+1}(x_{i+1}) \dots d\mu_{n}(x_{n}), \ i = 1, 2, \dots, n,$$

 $h = \phi_{nn}^{-q} F_n^{1-q}$, then for all non-negative measurable functions $f_i : \Omega \to \mathbb{R}$, i = 1, 2, ..., n, the inequalities

$$\int_{\Omega^{n}} K(x_{1}, \dots, x_{n}) \prod_{i=1}^{n} f_{i}(x_{i}) d\mu_{1}(x_{1}) \cdots d\mu_{n}(x_{n})$$

$$\leq \prod_{i=1}^{n} \left[\int_{\Omega} F_{i}(x_{i}) f_{i}^{p_{i}}(x_{i}) \phi_{ii}^{p_{i}}(x_{i}) d\mu_{i}(x_{i}) \right]^{\frac{1}{p_{i}}}$$
(1.73)

and

$$\int_{\Omega} h(x_n) \left[\int_{\Omega^{n-1}} K(x_1, \dots, x_n) \prod_{i=1}^{n-1} f_i(x_i) d\mu_1(x_1) \cdots d\mu_{n-1}(x_{n-1}) \right]^q d\mu_n(x_n) \\ \leq \prod_{i=1}^{n-1} \left[\int_{\Omega} F_i(x_i) f_i^{p_i}(x_i) \phi_{ii}^{p_i}(x_i) d\mu_i(x_i) \right]^{\frac{q}{p_i}}$$
(1.74)

hold and are equivalent.

If $p_i > 0$, $i \in \{1, 2, ..., n\}$, and $p_k < 0$, $k \neq i$, then the reverse inequality in (1.73) holds. Moreover, if $p_i > 0$, $i \in \{1, 2, ..., n - 1\}$, and $p_k < 0$, $k \neq i$, then the reverse inequality in (1.74) holds. In addition, inequality (1.74) holds also when $p_n > 0$ and $p_k < 0$, $k \neq n$.

Proof. We first prove inequality (1.73). Applying the Hölder inequality we have

$$\begin{split} &\int_{\Omega^n} K(x_1,...,x_n) \prod_{i=1}^n f_i(x_i) d\mu_1(x_1)...d\mu_n(x_n) \\ &= \int_{\Omega^n} K(x_1,...,x_n) \prod_{i=1}^n \left(f_i(x_i) \prod_{j=1}^n \phi_{ij}(x_j) \right) d\mu_1(x_1)...d\mu_n(x_n) \\ &\leq \prod_{i=1}^n \left[\int_{\Omega^n} K(x_1,...,x_n) f_i^{p_i}(x_i) \prod_{j=1}^n \phi_{ij}^{p_j}(x_j) d\mu_1(x_1)...d\mu_n(x_n) \right]^{\frac{1}{p_i}}, \end{split}$$

so the Fubini theorem and definitions of functions F_i , i = 1, 2, ..., n, yield (1.73). In order to prove inequality (1.74), we define $I(x_n)$ as

$$I(x_n) = \int_{\Omega^{n-1}} K(x_1, ..., x_n) \prod_{i=1}^{n-1} f_i(x_i) d\mu_1(x_1) ... d\mu_{n-1}(x_{n-1}).$$

Now, putting function

$$f_n(x_n) = h(x_n) \cdot (I(x_n))^{q-1}$$

in inequality (1.73), we get

$$\begin{split} I &= \int_{\Omega} h(x_n) (I(x_n))^q d\mu_n(x_n) \leq \prod_{i=1}^{n-1} \left[\int_{\Omega} \phi_{ii}^{p_i}(x_i) F_i(x_i) f_i^{p_i}(x_i) d\mu_i(x_i) \right]^{\frac{1}{p_i}} \\ & \times \left[\int_{\Omega} h^{p_n}(x_n) (I(x_n))^{p_n(q-1)} F_n(x_n) \phi_{nn}^{p_n}(x_n) d\mu_n(x_n) \right]^{1-\frac{1}{q}} \end{split}$$

Since $h(x_n) = \phi_{nn}^{-q}(x_n) F_n^{1-q}(x_n)$ and $\frac{1}{q} = \sum_{i=1}^{n-1} \frac{1}{p_i}$, the above inequality can be rewritten in the form,

$$I \leq \prod_{i=1}^{n-1} \left[\int_{\Omega} F_i(x_i) f_i^{p_i}(x_i) \phi_{ii}^{p_i}(x_i) d\mu_i(x_i) \right]^{\frac{1}{p_i}} \cdot I^{1-\frac{1}{q}},$$

that is, we obtain (1.74).

Reverse inequalities are derived in the same way, by virtue of the reverse Hölder inequality.

It remains to prove that inequalities (1.73) and (1.74) are equivalent. It is enough to check that inequality (1.73) follows from (1.74). For this purpose, suppose that inequality

(1.74) holds. Then we have

$$\begin{split} &\int_{\Omega^n} K(x_1,...,x_n) \prod_{i=1}^n f_i(x_i) d\mu_1(x_1)...d\mu_n(x_n) \\ &= \int_{\Omega} \phi_{nn}^{-1}(x_n) F_n^{-\frac{1}{p_n}}(x_n) \left[\int_{\Omega^{n-1}} K(x_1,...,x_n) \prod_{i=1}^{n-1} f_i(x_i) d\mu_1(x_1)...d\mu_{n-1}(x_{n-1}) \right] \\ &\times F_n^{\frac{1}{p_n}}(x_n) f_n(x_n) \phi_{nn}(x_n) d\mu_n(x_n). \end{split}$$

In addition, the Hölder inequality with conjugate exponents q and p_n yields

$$\begin{split} &\int_{\Omega^n} K(x_1,...,x_n) \prod_{i=1}^n f_i(x_i) d\mu_1(x_1)...d\mu_n(x_n) \\ &\leq \left\{ \int_{\Omega} h(x_n) \left[\int_{\Omega^{n-1}} K(x_1,...,x_n) \prod_{i=1}^{n-1} f_i(x_i) d\mu_1(x_1)...d\mu_{n-1}(x_{n-1}) \right]^q d\mu_n(x_n) \right\}^{\frac{1}{q}} \\ &\times \left[\int_{\Omega} \phi_{nn}^{p_n}(x_n) F_n(x_n) f_n^{p_n}(x_n) d\mu_n(x_n) \right]^{\frac{1}{p_n}}, \end{split}$$

and the result follows from (1.74).

Remark 1.21 Considering the proof of the previous theorem, it follows that the equality in (1.73) and (1.74) is possible if and only if it holds in the Hölder inequality. More precisely, this means that functions $f_i^{p_i}(x_i) \prod_{j=1}^n \phi_{ij}^{p_i}(x_j)$ are effectively proportional. Hence, equality in (1.73) and (1.74) holds if and only if

$$f_i(x_i) = C_i \phi_{ii}(x_i)^{\frac{p_i}{1-p_i}}, \ i = 1, \dots, n,$$
(1.75)

for some constants $C_i \ge 0$. That is possible only if the functions

$$\frac{\prod_{j=1, j\neq i}^{n} \phi_{jj}^{\frac{p_{j}}{1-p_{j}}}(x_{j})}{\prod_{j=1, j\neq i}^{n} \phi_{ij}^{p_{j}}(x_{j})}, \quad i=1,2,\ldots,n,$$

are appropriate constants, and

$$\int_{\Omega} F_i(x_i) \phi_{ii}^{\frac{p_i}{1-p_i}}(x_i) d\mu_i(x_i) < \infty, \quad i=1,2,\ldots n.$$

Otherwise, the inequalities in Theorem 1.18 are strict.

Remark 1.22 It should be emphasized here that Theorem 1.18 is a multidimensional extension of Theorem 1.9 from Section 1.2.

1.6.2 Application to homogeneous kernels

In the sequel we consider multidimensional versions of Hilbert inequality (i. e. Theorem 1.18) equipped with $\Omega = \mathbb{R}_+$, Lebesgue measures μ_i , i = 1, ..., n, a non-negative homogeneous kernel $K : \mathbb{R}^n_+ \to \mathbb{R}$, and the weight functions $\phi_{ij}(x_j) = x_j^{A_{ij}}$, where $A_{ij} \in \mathbb{R}$, i, j = 1, ..., n.

We define multidimensional version of integral $k(\alpha)$ (see definition (1.20)), but in this section it will be more convenient to slightly change the definition:

$$k(\alpha_1,\ldots,\alpha_{n-1}) = \int_{\mathbb{R}^{n-1}_+} K(1,t_1\ldots,t_{n-1}) t_1^{\alpha_1}\cdots t_{n-1}^{\alpha_{n-1}} dt_1\cdots dt_{n-1}, \qquad (1.76)$$

where we assume that $k(\alpha_1, \ldots, \alpha_{n-1}) < \infty$ for $\alpha_1, \ldots, \alpha_{n-1} > -1$ and $\alpha_1 + \cdots + \alpha_{n-1} + n < s+1$.

Theorem 1.19 Let $\sum_{i=1}^{n} \frac{1}{p_i} = 1$, $p_i > 1$, i = 1, 2, ..., n, and let $K : \mathbb{R}^n_+ \to \mathbb{R}$ be a nonnegative measurable homogeneous function of degree -s, s > 0. Further, let A_{ij} , i, j = 1, ..., n, be real parameters such that $\sum_{i=1}^{n} A_{ij} = 0$ for j = 1, ..., n. If $f_i : \mathbb{R}_+ \to \mathbb{R}$, $f_i \neq 0$, i = 1, ..., n, are non-negative measurable functions, then the inequalities

$$\int_{\mathbb{R}^{n}_{+}} K(x_{1},\dots,x_{n}) \prod_{i=1}^{n} f_{i}(x_{i}) dx_{1} \cdots dx_{n} < L \prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{n-s-1+p_{i}\alpha_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$
(1.77)

and

$$\int_{0}^{\infty} x_{n}^{(1-q)(n-1-s)-q\alpha_{n}} \left[\int_{\mathbb{R}^{n-1}_{+}} K(x_{1},\dots,x_{n}) \prod_{i=1}^{n-1} f_{i}(x_{i}) dx_{1} \cdots dx_{n-1} \right]^{q} dx_{n}$$

$$< L^{q} \prod_{i=1}^{n-1} \left[\int_{0}^{\infty} x_{i}^{n-1-s+p_{i}\alpha_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{q}{p_{i}}}$$
(1.78)

hold and are equivalent, where

$$L = k^{\frac{1}{p_1}} (p_1 A_{12}, \dots, p_1 A_{1n}) \cdot k^{\frac{1}{p_2}} (s - n - p_2(\alpha_2 - A_{22}), p_2 A_{23}, \dots, p_2 A_{2n})$$

$$\cdots k^{\frac{1}{p_n}} (p_n A_{n2}, \dots, p_n A_{n,n-1}, s - n - p_n(\alpha_n - A_{nn})), \qquad (1.79)$$

$$\frac{1}{q} = \sum_{i=1}^{n-1} \frac{1}{p_i}, \ \alpha_i = \sum_{j=1}^n A_{ij}, \ p_i A_{ij} > -1, \ i \neq j, \ p_i(A_{ii} - \alpha_i) > n - s - 1.$$

Proof. Set $\phi_{ij}(x_j) = x_j^{A_{ij}}$ in Theorem 1.18, where $\sum_{i=1}^n A_{ij} = 0$ for every j = 1, ..., n. It is enough to calculate the functions $F_i(x_i)$, i = 1, ..., n. Using homogeneity of the kernel *K* and obvious change of variables, we have

$$F_{1}(x_{1}) = \int_{\mathbb{R}^{n-1}_{+}} K(x_{1}, x_{2}, \dots, x_{n}) \prod_{j=2}^{n} x_{j}^{p_{1}A_{1j}} dx_{2} \cdots dx_{n}$$

= $\int_{\mathbb{R}^{n-1}_{+}} x_{1}^{-s} K(1, x_{2}/x_{1}, \dots, x_{n}/x_{1}) \prod_{j=2}^{n} x_{j}^{p_{1}A_{1j}} dx_{2} \cdots dx_{n}$
= $x_{1}^{n-1-s+p_{1}(\alpha_{1}-A_{11})} k(p_{1}A_{12}, \dots, p_{1}A_{1n}).$

On the other hand, using homogeneity of K and the change of variables

$$x_1 = x_2 \cdot \frac{1}{t_2}, x_i = x_2 \cdot \frac{t_i}{t_2}, i = 3, \dots, n$$
, so that $\frac{\partial(x_1, x_3, \dots, x_n)}{\partial(t_2, t_3, \dots, t_n)} = x_2^{n-1} t_2^{-n}$,

where $\frac{\partial(x_1, x_3, \dots, x_n)}{\partial(t_2, t_3, \dots, t_n)}$ denotes the Jacobian of the transformation, we have

$$F_{2}(x_{2}) = \int_{\mathbb{R}^{n-1}_{+}} K(x_{1}, x_{2}, \dots, x_{n}) \prod_{j=1, j\neq 2}^{n} x_{j}^{p_{2}A_{2j}} dx_{1} dx_{3} \cdots dx_{n}$$

$$= \int_{\mathbb{R}^{n-1}_{+}} x_{1}^{-s} K(1, x_{2}/x_{1}, \dots, x_{n}/x_{1}) \prod_{j=1, j\neq 2}^{n} x_{j}^{p_{2}A_{2j}} dx_{1} dx_{3} \cdots dx_{n}$$

$$= x_{2}^{n-1-s+p_{2}(\alpha_{2}-A_{22})} k(s-n-p_{2}(\alpha_{2}-A_{22}), p_{2}A_{23}, \dots, p_{2}A_{2n}).$$

In a similar manner we obtain

$$F_{i}(x_{i}) = x_{i}^{n-s-1+p_{i}(\alpha_{i}-A_{i})} \times k(p_{i}A_{i2},...,p_{i}A_{i,i-1},s-n-p_{i}(\alpha_{i}-A_{i}),p_{i}A_{i,i+1},...,p_{i}A_{i})$$

for i = 3, ..., n. This gives inequalities (1.77) and (1.78) with inequality sign \leq .

Finally, condition (1.75) immediately gives that nontrivial case of equality in (1.77) and (1.78) leads to the divergent integrals. This completes the proof.

Motivated by the ideas from Section 1.4, we can also establish conditions under which the constant factors L and L^q are the best possible in inequalities (1.77) and (1.78). In order to obtain such factors, it is natural to impose the following conditions on the parameters A_{ij} :

$$p_{j}A_{ji} = s - n - p_{i}(\alpha_{i} - A_{ii}), \ i = 2, \dots, n, \ j \neq i,$$

$$p_{i}A_{ik} = p_{j}A_{jk}, \ k \neq i, j, \ k \neq 1.$$
 (1.80)

The missing cases i = 1 and k = 1 can be deduced from (1.80) as follows:

$$p_1(\alpha_1 - A_{11}) = p_1 A_{1j} + p_1 \sum_{i \neq 1, j} A_{1i} = s - n - p_j(\alpha_j - A_{jj}) + p_j \sum_{i \neq j, 1} A_{ji} = s - n - p_j A_{j1}$$

where $j \neq 1$. Thus, the complete set of conditions is

$$p_{j}A_{ji} = s - n - p_{i}(\alpha_{i} - A_{ii}), i, j = 1, 2, \dots, n, i \neq j,$$

$$p_{i}A_{ik} = p_{j}A_{jk}, k \neq i, j.$$
(1.81)

Theorem 1.20 Suppose that real parameters A_{ij} , i, j = 1, ..., n, fulfill conditions from Theorem 1.19 and conditions given in (1.81). If the kernel $K : \mathbb{R}^n_+ \to \mathbb{R}$ is as in Theorem 1.19 and for every i = 2, ..., n

$$K(1, t_2, \dots, t_i, \dots, t_n) \le CK(1, t_2, \dots, 0, \dots, t_n), \ 0 \le t_i \le 1, \ t_j \ge 0, \ j \ne i,$$
(1.82)

for some C > 0, then the constants L and L^q are the best possible in inequalities (1.77) and (1.78). In this case $L = k(p_1A_{12}, p_1A_{13}, \dots, p_1A_{1n})$.

Proof. It is easy to see that $n - s + p_i \alpha_i = -p_i \widetilde{A}_i$, where $\widetilde{A}_i = p_1 A_{1i}$ for $i \neq 1$, and $\widetilde{A}_1 = p_n A_{n1}$. Hence, inequality (1.77) can be rewritten as

$$\int_{\mathbb{R}^{n}_{+}} K(x_{1},\dots,x_{n}) \prod_{i=1}^{n} f_{i}(x_{i}) dx_{1} \cdots dx_{n} < L \prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{-1-p_{i}\widetilde{A}_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}, \quad (1.83)$$

where $L = k(\widetilde{A}_2, \ldots, \widetilde{A}_n)$.

Now, suppose that the above constant factor L is not the best possible. Then, there exists a positive constant M, smaller than L, such that inequality (1.83) still holds after replacing L by M. For this purpose, set

$$\widetilde{f}_i(x_i) = \begin{cases} 0, & x \in (0,1) \\ x_i^{\widetilde{A}_i - \frac{\varepsilon}{p_i}}, & x \in [1,\infty) \end{cases}, \ i = 1, \dots, n,$$

where $0 < \varepsilon < \min_{1 \le i \le n} \{p_i + p_i \widetilde{A}_i\}$. If we substitute these functions in (1.83), then the right-hand side of the inequality becomes $\frac{M}{\varepsilon}$, since

$$\prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{-1-p_{i}\widetilde{A}_{i}} \widetilde{f}_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}} = \frac{1}{\varepsilon}.$$
(1.84)

Further, let *J* denotes the left-hand side of inequality (1.83), for the above choice of functions \tilde{f}_i . Utilizing substitutions $u_i = \frac{x_i}{x_1}$, i = 2, ..., n, in *J*, we find that

$$J = \int_{1}^{\infty} x_1^{-1-\varepsilon} \left[\int_{\frac{1}{x_1}}^{\infty} \cdots \int_{\frac{1}{x_1}}^{\infty} K(1, u_2, \dots, u_n) \prod_{i=2}^{n} u_i^{\widetilde{A}_i - \frac{\varepsilon}{p_i}} du_2 \dots du_n \right] dx_1.$$
(1.85)

Moreover, J can be estimated as

$$J \geq \int_{1}^{\infty} x_{1}^{-1-\varepsilon} \left[\int_{\mathbb{R}^{n-1}_{+}} K(1, u_{2}, \dots, u_{n}) \prod_{i=2}^{n} u_{i}^{\widetilde{A}_{i} - \frac{\varepsilon}{p_{i}}} du_{2} \dots du_{n} \right] dx_{1}$$
$$- \int_{1}^{\infty} x_{1}^{-1-\varepsilon} \sum_{j=2}^{n} I_{j}(x_{1}) dx_{1}$$
$$= \frac{1}{\varepsilon} k \left(\widetilde{A}_{2} - \frac{\varepsilon}{p_{2}}, \dots, \widetilde{A}_{n} - \frac{\varepsilon}{p_{n}} \right) - \int_{1}^{\infty} x_{1}^{-1-\varepsilon} \sum_{j=2}^{n} I_{j}(x_{1}) dx_{1}, \qquad (1.86)$$

where for $j = 2, ..., n, I_j(x_1)$ is defined by

$$I_j(x_1) = \int_{D_j} K(1, u_2, \dots, u_n) \prod_{i=2}^n u_i^{\widetilde{A}_i - \frac{\varepsilon}{p_i}} du_2 \dots du_n,$$

with $D_j = \{(u_2, u_3, \ldots, u_n); 0 < u_j \le \frac{1}{x_1}, 0 < u_k < \infty, k \neq j\}.$

Without losing generality, it is enough to estimate the integral $I_2(x_1)$. The case of n = 2 was proved in Theorem 1.13 (see Section 1.4). For $n \ge 3$ we have

$$I_{2}(x_{1}) \leq C \left[\int_{\mathbb{R}^{n-2}_{+}} K(1,0,u_{3},\ldots,u_{n}) \prod_{i=3}^{n} u_{i}^{\widetilde{A}_{i}-\frac{\varepsilon}{p_{i}}} du_{3}\ldots du_{n} \right]$$
$$\times \int_{0}^{\frac{1}{x_{1}}} u_{2}^{\widetilde{A}_{2}-\frac{\varepsilon}{p_{2}}} du_{2}$$
$$= C \left(1 - \frac{\varepsilon}{p_{2}} + \widetilde{A}_{2} \right)^{-1} x_{1}^{\frac{\varepsilon}{p_{2}}-\widetilde{A}_{2}-1} k \left(\widetilde{A}_{3} - \frac{\varepsilon}{p_{3}},\ldots,\widetilde{A}_{n} - \frac{\varepsilon}{p_{n}} \right)$$

where $k(\widetilde{A}_3 - \frac{\varepsilon}{p_3}, \dots, \widetilde{A}_n - \frac{\varepsilon}{p_n})$ is well-defined since obviously $\widetilde{A}_3 + \dots + \widetilde{A}_n < s - n + 2$. Hence, we have $I_j(x_1) = x_1^{\varepsilon/p_j - \widetilde{A}_j - 1}O_j(1)$ for $\varepsilon \to 0^+$, $j = 2, \dots, n$, and consequently

$$\int_{1}^{\infty} x_1^{-1-\varepsilon} \sum_{j=2}^{n} I_j(x_1) dx_1 = O(1).$$
(1.87)

Therefore, taking into account (1.84), (1.86), and (1.87), it follows that $L \leq M$ when $\varepsilon \to 0^+$, which is an obvious contradiction. This means that the constant *L* is the best possible in (1.83). Moreover, since the equivalence preserves the best possible constant, the proof is completed.

Remark 1.23 If the parameters A_{ij} are defined by $A_{ii} = \frac{(n-s)(p_i-1)}{p_i^2}$ and $A_{ij} = \frac{s-n}{p_i p_j}$, $i \neq j$, i, j = 1, ..., n, then we have

$$\sum_{i=1}^{n} A_{ij} = (s-n) \sum_{i \neq j} \frac{1}{p_i p_j} + (n-s) \frac{p_j - 1}{p_j^2} = \frac{s-n}{p_j} \left(\sum_{i=1}^{n} \frac{1}{p_i} - 1 \right) = 0,$$

for j = 1, 2, ..., n. Obviously, due to the symmetry, it follows that $\alpha_i = \sum_{j=1}^n A_{ij} = 0$, for i = 1, ..., n. Moreover, parameters A_{ij} , i, j = 1, ..., n, fulfill the set of conditions (1.81), hence, in this case Theorem 1.20 yields the following inequality

$$\int_{\mathbb{R}^{n}_{+}} K(x_{1},\ldots,n) \prod_{i=1}^{n} f_{i}(x_{i}) dx_{1} \cdots dx_{n} < L \prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{n-s-1} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}},$$

where $L = k(\frac{s-n}{p_2}, \dots, \frac{s-n}{p_n})$ is the best possible constant. For s = n-1 we obtain the nonweighted case with the best possible constant $L = k(-\frac{1}{p_2}, \dots, -\frac{1}{p_n})$ (compare with Remark 1.9).

1.6.3 Examples

We proceed with various examples of multidimensional Hilbert-type inequalities. In order to establish some particular results, we first indicate here a few lemmas. For the proof of the following lemma the reader is referred to [152].

Lemma 1.3 *If* $n \in \mathbb{N}$, $r_i > 0$, i = 1, ..., n, *then*

$$\int_{\mathbb{R}^{n-1}_{+}} \frac{\prod_{i=1}^{n-1} u_i^{r_i-1}}{\left(1 + \sum_{i=1}^{n-1} u_i\right)^{\sum_{i=1}^{n} r_i}} du_1 \dots du_{n-1} = \frac{\prod_{i=1}^{n} \Gamma(r_i)}{\Gamma(\sum_{i=1}^{n} r_i)},$$
(1.88)

where Γ is the usual Gama function.

Moreover, the trivial substitution $u_i = t_i^{\lambda}$, i = 1, ..., n-1, applied to relation (1.88), yields another integral formula:

Lemma 1.4 If $n \in \mathbb{N}$, $s, \lambda > 0$, $\beta_i > -1$, $i = 1, \dots, n-1$, and $\sum_{i=1}^{n-1} \beta_i < \lambda s - n + 1$, then

$$\int_{\mathbb{R}^{n-1}_{+}} \frac{\prod_{i=1}^{n-1} t_{i}^{\beta_{i}}}{\left(1 + \sum_{i=1}^{n-1} t_{i}^{\lambda}\right)^{s}} dt_{1} \dots dt_{n-1}$$

$$= \frac{1}{\Gamma(s)\lambda^{n-1}} \left[\prod_{i=1}^{n-1} \Gamma\left(\frac{\beta_{i}+1}{\lambda}\right)\right] \Gamma\left(s - \frac{1}{\lambda}\sum_{i=1}^{n-1} (\beta_{i}+1)\right).$$
(1.89)

The following lemma will be needed when considering a particular 3-dimensional case.

Lemma 1.5 *Let* s > 0, $a \ge 0$, and b > 0. *Further, let* α_1 , $\alpha_2 > -1$, $\alpha_1 + \alpha_2 < s - 2$, and

$$k(\alpha_1, \alpha_2) = \int_{\mathbb{R}^2_+} \frac{t_1^{\alpha_1} t_2^{\alpha_2}}{(a \min\{1, t_1, t_2\} + b \max\{1, t_1, t_2\})^s} dt_1 dt_2.$$

Then

$$k(\alpha_{1},\alpha_{2}) = \frac{b^{-s}}{(\alpha_{1}+1)(\alpha_{2}+1)} \sum_{i=1}^{2} F\left(s,\alpha_{i}+1;\alpha_{i}+2;-\frac{a}{b}\right)$$

$$-\frac{b^{-s}}{(\alpha_{1}+1)(\alpha_{2}+1)} F\left(s,\alpha_{1}+\alpha_{2}+2;\alpha_{1}+\alpha_{2}+3;-\frac{a}{b}\right)$$

$$+b^{-s} \sum_{i=1}^{2} \int_{0}^{1} t_{i}^{\alpha_{i}} F\left(s,s-\alpha_{i+1}-1;s-\alpha_{i+1};-\frac{a}{b}t_{i}\right) dt_{i}$$

$$+\frac{b^{-s}(\alpha_{1}+\alpha_{2}+2)}{(\alpha_{1}+1)(\alpha_{2}+1)(s-\alpha_{1}-\alpha_{2}-1)}$$

$$\times F\left(s,s-\alpha_{1}-\alpha_{2}-2;s-\alpha_{1}-\alpha_{2}-1;-\frac{a}{b}\right)$$

$$-b^{-s} \sum_{i=1}^{2} \frac{1}{(\alpha_{i}+1)(s-\alpha_{i+1}-1)} F\left(s,s-\alpha_{i+1}-1;s-\alpha_{i+1};-\frac{a}{b}\right),$$

(1.90)

where *F* denotes hypergeometric function (1.34) and the indices are taken modulo 2. *Proof.* By virtue of the Fubini theorem we have $k(\alpha_1, \alpha_2) = I_1 + I_2$, where

$$I_1 = \int_0^1 t_1^{\alpha_1} \left(\int_0^\infty \frac{t_2^{\alpha_2}}{(a\min\{t_1, t_2\} + b\max\{1, t_2\})^s} dt_2 \right) dt_1$$

and

$$I_{2} = \int_{1}^{\infty} t_{1}^{\alpha_{1}} \left(\int_{0}^{\infty} \frac{t_{2}^{\alpha_{2}}}{(a\min\{1,t_{2}\} + b\max\{t_{1},t_{2}\})^{s}} dt_{2} \right) dt_{1}.$$

In what follows, we shall express integral I_1 by a hypergeometric function. It is easy to see that $I_1 = I_{11} + I_{12}$, where

$$I_{11} = \int_0^1 t_1^{\alpha_1} \left(\int_0^1 \frac{t_2^{\alpha_2}}{(a\min\{t_1, t_2\} + b)^s} dt_2 \right) dt_1$$

and

$$I_{12} = \int_0^1 t_1^{\alpha_1} \left(\int_1^\infty \frac{t_2^{\alpha_2}}{(at_1 + bt_2)^s} dt_2 \right) dt_1.$$

The integral I_{11} can be transformed in the following way:

$$I_{11} = \int_0^1 t_1^{\alpha_1} \left(\int_0^{t_1} \frac{t_2^{\alpha_2}}{(at_2 + b)^s} dt_2 \right) dt_1 + \int_0^1 t_1^{\alpha_1} \left(\int_{t_1}^1 \frac{t_2^{\alpha_2}}{(at_1 + b)^s} dt_2 \right) dt_1.$$
(1.91)

Applying the classical calculus, we have

$$\int_{0}^{1} t_{1}^{\alpha_{1}} \left(\int_{0}^{t_{1}} \frac{t_{2}^{\alpha_{2}}}{(at_{2}+b)^{s}} dt_{2} \right) dt_{1}$$

$$= \int_{0}^{1} t_{2}^{\alpha_{2}} \left(\int_{t_{2}}^{1} \frac{t_{1}^{\alpha_{1}}}{(at_{2}+b)^{s}} dt_{1} \right) dt_{2} = \int_{0}^{1} t_{2}^{\alpha_{2}} (at_{2}+b)^{-s} \left(\int_{t_{2}}^{1} t_{1}^{\alpha_{1}} dt_{1} \right) dt_{2}$$

$$= \frac{b^{-s}}{\alpha_{1}+1} \int_{0}^{1} t_{2}^{\alpha_{2}} (1-t_{2}^{\alpha_{1}+1}) \left(1+\frac{a}{b}t_{2} \right)^{-s} dt_{2}$$

$$= \frac{b^{-s}}{\alpha_{1}+1} \left[\frac{1}{\alpha_{2}+1} F\left(s, \alpha_{2}+1; \alpha_{2}+2; -\frac{a}{b} \right) \right]$$

$$- \frac{1}{\alpha_{1}+\alpha_{2}+2} F\left(s, \alpha_{1}+\alpha_{2}+2; \alpha_{1}+\alpha_{2}+3; -\frac{a}{b} \right) \right], \qquad (1.92)$$

and similarly,

$$\int_{0}^{1} t_{1}^{\alpha_{1}} \left(\int_{t_{1}}^{1} \frac{t_{2}^{\alpha_{2}}}{(at_{1}+b)^{s}} dt_{2} \right) dt_{1}$$

$$= \frac{b^{-s}}{\alpha_{2}+1} \left[\frac{1}{\alpha_{1}+1} F\left(s, \alpha_{1}+1; \alpha_{1}+2; -\frac{a}{b}\right) -\frac{1}{\alpha_{1}+\alpha_{2}+2} F\left(s, \alpha_{1}+\alpha_{2}+2; \alpha_{1}+\alpha_{2}+3; -\frac{a}{b}\right) \right].$$
(1.93)

Now, setting (1.92) and (1.93) in (1.91), we obtain

$$I_{11} = \frac{b^{-s}}{(\alpha_1 + 1)(\alpha_2 + 1)} \sum_{i=1}^{2} F\left(s, \alpha_i + 1; \alpha_i + 2; -\frac{a}{b}\right) -\frac{b^{-s}}{(\alpha_1 + 1)(\alpha_2 + 1)} F\left(s, \alpha_1 + \alpha_2 + 2; \alpha_1 + \alpha_2 + 3; -\frac{a}{b}\right), \quad (1.94)$$

while the substitution $u = \frac{1}{t_2}$ yields

$$I_{12} = \int_{0}^{1} t_{1}^{\alpha_{1}} \left(\int_{1}^{\infty} \frac{t_{2}^{\alpha_{2}}}{(at_{1} + bt_{2})^{s}} dt_{2} \right) dt_{1}$$

= $b^{-s} \int_{0}^{1} t_{1}^{\alpha_{1}} \left(\int_{0}^{1} u^{s - \alpha_{2} - 2} \left(1 + \frac{a}{b} t_{1} u \right)^{-s} du \right) dt_{1}$
= $b^{-s} \int_{0}^{1} t_{1}^{\alpha_{1}} F\left(s, s - \alpha_{2} - 1; s - \alpha_{2}; -\frac{a}{b} t_{1} \right) dt_{1}.$ (1.95)

Finally, from (1.94) and (1.95) we have

$$I_{1} = I_{11} + I_{12} =$$

$$= \frac{b^{-s}}{(\alpha_{1} + 1)(\alpha_{2} + 1)} \sum_{i=1}^{2} F\left(s, \alpha_{i} + 1; \alpha_{i} + 2; -\frac{a}{b}\right)$$

$$- \frac{b^{-s}}{(\alpha_{1} + 1)(\alpha_{2} + 1)} F\left(s, \alpha_{1} + \alpha_{2} + 2; \alpha_{1} + \alpha_{2} + 3; -\frac{a}{b}\right)$$

$$+ b^{-s} \int_{0}^{1} t_{1}^{\alpha_{1}} F\left(s, s - \alpha_{2} - 1; s - \alpha_{2}; -\frac{a}{b}t_{1}\right) dt_{1}.$$
(1.96)

Repeating the above procedure for the integrals

$$I_{21} = \int_{1}^{\infty} t_{1}^{\alpha_{1}} \left(\int_{0}^{1} \frac{t_{2}^{\alpha_{2}}}{(at_{2} + bt_{1})^{s}} dt_{2} \right) dt_{1}$$

and

$$I_{22} = \int_{1}^{\infty} t_{1}^{\alpha_{1}} \left(\int_{1}^{\infty} \frac{t_{2}^{\alpha_{2}}}{(a+b\max\{t_{1},t_{2}\})^{s}} dt_{2} \right) dt_{1},$$

since $I_{21} + I_{22} = I_2$, we obtain (1.90), as claimed.

Multidimensional Hilbert-type inequalities that follow will be based on Theorems 1.19 and 1.20. More precisely, under the assumptions of Theorem 1.20, inequalities (1.77) and (1.78) can be rewritten as

$$\int_{\mathbb{R}^{n}_{+}} K(x_{1}, \dots, x_{n}) \prod_{i=1}^{n} f_{i}(x_{i}) dx_{1} \cdots dx_{n} < L \prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{-1-p_{i}\widetilde{A}_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$
(1.97)

and

$$\int_{0}^{\infty} x_{n}^{(1-q)(-1-p_{n}\widetilde{A}_{n})} \left[\int_{\mathbb{R}^{n-1}_{+}} K(x_{1},\dots,x_{n}) \prod_{i=1}^{n-1} f_{i}(x_{i}) dx_{1} \cdots dx_{n-1} \right]^{q} dx_{n} < L^{q} \prod_{i=1}^{n-1} \left[\int_{0}^{\infty} x_{i}^{-1-p_{i}\widetilde{A}_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{q}{p_{i}}},$$
(1.98)

where $\widetilde{A}_i = p_1 A_{1i}$ for $i \neq 1$, $\widetilde{A}_1 = p_n A_{n1}$, $\frac{1}{q} = \sum_{i=1}^{n-1} \frac{1}{p_i}$, $p_1 A_{1j} > -1$, $i \neq j$, and $p_1 (A_{11} - \alpha_1) > n - s - 1$. Moreover, constant factors $L = k(\widetilde{A}_2, \dots, \widetilde{A}_n)$ and L^q are the best possible in inequalities (1.97) and (1.98).

We first consider a particular case of parameters A_{ij} , i, j = 1, ..., n, which fulfill the set of conditions (1.81), necessary in establishing the inequalities with the best possible constant factors. These are the parameters

$$A_{ij} = \frac{s - p_j}{p_i p_j}, \quad i \neq j, \quad \text{and} \quad A_{ii} = \frac{(s - p_i)(1 - p_i)}{p_i^2}.$$
 (1.99)

Hence, considering inequalities (1.97) and (1.98) with parameters A_{ij} , i, j = 1, ..., n, defined by (1.99), we have the following consequence:

Corollary 1.4 Let $\sum_{i=1}^{n} \frac{1}{p_i} = 1$, $p_i > 1$, i = 1, 2, ..., n, and let $\frac{1}{q} = \sum_{i=1}^{n-1} \frac{1}{p_i}$. Further, suppose that $K : \mathbb{R}^n_+ \to \mathbb{R}$ is a non-negative measurable homogeneous function of degree -s, s > 0, fulfilling condition (1.82). If $f_i : \mathbb{R}_+ \to \mathbb{R}$, $f_i \neq 0$, i = 1, ..., n, are non-negative measurable functions, then inequalities

$$\int_{\mathbb{R}^{n}_{+}} K(x_{1},\dots,x_{n}) \prod_{i=1}^{n} f_{i}(x_{i}) dx_{1} \cdots dx_{n} < L \prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{p_{i}-s-1} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$
(1.100)

and

$$\int_{0}^{\infty} x_{n}^{\frac{s}{p_{n-1}}-1} \left[\int_{\mathbb{R}^{n-1}_{+}} K(x_{1}, \dots, x_{n}) \prod_{i=1}^{n-1} f_{i}(x_{i}) dx_{1} \cdots dx_{n-1} \right]^{q} dx_{n} \\ < L^{q} \prod_{i=1}^{n-1} \left[\int_{0}^{\infty} x_{i}^{p_{i}-s-1} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{q}{p_{i}}}$$
(1.101)

hold and are equivalent, where the constant factors $L = k\left(\frac{s-p_2}{p_2}, \dots, \frac{s-p_n}{p_n}\right)$ and L^q are the best possible in both inequalities.

Remark 1.24 Observe that the kernel $K(x_1, ..., x_n) = (x_1 + ... + x_n)^{-s}$, s > 0, fulfills condition (1.82) from Theorem 1.20 i.e. Corollary 1.4. In this case, having in mind integral formula (1.88), the above constant factor $L = k(\frac{s-p_2}{p_2}, ..., \frac{s-p_n}{p_n})$ can be expressed in terms of the Gamma function i.e. $L = \frac{1}{\Gamma(s)} \prod_{i=1}^{n} \Gamma(\frac{s}{p_i})$. This particular case was studied by B. Yang in paper [154]. Moreover, some other particular cases regarding this kernel were extensively studied in [156].

Remark 1.25 Setting n = 2, $K(x,y) = (a \min\{x,y\} + b \max\{x,y\})^{-1}$, p = q = 2 in Corollary 1.4, we obtain the result from [82] with the best possible constant factor $D(a,b) = k(-\frac{1}{2}) = \frac{4}{b}F(1,\frac{1}{2};\frac{3}{2};-\frac{a}{b})$. For each choice of parameters a, b we compute the best possible constant D(a,b):

- (i) $a = b = 1, D(1, 1) = \pi$, as in [82],
- (ii) a = 0, b = 1, D(0, 1) = 4, as in [82],

(iii) $a = 1, b = 2, D(1,2) = 2\sqrt{2} \arctan\left(\frac{1}{\sqrt{2}}\right),$

(iv)
$$a = 1, b = 3, D(1,3) = \frac{2\pi}{3\sqrt{3}},$$

(v) $a = 2, b = 1, D(2, 1) = 2\sqrt{2} \arctan \sqrt{2}$.

In order to exploit Lemma 1.4, in the sequel we consider the homogeneous kernel $K(x_1, \ldots, x_n) = \frac{\sum_{i=1}^n x_i^{s(\lambda-1)}}{(\sum_{i=1}^n x_i^{\lambda})^s}$, $s, \lambda > 0$. Clearly, this kernel fulfills the assumptions of Corollary 1.4, hence, as a consequence, we have the following pair of inequalities with the constant factors expressed in terms of the Gamma function:

Corollary 1.5 Let $\sum_{i=1}^{n} \frac{1}{p_i} = 1$, $p_i > 1$, i = 1, 2, ..., n, and let $\frac{1}{q} = \sum_{i=1}^{n-1} \frac{1}{p_i}$. If $f_i : \mathbb{R}_+ \to \mathbb{R}$, $f_i \neq 0$, i = 1, ..., n, are non-negative measurable functions, then inequalities

$$\int_{\mathbb{R}^{n}_{+}} \frac{\sum_{i=1}^{n} x_{i}^{s(\lambda-1)}}{\left(\sum_{i=1}^{n} x_{i}^{\lambda}\right)^{s}} \prod_{i=1}^{n} f_{i}(x_{i}) dx_{1} \dots dx_{n} < L_{1} \prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{p_{i}-s-1} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$
(1.102)

and

$$\left[\int_{0}^{\infty} x_{n}^{\frac{s}{p_{n}-1}-1} \left(\int_{\mathbb{R}^{n-1}_{+}} \frac{\sum_{i=1}^{n} x_{i}^{s(\lambda-1)}}{\left(\sum_{i=1}^{n} x_{i}^{\lambda}\right)^{s}} \prod_{i=1}^{n-1} f_{i}(x_{i}) dx_{1} \cdots dx_{n-1} \right)^{q} dx_{n} \right]^{\frac{1}{q}}$$

$$< L_{1} \prod_{i=1}^{n-1} \left[\int_{0}^{\infty} x_{i}^{p_{i}-s-1} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$

$$(1.103)$$

hold and are equivalent, where the constant factor

$$L_1 = \frac{1}{\Gamma(s)\lambda^{n-1}} \sum_{j=1}^n \left[\left(\prod_{i=1, i \neq j}^n \Gamma\left(\frac{s}{p_i \lambda}\right) \right) \cdot \Gamma\left(\frac{sp_j(\lambda-1)+s}{p_j \lambda}\right) \right]$$
(1.104)

is the best possible in both inequalities.

Proof. It is enough to calculate the constant $L_1 = k(\frac{s-p_2}{p_2}, \dots, \frac{s-p_n}{p_n})$. Using definition (1.76) of the integral $k(\alpha_1, \dots, \alpha_{n-1})$, L_1 can be represented in the form

$$L_{1} = \int_{\mathbb{R}^{n-1}_{+}} \frac{1 + t_{1}^{s(\lambda-1)} + \ldots + t_{n-1}^{s(\lambda-1)}}{\left(1 + \sum_{i=1}^{n} t_{i}^{\lambda}\right)^{s}} t_{1}^{\frac{s}{p_{2}}-1} \ldots t_{n-1}^{\frac{s}{p_{n}}-1} dt_{1} \ldots dt_{n-1} = \sum_{k=0}^{n-1} I_{k},$$

where

$$I_{0} = \int_{\mathbb{R}^{n-1}_{+}} \frac{t_{1}^{\frac{s}{p_{2}}-1} \dots t_{n-1}^{\frac{s}{p_{n}}-1}}{\left(1 + \sum_{i=1}^{n} t_{i}^{\lambda}\right)^{s}} dt_{1} \dots dt_{n-1}$$

and

$$I_{k} = \int_{\mathbb{R}^{n-1}_{+}} \frac{t_{1}^{\frac{s}{p_{2}}-1} \dots t_{k}^{s(\lambda-1)+\frac{s}{p_{k+1}}-1} \dots t_{n-1}^{\frac{s}{p_{n}}-1}}{\left(1+\sum_{i=1}^{n} t_{i}^{\lambda}\right)^{s}} dt_{1} \dots dt_{n-1}, \text{ for } k = 1, \dots, n-1.$$

Now, taking into account Lemma 1.4, it follows that

$$I_0 = \frac{1}{\Gamma(s)\lambda^{n-1}} \left[\prod_{i=2}^n \Gamma\left(\frac{s}{p_i\lambda}\right) \right] \cdot \Gamma\left(\frac{sp_1(\lambda-1)+s}{p_1\lambda}\right)$$

and

$$I_{k} = \frac{1}{\Gamma(s)\lambda^{n-1}} \left[\prod_{i=1, i \neq k+1}^{n} \Gamma\left(\frac{s}{p_{i}\lambda}\right) \right] \cdot \Gamma\left(\frac{sp_{k+1}(\lambda-1)+s}{p_{k+1}\lambda}\right), \ k = 1, \dots, n-1,$$

t is, we have (1.104).

that is, we have (1.104).

We conclude this subsection with some particular 3-dimensional Hilbert-type inequalities with constant factors expressed in terms of a hypergeometric function. The following result arises from Corollary 1.4 for the case of kernel $K(x_1, x_2, x_3) = (a \min\{x_1, x_2, x_3\})$ x_2, x_3 + $b \max\{x_1, x_2, x_3\}$)^{-s}, $a \ge 0, b > 0$. Clearly, K has degree of homogeneity -s, s > 0, and fulfills condition (1.82).

Corollary 1.6 Let $\frac{1}{p_1} + \frac{1}{p_2} + \frac{1}{p_3} = 1$, $p_i > 1$, i = 1, 2, 3, and let $\frac{1}{q} = \frac{1}{p_1} + \frac{1}{p_2}$. Then inequalities

$$\int_{\mathbb{R}^{3}_{+}} \frac{f_{1}(x_{1})f_{2}(x_{2})f_{3}(x_{3})}{(a\min\{x_{1},x_{2},x_{3}\}+b\max\{x_{1},x_{2},x_{3}\})^{s}} dx_{1}dx_{2}dx_{3}$$

$$< L_{2}\prod_{i=1}^{3} \left[\int_{0}^{\infty} x_{i}^{p_{i}-s-1}f_{i}^{p_{i}}(x_{i})dx_{i}\right]^{\frac{1}{p_{i}}}$$
(1.105)

and

$$\left[\int_{0}^{\infty} x_{3}^{\frac{s}{p_{3}-1}-1} \left(\int_{\mathbb{R}^{2}_{+}} \frac{f_{1}(x_{1})f_{2}(x_{2})}{(a\min\{x_{1},x_{2},x_{3}\}+b\max\{x_{1},x_{2},x_{3}\})^{s}} dx_{1} dx_{2} \right)^{q} dx_{3} \right]^{\frac{1}{q}}$$

$$< L_{2} \prod_{i=1}^{2} \left[\int_{0}^{\infty} x_{i}^{p_{i}-s-1} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$
(1.106)

hold for all non-negative measurable functions $f_i : \mathbb{R}_+ \to \mathbb{R}$, $f_i \neq 0$, i = 1, 2, 3, where

$$\begin{split} L_2 &= \frac{p_2 p_3}{b^s s^2} \sum_{i=2}^3 F\left(s, \frac{s}{p_i}; \frac{s}{p_i} + 1; -\frac{a}{b}\right) - \frac{p_2 p_3}{b^s s^2} F\left(s, \frac{s}{p_2} + \frac{s}{p_3}; \frac{s}{p_2} + \frac{s}{p_3} + 1; -\frac{a}{b}\right) \\ &+ b^{-s} \sum_{i=2}^3 \int_0^1 t^{\frac{s}{p_i} - 1} F\left(s, \frac{s}{p_1} + \frac{s}{p_i}; \frac{s}{p_1} + \frac{s}{p_i} + 1; -\frac{a}{b}t\right) dt \\ &+ \frac{p_1 (p_2 + p_3)}{b^s s(s + p_1)} F\left(s, \frac{s}{p_1}; \frac{s}{p_1} + 1; -\frac{a}{b}\right) \\ &- \frac{p_2 p_3}{b^s s^2} \sum_{i=2}^3 \frac{1}{p_i - 1} F\left(s, s - \frac{s}{p_i}; s - \frac{s}{p_i} + 1; -\frac{a}{b}\right). \end{split}$$

Moreover, the constant factor L_2 is the best possible in both inequalities. In particular, for s = 1, $p_1 = p_2 = p_3 = 3$ and a = b = 1, we have $L_2 = \frac{55 \log 2}{2} - \frac{3\pi\sqrt{3}}{2}$.

Proof. Making use of Corollary 1.4 in the case of homogeneous kernel $K(x_1, x_2, x_3) = (a\min\{x_1, x_2, x_3\} + b\max\{x_1, x_2, x_3\})^{-s}$, it is enough to calculate the constant $L_2 = k(\frac{s-p_2}{p_2}, \frac{s-p_3}{p_3})$. Now the result follows from Lemma 1.5.

Remark 1.26 Setting the kernel $K(x_1, x_2, x_3) = (x_1 + x_2 + x_3 - \min\{x_1, x_2, x_3\})^{-s}$, s > 0, in Corollary 1.4, we obtain the corresponding Hilbert-type inequalities with the best possible constant factor

$$L_{3} = k \left(\frac{s - p_{2}}{p_{2}}, \frac{s - p_{3}}{p_{3}} \right) = \frac{p_{2} + p_{3}}{s} F \left(s, \frac{s}{p_{2}} + \frac{s}{p_{3}}; \frac{s}{p_{2}} + \frac{s}{p_{3}} + 1; -1 \right)$$

+ $\frac{1}{s} \sum_{i=2}^{3} p_{i} F \left(s, \frac{s}{p_{1}} + \frac{s}{p_{i}}; \frac{s}{p_{1}} + \frac{s}{p_{i}} + 1; -1 \right)$
+ $\frac{p_{1}(p_{2} + p_{3})}{s^{2}} F \left(s, \frac{s}{p_{1}}; \frac{s}{p_{1}} + 1; -1 \right)$
- $\frac{p_{2}p_{3}}{s^{2}} \sum_{i=2}^{3} \frac{1}{p_{i} - 1} F \left(s, s - \frac{s}{p_{i}}; s - \frac{s}{p_{i}} + 1; -1 \right).$

In particular, if s = 1, $p_1 = p_2 = p_3 = 3$, we have $L_3 = \frac{10\pi\sqrt{3}}{3} + \log 4$.

1.6.4 Inequalities with product-type homogeneous kernels and Schur polynomials

In this subsection we study some particular product-type homogeneous kernels and investigate associated Hilbert-type inequalities. In some cases, the corresponding constant factors can be expressed in terms of Shur polynomials.

At the beginning, we shall be concerned with the homogeneous function $K : \mathbb{R}^n_+ \to \mathbb{R}$, defined by $K(x_1, \ldots, x_n) = \prod_{i=1}^{n-1} (x_1 + a_i^2 x_{i+1})$, where $a_i > 0$, $i = 1, \ldots, n-1$. Clearly, kernel *K* has degree of homogeneity equal to -(n-1), and, in addition, fulfills condition (1.82) from Theorem 1.20. In such a way, we obtain the corresponding Hilbert-type inequalities with the best possible constant factors.

Corollary 1.7 Let $\sum_{i=1}^{n} \frac{1}{p_i} = 1$, $p_i > 1$, i = 1, 2, ..., n, and let $\frac{1}{q} = \sum_{i=1}^{n-1} \frac{1}{p_i}$. If $a_i > 0$, i = 1, ..., n-1, and $f_i : \mathbb{R}_+ \to \mathbb{R}$, $f_i \neq 0$, i = 1, ..., n, are non-negative measurable functions, then inequalities

$$\int_{\mathbb{R}^{n}_{+}} \frac{\prod_{i=1}^{n} f_{i}(x_{i})}{\prod_{i=1}^{n-1} (x_{1} + a_{i}^{2} x_{i+1})} dx_{1} \dots dx_{n} < M_{1} \prod_{i=1}^{n} \left[\int_{0}^{\infty} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$
(1.107)

and

$$\int_{0}^{\infty} \left[\int_{\mathbb{R}^{n-1}_{+}} \frac{\prod_{i=1}^{n-1} f_{i}(x_{i})}{\prod_{i=1}^{n-1} (x_{1} + a_{i}^{2} x_{i+1})} dx_{1} \dots dx_{n-1} \right]^{q} dx_{n}$$

$$< M_{1}^{q} \prod_{i=1}^{n-1} \left[\int_{0}^{\infty} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{q}{p_{i}}}$$
(1.108)

hold and are equivalent, where

$$M_1 = \pi^{n-1} \prod_{i=2}^n \frac{a_{i-1}^{2(1-p_i)/p_i}}{\sin(\pi/p_i)}.$$
(1.109)

Moreover, constant factors M_1 and M_1^q are the best possible in inequalities (1.107) and (1.108).

Proof. Considering parameters $A_{ii} = \frac{p_i - 1}{p_i^2}$ and $A_{ij} = -\frac{1}{p_i p_j}$, $i \neq j, i, j = 1, ..., n$, we have

$$\sum_{i=1}^{n} A_{ij} = -\sum_{i \neq j} \frac{1}{p_i p_j} + \frac{p_j - 1}{p_j^2} = \frac{1}{p_j} \left(-\sum_{i=1}^{n} \frac{1}{p_i} + 1 \right) = 0,$$

for j = 1, 2, ..., n, that is, $\alpha_i = \sum_{j=1}^n A_{ij} = 0, i = 1, ..., n$, due to the symmetry. In addition, these parameters satisfy the set of conditions (1.81), which is necessary in obtaining the best possible constant factors.

Further, utilizing inequalities (1.97), (1.98), and the above parameters, we have -1 - 1 $p_i \widetilde{A}_i = 0, i = 1, \dots, n$, and $(1-q)(-1-p_n \widetilde{A}_n) = 0$. Hence, it is enough to calculate the constant $M_1 = k\left(-\frac{1}{p_2}, \dots, -\frac{1}{p_n}\right)$ in the case when $K(x_1, \dots, x_n) = \prod_{i=1}^{n-1} (x_1 + a_i^2 x_{i+1})$. Exploiting (1.76), we have

$$M_{1} = k\left(-\frac{1}{p_{2}}, \dots, -\frac{1}{p_{n}}\right) = \int_{\mathbb{R}^{n-1}_{+}} \frac{t_{1}^{-\frac{1}{p_{2}}} \dots t_{n-1}^{-\frac{1}{p_{n}}}}{\prod_{i=1}^{n-1} (1+a_{i}^{2}t_{i})} dt_{1} \dots dt_{n-1}$$
$$= \prod_{i=1}^{n-1} \left(\int_{0}^{\infty} \frac{t_{i}^{-\frac{1}{p_{i+1}}}}{1+a_{i}^{2}t_{i}} dt_{i}\right) = \pi^{n-1} \prod_{i=2}^{n} \frac{a_{i-1}^{2(1-p_{i})/p_{i}}}{\sin(\pi/p_{i})},$$

and the proof is completed.

The rest of this subsection is dedicated to the two-dimensional kernel $K(x,y) = \prod_{i=1}^{m} (x_i, y_i)$ $(a_i^2 y)^{-1}$, where m is an integer and $a_i > 0$, i = 1, ..., m, are real parameters. Observe that the function K(x,y) fulfills condition (1.82) from Theorem 1.20. In addition, we shall derive here two-dimensional Hilbert-type inequalities whose constant factors can be expressed in terms of Shur polynomials.

In order to establish the corresponding inequalities, we need the following auxiliary result. For the reader's convenience, let $f[x_1, \ldots, x_n]$ denote the well-known divided difference defined by

$$f[x_1,...,x_n] = \sum_{i=1}^n \frac{f(x_i)}{\prod_{j \neq i} (x_i - x_j)}.$$

Lemma 1.6 *Let m be an integer. If* $-1 < \alpha < m - 1$, $\alpha \neq 0, 1, ..., m - 2$, *and* $a_i > 0$, i = 1, ..., m, then

$$\int_0^\infty \frac{x^\alpha dx}{\prod_{i=1}^m (x+a_i^2)} = (-1)^m \frac{\pi}{\sin(\alpha\pi)} f[a_1^2, \dots, a_m^2], \quad f(x) = x^\alpha.$$
(1.110)

For k = 0, 1, ..., m - 2*, we have*

$$\int_0^\infty \frac{x^k dx}{\prod_{i=1}^m (x+a_i^2)} = (-1)^{m+k} f_k[a_1^2, \dots, a_m^2], \quad f_k(x) = x^k \log x.$$
(1.111)

Proof. The proof is based on mathematical induction. First, for m = 1, using the substitution $t = \frac{x}{a_1^2}$, we obtain

$$\int_{0}^{\infty} \frac{x^{\alpha} dx}{x + a_{1}^{2}} = a_{1}^{2\alpha} \int_{0}^{\infty} t^{\alpha} (1 + t)^{-1} dt, \qquad (1.112)$$

where $-1 < \alpha < 0$. Further, by using the substitution $x = \frac{1}{1+t}$, relation (1.112) and the definition of the usual Beta function, we have

$$\int_{0}^{\infty} \frac{x^{\alpha} dx}{x + a_{1}^{2}} = a_{1}^{2\alpha} \int_{0}^{1} (1 - x)^{\alpha} x^{-\alpha - 1} dx$$
$$= a_{1}^{2\alpha} B(1 + \alpha, -\alpha) = a_{1}^{2\alpha} \frac{\pi}{\sin(-\alpha\pi)}.$$
(1.113)

Starting from the following induction hypothesis

$$\int_0^\infty \frac{x^\alpha dx}{\prod_{i=1}^{m-1} (x+a_i^2)} = (-1)^{m-1} \frac{\pi}{\sin(\alpha\pi)} f[a_1^2, \dots, a_{m-1}^2], \quad f(x) = x^\alpha, \tag{1.114}$$

where $-1 < \alpha < m-2$, $\alpha \neq 0, 1, \dots, m-3$, we shall calculate the integral

$$I_m = \int_0^\infty \frac{x^{\alpha} dx}{\prod_{i=1}^m (x + a_i^2)}, \quad -1 < \alpha < m - 1, \quad \alpha \neq 0, 1, \dots, m - 2.$$

We treat three cases. If $-1 < \alpha < 0$, we use (1.113) and find that

$$I_{m} = \sum_{i=1}^{m} \frac{1}{\prod_{j \neq i} (a_{j}^{2} - a_{i}^{2})} \int_{0}^{\infty} \frac{x^{\alpha} dx}{x + a_{1}^{2}}$$

$$= \sum_{i=1}^{m} \frac{1}{\prod_{j \neq i} (a_{j}^{2} - a_{i}^{2})} a_{i}^{2\alpha} \frac{\pi}{\sin(-\alpha\pi)}$$

$$= (-1)^{m} \frac{\pi}{\sin(\alpha\pi)} f[a_{1}^{2}, \dots, a_{m}^{2}], f(x) = x^{\alpha}.$$
(1.115)

If $m-2 < \alpha < m-1$, we use the substitution $x = \frac{1}{t}$ and (1.115). More precisely, we have

$$I_m = \int_0^\infty \frac{t^{m-\alpha-2}}{\prod_{i=1}^m (\frac{1}{a_i^2} + t)} dt$$

= $\frac{1}{\prod_{i=1}^m a_i^2} (-1)^m \frac{\pi}{\sin[(m-\alpha-2)\pi]} g[1/a_1^2, \dots, 1/a_m^2],$ (1.116)

where $g(x) = x^{m-\alpha-2}$. It is easy to check that the following relation is valid:

$$g[1/a_1^2, \dots, 1/a_m^2] = \sum_{i=1}^m \frac{\left(\prod_{j=1}^m a_j^2\right) a_i^{2\alpha}}{\prod_{j \neq i} (a_j^2 - a_i^2)}.$$
(1.117)

Now, relations (1.116), (1.117), and the formula $\sin[(m - \alpha - 2)\pi] = (-1)^{m-1} \sin(\alpha \pi)$ imply (1.110).

On the other hand, if $0 < \alpha < m - 2$, we again use the induction hypothesis and obtain

$$I_m = \frac{1}{a_m^2 - a_1^2} \left[\int_0^\infty \frac{x^\alpha dx}{\prod_{i=1}^{m-1} (x + a_i^2)} - \int_0^\infty \frac{x^\alpha dx}{\prod_{i=2}^m (x + a_i^2)} \right]$$
$$= (-1)^m \frac{\pi}{\sin(\alpha \pi)} \frac{f[a_2^2, \dots, a_m^2] - f[a_1^2, \dots, a_{m-1}^2]}{a_m^2 - a_1^2}$$
$$= (-1)^m \frac{\pi}{\sin(\alpha \pi)} f[a_1^2, \dots, a_m^2], \quad f(x) = x^\alpha.$$

At the end of the proof, it is necessary to consider the critical cases of the integrals

$$\int_0^\infty \frac{x^k dx}{\prod_{i=1}^m (x+a_i^2)}, \quad k = 0, 1, \dots, m-2.$$

Applying the classical calculus, we have

$$\int_{0}^{\infty} \frac{x^{k} dx}{\prod_{i=1}^{m} (x+a_{i}^{2})} = \lim_{\alpha \to k} \int_{0}^{\infty} \frac{x^{\alpha} dx}{\prod_{i=1}^{m} (x+a_{i}^{2})}$$

= $(-1)^{m} \pi \lim_{\alpha \to k} \frac{1}{\sin(\alpha \pi)} \sum_{i=1}^{m} \frac{a_{i}^{2\alpha}}{\prod_{j \neq i} (a_{i}^{2}-a_{j}^{2})}$
= $\frac{(-1)^{m}}{\cos(k\pi)} \sum_{i=1}^{m} \frac{a_{i}^{2k} \log a_{i}^{2}}{\prod_{j \neq i} (a_{i}^{2}-a_{j}^{2})} = (-1)^{m+k} f_{k}[a_{1}^{2}, \dots, a_{m}^{2}],$

where $f_k(x) = x^k \log x$. That completes the proof.

The previous lemma enables us to derive the following two-dimensional Hilbert-type inequalities with the kernel $K(x,y) = \prod_{i=1}^{m} (x + a_i^2 y)^{-1}$, $a_i > 0$, i = 1, ..., m.

Corollary 1.8 Let *m* be an integer, let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let α be real parameter such that $-1 < \alpha < m - 1$. Further, let

$$L_2 = (-1)^m \frac{\pi}{\sin(\alpha \pi)} f[a_1^2, \dots, a_m^2], \quad f(x) = x^{\alpha},$$

for $\alpha \neq 0, 1, \ldots, m-2$, and

$$L_2 = (-1)^{m+\alpha} f_{\alpha}[a_1^2, \dots, a_m^2], \quad f_{\alpha}(x) = x^{\alpha} \log x,$$

for $\alpha = 0, 1, ..., m - 2$. If $a_i > 0$, i = 1, ..., m, and $f_1, f_2 : \mathbb{R}_+ \to \mathbb{R}$, $f_1, f_2 \neq 0$, are non-negative measurable functions, then the inequalities

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{f_{1}(x)f_{2}(y)}{(x+a_{1}^{2}y)\dots(x+a_{m}^{2}y)} dxdy$$

$$< L_{2} \left[\int_{0}^{\infty} x^{-1-\alpha p} f_{1}^{p}(x)dx \right]^{\frac{1}{p}} \left[\int_{0}^{\infty} y^{-1-(m-\alpha-2)q} f_{2}^{q}(y)dy \right]^{\frac{1}{q}}$$
(1.118)

$$\int_{0}^{\infty} y^{(1-p)(-1-(m-\alpha-2)q)} \left[\int_{0}^{\infty} \frac{f_1(x)}{(x+a_1^2y)\dots(x+a_m^2y)} dx \right]^p dy$$

$$< L_2^p \int_{0}^{\infty} x^{-1-\alpha p} f_1^p(x) dx$$
(1.119)

hold and are equivalent. Moreover, the constant factors L_2 and L_2^p are the best possible in (1.118) and (1.119).

Proof. Exploiting inequalities (1.97) and (1.98), for n = 2, with the kernel $K(x, y) = \prod_{i=1}^{m} (x + a_i^2 y)^{-1}$ and parameters $p_1 = p$, $p_2 = q$, $\widetilde{A}_1 = \alpha$, $\widetilde{A}_2 = m - \alpha - 2$, we see that it is enough to calculate the constant $L_2 = k(\widetilde{A}_2)$. Utilizing substitution $u = \frac{1}{x}$, we have

$$L_2 = k(m - \alpha - 2) = \int_0^\infty \frac{u^{m - \alpha - 2} du}{\prod_{i=1}^m (1 + a_i^2 u)} = \int_0^\infty \frac{x^{\alpha} dx}{\prod_{i=1}^m (x + a_i^2)},$$

and the result follows from Lemma 1.6.

A class of Hilbert-type inequalities derived in the previous corollary involves the best possible constant factor

$$L_2 = \int_0^\infty \frac{x^{\alpha} \, dx}{\prod_{i=1}^m \left(x + a_i^2\right)} = (-1)^m \frac{\pi}{\sin\left(\alpha\pi\right)} f\left[a_1^2, \dots, a_m^2\right],\tag{1.120}$$

where $f(x) = x^{\alpha}$, $-1 < \alpha < m-1$. In the sequel we study some other forms of expression (1.120), which will bring us to Shur polynomials.

It is well-known that the divided difference can be expressed via determinants:

$$f\left[a_{1}^{2},\ldots,a_{m}^{2}\right] = \frac{\det \begin{bmatrix} f\left(a_{1}^{2}\right) & f\left(a_{2}^{2}\right) & \cdots & f\left(a_{m}^{2}\right) \\ a_{1}^{2(m-2)} & a_{2}^{2(m-2)} & \cdots & a_{m}^{2(m-2)} \\ \vdots & \vdots & \vdots & \vdots \\ a_{1}^{2} & a_{2}^{2} & \cdots & a_{m}^{2} \\ 1 & 1 & \cdots & 1 \end{bmatrix}}{\det \begin{bmatrix} a_{1}^{2(m-1)} & a_{2}^{2(m-1)} & \cdots & a_{m}^{2(m-1)} \\ a_{1}^{2(m-2)} & a_{2}^{2(m-2)} & \cdots & a_{m}^{2(m-2)} \\ \vdots & \vdots & \vdots & \vdots \\ a_{1}^{2} & a_{2}^{2} & \cdots & a_{m}^{2} \\ 1 & 1 & \cdots & 1 \end{bmatrix}}.$$

$$(1.121)$$

and

Calculating the Vandermonde determinant, we have

$$\det \begin{bmatrix} a_1^{2(m-1)} & a_2^{2(m-1)} & \cdots & a_m^{2(m-1)} \\ a_1^{2(m-2)} & a_2^{2(m-2)} & \cdots & a_m^{2(m-2)} \\ \vdots & \vdots & \vdots & \vdots \\ a_1^2 & a_2^2 & \cdots & a_m^2 \\ 1 & 1 & \cdots & 1 \end{bmatrix} = \det \begin{bmatrix} a_1^{m-1} & a_2^{m-1} & \cdots & a_m^{m-1} \\ a_1^{m-2} & a_2^{m-2} & \cdots & a_m^{m-2} \\ \vdots & \vdots & \vdots & \vdots \\ a_1 & a_2 & \cdots & a_m \\ 1 & 1 & \cdots & 1 \end{bmatrix} \cdot \prod_{1 \le i < j \le m} (a_i + a_j).$$
(1.122)

Setting m = 3 and $\alpha = \frac{1}{2}$, we have (see also [133]):

1

$$\int_0^\infty \frac{x^{\frac{1}{2}} \, dx}{\left(x + a_1^2\right) \left(x + a_2^2\right) \left(x + a_3^2\right)} = \frac{\pi}{\left(a_1 + a_2\right) \left(a_1 + a_3\right) \left(a_2 + a_3\right)}$$

Similarly, for m = 4 and $\alpha = \frac{1}{2}$, by calculating the integral in (1.120) or resolving the above determinants, we have

$$\int_0^\infty \frac{x^{\frac{1}{2}} \, dx}{\prod_{i=1}^4 \left(x + a_i^2 \right)} = \frac{\sum_{i=1}^4 a_i}{\prod_{1 \le i < j \le 4} \left(a_i + a_j \right)} \pi.$$

Suppose that $\alpha = \frac{2l-3}{2}$, l = 2, ..., m (for l = 1 see below). The numerator of (1.121) is for l < m equal to

$$\det \begin{bmatrix} a_{1}^{2l-3} & a_{2}^{2l-3} & \cdots & a_{m}^{2l-3} \\ a_{1}^{2(m-2)} & a_{2}^{2(m-2)} & \cdots & a_{m}^{2(m-2)} \\ \vdots & \vdots & \vdots & \vdots \\ a_{1}^{2} & a_{2}^{2} & \cdots & a_{m}^{2} \\ 1 & 1 & \cdots & 1 \end{bmatrix} = (-1)^{m-l} \det \begin{bmatrix} a_{1}^{2(m-2)} & a_{2}^{2(m-2)} & \cdots & a_{m}^{2(m-2)} \\ \vdots & \vdots & \vdots & \vdots \\ a_{1}^{2(l-1)} & a_{2}^{2(l-1)} & \cdots & a_{m}^{2(l-1)} \\ a_{1}^{2l-3} & a_{2}^{2l-3} & \cdots & a_{m}^{2(l-3)} \\ a_{1}^{2(l-2)} & a_{2}^{2(l-2)} & \cdots & a_{m}^{2(l-2)} \\ \vdots & \vdots & \vdots & \vdots \\ a_{1}^{2} & a_{2}^{2} & \cdots & a_{m}^{2} \\ 1 & 1 & \cdots & 1 \end{bmatrix} .$$

$$(1.123)$$

Using $\sin \frac{2l-3}{2}\pi = (-1)^l$, we have

$$L_{2} = \int_{0}^{\infty} \frac{x^{\frac{2l-3}{2}} dx}{\prod_{i=1}^{m} (x+a_{i}^{2})} = \frac{\left[\begin{array}{cccc} a_{1}^{2(m-2)} & a_{2}^{2(m-2)} & \cdots & a_{m}^{2(m-2)} \\ \vdots & \vdots & \vdots & \vdots \\ a_{1}^{2(l-1)} & a_{2}^{2(l-1)} & \cdots & a_{m}^{2(l-1)} \\ a_{1}^{2l-3} & a_{2}^{2l-3} & \cdots & a_{m}^{2(l-3)} \\ a_{1}^{2(l-2)} & a_{2}^{2(l-2)} & \cdots & a_{m}^{2(l-2)} \\ \vdots & \vdots & \vdots & \vdots \\ a_{1}^{2} & a_{2}^{2} & \cdots & a_{m}^{2} \\ 1 & 1 & \cdots & 1 \\ \nabla (a_{1}, a_{2}, \dots, a_{m}) \cdot \prod_{1 \leq i < j \leq m} (a_{i} + a_{j}) \pi, \end{array} \right]$$
(1.124)

where $V(a_1, a_2, ..., a_m)$ denotes the Vandermonde determinant.

Recall that the Schur polynomial of a given integer partition $d = d_1 + d_2 + \cdots + d_m$, $d_1 \ge d_2 \ge \cdots \ge d_m \ge 0$, is defined by

$$S_{(d_1,d_2,\dots,d_m)}(x_1,x_2,\dots,x_m) = \frac{\det \begin{bmatrix} x_1^{d_1+m-1} & x_2^{d_1+m-1} & \cdots & x_m^{d_1+m-1} \\ x_1^{d_2+m-2} & x_2^{d_2+m-2} & \cdots & x_m^{d_2+m-2} \\ \vdots & \vdots & \vdots & \vdots \\ x_1^{d_m} & x_2^{d_m} & \cdots & x_m^{d_m} \end{bmatrix}}{\det \begin{bmatrix} x_1^{m-1} & x_2^{d_m} & \cdots & x_m^{m-1} \\ x_1^{m-2} & x_2^{m-2} & \cdots & x_m^{m-2} \\ \vdots & \vdots & \vdots & \vdots \\ 1 & 1 & \cdots & 1 \end{bmatrix}}.$$
 (1.125)

The Schur polynomial can be expanded as a sum of monomials

$$S_{(d_1,d_2,\dots,d_m)}(x_1,x_2,\dots,x_m) = \sum_T x_1^{t_1} \cdots x_m^{t_m},$$
(1.126)

where the summation is over all semi-standard Young tableaux T of shape $(d_1, d_2, ..., d_m)$. The exponents $t_1, ..., t_m$ give the weight of T, in which each t_i counts the occurrences of the number i in T (see [3] and [89]).

It is obvious from (1.121) (the case l = 2m - 3), (1.124) and (1.125) that

$$\int_{0}^{\infty} \frac{x^{\frac{2m-3}{2}} dx}{\prod_{i=1}^{m} (x+a_{i}^{2})} = \frac{S_{(m-2,m-3,\dots,1,0)} (a_{1},a_{2},\dots,a_{m})}{\prod_{i

$$\int_{0}^{\infty} \frac{x^{\frac{2l-3}{2}} dx}{\prod_{i=1}^{m} (x+a_{i}^{2})} = \frac{S_{(m-3,m-4,\dots,l-2,l-2,l-2,\dots,1,0)} (a_{1},a_{2},\dots,a_{m})}{\prod_{i

$$\int_{0}^{\infty} \frac{x^{\frac{1}{2}} dx}{\prod_{i=1}^{m} (x+a_{i}^{2})} = \frac{S_{(m-3,m-4,\dots,l,0,0)} (a_{1},a_{2},\dots,a_{m})}{\prod_{i$$$$$$

where $d = \frac{(m-3)(m-2)}{2} + 2(l-2), l = 2, ..., m$, is degree of the obtained Schur polynomials. To illustrate the above results we provide the following examples:

(i) For m = 2 the only admissible case is $\alpha = \frac{1}{2}$. It follows:

$$\int_0^\infty \frac{x^{\frac{1}{2}} \, dx}{\prod_{i=1}^2 \left(x + a_i^2\right)} = \frac{S_{(0,0)}\left(a_1, a_2\right)}{a_1 + a_2} \pi = \frac{\pi}{a_1 + a_2}.$$

(ii) For m = 3, the admissible cases are $\alpha = \frac{1}{2}$ and $\alpha = \frac{3}{2}$, that is, we have

$$\int_0^\infty \frac{x^{\frac{1}{2}} dx}{\prod_{i=1}^3 (x+a_i^2)} = \frac{S_{(0,0,0)}(a_1, a_2, a_3)}{\prod_{i
$$\int_0^\infty \frac{x^{\frac{3}{2}} dx}{\prod_{i=1}^3 (x+a_i^2)} = \frac{S_{(1,1,0)}(a_1, a_2, a_3)}{\prod_{i$$$$

(iii) For m = 4, the admissible cases are $\alpha = \frac{1}{2}$, $\alpha = \frac{3}{2}$, and $\alpha = \frac{5}{2}$, that is,

$$\begin{split} \int_{0}^{\infty} \frac{x^{\frac{1}{2}} dx}{\prod_{i=1}^{4} \left(x + a_{i}^{2}\right)} &= \frac{S_{(1,0,0,0)}\left(a_{1}, a_{2}, a_{3}, a_{4}\right)}{\prod_{i < j}\left(a_{i} + a_{j}\right)} \pi = \frac{\sum_{i=1}^{4} a_{i}}{\prod_{i < j}\left(a_{i} + a_{j}\right)} \pi, \\ \int_{0}^{\infty} \frac{x^{\frac{3}{2}} dx}{\prod_{i=1}^{4} \left(x + a_{i}^{2}\right)} &= \frac{S_{(1,1,1,0)}\left(a_{1}, a_{2}, a_{3}, a_{4}\right)}{\prod_{i < j}\left(a_{i} + a_{j}\right)} \pi = \frac{\sum_{i < j < k} a_{i} a_{j} a_{k}}{\prod_{i < j}\left(a_{i} + a_{j}\right)} \pi, \\ \int_{0}^{\infty} \frac{x^{\frac{5}{2}} dx}{\prod_{i=1}^{4} \left(x + a_{i}^{2}\right)} &= \frac{S_{(2,2,1,0)}\left(a_{1}, a_{2}, a_{3}, a_{4}\right)}{\prod_{i < j}\left(a_{i} + a_{j}\right)} \pi \\ &= \frac{\sum_{i < j < k} a_{i}^{2} a_{j}^{2} a_{k} + a_{i}^{2} a_{j} a_{k}^{2} + a_{i} a_{j}^{2} a_{k}^{2} + 2\sum_{i < j < k < l} a_{i}^{2} a_{j} a_{k} a_{l}}{\prod_{i < j}\left(a_{i} + a_{j}\right)} \pi \\ &+ \frac{a_{i} a_{j}^{2} a_{k} a_{l} + a_{i} a_{j} a_{k}^{2} a_{l} + a_{i} a_{j} a_{k}^{2} a_{l} + a_{i} a_{j} a_{k}^{2} a_{l}}{\prod_{i < j}\left(a_{i} + a_{j}\right)} \pi. \end{split}$$

The above method doesn't work in the case $\alpha = -\frac{1}{2}$. We proceed as follows. By virtue of the substitution $x = \frac{1}{t}$, using the case l = 2m - 3 and the basic properties of determinants, we have:

$$\begin{split} &\int_{0}^{\infty} \frac{x^{-\frac{1}{2}} dx}{\prod_{i=1}^{m} (x+a_{i}^{2})} \\ &= \frac{1}{\prod_{i=1}^{m} a_{i}^{2}} \int_{0}^{\infty} \frac{t^{\frac{2m-3}{2}} dt}{\prod_{i=1}^{m} \left(t+\frac{1}{a_{i}^{2}}\right)} = \frac{1}{\prod_{i=1}^{m} a_{i}^{2}} \frac{S_{(m-2,m-2,m-3,\dots,1,0)}\left(\frac{1}{a_{1}},\dots,\frac{1}{a_{m}}\right)}{\prod_{i$$

Remark 1.27 A general form of multidimensional Hilbert-type inequality (Subsection 1.6.1) is established in [156], while the application to homogeneous kernels and the existence of the best possible constant factors (Subsection 1.6.2) is derived in [111]. Most of the examples in Subsection 1.6.3 are taken from [113], while inequalities including Shur polynomials can be found in [112]. For some related results, the reader can also consult the following papers: [19], [23], [37], [52] [128], [152], and [156].

Chapter 2

Hilbert-type inequalities with non-conjugate exponents

The previous chapter was dedicated to a unified treatment of Hilbert-type inequalities and numerous applications of general results. All these results included conjugate exponents.

The question is whether it is possible to establish the corresponding inequalities related to Hilbert-type inequalities where the exponents are not conjugate. The answer to that question appeared to be true. This problem was dealt by some famous mathematicians, such as F. F. Bonsall, G. H. Hardy, V. Levin, J. Littlewood, G. Pólya, in the first half of the twentieth century, an later, by E. K. Godunova. This bring us to the concept of non-conjugate parameters.

Suppose that p and q are real parameters, such that

$$p > 1, q > 1, \frac{1}{p} + \frac{1}{q} \ge 1,$$
 (2.1)

and let $p' = \frac{p}{p-1}$ and $q' = \frac{q}{q-1}$ respectively be their conjugate exponents, that is, $\frac{1}{p} + \frac{1}{p'} = 1$ and $\frac{1}{q} + \frac{1}{q'} = 1$. Further, define

$$\lambda = \frac{1}{p'} + \frac{1}{q'} \tag{2.2}$$

and observe that $0 < \lambda \le 1$ holds for all p and q as in (2.1). In particular, equality $\lambda = 1$ holds in (2.2) if and only if q = p', that is, only if p and q are mutually conjugate. Otherwise, we have $0 < \lambda < 1$, and such parameters p and q will be referred to as non-conjugate exponents.

Considering p, q, and λ as in (2.1) and (2.2), Hardy, Littlewood, and Pólya [33], proved that there exists a constant $C_{p,q}$, dependent only on the parameters p and q, such that the following Hilbert-type inequality holds for all non-negative functions $f \in L^p(\mathbb{R}_+)$ and $g \in L^q(\mathbb{R}_+)$:

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{f(x)g(y)}{(x+y)^{\lambda}} dx dy \le C_{p,q} \|f\|_{L^{p}(\mathbb{R}_{+})} \|g\|_{L^{q}(\mathbb{R}_{+})}.$$
(2.3)

However, the original proof did not bring any information about the value of the best possible constant $C_{p,q}$. That drawback was improved by Levin [79], who obtained an explicit upper bound for $C_{p,q}$,

$$C_{p,q} \le \left(\pi \operatorname{cosec} \frac{\pi}{\lambda p'}\right)^{\lambda}.$$
(2.4)

This was an interesting result, since the right-hand side of (2.4) reduces to the previously known sharp constant $\pi \operatorname{cosec}(\pi/p')$ when the exponents p and q are conjugate (see Theorem 1.2, Chapter 1). A simpler proof of (2.4), based on a single application of the Hölder inequality, was given later by F. F. Bonsall [9].

In spite of its trivial appearance, Bonsall's idea was useful for investigating the inequalities for multiple integrals involving non-conjugate parameters. So, he obtained the following inequality for n = 3: Let $\frac{1}{p_1} + \frac{1}{p_2} + \frac{1}{p_3} \ge 1$ with $p_i > 1$, $\frac{1}{p_i} + \frac{1}{p'_i} = 1$, i = 1, 2, 3, and $\lambda = \frac{1}{2} \left(\frac{1}{p'_1} + \frac{1}{p'_2} + \frac{1}{p'_3} \right)$. Then,

$$\int_{0}^{\infty} \int_{0}^{\infty} \int_{0}^{\infty} \frac{f(x)g(y)h(z)}{(x+y+z)^{2\lambda}} dx dy dz \le k \|f\|_{L^{p_{1}}(\mathbb{R}_{+})} \|g\|_{L^{p_{2}}(\mathbb{R}_{+})} \|h\|_{L^{p_{3}}(\mathbb{R}_{+})},$$
(2.5)

with an explicit upper bound expressed in terms of the usual Gamma function:

$$k \leq \left[\Gamma\left(\frac{1}{\lambda p_1'}\right)\Gamma\left(\frac{1}{\lambda p_2'}\right)\Gamma\left(\frac{1}{\lambda p_3'}\right)\right]^{\lambda}.$$

Although Bonsall established the concept of n non-conjugate parameters, there were no results in that direction.

Moreover, in the same paper, with p, q and λ as in (2.1) and (2.2), Bonsall proved another interesting Hilbert-type inequality,

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{f(x)g(y)}{(x+y)^{\lambda}} dx dy \leq B^{\lambda} \left(\frac{1}{p'}, \frac{1}{q'}\right) \|f\|_{L^{p}(\mathbb{R}_{+})}^{\frac{p}{q'}} \|g\|_{L^{q}(\mathbb{R}_{+})}^{\frac{q}{p'}} \\ \times \left[\int_{0}^{\infty} \int_{0}^{\infty} \frac{x^{\frac{1}{p'}} y^{\frac{1}{q'}}}{(x+y)^{\lambda}} f^{p}(x) g^{q}(y) dx dy\right]^{1-\lambda},$$
(2.6)

with the best possible constant factor $B^{\lambda}(\frac{1}{p'}, \frac{1}{q'})$, expressed in terms of the usual Beta function.

Remark 2.1 In the above relations, $L^p(\mathbb{R}_+)$ denotes the Lebesgue space consisting of all measurable functions $f : \mathbb{R}_+ \to \mathbb{R}$ with a finite norm $||f||_{L^p(\mathbb{R}_+)} = [\int_0^\infty |f(x)|^p dx]^{\frac{1}{p}}$. More generally, when considering a measure space Ω with a positive σ -finite measure μ , the corresponding Lebesgue space will be denoted by $L^p(\mu)$, or simply L^p .

Although inequality (2.6) involves the best possible constant factor, there is still no evidence that the constant factors in inequalities (2.3) and (2.5) are the best possible. This problem seems to be very hard and remains still open. Hence, the problem of the best possible constant factors will not be considered in this chapter.

The main objective of this chapter is to extend Hilbert-type inequalities with conjugate exponents to the setting of non-conjugate exponents. First we derive the general forms of Hilbert-type inequalities with non-conjugate exponents in two-dimensional, and later on, in multidimensional case. Accordingly, all results from the previous chapter can be extended to this non-conjugate setting. In addition, we are going to consider some particular non-homogeneous kernels, yielding the constant factors expressed in terms of generalized hypergeometric functions. At the end of this chapter, we study some particular operators between the weighted Lebesgue spaces, that naturally arise from the established Hardy-Hilbert-type inequalities.

2.1 General form

To provide a basis for main results, in this section we first discuss general inequalities of the Hilbert-type and Hardy-Hilbert-type with non-conjugate exponents. These equivalent relations are stated and proved in the following theorem.

Theorem 2.1 Let p, q, and λ be real parameters as in (2.1) and (2.2), and let Ω_1 and Ω_2 be measure spaces with positive σ -finite measures μ_1 and μ_2 respectively. Let K be a non-negative measurable function on $\Omega_1 \times \Omega_2$, φ a measurable, a.e. positive function on Ω_1 , and ψ a measurable, a.e. positive function on Ω_2 . If the functions F on Ω_1 and G on Ω_2 are defined by

$$F(x) = \left[\int_{\Omega_2} K(x, y) \psi^{-q'}(y) \, d\mu_2(y) \right]^{\frac{1}{q'}}, \ x \in \Omega_1,$$
(2.7)

and

$$G(y) = \left[\int_{\Omega_1} K(x, y) \varphi^{-p'}(x) \, d\mu_1(x) \right]^{\frac{1}{p'}}, \ y \in \Omega_2,$$
(2.8)

then for all non-negative measurable functions f on Ω_1 and g on Ω_2 the inequalities

$$\int_{\Omega_1} \int_{\Omega_2} K^{\lambda}(x, y) f(x) g(y) \, d\mu_1(x) d\mu_2(y) \le \|\varphi F f\|_{L^p(\mu_1)} \|\psi G g\|_{L^q(\mu_2)} \tag{2.9}$$

and

$$\left\{ \int_{\Omega_2} \left[(\psi G)^{-1}(y) \int_{\Omega_1} K^{\lambda}(x, y) f(x) \, d\mu_1(x) \right]^{q'} d\mu_2(y) \right\}^{\frac{1}{q'}} \le \|\varphi F f\|_{L^p(\mu_1)} \tag{2.10}$$

hold and are equivalent.

Proof. We prove inequality (2.9) first. Let K, φ , and ψ be as in the statement of Theorem 2.1 and let f and g be arbitrary non-negative measurable functions on Ω_1 and Ω_2 respectively. Since $\frac{1}{q'} + \frac{1}{p'} + 1 - \lambda = 1$, the left-hand side of relation (2.9) can be written as

$$\begin{split} &\int_{\Omega_1} \int_{\Omega_2} K^{\lambda}(x, y) f(x) g(y) \, d\mu_1(x) d\mu_2(y) \\ &= \int_{\Omega_1} \int_{\Omega_2} \left[K(x, y) \psi^{-q'}(y) (\varphi^p F^{p-q'} f^p)(x) \right]^{\frac{1}{q'}} \left[K(x, y) \varphi^{-p'}(x) (\psi^q G^{q-p'} g^q)(y) \right]^{\frac{1}{p'}} \\ &\times \left[(\varphi F f)^p \left(x \right) (\psi G g)^q \left(y \right) \right]^{1-\lambda} d\mu_1(x) d\mu_2(y). \end{split}$$
(2.11)

Now, utilizing the Hölder inequality, either with the parameters $q', p', \frac{1}{1-\lambda} > 1$ in the case of non-conjugate exponents p and q, or with the parameters p and p' when q' = p (that is, when $\lambda = 1$), and then applying the Fubini theorem, we obtain that the right-hand side of (2.11) is not greater than

$$\begin{split} \left\{ \int_{\Omega_{1}} \left[\int_{\Omega_{2}} K(x,y) \psi^{-q'}(y) \, d\mu_{2}(y) \right] (\varphi^{p} F^{p-q'} f^{p})(x) \, d\mu_{1}(x) \right\}^{\frac{1}{q'}} \\ & \times \left\{ \int_{\Omega_{2}} \left[\int_{\Omega_{1}} K(x,y) \varphi^{-p'}(x) \, d\mu_{1}(x) \right] (\psi^{q} G^{q-p'} g^{q})(y) \, d\mu_{2}(y) \right\}^{\frac{1}{p'}} \\ & \times \left[\int_{\Omega_{1}} (\varphi F f)^{p}(x) \, d\mu_{1}(x) \right]^{1-\lambda} \left[\int_{\Omega_{2}} (\psi G g)^{q}(y) \, d\mu_{2}(y) \right]^{1-\lambda} \\ &= \left[\int_{\Omega_{1}} (\varphi F f)^{p}(x) \, d\mu_{1}(x) \right]^{\frac{1}{q'}+1-\lambda} \left[\int_{\Omega_{2}} (\psi G g)^{q}(y) \, d\mu_{2}(y) \right]^{\frac{1}{p'}+1-\lambda} \\ &= \|\varphi F f\|_{L^{p}(\mu_{1})} \|\psi G g\|_{L^{q}(\mu_{2})}, \end{split}$$

so (2.9) is proved. The further step is to prove that (2.9) implies (2.10) to hold for all non-negative measurable functions f on Ω_1 . In particular, for any such f and the function g defined by

$$g(y) = (\psi G)^{-q'}(y) \left[\int_{\Omega_1} K^{\lambda}(x, y) f(x) d\mu_1(x) \right]^{\frac{q'}{q}}, y \in \Omega_2,$$

applying the Fubini theorem, the left-hand side of (2.9) becomes

$$\begin{split} L &= \int_{\Omega_1} \int_{\Omega_2} K^{\lambda}(x, y) f(x)(\psi G)^{-q'}(y) \left[\int_{\Omega_1} K^{\lambda}(x, y) f(x) \, d\mu_1(x) \right]^{\frac{q'}{q}} d\mu_1(x) d\mu_2(y) \\ &= \int_{\Omega_2} \left[(\psi G)^{-1}(y) \int_{\Omega_1} K^{\lambda}(x, y) f(x) \, d\mu_1(x) \right]^{q'} d\mu_2(y), \end{split}$$

that is, the integral on the left-hand side of (2.10), while on the right-hand side of (2.9) we have

$$R = \|\varphi F f\|_{L^{p}(\mu_{1})} \left\{ \int_{\Omega_{2}} (\psi G)^{q(1-q')}(y) \left[\int_{\Omega_{1}} K^{\lambda}(x,y) f(x) d\mu_{1}(x) \right]^{q'} d\mu_{2}(y) \right\}^{\frac{1}{q}}$$

= $\|\varphi F f\|_{L^{p}(\mu_{1})} L^{\frac{1}{q}}.$

Hence,

$$L \leq \|\varphi F f\|_{L^p(\mu_1)} L^{\frac{1}{q}},$$

which directly yields (2.10), so the implication (2.9) \Rightarrow (2.10) is proved. Conversely, by using the Hölder inequality for the conjugate exponents q and q', together with the relation (2.10), for arbitrary $f, g \ge 0$ we have

$$\begin{split} &\int_{\Omega_{1}} \int_{\Omega_{2}} K^{\lambda}(x,y) f(x) g(y) \, d\mu_{1}(x) d\mu_{2}(y) \\ &= \int_{\Omega_{2}} (\psi Gg)(y) \left[(\psi G)^{-1}(y) \int_{\Omega_{1}} K^{\lambda}(x,y) f(x) \, d\mu_{1}(x) \right] d\mu_{2}(y) \\ &\leq \|\psi Gg\|_{L^{q}(\mu_{2})} \left\{ \int_{\Omega_{2}} \left[(\psi G)^{-1}(y) \int_{\Omega_{1}} K^{\lambda}(x,y) f(x) \, d\mu_{1}(x) \right]^{q'} d\mu_{2}(y) \right\}^{\frac{1}{q'}} \\ &\leq \|\varphi Ff\|_{L^{p}(\mu_{1})} \|\psi Gg\|_{L^{q}(\mu_{2})}. \end{split}$$

Thus, (2.10) implies (2.9), so these inequalities are equivalent. The proof is now completed. $\hfill \Box$

Remark 2.2 The sign of inequality in (2.9) depends only on the parameters p', q', and λ , since the crucial step in proving this relation was in applying the Hölder inequality. Therefore, we can consider exponents which provide the reverse inequality in (2.9). In particular, if the parameters p and q from Theorem 2.1 are such that

$$p < 0, \ 0 < q < 1, \ \frac{1}{p} + \frac{1}{q} \le 1,$$
 (2.12)

and λ is defined by (2.2), we have 0 < p' < 1, q' < 0, and $1 - \lambda \le 0$, so the inequality in (2.9) is reversed as a direct consequence of the reversed Hölder inequality. The same result is achieved also with the parameters p and q satisfying

$$0$$

since from (2.13) we obtain p' < 0, 0 < q' < 1, and $1 - \lambda \le 0$. Moreover, by using the same arguments, parameters $p, q \in (0, 1)$ give another sufficient condition for the reverse inequality in (2.9). In that case we have p', q' < 0, and $1 - \lambda > 0$.

Remark 2.3 Equality in (2.9) holds if and only if it holds in the Hölder inequality, that is, if the functions $K\psi^{-q'}\varphi^p F^{p-q'}f^p$, $K\varphi^{-p'}\psi^q G^{q-p'}g^q$, and $(\varphi F f)^p (\psi G g)^q$ are effectively proportional on $\Omega_1 \times \Omega_2$. Of course, this trivially happens if at least one of the functions involved in the left-hand side of (2.9) is the zero-function. To discuss other non-trivial cases of equality in (2.9), we can without loss of generality assume that the functions *K*, *f*, and *g* are positive. Under such assumptions, equality in (2.9) occurs if and only if there exist positive real constants α_1 , β_1 , and γ_1 , such that the relations

$$\alpha_1 K(x, y) \psi^{-q'}(y) (\varphi^p F^{p-q'} f^p)(x) = \beta_1 K(x, y) \varphi^{-p'}(x) (\psi^q G^{q-p'} g^q)(y)$$

= $\gamma_1 (\varphi F f)^p (x) (\psi G g)^q (y)$

hold for a. e. $(x,y) \in \Omega_1 \times \Omega_2$. These equalities can be written in a more suitable form, as

$$\alpha_1(\varphi^{p+p'}F^{p-q'}f^p)(x) = \beta_1(\psi^{q+q'}G^{q-p'}g^q)(y), \text{ for a.e. } (x,y) \in \Omega_1 \times \Omega_2, \qquad (2.14)$$

and

$$\alpha_1 K(x, y) = \gamma_1 F^{q'}(x) (\psi^{q+q'} G^q g^q)(y), \text{ for a.e. } (x, y) \in \Omega_1 \times \Omega_2.$$
 (2.15)

Since the left-hand side of (2.14) depends only on $x \in \Omega_1$, while the right-hand side is a single-variable function of $y \in \Omega_2$, (2.14) holds only if

$$\varphi^{p+p'}F^{p-q'}f^p = \alpha^p = const.$$
 a.e. on Ω_1

and

$$\psi^{q+q'}G^{q-p'}g^q = \beta^p = const.$$
 a.e. on Ω_2

for some positive real constants α and β . Taking into account $1 + \frac{p'}{p} = p'$ and $1 + \frac{q'}{q} = q'$, these identities can be finally transformed to

$$f = \alpha \varphi^{-p'} F^{\frac{q'}{p}-1} \quad \text{a.e. on } \Omega_1 \quad \text{and} \quad g = \beta \psi^{-q'} G^{\frac{p'}{q}-1} \quad \text{a.e. on } \Omega_2.$$
(2.16)

Moreover, combining (2.16) with (2.15), we obtain

$$K = \gamma F^{q'} G^{p'} \quad \text{a.e. on } \Omega_1 \times \Omega_2, \tag{2.17}$$

for some positive constant γ . Therefore, we proved that the conditions (2.16) and (2.17) are necessary and sufficient for equality in (2.9). Moreover, it is clear that the equality in (2.10) holds only if it holds in (2.9).

As an example of the function K which fulfills (2.17), here we mention

$$K(x,y) = \frac{\varphi^{p'}(x)\psi^{q'}(y)}{\mu_1(\Omega_1)\mu_2(\Omega_2)}, \quad (x,y) \in \Omega_1 \times \Omega_2,$$

where the sets Ω_1 and Ω_2 are such that $\mu_1(\Omega_1), \mu_2(\Omega_2) < \infty$ and the functions φ and ψ are arbitrary, as in Theorem 2.1. In particular, in this setting we have

$$F = \mu_1(\Omega_1)^{-\frac{1}{q'}} \varphi^{\frac{p'}{q'}}$$
 and $G = \mu_2(\Omega_2)^{-\frac{1}{p'}} \psi^{\frac{q'}{p'}}$,

so *K* fulfills (2.17) with $\gamma = 1$. Equality in (2.9) is attained for $f = \alpha \varphi^{-1 - \frac{p'}{q'}}$ and $g = \beta \psi^{-1 - \frac{q'}{p'}}$, where α and β are positive constants.

Remark 2.4 Considering the case of conjugate exponents, that is, when q = p' and $\lambda = 1$, Theorem 2.1 reduces to Theorem 1.9 from Section 1.2. In other words, Theorem 2.1 is the non-conjugate extension of Theorem 1.9. This extension was established recently in [16].

2.2 The case of a homogeneous kernel

Similarly as in Section 1.3, we apply here general results from Theorem 2.1 to non-negative homogeneous functions $K : \Omega \subseteq \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ with a negative degree of homogeneity. Recall that $K : \Omega \subseteq \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ is assumed to be homogeneous of degree -s, s > 0, such that $k(\alpha) = \int_0^\infty K(1, u)u^{-\alpha}du < \infty$ for $1 - s < \alpha < 1$ (see (1.20), Section 1.3).

In this way all results from Section 1.3 can be extended to the case of non-conjugate exponents. For example, non-conjugate version of Theorem 1.11 reads as follows.

Theorem 2.2 Let p, q, and λ be as in (2.1) and (2.2), and let $K : (a,b) \times (a,b) \to \mathbb{R}$ be a non-negative homogeneous function of degree -s, s > 0, strictly decreasing in both arguments. Further, suppose that A_1 and A_2 are real parameters such that $A_1 \in (\frac{1-s}{p'}, \frac{1}{p'})$, $A_2 \in (\frac{1-s}{q'}, \frac{1}{q'})$. If the functions φ_1 and φ_2 are defined as in the statement of Theorem 1.11, then for all non-negative measurable functions f and g on (a,b) the inequalities

$$\int_{a}^{b} \int_{a}^{b} K^{\lambda}(x,y) f(x)g(y) dx dy$$

$$\leq \left[\int_{a}^{b} \left(k(q'A_{2}) - \varphi_{1}(q'A_{2},x) \right)^{\frac{p}{q'}} x^{\frac{p}{q'}(1-s)+p(A_{1}-A_{2})} f^{p}(x) dx \right]^{\frac{1}{p}}$$

$$\times \left[\int_{a}^{b} \left(k(2-s-p'A_{1}) - \varphi_{2}(2-s-p'A_{1},y) \right)^{\frac{q}{p'}} y^{\frac{q}{p'}(1-s)+q(A_{2}-A_{1})} g^{q}(y) dy \right]^{\frac{1}{q}}$$
(2.18)

and

$$\begin{bmatrix}
\int_{a}^{b} y^{\frac{q'}{p'}(s-1)+q'(A_{1}-A_{2})} \left(k(2-s-p'A_{1})-\varphi_{2}(2-s-p'A_{1},y)\right)^{-\frac{q'}{p'}} \\
\times \left(\int_{a}^{b} K^{\lambda}(x,y)f(x)dx\right)^{q'}dy \end{bmatrix}^{\frac{1}{q'}} \\
\leq \left[\int_{a}^{b} \left(k(q'A_{2})-\varphi_{1}(q'A_{2},x)\right)^{\frac{p}{q'}}x^{\frac{p}{q'}(1-s)+p(A_{1}-A_{2})}f^{p}(x)dx\right]^{\frac{1}{p}} \qquad (2.19)$$

hold and are equivalent.

 \square

Proof. In order to prove (2.18) set $\varphi(x) = x^{A_1}$ and $\psi(y) = y^{A_2}$ in general inequality (2.9). Taking into account the substitution $u = \frac{y}{x}$, we obtain

$$\begin{split} &\int_{a}^{b} \int_{a}^{b} K^{\lambda}(x,y) f(x) g(y) dx dy \\ &\leq \left[\int_{a}^{b} x^{\frac{p}{q'}(1-s) + p(A_{1}-A_{2})} \left(\int_{\frac{a}{x}}^{\frac{b}{x}} K(1,u) u^{-q'A_{2}} du \right)^{\frac{p}{q'}} f^{p}(x) dx \right]^{\frac{1}{p}} \\ &\times \left[\int_{a}^{b} y^{\frac{q}{p'}(1-s) + q(A_{2}-A_{1})} \left(\int_{\frac{y}{b}}^{\frac{y}{a}} K(1,u) u^{p'A_{1}+s-2} du \right)^{\frac{q}{p'}} g^{q}(y) dy \right]^{\frac{1}{q}}. \end{split}$$

Now, the rest of the proof follows the same lines as the proof of Theorem 1.11.

Remark 2.5 According to Remark 2.2, we discuss here conditions under which the reverse inequalities in Theorem 2.2 are fulfilled. Firstly, if 0 , <math>0 < q < 1 and K is as in the statement of Theorem 2.2, then the reverse inequalities in (2.9) and (2.10) hold. On the other hand, if the conditions (2.12) are satisfied, then the reverse inequalities in Theorem 2.2 are valid if a = 0 and K is strictly increasing in first argument and strictly decreasing in second argument, or if $b = \infty$ and K is strictly decreasing in first argument and strictly increasing in second argument. The remaining case (2.13) which also provides reverse inequalities is analyzed similarly. Observe that in the case of reversed inequalities we have to adjust the intervals for parameters A_1 and A_2 .

Of course, the most important case of Theorem 2.2 is with integrals over \mathbb{R}_+ , that is, when a = 0 and $b = \infty$. The corresponding equivalent Hilbert-type and Hardy-Hilbert-type inequalities are given in the following corollary.

Corollary 2.1 Assume that p, q, and λ are as in (2.1) and (2.2), and $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ is a non-negative homogeneous function of degree -s, s > 0. Then the inequalities

$$\int_{0}^{\infty} \int_{0}^{\infty} K^{\lambda}(x,y) f(x)g(y) dx dy$$

$$\leq L' \left[\int_{0}^{\infty} x^{\frac{p}{q'}(1-s)+p(A_1-A_2)} f^p(x) dx \right]^{\frac{1}{p}} \left[\int_{0}^{\infty} y^{\frac{q}{p'}(1-s)+q(A_2-A_1)} g^q(y) dy \right]^{\frac{1}{q}}$$
(2.20)

and

$$\left[\int_{0}^{\infty} y^{\frac{q'}{p'}(s-1)+q'(A_{1}-A_{2})} \left(\int_{0}^{\infty} K^{\lambda}(x,y)f(x)dx\right)^{q'}dy\right]^{\frac{1}{q'}} \le L' \left[\int_{0}^{\infty} x^{\frac{p}{q'}(1-s)+p(A_{1}-A_{2})}f^{p}(x)dx\right]^{\frac{1}{p}}$$
(2.21)

hold for all parameters $A_1 \in \left(\frac{1-s}{p'}, \frac{1}{p'}\right)$, $A_2 \in \left(\frac{1-s}{q'}, \frac{1}{q'}\right)$, and for all non-negative measurable functions f and g on \mathbb{R}_+ , where $L' = k^{\frac{1}{q'}}(q'A_2)k^{\frac{1}{p'}}(2-s-p'A_1)$. Moreover, these inequalities are equivalent. In addition, reverse inequalities are valid under conditions appearing in Remark 2.2.
Remark 2.6 Considering the above Corollary 2.1, equipped with the kernel $K(x,y) = (x+y)^{-s}$, s > 0, the constant on the right-hand sides of (2.20) and (2.21) is expressed in terms of the Beta function, i.e.

$$L' = B^{\frac{1}{q'}} (1 - q'A_2, s + q'A_2 - 1) B^{\frac{1}{p'}} (1 - p'A_1, s + p'A_1 - 1).$$

For the kernel $K(x, y) = \max\{x, y\}^{-s}$, s > 0, this constant reads

$$L' = \frac{s^{\lambda}}{\left(1 - q'A_2\right)^{\frac{1}{q'}} \left(1 - p'A_1\right)^{\frac{1}{p'}} \left(s + q'A_2 - 1\right)^{\frac{1}{q'}} \left(s + p'A_1 - 1\right)^{\frac{1}{p'}}},$$

while for the homogeneous kernel of degree -1, given by $K(x,y) = \frac{\log y - \log x}{y-x}$, we have

$$L' = \pi^{2\lambda} \sin^{-\frac{2}{p'}}(A_1 p') \sin^{-\frac{2}{q'}}(A_2 q').$$

Note that in the conjugate case, these three constants coincide with the constants obtained in Subsection 1.4.1 (see Chapter 1).

Remark 2.7 As we have already mentioned, all results from Section 1.3 can be extended to non-conjugate case. Here they are omitted, and for more details the reader is referred to [16]. Moreover, diverse methods presented in Section 1.4, except the theorems with the best possible constant factors, can be combined with a general method for non-conjugate Hilbert-type inequalities. Namely, the problem of determining the best possible constants in non-conjugate case is not resolved yet and still remains open.

As an illustration, we only provide non-conjugate extension of Corollary 1.2. Namely, if p, q, and λ are as in (2.1) and (2.2), and s > 0, then the non-conjugate versions of inequalities (1.60) and (1.61) respectively read as

$$\begin{split} &\int_{a}^{b} \int_{a}^{b} \frac{f(x)g(y)}{(x+y)^{s\lambda}} dx dy \\ &\leq B^{\lambda} \left(\frac{s}{2}, \frac{s}{2}\right) \left\{ \int_{a}^{b} \left[1 - \frac{1}{2} \left(\frac{a}{x}\right)^{\frac{s}{2}} - \frac{1}{2} \left(\frac{x}{b}\right)^{\frac{s}{2}} \right]^{\frac{p}{q'}} x^{-\frac{s}{2}p\lambda + p - 1} f^{p}(x) dx \right\}^{\frac{1}{p}} \\ & \times \left\{ \int_{a}^{b} \left[1 - \frac{1}{2} \left(\frac{a}{y}\right)^{\frac{s}{2}} - \frac{1}{2} \left(\frac{y}{b}\right)^{\frac{s}{2}} \right]^{\frac{q}{p'}} y^{-\frac{s}{2}q\lambda + q - 1} g^{q}(y) dy \right\}^{\frac{1}{q}} \end{split}$$

and

$$\begin{cases} \int_{a}^{b} \left[1 - \frac{1}{2} \left(\frac{a}{y} \right)^{\frac{s}{2}} - \frac{1}{2} \left(\frac{y}{b} \right)^{\frac{s}{2}} \right]^{-\frac{q'}{p'}} y^{\frac{s}{2}q'\lambda - 1} \left[\int_{a}^{b} \frac{f(x)}{(x+y)^{s\lambda}} dx \right]^{q'} dy \end{cases}^{\frac{1}{q'}} \\ \leq B^{\lambda} \left(\frac{s}{2}, \frac{s}{2} \right) \left\{ \int_{a}^{b} \left[1 - \frac{1}{2} \left(\frac{a}{x} \right)^{\frac{s}{2}} - \frac{1}{2} \left(\frac{x}{b} \right)^{\frac{s}{2}} \right]^{\frac{p}{q'}} x^{-\frac{s}{2}p\lambda + p - 1} f^{p}(x) dx \end{cases}^{\frac{1}{p}}.$$

We conclude this section with some discrete examples. Namely, Theorem 2.1, rewritten with the counting measure on \mathbb{N} , leads to some interesting Hilbert-type inequalities related to sequences of non-negative real numbers. The following results include homogeneous kernel strictly decreasing in each argument.

Theorem 2.3 Let p, q, and λ be as in (2.1) and (2.2), $A, B, \alpha, \beta > 0$, and let $n, n' \in \mathbb{N}$, n < n'. Further, suppose that $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ is a non-negative homogeneous function of degree -s, s > 0, strictly decreasing in both arguments. If the functions ζ_1 , ζ_2 are defined by

$$\zeta_{1}(\gamma, x) = \left(\frac{a}{x}\right)^{\alpha(1-\gamma)} \int_{0}^{\frac{B}{A}a^{\beta-\alpha}} K(1, u)u^{-\gamma}du + \left(\frac{x}{b}\right)^{\alpha(s+\gamma-1)} \int_{0}^{\frac{A}{B}b^{\alpha-\beta}} K(u, 1)u^{s+\gamma-2}du$$
$$\zeta_{2}(\gamma, y) = \left(\frac{a}{y}\right)^{\beta(s+\gamma-1)} \int_{0}^{\frac{A}{B}a^{\alpha-\beta}} K(u, 1)u^{s+\gamma-2}du + \left(\frac{y}{b}\right)^{\beta(1-\gamma)} \int_{0}^{\frac{B}{A}b^{\beta-\alpha}} K(1, u)u^{-\gamma}du$$

and $M' = \alpha^{-\frac{1}{p'}} \beta^{-\frac{1}{q'}} A^{\frac{1-s}{q'} + A_1 - A_2 - \frac{1}{p'}} B^{\frac{1-s}{p'} + A_2 - A_1 - \frac{1}{q'}}$, then the inequalities

$$\sum_{i=n+1}^{n'} \sum_{j=n+1}^{n'} K^{\lambda}(Ai^{\alpha}, Bj^{\beta}) a_{i} b_{j}$$

$$\leq M' \left[\sum_{i=n+1}^{n'} \left(k(q'A_{2}) - \zeta_{1}(q'A_{2}, i) \right)^{\frac{p}{q'}} i^{\frac{\alpha p}{q'}(1-s) + \alpha p(A_{1}-A_{2}) - (p-1)(\alpha-1)} a_{i}^{p} \right]^{\frac{1}{p}}$$

$$\times \left[\sum_{j=n+1}^{n'} \left(k(2-s-p'A_{1}) - \zeta_{2}(2-s-p'A_{1}, j) \right)^{\frac{q}{p'}} j^{\frac{\beta q}{p'}(1-s) + \beta q(A_{2}-A_{1}) - (q-1)(\beta-1)} b_{j}^{q} \right]^{\frac{1}{q}}$$

$$(2.22)$$

and

$$\left[\sum_{j=n+1}^{n'} j^{\beta q'(A_1-A_2)+\frac{\beta q'}{p'}(s-1)+\beta-1} \left(k(2-s-p'A_1) - \zeta_2(2-s-p'A_1,j) \right)^{-\frac{q'}{p'}} \times \left(\sum_{i=n+1}^{n'} K^{\lambda}(Ai^{\alpha},Bj^{\beta})a_i \right)^{q'} \right]^{\frac{1}{q'}}$$

$$\leq M' \left[\sum_{i=n+1}^{n'} \left(k(q'A_2) - \zeta_1(q'A_2,i) \right)^{\frac{p}{q'}} i^{\frac{\alpha p}{q'}(1-s)+\alpha p(A_1-A_2)-(p-1)(\alpha-1)} a_i^p \right]^{\frac{1}{p}}$$
(2.23)

hold for all real parameters A_1 , A_2 such that $A_1 \in \left(\max\{\frac{1-s}{p'}, \frac{\alpha-1}{\alpha p'}\}, \frac{1}{p'}\right)$, $A_2 \in \left(\max\{\frac{1-s}{q'}, \frac{\beta-1}{\beta q'}\}, \frac{1}{q'}\right)$, and for all non-negative sequences $(a_n)_{n \in \mathbb{N}}$ and $(b_n)_{n \in \mathbb{N}}$. In addition, these inequalities are equivalent.

Proof. In order to prove inequality (2.22), rewrite Theorem 2.1 for the counting measure on \mathbb{N} , $K_{ij} = K(Ai^{\alpha}, Bj^{\beta})$, $\varphi_i = (Ai^{\alpha})^{A_1 + \frac{1}{p'\alpha} - \frac{1}{p'}}$, $\psi_j = (Bj^{\beta})^{A_2 + \frac{1}{q'\beta} - \frac{1}{q'}}$, and the sequences $(a_n)_{n \in \mathbb{N}}$ and $(b_n)_{n \in \mathbb{N}}$. In this setting, inequality (2.9) becomes

$$\sum_{i=n+1}^{n'} \sum_{j=n+1}^{n'} K^{\lambda}(Ai^{\alpha}, Bj^{\beta}) a_{i}b_{j}$$

$$\leq \left[\sum_{i=n+1}^{n'} (Ai^{\alpha})^{pA_{1}+(p-1)\frac{1-\alpha}{\alpha}} F_{i}^{p} a_{i}^{p}\right]^{\frac{1}{p}} \left[\sum_{j=n+1}^{n'} (Bj^{\beta})^{qA_{2}+(q-1)\frac{1-\beta}{\beta}} G_{j}^{q} b_{j}^{q}\right]^{\frac{1}{q}},$$

where

$$F_{i} = \left[\sum_{j=n+1}^{n'} \frac{K(Ai^{\alpha}, Bj^{\beta})}{(Bj^{\beta})^{q'A_{2} + \frac{1}{\beta} - 1}}\right]^{\frac{1}{q'}} \text{ and } G_{j} = \left[\sum_{i=n+1}^{n'} \frac{K(Ai^{\alpha}, Bj^{\beta})}{(Ai^{\alpha})^{p'A_{1} + \frac{1}{\alpha} - 1}}\right]^{\frac{1}{p'}}.$$

Now, since the kernel K is strictly decreasing in both variables and $p'A_1 + \frac{1}{\alpha} - 1 \ge 0$, $q'A_2 + \frac{1}{\beta} - 1 \ge 0$, we have that

$$F_i \leq \int_n^{n'} \frac{K(Ai^{\alpha}, By^{\beta})}{(By^{\beta})^{q'A_2 + \frac{1}{\beta} - 1}} dy \quad \text{and} \quad G_j \leq \int_n^{n'} \frac{K(Ax^{\alpha}, Bj^{\beta})}{(Ax^{\alpha})^{p'A_1 + \frac{1}{\alpha} - 1}} dx,$$

so the result follows due to the homogeneity of kernel K.

An important consequence of Theorem 2.3 is the following corollary for infinite series.

Corollary 2.2 Let p, q, and λ be as in (2.1) and (2.2), $A, B, \alpha, \beta > 0$, and let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a non-negative homogeneous function of degree -s, s > 0, strictly decreasing in both variables. Then the inequalities

$$\sum_{i=1}^{\infty} \sum_{j=1}^{\infty} K^{\lambda} (Ai^{\alpha}, Bj^{\beta}) a_{i} b_{j}$$

$$\leq L'M' \left[\sum_{i=1}^{\infty} i^{\frac{\alpha p}{q'}(1-s) + \alpha p(A_{1}-A_{2}) - (p-1)(\alpha-1)} a_{i}^{p} \right]^{\frac{1}{p}}$$

$$\times \left[\sum_{j=1}^{\infty} j^{\frac{\beta q}{p'}(1-s) + \beta q(A_{2}-A_{1}) - (q-1)(\beta-1)} b_{j}^{q} \right]^{\frac{1}{q}}$$
(2.24)

and

$$\left[\sum_{j=1}^{\infty} j^{\beta q'(A_1-A_2)+\frac{\beta q'}{p'}(s-1)+\beta-1} \left(\sum_{i=1}^{\infty} K^{\lambda}(Ai^{\alpha}, Bj^{\beta})a_i\right)^{q'}\right]^{\frac{1}{q'}} \le L'M' \left[\sum_{i=1}^{\infty} i^{\frac{\alpha p}{q'}(1-s)+\alpha p(A_1-A_2)-(p-1)(\alpha-1)}a_i^p\right]^{\frac{1}{p}}$$
(2.25)

hold for all real parameters A_1 , A_2 such that $A_1 \in \left(\max\{\frac{1-s}{p'}, \frac{\alpha-1}{\alpha p'}\}, \frac{1}{p'}\right)$, $A_2 \in \left(\max\{\frac{1-s}{q'}, \frac{\beta-1}{\beta q'}\}, \frac{1}{q'}\right)$, and for all non-negative sequences $(a_n)_{n \in \mathbb{N}}$ and $(b_n)_{n \in \mathbb{N}}$, where L' and M' are defined respectively in Corollary 2.1 and Theorem 2.3. Moreover, these inequalities are equivalent.

Remark 2.8 The Hilbert-type inequalities in Corollary 2.2 represent non-conjugate extensions of the corresponding inequalities from [52] and [53]. The results presented in this section are a part of the recent paper [16] by Čižmešija et. al.

2.3 Godunova-type inequalities

So far, we have considered integrals taken over certain subsets of \mathbb{R}_+ , that is, one-dimensional case. Since Theorem 2.1 covers more general settings, we apply that result to *n*-dimensional space \mathbb{R}^n_+ . Before presenting such results, it is necessary to introduce some notation for this section. For $\mathbf{x}, \mathbf{y} \in \mathbb{R}^n_+$, $\mathbf{x} = (x_1, x_2, \dots, x_n)$, $\mathbf{y} = (y_1, y_2, \dots, y_n)$, we define

$$\frac{\mathbf{y}}{\mathbf{x}} = \left(\frac{y_1}{x_1}, \frac{y_2}{x_2}, \dots, \frac{y_n}{x_n}\right) \quad \text{and} \quad \mathbf{x}^{\mathbf{y}} = x_1^{y_1} x_2^{y_2} \cdots x_n^{y_n}.$$

Moreover, 1 denotes the *n*-tuple (1, 1, ..., 1), and the vector is multiplied by a scalar in the usual way.

Motivated by the one-dimensional case (see relation (1.20), Chapter 1), we also define

$$k(\mathbf{a}) = \int_{\mathbb{R}^n_+} K(\mathbf{t}) \mathbf{t}^{-\mathbf{a}} d\mathbf{t}$$

where the function $K : \mathbb{R}^n_+ \to \mathbb{R}$ and the parameter $\mathbf{a} \in \mathbb{R}^n_+$ are such that the above integral converges.

Now, considering the weight functions $\varphi(\mathbf{x}) = \mathbf{x}^{\mathbf{A_1}}$, $\psi(\mathbf{y}) = \mathbf{y}^{\mathbf{A_2}}$, where $\mathbf{A_1} = (A_{11}, A_{12}, \dots, A_{1n})$, $\mathbf{A_2} = (A_{21}, A_{22}, \dots, A_{2n})$, and replacing $K(\mathbf{x}, \mathbf{y})$ with $\mathbf{x}^{-s}K(\frac{\mathbf{y}}{\mathbf{x}})$ in Theorem 2.1, we get the following result.

Theorem 2.4 Suppose that p, q, and λ are as in (2.1) and (2.2). If $\mathbf{s} \in \mathbb{R}^n_+$, $K : \mathbb{R}^n_+ \to \mathbb{R}$ is a non-negative measurable function, and the parameters $\mathbf{A_1}, \mathbf{A_2} \in \mathbb{R}^n$ are such that $L' = k^{\frac{1}{q'}}(q'\mathbf{A_2})k^{\frac{1}{p'}}(2 \cdot \mathbf{1} - \mathbf{s} - p'\mathbf{A_1}) < \infty$, then the inequalities

$$\begin{split} &\int_{\mathbb{R}^{n}_{+}} \int_{\mathbb{R}^{n}_{+}} \mathbf{x}^{-\lambda \mathbf{s}} K^{\lambda} \left(\frac{\mathbf{y}}{\mathbf{x}}\right) f(\mathbf{x}) g(\mathbf{y}) d\mathbf{x} d\mathbf{y} \\ &\leq L' \left[\int_{\mathbb{R}^{n}_{+}} \mathbf{x}^{\frac{p}{q'}(1-\mathbf{s})+p(\mathbf{A}_{1}-\mathbf{A}_{2})} f^{p}(\mathbf{x}) d\mathbf{x} \right]^{\frac{1}{p}} \left[\int_{\mathbb{R}^{n}_{+}} \mathbf{y}^{\frac{q}{p'}(1-\mathbf{s})+q(\mathbf{A}_{2}-\mathbf{A}_{1})} g^{q}(\mathbf{y}) d\mathbf{y} \right]^{\frac{1}{q}} \end{split}$$

and

$$\begin{split} & \left[\int_{\mathbb{R}^{n}_{+}} \mathbf{y}^{\frac{q'}{p'}(\mathbf{s}-\mathbf{1})+q'(\mathbf{A}_{1}-\mathbf{A}_{2})} \left(\int_{\mathbb{R}^{n}_{+}} \mathbf{x}^{-\lambda \mathbf{s}} K^{\lambda} \left(\frac{\mathbf{y}}{\mathbf{x}}\right) f(\mathbf{x}) d\mathbf{x} \right)^{q'} d\mathbf{y} \right]^{\frac{1}{q'}} \\ & \leq L' \left[\int_{\mathbb{R}^{n}_{+}} \mathbf{x}^{\frac{p}{q'}(\mathbf{1}-\mathbf{s})+p(\mathbf{A}_{1}-\mathbf{A}_{2})} f^{p}(\mathbf{x}) d\mathbf{x} \right]^{\frac{1}{p}} \end{split}$$

hold for all non-negative measurable functions $f, g : \mathbb{R}^n_+ \to \mathbb{R}$ and are equivalent.

Remark 2.9 For n = 1 we have that $x^{-s}K(\frac{y}{x})$ is a homogeneous function of degree -s, hence Theorem 2.4 may be regarded as an *n*-dimensional generalization of Corollary 2.1.

We explicitly state two particular cases of Theorem 2.4, obtained for some special choices of parameters. The first one considers $\mathbf{s} = s\mathbf{1}$, $\mathbf{A}_1 = A_1\mathbf{1}$, and $\mathbf{A}_2 = A_2\mathbf{1}$, where s, A_1, A_2 are real numbers.

Corollary 2.3 Suppose that p, q, and λ are as in (2.1) and (2.2). If $\mathbf{s} \in \mathbb{R}_+$, $K : \mathbb{R}^n_+ \to \mathbb{R}$ is a non-negative measurable function, and the parameters $A_1, A_2 \in \mathbb{R}$ are such that $L' = k^{\frac{1}{q'}}(q'A_2\mathbf{1})k^{\frac{1}{p'}}((2-s-p'A_1)\mathbf{1}) < \infty$, then the inequalities

$$\int_{\mathbb{R}^{n}_{+}} \int_{\mathbb{R}^{n}_{+}} \mathbf{x}^{-\lambda s \mathbf{1}} K^{\lambda} \left(\frac{\mathbf{y}}{\mathbf{x}}\right) f(\mathbf{x}) g(\mathbf{y}) d\mathbf{x} d\mathbf{y}$$

$$\leq L' \left[\int_{\mathbb{R}^{n}_{+}} \mathbf{x}^{\left[\frac{p}{q'}(1-s)+p(A_{1}-A_{2})\right] \mathbf{1}} f^{p}(\mathbf{x}) d\mathbf{x} \right]^{\frac{1}{p}} \left[\int_{\mathbb{R}^{n}_{+}} \mathbf{y}^{\left[\frac{q}{p'}(1-s)+q(A_{2}-A_{1})\right] \mathbf{1}} g^{q}(\mathbf{y}) d\mathbf{y} \right]^{\frac{1}{q}}$$

and

$$\begin{split} & \left[\int_{\mathbb{R}^{n}_{+}} \mathbf{y}^{\left[\frac{d'}{p'}(s-1)+q'(A_{1}-A_{2})\right]\mathbf{1}} \left(\int_{\mathbb{R}^{n}_{+}} \mathbf{x}^{-\lambda s} K^{\lambda}\left(\frac{\mathbf{y}}{\mathbf{x}}\right) f(\mathbf{x}) d\mathbf{x}\right)^{q'} d\mathbf{y}\right]^{\frac{1}{q'}} \\ & \leq L' \left[\int_{\mathbb{R}^{n}_{+}} \mathbf{x}^{\left[\frac{p}{q'}(1-s)+p(A_{1}-A_{2})\right]\mathbf{1}} f^{p}(\mathbf{x}) d\mathbf{x}\right]^{\frac{1}{p}} \end{split}$$

hold for all non-negative measurable functions $f, g : \mathbb{R}^n_+ \to \mathbb{R}$ and are equivalent.

The second special case of Theorem 2.4, and also the concluding result in this section, presents an inequality of E. K. Godunova from [24]. Namely, if we put $A_1 = \frac{2-s}{p'}$, $A_2 = 0$, $K = u^{\frac{1}{\lambda}}$ in Corollary 2.3, and consider the functions $\tilde{f}, \tilde{g} : \mathbb{R}^n_+ \to \mathbb{R}$, defined by $\tilde{f}(\mathbf{x}) = \mathbf{x}^{\left[-\frac{1}{p'}+(s-1)\lambda\right]\mathbf{1}}f(\mathbf{x}), \tilde{g}(\mathbf{y}) = \mathbf{y}^{\frac{1}{p'}\cdot\mathbf{1}}g(\mathbf{y})$, we get the following result.

Corollary 2.4 Let p, q, and λ be as in (2.1) and (2.2). If $\mathbf{s} \in \mathbb{R}_+$ and $u : \mathbb{R}^n_+ \to \mathbb{R}$ is a non-negative measurable function, then the inequalities

$$\int_{\mathbb{R}^n_+} \int_{\mathbb{R}^n_+} \mathbf{x}^{-\left(\frac{1}{p'}+\lambda\right)\mathbf{1}} \mathbf{y}^{\frac{1}{p'}\mathbf{1}} u\left(\frac{\mathbf{y}}{\mathbf{x}}\right) f(\mathbf{x}) g(\mathbf{y}) d\mathbf{x} d\mathbf{y} \leq ||u||_{L^{\frac{1}{\lambda}}(\mathbb{R}^n_+)} ||f||_{L^{p'}(\mathbb{R}^n_+)} ||g||_{L^{q'}(\mathbb{R}^n_+)}$$

and

$$\left[\int_{\mathbb{R}^n_+} \mathbf{y}^{\frac{q'}{p'}\mathbf{1}} \left(\int_{\mathbb{R}^n_+} \mathbf{x}^{-\left(\frac{1}{p'}+\lambda\right)\mathbf{1}} u\left(\frac{\mathbf{y}}{\mathbf{x}}\right) f(\mathbf{x}) d\mathbf{x}\right)^{q'} d\mathbf{y}\right]^{\frac{1}{q'}} \le ||u||_{L^{\frac{1}{\lambda}}(\mathbb{R}^n_+)} ||f||_{L^{p'}(\mathbb{R}^n_+)} ||f||_{L^{p'}(\mathbb{R}^n_+)}$$

hold for all non-negative measurable functions $f, g : \mathbb{R}^n_+ \to \mathbb{R}$ and are equivalent.

Remark 2.10 Since the first inequality in Corollary 2.4 was proved by E. K. Godunova in [24], all inequalities presented in this section will be referred to as the Godunova-type inequalities. These inequalities are established in [16]. In addition, Hilbert-type inequalities with vector variables will be extensively studied in Chapter 3.

2.4 Multidimensional case

The main objective of this section is to extend Theorem 2.1 to a multidimensional case. The three-dimensional version of the Hilbert-type inequality, that is, the relation (2.5), was given by F.F. Bonsall [9], in 1950s. Although Bonsall also established conditions for the set of *n* non-conjugate parameters, there were no results in that direction.

In order to obtain our general results we introduce here *n*-dimensional extension of nonconjugate exponents, defined in [9]. Let $n \in \mathbb{N}$, $n \ge 2$, and let real parameters p_1, \ldots, p_n be such that

$$p_1, \dots, p_n > 1, \quad \sum_{i=1}^n \frac{1}{p_i} \ge 1.$$
 (2.26)

Define

$$\lambda = \frac{1}{n-1} \sum_{i=1}^{n} \frac{1}{p'_i} \quad \text{and} \quad \frac{1}{q_i} = \lambda - \frac{1}{p'_i}, \quad i = 1, \dots, n,$$
(2.27)

where $\frac{1}{p_i} + \frac{1}{p'_i} = 1$, i = 1, ..., n. On the other hand, for any choice of parameters as in (2.26), it follows from (2.27) that

$$\frac{1}{q_i} + (1 - \lambda) = \frac{1}{p_i}, \quad i = 1, \dots, n,$$
(2.28)

and

$$\sum_{i=1}^{n} \frac{1}{q_i} + (1 - \lambda) = 1.$$
(2.29)

Hence, in order to apply the Hölder inequality with exponents $q_1, \ldots, q_n, \frac{1}{1-\lambda}$, we need to require

$$\frac{1}{q_i} > 0, \ i = 1, \dots, n.$$
 (2.30)

Note that for $n \ge 3$ conditions (2.26) and (2.27) do not automatically imply (2.30). More precisely, since (2.26) and (2.27) give only

$$\frac{1}{q_i} > \frac{2-n}{n-1} \frac{1}{p'_i}, \ i = 1, \dots, n,$$

some of q_i may be negative. For example, for $p_1 = 2$ and $p_2 = p_3 = \frac{20}{19}$ we have $\frac{1}{q_1} = -\frac{1}{5} < 0$. Therefore, the condition (2.30) is not redundant.

Observe that for $\lambda = 1$ the above parameters reduce to the conjugate case, that is, $\sum_{i=1}^{n} \frac{1}{p_i} = 1$ and $p_i = q_i, i = 1, 2, ..., n$.

Now, we are ready to state and prove general forms of multidimensional Hilbert-type inequalities with non-conjugate exponents. These inequalities will include integrals taken over general subsets of \mathbb{R}_+ , equipped with σ -finite measures. Of course, the following extension may also be regarded as a non-conjugate version of Theorem 1.18.

Theorem 2.5 Let λ , p_i , p'_i , q_i , i = 1, 2, ..., n, $n \ge 2$, fulfill relations (2.26), (2.27) and (2.30), and let Ω be a measure space with σ -finite measures μ_i , i = 1, 2, ..., n. Further, suppose that $K : \Omega^n \to \mathbb{R}$ and $\phi_{ij} : \Omega \to \mathbb{R}$, i, j = 1, ..., n, are non-negative measurable functions such that $\prod_{i,j=1}^n \phi_{ij}(x_j) = 1$ a.e. on Ω^n . If the functions F_i , i = 1, 2, ..., n, are defined by

$$F_{i}(x_{i}) = \left[\int_{\Omega^{n-1}} K(x_{1}, \dots, x_{n}) \prod_{j=1, j \neq i}^{n} \phi_{ij}^{q_{i}}(x_{j}) d\mu_{1}(x_{1}) \dots d\mu_{j-1}(x_{j-1}) \right. \\ \left. \times d\mu_{j+1}(x_{j+1}) \dots d\mu_{n}(x_{n}) \right]^{\frac{1}{q_{i}}},$$

then for all non-negative measurable functions $f_i: \Omega \to \mathbb{R}, i = 1, 2, ..., n$, the inequalities

$$\int_{\Omega^n} K^{\lambda}(x_1, ..., x_n) \prod_{i=1}^n f_i(x_i) d\mu_1(x_1) ... d\mu_n(x_n) \le \prod_{i=1}^n \|\phi_{ii} F_i f_i\|_{L^{p_i}(\mu_i)}$$
(2.31)

and

$$\left\{ \int_{\Omega} \left[\frac{1}{(\phi_{nn}F_n)(x_n)} \int_{\Omega^{n-1}} K^{\lambda}(x_1, ..., x_n) \prod_{i=1}^{n-1} f_i(x_i) d\mu_1(x_1) ... d\mu_{n-1}(x_{n-1}) \right]^{p'_n} d\mu_n(x_n) \right\}^{\frac{1}{p'_n}} \leq \prod_{i=1}^{n-1} \|\phi_{ii}F_if_i\|_{L^{p_i}(\mu_i)} \tag{2.32}$$

hold and are equivalent.

Proof. The left-hand side of inequality (2.31) can be rewritten as

$$\begin{split} &\int_{\Omega^n} K^{\lambda}(x_1,...,x_n) \prod_{i=1}^n f_i(x_i) d\mu_1(x_1)...d\mu_n(x_n) \\ &= \int_{\Omega^n} \prod_{i=1}^n \left[K(x_1,...,x_n) \phi_{ii}{}^{p_i}(x_i) \prod_{j=1,j\neq i}^n \phi_{ij}^{q_i}(x_j) F_i{}^{p_i-q_i}(x_i) f_i{}^{p_i}(x_i) \right]^{\frac{1}{q_i}} \\ &\times \left[\prod_{i=1}^n (\phi_{ii}F_if_i)^{p_i}(x_i) \right]^{1-\lambda} d\mu_1(x_1)...d\mu_n(x_n). \end{split}$$

In addition, since $\sum_{i=1}^{n} \frac{1}{q_i} + 1 - \lambda = 1$, $q_i > 1$, $0 < \lambda \le 1$, applying the Hölder inequality to the above relation yields

$$\int_{\Omega^{n}} K^{\lambda}(x_{1},...,x_{n}) \prod_{i=1}^{n} f_{i}(x_{i}) d\mu_{1}(x_{1})...d\mu_{n}(x_{n})$$

$$\leq \prod_{i=1}^{n} \left[\int_{\Omega} (\phi_{ii}F_{i}f_{i})^{p_{i}}(x_{i}) d\mu_{i}(x_{i}) \right]^{\frac{1}{q_{i}}} \prod_{i=1}^{n} \left[\int_{\Omega} (\phi_{ii}F_{i}f_{i})^{p_{i}}(x_{i}) d\mu_{i}(x_{i}) \right]^{1-\lambda}.$$

Finally, since $\frac{1}{q_i} + 1 - \lambda = \frac{1}{p_i}$, we obtain inequality (2.31). Now, we show that inequalities (2.31) and (2.32) are equivalent. Suppose that inequality (2.31) is valid. Setting the function $f_n : \Omega \to \mathbb{R}$, defined by

$$f_n(x_n) = \left[\frac{1}{(\phi_{nn}F_n)^{p_n}(x_n)} \int_{\Omega^{n-1}} K^{\lambda}(x_1,...,x_n) \prod_{i=1}^{n-1} f_i(x_i) d\mu_1(x_1)...d\mu_{n-1}(x_{n-1})\right]^{\frac{p'_n}{p_n}},$$

in inequality (2.31), we have

$$I(x_n)^{p'_n} \leq \prod_{i=1}^{n-1} \left[\int_{\Omega} (\phi_{ii}F_i f_i)^{p_i}(x_i) d\mu_i(x_i) \right]^{\frac{1}{p_i}} I(x_n)^{\frac{p'_n}{p_n}},$$

where $I(x_n)$ denotes the left-hand side of inequality (2.32). Clearly, this relation represents inequality (2.32).

It remains to prove that inequality (2.31) is a consequence of (2.32). Assume, therefore, that inequality (2.32) holds. The left-hand side of (2.31) can be rewritten as

$$\int_{\Omega^{n}} K^{\lambda}(x_{1},...,x_{n}) \prod_{i=1}^{n} f_{i}(x_{i}) d\mu_{1}(x_{1})...d\mu_{n}(x_{n})$$

$$= \int_{\Omega} (\phi_{nn}F_{n}f_{n})(x_{n}) \left[\frac{1}{(\phi_{nn}F_{n})(x_{n})} \int_{\Omega^{n-1}} K^{\lambda}(x_{1},...,x_{n}) \right] \times \prod_{i=1}^{n-1} f_{i}(x_{i}) d\mu_{1}(x_{1})...d\mu_{n-1}(x_{n-1}) d\mu_{n}(x_{n}),$$

hence, applying the Hölder inequality with conjugate exponents p_n and p'_n , we have

$$\int_{\Omega^n} K^{\lambda}(x_1,...,x_n) \prod_{i=1}^n f_i(x_i) d\mu_1(x_1)...d\mu_n(x_n) \le \|\phi_{nn}F_nf_n\|_{L^{p_n}(\mu_n)} I(x_n),$$

and the result follows from (2.32).

Remark 2.11 Since the crucial step in proving the previous theorem was in applying the Hölder inequality, equality in (2.31) holds if and only if the functions

$$K(x_1,...,x_n)\phi_{ii}^{p_i}(x_i)\prod_{j=1,j\neq i}^n\phi_{ij}^{q_i}(x_j)F_i^{p_i-q_i}(x_i)f_i^{p_i}(x_i), \quad i=1,2,...n,$$

and $\prod_{i=1}^{n} (\phi_{ii}F_if_i)^{p_i}(x_i)$ are effectively proportional. Clearly, this trivially happens if at least one of the functions f_i , i = 1, 2, ..., n, is a zero-function. Otherwise, these conditions can be rewritten in a more suitable form, yielding the explicit expressions for the functions and kernel, that is, $f_i(x_i) = C_i \phi_{ii}(x_i)^{\frac{q_i}{1-\lambda q_i}} F_i(x_i)^{(1-\lambda)q_i}$, i = 1, 2, ..., n, and $K(x_1, x_2, ..., x_n) = C \prod_{i=1}^{n} F_i^{q_i}(x_i)$, where *C* and *C_i* are positive constants. It is possible only if the functions

$$\frac{\prod_{j=1, j\neq i}^{n} \phi_{jj}^{\frac{\lambda q_j}{1-\lambda q_j}}(x_j)}{\prod_{j=1, j\neq i}^{n} \phi_{ij}^{\lambda q_j}(x_j)}, \quad i = 1, 2, \dots, n,$$

are appropriate constants, and

$$\int_{\Omega} F_i^{q_i}(x_i) \phi_{ii}^{\frac{q_i}{1-\lambda q_i}}(x_i) d\mu_i(x_i) < \infty, \quad i = 1, 2, \dots n$$

Otherwise, the inequalities in Theorem 2.5 are strict.

Remark 2.12 If the parameters p_i , i = 1, 2, ..., n, are chosen in such a way that

$$q_j > 0, j \in \{1, 2, \dots n\}, \quad q_i < 0, i \neq j, \text{ and } \lambda < 1,$$
 (2.33)

or

$$q_i < 0, \quad i = 1, 2, \dots, n,$$
 (2.34)

then the reverse inequalities in (2.31) and (2.32) hold, due to the reverse Hölder inequality.

As an application of Theorem 2.5, we consider the case of homogeneous kernel K: $\mathbb{R}^n_+ \to \mathbb{R}$, defined by $K(x_1, ..., x_n) = (x_1 + \dots + x_n)^{-s}$, s > 0, and the power weight functions $\phi_{ij}: \mathbb{R}^n_+ \to \mathbb{R}$, $\phi_{ij}(x_j) = x_{j}^{A_{ij}}$, $A_{ij} \in \mathbb{R}$, with respect to Lebesgue measures dx_i , i = 1, 2, ..., non \mathbb{R}_+ . The parameters A_{ij} fulfill relations $\sum_{i=1}^n A_{ij} = 0$, j = 1, ..., n, so that the condition $\prod_{i,j=1}^n \phi_{ij}(x_j) = 1$ is fulfilled. Regarding the form of the kernel, we obtain inequalities with the constant factor expressed in terms of the Gamma function.

Theorem 2.6 Let λ, p_i, p'_i, q_i , i = 1, 2, ..., n, $n \ge 2$, be as in (2.26), (2.27), and (2.30), and let A_{ij} , i, j = 1, ..., n, be real parameters such that $\sum_{i=1}^{n} A_{ij} = 0$ for j = 1, ..., n. If s > 0, $\alpha_i = \sum_{j=1}^{n} A_{ij}$, $A_{ij} > -\frac{1}{q_i}$, $i \ne j$, $A_{ii} - \alpha_i > \frac{n-s-1}{q_i}$, and

$$K = \frac{1}{\Gamma^{\lambda}(s)} \prod_{i=1}^{n} \Gamma^{\frac{1}{q_i}}(s - n + 1 - q_i \alpha_i + q_i A_{ii}) \prod_{i,j=1, i \neq j}^{n} \Gamma^{\frac{1}{q_i}}(q_i A_{ij} + 1),$$

then the inequalities

$$\int_{\mathbb{R}^{n}_{+}} \frac{\prod_{i=1}^{n} f_{i}(x_{i})}{(\sum_{j=1}^{n} x_{j})^{\lambda_{s}}} dx_{1} \dots dx_{n} < K \prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{\frac{p_{i}}{q_{i}}(n-1-s)+p_{i}\alpha_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$
(2.35)

and

$$\begin{bmatrix} \int_{0}^{\infty} x_{n}^{(1-\lambda p_{n}')(n-1-s)-p_{n}'\alpha_{n}} \left(\int_{\mathbb{R}^{n-1}_{+}} \frac{\prod_{i=1}^{n-1} f_{i}(x_{i})}{(\sum_{j=1}^{n} x_{j})^{\lambda s}} dx_{1} \dots dx_{n-1} \right)^{p_{n}'} dx_{n} \end{bmatrix}^{\frac{1}{p_{n}'}} \\ < K \prod_{i=1}^{n-1} \left[\int_{0}^{\infty} x_{i} \frac{p_{i}}{q_{i}}^{(n-1-s)+p_{i}\alpha_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$
(2.36)

hold for all non-negative measurable functions $f_i : \mathbb{R}_+ \to \mathbb{R}$, $f_i \neq 0$, i = 1, ..., n. Moreover, these inequalities are equivalent.

Another way of extending Hilbert-type inequalities with non-conjugate exponents to the multidimensional setting arises from inequality (2.6), presented at the beginning of this chapter.

Theorem 2.7 Let $n \in \mathbb{N}$, $n \ge 2$, and let parameters λ , p_i , q_i , i = 1, 2, ..., n, be as in (2.26), (2.27), and (2.30). Let $\mu_1, ..., \mu_n$ be positive σ -finite measures on Ω . If $K : \Omega^n \to \mathbb{R}$, $F_i : \Omega^n \to \mathbb{R}$, $\phi_{ij} : \Omega \to \mathbb{R}$, i, j = 1, ..., n, are non-negative measurable functions such that

$$\prod_{i,j=1}^{n} \phi_{ij}(x_j) = 1, \ a.e. \ on \ \Omega^n,$$
(2.37)

then the inequality

$$\int_{\Omega^{n}} K(x_{1},...,x_{n}) \prod_{i=1}^{n} f_{i}(x_{i}) d\mu_{1}(x_{1})...d\mu_{n}(x_{n})$$

$$\leq \prod_{i=1}^{n} \left[\int_{\Omega^{n}} \left(KF_{i}^{p_{i}-q_{i}} \right) (x_{1},...,x_{n}) (\phi_{ii}f_{i})^{p_{i}}(x_{i}) \prod_{j\neq i} \phi_{ij}^{q_{i}}(x_{j}) d\mu_{1}(x_{1})...d\mu_{n}(x_{n}) \right]^{\frac{1}{q_{i}}} \times \left[\int_{\Omega^{n}} K(x_{1},...,x_{n}) \prod_{i=1}^{n} F_{i}^{p_{i}}(x_{1},...,x_{n}) (\phi_{ii}f_{i})^{p_{i}}(x_{i}) d\mu_{1}(x_{1})...d\mu_{n}(x_{n}) \right]^{1-\lambda} (2.38)$$

holds for all non-negative measurable functions $f_i: \Omega \to \mathbb{R}, i = 1, ..., n$.

Proof. Note that from (2.28) we have $\frac{p_i}{q_i} + p_i(1 - \lambda) = 1$, i = 1, ..., n. Using this and (2.29), the left-hand side of (2.38) can be written as

$$\begin{split} &\int_{\Omega^n} K(x_1, \dots, x_n) \prod_{i=1}^n f_i(x_i) \, d\mu_1(x_1) \dots d\mu_n(x_n) \\ &= \int_{\Omega^n} K^{\sum_{i=1}^n \frac{1}{q_i} + 1 - \lambda}(x_1, \dots, x_n) \prod_{i=1}^n f_i^{\frac{p_i}{q_i} + p_i(1 - \lambda)}(x_i) \\ &\times \prod_{i=1}^n F_i^{\frac{p_i}{q_i} - 1 + p_i(1 - \lambda)}(x_1, \dots, x_n) \prod_{i=1}^n \phi_{ii}^{\frac{p_i}{q_i} + p_i(1 - \lambda)}(x_i) \\ &\times \prod_{j \neq i} \phi_{ij}(x_j) \, d\mu_1(x_1) \dots d\mu_n(x_n) \\ &= \int_{\Omega^n} \prod_{i=1}^n \left[\left(KF_i^{p_i - q_i} \right)(x_1, \dots, x_n)(\phi_{ii}f_i)^{p_i}(x_i) \prod_{j \neq i} \phi_{ij}^{q_i}(x_j) \right]^{\frac{1}{q_i}} \\ &\times \left[K(x_1, \dots, x_n) \prod_{i=1}^n F_i^{p_i}(x_1, \dots, x_n)(\phi_{ii}f_i)^{p_i}(x_i) \right]^{1 - \lambda} d\mu_1(x_1) \dots d\mu_n(x_n). \end{split}$$

The inequality (2.38) now follows by using the Hölder inequality with the exponents $q_1, \ldots, q_n, \frac{1}{1-\lambda}$.

Remark 2.13 Observe that without loss of generality the condition (2.37) from the statement of Theorem 2.7 can be replaced by $\prod_{i=1}^{n} \phi_{ij}(x_j) = 1$ a.e. on Ω , for j = 1, ..., n, since (2.37) implies that

$$\prod_{i=1}^{n} \phi_{ij}(x_j) = c_j = const, \ j = 1, \dots, n,$$
(2.39)

where $c_1 \cdots c_n = 1$.

Remark 2.14 Obviously, (2.38) becomes equality if at least one of the functions involved in its left-hand side is a zero-function. Otherwise, equality holds if and only if it holds in the Hölder inequality, that is, only if the functions $KF_i^{p_i-q_i}(\phi_{ii}f_i)^{p_i}\prod_{j\neq i}\phi_{ij}^{q_i}$, i = 1,...,n, and $K\prod_{i=1}^n (F_i\phi_{ii}f_i)^{p_i}$ are effectively proportional. Therefore, equality in (2.38) occurs if and only if there exist positive constants α_i , β_{ij} , i, j = 1,...,n, $j \neq i$, such that

$$KF_{i}^{p_{i}-q_{i}}\left(\phi_{ii}f_{i}\right)^{p_{i}}\prod_{l\neq i}\phi_{il}^{q_{i}}=\alpha_{i}K\prod_{l=1}^{n}\left(F_{l}\phi_{ll}f_{l}\right)^{p_{l}},\quad i=1,\ldots,n,$$
(2.40)

and

$$KF_{i}^{p_{i}-q_{i}}(\phi_{ii}f_{i})^{p_{i}}\prod_{l\neq i}\phi_{il}^{q_{i}}=\beta_{ij}KF_{j}^{p_{j}-q_{j}}(\phi_{jj}f_{j})^{p_{j}}\prod_{l\neq j}\phi_{jl}^{q_{j}}, \quad i\neq j.$$
(2.41)

Moreover, the relation (2.40) is equivalent to

$$F_i^{-q_i} = \alpha_i \prod_{l \neq i} \phi_{il}^{-q_i} \left(F_l \phi_{ll} f_l \right)^{p_l}, \quad i = 1, \dots, n.$$
(2.42)

In the special case when $F_i \equiv F_i(x_i)$, i = 1, ..., n, the functions f_i and ϕ_{ij} from (2.41) and (2.42) can be expressed explicitly in terms of ϕ_{ii} . More precisely, from (2.42) we have

$$F_i \equiv const, \quad i = 1, \dots, n, \tag{2.43}$$

directly, since the right-hand side of this relation depends on x_l , $l \neq i$, while its left-hand side in this setting is a function of x_i . Considering this, (2.41) becomes

$$(\phi_{ii}f_i)^{p_i}\phi_{ji}^{-q_j} = \gamma_{ij}(\phi_{jj}f_j)^{p_j}\prod_{l\neq i,j}\phi_{jl}^{q_j}\prod_{l\neq i}\phi_{il}^{-q_i}, \quad i\neq j,$$
(2.44)

for some positive constants γ_{ij} . Thus,

$$(\phi_{ii}f_i)^{p_i}\phi_{ji}^{-q_j} \equiv const, \quad i = 1, \dots, n, \ j \neq i,$$
(2.45)

where again we exploited the fact that the left-hand side of (2.44) depends only on x_i , while its right-hand side is a function of x_l , l = 1, ..., n, $l \neq i$. The relation (2.45) further implies that $\phi_{ji}^{q_j} \phi_{li}^{-q_l} = const$, i = 1, ..., n, $j, l \neq i$, which combined with (2.39) gives

$$\phi_{ii}\phi_{ji}^{q_j\sum_{l\neq i}\frac{1}{q_l}} \equiv const, \quad i = 1, \dots, n, \ j \neq i.$$
(2.46)

Since by (2.27) and (2.29) we have $q_j \sum_{l \neq i} \frac{1}{q_l} = \frac{q_j}{p'_i}$, the relation (2.46) can be transformed into

$$\phi_{ii}^{p_i} \phi_{ji}^{q_j} \equiv const, \quad i = 1, \dots, n, \ j \neq i,$$
(2.47)

while (2.45) becomes

$$f_i^{p_i} \equiv C_i \phi_{ii}^{-(p_i + p'_i)}, \quad i = 1, \dots, n,$$
(2.48)

for some positive constants C_i , i = 1, ..., n. Hence, if $F_i \equiv F_i(x_i)$, the conditions (2.43), (2.47), and (2.48) are necessary and sufficient for equality in (2.38).

Remark 2.15 If the parameters p_i , i = 1, 2, ..., n, in Theorem 2.7 are such that

$$0 < p_i < 1, \qquad \frac{n-1}{p_i} + 1 < \sum_{j=1}^n \frac{1}{p_j}, \ i = 1, \dots, n,$$
 (2.49)

and λ , and q_i , i = 1, 2, ..., n, are defined by (2.27), then the sign of inequality in (2.38) is reversed. To justify this assertion, observe that the first inequality in (2.49) gives $\frac{1}{p_i^2} < 0$, i = 1, ..., n, so we have $\lambda < 0$. Similarly, from the second relation in (2.49) it follows that

$$\frac{1}{q_i} = \lambda - \frac{1}{p'_i} = \frac{1}{n-1} \left(\frac{n-1}{p_i} + 1 - \sum_{j=1}^n \frac{1}{p_j} \right) < 0, \ i = 1, \dots, n.$$

Therefore, $q_i < 0$, i = 1, ..., n, and $0 < \frac{1}{1-\lambda} < 1$, so (2.38) holds with the reverse inequality as a direct consequence of the reverse Hölder inequality. The same result is also achieved with the parameters p_i , i = 1, 2, ..., n, satisfying

$$\sum_{i=1}^{n} \frac{1}{p_i} < 1 \quad \text{and} \quad 0 < p_l < 1, \quad \frac{n-1}{p_i} + 1 < \sum_{j=1}^{n} \frac{1}{p_j}, \quad i \neq l,$$
(2.50)

for some $l \in \{1, \ldots, n\}$, since from (2.50) we obtain $\frac{1}{1-\lambda} < 0$, $q_l > 0$, and $q_i < 0$, $i \neq l$.

To conclude this section, we restate Theorem 2.7 in the case of n = 2. This result is interesting in its own right, since it will be applied in the next chapter, where we shall consider Hilbert-type inequalities in some particular cases.

Theorem 2.8 Let p, q, and λ be as in (2.1) and (2.2). Let μ_1 and μ_2 be positive σ -finite measures on Ω . If K, F, and G are non-negative measurable functions on Ω^2 and φ and ψ are non-negative measurable functions on Ω , then the inequality

$$\int_{\Omega^{2}} K(x,y) f(x)g(y) d\mu_{1}(x) d\mu_{2}(y) \\
\leq \left[\int_{\Omega^{2}} (KF^{p-q'})(x,y) \psi^{-q'}(y) (\varphi f)^{p}(x) d\mu_{1}(x) d\mu_{2}(y) \right]^{\frac{1}{q'}} \\
\times \left[\int_{\Omega^{2}} (KG^{q-p'})(x,y) \varphi^{-p'}(x) (\psi g)^{q}(y) d\mu_{1}(x) d\mu_{2}(y) \right]^{\frac{1}{p'}} \\
\times \left[\int_{\Omega^{2}} (KF^{p}G^{q})(x,y) (\varphi f)^{p}(x) (\psi g)^{q}(y) d\mu_{1}(x) d\mu_{2}(y) \right]^{1-\lambda}$$
(2.51)

holds for all non-negative measurable functions f and g on Ω .

Proof. The proof follows directly from Theorem 2.7 using substitutions $p_1 = p$, $p_2 = q$, $q_1 = q'$, $q_2 = p'$, $\phi_{11} = \varphi$, and $\phi_{22} = \psi$. Observe that from $\phi_{11}\phi_{21} = 1$ and $\phi_{12}\phi_{22} = 1$ we have $\phi_{21} = \frac{1}{\varphi}$ and $\phi_{12} = \frac{1}{\psi}$.

Remark 2.16 If we rewrite Theorem 2.8 with $\Omega = \mathbb{R}_+$, Lebesgue measures, the kernel $K(x,y) = (x+y)^{-\lambda}$, and with functions $F(x,y) = G(x,y) \equiv 1$, $\varphi(x) = x^{\frac{1}{pp'}}$, and $\psi(y) = y^{\frac{1}{qq'}}$, we obtain (2.6). Hence, Theorem 2.8 can be viewed as a generalization of the mentioned Bonsall result (2.6) from [9].

Remark 2.17 Theorem 2.5 and its consequences are taken from [12], while Theorem 2.7 is obtained in [17].

2.5 Examples with hypergeometric functions

We have already discussed that the general method with non-conjugate exponents, developed in this chapter, can be combined with particular settings and diverse methods presented in Section 1.4. In addition, we have seen some Hilbert-type inequalities involving hypergeometric functions.

Therefore, we consider here some particular settings in which hypergeometric functions occur in a more general manner. We start with a classic example.

2.5.1 Hilbert-type inequalities and Gaussian hypergeometric function

Gaussian hypergeometric function is the formal power series in $z \in \mathbb{C}$ with three parameters, defined in terms of rising factorial powers:

$$F(a,b;c;z) = \sum_{k\geq 0} \frac{a^{\overline{k}} b^{\overline{k}}}{c^{\overline{k}}} \cdot \frac{z^{k}}{k!}, \ a,b,c,z \in \mathbb{R}, \ |z| < 1.$$
(2.52)

Here, the rising factorial power is $a^{\overline{k}} = a(a+1)(a+2)\cdots(a+k-1), k \in \mathbb{N}$, and $a^{\overline{0}} = 1$, $a \neq 0$.

To avoid division by zero, c is neither zero nor negative integer. The series (2.52) is often called the Gaussian hypergeometric function, because many of its interesting properties were first proved by Gauss. In fact, it was the only hypergeometric series until the second half of nineteenth century, when everything was generalized to arbitrary number of parameters. For more details, the reader is referred to [26].

Relation (1.34) from Chapter 1 represents the integral representation of the above power series. Moreover, knowing the relation between the Gamma and the Beta function, (1.34) can also be rewritten as

$$F(a,b;c;z) = \frac{1}{B(b,c-b)} \int_0^1 t^{b-1} (1-t)^{c-b-1} (1-zt)^{-a} dt, \qquad (2.53)$$

where c > b > 0 and |z| < 1.

In order to obtain Hilbert-type inequality with constant factor expressed in terms of F(a,b;c;z), we are going to rewrite relation (2.53) in a more suitable form.

Lemma 2.1 Suppose $a, b, c, \alpha, \gamma \in \mathbb{R}$ are such that a + c > b > 0 and $0 < \alpha < 2\gamma$. Then,

$$\int_{0}^{\infty} \frac{x^{b-1}}{(1+\alpha x)^{a}(1+\gamma x)^{c}} dx = \gamma^{-b} B(b,a+c-b) F\left(a,b;a+c;1-\frac{\alpha}{\gamma}\right).$$
(2.54)

Proof. Consider the integral $I = \int_0^1 t^{b-1} (1-t)^{c-b-1} (1-zt)^{-a} dt$. Using the substitutions $1-t = \frac{1}{1+u}$, $u = \gamma x$, $\gamma > 0$, and the abbreviation $\alpha = (1-z)\gamma$, we obtain

$$I = \gamma^b \int_0^\infty \frac{x^{b-1}}{(1+\alpha x)^a (1+\gamma x)^{c-a}} dx$$

Now, utilizing (2.53) we have

$$\int_0^\infty \frac{x^{b-1}}{(1+\alpha x)^a (1+\gamma x)^{c-a}} dx = \gamma^{-b} B(b,c-b) F\left(a,b;c;1-\frac{\alpha}{\gamma}\right).$$

Finally, replacing c - a with c in the previous formula, we get (2.54).

Now, considering Corollary 2.1 with the homogeneous kernel

$$K(x,y) = (x + \alpha_1 y)^{-s_1} (x + \alpha_2 y)^{-s_2}, \qquad (2.55)$$

where $\alpha_1, \alpha_2 > 0, \frac{1}{2} < \frac{\alpha_1}{\alpha_2} < 2$, and $s_1 + s_2 > 0$, it follows that the corresponding constant factor is a product of two integrals of the form (2.54). Moreover, the degree of homogeneity of the above kernel is $-(s_1 + s_2)$, so Corollary 2.1 yields the following consequence.

Corollary 2.5 Let p, q, and λ be as in (2.1) and (2.2), and let $\alpha_1, \alpha_2 > 0$, $\frac{1}{2} < \frac{\alpha_1}{\alpha_2} < 2$, $s_1 + s_2 > 0$. If $f, g : \mathbb{R}_+ \to \mathbb{R}$ are non-negative measurable functions, then the inequalities

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{f(x)g(y)}{(x+\alpha_{1}y)^{\lambda_{s_{1}}}(x+\alpha_{2}y)^{\lambda_{s_{2}}}} dxdy$$

$$\leq K' \left[\int_{0}^{\infty} x^{\frac{p}{q'}(1-s_{1}-s_{2})+p(A_{1}-A_{2})} f^{p}(x)dx \right]^{\frac{1}{p}}$$

$$\times \left[\int_{0}^{\infty} y^{\frac{q}{p'}(1-s_{1}-s_{2})+q(A_{2}-A_{1})} g^{q}(y)dy \right]^{\frac{1}{q}}$$
(2.56)

and

$$\left[\int_{0}^{\infty} y^{\frac{q'}{p'}(s_{1}+s_{2}-1)+q'(A_{1}-A_{2})} \left(\int_{0}^{\infty} \frac{f(x)}{(x+\alpha_{1}y)^{\lambda s_{1}}(x+\alpha_{2}y)^{\lambda s_{2}}} dx\right)^{q'} dy\right]^{\frac{1}{q'}} \leq K' \left[\int_{0}^{\infty} x^{\frac{p}{q'}(1-s_{1}-s_{2})+p(A_{1}-A_{2})} f^{p}(x) dx\right]^{\frac{1}{p}}$$
(2.57)

hold for all $A_1 \in (\frac{1-s_1-s_2}{p'}, \frac{1}{p'})$ and $A_2 \in (\frac{1-s_1-s_2}{q'}, \frac{1}{q'})$, where

$$K' = \alpha_1^{\frac{1-s_1}{p'} - A_1} \alpha_2^{A_2 - \frac{1}{q'} - \frac{s_2}{p'}} B^{\frac{1}{q'}} (1 - q'A_2, s_1 + s_2 + q'A_2 - 1) \times B^{\frac{1}{p'}} (1 - p'A_1, s_1 + s_2 + p'A_1 - 1) F^{\frac{1}{q'}} \left(s_1, 1 - q'A_2; s_1 + s_2; 1 - \frac{\alpha_1}{\alpha_2} \right) \times F^{\frac{1}{p'}} \left(s_2, 1 - p'A_1; s_1 + s_2; 1 - \frac{\alpha_1}{\alpha_2} \right).$$
(2.58)

Moreover, inequalities (2.56) and (2.57) are equivalent.

Remark 2.18 Every hypergeometric series always has the value 1 when z = 0. Hence, if $\alpha_1 = \alpha_2$, then the hypergeometric part of the above constant (2.58) takes the value of 1 and inequalities (2.56) and (2.57) reduce to already known cases, considered in Section 1.4.

On the other hand, hypergeometric series (2.52) converges also for z = 1 when b is a non-positive integer or c > a + b. In addition, since

$$F(a,b;c;1) = \frac{\Gamma(c-a-b)\Gamma(c)}{\Gamma(c-a)\Gamma(c-b)},$$
(2.59)

(see [26]), relation (2.54) also holds for $\alpha = 0$ and reduces to the well-known formula for the Beta function:

$$\int_0^\infty \frac{x^{b-1}}{(1+\gamma x)^c} dx = \gamma^{-b} B(b,c-b), \quad \text{where } c > b > 0.$$
 (2.60)

Therefore, if $\alpha_1 = 0$ or $\alpha_2 = 0$, the previous corollary also yields inequalities without hypergeometric part. Such cases are omitted here.

We conclude the previous discussion with a remark about the best possible constant factors appearing in inequalities (2.56) and (2.57), which can be achieved in the conjugate case.

Remark 2.19 If *p* and *q* are conjugate exponents, then the kernel (2.55) fulfills conditions as in Theorem 1.13 (Chapter 1). Moreover, since *K* is homogeneous of degree $-(s_1 + s_2)$, the parameters A_1 and A_2 , that provide the best possible constant factor, must fulfill the relation $pA_2 + qA_1 = 2 - s_1 - s_2$. Under the above assumptions and utilizing the so-called Euler identity $F(a,b;c;z) = (1-z)^{c-a-b}F(c-a,c-b;c;z)$ (see [26]), the constant factor *K'* from Corollary 2.5 reduces to

$$\alpha_2^{pA_2-1}B(1-pA_2,1-qA_1)F\left(s_1,1-pA_2;s_1+s_2;1-\frac{\alpha_1}{\alpha_2}\right)$$

and is the best possible.

2.5.2 Hilbert-type inequalities and generalized hypergeometric functions $_mF_n$

Gaussian hypergeometric function is naturally extended to an arbitrary number of parameters, which gives generalized hypergeometric function. Such generalized series also have integral representations. More precisely, we shall use the so-called Poisson-type integral representations, in order to obtain multidimensional Hilbert-type inequalities in such settings.

Hence, before obtaining such multidimensional inequalities, we introduce the notion of a generalized hypergeometric function $_mF_n$, as well as its integral representations.

By a generalized hypergeometric function $_mF_n$ we mean the sum of the series

$${}_{m}F_{n}(a_{1},\ldots,a_{m};b_{1},\ldots,b_{n};z) = \sum_{k=0}^{\infty} \frac{a_{1}^{\overline{k}}a_{2}^{\overline{k}}\ldots a_{\overline{m}}^{\overline{k}}}{b_{1}^{\overline{k}}b_{2}^{\overline{k}}\ldots b_{\overline{n}}^{\overline{k}}} \cdot \frac{z^{k}}{k!}$$

where $a_i^{\overline{k}}, b_i^{\overline{k}}$ are the rising factorial powers and $z \in \mathbb{C}$, in domain of its convergence: $\Omega = \{|z| < \infty\}$ for $m \le n$ and $\Omega = \{|z| < 1\}$ for m = n + 1, or its analytical continuation in $\{|z| > 1, |\arg(1-z)| < \pi\}$, in the latter case. One may also consider z as a real variable $z \in [0, \infty)$.

The paper [50] provides a unified treatment of generalized hypergeometric functions by means of a generalized fractional calculus. More precisely, hypergeometric functions ${}_{m}F_{n}$ are separated into three classes depending on whether m < n, m = n or m = n + 1. Further, hypergeometric functions of each class are represented as generalized fractional integrals or derivatives of three basic elementary functions:

$$\cos_{n-m+1}(z) \ (m < n), \qquad z^{\alpha} \exp z \ (m = n), \qquad z^{\alpha} (1-z)^{\beta} \ (m = n+1).$$

Here, $\cos_t(z)$ is the so-called generalized cosine function of order $t \ge 2$, defined as

$$\cos_t(z) = \sum_{k=0}^{\infty} \frac{(-1)^k z^{kt}}{(kt)!}, \ t \ge 2,$$

where $\cos z = \cos_2(z)$. The above mentioned representations lead to several new integral formulas for ${}_mF_n$ functions and allow their study in a unified way. Moreover, the generalized fractional calculus is developed in [49].

Now, we introduce the Poisson-type integral representations of the above classes of hypergeometric functions $_mF_n$, established in [50].

1° **First case**: m < n. If the conditions

$$b_k > \frac{k}{n-m+1}, \ k = 1, 2, \dots, n-m, \qquad b_{n-m+k} > a_k > 0, \ k = 1, 2, \dots, m,$$
 (2.61)

are fulfilled, then the following Poisson-type integral representation is valid:

$${}_{m}F_{n}(a_{1},...,a_{m};b_{1},...,b_{n};-z) = C \int_{0}^{1}...\int_{0}^{1}\prod_{k=1}^{n-m} \left[\frac{(1-t_{k})^{b_{k}-(k/(n-m+1))-1}}{\Gamma(b_{k}-(k/(n-m+1)))} t_{k}^{(k/(n-m+1))-1} \right] \times \prod_{k=n-m+1}^{n} \left[\frac{(1-t_{k})^{b_{k}-a_{k-n+m}-1}}{\Gamma(b_{k}-a_{k-n+m})} t_{k}^{a_{k-n+m}-1} \right] \times \cos_{n-m+1} \left[(n-m+1)(zt_{1}...t_{n})^{1/(n-m+1)} \right] dt_{1}...dt_{n}.$$
(2.62)

The constant *C* is defined by $C = \sqrt{\frac{n-m+1}{(2\pi)^{n-m}}} \frac{\prod_{j=1}^{n} \Gamma(b_j)}{\prod_{j=1}^{m} \Gamma(a_j)}$.

 2° Second case: m = n. Assuming that

$$b_k > a_k > 0, \ k = 1, 2, \dots, n,$$
 (2.63)

we have

$${}_{n}F_{n}(a_{1},\ldots,a_{n};b_{1},\ldots,b_{n};z) = E \int_{0}^{1} \ldots \int_{0}^{1} \prod_{k=1}^{n} \left[\frac{(1-t_{k})^{b_{k}-a_{k}-1}t_{k}^{a_{k}-1}}{\Gamma(b_{k}-a_{k})} \right] \exp(zt_{1}\ldots t_{n})dt_{1}\ldots dt_{n},$$
(2.64)

where the constant *E* is defined by $E = \prod_{j=1}^{n} \frac{\Gamma(b_j)}{\Gamma(a_j)}$.

 3° Third case: m = n + 1. If the conditions

$$b_k > a_{k+1} > 0, \ k = 1, 2, \dots, n,$$
 (2.65)

are satisfied, then the following Poisson-type integral representation is valid:

$$= M \int_{0}^{1} \dots \int_{0}^{1} \prod_{k=1}^{n} (1 - t_{k})^{b_{k} - a_{k+1} - 1} t_{k}^{a_{k+1} - 1} (1 \mp zt_{1} \dots t_{n})^{-a_{1}} dt_{1} \dots dt_{n}.$$
 (2.66)

The constant *M* is defined by $M = \prod_{j=1}^{n} \frac{\Gamma(b_j)}{\Gamma(a_{j+1})\Gamma(b_j - a_{j+1})}$.

Clearly, the above integral representations are valid for complex numbers z in the corresponding domains of convergence. These representations will be crucial in obtaining examples with multidimensional Hilbert-type inequalities. More precisely, our next step is to find appropriate kernels such that the formulas for the functions F_i , i = 1, 2, ..., n, from Theorem 2.5, reduce to the above stated Poisson-type integral representations.

The kernel involving exponential function

We are going to find a more suitable form of integral representation (2.64). Namely, utilizing substitutions $1 - t_i = \frac{1}{1+x_i}$, i = 1, 2, ..., n, and the well-known relationship between the Beta and the Gamma function, i.e. $B(x,y) = \frac{\Gamma(x)\Gamma(y)}{\Gamma(x+y)}$, x, y > 0, we have

$$\int_{\mathbb{R}^{n}_{+}} \prod_{i=1}^{n} \frac{x_{i}^{a_{i}-1}}{(1+x_{i})^{b_{i}}} \exp\left(x \prod_{i=1}^{n} \frac{x_{i}}{1+x_{i}}\right) dx_{1} dx_{2} \dots dx_{n}$$

= $_{n} F_{n}\left(\mathbf{a}; \mathbf{b}; x\right) \prod_{i=1}^{n} B\left(a_{i}, b_{i} - a_{i}\right),$ (2.67)

where $\mathbf{a} = (a_1, a_2, \dots, a_n)$ and $\mathbf{b} = (b_1, b_2, \dots, b_n)$. However, the coordinates a_i and b_i fulfill conditions as in (2.63).

Considering Theorem 2.5 with the kernel $K : \mathbb{R}^n_+ \to \mathbb{R}$, defined by

$$K(x_1, x_2, \dots, x_n) = \frac{\exp\left(\prod_{i=1}^n \frac{x_i}{1+x_i}\right)}{\prod_{i=1}^n (1+x_i)^{b_i}},$$
(2.68)

and the power weight functions $\varphi_{ij} : \mathbb{R}_+ \to \mathbb{R}$, i, j = 1, 2, ..., n, the above integral representation appears when calculating the constant factors involved in the corresponding inequality.

Theorem 2.9 Let λ , p_i , p'_i , q_i , i = 1, 2, ..., n, $n \ge 2$, be real parameters fulfilling (2.26), (2.27), and (2.30), and let A_{ij} , i, j = 1, ..., n, be real parameters such that $\sum_{i=1}^{n} A_{ij} = 0$ for j = 1, ..., n. If $\beta' = \prod_{i,j=1, i \ne j}^{n} B^{\frac{1}{q_i}} (1 + q_i A_{ij}, b_j - 1 - q_i A_{ij})$ and $f_i : \mathbb{R}_+ \to \mathbb{R}$, i = 1, ..., n, are non-negative measurable functions, then the inequalities

$$\int_{\mathbb{R}^{n}_{+}} \frac{\prod_{i=1}^{n} f_{i}(x_{i})}{\prod_{i=1}^{n} (1+x_{i})^{\lambda b_{i}}} \exp^{\lambda} \left(\prod_{i=1}^{n} \frac{x_{i}}{1+x_{i}}\right) dx_{1} dx_{2} \dots dx_{n} \\
\leq \beta' \prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{p_{i} A_{ii}} (1+x_{i})^{(1-\lambda)p_{i}b_{i}-b_{i}} \times_{n-1} F_{n-1}^{1-(1-\lambda)p_{i}} \left(\mathbf{1}+q_{i} \mathbf{A}_{i}; \mathbf{b}_{i}; \frac{x_{i}}{1+x_{i}}\right) f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}} (2.69)$$

$$\begin{cases} \int_{0}^{\infty} x_{n}^{-p_{n}'A_{nn}} (1+x_{n})^{b_{n}(\lambda p_{n}'-1)}_{n-1} F_{n-1}^{1-\lambda p_{n}'} \left(\mathbf{1}+q_{n}\mathbf{A_{n}}; \mathbf{b_{n}}; \frac{x_{n}}{1+x_{n}}\right) \\ \times \left[\int_{\mathbb{R}^{n-1}_{+}} \frac{\prod_{i=1}^{n-1} f_{i}(x_{i})}{\prod_{i=1}^{n} (1+x_{i})^{\lambda b_{i}}} \exp^{\lambda} \left(\prod_{i=1}^{n} \frac{x_{i}}{1+x_{i}}\right) dx_{1} dx_{2} \dots dx_{n-1} \right]^{p_{n}'} dx_{n} \end{cases}^{\frac{1}{p_{n}'}} \\ \leq \beta' \prod_{i=1}^{n-1} \left[\int_{0}^{\infty} x_{i}^{p_{i}A_{ii}} (1+x_{i})^{(1-\lambda)p_{i}b_{i}-b_{i}} \right]^{\frac{1}{p_{i}}} \\ \times_{n-1}F_{n-1}^{1-(1-\lambda)p_{i}} \left(\mathbf{1}+q_{i}\mathbf{A_{i}}; \mathbf{b_{i}}; \frac{x_{i}}{1+x_{i}}\right) f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$
(2.70)

hold for all parameters A_{ij} , $i \neq j$, such that $q_iA_{ij} \in (-1, b_j - 1)$, where $\mathbf{1} + q_i\mathbf{A}_i = (1 + q_iA_{i1}, 1 + q_iA_{i2}, \dots, 1 + q_iA_{i,i-1}, 1 + q_iA_{i,i+1}, \dots, 1 + q_iA_{in})$ and $\mathbf{b}_i = (b_1, b_2, \dots, b_{i-1}, b_{i+1}, \dots, b_n)$. Moreover, these inequalities are equivalent.

Proof. The proof is based on a simple use of Theorem 2.5. Taking into account notation from Theorem 2.5, as well as considering the kernel defined by (2.68) and the power weight functions $\phi_{ij}(x_j) = x_j^{A_{ij}}$, i, j = 1, 2, ..., n, we have

$$F_{i}(x_{i}) = (1+x_{i})^{-\frac{b_{i}}{q_{i}}} \left[\int_{\mathbb{R}^{n-1}_{+}} \prod_{\substack{j=1\\j\neq i}}^{n} \frac{x_{j}^{(1+q_{i}A_{ij})-1}}{(1+x_{j})^{b_{j}}} \exp\left(\frac{x_{i}}{1+x_{i}} \prod_{\substack{j=1\\j\neq i}}^{n} \frac{x_{j}}{1+x_{j}}\right) \right]$$

$$\times dx_{1} \dots dx_{i-1} dx_{i+1} \dots dx_{n} = \frac{1}{q_{i}}.$$

Clearly, the above power functions are well-defined, that is, $\prod_{i,j=1}^{n} \phi_{ij}(x_j) = 1$, since $\sum_{i=1}^{n} A_{ij} = 0$ for j = 1, ..., n. Now, exploiting integral representation (2.67), we have

$$F_{i}(x_{i}) = (1+x_{i})^{-\frac{b_{i}}{q_{i}}} \beta_{n-1} F_{n-1}^{\frac{1}{q_{i}}} \left(1+q_{i} \mathbf{A}_{i}; \mathbf{b}_{i}; \frac{x_{i}}{1+x_{i}} \right),$$

that is, after substituting the expressions for $F_i(x_i)$, i = 1, 2, ..., n, in (2.31) and (2.32) (see Theorem 2.5), we obtain desired inequalities. It should be noticed here that parameters A_{ij} , $i \neq j$, fulfill conditions $q_i A_{ij} \in (-1, b_j - 1)$, i, j = 1, 2, ..., n, since the arguments of the Beta function are positive.

Remark 2.20 Considering some particular values of parameters A_{ij} , i, j = 1, 2, ..., n, we can simplify the constant β' from Theorem 2.9. For example, taking the arithmetic mean of the borders of intervals defining parameters A_{ij} , $i \neq j$, we have $A_{ij} = \frac{b_j - 2}{2q_i}$, hence the constant factor becomes $\beta' = \prod_{i=1}^{n} B^{\frac{1}{p_i'}} \left(\frac{b_i}{2}, \frac{b_i}{2}\right)$. In that case, the parameters A_{ii} , i = 1, 2, ..., n, are defined by $A_{ii} = -\frac{b_j - 2}{2p_i'}$.

Since $0 < \frac{t}{1+t} < 1$, $t \in \mathbb{R}_+$, we have that $_{n-1}F_{n-1}\left(\mathbf{1}+q_i\mathbf{A_i}; \mathbf{b_i}; \frac{x_i}{1+x_i}\right) < _{n-1}F_{n-1}$ $(\mathbf{1}+q_i\mathbf{A_i}; \mathbf{b_i}; 1)$. Applying this estimate to inequalities (2.69) and (2.70), we obtain the constant factor expressed in terms of the hypergeometric function $_{n-1}F_{n-1}$.

Corollary 2.6 Under the assumptions of Theorem 2.9, the inequalities

$$\int_{\mathbb{R}^{n}_{+}} \frac{\prod_{i=1}^{n} f_{i}(x_{i})}{\prod_{i=1}^{n} (1+x_{i})^{\lambda b_{i}}} \exp^{\lambda} \left(\prod_{i=1}^{n} \frac{x_{i}}{1+x_{i}}\right) dx_{1} dx_{2} \dots dx_{n}$$

$$\leq \beta'_{H} \prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{p_{i}A_{ii}} (1+x_{i})^{(1-\lambda)p_{i}b_{i}-b_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i}\right]^{\frac{1}{p_{i}}}$$
(2.71)

and

$$\begin{cases} \int_{0}^{\infty} x_{n}^{-p_{n}'A_{nn}} (1+x_{n})^{b_{n}(\lambda p_{n}'-1)} \\ \times \left[\int_{\mathbb{R}^{n-1}_{+}} \frac{\prod_{i=1}^{n-1} f_{i}(x_{i})}{\prod_{i=1}^{n} (1+x_{i})^{\lambda b_{i}}} \exp^{\lambda} \left(\prod_{i=1}^{n} \frac{x_{i}}{1+x_{i}} \right) dx_{1} dx_{2} \dots dx_{n-1} \right]^{p_{n}'} dx_{n} \end{cases}^{\frac{1}{p_{n}'}} \\ \leq \beta_{H}' \prod_{i=1}^{n-1} \left[\int_{0}^{\infty} x_{i}^{p_{i}A_{ii}} (1+x_{i})^{(1-\lambda)p_{i}b_{i}-b_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$
(2.72)

hold and are equivalent, where

$$\beta'_{H} = \beta' \prod_{i=1}^{n} {}_{n-1}F_{n-1}^{\frac{1}{q_{i}}} \left(\mathbf{1} + q_{i}\mathbf{A_{i}}; \mathbf{b_{i}}; 1 \right)$$
(2.73)

and β' is defined in Theorem 2.9.

It remains to investigate the other two integral representations (2.62) and (2.66). We use the same procedure as for the kernel involving the exponential function.

The kernel involving cosine function

We consider here integral representation (2.62) for m = n - 1. Applying the substitutions $1 - t_i = \frac{1}{1+x_i}$, i = 1, 2, ..., n, this representation can be expressed in a more suitable form

$$\int_{\mathbb{R}^{n}_{+}} \prod_{i=1}^{n} \frac{x_{i}^{a_{i-1}-1}}{(1+x_{i})^{b_{i}}} \cos\left[2\left(x\prod_{i=1}^{n} \frac{x_{i}}{1+x_{i}}\right)^{\frac{1}{2}}\right] dx_{1} dx_{2} \dots dx_{n}$$
$$= \left(\pi + \frac{1}{2}\right)^{-\frac{1}{2}}{}_{n-1}F_{n}\left(\mathbf{a};\mathbf{b};-x\right)B\left(b_{1} - \frac{1}{2},\frac{1}{2}\right)\prod_{i=2}^{n}B\left(a_{i-1},b_{i} - a_{i-1}\right), \quad (2.74)$$

where $\mathbf{a} = (a_1, a_2, \dots, a_{n-1})$, $\mathbf{b} = (b_1, b_2, \dots, b_n)$ and $a_0 = \frac{1}{2}$. Of course, the coordinates of vectors \mathbf{a} and \mathbf{b} fulfill conditions as in (2.61).

The previous integral formula will be needed when applying Theorem 2.5 to the kernel $K : \mathbb{R}^n_+ \to \mathbb{R}$ defined by

$$K(x_1, x_2, \dots, x_n) = \frac{\cos\left(\prod_{i=1}^n \frac{x_i}{1+x_i}\right)^{\frac{1}{2}}}{\prod_{i=1}^n (1+x_i)^{b_i}},$$
(2.75)

and the power weight functions $\varphi_{ij} : \mathbb{R}_+ \to \mathbb{R}, i, j = 1, 2, ..., n$.

Theorem 2.10 Let λ , p_i , p'_i , q_i , i = 1, 2, ..., n, $n \ge 2$, be as in (2.26), (2.27) and (2.30), and let A_{ij} , i, j = 1, ..., n, be real parameters such that $\sum_{i=1}^{n} A_{ij} = 0$ for j = 1, ..., n. If $\gamma' = (\pi + \frac{1}{2})^{-\frac{\lambda}{2}} \beta' \prod_{i=1}^{n} B^{\frac{1}{q_i}} (b_{i+1} - \frac{1}{2}, \frac{1}{2}) (\beta' \text{ is defined in Theorem 2.9})$, and $f_i : \mathbb{R}_+ \to \mathbb{R}$, i = 1, ..., n, are non-negative measurable functions, then the inequalities

$$\int_{\mathbb{R}^{n}_{+}} \frac{\prod_{i=1}^{n} f_{i}(x_{i})}{\prod_{i=1}^{n} (1+x_{i})^{\lambda b_{i}}} \cos^{\lambda} \left(\prod_{i=1}^{n} \frac{x_{i}}{1+x_{i}}\right)^{\frac{1}{2}} dx_{1} dx_{2} \dots dx_{n}$$

$$\leq \gamma' \prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{p_{i}A_{ii}} (1+x_{i})^{(1-\lambda)p_{i}b_{i}-b_{i}} \times_{n-2} F_{n-1}^{1-(1-\lambda)p_{i}} \left(\mathbf{1}+q_{i}\mathbf{A}_{i,i+1}; \mathbf{b}_{i}; \frac{x_{i}}{4(1+x_{i})}\right) f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$
(2.76)

and

$$\begin{cases} \int_{0}^{\infty} x_{n}^{-p_{n}'A_{nn}} (1+x_{n})^{b_{n}(\lambda p_{n}'-1)}_{n-2} F_{n-1}^{1-\lambda p_{n}'} \left(\mathbf{1}+q_{n}\mathbf{A_{n1}}; \mathbf{b_{n}}; \frac{x_{n}}{4(1+x_{n})}\right) \\ \times \left[\int_{\mathbb{R}^{n-1}_{+}} \frac{\prod_{i=1}^{n-1} f_{i}(x_{i})}{\prod_{i=1}^{n} (1+x_{i})^{\lambda b_{i}}} \cos^{\lambda} \left(\prod_{i=1}^{n} \frac{x_{i}}{1+x_{i}}\right)^{\frac{1}{2}} dx_{1} dx_{2} \dots dx_{n-1} \right]^{p_{n}'} dx_{n} \end{cases}^{\frac{1}{p_{n}'}} \\ \leq \gamma' \prod_{i=1}^{n-1} \left[\int_{0}^{\infty} x_{i}^{p_{i}A_{ii}} (1+x_{i})^{(1-\lambda)p_{i}b_{i}-b_{i}} \right]^{\frac{1}{p_{i}}} \times (1+x_{i})^{n-1} \left(\mathbf{1}+q_{i}\mathbf{A_{i,i+1}}; \mathbf{b_{i}}; \frac{x_{i}}{4(1+x_{i})}\right) f_{i}^{p_{i}}(x_{i}) dx_{i} \end{cases}^{\frac{1}{p_{i}}}$$
(2.77)

hold for all parameters A_{ij} , i, j = 1, 2, ..., n, and b_i , i = 1, 2, ..., n, such that $q_i A_{ij} \in (-1, b_j - 1)$, $j - i \notin \{0, 1, 1 - n\}$, $q_i A_{i,i+1} = -\frac{1}{2}$, $b_i > \frac{1}{2}$, i = 1, 2, ..., n, and where $\mathbf{1} + q_i \mathbf{A}_{i,i+1} = (1 + q_i A_{i,1}, ..., 1 + q_i A_{i,i-1}, 1 + q_i A_{i,i+2}, ..., 1 + q_i A_{in})$, $\mathbf{b_i} = (b_1, b_2, ..., b_{i-1}, b_{i+1}, ..., b_n)$. Moreover, these inequalities are equivalent.

Proof. Considering Theorem 2.5 with the kernel defined by (2.75) and the functions

 $\phi_{ij}(x_j) = x_j^{A_{ij}}, i, j = 1, 2, ..., n$, we have

$$F_{i}(x_{i}) = (1+x_{i})^{-\frac{b_{i}}{q_{i}}} \left[\int_{\mathbb{R}^{n-1}_{+}} \frac{x_{i+1}^{(1+q_{i}A_{i,i+1})-1}}{(1+x_{i+1})^{b_{i+1}}} \prod_{\substack{j=1\\j\neq i,i+1}}^{n} \frac{x_{j}^{(1+q_{i}A_{ij})-1}}{(1+x_{j})^{b_{j}}} \right] \\ \times \cos \left[2 \left(\frac{x_{i}}{4(1+x_{i})} \prod_{\substack{j=1\\j\neq i}}^{n} \frac{x_{j}}{1+x_{j}} \right)^{\frac{1}{2}} \right] dx_{1} \dots dx_{i-1} dx_{i+1} \dots dx_{n} \right]^{\frac{1}{q_{i}}}.$$

Hence, the above integral representation (2.74) yields

$$F_{i}(x_{i}) = (1+x_{i})^{-\frac{b_{i}}{q_{i}}}\beta'\left(\pi+\frac{1}{2}\right)^{-\frac{1}{2q_{i}}}B^{\frac{1}{q_{i}}}\left(b_{i+1}-\frac{1}{2},\frac{1}{2}\right)$$
$$\times_{n-2}F_{n-1}^{\frac{1}{q_{i}}}\left(1+q_{i}\mathbf{A}_{\mathbf{i},\mathbf{i}+1};\mathbf{b}_{\mathbf{i}};\frac{x_{i}}{4(1+x_{i})}\right),$$

where β' is defined in Theorem 2.9, and the result follows. Note also that we assume congruence modulo *n* for the parameters A_{ij} , i.e. $A_{n,n+1} = A_{n1}$.

Remark 2.21 Taking into account the obvious estimate

$$_{n-2}F_{n-1}\left(\mathbf{1}+q_{i}\mathbf{A}_{i,i+1};\mathbf{b}_{i};\frac{x_{i}}{4(1+x_{i})}\right) < _{n-2}F_{n-1}\left(\mathbf{1}+q_{i}\mathbf{A}_{i,i+1};\mathbf{b}_{i};1\right),$$

which holds for all $x_i \in \mathbb{R}_+$, we also obtain the inequalities as those from Corollary 2.6, with the kernel (2.75) and the corresponding constant factor

$$\gamma'\prod_{i=1}^{n} \sum_{n-2}F_{n-1}^{\frac{1}{q_i}}\left(\mathbf{1}+q_i\mathbf{A_{i,i+1}};\mathbf{b_i};1\right),$$

where γ' is defined in Theorem 2.10.

Fractional kernel

It remains to consider the remaining case, that is, the integral representation (2.66). It can be rewritten in a more convenient form, that is,

$$\int_{\mathbb{R}^{n}_{+}} \prod_{i=1}^{n} \frac{x_{i}^{a_{i+1}-1}}{(1+x_{i})^{b_{i}}} \left(1 - x \prod_{i=1}^{n} \frac{x_{i}}{1+x_{i}}\right)^{-a_{1}} dx_{1} dx_{2} \dots dx_{n}$$
$$= {}_{n+1}F_{n}\left(\mathbf{a};\mathbf{b};x\right) \prod_{i=1}^{n} B\left(a_{i+1}, b_{i} - a_{i+1}\right), \qquad (2.78)$$

where $\mathbf{a} = (a_1, a_2, \dots, a_{n+1})$ and $\mathbf{b} = (b_1, b_2, \dots, b_n)$ are such that $b_i > a_{i+1} > 0$, $i = 1, 2, \dots, n$.

The previous integral representation is essential when considering Theorem 2.5 with the fractional kernel $K : \mathbb{R}^n_+ \to \mathbb{R}$, defined by

$$K(x_1, x_2, \dots, x_n) = \frac{\left(1 - \prod_{i=1}^n \frac{x_i}{1 + x_i}\right)^{-s}}{\prod_{i=1}^n (1 + x_i)^{b_i}}, \quad s > 0,$$
(2.79)

and the power weight functions.

Theorem 2.11 Let λ , p_i , p'_i , q_i , i = 1, 2, ..., n, $n \ge 2$, be as in (2.26), (2.27) and (2.30), and let A_{ij} , i, j = 1, ..., n, be real parameters such that $\sum_{i=1}^{n} A_{ij} = 0$ for j = 1, ..., n. If the constant β' is as in Theorem 2.9 and $f_i : \mathbb{R}_+ \to \mathbb{R}$, i = 1, ..., n, are non-negative measurable functions, then the inequalities

$$\int_{\mathbb{R}^{n}_{+}} \frac{\prod_{i=1}^{n} f_{i}(x_{i})}{\prod_{i=1}^{n} (1+x_{i})^{\lambda b_{i}}} \left(1 - \prod_{i=1}^{n} \frac{x_{i}}{1+x_{i}}\right)^{-\lambda s} dx_{1} dx_{2} \dots dx_{n}$$

$$\leq \beta' \prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{p_{i}A_{ii}} (1+x_{i})^{(1-\lambda)p_{i}b_{i}-b_{i}} \times_{n} F_{n-1}^{1-(1-\lambda)p_{i}} \left(s, 1+q_{i}\mathbf{A}_{i}; \mathbf{b}_{i}; \frac{x_{i}}{1+x_{i}}\right) dx_{i}\right]^{\frac{1}{p_{i}}}$$
(2.80)

and

$$\begin{cases} \int_{0}^{\infty} x_{n}^{-p_{n}'A_{nn}} (1+x_{n})^{b_{n}(\lambda p_{n}'-1)} {}_{n}F_{n-1}^{1-\lambda p_{n}'} \left(s,\mathbf{1}+q_{n}\mathbf{A_{n}};\mathbf{b_{n}};\frac{x_{n}}{1+x_{n}}\right) \\ \times \left[\int_{\mathbb{R}^{n-1}_{+}} \frac{\prod_{i=1}^{n-1} f_{i}(x_{i})}{\prod_{i=1}^{n} (1+x_{i})^{\lambda b_{i}}} \left(1-\prod_{i=1}^{n} \frac{x_{i}}{1+x_{i}}\right)^{-\lambda s} dx_{1} dx_{2} \dots dx_{n-1}\right]^{p_{n}'} dx_{n} \end{cases} \overset{1}{\xrightarrow{p_{n}'}} \\ \leq \beta' \prod_{i=1}^{n-1} \left[\int_{0}^{\infty} x_{i}^{p_{i}A_{ii}} (1+x_{i})^{(1-\lambda)p_{i}b_{i}-b_{i}} \\ \times {}_{n}F_{n-1}^{1-(1-\lambda)p_{i}} \left(s,\mathbf{1}+q_{i}\mathbf{A_{i}};\mathbf{b_{i}};\frac{x_{i}}{1+x_{i}}\right) dx_{i}\right]^{\frac{1}{p_{i}}} \tag{2.81}$$

hold for all parameters A_{ij} , $i \neq j$, such that $q_i A_{ij} \in (-1, b_j - 1)$, i, j = 1, 2, ..., n, where $\mathbf{1} + q_i \mathbf{A_i} = (1 + q_i A_{i1}, 1 + q_i A_{i2}, ..., 1 + q_i A_{i,i-1}, 1 + q_i A_{i,i+1}, ..., 1 + q_i A_{in})$ and $\mathbf{b_i} = (b_1, b_2, ..., b_{i-1}, b_{i+1}, ..., b_n)$. Moreover, these inequalities are equivalent.

Proof. Applying Theorem 2.5 with the fractional kernel (2.79) and the power weight functions $\phi_{ij}(x_j) = x_j^{A_{ij}}$, i, j = 1, 2, ..., n, and taking into account the integral representation (2.78), we have that

$$F_i(x_i) = (1+x_i)^{-\frac{b_i}{q_i}} \beta_n F_{n-1}^{\frac{1}{q_i}} \left(s, \mathbf{1}+q_i \mathbf{A_i}; \mathbf{b_i}; \frac{x_i}{1+x_i} \right),$$

so the result follows.

Remark 2.22 It is obvious that the estimate

$$_{n}F_{n-1}\left(s,\mathbf{1}+q_{i}\mathbf{A_{i}};\mathbf{b_{i}};\frac{x_{i}}{1+x_{i}}\right) < _{n}F_{n-1}\left(s,\mathbf{1}+q_{i}\mathbf{A_{i}};\mathbf{b_{i}};1\right)$$

holds for all $x_i \in \mathbb{R}_+$. Therefore, we also obtain the inequalities as those from Corollary 2.6, with the fractional kernel (2.79) and the corresponding constant factor $\beta' \prod_{i=1}^{n} {}_{n}F_{n-1} \frac{1}{q_i} (s, 1+q_i\mathbf{A_i}; \mathbf{b_i}; 1)$, where β' is defined in Theorem 2.9.

Remark 2.23 Regarding Remark 2.11, it follows that the equality in any of inequalities from this subsection is possible if and only if at least one of the functions $f_i : \mathbb{R}_+ \to \mathbb{R}$, i = 1, ..., n, is the zero function. Namely, the equality is possible only for the kernels with separated variables, which is not the case here.

Remark 2.24 Hilbert-type inequalities from this section related to Gaussian hypergeometric function are derived in [62], while the multidimensional extension via the Poisson-type integral representations is developed in [60].

2.6 Hilbert-type inequalities and related operators

So far, we have considered Hilbert-type inequalities with homogeneous kernels of negative degree of homogeneity. The reason for this was in the fact that we required the kernel to be decreasing in both arguments (see Remark 1.6, Chapter 1). Such requirement was essential in order to obtain some estimates when considering inequalities with integrals taken over bounded intervals in \mathbb{R}_+ . However, when considering the integrals taken over \mathbb{R}_+ , such requirement was redundant.

On the other hand, assuming the convergence, Hilbert-type inequalities with integrals taken over the set \mathbb{R}_+ can also be considered for homogeneous kernels of zero-degree.

In this section we shall be more concerned with an equivalent form of the Hilbert-type inequality, that is, with the Hardy-Hilbert-type inequality. Namely, the Hardy-Hilbert form of inequality provides the possibility of defining certain integral operators between the weighted Lebesgue spaces and determining their norms in some particular cases.

The previous program will be carried out for the Hardy-Hilbert-type inequalities including a homogeneous kernel with zero-degree of homogeneity. First, we are going to derive appropriate inequalities, and then, to consider the related operators between the weighted Lebesgue spaces, which naturally arise from these inequalities.

2.6.1 Hilbert-type inequalities involving a homogeneous function of zero-degree

In this subsection we give a unified treatment of Hilbert-type inequalities with homogeneous kernels of zero-degree.

The results that follow are considered in the setting with non-conjugate exponents, in a slightly generalized form. More precisely, the kernel includes two differentiable functions with some additional properties. We start with some definitions and notation that will be valid throughout this section.

Let (a,b) be an interval on the real line and let $u,v:(a,b) \to \mathbb{R}$ be non-negative measurable functions satisfying the following conditions:

- (i) u and v are differentiable on (a, b);
- (ii) u and v are strictly increasing on (a,b);
- (iii) $\lim_{x \to a^+} u(t) = v(t) = 0$ and $\lim_{x \to b^-} u(t) = v(t) = \infty$.

In this section, by $k_0 : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ we denote a non-negative measurable homogeneous function of zero-degree. Also, we deal with the integral

$$c_0(\alpha) = \int_0^\infty k_0(1,t) t^{-\alpha} dt.$$
 (2.82)

We consider only the parameters α such that (2.82) converges.

The above functions u, v and k_0 will be essential in defining the corresponding kernel. More precisely, $K_0: (a,b) \times (a,b) \rightarrow \mathbb{R}$ denotes a non-negative measurable function defined by

$$K_0(x,y) = k_0(u(x), v(y)), \qquad (2.83)$$

where *u* and *v* are assumed to fulfill conditions (i)-(iii).

First we provide Hilbert-type and Hardy-Hilbert-type inequalities involving the above kernel K_0 .

Theorem 2.12 Let p, q, and λ be as in (2.1) and (2.2), and let $u, v : (a,b) \to \mathbb{R}$ be nonnegative measurable functions fulfilling conditions (i)-(iii). If $K_0 : (a,b) \times (a,b) \to \mathbb{R}$ is a non-negative measurable function defined by (2.83), and A_1 , A_2 are real parameters such that $c_0(2 - p'A_1) < \infty$, $c_0(q'A_2) < \infty$, then the inequalities

$$\int_{a}^{b} \int_{a}^{b} K_{0}^{\lambda}(x,y) f(x)g(y) \, dx \, dy \leq c_{0}^{\frac{1}{p'}} (2 - p'A_{1}) c_{0}^{\frac{1}{q'}}(q'A_{2}) \\ \times \left[\int_{a}^{b} \frac{u^{(A_{1} - A_{2})p + \frac{p}{q'}}(x)}{(u')^{p-1}(x)} f^{p}(x) \, dx \right]^{\frac{1}{p}} \left[\int_{a}^{b} \frac{v^{(A_{2} - A_{1})q + \frac{q}{p'}}(y)}{(v')^{q-1}(y)} g^{q}(y) \, dy \right]^{\frac{1}{q}}$$
(2.84)

and

$$\left\{ \int_{a}^{b} \frac{v'(y)}{v^{(A_{2}-A_{1})q'+\frac{q'}{p'}(y)}} \left[\int_{a}^{b} K_{0}^{\lambda}(x,y)f(x) \, dx \right]^{q'} \, dy \right\}^{\frac{1}{q'}} \\
\leq c_{0}^{\frac{1}{p'}}(2-p'A_{1})c_{0}^{\frac{1}{q'}}(q'A_{2}) \left[\int_{a}^{b} \frac{u^{(A_{1}-A_{2})p+\frac{p}{q'}}(x)}{(u')^{p-1}(x)} f^{p}(x) \, dx \right]^{\frac{1}{p}}$$
(2.85)

hold for all non-negative measurable functions $f, g: (a,b) \to \mathbb{R}$ and are equivalent. Equalities in (2.84) and (2.85) hold if and only if f = 0 or g = 0 a.e. on (a,b).

Proof. Similarly as for homogeneous kernels of negative degree, this theorem is also a consequence of Theorem 2.1. Namely, using notation from Theorem 2.1, we put the kernel K_0 , defined by (2.83), and the weight functions

$$\varphi(x) = \frac{u^{A_1}(x)}{(u')^{\frac{1}{p'}}(x)}, \qquad \psi(y) = \frac{v^{A_2}(y)}{(v')^{\frac{1}{q'}}(y)}, \qquad x \in (a,b),$$

in inequalities (2.9) and (2.10). Then, defining v(y) = tu(x) and using the change of variable, we get

$$F(x) = \left[\int_{a}^{b} k_{0}(u(x), v(y)) v^{-q'A_{2}}(y) v'(y) dy \right]^{\frac{1}{q'}}$$

= $u^{\frac{1}{q'} - A_{2}}(x) \left[\int_{0}^{\infty} k_{0}(1, t) t^{-q'A_{2}} dt \right]^{\frac{1}{q'}} = c_{0}^{\frac{1}{q'}}(q'A_{2}) u^{\frac{1}{q'} - A_{2}}(x), \quad x \in (a, b).$

Using the same argument, we also have

$$G(y) = c_0^{\frac{1}{p'}} (2 - p'A_1) v^{\frac{1}{p'} - A_1}(x), \quad y \in (a, b),$$

which proves relations (2.84) and (2.85).

The case of equality follows immediately from Remark 2.3.

We continue with some basic examples of functions u and v, fulfilling conditions (i)-(iii). For example, functions $u, v : \mathbb{R}_+ \to \mathbb{R}$, defined by $u(x) = Ax^{\mu}$, $v(y) = By^{\nu}$, where $A, B, \mu, \nu > 0$, fulfill the above mentioned conditions, hence the following result is a direct consequence of Theorem 2.12.

Corollary 2.7 Suppose p, q, and λ are as in (2.1) and (2.2), and $k_0 : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ is a non-negative measurable homogeneous function of zero-degree. Let A_1 , A_2 be real parameters such that $c_0(2 - p'A_1) < \infty$ and $c_0(q'A_2) < \infty$, and let $A, B, \mu, \nu > 0$. Then the inequalities

$$\int_{0}^{\infty} \int_{0}^{\infty} k_{0}^{\lambda} (Ax^{\mu}, By^{\nu}) f(x) g(y) \, dx \, dy$$

$$\leq c_{0}^{\frac{1}{p'}} (2 - p'A_{1}) c_{0}^{\frac{1}{q'}} (q'A_{2}) C_{0} \left[\int_{0}^{\infty} x^{\left(A_{1} - A_{2} + \frac{1}{q'}\right) p\mu + (p-1)(1-\mu)} f^{p}(x) \, dx \right]^{\frac{1}{p}}$$

$$\times \left[\int_{0}^{\infty} y^{\left(A_{2} - A_{1} + \frac{1}{p'}\right) q\nu + (q-1)(1-\nu)} g^{q}(y) \, dy \right]^{\frac{1}{q}}$$
(2.86)

$$\begin{cases} \int_{0}^{\infty} y^{\left(A_{1}-A_{2}-\frac{1}{p'}\right)q'\nu+\nu-1} \left[\int_{0}^{\infty} k_{0}^{\lambda}(Ax^{\mu}, By^{\nu})f(x)\,dx\right]^{q'}\,dy \end{cases}^{\frac{1}{q'}} \\ \leq c_{0}^{\frac{1}{p'}}(2-p'A_{1})c_{0}^{\frac{1}{q'}}(q'A_{2})C_{0}\left[\int_{0}^{\infty} x^{\left(A_{1}-A_{2}+\frac{1}{q'}\right)p\mu+(p-1)(1-\mu)}f^{p}(x)\,dx\right]^{\frac{1}{p}} (2.87) \end{cases}$$

hold for all non-negative measurable functions $f, g : \mathbb{R}_+ \to \mathbb{R}$, where

$$C_0 = \mu^{-\frac{1}{p'}} v^{-\frac{1}{q'}} A^{A_1 - A_2 + \frac{1}{q'} - \frac{1}{p'}} B^{A_2 - A_1 + \frac{1}{p'} - \frac{1}{q'}}.$$

Moreover, these inequalities are equivalent. In addition, equalities in (2.86) *and* (2.87) *hold if and only if* f = 0 *or* g = 0 *a.e. on* \mathbb{R}_+ .

In the sequel we consider another interesting kernel dependent on homogeneous function $k_0 : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ of zero-degree. Namely, let $\widetilde{K}_0 : (a,b) \times (a,b) \to \mathbb{R}$ be a nonnegative measurable function defined by

$$K_0(x,y) = k_0(1,u(x)v(y)), \qquad (2.88)$$

where the functions u and v fulfill conditions (i)-(iii). The following result is an analogue of Theorem 2.12.

Theorem 2.13 Let p, q, and λ be as in (2.1) and (2.2), and let $u, v : (a, b) \to \mathbb{R}$ be nonnegative measurable functions fulfilling conditions (i)-(iii). Further, suppose $\widetilde{K}_0 : (a, b) \times (a, b) \to \mathbb{R}$ is a non-negative measurable function defined by (2.88), and A_1 , A_2 are real parameters such that $c_0(p'A_1) < \infty$, $c_0(q'A_2) < \infty$. Then the inequalities

$$\int_{a}^{b} \int_{a}^{b} \widetilde{K}_{0}^{\lambda}(x,y) f(x)g(y) \, dx \, dy \leq c_{0}^{\frac{1}{p'}}(p'A_{1}) c_{0}^{\frac{1}{q'}}(q'A_{2}) \\ \times \left[\int_{a}^{b} \frac{u^{(A_{1}+A_{2})p-\frac{p}{q'}}(x)}{(u')^{p-1}(x)} f^{p}(x) \, dx \right]^{\frac{1}{p}} \left[\int_{a}^{b} \frac{v^{(A_{1}+A_{2})q-\frac{q}{p'}}(y)}{(v')^{q-1}(y)} g^{q}(y) \, dy \right]^{\frac{1}{q}} (2.89)$$

and

$$\left\{ \int_{a}^{b} \frac{v'(y)}{v^{(A_{1}+A_{2})q'-\frac{q'}{p'}(y)}} \left[\int_{a}^{b} \widetilde{K}_{0}^{\lambda}(x,y)f(x) \, dx \right]^{q'} \, dy \right\}^{\frac{1}{q'}} \\
\leq c_{0}^{\frac{1}{p'}}(p'A_{1})c_{0}^{\frac{1}{q'}}(q'A_{2}) \left[\int_{a}^{b} \frac{u^{(A_{1}+A_{2})p-\frac{p}{q'}}(x)}{(u')^{p-1}(x)} f^{p}(x) \, dx \right]^{\frac{1}{p}}$$
(2.90)

hold for all non-negative measurable functions $f, g: (a,b) \to \mathbb{R}$ and are equivalent. Equalities in (2.89) and (2.90) hold if and only if f = 0 or g = 0 a.e. on (a,b).

Proof. It follows directly from Theorem 2.1.

As an application of Theorem 2.13, we consider exponential functions $u, v : \mathbb{R} \to \mathbb{R}$, that is, $u(x) = \exp x$ and $v(y) = \exp y$.

Corollary 2.8 Suppose p, q, and λ are as in (2.1) and (2.2), and $k_0 : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ is a non-negative measurable homogeneous function of zero-degree. If A_1 and A_2 are real parameters such that $c_0(p'A_1) < \infty$ and $c_0(q'A_2) < \infty$, then the inequalities

$$\int_{-\infty}^{\infty} \int_{-\infty}^{\infty} k_0^{\lambda} (1, \exp(x+y)) f(x) g(y) \, dx \, dy \le c_0^{\frac{1}{p'}} (p'A_1) c_0^{\frac{1}{q'}} (q'A_2) \\ \times \left[\int_{-\infty}^{\infty} \exp\left[\left((A_1 + A_2)p + 1 - 2p + \frac{p}{q} \right) x \right] f^p(x) \, dx \right]^{\frac{1}{p}} \\ \times \left[\int_{-\infty}^{\infty} \exp\left[\left((A_1 + A_2)q + 1 - 2q + \frac{q}{p} \right) y \right] g^q(y) \, dy \right]^{\frac{1}{q}}$$
(2.91)

and

$$\left\{ \int_{-\infty}^{\infty} \exp\left[\left(1 - (A_1 + A_2)q' + \frac{q'}{p'} \right) y \right] \left[\int_{-\infty}^{\infty} k_0^{\lambda} (1, \exp(x + y)) f(x) \, dx \right]^{q'} \, dy \right\}^{\frac{1}{q'}} \\
\leq c_0^{\frac{1}{p'}} (p'A_1) c_0^{\frac{1}{q'}} (q'A_2) \\
\times \left[\int_{-\infty}^{\infty} \exp\left[\left((A_1 + A_2)p + 1 - 2p + \frac{p}{q} \right) x \right] f^p(x) \, dx \right]^{\frac{1}{p}}$$
(2.92)

hold for all non-negative measurable functions $f, g : \mathbb{R} \to \mathbb{R}$ and are equivalent. Equalities in (2.91) and (2.92) hold if and only if f = 0 or g = 0 a.e on \mathbb{R} .

2.6.2 On some related Hilbert-type operators

The Hardy-Hilbert-type inequalities (2.85) and (2.90) allow us precise definition of Hilbert-type operators and some conclusions about their norms as well. We consider here the weighted Lebesgue space $L_w^r(a,b)$ consisting of all measurable functions $f:(a,b) \to \mathbb{R}$ with a finite norm $||f||_{L_w^r(a,b)} = \left[\int_a^b w(x)|f^r(x)|\,dx\right]^{\frac{1}{r}}$. Here, r > 1 and $w:(a,b) \to \mathbb{R}$ is a non-negative measurable weight function.

Hence, in the setting with p, q and λ as in (2.1) and (2.2), the kernel K_0 defined by (2.83), and the weight functions

$$\Phi(x) = \frac{u^{(A_1 - A_2)p + \frac{p}{q'}}(x)}{(u')^{p-1}(x)} \text{ and } \Psi(y) = \frac{v^{(A_2 - A_1)q + \frac{q}{p'}}(y)}{(v')^{q-1}(y)}, \quad x, y \in (a, b),$$

we can define the operator $\mathscr{T}_0: L^p_{\Phi}(a,b) \to L^{q'}_{\Psi^{1-q'}}(a,b)$ as

$$\left(\mathscr{T}_0 f\right)(y) = \int_a^b K_0^\lambda(x, y) f(x) \, dx, \quad y \in (a, b).$$
(2.93)

Clearly, the operator \mathscr{T}_0 is well-defined since inequality (2.85) implies that $\mathscr{T}_0 f \in L^{q'}_{\Psi^{1-q'}}(a,b)$. Moreover, considering the norm of operator \mathscr{T}_0 , that is,

$$\|\mathscr{T}_{0}\| = \sup_{f \in L^{p}_{\Phi}(a,b), f \neq 0} \frac{\|\mathscr{T}_{0}f\|_{L^{q'}_{\Psi^{1-q'}}(a,b)}}{\|f\|_{L^{p}_{\Phi}(a,b)}},$$
(2.94)

it follows that the operator \mathscr{T}_0 is bounded. In other words, inequality (2.85) yields the upper bound for the norm of this operator, i.e.

$$\|\mathscr{T}_0\| \le c_0^{\frac{1}{p'}} (2 - p'A_1) c_0^{\frac{1}{q'}} (q'A_2).$$

The same type of discussion can also be applied when considering the Hilbert-type operator related to kernel \tilde{K}_0 , defined by (2.88). More precisely, if we denote

$$\widetilde{\Phi}(x) = \frac{u^{(A_1 + A_2)p - \frac{p}{q'}}(x)}{(u')^{p-1}(x)} \text{ and } \widetilde{\Psi}(y) = \frac{v^{(A_1 + A_2)q - \frac{q}{p'}}(y)}{(v')^{q-1}(y)}, \qquad x, y \in (a, b),$$

we can define the operator $\widetilde{\mathscr{T}_0}: L^p_{\widetilde{\Phi}}(a,b) \to L^{q'}_{\widetilde{\Psi}^{1-q'}}(a,b)$ as

$$\left(\widetilde{\mathscr{T}}_0 f\right)(y) = \int_a^b \widetilde{K}_0^\lambda(x, y) f(x) \, dx, \quad y \in (a, b).$$
(2.95)

The operator $\widetilde{\mathscr{T}_0}$ is well-defined since inequality (2.90) implies $\widetilde{\mathscr{T}_0} f \in L^{q'}_{\widetilde{\Psi}^{1-q'}}(a,b)$. In addition, since the norm of operator $\widetilde{\mathscr{T}_0}$ is defined by

$$\|\widetilde{\mathscr{T}}_{0}\| = \sup_{f \in L^{p}_{\widehat{\Phi}}(a,b), f \neq 0} \frac{\|\widetilde{\mathscr{T}}_{0}f\|_{L^{q'}_{\widehat{\Psi}^{1-q'}}(a,b)}}{\|f\|_{L^{p}_{\widehat{\Phi}}(a,b)}},$$
(2.96)

inequality (2.90) yields the upper bound for the norm, that is,

$$\|\widetilde{\mathscr{T}}_{0}\| \leq c_{0}^{\frac{1}{p'}}(p'A_{1})c_{0}^{\frac{1}{q'}}(q'A_{2}).$$

We consider now some particular cases in which we are able to find the norm of operators \tilde{T} and \tilde{T}_0 . Obviously, this problem is equivalent to the problem of finding the best possible constants in inequalities (2.85) and (2.90).

The problem of finding the best possible constants in Hilbert-type inequalities with non-conjugate exponents is still open. We solved the mentioned problem in the case of conjugate exponents, as in Section 1.4 (Chapter 1).

Hence, in order to obtain the best possible constants in Theorems 2.12 and 2.13, in the case of conjugate exponents, we exploit Theorem 1.13. The parameters A_1 and A_2 should fulfill condition (1.29) for s = 0, that is, $pA_2 + qA_1 = 2$, and the homogeneous kernel of zero-degree should be such that $k_0(1,t)$ is bounded on (0,1). Under these assumptions, the constant factor on the right-hand sides of inequalities (2.84) and (2.85) reduces to $c_0(pA_2)$.

Theorem 2.14 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, let A_1 and A_2 be real parameters such that $qA_1 + pA_2 = 2$, and let $c_0(pA_2) < \infty$. If the function $k_0(1,t)$ is bounded on (0,1), then the constant $c_0(pA_2)$ is the best possible in both inequalities (2.84) and (2.85).

According to the above discussion, the previous result also provides the norm of the operator \mathscr{T}_0 , defined by (2.93), in the case of conjugate exponents.

Corollary 2.9 Suppose that the assumptions of Theorem 2.14 are fulfilled. Then the norm of operator $\mathscr{T}_0: L^p_{\Phi}(a,b) \to L^p_{\Psi^{1-p}}(a,b)$, defined by the formula $(\mathscr{T}_0f)(y) = \int_a^b K_0(x,y)f(x)dx$, $y \in (a,b)$, is $\|\mathscr{T}_0\| = c_0(pA_2)$.

Utilizing a suitable change of variables, Theorem 1.13 can also be adjusted in establishing the best possible constant in Theorem 2.13. It turns out that the parameters A_1 and A_2 satisfy $pA_2 = qA_1$, providing the same constant factor $c_0(pA_2)$ as in the previous case.

Theorem 2.15 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, let A_1 and A_2 be real parameters such that $qA_1 = pA_2$, and let $c_0(pA_2) < \infty$. If the function $k_0(1,t)$ is bounded on (0,1), then the constant factor $c_0(pA_2)$ is the best possible in both inequalities (2.89) and (2.90).

We conclude this subsection with the operator analogue of Theorem 2.15.

Corollary 2.10 Suppose that the assumptions as in Theorem 2.15 are fulfilled. Then, the norm of the operator $\widetilde{\mathscr{T}}_0: L^p_{\widehat{\Phi}}(a,b) \to L^p_{\widehat{\Psi}^{1-p}}(a,b)$, defined by $(\widetilde{\mathscr{T}}_0f)(y) = \int_a^b \widetilde{K}_0(x,y)f(x)dx$, $y \in (a,b)$, is $\|\widetilde{\mathscr{T}}_0\| = c_0(pA_2)$.

2.6.3 On some related Hardy-type operators

In the previous subsection we have considered two examples of Hilbert-type operators. We can also generate some other operators related to inequalities (2.85) and (2.90). More precisely, if $k_0 : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ is a homogeneous function of zero-degree, then the function $\overline{k}_0 : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$, defined by

$$\overline{k}_0(x,y) = k_0(x,y)\chi_{x \ge y}(x,y) = \begin{cases} 0, & x < y, \\ k_0(x,y), & x \ge y, \end{cases}$$
(2.97)

is also homogeneous of zero-degree. Now, we use the same procedure as in the previous subsection, but with this function \overline{k}_0 . The kernel \overline{k}_0 looks like the classical Hardy kernel (see Chapter 8), so the corresponding operators will be referred to as the Hardy-type operators. Moreover, since

$$c_0(\alpha) = \int_0^\infty \overline{k}_0(1,t) t^{-\alpha} dt = \int_0^1 k_0(1,t) t^{-\alpha} dt,$$

we define

$$\overline{c}_0(\alpha) = \int_0^1 k_0(1,t) t^{-\alpha} dt.$$
(2.98)

Bearing in mind the notation from the previous subsection, we define the operator $\overline{\mathscr{T}}_0: L^p_{\Phi}(a,b) \to L^{q'}_{\Psi^{1-q'}}(a,b)$ by

$$\left(\overline{\mathscr{T}}_{0}f\right)(y) = \int_{u^{(-1)}(v(y))}^{b} K_{0}^{\lambda}(x,y)f(x)\,dx, \quad y \in (a,b),$$
(2.99)

and $\overline{\widetilde{\mathscr{T}}}_0: L^p_{\widetilde{\Phi}}(a,b) \to L^{q'}_{\widetilde{\Psi}^{1-q'}}(a,b)$ by

$$\left(\overline{\widetilde{\mathscr{T}}}_{0}f\right)(y) = \int_{a}^{u^{(-1)\left(\frac{1}{\nu(y)}\right)}} \widetilde{K}_{0}^{\lambda}(x,y)f(x)\,dx, \quad y \in (a,b),$$
(2.100)

where $u^{(-1)}$ denotes the inverse of the function *u*.

The following two corollaries show that these definitions are meaningful. In other words, we provide the corresponding analogues of Theorems 2.12 and 2.13 for the above Hardy-type kernel of zero-degree.

Corollary 2.11 Let p, q, and λ be as in (2.1) and (2.2), and let $u, v : (a,b) \to \mathbb{R}$ be nonnegative measurable functions fulfilling conditions (i)-(iii). Further, suppose $K_0 : (a,b) \times (a,b) \to \mathbb{R}$ is a non-negative measurable function defined by (2.83). If A_1 and A_2 are real parameters such that $\overline{c}_0(2 - p'A_1) < \infty$ and $\overline{c}_0(q'A_2) < \infty$, then the inequalities

$$\int_{a}^{b} \int_{u^{(-1)}(v(y))}^{b} K_{0}^{\lambda}(x,y) f(x)g(y) \, dx \, dy \leq \overline{c}_{0}^{\frac{1}{p'}}(2-p'A_{1})\overline{c}_{0}^{\frac{1}{q'}}(q'A_{2}) \\ \times \left[\int_{a}^{b} \frac{u^{(A_{1}-A_{2})p+\frac{p}{q'}}(x)}{(u')^{p-1}(x)} f^{p}(x) \, dx \right]^{\frac{1}{p}} \left[\int_{a}^{b} \frac{v^{(A_{2}-A_{1})q+\frac{q}{p'}}(y)}{(v')^{q-1}(y)} g^{q}(y) \, dy \right]^{\frac{1}{q}}$$
(2.101)

and

$$\left\{ \int_{a}^{b} \frac{v'(y)}{v^{(A_{2}-A_{1})q'+\frac{q'}{p'}(y)}} \left[\int_{u^{(-1)}(v(y))}^{b} K_{0}^{\lambda}(x,y)f(x) \, dx \right]^{q'} \, dy \right\}^{\frac{1}{q'}} \\
\leq \overline{c}_{0}^{\frac{1}{p'}}(2-p'A_{1})\overline{c}_{0}^{\frac{1}{q'}}(q'A_{2}) \left[\int_{a}^{b} \frac{u^{(A_{1}-A_{2})p+\frac{p}{q'}}(x)}{(u')^{p-1}(x)} f^{p}(x) \, dx \right]^{\frac{1}{p}}$$
(2.102)

hold for all non-negative measurable functions $f, g: (a,b) \to \mathbb{R}$ and are equivalent. Equalities in (2.101) and (2.102) hold if and only if f = 0 or g = 0 a.e. on (a,b).

Corollary 2.12 Let p, q, and λ be as in (2.1) and (2.2), and let $u, v : (a,b) \to \mathbb{R}$ be nonnegative measurable functions fulfilling conditions (i)-(iii). Further, suppose $\widetilde{K}_0 : (a,b) \times (a,b) \to \mathbb{R}$ is a non-negative measurable function defined by (2.88) and A_1 , A_2 are real parameters such that $\overline{c}_0(p'A_1) < \infty$ and $\overline{c}_0(q'A_2) < \infty$. Then the inequalities

$$\int_{a}^{b} \int_{a}^{u^{(-1)}\left(\frac{1}{v(y)}\right)} \widetilde{K}_{0}^{\lambda}(x,y) f(x)g(y) \, dx \, dy \leq \overline{c}_{0}^{\frac{1}{p'}}(p'A_{1})\overline{c}_{0}^{\frac{1}{q'}}(q'A_{2}) \\ \times \left[\int_{a}^{b} \frac{u^{(A_{1}+A_{2})p-\frac{p}{q'}}(x)}{(u')^{p-1}(x)} f^{p}(x) \, dx\right]^{\frac{1}{p}} \left[\int_{a}^{b} \frac{v^{(A_{1}+A_{2})q-\frac{q}{p'}}(y)}{(v')^{q-1}(y)} g^{q}(y) \, dy\right]^{\frac{1}{q}}$$
(2.103)

and

$$\left\{ \int_{a}^{b} \frac{v'(y)}{v^{(A_{1}+A_{2})q'-\frac{q'}{p'}}(y)} \left[\int_{a}^{u^{(-1)}\left(\frac{1}{v(y)}\right)} \widetilde{K}_{0}^{\lambda}(x,y) f(x) \, dx \right]^{q'} \, dy \right\}^{\frac{1}{q'}} \\
\leq \overline{c}_{0}^{\frac{1}{p'}}(p'A_{1}) \overline{c}_{0}^{\frac{1}{q'}}(q'A_{2}) \left[\int_{a}^{b} \frac{u^{(A_{1}+A_{2})p-\frac{p}{q'}}(x)}{(u')^{p-1}(x)} f^{p}(x) \, dx \right]^{\frac{1}{p}}$$
(2.104)

hold for all non-negative measurable functions $f, g: (a,b) \to \mathbb{R}$ and are equivalent. Equalities in (2.103) and (2.104) hold if and only if f = 0 or g = 0 a.e. on (a,b).

Remark 2.25 Regarding Corollaries 2.11 and 2.12, one can easily obtain analogues of Corollaries 2.7 and 2.8. It suffices to replace the constant factor $c_0(\alpha)$ with $\overline{c}_0(\alpha)$ and change the integration intervals according to definitions (2.99) and (2.100). For example, if $u(x) = \exp x$ and $v(y) = \exp y$, then $u^{(-1)}(v(y)) = y$ and $u^{(-1)}(\frac{1}{v(y)}) = -y$.

Remark 2.26 The discussion about the best constants, carried out in the previous subsection, holds for the Hardy-type operators (2.99) and (2.100), as well. More precisely, if p,q > 1 are conjugate exponents and A_1, A_2 are such that $pA_2 + qA_1 = 2$, then $\overline{c}_0(pA_2)$ is the best possible constant in inequalities (2.101) and (2.102). Moreover, the same constant is also the best possible in (2.103) and (2.104), provided that $qA_1 = pA_2$ in the conjugate setting. Of course, under the above assumptions, the norms of operators $\overline{\mathcal{T}}_0$ and $\overline{\widetilde{\mathcal{T}}}_0$ are both equal to $\overline{c}_0(pA_2)$.

2.6.4 Applications

We conclude this section with some consequences of Theorem 2.12, Theorem 2.13, Corollary 2.11 and Corollary 2.12, obtained by a suitable choice of parameters A_1 and A_2 . Namely, if $A_1 = \frac{1}{p'}$ and $A_2 = \frac{1}{q'}$, then the inequalities (2.84) and (2.85) respectively read as

$$\int_{a}^{b} \int_{a}^{b} k_{0}^{\lambda}(u(x), v(y)) f(x)g(y) \, dx \, dy$$

$$\leq c_{0}^{\lambda}(1) \left[\int_{a}^{b} \left(\frac{u}{u'} \right)^{p-1}(x) f^{p}(x) \, dx \right]^{\frac{1}{p}} \left[\int_{a}^{b} \left(\frac{v}{v'} \right)^{p-1}(y) g^{q}(y) \, dy \right]^{\frac{1}{q}} (2.105)$$

and

$$\left\{ \int_{a}^{b} \left(\frac{v'}{v}\right)(y) \left[\int_{a}^{b} k_{0}^{\lambda}(u(x), v(y)) f(x) \, dx \right]^{q'} \, dy \right\}^{\frac{1}{q'}} \\ \leq c_{0}^{\lambda}(1) \left[\int_{a}^{b} \left(\frac{u}{u'}\right)^{p-1}(x) f^{p}(x) \, dx \right]^{\frac{1}{p}}.$$

$$(2.106)$$

Remark 2.27 For the same choice of parameters A_1 and A_2 , i.e. $A_1 = \frac{1}{p'}$ and $A_2 = \frac{1}{q'}$, Theorem 2.13 yields the same inequalities as (2.105) and (2.106), with the kernel $k_0(1, u(x)v(y))$ instead of $k_0(u(x), v(y))$. In other words, inequalities (2.105) and (2.106) also hold after replacing the kernel $k_0(u(x), v(y))$ with $k_0(1, u(x)v(y))$.

The same setting also yields the corresponding result for the Hardy-type kernel (2.97). Namely, for $A_1 = \frac{1}{p'}$ and $A_2 = \frac{1}{q'}$, Corollary 2.11 yields inequalities

$$\int_{a}^{b} \int_{u^{(-1)}(v(y))}^{b} k_{0}^{\lambda}(u(x), v(y)) f(x)g(y) \, dx \, dy$$

$$\leq \overline{c}_{0}^{\lambda}(1) \left[\int_{a}^{b} \left(\frac{u}{u'} \right)^{p-1}(x) f^{p}(x) \, dx \right]^{\frac{1}{p}} \left[\int_{a}^{b} \left(\frac{v}{v'} \right)^{p-1}(y) g^{q}(y) \, dy \right]^{\frac{1}{q}} (2.107)$$

and

$$\left\{ \int_{a}^{b} \left(\frac{v'}{v}\right)(y) \left[\int_{u^{(-1)}(v(y))}^{b} k_{0}^{\lambda}(u(x), v(y))f(x) dx \right]^{q'} dy \right\}^{\frac{1}{q'}} \leq \overline{c}_{0}^{\lambda}(1) \left[\int_{a}^{b} \left(\frac{u}{u'}\right)^{p-1}(x)f^{p}(x) dx \right]^{\frac{1}{p}}.$$
(2.108)

Remark 2.28 According to Corollary 2.12, inequalities (2.107) and (2.108) also hold after replacing the kernel $k_0(u(x), v(y))$ with $k_0(1, u(x)v(y))$ and replacing integration interval $(u^{(-1)}(v(y)), b)$ with $(a, u^{(-1)}(\frac{1}{v(y)}))$.

Remark 2.29 Considering the conjugate case, parameters that generate inequalities (2.105), (2.106), (2.107), and (2.108) become $A_1 = \frac{1}{q}$ and $A_2 = \frac{1}{p}$. These parameters fulfill condition $qA_1 + pA_2 = 2$, providing the best possible constant factors $c_0(1)$ and $\overline{c_0}(1)$.

In order to complete the previous discussion, we provide some examples of homogeneous kernels with zero-degree of homogeneity, which generate the corresponding constant factors expressed in terms of the Beta and the Gamma function.

Example 2.1 Let $\alpha > 0$, $\beta > -1$ and

$$k_0(x,y) = \left(\frac{\min\{x,y\}}{\max\{x,y\}}\right)^{\alpha} \left|\log\left(\frac{y}{x}\right)\right|^{\beta}.$$

Then,

$$\int_0^\infty \left(\frac{\min\{1,t\}}{\max\{1,t\}}\right)^\alpha |\log t|^\beta t^{-1} dt = \int_0^1 t^{\alpha-1} (-\log t)^\beta dt + \int_1^\infty t^{-\alpha-1} (\log t)^\beta dt.$$

Since,

$$\int_{0}^{1} t^{\alpha - 1} (-\log t)^{\beta} dt = \int_{1}^{\infty} t^{-\alpha - 1} (\log t)^{\beta} dt = \int_{0}^{\infty} e^{-\alpha v} v^{\beta} dv = \frac{\Gamma(\beta + 1)}{\alpha^{\beta + 1}},$$

the above constant factors become

$$c_0(1) = \frac{2\Gamma(\beta+1)}{\alpha^{\beta+1}}$$
 and $\overline{c}_0(1) = \frac{\Gamma(\beta+1)}{\alpha^{\beta+1}}.$

Example 2.2 For the homogeneous function defined by

$$k_0(x,y) = \frac{\min\{x,y\}}{\max\{x,y\}} \arctan\left(\frac{y}{x}\right),$$

we have

$$\int_0^\infty \frac{\min\{1,t\}}{\max\{1,t\}} t^{-1} \arctan t dt = \int_0^1 \arctan t dt + \int_1^\infty t^{-2} \arctan t dt.$$

Clearly, the above two integrals can be resolved by using integration by parts, yielding the constant factors

$$c_0(1) = \frac{\pi}{2}$$
 and $\overline{c}_0(1) = \frac{\pi}{4} - \frac{\log 2}{2}$.

Example 2.3 *Let* $0 < \alpha < 1$ *and*

. 1

$$k_0(x,y) = \left(\frac{\min\{x,y\}}{|x-y|}\right)^{\alpha}.$$

Then,

$$\int_0^\infty \left(\frac{\min\{1,t\}}{|1-t|}\right)^\alpha t^{-1}dt = \int_0^1 t^{\alpha-1}(1-t)^{-\alpha}dt + \int_1^\infty t^{-1}(t-1)^{-\alpha}dt.$$

Since

$$\int_{0}^{1} t^{\alpha - 1} (1 - t)^{-\alpha} dt = \int_{1}^{\infty} t^{-1} (t - 1)^{-\alpha} dt = B(1 - \alpha, \alpha),$$

we obtain the constant factors expressed in terms of the Beta function:

$$c_0(1) = 2B(1-\alpha,\alpha)$$
 and $\overline{c}_0(1) = B(1-\alpha,\alpha)$

Remark 2.30 Results in this section are taken from [166]. However, similar Hilbert-type and Hardy-type operators can also be derived for homogeneous kernels of arbitrary degree of homogeneity. For more details about similar results, the reader is referred to [37], [158], [160], [163], [165], [167].

Chapter 3

Hilbert-type inequalities with vector variables

This chapter deals with Hilbert-type inequalities involving real valued functions with vector arguments. We start this overview with the so-called doubly weighted Hardy-Littlewood-Sobolev inequality of Stein and Weiss, [118],

$$\int_{\mathbb{R}^n} \int_{\mathbb{R}^n} \frac{f(x)g(y)}{|x|^{\alpha}|x-y|^s|y|^{\beta}} \, dxdy \le C_{\alpha,\beta,p,q,n} \, \|f\|_{L^p(\mathbb{R}^n)} \|g\|_{L^q(\mathbb{R}^n)}, \tag{3.1}$$

which holds for $n \in \mathbb{N}$, p, q > 1 such that $\frac{1}{p} + \frac{1}{q} > 1$, λ as in (2.2), $0 \le \alpha < \frac{n}{p'}$, $0 \le \beta < \frac{n}{q'}$, $s = n\lambda - \alpha - \beta$, and all non-negative functions $f \in L^p(\mathbb{R}^n)$ and $g \in L^q(\mathbb{R}^n)$. Here, and throughout this chapter, |x| denotes the Euclidean norm of the vector $x \in \mathbb{R}^n$.

In [83], Lieb proved the existence of optimizers for (3.1), that is, functions f and g which, when inserted into (3.1), yield equality with the smallest possible constant $C_{\alpha,\beta,p,q,n}$. Moreover, for p = q and $\alpha = \beta = 0$, the constant and maximizing functions were explicitly computed in [83]. In particular, Lieb proved that

$$C_{0,0,p,p,n} = \pi^{\frac{n}{p'}} \frac{\Gamma\left(\frac{n}{2} - \frac{n}{p'}\right)}{\Gamma\left(\frac{n}{p}\right)} \left[\frac{\Gamma\left(\frac{n}{2}\right)}{\Gamma(n)}\right]^{\frac{2}{p'}-1}$$

where Γ is the Gamma function. Unfortunately, neither $C_{\alpha,\beta,p,q,n}$ nor the optimizers are known for any other choice of the parameters appearing in (3.1). It was only shown (see e. g. [84]) that for the classical Hardy-Littlewood-Sobolev inequality, that is, for (3.1) with

 $\alpha = \beta = 0$ and $s = n\lambda$, the estimate

$$C_{0,0,p,q,n} \le \frac{(p')^{\lambda} + (q')^{\lambda}}{(1-\lambda)pq} \left(\frac{\lambda}{n} |\mathbb{S}^{n-1}|\right)^{\lambda}$$
(3.2)

holds, where

$$\mathbb{S}^{n-1}| = \frac{2\pi^{\frac{n}{2}}}{\Gamma\left(\frac{n}{2}\right)} \tag{3.3}$$

is the area of the unit sphere \mathbb{S}^{n-1} in \mathbb{R}^n . For more details about the Hardy-Littlewood-Sobolev inequality, the reader is referred to [83] and [84].

One of the most important tasks in this chapter is to derive the explicit upper bounds for the general case of the doubly weighted Hardy-Littlewood-Sobolev inequality (3.1). This can be done by virtue of Theorem 2.8 from the previous chapter and the Selberg integral formula (see e. g. [119]). Moreover, the Selberg integral formula and the appropriate results from the previous chapter will also be utilized in deriving numerous particular Hilbert-type inequalities with vector variables.

On the other hand, some vector extensions of the usual Beta function will also be utilized in obtaining some particular Hilbert-type inequalities. Finally, at the end of this chapter we shall present some multidimensional Hilbert-type inequalities including a general homogeneous kernel and power weight functions whose arguments are norms of the corresponding vectors.

3.1 Explicit upper bounds for the doubly weighted Hardy-Littlewood-Sobolev inequality

The main goal of this section is to derive a form of the doubly weighted Hardy-Littlewood-Sobolev inequality (3.1) with an explicit constant factor on its right-hand side. In fact, we derive explicit upper bounds for the sharp constant $C_{\alpha,\beta,p,q,n}$ in (3.1).

Main results in this section will be based on Theorem 2.8 (see Section 2.4), and the well-known Selberg integral formula

$$\int_{\mathbb{R}^{k_n}} |x_k|^{\alpha_k - n} \left(\prod_{i=1}^{k-1} |x_{i+1} - x_i|^{\alpha_i - n} \right) |x_1 - y|^{\alpha_0 - n} dx_1 \dots dx_k$$
$$= \frac{\Gamma_n(\alpha_0) \cdots \Gamma_n(\alpha_k)}{\Gamma_n(\alpha_0 + \dots + \alpha_k)} |y|^{\alpha_0 + \dots + \alpha_k - n},$$
(3.4)

for arbitrary $k, n \in \mathbb{N}$, $y \in \mathbb{R}^n$, and $0 < \alpha_0, \dots, \alpha_k < n$ such that $0 < \sum_{i=0}^k \alpha_i < n$, where $\Gamma_n(\alpha) = \frac{\pi^{\frac{n}{2}} 2^{\alpha} \Gamma(\frac{\alpha}{2})}{\Gamma(\frac{n}{2} - \frac{\alpha}{2})}$ is the *n*-dimensional Gamma function. For k = 1, the product appearing in (3.4) is defined to be equal to 1. In [119], Stein derived the above Selberg integral
formula with two parameters using the Riesz potential (see also [25]). It should be noticed here that

$$\Gamma_n(n-\alpha) = \frac{(2\pi)^n}{\Gamma_n(\alpha)}, \quad 0 < \alpha < n.$$
(3.5)

The Selberg integral formula is very useful in numerous parts of mathematics, especially in representation theory and in mathematical physics.

In order to establish our main results, we first reformulate (3.4) in a form which will be more suitable for our computations.

Lemma 3.1 Suppose $k, n \in \mathbb{N}$, $0 < \beta_1, ..., \beta_k, s < n$ are such that $\sum_{i=1}^k \beta_i + s > kn$, and $y \in \mathbb{R}^n$. Then

$$\int_{\mathbb{R}^{k_n}} \frac{|x_1|^{-\beta_1} \cdots |x_k|^{-\beta_k}}{|x_1 + \dots + x_k + y|^s} dx_1 \dots dx_k = \frac{\Gamma_n (n - \beta_1) \cdots \Gamma_n (n - \beta_k) \Gamma_n (n - s)}{\Gamma_n ((k+1)n - \beta_1 - \dots - \beta_k - s)} |y|^{k_n - \beta_1 - \dots - \beta_k - s}.$$
(3.6)

Proof. Set $\alpha_i = n - \beta_{i+1}$, i = 0, ..., k - 1, and $\alpha_k = n - s$. Substituting first $t_1 = x_1 + y$ and then $t_2 = t_1 + x_2$, the left-hand side of (3.6) becomes

$$\int_{\mathbb{R}^{k_n}} \frac{|x_1|^{-\beta_1} \dots |x_k|^{-\beta_k}}{|x_1 + \dots + x_k + y|^s} dx_1 \dots dx_k$$

=
$$\int_{\mathbb{R}^{k_n}} \frac{|t_1 - y|^{\alpha_0 - n} |x_2|^{\alpha_1 - n} \dots |x_k|^{\alpha_{k-1} - n}}{|t_1 + x_2 + \dots + x_k|^{n - \alpha_k}} dt_1 dx_2 \dots dx_k$$

=
$$\int_{\mathbb{R}^{k_n}} \frac{|t_1 - y|^{\alpha_0 - n} |t_2 - t_1|^{\alpha_1 - n} |x_3|^{\alpha_2 - n} \dots |x_k|^{\alpha_{k-1} - n}}{|t_2 + x_3 + \dots + x_k|^{n - \alpha_k}} dt_1 dt_2 dx_3 \dots dx_k.$$
(3.7)

After the sequence of similar substitutions $t_i = t_{i-1} + x_i$, i = 2, ..., k, the last line of (3.7) is finally equal to

$$\int_{\mathbb{R}^{kn}} |t_k|^{\alpha_k - n} \left(\prod_{i=1}^{k-1} |t_{i+1} - t_i|^{\alpha_i - n} \right) |t_1 - y|^{\alpha_0 - n} dt_1 \dots dt_k$$
$$= \frac{\Gamma_n(\alpha_0) \cdots \Gamma_n(\alpha_k)}{\Gamma_n(\alpha_0 + \dots + \alpha_k)} |y|^{\alpha_0 + \dots + \alpha_k - n}$$
$$= \frac{\Gamma_n(n - \beta_1) \cdots \Gamma_n(n - \beta_k) \Gamma_n(n - s)}{\Gamma_n((k+1)n - \beta_1 - \dots - \beta_k - s)} |y|^{kn - \beta_1 - \dots - \beta_k - s},$$

where the last two equalities are obtained by the Selberg integral formula (3.4) and by replacing α_i by the corresponding expressions including β_i .

Since the case k = 1 of Lemma 3.1 will be of our special interest, we state it as a separate result.

Lemma 3.2 Let $n \in \mathbb{N}$ and $y \in \mathbb{R}^n$. If $0 < \beta$, s < n are such that $\beta + s > n$, then

$$\int_{\mathbb{R}^n} \frac{|x|^{-\beta}}{|x+y|^s} \, dx = \frac{\Gamma_n(n-\beta)\Gamma_n(n-s)}{\Gamma_n(2n-\beta-s)} \, |y|^{n-\beta-s}.$$

We can now obtain the doubly weighted Hardy-Littlewood-Sobolev inequality (3.1) mentioned above. More precisely, we utilize Theorem 2.8, that is, inequality (2.51) with $\Omega = \mathbb{R}^n$ and the kernel $K(x,y) = |x|^{-\alpha} |x-y|^{-s} |y|^{-\beta}$.

The first step is to consider the case when the function $g \in L^q(\mathbb{R}^n)$ on the left-hand side of (2.51) is symmetric-decreasing, that is, $g(x) \ge g(y)$ whenever $|x| \le |y|$. For such function and $y \in \mathbb{R}^n$, $y \ne 0$, we have

$$g^{q}(y) \leq \frac{1}{|B(|y|)|} \int_{B(|y|)} g^{q}(x) dx$$

$$\leq \frac{1}{|B(|y|)|} \int_{\mathbb{R}^{n}} g^{q}(x) dx = \frac{n}{|\mathbb{S}^{n-1}|} |y|^{-n} \|g\|_{L^{q}(\mathbb{R}^{n})}^{q}, \qquad (3.8)$$

where B(|y|) denotes the ball of radius |y| in \mathbb{R}^n , centered at the origin, and $|B(|y|)| = |y|^n \frac{|\mathbb{S}^{n-1}|}{n}$ is its volume.

Theorem 3.1 Let $n \in \mathbb{N}$, p > 1 and q > 1 be such that $\frac{1}{p} + \frac{1}{q} > 1$, and let λ be defined by (2.2). Let $0 \le \alpha < \frac{n}{p'}$, $0 \le \beta < \frac{n}{q'}$, and $s = n\lambda - \alpha - \beta$. Then the inequality

$$\int_{\mathbb{R}^{n}} \int_{\mathbb{R}^{n}} \frac{f(x)g(y)}{|x|^{\alpha}|x-y|^{s}|y|^{\beta}} dxdy$$

$$\leq \frac{(2\pi)^{2n} \left(\frac{|\mathbb{S}^{n-1}|}{n}\right)^{\lambda-1}}{\Gamma_{n}\left(\frac{n}{p}+\alpha\right)\Gamma_{n}\left(\frac{n}{q}+\beta\right)\Gamma_{n}(s)} \|f\|_{L^{p}(\mathbb{R}^{n})} \|g\|_{L^{q}(\mathbb{R}^{n})}$$
(3.9)

holds for all non-negative functions $f \in L^p(\mathbb{R}^n)$ and symmetric-decreasing functions $g \in L^q(\mathbb{R}^n)$.

Proof. Suppose that in Theorem 2.8 (see Section 2.4) we have $\Omega = \mathbb{R}^n$, $K(x, y) = |x|^{-\alpha} |x - y|^{-s} |y|^{-\beta}$, $F(x, y) = G(x, y) \equiv 1$, $\varphi(x) = |x|^{\frac{n}{pp'}}$, $\psi(y) = |y|^{\frac{n}{qq'}}$, and the Lebesgue measure dx. In this case, the left-hand side of inequality (2.51) reads

$$L = \int_{\mathbb{R}^n} \int_{\mathbb{R}^n} \frac{f(x)g(y)}{|x|^{\alpha} |x-y|^s |y|^{\beta}} \, dx dy, \tag{3.10}$$

while its right-hand side is a product $I_1^{\frac{1}{q'}} I_2^{\frac{1}{p'}} I_3^{1-\lambda}$, where

$$I_{1} = \int_{\mathbb{R}^{2n}} \frac{|x|^{\frac{n}{p'}} |y|^{-\frac{n}{q}}}{|x|^{\alpha} |x-y|^{s} |y|^{\beta}} f^{p}(x) \, dx dy, \quad I_{2} = \int_{\mathbb{R}^{2n}} \frac{|x|^{-\frac{n}{p}} |y|^{\frac{n}{q'}}}{|x|^{\alpha} |x-y|^{s} |y|^{\beta}} \, g^{q}(y) \, dx dy.$$
$$I_{3} = \int_{\mathbb{R}^{2n}} \frac{|x|^{\frac{n}{p'}} |y|^{\frac{n}{q'}}}{|x|^{\alpha} |x-y|^{s} |y|^{\beta}} \, f^{p}(x) g^{q}(y) \, dx dy.$$

Therefore, applying the Fubini theorem, Lemma 3.2, identity (3.5), and the fact that $\alpha + \beta + s = n\lambda$, we have

$$\begin{split} I_{1} &= \int_{\mathbb{R}^{n}} |x|^{\frac{n}{p'}-\alpha} f^{p}(x) \int_{\mathbb{R}^{n}} \frac{|y|^{-\left(\frac{n}{q}+\beta\right)}}{|x-y|^{s}} dy dx \\ &= \int_{\mathbb{R}^{n}} |x|^{\frac{n}{p'}-\alpha} f^{p}(x) \int_{\mathbb{R}^{n}} \frac{|z|^{-\left(\frac{n}{q}+\beta\right)}}{|z+x|^{s}} dz dx \\ &= \frac{\Gamma_{n}\left(n-\frac{n}{q}-\beta\right)\Gamma_{n}(n-s)}{\Gamma_{n}\left(2n-\frac{n}{q}-\beta-s\right)} \int_{\mathbb{R}^{n}} |x|^{\frac{n}{p'}-\alpha+n-\frac{n}{q}-\beta-s} f^{p}(x) dx \\ &= \frac{(2\pi)^{2n}}{\Gamma_{n}\left(\frac{n}{p}+\alpha\right)\Gamma_{n}\left(\frac{n}{q}+\beta\right)\Gamma_{n}(s)} \parallel f \parallel_{L^{p}(\mathbb{R}^{n})}^{p}. \end{split}$$

Analogously,

$$I_2 = \frac{(2\pi)^{2n}}{\Gamma_n\left(\frac{n}{p} + \alpha\right)\Gamma_n\left(\frac{n}{q} + \beta\right)\Gamma_n(s)} \parallel g \parallel_{L^q(\mathbb{R}^n)}^q,$$

and, by (3.8),

$$I_{3} \leq \frac{n}{|\mathbb{S}^{n-1}|} \|g\|_{L^{q}(\mathbb{R}^{n})}^{q} \int_{\mathbb{R}^{n}} |x|^{\frac{n}{p'}-\alpha} f^{p}(x) \int_{\mathbb{R}^{n}} \frac{|y|^{-\left(\frac{n}{q}+\beta\right)}}{|x-y|^{s}} dy dx$$
$$= \frac{n}{|\mathbb{S}^{n-1}|} \frac{(2\pi)^{2n}}{\Gamma_{n}\left(\frac{n}{p}+\alpha\right)\Gamma_{n}\left(\frac{n}{q}+\beta\right)\Gamma_{n}(s)} \|f\|_{L^{p}(\mathbb{R}^{n})}^{p} \|g\|_{L^{q}(\mathbb{R}^{n})}^{q}$$

Finally, (3.9) follows by combining (3.10) and the expressions we have obtained for the integrals I_1 , I_2 , and I_3 .

To obtain an analogous result for arbitrary non-negative functions $f \in L^p(\mathbb{R}^n)$ and $g \in L^q(\mathbb{R}^n)$, we utilize the general rearrangement inequality. Recall that for a Borel set $A \subset \mathbb{R}^n$ of finite Lebesgue measure, we define A^* , the symmetric rearrangement of A, to be the open ball centered at origin whose volume is that of A. The symmetric-decreasing rearrangement of a characteristic function of a set A is $\chi_A^* = \chi_{A^*}$, so if $f : \mathbb{R}^n \to \mathbb{C}$ is a Borel measurable function vanishing at infinity, the symmetric-decreasing rearrangement f^* of a function f is defined by

$$f^*(x) = \int_0^\infty \chi^*_{\{|f|>t\}}(x) dt.$$

Now, the general rearrangement inequality asserts that

$$I(f_1, f_2, \dots, f_m) \le I(f_1^*, f_2^*, \dots, f_m^*), \tag{3.11}$$

where f_1, f_2, \ldots, f_m are non-negative functions on \mathbb{R}^n , vanishing at infinity, and

$$I(f_1, f_2, \dots, f_m) = \int_{\mathbb{R}^n} \cdots \int_{\mathbb{R}^n} \prod_{j=1}^m f_j\left(\sum_{i=1}^k b_{ij} x_i\right) dx_1 dx_2 \dots dx_k$$

 $k \leq m$, $b_{ij} \in \mathbb{R}$, $1 \leq i \leq k$, $1 \leq j \leq m$. In the following result we also use the fact that for parameter $\gamma > 0$ the function $h : \mathbb{R}^n \to \mathbb{R}$, $h(x) = |x|^{-\gamma}$, is symmetric-decreasing and vanishes at infinity, which implies that $h^* = h$. For more details about rearrangements of sets and functions, the reader is referred to [84].

Theorem 3.2 Let $n \in \mathbb{N}$, p > 1 and q > 1 be such that $\frac{1}{p} + \frac{1}{q} > 1$, and set λ as in (2.2). Suppose $0 \le \alpha < \frac{n}{p'}$, $0 \le \beta < \frac{n}{q'}$, and $s = n\lambda - \alpha - \beta$. Then the inequality (3.9) holds for all non-negative functions $f \in L^p(\mathbb{R}^n)$ and $g \in L^q(\mathbb{R}^n)$.

Proof. Since $x \to |x|^{-\alpha}$, $x \to |x|^{-s}$, and $x \to |x|^{-\beta}$ are symmetric-decreasing functions vanishing at infinity, the general rearrangement inequality (3.11) implies that

$$\int_{\mathbb{R}^n} \int_{\mathbb{R}^n} \frac{f(x)g(y)}{|x|^{\alpha}|x-y|^s|y|^{\beta}} \, dxdy \le \int_{\mathbb{R}^n} \int_{\mathbb{R}^n} \frac{f^*(x)g^*(y)}{|x|^{\alpha}|x-y|^s|y|^{\beta}} \, dxdy.$$
(3.12)

Clearly, by Theorem 3.1, the right-hand side of (3.12) is not greater than

$$K_{\alpha,\beta,p,q,n} \| f^* \|_{L^p(\mathbb{R}^n)} \| g^* \|_{L^q(\mathbb{R}^n)} = K_{\alpha,\beta,p,q,n} \| f \|_{L^p(\mathbb{R}^n)} \| g \|_{L^q(\mathbb{R}^n)},$$
(3.13)

where $K_{\alpha,\beta,p,q,n}$ is the constant from the right-hand side of (3.9). To obtain equality in (3.13), we have used the fact that the symmetric-decreasing rearrangement is norm preserving (see e.g. [84]).

Remark 3.1 Note that $C_{\alpha,\beta,p,q,n} \leq K_{\alpha,\beta,p,q,n}$, where $C_{\alpha,\beta,p,q,n}$ is the sharp constant for (3.1) and $K_{\alpha,\beta,p,q,n}$ is the constant factor involved in the right-hand side of (3.9). Hence, Theorem 3.2 provides new explicit upper bounds for the doubly weighted Hardy-Littlewood-Sobolev inequality. In particular, for $\alpha = \beta = 0$ we have

$$K_{0,0,p,q,n} = \frac{(2\pi)^{2n} \left(\frac{|\mathbb{S}^{n-1}|}{n}\right)^{\lambda-1}}{\Gamma_n\left(\frac{n}{p}\right) \Gamma_n\left(\frac{n}{q}\right) \Gamma_n(n\lambda)},$$
(3.14)

while for p = q the above constant reduces to

$$K_{0,0,p,p,n} = \frac{(2\pi)^{2n} \left(\frac{|\mathbb{S}^{n-1}|}{n}\right)^{\frac{2}{p'}-1}}{\Gamma_n^2\left(\frac{n}{p}\right) \Gamma_n\left(\frac{2n}{p'}\right)} = \pi^{\frac{n}{p'}} \frac{\Gamma\left(\frac{n}{2} - \frac{n}{p'}\right)}{\Gamma\left(\frac{n}{p'}\right)} \left[\frac{\Gamma\left(\frac{n}{2p'}\right)}{\Gamma\left(\frac{n}{2p}\right)}\right]^2 \left[\Gamma\left(\frac{n}{2}+1\right)\right]^{1-\frac{2}{p'}}$$

Although (3.2) provides a better estimate for $C_{0,0,p,q,n}$ than (3.14), it is important to note that Theorem 3.2 covers all admissible choices of the parameters p, q, α , and β in (3.1), so the main contribution of this section is in extending the mentioned Lieb result, presented at the beginning of this chapter.

3.2 Trilinear version of standard Beta integral

Considering Lemma 3.2, we may regard the Selberg integral formula as the k-fold generalization of the standard Beta integral on \mathbb{R}^n . On the other hand, by virtue of the Fourier transform (see [25]), the so-called trilinear version of the standard Beta integral is obtained, that is,

$$\int_{\mathbb{R}^n} \frac{|t|^{\alpha+\beta-2n}}{|x-t|^{\alpha}|y-t|^{\beta}} dt = B(\alpha,\beta,n) \frac{|x-y|^{n-\alpha-\beta}}{|x|^{n-\beta}|y|^{n-\alpha}},$$
(3.15)

where $x, y \in \mathbb{R}^n$, $x \neq y \neq 0$, $0 < \alpha, \beta < n, \alpha + \beta > n$, and

$$B(\alpha,\beta,n) = \pi^{\frac{n}{2}} \frac{\Gamma\left(\frac{n-\alpha}{2}\right)\Gamma\left(\frac{n-\beta}{2}\right)\Gamma\left(\frac{\alpha+\beta-n}{2}\right)}{\Gamma\left(\frac{\alpha}{2}\right)\Gamma\left(\frac{\beta}{2}\right)\Gamma\left(n-\frac{\alpha+\beta}{2}\right)}$$

Taking into account definition (3.5) of *n*-dimensional Gamma function, the previous formula can be rewritten as

$$B(\alpha,\beta,n) = \frac{\Gamma_n(n-\alpha)\Gamma_n(n-\beta)}{\Gamma_n(2n-\alpha-\beta)}.$$
(3.16)

It is still unclear whether or not there is a corresponding *k*-fold analogue of (3.15). In spite of that, we shall use the trilinear formula (3.15) to obtain a 2-fold inequality of the Hilbert-type for the kernel $K(x,y) = |x-y|^{\alpha-n}|x+y|^{\beta-n}$, where $0 < \alpha, \beta < n, \alpha + \beta < n$. Of course, we shall rewrite formula (3.15) in a more suitable form. Namely, replacing *y* with -x in (3.15), we obtain

$$\int_{\mathbb{R}^{n}} \frac{|t|^{-\alpha-\beta}}{|x-t|^{n-\alpha}|x+t|^{n-\beta}} dt = 2^{\alpha+\beta-n} B^{*}(\alpha,\beta,n)|x|^{-n},$$
(3.17)

where

$$B^*(\alpha,\beta,n) = \frac{\Gamma_n(\alpha)\Gamma_n(\beta)}{\Gamma_n(\alpha+\beta)}.$$
(3.18)

Now, the corresponding Hilbert-type inequality and its equivalent form are the immediate consequences of Theorem 2.1 and the above relation (3.17).

Theorem 3.3 Let p, q, and λ be as in (2.1) and (2.2), and let α and β be real parameters satisfying $0 < \alpha, \beta < n$ and $\alpha + \beta < n$. Then, the inequalities

$$\int_{\mathbb{R}^{2n}} \frac{f(x)g(y)}{|x-y|^{\lambda(n-\alpha)}|x+y|^{\lambda(n-\beta)}} dxdy$$

$$\leq N \left[\int_{\mathbb{R}^{n}} |x|^{(p-1)(\alpha+\beta+n)-pn\lambda} f^{p}(x) dx \right]^{\frac{1}{p}} \left[\int_{\mathbb{R}^{n}} |y|^{(q-1)(\alpha+\beta+n)-qn\lambda} g^{q}(y) dy \right]^{\frac{1}{q}}$$
(3.19)

and

$$\left\{ \int_{\mathbb{R}^{n}} |y|^{n(\lambda q'-1)-\alpha-\beta} \left[\int_{\mathbb{R}^{n}} \frac{f(x)dx}{|x-y|^{\lambda(n-\alpha)}|x+y|^{\lambda(n-\beta)}} \right]^{q'} dy \right\}^{\frac{1}{q'}} \qquad (3.20)$$

$$\leq N \left[\int_{\mathbb{R}^{n}} |x|^{(p-1)(\alpha+\beta+n)-pn\lambda} f^{p}(x)dx \right]^{\frac{1}{p}}$$

hold and are equivalent, where $N = 2^{\lambda(\alpha+\beta-n)}B^*(\alpha,\beta,n)^{\lambda}$.

Proof. Putting $K(x,y) = |x-y|^{\alpha-n}|x+y|^{\beta-n}$, $\varphi(x) = |x|^{\frac{\alpha+\beta}{p'}}$, $\psi(y) = |y|^{\frac{\alpha+\beta}{q'}}$ in (2.9) and (2.10) (see Theorem 2.1), and utilizing the formula (3.17), we obtain (3.19) and (3.20) respectively, as claimed.

Real parameters α and β from the previous theorem fulfill the condition $\alpha + \beta < n$. In what follows we shall obtain similar inequalities which are, in some way, complementary to inequalities (3.19) and (3.20). Such inequalities will be derived by virtue of Theorem 2.8 (see Section 2.4).

Theorem 3.4 Let p, q, and λ be as in (2.1) and (2.2), and let α and β be real parameters satisfying $0 < \alpha < n$, $0 < \beta < n$, $\alpha + \beta = n\left(\frac{1}{p} + \frac{1}{q}\right) > n$. If f and g are non-negative functions such that $f \in L^p(\mathbb{R}^n)$, $g \in L^q(\mathbb{R}^n)$, then the following inequalities hold and are equivalent:

$$\int_{\mathbb{R}^{2n}} \frac{f(x)g(y)}{|x-y|^{n-\alpha}|x+y|^{n-\beta}} dx dy \le \left(\frac{n}{|\mathbb{S}^{n-1}|}\right)^{1-\lambda} C(p,q;\alpha,\beta;n) \|f\|_{L^{p}(\mathbb{R}^{n})} \|g\|_{L^{q}(\mathbb{R}^{n})},$$
(3.21)

$$\left\{\int_{\mathbb{R}^n} \left[\int_{\mathbb{R}^n} \frac{f(x)dx}{|x-y|^{n-\alpha}|x+y|^{n-\beta}}\right]^{q'} dy\right\}^{\frac{1}{q'}} \le \left(\frac{n}{|\mathbb{S}^{n-1}|}\right)^{1-\lambda} C(p,q;\alpha,\beta;n) \|f\|_{L^p(\mathbb{R}^n)}.$$
(3.22)

Here,

$$C(p,q;\alpha,\beta;n) = \int_{\mathbb{R}^n} \frac{|x|^{-\frac{n}{q}} dx}{|e_1 - x|^{n-\alpha} |e_1 + x|^{n-\beta}},$$
(3.23)

where $e_1 = (1, 0, ..., 0) \in \mathbb{R}^n$ and $|\mathbb{S}^{n-1}|$ is the Lebesgue measure of the unit sphere in \mathbb{R}^n .

Proof. Since we shall use the general rearrangement inequality (3.11), it is enough to prove the inequalities for symmetric-decreasing functions f and g. First, applying Theorem 2.8, we have

$$\int_{\mathbb{R}^{2n}} \frac{f(x)g(y)}{|x-y|^{n-\alpha}|x+y|^{n-\beta}} dx dy \le I_1^{\frac{1}{q'}} I_2^{\frac{1}{p'}} I_3^{1-\lambda},$$
(3.24)

where

$$I_{1} = \int_{\mathbb{R}^{2n}} \frac{|x|^{\frac{n}{p'}}|y|^{-\frac{n}{q}}}{|x-y|^{n-\alpha}|x+y|^{n-\beta}} f^{p}(x) dxdy,$$

$$I_{2} = \int_{\mathbb{R}^{2n}} \frac{|x|^{-\frac{n}{p}}|y|^{\frac{n}{q'}}}{|x-y|^{n-\alpha}|x+y|^{n-\beta}} g^{q}(y) dxdy,$$

$$I_{3} = \int_{\mathbb{R}^{2n}} \frac{|x|^{\frac{n}{p'}}|y|^{\frac{n}{q'}}}{|x-y|^{n-\alpha}|x+y|^{n-\beta}} f^{p}(x)g^{q}(y) dxdy.$$

Further, using the change of variables y = |x|u (so $dy = |x|^n du$) and rotational invariance of the Lebesgue integral in \mathbb{R}^n , we easily get:

$$\begin{split} I_{1} &= \int_{\mathbb{R}^{n}} |x|^{\frac{n}{p'}} f^{p}(x) \int_{\mathbb{R}^{n}} \frac{|y|^{-\frac{n}{q}}}{|x-y|^{n-\alpha}|x+y|^{n-\beta}} \, dy dx \\ &= \int_{\mathbb{R}^{n}} |x|^{\frac{n}{p'} - \frac{n}{q} + \alpha + \beta - n} f^{p}(x) \int_{\mathbb{R}^{n}} \frac{|u|^{-\frac{n}{q}}}{\left|\frac{x}{|x|} - u\right|^{n-\alpha} \left|\frac{x}{|x|} + u\right|^{n-\beta}} \, du dx \\ &= \int_{\mathbb{R}^{n}} \frac{|u|^{-\frac{n}{q}} du}{|e_{1} - u|^{n-\alpha} |e_{1} + u|^{n-\beta}} \, \|f\|_{L^{p}(\mathbb{R}^{n})}^{p}. \end{split}$$

Analogously,

$$I_{2} = \int_{\mathbb{R}^{n}} \frac{|u|^{-\frac{n}{p}} du}{|e_{1} - u|^{n-\alpha} |e_{1} + u|^{n-\beta}} \|g\|_{L^{q}(\mathbb{R}^{n})}^{q},$$

and by (3.8),

$$I_{3} \leq \frac{n}{|\mathbb{S}^{n-1}|} \int_{\mathbb{R}^{n}} \frac{|u|^{-\frac{n}{q}} du}{|e_{1}-u|^{n-\alpha} |e_{1}+u|^{n-\beta}} \|f\|_{L^{p}(\mathbb{R}^{n})}^{p} \|g\|_{L^{q}(\mathbb{R}^{n})}^{q}.$$

It remains to prove that

$$\int_{\mathbb{R}^n} \frac{|x|^{-\frac{n}{p}} dx}{|e_1 - x|^{n - \alpha} |e_1 + x|^{n - \beta}} = \int_{\mathbb{R}^n} \frac{|x|^{-\frac{n}{q}} dx}{|e_1 - x|^{n - \alpha} |e_1 + x|^{n - \beta}}.$$

We transform the left integral in polar coordinates using $x = t\theta$, $t \ge 0$, $\theta \in |\mathbb{S}^{n-1}|$, and $t = \frac{1}{u}$ to obtain:

$$\begin{split} \int_{\mathbb{R}^n} \frac{|x|^{-\frac{n}{p}} dx}{|e_1 - x|^{n-\alpha} |e_1 + x|^{n-\beta}} &= \int_{\mathbb{S}^{n-1}} d\theta \int_0^\infty \frac{t^{-\frac{n}{p}} t^{n-1} dt}{|e_1 - t\theta|^{n-\alpha} |e_1 + t\theta|^{n-\beta}} \\ &= \int_{\mathbb{S}^{n-1}} d\theta \int_0^\infty \frac{t^{-\frac{n}{p}} t^{n-1} dt}{(1 + t^2 - 2t(e_1, \theta))^{\frac{n-\alpha}{2}} (1 + t^2 + 2t(e_1, \theta))^{\frac{n-\beta}{2}}} \\ &= \int_{\mathbb{S}^{n-1}} d\theta \int_0^\infty \frac{u^{\frac{n}{p} - \alpha - \beta} u^{n-1} du}{(1 + u^2 - 2u(e_1, \theta))^{\frac{n-\alpha}{2}} (1 + u^2 + 2u(e_1, \theta))^{\frac{n-\beta}{2}}} \\ &= \int_{\mathbb{R}^n} \frac{|x|^{-\frac{n}{q}} dx}{|e_1 - x|^{n-\alpha} |e_1 + x|^{n-\beta}}. \end{split}$$

To complete the proof, we need to consider the general case, that is, arbitrary nonnegative functions f and g. Since $x \to |x|^{n-\alpha}$, $x \to |x|^{n-\beta}$ are symmetric-decreasing functions vanishing at infinity, the general rearrangement inequality implies that

$$\int_{\mathbb{R}^{2n}} \frac{f(x)g(y)}{|x-y|^{n-\alpha}|x+y|^{n-\beta}} \, dx dy \le \int_{\mathbb{R}^{2n}} \frac{f^*(x)g^*(y)}{|x-y|^{n-\alpha}|x+y|^{n-\beta}} \, dx dy.$$
(3.25)

Clearly, by (3.24), the right-hand side of (3.25) is not greater than

$$\left(\frac{n}{|\mathbb{S}^{n-1}|}\right)^{1-\lambda} C(p,q;\alpha,\beta;n) \|f^*\|_{L^p(\mathbb{R}^n)} \|g^*\|_{L^q(\mathbb{R}^n)} = \left(\frac{n}{|\mathbb{S}^{n-1}|}\right)^{1-\lambda} C(p,q;\alpha,\beta;n) \|f\|_{L^p(\mathbb{R}^n)} \|g\|_{L^q(\mathbb{R}^n)}, \qquad (3.26)$$

where $C(p,q;\alpha,\beta;n)$ is the constant from the right-hand side of (3.21). To get equality in (3.26), we used the fact that the symmetric-decreasing rearrangement is norm preserving.

On the other hand, substituting the function

$$g(y) = \left[\int_{\mathbb{R}^n} \frac{f(x)}{|x - y|^{n - \alpha} |x + y|^{n - \beta}} dx \right]^{\frac{d'}{q}}$$

in (3.21), we obtain inequality (3.22). To show that inequality (3.22) implies (3.21), we proceed in the same way as in the proof of Theorem 2.1. \Box

It is very interesting to consider the case n = 1 in the previous theorem. Namely, in that case the constant $C(p,q;\alpha,\beta;n)$ can be expressed in terms of the usual Beta function and the Gaussian hypergeometric function (see formulas (1.5) and (1.34), Chapter 1). More precisely, utilizing the above mentioned definitions, it follows easily that the following identity holds for $0 < d_1, d_2, d_3 < 1$ and $d_1 + d_2 + d_3 > 1$:

$$\begin{split} &\int_{\mathbb{R}} |t|^{-d_2} |1-t|^{-d_3} |1+t|^{-d_1} dt = \\ & B(1-d_2,1-d_3) F(d_1,1-d_2;2-d_2-d_3;-1) \\ & + B(1-d_2,1-d_1) F(d_3,1-d_2;2-d_2-d_1;-1) \\ & + B(d_1+d_2+d_3-1,1-d_3) F(d_1,d_1+d_2+d_3-1;d_1+d_2;-1) \\ & + B(d_1+d_2+d_3-1,1-d_1) F(d_3,d_1+d_2+d_3-1;d_3+d_2;-1). \end{split}$$

Hence, for n = 1 we have:

Corollary 3.1 Let p, q, and λ be as in (2.1) and (2.2), and let α and β be real parameters satisfying $0 < \alpha < 1$, $0 < \beta < 1$, and $\alpha + \beta = \frac{1}{p} + \frac{1}{q} > 1$. If $f \in L^p(\mathbb{R})$ and $g \in L^q(\mathbb{R})$ are non-negative functions, then the inequalities

$$\int_{\mathbb{R}^2} \frac{f(x)g(y)}{|x-y|^{1-\alpha}|x+y|^{1-\beta}} dx dy \le 2^{\lambda-1} C(p,q;\alpha,\beta) \|f\|_{L^p(\mathbb{R})} \|g\|_{L^q(\mathbb{R})}$$
(3.27)

and

$$\left\{\int_{\mathbb{R}} \left[\int_{\mathbb{R}} \frac{f(x)dx}{|x-y|^{1-\alpha}|x+y|^{1-\beta}}\right]^{q'} dy\right\}^{\frac{1}{q'}} \le 2^{\lambda-1} C(p,q;\alpha,\beta) \|f\|_{L^{p}(\mathbb{R})}$$
(3.28)

hold and are equivalent, where

$$C(p,q;\alpha,\beta) = B\left(\frac{1}{q'},\alpha\right)F\left(1-\beta,\frac{1}{q'};\frac{1}{q'}+\alpha;-1\right) +B\left(\frac{1}{q'},\beta\right)F\left(1-\alpha,\frac{1}{q'};\frac{1}{q'}+\beta;-1\right)+B\left(\frac{1}{p'},\alpha\right)F\left(1-\beta,\frac{1}{p'};\frac{1}{p'}+\alpha;-1\right) +B\left(\frac{1}{p'},\beta\right)F\left(1-\alpha,\frac{1}{p'};\frac{1}{p'}+\beta;-1\right).$$
(3.29)

The following corollary should be compared with Theorem 3.3.

Corollary 3.2 If p, q, and λ are as in (2.1) and (2.2), and $f \in L^p(\mathbb{R})$, $g \in L^q(\mathbb{R})$ are non-negative functions, then the following inequalities hold and are equivalent:

$$\int_{\mathbb{R}^{2}} \frac{f(x)g(y)dxdy}{|x^{2}-y^{2}|^{\frac{\lambda}{2}}} \leq 2^{\lambda-1} \left(B\left(1-\frac{\lambda}{2},\frac{1}{2p'}\right) + B\left(1-\frac{\lambda}{2},\frac{1}{2q'}\right) \right) \|f\|_{L^{p}(\mathbb{R})} \|g\|_{L^{q}(\mathbb{R})}, \tag{3.30}$$

$$\left[\int_{\mathbb{R}} \left(\int_{\mathbb{R}} \frac{f(x)dx}{|x^{2}-y^{2}|^{\frac{\lambda}{2}}} \right)^{q'} dy \right]^{\frac{1}{q'}} \leq 2^{\lambda-1} \left(B\left(1-\frac{\lambda}{2},\frac{1}{2p'}\right) + B\left(1-\frac{\lambda}{2},\frac{1}{2q'}\right) \right) \|f\|_{L^{p}(\mathbb{R})}. \tag{3.31}$$

Proof. Set $\alpha = \beta = 1 - \frac{\lambda}{2}$ in the previous corollary.

It should be noticed here that inequalities (3.30) and (3.31) could not be obtained from Theorem 3.3. In other words, there are no α and β for which the kernel appearing in inequalities (3.19) and (3.20) would reduce to $|x^2 - y^2|^{-\frac{\lambda}{2}}$.

3.3 Multiple Hilbert-type inequalities via the Selberg integral formula

In this section we derive some multidimensional versions of Hilbert-type inequalities with the help of the Selberg integral formula. The following result can be regarded as the corresponding analogue of Theorem 2.6.

Theorem 3.5 Let λ , p_i , p'_i , q_i , i = 1, 2, ..., k, $k \ge 2$, be real numbers satisfying (2.26), (2.27), and (2.30), and let A_{ij} , i, j = 1, ..., k, be real parameters such that $\sum_{i=1}^{k} A_{ij} = 0$ for j = 1, ..., k. If $n \in \mathbb{N}$, 0 < s < n, $A_{ij} \in (-\frac{n}{q_i}, 0)$, $\alpha_i - A_{ii} < \frac{s - (k - 1)n}{q_i}$, and

$$K = \frac{1}{\Gamma_n^{\lambda}(s)} \prod_{i,j=1, i \neq j}^k \Gamma_n^{\frac{1}{q_i}} (n+q_i A_{ij}) \prod_{i=1}^k \Gamma_n^{\frac{1}{q_i}} (s-(k-1)n-q_i \alpha_i + q_i A_{ii})$$

then the inequalities

$$\int_{\mathbb{R}^{k_n}} \frac{\prod_{i=1}^k f_i(x_i)}{|\sum_{i=1}^k x_i|^{\lambda_s}} dx_1 dx_2 \cdots dx_k \le K \prod_{i=1}^k \left[\int_{\mathbb{R}^n} |x_i|^{\frac{p_i(k-1)n - p_i s}{q_i}} + p_i \alpha_i f_i^{p_i}(x_i) dx_i \right]^{\frac{1}{p_i}}$$
(3.32)

and

$$\begin{cases} \int_{\mathbb{R}^{n}} |x_{k}|^{-\frac{p_{k}'}{q_{k}}[(k-1)n-s]-p_{k}'\alpha_{k}} \left[\int_{\mathbb{R}^{(k-1)n}} \frac{\prod_{i=1}^{k-1} f_{i}(x_{i})}{|\sum_{i=1}^{k} x_{i}|^{\lambda_{s}}} dx_{1} dx_{2} \cdots dx_{k-1} \right]^{p_{k}'} dx_{k} \end{cases}^{\frac{1}{p_{k}'}} \\ \leq K \prod_{i=1}^{k-1} \left[\int_{\mathbb{R}^{n}} |x_{i}|^{\frac{p_{i}(k-1)n-p_{i}s}{q_{i}}+p_{i}\alpha_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}} \tag{3.33}$$

hold for all non-negative measurable functions $f_i : \mathbb{R}^n \to \mathbb{R}$, i = 1, ..., k. Here $\alpha_i = \sum_{j=1}^k A_{ij}$, i = 1, 2, ..., k. Moreover, these inequalities are equivalent.

Proof. We use the general result in the non-conjugate multidimensional setting, that is, Theorem 2.5. Namely, we consider inequalities (2.31) and (2.32) with $\Omega = \mathbb{R}^n$, $\phi_{ij}(x_j) = |x_j|^{A_{ij}}$, i, j = 1, ..., k, $K(x_1, ..., x_k) = |x_1 + \cdots + x_k|^{-s}$, and the Lebesgue measure

Now, using the notation from Theorem 2.5, the Selberg integral formula (3.6) yields

$$F_{i}(x_{i}) = \left[\frac{\prod_{j=1, j\neq i}^{k} \Gamma_{n}(n+q_{i}A_{ij})\Gamma_{n}(n-s)}{\Gamma_{n}(kn+q_{i}\alpha_{i}-q_{i}A_{ii}-s)}\right]^{\frac{1}{q_{i}}} |x_{i}|^{\frac{(k-1)n-s}{q_{i}}+\alpha_{i}-A_{ii}},$$

so inequalities (2.31) and (2.32) reduce to (3.32) and (3.33) respectively.

Remark 3.2 Observe that, according to Remark 2.11, equalities in (3.32) and (3.33) hold if and only if at least one of the functions f_i is equal to zero a.e. on \mathbb{R}^n . Otherwise, inequalities (3.32) and (3.33) are strict.

As an application of Theorem 3.5, we consider some particular choices of parameters A_{ij} , i, j = 1, 2, ..., k. For example, if $A_{ii} = (nk - s)\frac{\lambda q_i - 1}{q_i^2}$ and $A_{ij} = (s - nk)\frac{1}{q_i q_j}$, $i \neq j$, i, j = 1, 2, ..., k, then

$$\sum_{i=1}^{k} A_{ij} = \sum_{i \neq j} \frac{s - nk}{q_i q_j} + (nk - s) \left(\frac{\lambda q_j - 1}{q_j^2}\right) = \frac{s - nk}{q_j} \left(\sum_{i=1}^{k} \frac{1}{q_i} - \lambda\right) = 0,$$

for j = 1, 2, ..., k, that is, these parameters fulfill conditions of Theorem 3.5. Moreover, due to the symmetry, we also have $\alpha_i = \sum_{j=1}^k A_{ij} = 0$ for i = 1, 2, ..., k, and hence, as a consequence we obtain the following result.

Corollary 3.3 Let λ , p_i , p'_i , q_i , i = 1, 2, ..., k, $k \ge 2$, be real numbers satisfying (2.26), (2.27), and (2.30). If $n \in \mathbb{N}$, $0 < nk - s < n \min\{p_i, q_j, i, j = 1, 2, ..., k\}$, and

$$L = \frac{1}{\Gamma_n^{\lambda}(s)} \prod_{i=1}^k \Gamma_n^{\frac{1}{p_i'}} \left(n - \frac{nk-s}{q_i} \right) \prod_{i=1}^k \Gamma_n^{\frac{1}{q_i}} \left(n - \frac{nk-s}{p_i} \right),$$

then the inequalities

$$\int_{\mathbb{R}^{kn}} \frac{\prod_{i=1}^{k} f_i(x_i)}{|\sum_{i=1}^{k} x_i|^{\lambda s}} dx_1 dx_2 \cdots dx_k \le L \prod_{i=1}^{k} \left[\int_{\mathbb{R}^n} |x_i|^{\frac{p_i(k-1)n - p_i s}{q_i}} f_i^{p_i}(x_i) dx_i \right]^{\frac{1}{p_i}}$$
(3.34)

and

$$\begin{cases} \int_{\mathbb{R}^{n}} |x_{k}|^{-\frac{p_{k}'}{q_{k}}[(k-1)n-s]} \left[\int_{\mathbb{R}^{(k-1)n}} \frac{\prod_{i=1}^{k-1} f_{i}(x_{i})}{|\sum_{i=1}^{k} x_{i}|^{\lambda_{s}}} dx_{1} dx_{2} \cdots dx_{k-1} \right]^{p_{k}'} dx_{k} \end{cases}^{\frac{1}{p_{k}'}} \\ \leq L \prod_{i=1}^{k-1} \left[\int_{\mathbb{R}^{n}} |x_{i}|^{\frac{p_{i}(k-1)n-p_{i}s}{q_{i}}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}} \tag{3.35}$$

hold for all non-negative measurable functions $f_i : \mathbb{R}^n \to \mathbb{R}$, i = 1, ..., k. Moreover, these inequalities are equivalent. The equality in both inequalities holds if and only if at least one of the functions f_i , i = 1, 2, ..., k, is equal to zero a.e. on \mathbb{R}^n .

Remark 3.3 Similarly as in the previous corollary, defining $A_{ii} = \frac{n(\lambda q_i - 1)}{\lambda q_i^2}$ and $A_{ij} = -\frac{n}{\lambda q_i q_j}$, $i \neq j, i, j = 1, 2, ..., k$, we have

$$\sum_{i=1}^{k} A_{ij} = \sum_{j=1}^{k} A_{ij} = \sum_{i \neq j} -\frac{n}{\lambda q_i q_j} + \frac{n(\lambda q_j - 1)}{\lambda q_j^2} = -\frac{n}{\lambda q_j} \left(\sum_{i=1}^{k} \frac{1}{q_i} - \lambda \right) = 0,$$

for j = 1, 2, ..., k. Hence, substituting these parameters in Theorem 3.5, assuming that $(k-1)n - s < \frac{n}{\lambda p'_i} < n, i = 1, 2, ..., k$, we obtain the same inequalities as in Corollary 3.3, with the constant

$$L' = \frac{1}{\Gamma_n^{\lambda}(s)} \prod_{i=1}^k \Gamma_n^{\lambda - \frac{1}{q_i}} \left(\frac{n}{\lambda p_i'}\right) \prod_{i=1}^k \Gamma_n^{\frac{1}{q_i}} \left(s + \frac{n}{\lambda p_i'} - (k-1)n\right),$$

instead of L.

Inequalities in Corollary 3.3 and Remark 3.3 are interesting since they provide the best possible constant factors in the conjugate case. This will be explained in the next section. It should also be noticed here that these inequalities can be regarded as the *n*-fold extensions of the corresponding results from [12] and [156].

We proceed with multidimensional Hilbert-type inequalities related to those in Section 3.1. Namely, in Section 3.1, by virtue of Theorem 2.8 and Lemma 3.2, we have

obtained Hardy-Littlewood-Sobolev inequality which includes a pair of non-conjugate parameters. However, the method used in deriving Hardy-Littlewood-Sobolev inequality can also be extended to a multidimensional non-conjugate setting. More precisely, we shall utilize Theorem 2.7 applied to the kernel $K(x_1, \ldots, x_k) = |x_1 + \ldots + x_k|^{-(k-1)n\lambda}$ on \mathbb{R}^{kn} , and Lemma 3.1. The corresponding Hilbert-type inequality is given in the following theorem.

Theorem 3.6 *Let* $n \in \mathbb{N}$, $k \in \mathbb{N}$, $k \ge 2$, and assume that $\lambda, p_i, q_i, i = 1, 2, ..., k$, are as in (2.26), (2.27), and (2.30). If $0 < \lambda < \frac{1}{k-1}$, then the inequality

$$\int_{\mathbb{R}^{kn}} \frac{f_1(x_1) \cdots f_k(x_k)}{|x_1 + \dots + x_k|^{(k-1)n\lambda}} dx_1 \dots dx_k
\leq \frac{(2\pi)^{kn} \left(\frac{|\mathbb{S}^{n-1}|}{n}\right)^{(k-1)(\lambda-1)}}{\Gamma_n((k-1)n\lambda) \prod_{i=1}^k \Gamma_n\left(\frac{n}{p_i}\right)} \|f_1\|_{L^{p_1}(\mathbb{R}^n)} \cdots \|f_k\|_{L^{p_k}(\mathbb{R}^n)}$$
(3.36)

holds for all non-negative functions $f_i \in L^{p_i}(\mathbb{R}^n)$, i = 1, ..., k.

Proof. First, we consider a simpler special case of the functions appearing in (3.36). Namely, suppose that f_2, \ldots, f_k are symmetric-decreasing functions. To prove the above assertion, we rewrite Theorem 2.7 with $\Omega = \mathbb{R}^n$, $K(x_1, \ldots, x_k) = |x_1 + \ldots + x_k|^{-(k-1)n\lambda}$, $F_i(x_1, \ldots, x_k) \equiv 1$, $\phi_{ij}(x_j) = |x_j|^{A_{ij}}$, where

$$A_{ij} = \begin{cases} \frac{n}{p_i p'_i}, & i = j \\ -\frac{n}{q_i p_j}, & i \neq j, \end{cases}$$
(3.37)

and with Lebesgue measures dx_i , for i, j = 1, ..., k. Then the left-hand side of (2.38) in Theorem 2.7 becomes

$$L = \int_{\mathbb{R}^{k_n}} \frac{f_1(x_1) \cdots f_k(x_k)}{|x_1 + \dots + x_k|^{(k-1)n\lambda}} \, dx_1 \dots dx_k,$$
(3.38)

while the right-hand side of this inequality is the product of k + 1 factors,

$$R = I_1^{\frac{1}{q_1}} \cdots I_k^{\frac{1}{q_k}} I_{k+1}^{1-\lambda}, \tag{3.39}$$

where

$$I_{i} = \int_{\mathbb{R}^{kn}} \frac{|x_{i}|^{\frac{n}{p_{i}}} \prod_{j \neq i} |x_{j}|^{-\frac{n}{p_{j}}}}{|x_{1} + \ldots + x_{k}|^{(k-1)n\lambda}} f_{i}^{p_{i}}(x_{i}) dx_{1} \ldots dx_{k}, \ i = 1, \ldots, k,$$

and

$$I_{k+1} = \int_{\mathbb{R}^{k_n}} \frac{\prod_{i=1}^k |x_i|^{\frac{n}{p_i'}}}{|x_1 + \ldots + x_k|^{(k-1)n\lambda}} \prod_{i=1}^k f_i^{p_i}(x_i) \, dx_1 \dots dx_k$$

Before calculating these integrals, observe that from (2.26), (2.27), and (2.30) (see Section 2.4), we obtain that $0 < \frac{n}{p_i} < n$ and

$$\sum_{j \neq i} \frac{n}{p_j} + (k-1)n\lambda = n \sum_{j \neq i} \left(\frac{1}{p_j} + \lambda\right) = n \sum_{j \neq i} \left(\frac{1}{q_j} + 1\right)$$
$$= (k-1)n + n \sum_{j \neq i} \frac{1}{q_j} > (k-1)n,$$

for all $i \in \{1, ..., k\}$. Moreover, the conditions of Theorem 3.6 imply also that $0 < (k - 1)n\lambda < n$. Therefore, applying the Fubini theorem, Lemma 3.1, and (3.5), for i = 1, ..., k, we get

$$I_{i} = \int_{\mathbb{R}^{n}} |x_{i}|^{\frac{n}{p_{i}^{i}}} f_{i}^{p_{i}}(x_{i}) \int_{\mathbb{R}^{(k-1)n}} \frac{\prod_{j \neq i} |x_{j}|^{-\frac{n}{p_{j}}}}{|x_{1} + \ldots + x_{k}|^{(k-1)n\lambda}} dx_{1} \ldots dx_{i-1} dx_{i+1} \ldots dx_{k} dx_{i}$$

$$= \frac{\Gamma_{n}(n - (k-1)n\lambda) \prod_{j \neq i} \Gamma_{n}\left(n - \frac{n}{p_{j}}\right)}{\Gamma_{n}\left(kn - \sum_{j \neq i} \frac{n}{p_{j}} - (k-1)n\lambda\right)} \int_{\mathbb{R}^{n}} |x_{i}|^{\frac{n}{p_{i}^{i}} + (k-1)n - \sum_{j \neq i} \frac{n}{p_{j}} - (k-1)n\lambda} f_{i}^{p_{i}}(x_{i}) dx_{i}$$

$$= \frac{(2\pi)^{kn}}{\Gamma_{n}((k-1)n\lambda) \prod_{j=1}^{k} \Gamma_{n}\left(\frac{n}{p_{j}}\right)} \|f_{i}\|_{L^{p_{i}}(\mathbb{R}^{n})}^{p_{i}}.$$
(3.40)

To estimate the last integral I_{k+1} in (3.39), we use the assumption that the functions f_2, \ldots, f_k are symmetric-decreasing. Hence, we can use relation (3.8) to obtain that

$$f_i^{p_i}(x_i) \le \frac{n}{|\mathbb{S}^{n-1}|} |x_i|^{-n} \parallel f_i \parallel_{L^{p_i}(\mathbb{R}^n)}^{p_i}$$

holds for all $x_i \in \mathbb{R}^n$, $x_i \neq 0$. Again, according to the Fubini theorem, Lemma 3.1, and identity (3.5), similarly to the procedure used in (3.40), we obtain

$$\begin{split} I_{k+1} &\leq \left(\frac{n}{|\mathbb{S}^{n-1}|}\right)^{k-1} \prod_{i=2}^{k} \|f_{i}\|_{L^{p_{i}}(\mathbb{R}^{n})}^{p_{i}} \\ &\qquad \times \int_{\mathbb{R}^{kn}} \frac{|x_{1}|^{\frac{n}{p_{1}'}} \prod_{i=2}^{k} |x_{i}|^{\frac{n}{p_{i}'}-n}}{|x_{1}+\dots+x_{k}|^{(k-1)n\lambda}} f_{1}^{p_{1}}(x_{1}) dx_{1}\dots dx_{k} \\ &= \left(\frac{n}{|\mathbb{S}^{n-1}|}\right)^{k-1} \prod_{i=2}^{k} \|f_{i}\|_{L^{p_{i}}(\mathbb{R}^{n})}^{p_{i}} \\ &\qquad \times \int_{\mathbb{R}^{n}} |x_{1}|^{\frac{n}{p_{1}'}} f_{1}^{p_{1}}(x_{1}) \int_{\mathbb{R}^{(k-1)n}} \frac{\prod_{i=2}^{k} |x_{i}|^{-\frac{n}{p_{i}}}}{|x_{1}+\dots+x_{k}|^{(k-1)n\lambda}} dx_{2}\dots dx_{k} dx_{1} \\ &= \left(\frac{n}{|\mathbb{S}^{n-1}|}\right)^{k-1} \frac{\Gamma_{n}(n-(k-1)n\lambda) \prod_{i=2}^{k} \Gamma_{n}\left(n-\frac{n}{p_{i}}\right)}{\Gamma_{n}\left(kn-\sum_{i=2}^{k} \frac{n}{p_{i}}-(k-1)n\lambda\right)} \prod_{i=1}^{k} \|f_{i}\|_{L^{p_{i}}(\mathbb{R}^{n})}^{p_{i}} \\ &= \left(\frac{n}{|\mathbb{S}^{n-1}|}\right)^{k-1} \frac{(2\pi)^{kn}}{\Gamma_{n}((k-1)n\lambda) \prod_{i=1}^{k} \Gamma_{n}\left(\frac{n}{p_{i}}\right)} \prod_{i=1}^{k} \|f_{i}\|_{L^{p_{i}}(\mathbb{R}^{n})}^{p_{i}} . \end{split}$$
(3.41)

Now, we arrive at inequality (3.36) for this case by combining (3.38), (3.39), (3.40), and (3.41). To complete the proof, we need to consider the general case, that is, arbitrary non-negative functions f_2, \ldots, f_k . Since the function $x \to |x|^{-(k-1)n\lambda}$ is symmetric-decreasing and vanishes at infinity, by the general rearrangement inequality we have

$$\begin{split} \int_{\mathbb{R}^{kn}} \frac{f_1(x_1)\cdots f_k(x_k)}{|x_1+\cdots+x_k|^{(k-1)n\lambda}} dx_1 \dots dx_k &\leq \int_{\mathbb{R}^{kn}} \frac{f_1^*(x_1)\cdots f_k^*(x_k)}{|x_1+\cdots+x_k|^{(k-1)n\lambda}} dx_1 \dots dx_k \\ &\leq \frac{(2\pi)^{kn} \left(\frac{|\underline{\mathbb{S}^{n-1}}|}{n}\right)^{(k-1)(\lambda-1)}}{\Gamma_n((k-1)n\lambda) \prod_{i=1}^k \Gamma_n\left(\frac{n}{p_i}\right)} \parallel f_1^* \parallel_{L^{p_1}(\mathbb{R}^n)} \dots \parallel f_k^* \parallel_{L^{p_k}(\mathbb{R}^n)} \\ &= \frac{(2\pi)^{kn} \left(\frac{|\underline{\mathbb{S}^{n-1}}|}{n}\right)^{(k-1)(\lambda-1)}}{\Gamma_n((k-1)n\lambda) \prod_{i=1}^k \Gamma_n\left(\frac{n}{p_i}\right)} \parallel f_1 \parallel_{L^{p_1}(\mathbb{R}^n)} \dots \parallel f_k \parallel_{L^{p_k}(\mathbb{R}^n)}. \end{split}$$

As in the proof of Theorem 3.2, here we used the fact that f_2^*, \ldots, f_k^* are symmetricdecreasing functions and that the mapping $f \to f^*$ is norm preserving.

Remark 3.4 Note that the proof of Theorem 3.6 is, in fact, based on the Selberg integral formula (3.4). Some further applications of this formula can be found in [25].

Setting k = 2 and k = 3 in Theorem 3.6, we get the following consequences.

Corollary 3.4 Let $n \in \mathbb{N}$, p > 1 and q > 1 be such that $\frac{1}{p} + \frac{1}{q} > 1$, and let λ be defined by (2.2). Then

$$\int_{\mathbb{R}^n} \int_{\mathbb{R}^n} \frac{f(x)g(y)}{|x+y|^{n\lambda}} dx dy \leq \frac{(2\pi)^{2n} \left(\frac{|\mathbb{S}^{n-1}|}{n}\right)^{\lambda-1}}{\Gamma_n(n\lambda)\Gamma_n\left(\frac{n}{p}\right)\Gamma_n\left(\frac{n}{q}\right)} \|f\|_{L^p(\mathbb{R}^n)} \|g\|_{L^q(\mathbb{R}^n)}$$
(3.42)

holds for all non-negative functions $f \in L^p(\mathbb{R}^n)$ and $g \in L^q(\mathbb{R}^n)$. In particular, if n = 1, then (3.42) becomes

$$\int_{-\infty}^{\infty} \int_{-\infty}^{\infty} \frac{f(x)g(y)}{|x+y|^{\lambda}} dx dy \leq 2^{\lambda-1} \sqrt{\pi} \frac{B\left(\frac{1}{2p'}, \frac{1}{2q'}\right)}{B\left(\frac{1}{2p}, \frac{1}{2q}\right)} \frac{\Gamma\left(\frac{1}{2} - \frac{\lambda}{2}\right)}{\Gamma\left(1 - \frac{\lambda}{2}\right)} \parallel f \parallel_{L^{p}(\mathbb{R})} \parallel g \parallel_{L^{q}(\mathbb{R})}.$$

Corollary 3.5 *Let* $n \in \mathbb{N}$ *and assume that parameters* $p_1, p_2, p_3, \lambda, q_1, q_2, q_3$ *satisfy* (2.26), (2.27), *and* (2.30). *If* $0 < \lambda < \frac{1}{2}$, *then*

$$\int_{\mathbb{R}^{3n}} \frac{f(x)g(y)h(z)}{|x+y+z|^{2n\lambda}} dx dy dz$$

$$\leq \frac{(2\pi)^{3n} \left(\frac{|\mathbb{S}^{n-1}|}{n}\right)^{2(\lambda-1)}}{\Gamma_n(2n\lambda)\Gamma_n\left(\frac{n}{p_1}\right)\Gamma_n\left(\frac{n}{p_2}\right)\Gamma_n\left(\frac{n}{p_3}\right)} \|f\|_{L^{p_1}(\mathbb{R}^n)} \|g\|_{L^{p_2}(\mathbb{R}^n)} \|h\|_{L^{p_3}(\mathbb{R}^n)}$$
(3.43)

holds for all non-negative functions $f \in L^{p_1}(\mathbb{R}^n)$, $g \in L^{p_2}(\mathbb{R}^n)$, and $h \in L^{p_3}(\mathbb{R}^n)$. In particular, if n = 1, then (3.43) reads

$$\begin{split} &\int_{-\infty}^{\infty} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} \frac{f(x)g(y)h(z)}{|x+y+z|^{2\lambda}} dx dy dz \\ &\leq 2^{2(\lambda-1)} \pi \frac{\Gamma\left(\frac{1}{2}-\lambda\right)}{\Gamma(\lambda)} \prod_{i=1}^{3} \frac{\Gamma\left(\frac{1}{2p'_{i}}\right)}{\Gamma\left(\frac{1}{2p_{i}}\right)} \parallel f \parallel_{L^{p_{1}}(\mathbb{R})} \parallel g \parallel_{L^{p_{2}}(\mathbb{R})} \parallel h \parallel_{L^{p_{3}}(\mathbb{R})}. \end{split}$$

Remark 3.5 Note that in all presented applications of Theorem 2.7 we considered $F_i \equiv 1$, i = 1, ..., k, while Theorem 2.8 was applied with $F = G \equiv 1$ (see Section 3.1). Obviously, according to the conditions from the statements of these theorems, we can use any other non-negative functions F_i and, consequently, take the infimum of the right-hand sides of the obtained inequalities over all such functions. Therefore, to conclude this section, we mention the following open problem: Can this approach give sharp Hilbert-type inequalities, that is, do there exist such functions F_i that the related inequalities are obtained with the best possible constants on their right-hand sides?

3.4 The best constants

In this section we are focused on Theorem 3.5 in the conjugate setting, that is, when $\sum_{i=1}^{k} \frac{1}{p_i} = 1$ and $\lambda = 1$. The main task is to determine the conditions on parameters A_{ij} , i, j = 1, ..., k, under which the constant factor on the right-hand sides of inequalities (3.32) and (3.33) is the best possible. In the conjugate case, the constant *K*, involved in the above mentioned inequalities, reduces to

$$K = \frac{1}{\Gamma_n(s)} \prod_{i,j=1, i \neq j}^k \Gamma_n^{\frac{1}{p_i}} (n + p_i A_{ij}) \prod_{i=1}^k \Gamma_n^{\frac{1}{p_i}} (s - (k-1)n - p_i \alpha_i + p_i A_{ii}).$$

Taking into account the method as in Section 1.6 (Chapter 1), we are going to simplify the above constant K, to obtain the expression without exponents. Hence, we impose the following conditions on the parameters A_{ii} :

$$s - (k-1)n + p_i A_{ii} - p_i \alpha_i = n + p_j A_{ji}, \quad j \neq i, \quad i, j = 1, 2, \dots, k.$$
(3.44)

In this case the above constant *K* reads

$$K^* = \frac{1}{\Gamma_n(s)} \prod_{i=1}^k \Gamma_n(n + \widetilde{A}_i), \qquad (3.45)$$

where

$$\widetilde{A}_i = p_j A_{ji}, \ j \neq i, \ \text{and} \ -n < \widetilde{A}_i < 0.$$
 (3.46)

It follows easily that then

$$\sum_{i=1}^{k} \widetilde{A}_i = s - kn, \tag{3.47}$$

so that inequalities (3.32) and (3.33) with the parameters A_{ij} fulfilling (3.44), reduce to

$$\int_{\mathbb{R}^{k_n}} \frac{\prod_{i=1}^k f_i(x_i)}{|\sum_{i=1}^k x_i|^s} dx_1 dx_2 \cdots dx_k \le K^* \prod_{i=1}^k \left[\int_{\mathbb{R}^n} |x_i|^{-n-p_i \widetilde{A}_i} f_i^{p_i}(x_i) dx_i \right]^{\frac{1}{p_i}}$$
(3.48)

and

$$\left\{ \int_{\mathbb{R}^{n}} |x_{k}|^{(1-p_{k}')(-n-p_{k}\widetilde{A}_{k})} \left[\int_{\mathbb{R}^{(k-1)n}} \frac{\prod_{i=1}^{k-1} f_{i}(x_{i})}{|\sum_{i=1}^{k} x_{i}|^{s}} dx_{1} dx_{2} \cdots dx_{k-1} \right]^{p_{k}'} dx_{k} \right\}^{\frac{1}{p_{k}'}} \\
\leq K^{*} \prod_{i=1}^{k-1} \left[\int_{\mathbb{R}^{n}} |x_{i}|^{-n-p_{i}\widetilde{A}_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}.$$
(3.49)

Inequalities (3.48) and (3.49) involve the best possible constant factor K^* on their right-hand sides. To show our assertion, we first need to establish some auxiliary results.

Lemma 3.3 Let $k \ge 2$ be an integer, $x_k \in \mathbb{R}^n$, and $x_k \ne 0$. We define

$$I_{1}^{\varepsilon}(x_{k}) = \int_{K^{n}(\varepsilon)} |x_{1}|^{\widetilde{A}_{1}} \left[\int_{\mathbb{R}^{(k-2)n}} \frac{\prod_{i=2}^{k-1} |x_{i}|^{\widetilde{A}_{i}}}{|\sum_{i=1}^{k} x_{i}|^{s}} dx_{2} \dots dx_{k-1} \right] dx_{1},$$

where $\varepsilon > 0$, $K^n(\varepsilon)$ is the closed n-dimensional ball of radius ε , and parameters \widetilde{A}_i , i = 1, 2, ..., k, are defined by (3.46). Then there exists a positive constant C_k such that

$$I_1^{\varepsilon}(x_k) \le C_k \varepsilon^{n+A_1} |x_k|^{-2n-A_1-A_k}, \quad when \quad \varepsilon \to 0.$$
(3.50)

Proof. We distinguish two cases. If k = 2 we have

$$I_1^{\varepsilon}(x_2) = \int_{K^n(\varepsilon)} \frac{|x_1|^{A_1}}{|x_1 + x_2|^s} dx_1.$$

Letting $\varepsilon \to 0$, it follows that there exist a positive constant c_2 such that $I_1^{\varepsilon}(x_2) \le c_2 |x_2|^{-s} \int_{K^n(\varepsilon)} |x_1|^{\widetilde{A}_1} dx_1$. The previous integral can be calculated by using *n*-dimensional spherical coordinates. More precisely, we have

$$\int_{K^{n}(\varepsilon)} |x_{1}|^{\widetilde{A}_{1}} dx_{1}$$

$$= \int_{0}^{\pi} \cdots \int_{0}^{\pi} \int_{0}^{2\pi} \int_{0}^{\varepsilon} r^{n+\widetilde{A}_{1}-1} \sin^{n-2}\theta_{n-1} \sin^{n-3}\theta_{n-2} \cdots \sin\theta_{2} dr d\theta_{1} \dots d\theta_{n-1}$$

$$= \int_{0}^{\varepsilon} r^{n+\widetilde{A}_{1}-1} dr \int_{|\mathbb{S}^{n-1}|} dS = \frac{|\mathbb{S}^{n-1}|\varepsilon^{n+\widetilde{A}_{1}}}{n+\widetilde{A}_{1}}.$$
(3.51)

Consequently,

$$I_1^{\varepsilon}(x_2) \le \frac{c_2 |\mathbb{S}^{n-1}| \varepsilon^{n+\widetilde{A}_1}}{n+\widetilde{A}_1} |x_2|^{-s}$$

so the inequality holds when $\varepsilon \to 0$, since $-2n - \widetilde{A}_1 - \widetilde{A}_2 = -s$ holds for k = 2. Further, if k > 2, then by letting $\varepsilon \to 0$, since $|x_1| \to 0$, we easily conclude that there exists a positive constant c_k such that

$$I_{1}^{\varepsilon}(x_{k}) \leq c_{k} \left[\int_{K^{n}(\varepsilon)} |x_{1}|^{\widetilde{A}_{1}} dx_{1} \right] \left[\int_{\mathbb{R}^{(k-2)n}} \frac{\prod_{i=2}^{k-1} |x_{i}|^{\widetilde{A}_{i}}}{|\sum_{i=2}^{k} x_{i}|^{s}} dx_{2} \dots dx_{k-1} \right].$$
(3.52)

We have already calculated the first integral in inequality (3.52), and the second one is the Selberg integral. Namely, utilizing formulas (3.5), (3.6) (see Section 3.1), and (3.47) we have

$$\int_{\mathbb{R}^{(k-2)n}} \frac{\prod_{i=2}^{k-1} |x_i|^{\widetilde{A}_i}}{|\sum_{i=2}^{k} x_i|^s} dx_2 \dots dx_{k-1}$$

= $\frac{\Gamma_n (2n + \widetilde{A}_1 + \widetilde{A}_k) \prod_{i=2}^{k-1} \Gamma_n (n + \widetilde{A}_i)}{\Gamma_n (s)} |x_k|^{-2n - \widetilde{A}_1 - \widetilde{A}_k}.$ (3.53)

Finally, combining (3.51), (3.52), and (3.53) we obtain inequality (3.50) and the proof is completed. $\hfill \Box$

Lemma 3.4 *Let* $k \ge 2$ *be an integer and* $x_k \in \mathbb{R}^n$ *. We define*

$$I_{1}^{\varepsilon^{-1}}(x_{k}) = \int_{\mathbb{R}^{n} \setminus K^{n}(\varepsilon^{-1})} |x_{1}|^{\widetilde{A}_{1}} \left[\int_{\mathbb{R}^{(k-2)n}} \frac{\prod_{i=2}^{k-1} |x_{i}|^{\widetilde{A}_{i}}}{|\sum_{i=1}^{k} x_{i}|^{s}} dx_{2} \dots dx_{k-1} \right] dx_{1}$$

where $\varepsilon > 0$, $K^n(\varepsilon^{-1})$ is the closed n-dimensional ball of radius ε^{-1} , and parameters \widetilde{A}_i , i = 1, 2, ..., k, are defined by (3.46). Then there exists a positive constant D_k such that

$$I_1^{\varepsilon^{-1}}(x_k) \le D_k \varepsilon^{n+A_k}, \quad \text{when} \quad \varepsilon \to 0.$$
(3.54)

Proof. We again consider two cases. If k = 2 we have

$$I_1^{\varepsilon^{-1}}(x_2) = \int_{\mathbb{R}^n \setminus K^n(\varepsilon^{-1})} \frac{|x_1|^{A_1}}{|x_1 + x_2|^s} dx_1.$$

If $\varepsilon \to 0$, then $|x_1| \to \infty$, hence there exist a positive constant d_2 such that

$$I_1^{\varepsilon^{-1}}(x_2) \leq d_2 \int_{\mathbb{R}^n \setminus K^n(\varepsilon^{-1})} |x_1|^{\widetilde{A}_1 - s} dx_1.$$

Now, utilizing spherical coordinates for calculating the integral on the right-hand side of the above inequality, we obtain

$$I_1^{\varepsilon^{-1}}(x_2) \leq \frac{d_2|\mathbb{S}^{n-1}|}{n+\widetilde{A}_2}\varepsilon^{n+\widetilde{A}_2}.$$

Moreover, for k > 2, using (3.5), (3.6), and (3.47) we have

$$\int_{\mathbb{R}^{(k-2)n}} \frac{\prod_{i=2}^{k-1} |x_i|^{A_i}}{|\sum_{i=1}^k x_i|^s} dx_2 \dots dx_{k-1}$$

=
$$\frac{\Gamma_n (2n + \widetilde{A}_1 + \widetilde{A}_k) \prod_{i=2}^{k-1} \Gamma_n (n + \widetilde{A}_i)}{\Gamma_n (s)} |x_1 + x_k|^{-2n - \widetilde{A}_1 - \widetilde{A}_k}.$$
 (3.55)

Therefore, we obtain

=

$$I_{1}^{\varepsilon^{-1}}(x_{k}) = \frac{\Gamma_{n}(2n+\widetilde{A}_{1}+\widetilde{A}_{k})\prod_{i=2}^{k-1}\Gamma_{n}(n+\widetilde{A}_{i})}{\Gamma_{n}(s)} \times \int_{\mathbb{R}^{n}\setminus K^{n}(\varepsilon^{-1})} |x_{1}|^{\widetilde{A}_{1}}|x_{1}+x_{k}|^{-2n-\widetilde{A}_{1}-\widetilde{A}_{k}}dx_{1}.$$
(3.56)

Letting $\varepsilon \to 0$, it follows that $|x_1| \to \infty$, so there exist a positive constant d_k such that

$$I_1^{\varepsilon^{-1}}(x_k) \le d_k \int_{\mathbb{R}^n \setminus K^n(\varepsilon^{-1})} |x_1|^{-2n - \widetilde{A}_k} dx_1.$$

Finally, since

$$\int_{\mathbb{R}^n \setminus K^n(\varepsilon^{-1})} |x_1|^{-2n - \widetilde{A}_k} dx_1 = \frac{|\mathbb{S}^{n-1}|\varepsilon^{n+\widetilde{A}_k}}{n + \widetilde{A}_k},$$

inequality (3.54) holds.

Now, we are able to obtain the best possible constant factors in (3.48) and (3.49). Clearly, inequalities (3.48) and (3.49) do not contain parameters A_{ij} , i, j = 1, 2, ..., k, so we can regard these inequalities with \tilde{A}_i , i = 1, 2, ..., k, as primitive parameters.

Theorem 3.7 The constant K^* is the best possible in both inequalities (3.48) and (3.49).

Proof. Let $K^n(\varepsilon)$ be the closed *n*-dimensional ball of radius ε , centered at the origin, and let $0 < \varepsilon < 1$. We define the functions $\tilde{f}_i : \mathbb{R}^n \to \mathbb{R}, i = 1, 2, ..., k$, in the following way:

$$\widetilde{f}_i(x_i) = \begin{cases} |x_i|^{\widetilde{A}_i}, \ x_i \in K^n(\varepsilon^{-1}) \setminus K^n(\varepsilon) \\ 0, & \text{otherwise.} \end{cases}$$

Inserting the above functions in (3.48), the right-hand side of (3.48) becomes

$$K^* \prod_{i=1}^k \left(\int_{K^n(\varepsilon^{-1}) \setminus K^n(\varepsilon)} |x_i|^{-n} dx_i \right)^{\frac{1}{p_i}} = K^* \int_{K^n(\varepsilon^{-1}) \setminus K^n(\varepsilon)} |x_i|^{-n} dx_i.$$

In addition, utilizing n-dimensional spherical coordinates we have

$$\int_{K^n(\varepsilon^{-1})\setminus K^n(\varepsilon)} |x_i|^{-n} dx_i = \int_{\varepsilon}^{\varepsilon^{-1}} r^{-1} dr \int_{|\mathbb{S}^{n-1}|} dS = |\mathbb{S}^{n-1}| \log \frac{1}{\varepsilon^2},$$

114

so that the right-hand side of inequality (3.48) reads

$$K^* |\mathbb{S}^{n-1}| \log \frac{1}{\varepsilon^2}.$$
(3.57)

Now, let J denotes the left-hand side of inequality (3.48). Using the Fubini theorem, for the above choice of functions \tilde{f}_i , we have

$$J = \int_{\left(K^n(\varepsilon^{-1})\setminus K^n(\varepsilon)\right)^k} \frac{\prod_{i=1}^k |x_i|^{\widetilde{A}_i}}{|\sum_{i=1}^k x_i|^s} dx_1 dx_2 \dots dx_k$$

=
$$\int_{K^n(\varepsilon^{-1})\setminus K^n(\varepsilon)} |x_k|^{\widetilde{A}_k} \left[\int_{\left(K^n(\varepsilon^{-1})\setminus K^n(\varepsilon)\right)^{k-1}} \frac{\prod_{i=1}^{k-1} |x_i|^{\widetilde{A}_i}}{|\sum_{i=1}^k x_i|^s} dx_1 dx_2 \dots dx_{k-1} \right] dx_k.$$

Note that the integral J can be transformed in the following way: $J = J_1 - J_2 - J_3$, where

$$J_{1} = \int_{K^{n}(\varepsilon^{-1})\setminus K^{n}(\varepsilon)} |x_{k}|^{\widetilde{A}_{k}} \left[\int_{\mathbb{R}^{(k-1)n}} \frac{\prod_{i=1}^{k-1} |x_{i}|^{\widetilde{A}_{i}}}{|\sum_{i=1}^{k} x_{i}|^{s}} dx_{1} dx_{2} \dots dx_{k-1} \right] dx_{k},$$

$$J_{2} = \int_{K^{n}(\varepsilon^{-1})\setminus K^{n}(\varepsilon)} |x_{k}|^{\widetilde{A}_{k}} \sum_{j=1}^{k-1} I_{j}^{\varepsilon}(x_{k}) dx_{k},$$

$$J_{3} = \int_{K^{n}(\varepsilon^{-1})\setminus K^{n}(\varepsilon)} |x_{k}|^{\widetilde{A}_{k}} \sum_{j=1}^{k-1} I_{j}^{\varepsilon^{-1}}(x_{k}) dx_{k}.$$

Here, for j = 1, 2, ..., k - 1, the integrals $I_j^{\varepsilon}(x_k)$ and $I_j^{\varepsilon^{-1}}(x_k)$ are defined by

$$I_j^{\varepsilon}(x_k) = \int_{\mathbb{P}_j} \frac{\prod_{i=1}^{k-1} |x_i|^{\widetilde{A}_i}}{|\sum_{i=1}^k x_i|^s} dx_1 dx_2 \dots dx_{k-1},$$

where $\mathbb{P}_{j} = \{(U_{1}, U_{2}, \dots, U_{k-1}); U_{j} = K^{n}(\varepsilon), U_{l} = \mathbb{R}^{n}, l \neq j\}$, and

$$I_{j}^{\varepsilon^{-1}}(x_{k}) = \int_{\mathbb{Q}_{j}} \frac{\prod_{i=1}^{k-1} |x_{i}|^{\widetilde{A}_{i}}}{|\sum_{i=1}^{k} x_{i}|^{s}} dx_{1} dx_{2} \dots dx_{k-1},$$

where $\mathbb{Q}_j = \{(U_1, U_2, \dots, U_{k-1}); U_j = \mathbb{R}^n \setminus K^n(\varepsilon^{-1}), U_l = \mathbb{R}^n, l \neq j\}.$ Now, our aim is to find the lower bound for *J*. The first integral J_1 can easily be computed by virtue of the Selberg integral formula (3.6). More precisely, since parameters \widetilde{A}_i fulfill relation (3.47), it follows that

$$\int_{\mathbb{R}^{(k-1)n}} \frac{\prod_{i=1}^{k-1} |x_i|^{\widetilde{A}_i}}{|\sum_{i=1}^k x_i|^s} dx_1 dx_2 \dots dx_{k-1} = K^* |x_k|^{-\widetilde{A}_k - n},$$

and consequently,

$$J_1 = K^* |\mathbb{S}^{n-1}| \log \frac{1}{\varepsilon^2}.$$
 (3.58)

In the sequel, we show that the parts J_2 and J_3 converge when $\varepsilon \to 0$. For that sake, without loss of generality, it is enough to estimate the integrals

$$\int_{K^n(\varepsilon^{-1})\setminus K^n(\varepsilon)} |x_k|^{\widetilde{A}_k} I_1^{\varepsilon}(x_k) dx_k \quad \text{and} \quad \int_{K^n(\varepsilon^{-1})\setminus K^n(\varepsilon)} |x_k|^{\widetilde{A}_k} I_1^{\varepsilon^{-1}}(x_k) dx_k.$$

Utilizing Lemma 3.4 and n-dimensional spherical coordinates we have

$$\begin{split} \int_{K^{n}(\varepsilon^{-1})\setminus K^{n}(\varepsilon)} |x_{k}|^{\widetilde{A}_{k}} I_{1}^{\varepsilon}(x_{k}) dx_{k} &\leq C_{k} \varepsilon^{n+\widetilde{A}_{1}} \int_{K^{n}(\varepsilon^{-1})\setminus K^{n}(\varepsilon)} |x_{k}|^{-2n-\widetilde{A}_{1}} dx_{k} \\ &= C_{k} |\mathbb{S}^{n-1}| \varepsilon^{n+\widetilde{A}_{1}} \int_{\varepsilon}^{\varepsilon^{-1}} r^{-n-\widetilde{A}_{1}-1} dr \\ &= \frac{C_{k} |\mathbb{S}^{n-1}|}{n+\widetilde{A}_{1}} \left(1 - \varepsilon^{2(n+\widetilde{A}_{1})}\right). \end{split}$$

Similarly, we also have

$$egin{aligned} &\int_{K^n(arepsilon^{-1})ackslash K^n(arepsilon^{-1})ackslash K^n(arepsilon^{-1})allon^{-1}allon^{-1} allon^{-1} allon^{$$

Clearly, since $n + \widetilde{A}_i > 0$, i = 1, 2, ..., k, the above computation shows that $J_2 + J_3 \le O(1)$, when $\varepsilon \to 0$. Hence, taking into account relation (3.58), it follows that *J* is bounded from below by

$$K^* |\mathbb{S}^{n-1}| \log \frac{1}{\varepsilon^2} - O(1), \text{ when } \varepsilon \to 0.$$
 (3.59)

Now, suppose that there exists a positive constant L^* , $0 < L^* < K^*$, such that inequality (3.48) holds with the constant L^* instead of K^* . In that case, for the above choice of functions \tilde{f}_i , the right-hand side of inequality (3.48) becomes $L^*|\mathbb{S}^{n-1}|\log \frac{1}{\varepsilon^2}$. Since $L^*|\mathbb{S}^{n-1}|\log \frac{1}{\varepsilon^2} \ge J$, relation (3.59) implies that

$$(K^* - L^*) |\mathbb{S}^{n-1}| \log \frac{1}{\varepsilon^2} \le O(1), \quad \text{when} \quad \varepsilon \to 0.$$
(3.60)

Finally, letting $\varepsilon \to 0$, we obtain a contradiction, since the left-hand side of inequality (3.60) goes to infinity. This shows that the constant K^* is the best possible in (3.48). Due to the equivalence, K^* is also the best possible in inequality (3.49).

Remark 3.6 A straightforward computation shows that the parameters A_{ij} , providing the inequalities from Corollary 3.3, fulfill the set of conditions (3.44) in the conjugate case. Then, the constant *L* from Corollary 3.3 becomes

$$L = \frac{1}{\Gamma_n(s)} \prod_{i=1}^k \Gamma_n\left(n - \frac{nk - s}{p_i}\right),$$

and that is the best possible constant in the corresponding inequalities.

Remark 3.7 Hilbert-type inequalities from this chapter, obtained by virtue of the Selberg integral formula, are taken from [17] and [61].

3.5 Some related inequalities with norms

So far, in this chapter we have studied Hilbert-type inequalities established by virtue of the Selberg integral formula. As distinguished from the previous sections, we consider here multidimensional Hilbert-type inequalities with a homogeneous kernel and the power weight functions whose arguments are α -norms of the corresponding vectors. Recall, if $t = (t_1, t_2, \dots, t_n) \in \mathbb{R}^n$, then α -norm ($\alpha \ge 1$) of vector t is $|t|_{\alpha} = (t_1^{\alpha} + t_2^{\alpha} + \dots + t_n^{\alpha})^{\frac{1}{\alpha}}$.

We shall be concerned here with a general homogeneous kernel $K_{\alpha} : \mathbb{R}^k_+ \to \mathbb{R}$ of degree -s, s > 0. Hence, the constant factors in the corresponding inequalities will include the integral

$$k_{\alpha}(\beta_{1},\ldots,\beta_{k-1}) = \int_{\mathbb{R}^{k-1}_{+}} K_{\alpha}(1,t_{1}\ldots,t_{k-1})t_{1}^{\beta_{1}}\cdots t_{k-1}^{\beta_{k-1}}dt_{1}\cdots dt_{k-1}.$$

The above integral is assumed to converge for $\beta_1, \ldots, \beta_{k-1} > -1$ and $\beta_1 + \cdots + \beta_{k-1} + k < s + 1$ (see relation (1.76), Section 1.6, Chapter 1).

On the other hand, since the arguments of the kernel and the weight functions will be expressed by means of α -norm, constant factors in the corresponding Hilbert-type inequalities will also include the area of the unit sphere in \mathbb{R}^n . The area of the unit sphere in view of the α -norm, denoted here by $|\mathbb{S}^{n-1}|_{\alpha}$, is

$$|\mathbb{S}^{n-1}|_{\alpha} = \frac{2^n \Gamma^n(\frac{1}{\alpha})}{\alpha^{n-1} \Gamma(\frac{n}{\alpha})}.$$
(3.61)

It should be noticed here that the above formula (3.61) coincides with (3.3) when $\alpha = 2$.

The following Theorem can be regarded as a vector extension of Theorem 1.19 (see Section 1.6, Chapter 1), in view of α -norm.

Theorem 3.8 Let λ , p_i , p'_i , q_i , i = 1, 2, ..., k, $k \ge 2$, be parameters satisfying conditions (2.26), (2.27), and (2.30), and let A_{ij} , i, j = 1, ..., k, be such that $\sum_{i=1}^{k} A_{ij} = 0$ for j = 1, ..., k, $q_i A_{ij} > -n$, $i \ne j$, and $q_i (A_{ii} - \alpha_i) > (k - 1)n - s$, where $n \in \mathbb{N}$, s > 0, and $\alpha_i = \sum_{j=1}^{k} A_{ij}$, i = 1, 2, ..., k. If $K_{\alpha} : \mathbb{R}^k_+ \to \mathbb{R}$ is a non-negative measurable homogeneous function of degree -s and

$$L = \frac{|\mathbb{S}^{n-1}|_{\alpha}^{(k-1)\lambda}}{2^{(k-1)n\lambda}} k_{\alpha}^{\frac{1}{q_1}} (n-1+q_1A_{12},\dots,n-1+q_1A_{1k}) \\ \times k_{\alpha}^{\frac{1}{q_2}} (s-(k-1)n-1-q_2(\alpha_2-A_{22}),n-1+q_2A_{23},\dots,n-1+q_2A_{2k}) \\ \cdots k_{\alpha}^{\frac{1}{q_k}} (n-1+q_kA_{k2},\dots,n-1+q_kA_{k,k-1},s-(k-1)n-1-q_k(\alpha_k-A_{kk})),$$
(3.62)

then the inequalities

$$\int_{\mathbb{R}^{nk}_{+}} K_{\alpha}^{\lambda}(|x_{1}|_{\alpha}, \dots, |x_{k}|_{\alpha}) \prod_{i=1}^{k} f_{i}(x_{i}) dx_{1} \dots dx_{k}$$

$$< L \prod_{i=1}^{k} \left[\int_{\mathbb{R}^{n}_{+}} |x_{i}|_{\alpha}^{\frac{p_{i}}{q_{i}}[(k-1)n-s]+p_{i}\alpha_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$

$$(3.63)$$

and

$$\begin{bmatrix}
\int_{\mathbb{R}^{n}_{+}} |x_{k}|_{\alpha}^{-\frac{p'_{k}}{q_{k}}[(k-1)n-s]-p'_{k}\alpha_{k}} \left(\int_{\mathbb{R}^{n(k-1)}_{+}} K_{\alpha}^{\lambda}(|x_{1}|_{\alpha}, \dots, |x_{k}|_{\alpha}) \\
\times \prod_{i=1}^{k-1} f_{i}(x_{i})dx_{1}\dots dx_{k-1} \right)^{p'_{k}} dx_{k} \end{bmatrix}^{\frac{1}{p'_{k}}} \\
< L \prod_{i=1}^{k-1} \left[\int_{\mathbb{R}^{n}_{+}} |x_{i}|_{\alpha}^{\frac{p_{i}}{q_{i}}[(k-1)n-s]+p_{i}\alpha_{i}} f_{i}^{p_{i}}(x_{i})dx_{i} \right]^{\frac{1}{p_{i}}} (3.64)$$

hold for all non-negative measurable functions $f_i : \mathbb{R}^n_+ \to \mathbb{R}$, $f_i \neq 0$, i = 1, ..., k. Moreover, these inequalities are equivalent.

Proof. We utilize Theorem 1.18 (see Section 1.6) with the homogeneous kernel $K_{\alpha}(|x_1|_{\alpha},...,|x_k|_{\alpha})$, the weight functions $\phi_{ij}(x_j) = |x_j|^{A_{ij}}$, and Lebesgue measures on \mathbb{R}^n_+ . Using notation from Theorem 1.18, it is enough to calculate the functions $F_i(x_i)$, i = 1,...,k. Utilizing *n*-dimensional spherical coordinates we find that

$$F_1^{q_1}(x_1) = \int_{\mathbb{R}^{n(k-1)}_+} K_\alpha(|x_1|_\alpha, \dots, |x_k|_\alpha) \prod_{j=2}^k |x_j|^{q_1 A_{1j}} dx_2 \cdots dx_k$$

= $\frac{|\mathbb{S}^{n-1}|_\alpha^{k-1}}{2^{(k-1)n}} \int_{\mathbb{R}^{k-1}_+} K_\alpha(|x_1|_\alpha, t_2, \dots, t_k) \prod_{j=2}^k t_j^{n-1+q_1 A_{1j}} dt_2 \cdots dt_k.$

Moreover, due to homogeneity of the kernel K_{α} and using the change of variables $u_i = \frac{l_i}{|x_i|_{\alpha}}$, i = 2, ..., k, we obtain

$$F_{1}^{q_{1}}(x_{1}) = \frac{\left|\mathbb{S}^{n-1}\right|_{\alpha}^{k-1}}{2^{(k-1)n}} \int_{\mathbb{R}^{k-1}_{+}} |x_{1}|_{\alpha}^{-s} K_{\alpha}(1, u_{2}, \dots, u_{k})$$

$$\times \prod_{j=2}^{k} (|x_{1}|_{\alpha} u_{j})^{n-1+q_{1}A_{1j}} |x_{1}|_{\alpha}^{k-1} du_{2} \dots du_{k}$$

$$= \frac{\left|\mathbb{S}^{n-1}\right|_{\alpha}^{k-1}}{2^{(k-1)n}} |x_{1}|_{\alpha}^{(k-1)n-s+q_{1}(\alpha_{1}-A_{11})} k_{\alpha}(n-1+q_{1}A_{12}, \dots, n-1+q_{1}A_{1k}).$$

Similarly,

$$\begin{split} F_2^{q_2}(x_2) &= \int_{\mathbb{R}^{n(k-1)}_+} K_\alpha(|x_1|_\alpha, \dots, |x_k|_\alpha) \prod_{j=1, j \neq 2}^k |x_j|^{q_2 A_{2j}} dx_1 dx_3 \dots dx_k \\ &= \frac{|\mathbb{S}^{n-1}|_\alpha^{k-1}}{2^{(k-1)n}} \int_{\mathbb{R}^{k-1}_+} t_1^{-s} K_\alpha(1, |x_2|_\alpha/t_1, t_3/t_1, \dots, t_k/t_1) \\ &\qquad \times \prod_{j=1, j \neq 2}^k t_j^{n-1+q_2 A_{2j}} dt_1 dt_3 \dots dt_k. \end{split}$$

Hence, the change of variables $t_1 = |x_2|_{\alpha} u_2^{-1}$, $t_i = |x_2|_{\alpha} u_2^{-1} u_i$, i = 3, ..., k, with the Jacobian

$$\frac{\partial(t_1,t_3,\ldots,t_k)}{\partial(u_2,u_3,\ldots,u_k)} = |x_2|_{\alpha}^{k-1}u_2^{-k},$$

yields

$$F_{2}^{q_{2}}(x_{2}) = \frac{\left|\mathbb{S}^{n-1}\right|_{\alpha}^{k-1}}{2^{(k-1)n}} |x_{2}|_{\alpha}^{(k-1)n-s+q_{2}(\alpha_{2}-A_{22})} \\ \times \int_{\mathbb{R}^{k-1}_{+}} K_{\alpha}(1, u_{2}, \dots, u_{k}) u_{2}^{s-(k-1)n-q_{2}(\alpha_{2}-A_{22})} \prod_{j=3}^{k} u_{j}^{n-1+q_{2}A_{2j}} du_{2} \dots du_{k} \\ = \frac{\left|\mathbb{S}^{n-1}\right|_{\alpha}^{k-1}}{2^{(k-1)n}} |x_{2}|_{\alpha}^{(k-1)n-s-q_{2}(\alpha_{2}-A_{22})} \\ \times k_{\alpha}(s-(k-1)n-1-q_{2}(\alpha_{2}-A_{22}), n-1+q_{2}A_{23}, \dots, n-1+q_{2}A_{2k}).$$

In a similar manner we obtain

$$F_i^{q_i}(x_i) = \frac{|\mathbb{S}^{n-1}|_{\alpha}^{k-1}}{2^{(k-1)n}} |x_i|_{\alpha}^{(k-1)n-s+q_i(\alpha_i-A_{ii})} \\ \times k_{\alpha}(n-1+q_iA_{i2},\ldots,n-1+q_iA_{i,i-1},s-(k-1)n-1-q_i(\alpha_i-A_{ii}), \\ n-1+q_iA_{i,i+1},\ldots,n-1+q_iA_{ik}), \quad i=3,\ldots,k,$$

which yields inequalities (3.63) and (3.64) with the sign \leq . Finally, Remark 1.21 (see Section 1.6) provides the sharpness of the obtained inequalities.

In order to obtain the best constants in (3.63) and (3.64), we consider now these inequalities in the conjugate setting. Similarly to the previous section, we impose certain conditions on parameters A_{ij} , i, j = 1, 2, ..., k, to obtain a simpler form of the constant *L* defined by (3.62). More precisely, if the parameters A_{ij} fulfill conditions

$$n + p_{j}A_{ji} = s - (k - 1)n - p_{i}(\alpha_{i} - A_{ii}), \quad j \neq i, \quad i, j = 1, 2, \dots, k,$$
(3.65)

then, in the conjugate case, constant L reduces to

$$L^* = \frac{|\mathbb{S}^{n-1}|_{\alpha}^{(k-1)}}{2^{(k-1)n}} k_{\alpha} (n-1+\widetilde{A}_2,\dots,n-1+\widetilde{A}_k),$$
(3.66)

where $\widetilde{A}_i = p_1 A_{1i}$, $i \neq 1$, and $\widetilde{A}_1 = p_k A_{k1}$. Moreover, inequalities (3.63) and (3.64) reduce to

$$\int_{\mathbb{R}^{nk}_{+}} K_{\alpha}(|x_{1}|_{\alpha}, \dots, |x_{k}|_{\alpha}) \prod_{i=1}^{k} f_{i}(x_{i}) dx_{1} \dots dx_{k}$$
$$< L^{*} \prod_{i=1}^{k} \left[\int_{\mathbb{R}^{n}_{+}} |x_{i}|_{\alpha}^{-n-p_{i}\widetilde{A}_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$
(3.67)

and

$$\begin{bmatrix}
\int_{\mathbb{R}^{n}_{+}} |x_{k}|_{\alpha}^{(1-p_{k}')(-n-p_{k}\widetilde{A}_{k})} \left(\int_{\mathbb{R}^{n(k-1)}_{+}} K_{\alpha}(|x_{1}|_{\alpha}, \dots, |x_{k}|_{\alpha}) \\
\times \prod_{i=1}^{k-1} f_{i}(x_{i}) dx_{1} \dots dx_{k-1} \right)^{p_{k}'} dx_{k} \end{bmatrix}^{\frac{1}{p_{k}'}} \\
< L^{*} \prod_{i=1}^{k-1} \left[\int_{\mathbb{R}^{n}_{+}} |x_{i}|_{\alpha}^{-n-p_{i}\widetilde{A}_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}.$$
(3.68)

The following result asserts that L^* is the best possible constant in inequalities (3.67) and (3.68), under some weak conditions on the kernel K_{α} .

Theorem 3.9 Suppose that real parameters A_{ij} , i, j = 1, ..., k, fulfill conditions of Theorem 3.8 (conjugate case) and conditions given in (3.65). If the kernel $K_{\alpha} : \mathbb{R}^k_+ \to \mathbb{R}$ satisfies conditions of Theorem 3.8 and for every i = 2, ..., k,

$$K_{\alpha}(1, t_2, \dots, t_i, \dots, t_k) \leq CK_{\alpha}(1, t_2, \dots, 0, \dots, t_k), \ 0 \leq t_i \leq 1, \ t_j \geq 0, \ j \neq i,$$

for some C > 0, then L^* is the best possible constant in inequalities (3.67) and (3.68).

Proof. Suppose that inequality (3.67) holds for all non-negative measurable functions f_i , i = 1, 2, ..., k, when L^* is replaced with a smaller positive constant L_1 . To prove our assertion, we consider inequality (3.67) with the constant L_1 and the functions $\tilde{f}_{i,\varepsilon} : \mathbb{R}^n_+ \to \mathbb{R}$, defined by

$$\widetilde{f}_{i,\varepsilon}(x_i) = \begin{cases} 0, & |x_i|_{\alpha} < 1\\ |x_i|_{\alpha}^{\widetilde{A}_i - \frac{\varepsilon}{p_i}}, & |x_i|_{\alpha} \ge 1 \end{cases}, \quad i = 1, \dots, k,$$

where $0 < \varepsilon < \min_{1 \le i \le k} \{p_i + p_i \widetilde{A}_i\}$. Utilizing *n*-dimensional spherical coordinates, the right-hand side of (3.67) becomes

$$L_{1}\prod_{i=1}^{k} \left[\int_{|x_{i}|_{\alpha} \ge 1} |x_{i}|_{\alpha}^{-n-\varepsilon} dx_{i} \right]^{\frac{1}{p_{i}}} = \frac{L_{1}|\mathbb{S}^{n-1}|_{\alpha}}{2^{n}} \int_{1}^{\infty} t^{-1-\varepsilon} dt = \frac{L_{1}|\mathbb{S}^{n-1}|_{\alpha}}{2^{n}\varepsilon}.$$
 (3.69)

Further, let *J* denote the left-hand side of inequality (3.67), for the above choice of functions $\tilde{f}_{i,\varepsilon}$. Using the change of variables $u_i = \frac{t_i}{t_1}$, $i \neq 2$, we have

$$J = \int_{|x_1|_{\alpha \ge 1}} \cdots \int_{|x_k|_{\alpha \ge 1}} K_{\alpha}(|x_1|_{\alpha}, \dots, |x_k|_{\alpha}) \prod_{i=1}^k |x_i|_{\alpha}^{\widetilde{A}_i - \frac{\varepsilon}{p_i}} dx_1 \dots dx_k$$

$$= \frac{|\mathbb{S}^{n-1}|_{\alpha}^k}{2^{kn}} \int_1^{\infty} \cdots \int_1^{\infty} K_{\alpha}(t_1, \dots, t_k) \prod_{i=1}^k t_i^{n-1+\widetilde{A}_i - \frac{\varepsilon}{p_i}} dt_1 \dots dt_k$$

$$= \frac{|\mathbb{S}^{n-1}|_{\alpha}^k}{2^{kn}} \int_1^{\infty} t_1^{-1-\frac{\varepsilon}{\beta}} \left(\int_{\frac{1}{t_1}}^{\infty} \cdots \int_{\frac{1}{t_1}}^{\infty} K_{\alpha}(1, u_2, \dots, u_k) \prod_{i=2}^k u_i^{n-1+\widetilde{A}_i - \frac{\varepsilon}{p_i}} du_2 \dots du_k \right) dt_1,$$

so, J can be estimated from below as follows:

$$J \ge \frac{|\mathbb{S}^{n-1}|_{\alpha}^{k}}{2^{kn}} \int_{1}^{\infty} t_{1}^{-1-\varepsilon} \left(\int_{0}^{\infty} \cdots \int_{0}^{\infty} K_{\alpha}(1, u_{2}, \dots, u_{k}) \prod_{i=2}^{k} u_{i}^{n-1+\widetilde{A}_{i}-\frac{\varepsilon}{p_{i}}} du_{2} \dots du_{k} \right) dt_{1} - \frac{|\mathbb{S}^{n-1}|_{\alpha}^{k}}{2^{kn}} \int_{1}^{\infty} t_{1}^{-1-\varepsilon} \sum_{j=2}^{k} I_{j}(t_{1}) dt_{1},$$
(3.70)

where

$$I_j(t_1) = \int_{\mathbb{D}_j} K_\alpha(1, u_2, \dots, u_k) \prod_{i=2}^k u_i^{n-1+\widetilde{A}_i - \frac{\varepsilon}{p_i}} du_2 \dots du_k, \ j = 2, \dots, k,$$

and $\mathbb{D}_j = \{(u_2, \dots, u_k); 0 < u_j < \frac{1}{t_1}, 0 < u_l < \infty, l \neq j\}.$

Now, the rest of the proof follows the same lines as the proof of Theorem 1.20 (see Section 1.6, Chapter 1). Namely, utilizing the estimate $\int_1^{\infty} t_1^{-1-\varepsilon} \sum_{j=2}^k I_j(t_1) dt_1 \leq O(1)$, (3.69), and (3.70), it follows that $L^* \leq L_1$, which contradicts assumption that L_1 is smaller than L^* .

Observe that Theorem 3.9 may be regarded as an extension of Theorem 1.20 from Section 1.6.

Remark 3.8 Let $k_s : \mathbb{R}^2_+ \to \mathbb{R}$ be a non-negative homogeneous kernel of degree -s, s > 0, and let parameters A_1 and A_2 fulfill relation $qA_1 + pA_2 = m + n - s$, where p and q are conjugate parameters, and $m, n \in \mathbb{N}$. Assuming that there exists $\delta > 0$ such that $c_s(\eta) = \int_0^\infty k_s(1,t)t^{-\eta}dt < \infty$ for $\eta \in [pA_2 + 1 - n - \delta, pA_2 + 1 - n]$, Yang and Krnić [167], obtained a pair of equivalent inequalities

$$\begin{split} &\int_{\mathbb{R}^m_+} \int_{\mathbb{R}^n_+} k_s \left(|x|_{\alpha}, |y|_{\beta} \right) f(x) g(y) dx dy \\ &\leq C_s \left[\int_{\mathbb{R}^m_+} |x|_{\alpha}^{pqA_1 - m} f^p(x) dx \right]^{\frac{1}{p}} \left[\int_{\mathbb{R}^n_+} |y|_{\beta}^{pqA_2 - n} g^q(y) dy \right]^{\frac{1}{q}} \end{split}$$

and

$$\begin{split} &\left\{\int_{\mathbb{R}^{n}_{+}}|y|\beta^{pqA_{1}+p(s-m)-n}\left[\int_{\mathbb{R}^{m}_{+}}k_{s}\left(|x|_{\alpha},|y|_{\beta}\right)f(x)dx\right]^{p}dy\right\}^{\frac{1}{p}}\\ &\leq C_{s}\left[\int_{\mathbb{R}^{m}_{+}}|x|_{\alpha}^{pqA_{1}-m}f^{p}(x)dx\right]^{\frac{1}{p}}, \end{split}$$

with the best possible constant expressed in terms of the usual Gamma function:

$$C_{s} = \left[\frac{\Gamma^{m}\left(\frac{1}{\alpha}\right)}{\alpha^{m-1}\Gamma\left(\frac{m}{\alpha}\right)}\right]^{\frac{1}{q}} \left[\frac{\Gamma^{n}\left(\frac{1}{\beta}\right)}{\beta^{n-1}\Gamma\left(\frac{n}{\beta}\right)}\right]^{\frac{1}{p}} c_{s}(pA_{2}+1-n).$$

It should be noticed here that these inequalities include two different norms $|\cdot|_{\alpha}$ and $|\cdot|_{\beta}$. Obviously, the above inequalities may also be extended to a multidimensional setting, as in Theorem 3.8.

To conclude this section we consider Theorem 3.8 for some particular choices of the kernel K_{α} and parameters A_{ij} , i, j = 1, 2, ..., k. More precisely, we are concerned here with the homogeneous kernel $K_{\alpha}(t_1, t_2, ..., t_k) = (\sum_{i=1}^{k} t_i^{\beta})^{-s}$ of degree $-\beta s$. In this case we have

$$k_{\alpha}(\beta_1,\ldots,\beta_{k-1}) = \frac{1}{\beta^{k-1}\Gamma(s)}\Gamma\left(s - \sum_{i=1}^{k-1}\frac{\beta_i+1}{\beta}\right)\prod_{i=1}^{k-1}\Gamma\left(\frac{\beta_i+1}{\beta}\right)$$

(see Lemma 1.4, Section 1.6, Chapter 1), so Theorem 3.8 yields the following result:

Corollary 3.6 Let λ , p_i , p'_i , q_i , i = 1, 2, ..., k, $k \ge 2$, be parameters satisfying (2.26), (2.27), and (2.30), and let A_{ij} , i, j = 1, ..., k, be real parameters such that $\sum_{i=1}^{k} A_{ij} = 0$ for j = 1, ..., k, $q_i A_{ij} > -n$, $i \ne j$, and $q_i (A_{ii} - \alpha_i) > (k - 1)n - \beta s$, where $\alpha_i = \sum_{j=1}^{k} A_{ij}$, i = 1, 2, ..., k, $n \in \mathbb{N}$, and s > 0. If

$$\begin{split} K &= \frac{\Gamma^{\lambda(k-1)n}\left(\frac{1}{\alpha}\right)}{\beta^{\lambda(k-1)}\alpha^{\lambda(n-1)(k-1)}\Gamma^{\lambda(k-1)}\left(\frac{n}{\alpha}\right)\Gamma^{\lambda}(s)} \\ &\times \prod_{i=1}^{k} \Gamma^{\frac{1}{q_i}}\left(s - \frac{(k-1)n + q_i\alpha_i - q_iA_{ii}}{\beta}\right) \prod_{i,j=1, i \neq j}^{k} \Gamma^{\frac{1}{q_i}}\left(\frac{n + q_iA_{ij}}{\beta}\right), \end{split}$$

then the inequalities

$$\int_{\mathbb{R}^{nk}_{+}} \frac{\prod_{i=1}^{k} f_i(x_i)}{\left(\sum_{i=1}^{k} |x_i| \alpha^{\beta}\right)^{\lambda_s}} dx_1 dx_2 \cdots dx_k$$
$$\leq K \prod_{i=1}^{k} \left[\int_{\mathbb{R}^{n}_{+}} |x_i| \alpha^{\frac{p_i(k-1)n-p_i\beta_s}{q_i}} + p_i\alpha_i f_i^{p_i}(x_i) dx_i \right]^{\frac{1}{p_i}}$$
(3.71)

and

$$\left\{ \int_{\mathbb{R}^{n}_{+}} |x_{k}|_{\alpha} - \frac{p_{k}'}{q_{k}} [(k-1)n - \beta s] - p_{k}' \alpha_{k} \\
\times \left[\int_{\mathbb{R}^{n}_{+}} \frac{\prod_{i=1}^{k-1} f_{i}(x_{i})}{\left(\sum_{i=1}^{k} |x_{i}|_{\alpha}^{\beta}\right)^{\lambda s}} dx_{1} dx_{2} \cdots dx_{k-1} \right]^{p_{k}'} dx_{k} \right\}^{\frac{1}{p_{k}'}} \\
\leq K \prod_{i=1}^{k-1} \left[\int_{\mathbb{R}^{n}_{+}} |x_{i}|_{\alpha} \frac{p_{i}(k-1)n - p_{i}\beta s}{q_{i}} + p_{i}\alpha_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$
(3.72)

hold for all non-negative measurable functions $f_i : \mathbb{R}^n_+ \to \mathbb{R}$, i = 1, ..., k, and are equivalent.

Remark 3.9 Introducing the parameters $A_{ii} = (nk-s)\frac{\lambda q_i-1}{q_i^2}$ and $A_{ij} = (s-nk)\frac{1}{q_iq_j}$, $i \neq j$, it follows that $\sum_{i=1}^k A_{ij} = \sum_{j=1}^k A_{ij} = 0$ for i, j = 1, 2, ..., k. Hence, in this setting inequalities (3.71) and (3.72) become

$$\int_{\mathbb{R}^{nk}_+} \frac{\prod_{i=1}^k f_i(x_i)}{\left(\sum_{i=1}^k |x_i| \alpha^{\beta}\right)^{\lambda_s}} dx_1 dx_2 \cdots dx_k \le L \prod_{i=1}^k \left[\int_{\mathbb{R}^n_+} |x_i| \alpha^{\frac{p_i(k-1)n-p_i\beta_s}{q_i}} f_i^{p_i}(x_i) dx_i \right]^{\frac{1}{p_i}}$$

and

$$\begin{cases} \int_{\mathbb{R}^{n}_{+}} |x_{k}|_{\alpha} - \frac{p_{k}'}{q_{k}} [(k-1)n - \beta s] \left[\int_{(\mathbb{R}^{n}_{+})^{k-1}} \frac{\prod_{i=1}^{k-1} f_{i}(x_{i})}{\left(\sum_{i=1}^{k} |x_{i}|_{\alpha}^{\beta}\right)^{\lambda s}} dx_{1} dx_{2} \cdots dx_{k-1} \right]^{p_{k}'} dx_{k} \end{cases}^{\frac{1}{p_{k}'}} \\ \leq L \prod_{i=1}^{k-1} \left[\int_{\mathbb{R}^{n}_{+}} |x_{i}|_{\alpha} \frac{p_{i}(k-1)n - p_{i}\beta s}{q_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}, \end{cases}$$

with the constant factor

$$L = \frac{\Gamma^{\lambda(k-1)n}\left(\frac{1}{\alpha}\right)}{\beta^{\lambda(k-1)}\alpha^{\lambda(n-1)(k-1)}\Gamma^{\lambda(k-1)}\left(\frac{n}{\alpha}\right)\Gamma^{\lambda}(s)} \prod_{i=1}^{k} \Gamma^{\frac{1}{q_i}}\left(\frac{\beta s + n(p_i - k)}{\beta p_i}\right) \times \prod_{i=1}^{k} \Gamma^{\lambda - \frac{1}{q_i}}\left(\frac{\beta s + n(q_i - k)}{\beta q_i}\right),$$

where we assume that $nk - \beta s < n \min\{p_i, q_j, i, j = 1, 2, \dots k\}$.

Observe that the above parameters A_{ij} fulfill the set of conditions (3.65) in the conjugate case, that is, when $\lambda = 1$ and $p_i = q_i, i = 1, 2, ..., n$. Therefore, in the conjugate case, the above constant *L* is the best possible in the corresponding inequalities.

Remark 3.10 Let $A_i \in \mathbb{R}$, i = 1, 2, ..., k. Defining the parameters $A_{ii} = A_i, A_{ii+1} = -A_{i+1}$, $A_{ij} = 0$ for |i - j| > 1, i, j = 1, 2, ..., k, where indices are taken modulo k, we have $\sum_{i=1}^{n} A_{ij} = A_{j-1j} + A_{jj} = -A_j + A_j = 0$. Hence, Corollary 3.6 yields

$$\int_{\mathbb{R}^{nk}_{+}} \frac{\prod_{i=1}^{k} f_{i}(x_{i})}{\left(\sum_{i=1}^{k} |x_{i}|_{\alpha}^{\beta}\right)^{\lambda_{s}}} dx_{1} dx_{2} \cdots dx_{k} \\
\leq M \prod_{i=1}^{k} \left[\int_{\mathbb{R}^{n}_{+}} |x_{i}|_{\alpha} \frac{p_{i}(k-1)n - p_{i}\beta_{s}}{q_{i}} + p_{i}(A_{i} - A_{i+1})} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}$$
(3.73)

and

$$\begin{cases} \int_{\mathbb{R}^{n}_{+}} |x_{k}|_{\alpha}^{-\frac{p_{k}'}{q_{k}}[(k-1)n-\beta s]-p_{k}'(A_{k}-A_{1})} \\ \times \left[\int_{\mathbb{R}^{n}_{+}(k-1)} \frac{\prod_{i=1}^{k-1} f_{i}(x_{i})}{\left(\sum_{i=1}^{k} |x_{i}|_{\alpha}^{\beta}\right)^{\lambda s}} dx_{1} dx_{2} \cdots dx_{k-1} \right]^{p_{k}'} dx_{k} \end{cases}^{\frac{1}{p_{k}'}} \\ \leq M \prod_{i=1}^{k-1} \left[\int_{\mathbb{R}^{n}_{+}} |x_{i}|_{\alpha}^{\frac{p_{i}(k-1)n-p_{i}\beta s}{q_{i}}+p_{i}(A_{i}-A_{i+1})} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}, \qquad (3.74)$$

with the constant factor

$$M = \frac{\Gamma^{\lambda(k-1)n}\left(\frac{1}{\alpha}\right)\Gamma^{\lambda(k-2)}\left(\frac{n}{\beta}\right)}{\beta^{\lambda(k-1)}\alpha^{\lambda(n-1)(k-1)}\Gamma^{\lambda(k-1)}\left(\frac{n}{\alpha}\right)\Gamma^{\lambda}(s)}\prod_{i=1}^{k}\Gamma^{\frac{1}{q_{i}}}\left(\frac{\beta s + q_{i}A_{i+1} - (k-1)n}{\beta}\right)$$
$$\times \prod_{i=1}^{k}\Gamma^{\frac{1}{q_{i}}}\left(\frac{n - q_{i}A_{i+1}}{\beta}\right),$$

provided that $A_i \in \left(\frac{(k-1)n-\beta s}{q_{i-1}}, \frac{n}{q_{i-1}}\right), i = 1, 2, \dots, k.$

Remark 3.11 Hilbert-type inequalities in this section are established in [59] and [127]. For some related particular results, the reader is referred to [125] and [174]. In addition, results from this section for n = 1 and $\beta = 1$ were also studied in [12], [111], and [156].



Applying the Euler-Maclaurin summation formula

The starting point in this chapter is the well-known Euler-Maclaurin summation formula which asserts that

$$\int_{a}^{b} f(x)dx = \frac{(b-a)}{2} [f(a) + f(b)] -\sum_{k=1}^{n-1} \frac{(b-a)^{2k}}{(2k)!} B_{2k} \left[f^{(2k-1)}(b) - f^{(2k-1)}(a) \right] + \rho_{2n}(f), \quad (4.1)$$

where B_k is the corresponding Bernoulli number and the remainder $\rho_{2n}(f)$ is

$$\rho_{2n}(f) = \frac{(b-a)^{2n-1}}{(2n-1)!} \int_{a}^{b} f^{(2n-1)}(t) \gamma_{2n-1}^{*}\left(\frac{b-t}{b-a}\right) dt$$

or

$$\rho_{2n}(f) = \frac{(b-a)^{2n}}{(2n)!} \int_a^b f^{(2n)}(t) \gamma_{2n}^*\left(\frac{b-t}{b-a}\right) dt,$$

depending on whether the function $f:(a,b) \to \mathbb{R}$ has a continuous derivative of order 2n-1 or 2n, that is, f is of class C^{2n-1} or C^{2n} . Here, γ_n^* denotes the periodic function $\gamma_n^*(x) = B_n^*(x) - B_n$, where $B_n^*(x)$ and B_n are the corresponding Bernoulli polynomial and the Bernoulli number. For a comprehensive inspection on Bernoulli polynomials and numbers, as well as on the above summation formula, the reader is referred to [1] and [67].

The Euler-Maclaurin summation formula is a very useful tool in obtaining refinements of discrete Hilbert-type inequalities. Recently, Mingzhe and Xuemei [100], obtained the following refinement of the Hilbert-type inequality by means of the above summation formula.

Theorem 4.1 Suppose $p \ge q > 1$ are conjugate exponents. If $1 - \frac{q}{p} < s \le 2$, then the inequality

$$\sum_{m=0}^{\infty}\sum_{n=0}^{\infty}\frac{a_mb_n}{(m+n+1)^s} < \left[\sum_{m=0}^{\infty}\omega_q(s,m)a_m^p\right]^{\frac{1}{p}} \left[\sum_{n=0}^{\infty}\omega_p(s,n)b_n^q\right]^{\frac{1}{q}}$$
(4.2)

holds for all non-negative sequences $(a_m)_{m \in \mathbb{N}_0}$, $(b_n)_{n \in \mathbb{N}_0} \neq 0$, where

$$\omega_r(s,n) = \left(n + \frac{1}{2}\right)^{1-s} B\left(\frac{p-2+s}{p}, \frac{q-2+s}{q}\right) - \frac{(2-s)(r+2-s)}{4r(r+s-2)(2n+1)^{s-\frac{2-s}{r}}}$$

Since $B(\frac{p-2+s}{p}, \frac{q-2+s}{q}) = \pi/\sin(\pi/p)$ for s = 1, the above inequality (4.2) yields a refinement of the Hilbert double series theorem (1.4) (see Chapter 1).

On the other hand, Jichang and Debnath obtained in [41] the following refinement dealing with a symmetric homogeneous kernel of class C^4 .

Theorem 4.2 Suppose $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and $\frac{1}{2} \le \mu < \frac{1}{2}\min\{p,q\}$. Further, let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a non-negative homogeneous symmetric function of degree -s, s > 0, and of class C^4 . If $(-1)^n \frac{\partial^n K}{\partial y^n}(1,y) \ge 0$, n = 0, 1, 2, 3, 4, and $\frac{\partial^m K}{\partial y^m}(1,y)y^{-\frac{2\mu}{r}} \to 0$ when $y \to \infty$, m = 0, 1, then the inequality

$$\sum_{m=0}^{\infty} \sum_{n=0}^{\infty} K(m+\mu, n+\mu) a_m b_n$$

$$< \left\{ \sum_{m=0}^{\infty} \left[I(q,\mu) - \phi_q(m,s,\mu) \right] (m+\mu)^{1-s} a_m^p \right\}^{\frac{1}{p}}$$

$$\times \left\{ \sum_{n=0}^{\infty} \left[I(p,\mu) - \phi_p(n,s,\mu) \right] (n+\mu)^{1-s} b_n^q \right\}^{\frac{1}{q}}$$
(4.3)

holds for all non-negative sequences $(a_m)_{m \in \mathbb{N}_0}$, $(b_n)_{n \in \mathbb{N}_0} \neq 0$, where

$$\phi_r(m,s,\mu) = \left(\frac{\mu}{m+\mu}\right)^{1-\frac{2\mu}{r}} \left\{ K\left(1,\frac{\mu}{m+\mu}\right) \left[\frac{1}{1-\frac{2\mu}{r}} - \frac{1}{2\mu}\left(1+\frac{1}{3r}\right)\right] -\frac{1}{24\mu(m+\mu)}\frac{\partial K}{\partial y}\left(1,\frac{\mu}{m+\mu}\right) \right\} > 0$$

and $I(r,\mu) = \int_0^\infty K(1,u) u^{-\frac{2\mu}{r}} du < \infty$.

Motivated by Theorems 4.1 and 4.2, in this chapter we develop some general methods for improving discrete Hilbert-type inequalities via the Euler-Maclaurin summation formula. More precisely, the method used in the proof of Theorem 4.1 can be utilized in obtaining refinements of Hilbert-type inequalities with a general homogeneous kernel of class C^2 , fulfilling some additional properties. Similarly, Theorem 4.2 can also be extended to include homogeneous kernels of class C^4 which are generally not symmetric. Such extensions will be given in both equivalent forms, as in the previous chapters, and in the setting of non-conjugate parameters.

Finally, the last section of this chapter is dedicated to some particular refinements of Hilbert-type inequalities with the kernel $K(x,y) = (x+y)^{-s}$, s > 0. Such extensions will also be established by virtue of the Euler-Maclaurin summation formula.

4.1 Inequalities for kernels of class C²

4.1.1 Auxiliary results

In order to obtain refinements of discrete Hilbert-type inequalities with a homogeneous kernel of class C^2 , we first provide some auxiliary results, derived by virtue of the Euler-Maclaurin summation formula.

Lemma 4.1 Let $f : [M,N] \to \mathbb{R}$, $M,N \in \mathbb{N}$, be a continuously differentiable function. *Then the following equality holds*

$$\sum_{k=M}^{N} f(k) = \int_{M}^{N} f(x) dx + \frac{1}{2} \left[f(M) + f(N) \right] + \int_{M}^{N} \gamma_{1}^{*}(x) f'(x) dx,$$
(4.4)

where $\gamma_1^*(x) = x - \lfloor x \rfloor - \frac{1}{2}$.

Proof. Making use of the Euler-Maclaurin summation formula (4.1) with a = K and b = K + 1, where k = M, M + 1, ..., N - 1, we obtain

$$\int_{K}^{K+1} f(x)dx = \frac{f(K) + f(K+1)}{2} + \int_{K}^{K+1} \gamma_{1}^{*}(x)f'(x)dx,$$

so the result follows by summing the above equalities.

Remark 4.1 In particular, if the function *f* from the previous lemma is defined on $[0,\infty)$, and $f(x) \to 0^+$ when $x \to \infty$, then relation (4.4) yields (see also [100]):

$$\sum_{k=0}^{\infty} f(k) = \int_0^{\infty} f(x) dx + \frac{1}{2} f(0) + \int_0^{\infty} \gamma_1^*(x) f'(x) dx.$$

Utilizing the fact that $\gamma_1^*(x) = x - \lfloor x \rfloor - \frac{1}{2}$ is a periodic function with the period equal to 1, we also obtain the following estimate:

Lemma 4.2 Let $\varphi : [M,N] \to \mathbb{R}$, $M,N \in \mathbb{N}$, be strictly decreasing function. Then,

$$-\int_{M}^{N}\gamma_{1}^{*}(x)\varphi(x)dx < \frac{1}{8}\left[\varphi(M) - \varphi(N)\right],\tag{4.5}$$

where $\gamma_1^*(x) = x - \lfloor x \rfloor - \frac{1}{2}$.

Proof. Since $\int_{K}^{K+1} \gamma_{1}^{*}(x) dx = 0, K \in \mathbb{N}$, we have

$$\begin{split} -\int_{M}^{N} \gamma_{1}^{*}(x)\varphi(x)dx &= \sum_{k=M}^{N-1} \int_{k}^{k+1} -\gamma_{1}^{*}(x) \left(\varphi(x) - \varphi(k + \frac{1}{2})\right) dx \\ &= \sum_{k=M}^{N-1} \int_{k}^{k+\frac{1}{2}} -\gamma_{1}^{*}(x) \left(\varphi(k) - \varphi(k + \frac{1}{2})\right) dx \\ &\quad + \sum_{k=M}^{N-1} \int_{k+\frac{1}{2}}^{k+1} \gamma_{1}^{*}(x) \left(\varphi(k + \frac{1}{2}) - \varphi(k + 1)\right) dx + \sum_{k=M}^{N-1} \alpha_{k} \\ &= \frac{1}{8} \left(\varphi(M) - \varphi(N)\right) + \sum_{k=M}^{N-1} \alpha_{k}, \end{split}$$

where

$$\alpha_{k} = \int_{k}^{k+\frac{1}{2}} -\gamma_{1}^{*}(x) \left(\varphi(x) - \varphi(k)\right) dx + \int_{k+\frac{1}{2}}^{k+1} \gamma_{1}^{*}(x) \left(\varphi(k+1) - \varphi(x)\right) dx.$$

Finally, since φ is strictly decreasing, it follows that $\alpha_k < 0$ and (4.5) is proved.

Remark 4.2 Assuming that φ is defined on $[0,\infty)$ and $\varphi(x) \to 0^+$ when $x \to \infty$, inequality (4.5) reads:

$$-\int_0^\infty \gamma_1^*(x)\varphi(x)dx < \frac{1}{8}\varphi(0).$$

Now, by virtue of the above lemmas, we derive the following estimate referring to a homogeneous kernel of class C^2 .

Lemma 4.3 Let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a non-negative homogeneous function of degree -s, s > 0, and of class C^2 , such that

$$\frac{\partial K}{\partial y}(x,y) < 0, \ \frac{\partial^2 K}{\partial y^2}(x,y) > 0 \ \lim_{y \to \infty} K(x,y) = \lim_{y \to \infty} \frac{\partial K}{\partial y}(x,y) = 0, \ x \in \mathbb{R}_+.$$
(4.6)

Further, let f *and* $F_{q'}$ *be defined by*

$$f(y) = K(m+\mu, y+\mu) \frac{(m+\mu)^{q'A_1}}{(y+\mu)^{q'A_2}}$$
(4.7)

and

$$F_{q'}(m,s,\mu) = (m+\mu)^{1-s+q'(A_1-A_2)} \int_0^{\frac{\mu}{m+\mu}} K(1,y) y^{-q'A_2} dy,$$
(4.8)

where A_1 , A_2 , μ , and q' are real parameters such that $0 \le A_2 < \frac{1}{q'}$ and $\mu \ge \frac{1}{2}$. Then,

$$F_{q'}(m,s,\mu) - \frac{1}{2}f(0) - \int_{0}^{\infty} \gamma_{1}^{*}(y)f'(y)dy$$

> $\frac{(m+\mu)^{q'A_{1}}}{\mu^{q'A_{2}}} \left[K(m+\mu,m+\mu) \left(\frac{\mu}{1-q'A_{2}} - \frac{q'A_{2}}{8\mu} - \frac{1}{2} \right) - \frac{\partial K}{\partial y}(m+\mu,m+\mu) \left(\frac{\mu^{2}}{2} - \frac{1}{8} \right) \right],$ (4.9)

where $\gamma_1^*(y) = y - \lfloor y \rfloor - \frac{1}{2}$.

Proof. Utilizing the integration by parts, we obtain the following identity:

$$\int_{0}^{\frac{\mu}{m+\mu}} K(1,y) y^{-q'A_2} dy$$

= $K\left(1, \frac{\mu}{m+\mu}\right) \frac{\mu^{1-q'A_2}}{(m+\mu)^{1-q'A_2}(1-q'A_2)} - \frac{1}{1-q'A_2} \int_{0}^{\frac{\mu}{m+\mu}} \frac{\partial K}{\partial y}(1,y) y^{1-q'A_2} dy$

In addition, since $\frac{\partial^2 K}{\partial y^2}(x,y) > 0$ for all $x \in \mathbb{R}_+$, it follows that the function $\frac{\partial K}{\partial y}(1,y)$ is strictly increasing, so that

$$\begin{split} &\int_{0}^{\frac{\mu}{m+\mu}} K(1,y) y^{-q'A_2} dy \\ &> K \Big(1, \frac{\mu}{m+\mu} \Big) \frac{\mu^{1-q'A_2}}{(m+\mu)^{1-q'A_2}(1-q'A_2)} - \frac{\partial K}{\partial y} \Big(1, \frac{\mu}{m+\mu} \Big) \int_{0}^{\frac{\mu}{m+\mu}} y^{1-q'A_2} dy \\ &\ge K \Big(1, \frac{\mu}{m+\mu} \Big) \frac{\mu^{1-q'A_2}}{(m+\mu)^{1-q'A_2}(1-q'A_2)} - \frac{1}{2} \frac{\partial K}{\partial y} \Big(1, \frac{\mu}{m+\mu} \Big) \frac{\mu^{2-q'A_2}}{(m+\mu)^{2-q'A_2}}. \end{split}$$

Now, taking into account the homogeneity of *K*, the above estimate yields:

$$F_{q'}(m,s,\mu) > \frac{(m+\mu)^{q'A_1}}{\mu^{q'A_2-1}} \left[\frac{1}{1-q'A_2} K(m+\mu,\mu) - \frac{\mu}{2} \frac{\partial K}{\partial y}(m+\mu,\mu) \right].$$
(4.10)

On the other hand, it is obvious that the function -f' fulfills conditions of Lemma 4.2. Moreover, since $-f'(x) \to 0^+$, when $x \to \infty$, relation (4.5) provides the estimate

$$\int_{0}^{\infty} \gamma_{1}^{*}(y) f'(y) dy < \frac{(m+\mu)^{q'A_{1}}}{\mu^{q'A_{2}}} \left[\frac{q'A_{2}}{8\mu} K(m+\mu,\mu) - \frac{1}{8} \frac{\partial K}{\partial y}(m+\mu,\mu) \right].$$
(4.11)

Clearly, the above estimates (4.10) and (4.11) imply the inequality

$$\begin{split} F_{q'}(m,s,\mu) &- \frac{1}{2}f(0) - \int_{0}^{\infty} \gamma_{1}^{*}(y)f'(y)dy \\ &> \frac{(m+\mu)^{q'A_{1}}}{\mu^{q'A_{2}}} \bigg[K(m+\mu,\mu) \left(\frac{\mu}{1-q'A_{2}} - \frac{q'A_{2}}{8\mu} - \frac{1}{2} \right) \\ &- \frac{\partial K}{\partial y}(m+\mu,\mu) \left(\frac{\mu^{2}}{2} - \frac{1}{8} \right) \bigg]. \end{split}$$

Finally, since $\frac{\partial K}{\partial y}(x,y) < 0$ and $\frac{\partial^2 K}{\partial y^2}(x,y) > 0$, we have $K(m+\mu,\mu) > K(m+\mu,m+\mu)$ and $\frac{\partial K}{\partial y}(m+\mu,\mu) < \frac{\partial K}{\partial y}(m+\mu,m+\mu)$, so (4.9) holds.

Remark 4.3 It should be noticed here that the right-hand side of inequality (4.9) is non-negative since $\frac{\mu}{1-q'A_2} - \frac{q'A_2}{8\mu} - \frac{1}{2} \ge 0$ and $\mu \ge \frac{1}{2}$.

Since we deal with homogeneous kernels which are in general not symmetric, we shall also use the following estimate complementary to (4.9).

Lemma 4.4 Let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a non-negative homogeneous function of degree -s, s > 0, and of class C^2 , such that

$$\frac{\partial K}{\partial x}(x,y) < 0, \ \frac{\partial^2 K}{\partial x^2}(x,y) > 0, \ \lim_{x \to \infty} K(x,y) = \lim_{x \to \infty} \frac{\partial K}{\partial x}(x,y) = 0, \ y \in \mathbb{R}_+.$$
(4.12)

Further, let f *and* $F_{p'}$ *be defined by*

$$f(x) = K(x+\mu, n+\mu) \frac{(n+\mu)^{p'A_2}}{(x+\mu)^{p'A_1}}$$
(4.13)

and

$$F_{p'}(n,s,\mu) = (n+\mu)^{1-s+p'(A_2-A_1)} \int_0^{\frac{\mu}{n+\mu}} K(x,1) x^{-p'A_1} dx, \tag{4.14}$$

where A_1 , A_2 , μ , and p' are real parameters such that $0 \le A_1 < \frac{1}{p'}$ and $\mu \ge \frac{1}{2}$. Then,

$$F_{p'}(n,s,\mu) - \frac{1}{2}f(0) - \int_{0}^{\infty} \gamma_{1}^{*}(x)f'(x)dx$$

> $\frac{(n+\mu)^{p'A_{2}}}{\mu^{p'A_{1}}} \left[K(m+\mu,m+\mu) \left(\frac{\mu}{1-p'A_{1}} - \frac{p'A_{1}}{8\mu} - \frac{1}{2} \right) - \frac{\partial K}{\partial x}(m+\mu,m+\mu) \left(\frac{\mu^{2}}{2} - \frac{1}{8} \right) \right],$ (4.15)

where $\gamma_1^*(x) = x - \lfloor x \rfloor - \frac{1}{2}$.

Remark 4.4 Similarly to (4.9), the right-hand side of inequality (4.15) is also non-negative since $\frac{\mu}{1-p'A_1} - \frac{p'A_1}{8\mu} - \frac{1}{2} \ge 0$ and $\mu \ge \frac{1}{2}$.

4.1.2 Refined discrete Hilbert-type inequalities

Estimates from the previous subsection enable us to derive refinements of discrete Hilberttype inequalities with homogeneous kernels. Our aim here is to extend Theorem 4.1 to include homogeneous kernels of class C^2 . Similarly to Section 1.3 (Chapter 1) we use the notation $k(\alpha) = \int_0^\infty K(1, u)u^{-\alpha} du$, where $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ is a non-negative homogeneous function of degree -s, s > 0, provided that $k(\alpha) < \infty$ for min $\{1 - s, 0\} < \alpha <$ max $\{1, 2 - s\}$. The main result is presented in the setting of non-conjugate exponents.

Theorem 4.3 Let p, q, and λ be as in (2.1) and (2.2), and let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a nonnegative homogeneous function of degree -s, s > 0, and of class C^2 , fulfilling conditions (4.6) and (4.12). If $A_1 \in (0, \frac{1}{p'})$, $A_2 \in (0, \frac{1}{q'})$, and $\mu \ge \frac{1}{2}$, then the inequalities

$$\sum_{m=0}^{\infty} \sum_{n=0}^{\infty} K^{\lambda}(m+\mu, n+\mu) a_{m} b_{n}$$

$$< \left[\sum_{m=0}^{\infty} \Omega_{1}^{\frac{p}{q'}}(m, q') a_{m}^{p} \right]^{\frac{1}{p}} \left[\sum_{n=0}^{\infty} \Omega_{2}^{\frac{q}{p'}}(n, p') b_{n}^{q} \right]^{\frac{1}{q}}$$
(4.16)

and

$$\left[\sum_{n=0}^{\infty} \Omega_{2}^{-\frac{q'}{p'}}(n,p') \left(\sum_{m=0}^{\infty} K^{\lambda}(m+\mu,n+\mu)a_{m}\right)^{q'}\right]^{\frac{1}{q'}} < \left[\sum_{m=0}^{\infty} \Omega_{1}^{\frac{p}{q'}}(m,q')a_{m}^{p}\right]^{\frac{1}{p}}$$
(4.17)

hold for all non-negative sequences $(a_m)_{m \in \mathbb{N}_0}$ and $(b_n)_{n \in \mathbb{N}_0} \neq 0$, where $\Omega_i(h,r) = (h + \mu)^{1-s+r(A_i-A_{i+1})}k(\alpha_i) - \Gamma_i(h,r)$, $i = 1, 2, \alpha_1 = q'A_2$, $\alpha_2 = 2 - s - p'A_1$, and

$$\begin{split} \Gamma_i(h,r) \ &= \ \frac{(h+\mu)^{rA_i}}{\mu^{rA_{i+1}}} \cdot \left[K(h+\mu,h+\mu) \left(\frac{\mu}{1-rA_{i+1}} - \frac{rA_{i+1}}{8\mu} - \frac{1}{2} \right) \right. \\ &\left. - \frac{\partial K}{\partial x_{i+1}} (h+\mu,h+\mu) \left(\frac{\mu^2}{2} - \frac{1}{8} \right) \right], \quad x_1 = x, x_2 = y. \end{split}$$

Moreover, these inequalities are equivalent.

Proof. It is enough to show inequality (4.16) since the equivalent form (4.17) follows from Theorem 2.1 (see also Theorem 1.10, Chapter 1). If we rewrite (2.9) (Theorem 2.1) with the weight functions $\varphi(x) = (x + \mu)^{A_1}$, $\psi(y) = (y + \mu)^{A_2}$, and the counting measure, we obtain the inequality

$$\sum_{m=0}^{\infty}\sum_{n=0}^{\infty}K^{\lambda}(m+\mu,n+\mu)a_{m}b_{n}\leq\left[\sum_{m=0}^{\infty}\Phi^{\frac{p}{q'}}(q')a_{m}^{p}\right]^{\frac{1}{p}}\left[\sum_{n=0}^{\infty}\Phi^{\frac{q}{p'}}(p')b_{n}^{q}\right]^{\frac{1}{q}},$$

where the weight functions are defined by $\Phi(q') = \sum_{n=0}^{\infty} K(m+\mu, n+\mu) \frac{(m+\mu)q'A_1}{(n+\mu)q'A_2}$ and $\Phi(q') = \sum_{n=0}^{\infty} K(m+\mu, n+\mu) \frac{(m+\mu)q'A_2}{(n+\mu)q'A_2}$

$$\Phi(p') = \sum_{m=0}^{\infty} K(m+\mu, n+\mu) \frac{(n+\mu)^{p/A_2}}{(m+\mu)^{p'A_1}}.$$

On the other hand, the function f defined by (4.7) is continuously differentiable on $[0,\infty)$ and $f(y) \to 0^+$, when $y \to \infty$. Hence, in view of Lemma 4.1 we have

$$\Phi(q') = \int_0^\infty f(y) dy + \frac{1}{2}f(0) + \int_0^\infty \gamma_1^*(y) f'(y) dy.$$

Moreover, the homogeneity of the kernel K implies that

$$\int_0^\infty f(y)dy = (m+\mu)^{1-s+q'(A_1-A_2)}k(q'A_2) - F_{q'}(m,s,\mu),$$

where $F_{q'}(m, s, \mu)$ is defined by (4.8). Clearly, the above two relations yield

$$\Phi(q') = (m+\mu)^{1-s+q'(A_1-A_2)}k(q'A_2) - \left(F_{q'}(m,s,\mu) - \frac{1}{2}f(0) - \int_0^\infty \gamma_1^*(y)f'(y)dy\right),$$

that is, $\Phi(q') < \Omega_1(m,q')$, by virtue of inequality (4.9). Analogously, by virtue of (4.15) we obtain $\Phi(p') < \Omega_2(n,p')$, so (4.16) is proved.

The kernel $K(x,y) = (x+y)^{-s}$, s > 0, fulfills conditions of Theorem 4.3. Now, considering the above theorem with this kernel and the parameters $A_1 = A_2 = \frac{2-s}{p'q'}$, $\mu = \frac{1}{2}$, we obtain an extension of Theorem 4.1 to non-conjugate case.

Corollary 4.1 Let p, q, and λ satisfy (2.1) and (2.2), and let

$$\max\left\{1 - \min\left\{\frac{1}{p'-1}, \frac{1}{q'-1}\right\}, 2 - \min\{p', q'\}\right\} < s \le 2.$$

Then the inequalities

$$\sum_{n=0}^{\infty} \sum_{n=0}^{\infty} \frac{a_m b_n}{(m+n+1)^{\lambda_s}} < \left[\sum_{m=0}^{\infty} \Omega_{p'}^{\frac{p}{q'}}(m) a_m^p \right]^{\frac{1}{p}} \left[\sum_{n=0}^{\infty} \Omega_{q'}^{\frac{q}{p'}}(n) b_n^q \right]^{\frac{1}{q}}$$
(4.18)

and

$$\left[\sum_{n=0}^{\infty} \Omega_{q'}^{-\frac{q'}{p'}}(n) \left(\sum_{m=0}^{\infty} \frac{a_m}{(m+n+1)^{\lambda_s}}\right)^{q'}\right]^{\frac{1}{q'}} < \left[\sum_{m=0}^{\infty} \Omega_{p'}^{\frac{p}{q'}}(m) a_m^p\right]^{\frac{1}{p}}$$
(4.19)

hold for all non-negative sequences $(a_m)_{m \in \mathbb{N}_0}$, $(b_n)_{n \in \mathbb{N}_0} \neq 0$, where

$$\Omega_r(h) = \left(h + \frac{1}{2}\right)^{1-s} B\left(\frac{r+s-2}{r}, \frac{rs-r-s+2}{r}\right) - \frac{(2-s)(r+2-s)}{4r(r+s-2)(2h+1)^{s-\frac{2-s}{r}}}.$$

Moreover, these inequalities are equivalent.

In the sequel we derive a version of Theorem 4.3 dealing with finite sums. The corresponding results will include the weight functions

$$\Phi_{i}(h,r) = (h+\mu)^{1-s+r(A_{i}-A_{i+1})}k(\alpha_{i}) - \frac{(h+\mu)^{rA_{i}}}{(M+\mu)^{rA_{i+1}}}\Lambda_{i}(h,r) - \frac{(h+\mu)^{rA_{i}}}{(N+\mu)^{rA_{i+1}}}\Delta_{i}(h,r), \quad i = 1, 2,$$
(4.20)
where

$$\begin{split} \Lambda_{i}(h,r) &= K_{i}(h,M) \left[\frac{M+\mu}{1-rA_{i+1}} - \frac{rA_{i+1}}{8(M+\mu)} - \frac{1}{2} \right] - \frac{\partial K_{i}}{\partial x_{i+1}}(h,M) \left[\frac{(M+\mu)^{2}}{2} - \frac{1}{8} \right] \\ \Delta_{i}(h,r) &= K_{i}(h,N) \left[\frac{N+\mu}{s+rA_{i+1}-1} + \frac{rA_{i+1}}{8(N+\mu)} - \frac{1}{2} \right] - \frac{(h+\mu)(N+\mu)}{s(s+1)} \frac{\partial K_{i}}{\partial x_{i}}(h,N) \\ &- \frac{1}{8} \frac{\partial K_{i}}{\partial x_{i+1}}(h,N), \end{split}$$

 $\alpha_1 = q'A_2, \ \alpha_2 = 2 - s - p'A_1, \ x_1 = x, \ x_2 = y, \ K_1(h, H) = K(h + \mu, H + \mu), \ \text{and} \ K_2(h, H) = K(H + \mu, h + \mu).$

Theorem 4.4 Let p, q, and λ satisfy (2.1) and (2.2), and let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a nonnegative homogeneous function of degree -s, s > 0, and of class C^2 , fulfilling conditions

$$\frac{\partial K}{\partial x}(x,y) < 0, \ \frac{\partial^2 K}{\partial x^2}(x,y) > 0, \ \frac{\partial K}{\partial y}(x,y) < 0, \ \frac{\partial^2 K}{\partial y^2}(x,y) > 0.$$

If $M, N \in \mathbb{N}$, and $A_1 \in \left(\max\{\frac{1-s}{p'}, 0\}, \frac{1}{p'}\right)$, $A_2 \in \left(\max\{\frac{1-s}{q'}, 0\}, \frac{1}{q'}\right)$, $\mu \ge 0$, then the inequalities

$$\sum_{m=M}^{N} \sum_{n=M}^{N} K^{\lambda}(m+\mu, n+\mu) a_{m} b_{n} \\ < \left[\sum_{m=M}^{N} \Phi_{1}^{\frac{p}{q'}}(m, q') a_{m}^{p} \right]^{\frac{1}{p}} \left[\sum_{n=M}^{N} \Phi_{2}^{\frac{q}{p'}}(n, p') b_{n}^{q} \right]^{\frac{1}{q}}$$
(4.21)

and

$$\left[\sum_{n=M}^{N} \Phi_{2}^{-\frac{q'}{p'}}(n,p') \left(\sum_{m=M}^{N} K^{\lambda}(m+\mu,n+\mu)a_{m}\right)^{q'}\right]^{\frac{1}{q'}} < \left[\sum_{m=M}^{N} \Phi_{1}^{\frac{p}{q'}}(m,q')a_{m}^{p}\right]^{\frac{1}{p}}$$
(4.22)

hold for all non-negative sequences $(a_m)_{m \in \mathbb{N}}$, $(b_n)_{n \in \mathbb{N}} \neq 0$, where the weight functions are defined by (4.20). Moreover, these inequalities are equivalent.

Proof. The proof follows the same lines as the proof of Theorem 4.3, except that we consider the bounded interval [M, N] instead of $[0, \infty)$. Applying Theorem 2.1 (Chapter 2), we obtain inequality

$$\sum_{m=M}^{N} \sum_{n=M}^{N} K^{\lambda}(m+\mu, n+\mu) a_{m} b_{n} \leq \left(\sum_{m=M}^{N} \Phi^{\frac{p}{q'}}(q') a_{m}^{p}\right)^{\frac{1}{p}} \left(\sum_{n=M}^{N} \Phi^{\frac{q}{p'}}(p') b_{n}^{q}\right)^{\frac{1}{q}},$$

with the weight functions $\Phi(q') = \sum_{n=M}^{N} K(m+\mu, n+\mu) \frac{(m+\mu)q'A_1}{(n+\mu)q'A_2}$ and $\Phi(p') = \sum_{m=M}^{N} K(m+\mu, n+\mu) \frac{(m+\mu)q'A_1}{(n+\mu)q'A_2}$

$$+\mu, n+\mu) \frac{(n+\mu)^{p'A_2}}{(m+\mu)^{p'A_1}}.$$

On the other hand, applying Lemma 4.1 to the function f defined by (4.7), we have

$$\Phi(q') = \int_{M}^{N} f(y) dy + \frac{1}{2} \left(f(M) + f(N) \right) + \int_{M}^{N} \gamma_{1}^{*}(y) f'(y) dy,$$
(4.23)

while the homogeneity of the kernel K implies

$$\begin{split} \int_{M}^{N} f(y) dy &= (m+\mu)^{1-s+q'(A_{1}-A_{2})} k(q'A_{2}) - (m+\mu)^{1-s+q'(A_{1}-A_{2})} \\ &\times \left[\int_{0}^{\frac{M+\mu}{m+\mu}} K(1,t) t^{-q'A_{2}} dt + \int_{0}^{\frac{m+\mu}{N+\mu}} K(t,1) t^{s+q'A_{2}-1} dt \right]. \end{split}$$

Finally, using (4.23), Lemma 4.2 and the same estimates as in Lemma 4.3 we have $\Phi(q') < \Phi_1(m,q')$, and similarly, $\Phi(p') < \Phi_2(n,p')$, which completes the proof. \Box

Remark 4.5 It should be noticed here that Theorem 4.4 also covers Theorem 4.3. Namely, let M = 0 and $N \rightarrow \infty$. Then, taking into account the conditions

$$\lim_{x \to 0} K(x, y) = \lim_{y \to 0} K(x, y) = \lim_{x \to 0} \frac{\partial K}{\partial x}(x, y) = \lim_{y \to 0} \frac{\partial K}{\partial y}(x, y) = 0$$

and $\mu \geq \frac{1}{2}$, Theorem 4.4 reduces to Theorem 4.3.

We conclude this section with the finite sum version of Corollary 4.1.

Corollary 4.2 Let p, q, and λ satisfy (2.1) and (2.2), and let

$$\max\left\{1 - \min\left\{\frac{1}{p' - 1}, \frac{1}{q' - 1}\right\}, 2 - \min\{p', q'\}\right\} < s \le 2.$$

Then,

$$\sum_{m=M}^{N} \sum_{n=M}^{N} \frac{a_m b_n}{(m+n+1)^{\lambda_s}} < \left[\sum_{m=M}^{N} \Phi_{p'}^{\frac{p}{q'}}(m) a_m^p \right]^{\frac{1}{p}} \left[\sum_{n=M}^{N} \Phi_{q'}^{\frac{q}{p'}}(n) b_n^q \right]^{\frac{1}{q}}$$
(4.24)

and

$$\left[\sum_{n=M}^{N} \Phi_{q'}^{-\frac{q'}{p'}}(n) \left(\sum_{m=M}^{N} \frac{a_m}{(m+n+1)^{\lambda_s}}\right)^{q'}\right]^{\frac{1}{q'}} < \left[\sum_{m=M}^{N} \Phi_{p'}^{\frac{p}{q'}}(m) a_m^p\right]^{\frac{1}{p}},$$
(4.25)

1

where the weight functions are defined by

$$\Phi_r(h) = \left(h + \frac{1}{2}\right)^{1-s} B\left(\frac{r+s-2}{r}, \frac{rs-r-s+2}{r}\right) \\ - \left(\frac{2h+1}{2M+1}\right)^{\frac{2-s}{r}} \phi_{M,r}(h) - \left(\frac{2h+1}{2N+1}\right)^{\frac{2-s}{r}} \phi_{N,r}(h), \quad r = p', q',$$

$$\begin{split} \phi_{M,r}(h) &= \frac{1}{(M+h+1)^s} \left[\frac{r(2M+1)}{2(r+s-2)} - \frac{2-s}{4r(2M+1)} + \frac{sM(M+1)}{2(M+h+1)} - \frac{1}{2} \right], \\ \phi_{N,r}(h) &= \frac{1}{(N+h+1)^s} \left[\frac{r(2N+1)}{2(rs-r-s+2)} + \frac{2-s}{4r(2N+1)} + \frac{2(2N+1)(2h+1) + s(s+1)}{8(s+1)(N+h+1)} - \frac{1}{2} \right]. \end{split}$$

Moreover, these inequalities are equivalent.

4.2 Inequalities for kernels of class C^4

4.2.1 Auxiliary results

Motivated by Theorem 4.2, in this section we deal with Hilbert-type inequalities with a homogeneous kernel of class C^4 . Similarly to the previous section, we first prove a few auxiliary results, derived by means of the Euler-Maclaurin summation formula.

Lemma 4.5 Suppose $f : [M,N] \to \mathbb{R}$, $M,N \in \mathbb{N}$, is of class C^4 and let $f^{(4)}(x) \ge 0$, $x \in [M,N]$. Then the following inequality holds:

$$\sum_{k=M}^{N} f(k) < \int_{M}^{N} f(x) dx + \frac{1}{2} \left[f(M) + f(N) \right] + \frac{1}{12} \left[f'(N) - f'(M) \right].$$
(4.26)

Proof. Applying the Euler-Maclaurin summation formula to the function $f : [M, N] \to \mathbb{R}$, we have

$$\sum_{k=M}^{N} f(k) = \int_{M}^{N} f(x) dx + \frac{1}{2} [f(M) + f(N)] + \frac{1}{12} [f'(N) - f'(M)] + \frac{1}{24} \int_{M}^{N} f^{(4)}(t) (B_4 - B_4(t - \lfloor t \rfloor)) dt,$$

where B_4 and $B_4(t)$ respectively denote the corresponding Bernoulli number and the Bernoulli polynomial. Since $B_4 = -\frac{1}{30}$ and the sign of $B_4 - B_4(t - \lfloor t \rfloor)$ is the same as the sign of B_4 , we obtain (4.26).

Remark 4.6 In particular, if the function *f* from the previous lemma is defined on $[0, \infty)$, and $f(x), f'(x) \to 0$ when $x \to \infty$, then relation (4.26) yields (see also [41]):

$$\sum_{k=0}^{\infty} f(k) < \int_0^{\infty} f(t)dt + \frac{1}{2}f(0) - \frac{1}{12}f'(0).$$

The following two estimates refer to a homogeneous kernel $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ of degree -s, s > 0, and of class C^4 . Like in the previous section, the integral $k(\alpha) = \int_0^\infty K(1, u) u^{-\alpha} du$ is assumed to converge for $\min\{1 - s, 0\} < \alpha < \max\{1, 2 - s\}$.

Lemma 4.6 Let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a non-negative homogeneous function of degree -s, s > 0, and of class C^4 , fulfilling

$$(-1)^{n} \frac{\partial^{n} K}{\partial y^{n}}(x, y) > 0, \ n = 1, 2, 3, 4, \ and \ \lim_{y \to \infty} K(x, y) = \lim_{y \to \infty} \frac{\partial K}{\partial y}(x, y) = 0,$$
(4.27)

for all $x, y \in \mathbb{R}_+$. Further, let $\Theta_{q'}$ be defined by

$$\Theta_{q'}(m,s,\mu) = (m+\mu)^{q'A_1} \sum_{n=0}^{\infty} K(m+\mu,n+\mu)(n+\mu)^{-q'A_2},$$
(4.28)

where A_1 , A_2 , μ , and q' are real parameters such that $0 \le A_2 < \frac{1}{q'}$ and $\mu \ge \frac{1}{2}$. Then,

$$\Theta_{q'}(m,s,\mu) < (m+\mu)^{1-s+q'(A_1-A_2)} \left[k(q'A_2) - \theta_{q'}(m,s,\mu) \right],$$
(4.29)

where

$$\begin{aligned} \theta_{q'}(m,s,\mu) &= \left(\frac{\mu}{m+\mu}\right)^{1-q'A_2} \left\{ K\left(1,\frac{\mu}{m+\mu}\right) \left[\frac{1}{1-q'A_2} - \frac{1}{2\mu}\left(1+\frac{q'A_2}{6\mu}\right)\right] \\ &- \frac{1}{24\mu(m+\mu)} \frac{\partial K}{\partial y}\left(1,\frac{\mu}{m+\mu}\right) \right\}. \end{aligned}$$

Proof. Setting $f(y) = K(m + \mu, y + \mu)(m + \mu)^{q'A_1}(y + \mu)^{-q'A_2}$, we have $\Theta_{q'}(m, s, \mu) = \sum_{n=0}^{\infty} f(n)$, so Lemma 4.5 yields the inequality

$$\Theta_{q'}(m,s,\mu) < (m+\mu)^{1-s+q'(A_1-A_2)} \left[k(q'A_2) - \omega_{q'}(m,s,\mu) \right],$$

where

$$\begin{split} \omega_{q'}(m,s,\mu) \, &= \, \int_{0}^{\frac{\mu}{m+\mu}} K(1,t) t^{-q'A_2} dt - \left(\frac{1}{2\mu} + \frac{q'A_2}{12\mu^2}\right) \left(\frac{\mu}{m+\mu}\right)^{1-q'A_2} K\left(1,\frac{\mu}{m+\mu}\right) \\ &+ \frac{1}{12\mu^2} \left(\frac{\mu}{m+\mu}\right)^{2-q'A_2} \frac{\partial K}{\partial y} \left(1,\frac{\mu}{m+\mu}\right). \end{split}$$

Moreover, applying the integration by parts twice, we have

$$\begin{split} \int_{0}^{\frac{\mu}{m+\mu}} K(1,t)t^{-q'A_2}dt &= \frac{1}{1-q'A_2} \left(\frac{\mu}{m+\mu}\right)^{1-q'A_2} K\left(1,\frac{\mu}{m+\mu}\right) \\ &- \frac{1}{(1-q'A_2)(2-q'A_2)} \left(\frac{\mu}{m+\mu}\right)^{2-q'A_2} \frac{\partial K}{\partial y} \left(1,\frac{\mu}{m+\mu}\right) \\ &+ \frac{1}{(1-q'A_2)(2-q'A_2)} \int_{0}^{\frac{\mu}{m+\mu}} \frac{\partial^2 K}{\partial y^2}(1,t)t^{2-q'A_2} dt. \end{split}$$

Finally, since the last term in the above relation is non-negative and since the inequality

$$\frac{1}{(1-q'A_2)(2-q'A_2)} - \frac{1}{12\mu^2} > \frac{1}{24\mu^2}$$

holds for $0 < q'A_2 < 1$ and $\mu \ge \frac{1}{2}$, it follows that $\omega_{q'}(m, s, \mu) > \theta_{q'}(m, s, \mu)$, which completes the proof.

Since we are concerned with homogeneous kernels which are in general not symmetric, we shall also utilize the result which is, in some way, complementary to Lemma 4.6.

Lemma 4.7 Let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a non-negative homogeneous function of degree -s, s > 0, and of class C^4 , fulfilling

$$(-1)^{n} \frac{\partial^{n} K}{\partial x^{n}}(x,y) > 0, \ n = 1,2,3,4, \ and \ \lim_{x \to \infty} K(x,y) = \lim_{x \to \infty} \frac{\partial K}{\partial x}(x,y) = 0,$$
 (4.30)

for all $x, y \in \mathbb{R}_+$. Further, let $\Theta_{p'}$ be defined by

$$\Theta_{p'}(n,s,\mu) = (n+\mu)^{p'A_2} \sum_{m=0}^{\infty} K(m+\mu,n+\mu)(m+\mu)^{-p'A_1},$$
(4.31)

where A_1 , A_2 , μ , and p' are real parameters such that $0 \le A_1 < \frac{1}{p'}$ and $\mu \ge \frac{1}{2}$. Then,

$$\Theta_{p'}(n,s,\mu) < (n+\mu)^{1-s+p'(A_2-A_1)} \left[k(2-s-p'A_1) - \theta_{p'}(n,s,\mu) \right],$$
(4.32)

where

$$\begin{split} \theta_{p'}(n,s,\mu) &= \left(\frac{\mu}{n+\mu}\right)^{1-p'A_1} \left\{ K\left(\frac{\mu}{n+\mu},1\right) \left[\frac{1}{1-p'A_1} - \frac{1}{2\mu}\left(1 + \frac{p'A_1}{6\mu}\right)\right] \\ &- \frac{1}{24\mu(n+\mu)} \frac{\partial K}{\partial x}\left(\frac{\mu}{n+\mu},1\right) \right\}. \end{split}$$

Remark 4.7 It should be noticed here that the terms $\theta_{q'}(m, s, \mu)$ and $\theta_{p'}(n, s, \mu)$ in (4.29) and (4.32) are non-negative.

4.2.2 Refined discrete Hilbert-type inequalities

In the sequel we extend Theorem 4.2 to hold for a homogeneous kernel of class C^4 , in the non-conjugate case. Such extension is a simple consequence of the general Hilbert-type inequality in non-conjugate setting (Theorem 2.1, Chapter 2) and Lemmas 4.6 and 4.7.

Theorem 4.5 Let p, q, and λ satisfy (2.1) and (2.2), and let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a nonnegative homogeneous function of degree -s, s > 0, and of class C^4 , fulfilling conditions (4.27) and (4.30). If $A_1 \in (0, \frac{1}{p'})$, $A_2 \in (0, \frac{1}{q'})$, and $\mu \geq \frac{1}{2}$, then the inequalities

$$\sum_{m=0}^{\infty} \sum_{n=0}^{\infty} K^{\lambda}(m+\mu,n+\mu)a_{m}b_{n}$$

$$< \left\{ \sum_{m=0}^{\infty} (m+\mu)^{\frac{p}{q'}(1-s)+p(A_{1}-A_{2})} \left[k(q'A_{2}) - \theta_{q'}(m,s,\mu) \right]^{\frac{p}{q'}} a_{m}^{p} \right\}^{\frac{1}{p}}$$

$$\times \left\{ \sum_{n=0}^{\infty} (n+\mu)^{\frac{q}{p'}(1-s)+q(A_{2}-A_{1})} \left[k(2-s-p'A_{1}) - \theta_{p'}(n,s,\mu) \right]^{\frac{q}{p'}} b_{n}^{q} \right\}^{\frac{1}{q}}$$
(4.33)

and

$$\begin{cases}
\sum_{n=0}^{\infty} (n+\mu)^{q'(A_1-A_2)+\frac{q'}{p'}(s-1)} \left[k(2-s-p'A_1)-\theta_{p'}(n,s,\mu)\right]^{-\frac{q'}{p'}} \\
\times \left[\sum_{m=0}^{\infty} K^{\lambda}(m+\mu,n+\mu)a_m\right]^{q'} \right\}^{\frac{1}{q'}} \\
< \left\{\sum_{m=0}^{\infty} (m+\mu)^{\frac{p}{q'}(1-s)+p(A_1-A_2)} \left[k(q'A_2)-\theta_{q'}(m,s,\mu)\right]^{\frac{p}{q'}}a_m^p \right\}^{\frac{1}{p}}$$
(4.34)

hold for all non-negative sequences $(a_m)_{m \in \mathbb{N}_0}$, $(b_n)_{n \in \mathbb{N}_0} \neq 0$, where the functions $\theta_{q'}(m,s,\mu)$ and $\theta_{p'}(n,s,\mu)$ are defined in Lemmas 4.6 and 4.7. Moreover, these inequalities are equivalent.

Proof. Applying Theorem 2.1 (Chapter 2) to discrete setting with a suitable power weight functions, we have

$$\sum_{m=0}^{\infty}\sum_{n=0}^{\infty}K^{\lambda}(m+\mu,n+\mu)a_{m}b_{n} < \left[\sum_{m=0}^{\infty}\Theta_{q'}^{\frac{p}{q'}}(m,s,\mu)a_{m}^{p}\right]^{\frac{1}{p}}\left[\sum_{n=0}^{\infty}\Theta_{p'}^{\frac{q}{p'}}(n,s,\mu)b_{n}^{q}\right]^{\frac{1}{q}},$$

where $\Theta_{q'}(m,s,\mu)$ and $\Theta_{p'}(n,s,\mu)$ are defined by (4.28) and (4.31) respectively. Now, inequality (4.33) follows from estimates (4.29) and (4.32).

The previous theorem with the parameters $A_1 = A_2 = \frac{2\mu}{p'q'}$, implies the following:

Corollary 4.3 Let p, q, and λ satisfy (2.1) and (2.2), and let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a nonnegative homogeneous function of degree -s, s > 0, and of class C^4 , fulfilling conditions (4.27) and (4.30). If $\frac{1}{2} \le \mu < \frac{1}{2} \min\{p', q'\}$, then the inequalities

$$\sum_{m=0}^{\infty} \sum_{n=0}^{\infty} K^{\lambda}(m+\mu,n+\mu) a_{m} b_{n}$$

$$< \left\{ \sum_{m=0}^{\infty} (m+\mu)^{\frac{p}{q'}(1-s)} \left[k\left(\frac{2\mu}{p'}\right) - \phi_{p'}(m,s,\mu) \right]^{\frac{p}{q'}} a_{m}^{p} \right\}^{\frac{1}{p}}$$

$$\times \left\{ \sum_{n=0}^{\infty} (n+\mu)^{\frac{q}{p'}(1-s)} \left[k\left(2-s-\frac{2\mu}{q'}\right) - \phi_{q'}(m,s,\mu) \right]^{\frac{q}{p'}} b_{n}^{q} \right\}^{\frac{1}{q}}$$
(4.35)

and

$$\begin{cases}
\sum_{n=0}^{\infty} (n+\mu)^{\frac{q'}{p'}(s-1)} \left[k \left(2-s-\frac{2\mu}{q'} \right) - \phi_{q'}(n,s,\mu) \right]^{-\frac{q'}{p'}} \\
\times \left[\sum_{m=0}^{\infty} K^{\lambda}(m+\mu,n+\mu)a_m \right]^{q'} \right\}^{\frac{1}{q'}} \\
< \left\{ \sum_{m=0}^{\infty} (m+\mu)^{\frac{p}{q'}(1-s)} \left[k \left(\frac{2\mu}{p'} \right) - \theta_{p'}(m,s,\mu) \right]^{\frac{p}{q'}} a_m^p \right\}^{\frac{1}{p}}$$
(4.36)

hold for all non-negative sequences $(a_m)_{m \in \mathbb{N}_0}$, $(b_n)_{n \in \mathbb{N}_0} \neq 0$, where the functions $\theta_{q'}(m,s,\mu)$ and $\theta_{p'}(n,s,\mu)$ are defined in Lemmas 4.6 and 4.7. Moreover, these inequalities are equivalent.

Remark 4.8 Assuming that the kernel $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ from Corollary 4.3 is symmetric, we have that $k(2 - s - \alpha) = k(\alpha)$. Then, in the conjugate case, that is, when $\lambda = 1$, inequality (4.35) reduces to (4.3) from Theorem 4.2.

The rest of this section is dedicated to some particular homogeneous kernels fulfilling conditions (4.27) and (4.30). We start with the well-known example $K(x,y) = (x+y)^{-s}$, s > 0. In this case, the corresponding Hilbert-type inequalities involve the constant factors expressed in terms of the Beta function. Hence, denoting $b(\alpha) = B(1 - \alpha, s + \alpha - 1)$, we have the following result:

Corollary 4.4 Suppose p, q, and λ satisfy conditions (2.1) and (2.2), and let $A_1 \in \left(\max\{\frac{1-s}{p'}, 0\}, \frac{1}{p'}\right), A_2 \in \left(\max\{\frac{1-s}{q'}, 0\}, \frac{1}{q'}\right), and \mu \geq \frac{1}{2}$. Then the inequalities

$$\sum_{m=0}^{\infty} \sum_{n=0}^{\infty} \frac{a_m b_n}{(m+n+2\mu)^{\lambda s}} \\ < \left\{ \sum_{m=0}^{\infty} (m+\mu)^{\frac{p}{q'}(1-s)+p(A_1-A_2)} \left[b(q'A_2) - \psi_{q'}(m,s,\mu) \right]^{\frac{p}{q'}} a_m^p \right\}^{\frac{1}{p}} \\ \times \left\{ \sum_{n=0}^{\infty} (n+\mu)^{\frac{q}{p'}(1-s)+q(A_2-A_1)} \left[b(p'A_1) - \psi_{p'}(n,s,\mu) \right]^{\frac{q}{p'}} b_n^q \right\}^{\frac{1}{q}}$$
(4.37)

and

$$\left\{\sum_{n=0}^{\infty} (n+\mu)^{q'(A_1-A_2)+\frac{q'}{p'}(s-1)} \left[b(p'A_1)-\psi_{p'}(n,s,\mu)\right]^{-\frac{q'}{p'}} \times \left[\sum_{m=0}^{\infty} \frac{a_m}{(m+n+2\mu)^{\lambda s}}\right]^{q'}\right\}^{\frac{1}{q'}} \\
< \left\{\sum_{m=0}^{\infty} (m+\mu)^{\frac{p}{q'}(1-s)+p(A_1-A_2)} \left[b(q'A_2)-\psi_{q'}(m,s,\mu)\right]^{\frac{p}{q'}} a_m^p\right\}^{\frac{1}{p}} \quad (4.38)$$

hold for all non-negative sequences $(a_m)_{m \in \mathbb{N}_0}$, $(b_n)_{n \in \mathbb{N}_0} \neq 0$, where

$$\psi_{q'}(m,s,\mu) = \left(\frac{\mu}{m+\mu}\right)^{1-q'A_2} \frac{1}{2^s} \left[\frac{1}{1-q'A_2} - \frac{1}{2\mu} \left(1 + \frac{q'A_2}{6\mu}\right)\right]$$

and

$$\psi_{p'}(n,s,\mu) = \left(\frac{\mu}{n+\mu}\right)^{1-p'A_1} \frac{1}{2^s} \left[\frac{1}{1-p'A_1} - \frac{1}{2\mu}\left(1+\frac{p'A_1}{6\mu}\right)\right].$$

In addition, these inequalities are equivalent.

Proof. It follows from Theorem 4.5, setting $K(x,y) = (x+y)^{-s}$ and observing that $\psi_{q'}(m,s,\mu) < \theta_{q'}(m,s,\mu)$ and $\psi_{p'}(n,s,\mu) < \theta_{p'}(n,s,\mu)$.

Remark 4.9 It should be noticed here that Corollary 4.4 provides refinements of inequalities (1.58) and (1.59) (see Remark 1.14, Chapter 1). This can be seen by considering inequalities (4.37) and (4.38) in the conjugate case, with the parameters $\mu = \frac{1}{2}$ and $A_1 = A_2 = \frac{2-s}{pq}$, provided that max $\{1 - \min\{\frac{1}{p-1}, \frac{1}{q-1}\}, 2 - \min\{p,q\}\} < s \le 2$. In such a way we also obtain refinement of the Hilbert double series theorem (1.4) (see Chapter 1).

Our next application of Theorem 4.5 refers to the homogeneous kernel $K(x,y) = (x^s + y^s)^{-1}$, s > 0. Namely, this kernel fulfills conditions (4.27) and (4.30), hence, denoting $b_s(\alpha) = \frac{1}{s}B(\frac{1-\alpha}{s}, \frac{s+\alpha-1}{s})$, we obtain the following consequence:

Corollary 4.5 Let p, q, and λ satisfy (2.1) and (2.2), and let $A_1 \in \left(\max\{\frac{1-s}{p'}, 0\}, \frac{1}{p'}\right)$, $A_2 \in \left(\max\{\frac{1-s}{q'}, 0\}, \frac{1}{q'}\right)$, and $\mu \geq \frac{1}{2}$. Then the inequalities

$$\sum_{m=0}^{\infty} \sum_{n=0}^{\infty} \frac{a_m b_n}{[(m+\mu)^s + (n+\mu)^s]^{\lambda}} < \left\{ \sum_{m=0}^{\infty} (m+\mu)^{\frac{p}{q'}(1-s) + p(A_1 - A_2)} \left[b_s(q'A_2) - 2^{s-1} \psi_{q'}(m,s,\mu) \right]^{\frac{p}{q'}} a_m^p \right\}^{\frac{1}{p}} \times \left\{ \sum_{n=0}^{\infty} (n+\mu)^{\frac{q}{p'}(1-s) + q(A_2 - A_1)} \left[b_s(p'A_1) - 2^{s-1} \psi_{p'}(n,s,\mu) \right]^{\frac{q}{p'}} b_n^q \right\}^{\frac{1}{q}}$$
(4.39)

and

$$\left\{ \sum_{n=0}^{\infty} (n+\mu)^{q'(A_1-A_2)+\frac{q'}{p'}(s-1)} \left[b_s(p'A_1) - 2^{s-1} \psi_{p'}(n,s,\mu) \right]^{-\frac{q'}{p'}} \times \left[\sum_{m=0}^{\infty} \frac{a_m}{[(m+\mu)^s + (n+\mu)^s]^{\lambda}} \right]^{q'} \right\}^{\frac{1}{q'}} \\
< \left\{ \sum_{m=0}^{\infty} (m+\mu)^{\frac{p}{q'}(1-s)+p(A_1-A_2)} \left[b_s(q'A_2) - 2^{s-1} \psi_{q'}(m,s,\mu) \right]^{\frac{p}{q'}} a_m^p \right\}^{\frac{1}{p}} \quad (4.40)$$

hold for all non-negative sequences $(a_m)_{m \in \mathbb{N}_0}$, $(b_n)_{n \in \mathbb{N}_0} \neq 0$, where the functions $\psi_{q'}(m,s,\mu)$ and $\psi_{p'}(n,s,\mu)$ are defined in the previous corollary. In addition, these inequalities are equivalent.

Remark 4.10 Observe that special cases of Corollaries 4.4 and 4.5, involving the conjugate setting and the parameters $A_1 = A_2 = \frac{2\mu}{pq}$, were studied in [41].

To conclude this section, we consider Corollary 4.5 with $A_1 = \frac{1}{p'} \left(1 - \frac{s}{r}\right)$ and $A_2 = \frac{1}{q'} \left(1 - \frac{s}{t}\right)$, where $\frac{1}{r} + \frac{1}{t} = 1$, r > 1, and $0 < s \le 1$. A weaker version of the following result (without weight functions $\psi_{q'}$ and $\psi_{p'}$) was obtained in [156].

Corollary 4.6 Suppose p, q, and λ are as in (2.1) and (2.2). If $\frac{1}{r} + \frac{1}{t} = 1$, r > 1, and $0 < s \le 1$, then the inequalities

$$\sum_{m=0}^{\infty} \sum_{n=0}^{\infty} \frac{a_m b_n}{[(m+\mu)^s + (n+\mu)^s]^{\lambda}} < \left\{ \sum_{m=0}^{\infty} (m+\mu)^{p(\frac{1}{p'} - \frac{s\lambda}{r})} \left[\frac{\pi}{s\sin(\frac{\pi}{r})} - 2^{s-1} \psi_{q'}(m,s,\mu) \right]^{\frac{p}{q'}} a_m^p \right\}^{\frac{1}{p}} \times \left\{ \sum_{n=0}^{\infty} (n+\mu)^{q(\frac{1}{q'} - \frac{s\lambda}{r})} \left[\frac{\pi}{s\sin(\frac{\pi}{r})} - 2^{s-1} \psi_{p'}(n,s,\mu) \right]^{\frac{q}{p'}} b_n^q \right\}^{\frac{1}{q}}$$
(4.41)

and

$$\begin{cases} \sum_{n=0}^{\infty} (n+\mu)^{\frac{sq'\lambda}{t}-1} \left[\frac{\pi}{s\sin\left(\frac{\pi}{t}\right)} - 2^{s-1} \psi_{p'}(n,s,\mu) \right]^{-\frac{q'}{p'}} \\ \times \left[\sum_{m=0}^{\infty} \frac{a_m}{[(m+\mu)^s + (n+\mu)^s]^{\lambda}} \right]^{q'} \end{cases}^{\frac{1}{q'}} \\ < \left\{ \sum_{m=0}^{\infty} (m+\mu)^{p(\frac{1}{p'} - \frac{s\lambda}{r})} \left[\frac{\pi}{s\sin\left(\frac{\pi}{r}\right)} - 2^{s-1} \psi_{q'}(m,s,\mu) \right]^{\frac{p}{q'}} a_m^p \right\}^{\frac{1}{p}} \tag{4.42}$$

hold for all non-negative sequences $(a_m)_{m \in \mathbb{N}_0}$, $(b_n)_{n \in \mathbb{N}_0} \neq 0$, where $\psi_{q'}(m, s, \mu)$ and $\psi_{p'}(n, s, \mu)$ are defined in Corollary 4.4. In addition, these inequalities are equivalent.

4.3 Some particular refinements

Discrete Hilbert-type inequalities are more complicated than the corresponding integral inequalities, since we use some additional estimates in order to obtain a suitable form of inequality. Of course, this causes some extra conditions on functions and parameters involved in the corresponding discrete inequality. For example, the general Theorem 1.14 (Chapter 1) dealing with homogeneous kernels, includes kernels which are strictly decreasing in both arguments, while the corresponding integral result, that is, Corollary 1.1, holds for an arbitrary non-negative homogeneous kernel.

The starting point in this section is the following consequence of the above mentioned Theorem 1.14, that is, a pair of equivalent inequalities

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \frac{a_m b_n}{(m+n)^s} \le L \left[\sum_{m=1}^{\infty} m^{1-s+p(A_1-A_2)} a_m^p \right]^{\frac{1}{p}} \left[\sum_{n=1}^{\infty} n^{1-s+q(A_2-A_1)} b_n^q \right]^{\frac{1}{q}}$$
(4.43)

and

$$\sum_{n=1}^{\infty} n^{(s-1)(p-1)+p(A_1-A_2)} \left[\sum_{m=1}^{\infty} \frac{a_m}{(m+n)^s} \right]^p \le L^p \sum_{m=1}^{\infty} m^{1-s+p(A_1-A_2)} a_m^p, \quad (4.44)$$

where $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, s > 0, $A_1 \in (\max\{0, \frac{1-s}{q}\}, \frac{1}{q})$, $A_2 \in (\max\{0, \frac{1-s}{p}, \frac{1}{p}\}, \frac{1}{p})$, and the constant *L* is expressed in terms of the Beta function, i.e. $L = B^{\frac{1}{p}}(1 - A_2p, s - 1 + A_2p)B^{\frac{1}{q}}(1 - A_1q, s - 1 + A_1q)$.

Note that parameters A_1 and A_2 in (4.43) and (4.44) are non-negative, while the corresponding result in the integral case (Corollary 1.1, Chapter 1) refers to parameters which can be negative, that is, $A_1 \in (\frac{1-s}{q}, \frac{1}{q})$, $A_2 \in (\frac{1-s}{p}, \frac{1}{p})$. In other words, the intervals for parameters A_1 and A_2 in the discrete case are smaller than the corresponding intervals in the integral case.

On the other hand, the above intervals $(\max\{0, \frac{1-s}{q}\}, \frac{1}{q})$ and $(\max\{0, \frac{1-s}{p}, \frac{1}{p})$ for parameters A_1 and A_2 can also be extended so that inequalities (4.43) and (4.44) still hold. This can be done by virtue of the Euler-Maclaurin summation formula, utilizing similar methods as in the previous two sections. Hence, in order to establish extensions of inequalities (4.43) and (4.44), we first mention some preliminary estimates.

Let $f:[1,\infty) \to \mathbb{R}$ be a non-negative continuously differentiable function such that $\sum_{k=1}^{\infty} f(k) < \infty$ and $\int_{1}^{\infty} f(t) dt < \infty$. It follows from Lemma 4.1 that

$$\sum_{k=1}^{\infty} f(k) = \int_{1}^{\infty} f(t)dt + \frac{1}{2}f(1) + \int_{1}^{\infty} \gamma_{1}^{*}(t)f'(t)dt, \qquad (4.45)$$

where $\gamma_1^*(t) = t - \lfloor t \rfloor - \frac{1}{2}$.

Moreover, if $f: [1,\infty) \to \mathbb{R}$ is of class C^4 such that $f^{(r)}(\infty) = 0$, r = 0, 1, 2, 3, 4, $f^{(2r)}(x) > 0$, and $f^{(2r-1)}(x) < 0$, r = 1, 2, then the following sequence of inequalities hold:

$$\frac{1}{12}f(1) < \int_{1}^{\infty} \gamma_{1}^{*}(t)f(t)dt < -\frac{1}{12}f(1) + \frac{1}{720}f''(1) < 0.$$
(4.46)

The above sequence of inequalities was established in [147], by virtue of the Euler-Maclaurin summation formula. Relations (4.45) and (4.46) will be utilized in extending inequalities (4.43) and (4.44).

First, we define the functions $f_{s,\alpha,n}(t) = t^{-\alpha}(t+n)^{-s}$, where $-1 \le \alpha < 1, 0 < s \le 14$, $n \in \mathbb{N}$, and

$$Q_{s,\alpha}(n) = \int_0^1 f_{s,\alpha,n}(t) dt - \frac{1}{2} f_{s,\alpha,n}(1) - \int_1^\infty f'_{s,\alpha,n}(t) \gamma_1^*(t) dt,$$

where $\gamma_1^*(t) = t - \lfloor t \rfloor - \frac{1}{2}$. The following lemma asserts that under the above assumptions $Q_{s,\alpha}(n)$ is a non-negative function.

Lemma 4.8 If $-1 \le \alpha < 1$ and $0 < s \le 14$, then $Q_{s,\alpha}(n) > 0$.

Proof. Utilizing the integration by parts three times, we have

$$\begin{split} \int_0^1 f_{s,\alpha,n}(t) dt &= \frac{1}{(1-\alpha)(n+1)^s} + \frac{s}{1-\alpha} \int_0^1 \frac{t^{1-\alpha}}{(t+n)^{s+1}} dt \\ &= \frac{1}{(1-\alpha)(n+1)^s} + \frac{s}{(1-\alpha)(2-\alpha)(n+1)^{s+1}} \\ &+ \frac{s(s+1)}{(1-\alpha)(2-\alpha)} \int_0^1 \frac{t^{2-\alpha}}{(t+n)^{s+2}} dt \\ &= \frac{1}{(1-\alpha)(n+1)^s} + \frac{s}{(1-\alpha)(2-\alpha)(n+1)^{s+1}} \\ &+ \frac{s(s+1)}{(1-\alpha)(2-\alpha)(3-\alpha)(n+1)^{s+2}} \\ &+ \frac{s(s+1)(s+2)}{(1-\alpha)(2-\alpha)(3-\alpha)} \int_0^1 \frac{t^{3-\alpha}}{(t+n)^{s+3}} dt. \end{split}$$

Further, since the function $f_{s,\alpha,n} : \mathbb{R}_+ \to \mathbb{R}$ is non-negative, we obtain the inequality

$$\int_{0}^{1} f_{s,\alpha,n}(t) dt > \frac{1}{(1-\alpha)} \left[\frac{1}{(n+1)^{s}} + \frac{s}{(2-\alpha)(n+1)^{s+1}} + \frac{s(s+1)}{(2-\alpha)(3-\alpha)(n+1)^{s+2}} \right].$$
(4.47)

In addition,

$$-\frac{1}{2}f_{s,\alpha,n}(1) = -\frac{1}{2(n+1)^s}.$$
(4.48)

It remains to estimate the last term in the expression for $Q_{s,\alpha}(n)$. The first derivative of $f_{s,\alpha,n}$ is

$$f'_{s,\alpha,n}(t) = \frac{nst^{-\alpha-1}}{(t+n)^{s+1}} - \frac{(s+\alpha)t^{-\alpha-1}}{(t+n)^s},$$

that is,

$$-\int_{1}^{\infty} f_{s,\alpha,n}'(t)\gamma_{1}^{*}(t)dt = \int_{1}^{\infty} g_{1}(t)\gamma_{1}^{*}(t)dt - \int_{1}^{\infty} g_{2}(t)\gamma_{1}^{*}(t)dt,$$

where

$$g_1(t) = \frac{(s+\alpha)t^{-\alpha-1}}{(t+n)^s}$$
 and $g_2(t) = \frac{nst^{-\alpha-1}}{(t+n)^{s+1}}.$

Now, using (4.46), we obtain

$$\int_{1}^{\infty} g_1(t)\gamma_1^*(t)dt > -\frac{1}{12}g_1(1) = -\frac{s+\alpha}{12(n+1)^s}$$

and

$$\begin{split} &-\int_{1}^{\infty} g_{2}(t)\gamma_{1}^{*}(t)dt \\ &> \frac{1}{12}g_{2}(1) - \frac{1}{720}g_{2}^{\prime\prime}(1) \\ &= \frac{ns}{12(n+1)^{s+1}} - \frac{ns}{720} \left[\frac{(s+1)(s+2)}{(n+1)^{s+3}} + \frac{2(s+1)(\alpha+1)}{(n+1)^{s+2}} + \frac{(\alpha+1)(\alpha+2)}{(n+1)^{s+1}} \right] \\ &> \frac{(n+1)s-s}{12(n+1)^{s+1}} - \frac{s}{720} \left[\frac{(s+1)(s+2)}{(n+1)^{s+2}} + \frac{2(s+1)(\alpha+1)}{(n+1)^{s+1}} + \frac{(\alpha+1)(\alpha+2)}{(n+1)^{s}} \right] \end{split}$$

which implies

$$-\int_{1}^{\infty} f_{s,\alpha,n}'(t)\gamma_{1}^{*}(t)dt$$

$$> -\frac{\alpha}{12(n+1)^{s}} - \frac{s}{12(n+1)^{s+1}}$$

$$-\frac{s}{720} \left[\frac{(s+1)(s+2)}{(n+1)^{s+2}} + \frac{2(s+1)(\alpha+1)}{(n+1)^{s+1}} + \frac{(\alpha+1)(\alpha+2)}{(n+1)^{s}} \right].$$
(4.49)

,

Finally, taking into account (4.47), (4.48), and (4.49), we obtain

$$Q_{s,\alpha}(n) > \frac{1}{(n+1)^s} Q_0(s,\alpha) + \frac{1}{(n+1)^{s+1}} Q_1(s,\alpha) + \frac{1}{(n+1)^{s+2}} Q_2(s,\alpha),$$

where

$$\begin{aligned} Q_0(s,\alpha) &= \frac{1}{1-\alpha} - \frac{1}{2} - \frac{\alpha}{12} - \frac{s(\alpha+1)(\alpha+2)}{720}, \\ Q_1(s,\alpha) &= \frac{s}{(1-\alpha)(2-\alpha)} - \frac{s}{12} - \frac{s(s+1)(\alpha+1)}{360}, \\ Q_2(s,\alpha) &= \frac{s(s+1)}{(1-\alpha)(2-\alpha)(3-\alpha)} - \frac{s(s+1)(s+2)}{720} \end{aligned}$$

It is enough to show that the expressions $Q_0(s, \alpha)$, $Q_1(s, \alpha)$, and $Q_2(s, \alpha)$ are non-negative. Obviously, we have $Q_2(s, \alpha) > s(s+1)\left(\frac{1}{24} - \frac{1}{45}\right) > 0$ and $Q_1(s, \alpha) > s\left(\frac{1}{6} - \frac{1}{12} - \frac{1}{12}\right) = 0$. In addition, if $s \le 14$ then $\frac{s}{720} \le \frac{1}{24}$, so that

$$Q_0(s,\alpha) \ge \frac{1}{1-\alpha} - \frac{1}{2} - \frac{\alpha}{12} - \frac{(\alpha+1)(\alpha+2)}{24} = \frac{\alpha^3 + 4\alpha^2 + 9\alpha + 10}{24(1-\alpha)}.$$

It is easy to check that the function $f(\alpha) = \alpha^3 + 4\alpha^2 + 9\alpha + 10$ is strictly increasing. Since f(-1) = 4, it follows that $Q_0(s, \alpha) > 0$. The proof is now completed.

In order to extend inequalities (4.43) and (4.44), it is necessary to establish a suitable upper bound on the weight function

$$\omega_{s,\alpha_1,\alpha_2}(n) = \sum_{m=1}^{\infty} \frac{1}{(m+n)^s} \frac{n^{\alpha_1}}{m^{\alpha_2}},$$
(4.50)

where $0 < s \le 14$, and either $\alpha_2 \in (1 - s, 1)$ for $s \le 2$, or $\alpha_2 \in [-1, 1)$ for s > 2. Of course, due to the form of the weight function, such a bound will involve the Beta function.

Lemma 4.9 *If* $0 < s \le 14$, and either $\alpha_2 \in (1 - s, 1)$ for $s \le 2$, or $\alpha_2 \in [-1, 1)$ for s > 2, *then*

$$\omega_{s,\alpha_1,\alpha_2}(n) < n^{1-s+\alpha_1-\alpha_2} B(1-\alpha_2, s+\alpha_2-1).$$
(4.51)

Proof. Since $f_{s,\alpha_2,n}(t) = t^{-\alpha_2}(t+n)^{-s}$, we have $\omega_{s,\alpha_1,\alpha_2}(n) = n^{\alpha_1} \sum_{m=1}^{\infty} f_{s,\alpha_2,n}(m)$. Further, by virtue of Lemma 4.8, it follows that

$$\omega_{s,\alpha_1,\alpha_2}(n) = n^{\alpha_1} \left[\int_0^\infty f_{s,\alpha_2,n}(t) dt - Q_{s,\alpha_2}(n) \right] < n^{\alpha_1} \int_0^\infty f_{s,\alpha_2,n}(t) dt$$

Finally, utilizing the change of variables x = nt, we have

$$\int_0^{\infty} f_{s,\alpha_2,n}(t) dt = n^{1-s-\alpha_2} \int_0^{\infty} \frac{x^{-\alpha_2}}{(1+x)^s} dx,$$

so that (4.51) holds due to the well-known formula

$$\int_0^\infty \frac{x^{-\alpha_2}}{(1+x)^s} dx = B(1-\alpha_2, s+\alpha_2-1).$$

The following result extends inequalities (4.43) and (4.44) in the sense that the parameters A_1 and A_2 can be chosen from a larger interval.

Theorem 4.6 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and $0 < s \le 14$. If either $A_1 \in \left(\frac{1-s}{q}, \frac{1}{q}\right)$, $A_2 \in \left(\frac{1-s}{p}, \frac{1}{p}\right)$ for $s \le 2$, or $A_1 \in \left[-\frac{1}{q}, \frac{1}{q}\right)$, $A_2 \in \left[-\frac{1}{p}, \frac{1}{p}\right)$ for s > 2, then inequalities (4.43) and (4.44) hold for all non-negative sequences $(a_m)_{m \in \mathbb{N}}$, $(b_n)_{n \in \mathbb{N}} \neq 0$. In addition, these inequalities are equivalent.

Proof. Making use of Theorem 1.9 (Chapter 1), it is enough to show that (4.43) holds under conditions of this theorem.

Clearly, considering the discrete form of inequality (1.12) (Theorem 1.9) with the kernel $K(m,n) = (m+n)^{-s}$, and the weight functions $\varphi(m) = m^{A_1}$, $\varphi(n) = n^{A_2}$, it follows that

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \frac{a_m b_n}{(m+n)^s} \le \left[\sum_{m=1}^{\infty} \omega_{s,pA_1,pA_2}(m) a_m^p \right]^{\frac{1}{p}} \left[\sum_{n=1}^{\infty} \omega_{s,qA_2,qA_1}(n) b_n^q \right]^{\frac{1}{q}},$$

where $\omega_{s,\alpha_1,\alpha_2}$ is defined by (4.50). Now, the result follows from Lemma 4.9.

Remark 4.11 Obviously, Theorem 4.6 can also be extended to the case of non-conjugate exponents. For more details, the reader is referred to [55].

We know from Section 1.4 (Chapter 1) that the constant *L* appearing in (4.43) and (4.44) is the best possible if the parameters $A_1 \in \left(\max\left\{0, \frac{1-s}{q}\right\}, \frac{1}{q}\right)$ and $A_2 \in \left(\max\left\{0, \frac{1-s}{p}, \right\}, \frac{1}{p}\right)$ fulfill condition $pA_2 + qA_1 = 2 - s$. The following theorem asserts that *L* is also the best constant in the context of Theorem 4.6.

Theorem 4.7 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let A_1 , A_2 , s be real parameters fulfilling conditions of Theorem 4.6. If $pA_2 + qA_1 = 2 - s$, then L is the best constant in inequalities (4.43) and (4.44).

Proof. Due to the equivalence, it is enough to show that *L* is the best constant in (4.43), under conditions of the theorem. Clearly, if $pA_2 + qA_1 = 2 - s$, this constant reduces to $L = B(1 - pA_2, pA_2 + s - 1)$.

Applying (4.43) to the sequences $a_m = m^{-qA_1 - \frac{\varepsilon}{p}}$ and $b_n = n^{-pA_2 - \frac{\varepsilon}{q}}$, $\varepsilon > 0$, its righthand side becomes $L\sum_{n=1}^{\infty} n^{-1-\varepsilon}$.

On the other hand, for the above choice of sequences $(a_m)_{m \in \mathbb{N}}$ and $(b_n)_{n \in \mathbb{N}}$ we have

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \frac{a_m b_n}{(m+n)^s} = \sum_{n=1}^{\infty} n^{-pA_2 - \frac{\varepsilon}{q}} \left(\sum_{m=1}^{\infty} h(m) \right),$$
(4.52)

where $h(t) = \frac{t^{-qA_1 - \frac{\varepsilon}{p}}}{(n+t)^s}$. Further, it follows that

$$\int_0^\infty h(t)dt = n^{1-s-qA_1-\frac{\varepsilon}{p}} B\left(1-qA_1-\frac{\varepsilon}{p}, qA_1+\frac{\varepsilon}{p}+s-1\right)$$
(4.53)

and

$$\int_{0}^{1} h(t)dt = n^{1-s-qA_{1}-\frac{\varepsilon}{p}} \int_{0}^{\frac{1}{n}} \frac{x^{-qA_{1}-\frac{\varepsilon}{p}}}{(1+x)^{s}} dx$$
$$< n^{1-s-qA_{1}-\frac{\varepsilon}{p}} \int_{0}^{\frac{1}{n}} x^{-qA_{1}-\frac{\varepsilon}{p}} dx = n^{-s}.$$
(4.54)

Moreover, taking into account that

$$h'(t) = \frac{ns}{(n+t)^{s+1}t^{qA_1 + \frac{\varepsilon}{p} + 1}} - \frac{s + qA_1 + \frac{\varepsilon}{p}}{(n+t)^{s}t^{qA_1 + \frac{\varepsilon}{p} + 1}},$$

we have

$$\int_{1}^{\infty} h'(t)\gamma_{1}^{*}(t)dt > -\frac{ns}{12(n+1)^{s+1}} > -\frac{s}{12n^{s}},$$
(4.55)

by virtue of (4.46). Now, utilizing the summation formula

$$\sum_{m=1}^{\infty} h(m) = \int_0^{\infty} h(t)dt - \int_0^1 h(t)dt + \frac{1}{2}h(1) + \int_1^{\infty} h'(t)\gamma_1^*(t)dt$$

and relations (4.53)–(4.55), we obtain

$$\sum_{m=1}^{\infty} h(m) > B(\varepsilon) n^{1-s-qA_1 - \frac{\varepsilon}{p}} - \frac{1}{n^s} - \frac{s}{12n^s},$$
(4.56)

where $B(\varepsilon) = B(1 - qA_1 - \frac{\varepsilon}{p}, qA_1 + \frac{\varepsilon}{p} + s - 1)$. Moreover, relations (4.52) and (4.56) yield the estimate

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \frac{a_m b_n}{(m+n)^s} > B(\varepsilon) \sum_{n=1}^{\infty} \frac{1}{n^{1+\varepsilon}} - \left(1 + \frac{s}{12}\right) \sum_{n=1}^{\infty} \frac{1}{n^{s+pA_2 + \frac{\varepsilon}{q}}}.$$
(4.57)

Now, assuming that there exists a constant *C*, 0 < C < L, such that (4.43) holds after replacing *L* by *C*, the above discussion implies that

$$(L-C)\sum_{n=1}^{\infty}\frac{1}{n^{1+\varepsilon}} < \left(1+\frac{s}{12}\right)\sum_{n=1}^{\infty}\frac{1}{n^{s+pA_2+\frac{\varepsilon}{q}}}.$$

Finally, letting $\varepsilon \to 0$, it follows that $L \leq C$, which contradicts the assumption 0 < C < L. \Box

Our first application of Theorem 4.6 refers to parameters $A_1 = A_2 = \frac{2-s}{pq}$. These parameters satisfy $pA_2 + qA_1 = 2 - s$, and hence, yield the inequalities involving the best constants.

Corollary 4.7 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $2 - \min\{p,q\} < s \le 2 + \max\{p,q\}$. If $L_1 = B(\frac{p+s-2}{p}, \frac{q+s-2}{q})$, then the inequalities

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \frac{a_m b_n}{(m+n)^s} < L_1 \left[\sum_{m=1}^{\infty} m^{1-s} a_m^p \right]^{\frac{1}{p}} \left[\sum_{n=1}^{\infty} n^{1-s} b_n^q \right]^{\frac{1}{q}}$$
(4.58)

and

$$\sum_{n=1}^{\infty} n^{(s-1)(p-1)} \left[\sum_{m=1}^{\infty} \frac{a_m}{(m+n)^s} \right]^p < L_1^p \sum_{m=1}^{\infty} m^{1-s} a_m^p$$
(4.59)

hold for all non-negative sequences $(a_m)_{m \in \mathbb{N}}$, $(b_n)_{n \in \mathbb{N}} \neq 0$. In addition, these inequalities are equivalent and L_1 is the best constant in both of them.

Remark 4.12 It should be noticed here that inequalities (4.58) and (4.59) were also obtained by Yang (see [148], [152], and [153]), but only for $2 - \min\{p,q\} < s \le 2$. On the other hand, considering the special case when p = q = 2 and utilizing (4.46), Yang (see [147] and [152]) showed that inequalities (4.58) and (4.59) hold for $0 < s \le 4$, which obviously coincides with the above result.

Another application of Theorems 4.6 and 4.7 refers to parameters $A_1 = \frac{2-s}{2q}$, $A_2 = \frac{2-s}{2p}$, and is closely connected with the above mentioned papers [147] and [152].

Corollary 4.8 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $0 < s \le 4$. If $L_2 = B(\frac{s}{2}, \frac{s}{2})$, then the inequalities

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \frac{a_m b_n}{(m+n)^s} < L_2 \left[\sum_{m=1}^{\infty} m^{-\frac{ps}{2}+p-1} a_m^p \right]^{\frac{1}{p}} \left[\sum_{n=1}^{\infty} n^{-\frac{qs}{2}+q-1} b_n^q \right]^{\frac{1}{q}}$$
(4.60)

and

$$\sum_{n=1}^{\infty} n^{\frac{ps}{2}-1} \left[\sum_{m=1}^{\infty} \frac{a_m}{(m+n)^s} \right]^p < L_2^p \sum_{m=1}^{\infty} m^{-\frac{ps}{2}+p-1} a_m^p$$
(4.61)

hold for all non-negative sequences $(a_m)_{m \in \mathbb{N}}$, $(b_n)_{n \in \mathbb{N}} \neq 0$. In addition, these inequalities are equivalent and L_2 is the best constant in both of them.

Remark 4.13 The main results presented in this section are derived in [55] by Krnić and Pečarić. In addition, general refinements of Hilbert-type inequalities dealing with homogeneous kernels of class C^2 and C^4 (Sections 4.1 and 4.2) are established in [56] and [57] by the same authors.

Chapter 5

Applying the Hermite-Hadamard inequality

This chapter provides a different approach for improving discrete Hilbert-type inequalities. The method we develop here, is based on the famous Hermite-Hadamard inequality regarding convex functions.

Recall that $f : [a,b] \to \mathbb{R}$ is a convex function if

$$f(tx + (1-t)y) \le tf(x) + (1-t)f(y), \tag{5.1}$$

for all $x, y \in [a, b]$ and $t \in [0, 1]$. The Hermite-Hadamard inequality asserts that

$$f\left(\frac{a+b}{2}\right) \le \frac{1}{b-a} \int_{a}^{b} f(t)dt \le \frac{f(a)+f(b)}{2},\tag{5.2}$$

where $f : [a,b] \to \mathbb{R}$ is a convex function.

The main objective of this chapter is to establish a general method for refining discrete Hilbert-type inequalities via the above inequality (5.2).

By virtue of the Hermite-Hadamard inequality, in Section 5.1 we establish the basic theorem regarding discrete Hilbert-type inequalities with a general kernel and weight functions. Further, in Section 5.2 the general result is then applied to discrete inequalities with homogeneous kernels. Moreover, in Section 5.3 some particular homogeneous kernels are considered and compared with actual results, known from the literature. Finally, in Section 5.4 some cases of non-homogeneous kernels are also considered.

Observe that refined Hilbert-type inequalities that we derive in this chapter, are considered in the case of non-conjugate parameters.

5.1 Basic theorem and some remarks

When dealing with discrete Hilbert-type inequalities, some integral bounds are used for certain sums. In Sections 1.4 and 2.2 such sums were recognized as the lower Darboux sums for the corresponding integrals. As we have already seen, this fact required monotonic decrease of the function that defines the integral sum. For example, in the statement of Theorem 1.14 (Chapter 1) the kernel was strictly decreasing in each argument.

As distinguished from the above discussion, in this section we are going to adjust the Hermite-Hadamard inequality, in order to bound the integral sum with the integral. Of course, this requires some extra assumptions concerning convexity, but as a consequence, we shall obtain better results than in the previously discussed case.

Now, we state and prove the main result that will be the basis for the results in this chapter.

Theorem 5.1 Let p, q, and λ satisfy (2.1) and (2.2), and let $m, M, n, N \in \mathbb{N}$ be such that m < M and n < N. Suppose that $K : [m - \frac{1}{2}, M + \frac{1}{2}] \times [n - \frac{1}{2}, N + \frac{1}{2}] \rightarrow \mathbb{R}$, $\varphi : [m - \frac{1}{2}, M + \frac{1}{2}] \rightarrow \mathbb{R}$, $\psi : [n - \frac{1}{2}, N + \frac{1}{2}] \rightarrow \mathbb{R}$ are non-negative measurable functions fulfilling the following conditions:

- (i) the functions $K(i,t)\psi^{-q'}(t)$ are convex on interval $[n-\frac{1}{2}, N+\frac{1}{2}]$ for every $i = m, m+1, \ldots, M$;
- (ii) the functions $K(t, j)\varphi^{-p'}(t)$ are convex on interval $[m \frac{1}{2}, M + \frac{1}{2}]$ for every j = n, n+1, ..., N.

Then the inequality

$$\sum_{i=m}^{M} \sum_{j=n}^{N} K^{\lambda}(i,j) a_{i} b_{j}$$

$$\leq \left[\sum_{i=m}^{M} \varphi^{p}(i) \left[\int_{n-\frac{1}{2}}^{N+\frac{1}{2}} \frac{K(i,t)}{\psi^{q'}(t)} dt \right]^{\frac{p}{q'}} a_{i}^{p} \right]^{\frac{1}{p}} \left[\sum_{j=n}^{N} \psi^{q}(j) \left[\int_{m-\frac{1}{2}}^{M+\frac{1}{2}} \frac{K(t,j)}{\varphi^{p'}(t)} dt \right]^{\frac{q}{p'}} b_{j}^{q} \right]^{\frac{1}{q}}$$
(5.3)

holds for all sequences $(a_i)_{i \in \mathbb{N}}$ and $(b_j)_{i \in \mathbb{N}}$ of non-negative real numbers.

Proof. We utilize the fundamental Hilbert-type inequality in the non-conjugate case, that is, Theorem 2.1. More precisely, using inequality (2.9) in a suitable discrete setting, it follows that

$$\sum_{i=m}^{M} \sum_{j=n}^{N} K^{\lambda}(i,j) a_{i} b_{j} \leq \left[\sum_{i=m}^{M} \varphi^{p}(i) F_{i}^{p} a_{i}^{p} \right]^{\frac{1}{p}} \left[\sum_{j=n}^{N} \psi^{q}(j) G_{j}^{q} b_{j}^{q} \right]^{\frac{1}{q}},$$
(5.4)

where

$$F_i = \left[\sum_{j=n}^N \frac{K(i,j)}{\psi^{q'}(j)}\right]^{\frac{1}{q'}}, \quad i = m, m+1, \dots, M,$$

and

$$G_j = \left[\sum_{i=m}^M \frac{K(i,j)}{\varphi^{p'}(i)}\right]^{\frac{1}{p'}}, \quad j = n, n+1, \dots, N.$$

The further step is to estimate the terms of the sequences F_i , i = m, m + 1, ..., M, and G_j , j = n, n + 1, ..., N, via the Hermite-Hadamard inequality. Since the functions $K(i,t)\psi^{-q'}(t)$ are convex on interval $[n - \frac{1}{2}, N + \frac{1}{2}]$ for every i = 1

Since the functions $K(i,t)\psi^{-q'}(t)$ are convex on interval $[n-\frac{1}{2}, N+\frac{1}{2}]$ for every $i = m, m+1, \ldots, M$, applying the Hermite-Hadamard inequality to intervals $[j-\frac{1}{2}, j+\frac{1}{2}]$ yields the following inequalities:

$$\frac{K(i,j)}{\psi^{q'}(j)} \le \int_{j-\frac{1}{2}}^{j+\frac{1}{2}} \frac{K(i,t)}{\psi^{q'}(t)} dt, \quad j=n,n+1,\dots,N.$$

Now, summing these inequalities we have

$$\sum_{j=n}^{N} \frac{K(i,j)}{\psi^{q'}(j)} \le \int_{n-\frac{1}{2}}^{N+\frac{1}{2}} \frac{K(i,t)}{\psi^{q'}(t)} dt, \quad i=m,m+1,\ldots,M,$$

that is,

$$F_{i} \leq \left[\int_{n-\frac{1}{2}}^{N+\frac{1}{2}} \frac{K(i,t)}{\psi^{q'}(t)} dt \right]^{\frac{1}{q'}}, \quad i = m, m+1, \dots, M.$$
(5.5)

Finally, the same conclusion can be drawn by exploiting the convexity of functions K(t, j) $\varphi^{-p'}(t)$, j = n, n+1, ..., N, on interval $[m - \frac{1}{2}, M + \frac{1}{2}]$. In that case we have the estimates

$$G_{j} \leq \left[\int_{m-\frac{1}{2}}^{M+\frac{1}{2}} \frac{K(t,j)}{\varphi^{p'}(t)} dt \right]^{\frac{1}{p'}}, \quad j = n, n+1, \dots, N,$$
(5.6)

and the proof is now completed.

Remark 5.1 Assuming the convergence of the series and integrals, the proof of Theorem 5.1 covers inequality (5.3) also for $M = N = \infty$. In that case, inequality (5.3) reads

$$\sum_{i=m}^{\infty} \sum_{j=n}^{\infty} K^{\lambda}(i,j) a_{i} b_{j}$$

$$\leq \left[\sum_{i=m}^{\infty} \varphi^{p}(i) \left[\int_{n-\frac{1}{2}}^{\infty} \frac{K(i,t)}{\psi^{q'}(t)} dt \right]^{\frac{p}{q'}} a_{i}^{p} \right]^{\frac{1}{p}} \left[\sum_{j=n}^{\infty} \psi^{q}(j) \left[\int_{m-\frac{1}{2}}^{\infty} \frac{K(t,j)}{\varphi^{p'}(t)} dt \right]^{\frac{q}{p'}} b_{j}^{q} \right]^{\frac{1}{q}}.$$
(5.7)

Remark 5.2 Let us explain here why the method presented in Theorem 5.1 yields a better result than the method used in Sections 1.4 and 2.2, regarding discrete case. Namely, using the notation of Theorem 5.1, the method exploited in the above mentioned sections is based on the assumption that the functions $K(i,t)\psi^{-q'}(t)$ and $K(t,j)\varphi^{-p'}(t)$, $i = m, m+1, \ldots, M$, $j = n, n+1, \ldots, N$, are strictly decreasing on their domains. Hence, $\sum_{i=m}^{M} K(i,t)\psi^{-q'}(t)$ and $\sum_{j=n}^{N} K(t,j)\varphi^{-p'}(t)$ are the lower Darboux sums for the corresponding integrals, that is,

$$\sum_{j=n}^{N} \frac{K(i,j)}{\psi^{q'}(j)} \leq \int_{n-1}^{N} \frac{K(i,t)}{\psi^{q'}(t)} dt, \quad \sum_{i=m}^{M} \frac{K(i,j)}{\varphi^{p'}(i)} \leq \int_{m-1}^{M} \frac{K(t,j)}{\varphi^{p'}(t)} dt,$$

i = m, m + 1, ..., M, j = n, n + 1, ..., N, provided that all functions are defined on the corresponding intervals. Clearly, due to the described monotonicity, these estimates are less precise than estimates (5.5) and (5.6).

Remark 5.3 The inequality sign in (5.4) depends only on the parameters p', q', and λ , since the crucial step in proving this relation was in applying the Hölder inequality (see Theorem 2.1). Therefore, the reverse inequality in (5.4) holds under conditions of Remark 2.2. On the other hand, in order to obtain the reverse inequality of (5.3), the assumptions of Remark 2.2 should be consistent with the estimates for F_i , $i = m, m + 1, \ldots, M$, and G_j , $j = n, n+1, \ldots, N$ (see the proof of Theorem 5.1). In other words, estimates (5.5) and (5.6) should also hold with the reverse inequality, and that is possible only if p', q' < 0, which implies that 0 < p, q < 1. Comparing this with Remark 2.2, it follows that the reverse inequality in (5.3) holds if and only if 0 < p, q < 1.

Remark 5.4 The equivalent form assigned to (5.3), that is, the Hardy-Hilbert-type inequality, reads

$$\begin{cases} \sum_{j=n}^{N} \psi^{-q'}(j) \left[\int_{m-\frac{1}{2}}^{M+\frac{1}{2}} \frac{K(t,j)}{\varphi^{p'}(t)} dt \right]^{-\frac{q'}{p'}} \left[\sum_{i=m}^{M} K^{\lambda}(i,j) a_{i} \right]^{q'} \end{cases}^{\frac{1}{q'}} \\ \leq \left[\sum_{i=m}^{M} \varphi^{p}(i) \left[\int_{n-\frac{1}{2}}^{N+\frac{1}{2}} \frac{K(i,t)}{\psi^{q'}(t)} dt \right]^{\frac{p}{q'}} a_{i}^{p} \right]^{\frac{1}{p}}, \tag{5.8}$$

and is established by substituting the sequence

$$b_{j} = \psi^{-q'}(j) \left[\int_{m-\frac{1}{2}}^{M+\frac{1}{2}} \frac{K(t,j)}{\varphi^{p'}(t)} dt \right]^{-\frac{q'}{p'}} \left[\sum_{i=m}^{M} K^{\lambda}(i,j) a_{i} \right]^{\frac{q'}{q}}, \quad j = n, n+1, \dots, N,$$

in (5.3). Such inequalities will not be considered in this chapter.

5.2 A unified approach to inequalities with a homogeneous kernel

In this section we are concerned with Hilbert-type inequalities involving homogeneous kernels with negative degree of homogeneity, which are defined for all positive arguments. Moreover, the weight functions will be chosen to be the power functions.

Remark 5.5 When dealing with a homogeneous kernel and the power weight functions, conditions (i) and (ii) from Theorem 5.1, referring to convexity, can be rewritten in a more suitable form. Namely, suppose that the function $f(t) = K(1,t)t^{-a}$ is convex on interval $\left[\frac{2n-1}{2M}, \frac{2N+1}{2m}\right]$, where K is homogeneous function of degree -s, s > 0 and n < N, m < M are positive integers. Then, introducing the functions $f_i(t) = K(i,t)t^{-a}$, $i = m, m+1, \dots, M$, we have

$$f_i(\lambda t_1 + (1-\lambda)t_2) = i^{-a-s} f\left(\lambda \frac{t_1}{i} + (1-\lambda)\frac{t_2}{i}\right)$$

$$\leq i^{-a-s} \lambda f\left(\frac{t_1}{i}\right) + i^{-a-s} (1-\lambda) f\left(\frac{t_2}{i}\right) = \lambda f_i(t_1) + (1-\lambda) f_i(t_2)$$

for $t_1, t_2 \in [n - \frac{1}{2}, N + \frac{1}{2}]$ and $0 \le \lambda \le 1$. This means that the functions $f_i, i = m, m + \frac{1}{2}$ 1,...,*M*, are also convex on interval $[n - \frac{1}{2}, N + \frac{1}{2}]$. By the same arguments, convexity of the function $g(t) = K(t, 1)t^{-a}$ on the interval $[\frac{2m-1}{2N}, \frac{2M+1}{2n}]$ implies convexity of the functions $g_j(t) = K(t, j)t^{-a}$, j = n, n+1, ..., N, on interval $[m - \frac{1}{2}, M + \frac{1}{2}]$.

Therefore, if $\varphi(t) = t^{A_1}$ and $\psi(t) = t^{A_2}$, then, using the notation of Theorem 5.1, the conditions

- (i') function $K(1,t)\psi^{-q'}(t)$ is convex on interval $\left[\frac{2n-1}{2M}, \frac{2N+1}{2m}\right]$;
- (i'') function $K(t,1)\varphi^{-p'}(t)$ is convex on interval $\left[\frac{2m-1}{2N},\frac{2M+1}{2n}\right]$;

imply conditions (i) and (ii) from Theorem 5.1. In particular, if m = n = 1 and $M = N = \infty$, then convexity of the functions $K(1,t)t^{-A_2q'}$ and $K(t,1)t^{-A_1p'}$ on \mathbb{R}_+ implies convexity of functions $K(i,t)t^{-A_2q'}$ and $K(t,j)t^{-A_1p'}$ on \mathbb{R}_+ for all $i, j \in \mathbb{N}$. This fact will frequently be used in order to make checking of the convexity conditions easier.

Now, in order to present the main result concerning homogeneous kernels, we define the integral

$$k(\alpha; r_1, r_2) = \int_{r_1}^{r_2} K(1, t) t^{-\alpha} dt, \quad 0 \le r_1 < r_2 \le \infty,$$
(5.9)

where the arguments α , r_1 and r_2 are such that (5.9) converges. In addition, if $r_1 = 0$ and $r_2 = \infty$, then the integral $k(\alpha; 0, \infty)$ will be denoted by $k(\alpha)$, for short, as in the previous chapters.

Theorem 5.2 Let p, q, and λ satisfy (2.1) and (2.2), let $m, M, n, N \in \mathbb{N}$ be such that m < M, n < N and let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a non-negative homogeneous function of degree -s, s > 0. If A_1 and A_2 are real parameters such that the functions $K(1,t)t^{-A_2q'}$ and $K(t,1)t^{-A_1p'}$ are convex on intervals $\left[\frac{2n-1}{2M}, \frac{2N+1}{2m}\right]$ and $\left[\frac{2m-1}{2N}, \frac{2M+1}{2n}\right]$ respectively, then the inequality

$$\sum_{i=m}^{M} \sum_{j=n}^{N} K^{\lambda}(i,j) a_{i} b_{j}$$

$$\leq \left[\sum_{i=m}^{M} i^{\frac{p}{q'}(1-s)+p(A_{1}-A_{2})} k^{\frac{p}{q'}} \left(A_{2}q'; \frac{2n-1}{2i}, \frac{2N+1}{2i} \right) a_{i}^{p} \right]^{\frac{1}{p}}$$

$$\times \left[\sum_{j=n}^{N} j^{\frac{q}{p'}(1-s)+q(A_{2}-A_{1})} k^{\frac{q}{p'}} \left(2-A_{1}p'-s; \frac{2j}{2M+1}, \frac{2j}{2m-1} \right) b_{j}^{q} \right]^{\frac{1}{q}}$$
(5.10)

holds for all sequences $(a_i)_{i \in \mathbb{N}}$ and $(b_j)_{j \in \mathbb{N}}$ of non-negative real numbers.

Proof. Exploiting Theorem 5.1 with the power weight functions $\varphi(i) = i^{A_1}$ and $\psi(j) = j^{A_2}$, and taking into account the homogeneity of kernel $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$, we have

$$\int_{n-\frac{1}{2}}^{N+\frac{1}{2}} K(i,t) t^{-A_2q'} dt = i^{1-s-A_2q'} \int_{\frac{2n-1}{2i}}^{\frac{2N+1}{2i}} K(1,t) t^{-A_2q'} dt$$
$$= i^{1-s-A_2q'} k \left(A_2q'; \frac{2n-1}{2i}, \frac{2N+1}{2i}\right)$$

and

$$\int_{m-\frac{1}{2}}^{M+\frac{1}{2}} K(t,j) t^{-A_1 p'} dt = j^{1-s-A_1 p'} \int_{\frac{2j}{2m-1}}^{\frac{2j}{2m-1}} K(1,t) t^{s+A_1 p'-2} dt$$
$$= j^{1-s-A_1 p'} k \left(2A_1 p' - s; \frac{2j}{2M+1}, \frac{2j}{2m-1} \right).$$

Hence, the result follows from (5.3).

An important consequence of Theorem 5.2 is the corresponding result for infinite series, that is, when m = n = 1 and $M = N = \infty$.

Corollary 5.1 Let p, q, and λ satisfy (2.1) and (2.2), and let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a nonnegative homogeneous function of degree -s, s > 0. If A_1 and A_2 are real parameters such that the functions $K(1,t)t^{-A_2q'}$ and $K(t,1)t^{-A_1p'}$ are convex on \mathbb{R}_+ , then the inequality

$$\sum_{i=1}^{\infty} \sum_{j=1}^{\infty} K^{\lambda}(i,j) a_{i} b_{j}$$

$$\leq \left[\sum_{i=1}^{\infty} i^{\frac{p}{q'}(1-s)+p(A_{1}-A_{2})} k^{\frac{p}{q'}} (A_{2}q';\frac{1}{2i},\infty) a_{i}^{p} \right]^{\frac{1}{p}}$$

$$\times \left[\sum_{j=1}^{\infty} j^{\frac{q}{p'}(1-s)+q(A_{2}-A_{1})} k^{\frac{q}{p'}} (2-A_{1}p'-s;0,2j) b_{j}^{q} \right]^{\frac{1}{q}}$$
(5.11)

holds for all sequences $(a_i)_{i \in \mathbb{N}}$ and $(b_j)_{i \in \mathbb{N}}$ of non-negative real numbers.

Remark 5.6 According to the obvious estimates

 $k(A_2q'; \frac{1}{2i}, \infty) \le k(A_2q')$ and $k(2 - A_1p' - s; 0, 2j) \le k(2 - A_1p' - s)$,

which are valid for all $i, j \in \mathbb{N}$, it follows that the right-hand side of inequality (5.11) is not greater than the right-hand side of inequality (2.24) for $A = B = \alpha = \beta = 1$ (see Section 2.2). In such a way we get the interpolating sequence of inequalities, that is, inequality (5.11) refines inequality (2.24). As we have already discussed, the convexity assumptions in Corollary 5.1 yield a better result than the monotonicity assumptions of the kernel in each of its arguments.

5.3 Examples with homogeneous kernels

In the sequel we consider Corollary 5.1 in some particular cases of homogeneous kernels. Our first example refers to the well-known homogeneous kernel $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$, defined by $K(x,y) = (x+y)^{-s}$, s > 0. In such a way we shall obtain the weight functions expressed in terms of the incomplete Beta function. Recall that the incomplete Beta function is defined by

$$B_r(a,b) = \int_0^r t^{a-1} (1-t)^{b-1} dt, \quad a,b > 0.$$
(5.12)

Clearly, if r = 1 the incomplete Beta function reduces to the usual Beta function (1.5) and obviously, $B_r(a,b) \le B(a,b)$, a,b > 0, $0 \le r \le 1$. Hence, in the above setting, Corollary 5.1 reduces to the following form:

Corollary 5.2 Let p, q, and λ satisfy (2.1) and (2.2), and let s > 0. If A_1 and A_2 are real parameters such that $A_1 \in \left(\max\{\frac{1-s}{p'}, 0\}, \frac{1}{p'}\right)$ and $A_2 \in \left(\max\{\frac{1-s}{q'}, 0\}, \frac{1}{q'}\right)$, then the inequality

$$\sum_{i=1}^{\infty} \sum_{j=1}^{\infty} \frac{a_i b_j}{(i+j)^{\lambda_s}} \\ \leq \left[\sum_{i=1}^{\infty} i^{\frac{p}{q'}(1-s)+p(A_1-A_2)} B^{\frac{p}{q'}}_{\frac{2i}{2i+1}}(s+A_2q'-1,1-A_2q')a^p_i \right]^{\frac{1}{p}} \\ \times \left[\sum_{j=1}^{\infty} j^{\frac{q}{p'}(1-s)+q(A_2-A_1)} B^{\frac{p}{q'}}_{\frac{2j}{2j+1}}(s+A_1p'-1,1-A_1p')b^q_j \right]^{\frac{1}{q}}$$
(5.13)

holds for all sequences $(a_i)_{i \in \mathbb{N}}$ and $(b_j)_{i \in \mathbb{N}}$ of non-negative real numbers.

Proof. In order to apply Corollary 5.1, we have to check that the functions $K(1,t)t^{-A_2q'}$ and $K(t,1)t^{-A_1p'}$ are convex on \mathbb{R}_+ , where $K(x,y) = (x+y)^{-s}$. Due to the symmetry of K, it suffices to show that the function $f(t) = (1+t)^{-s}t^{-a}$ is convex on \mathbb{R}_+ for a > 0. Its second derivative is equal to

$$f''(t) = \frac{(s+a)(s+a+1)t^2 + 2a(s+a+1)t + a(a+1)}{t^{a+2}(1+t)^{s+2}},$$
(5.14)

that is, f''(t) > 0 for $t \in \mathbb{R}_+$, since a > 0 and s > 0.

Since the assumptions of Corollary 5.1 are fulfilled, we are ready to apply inequality (5.11) in the case of homogeneous kernel $K(x,y) = (x+y)^{-s}$.

From the definition of the incomplete Beta function and passing to the new variable $t = \frac{1}{u} - 1$, we have

$$k(A_2q'; \frac{1}{2i}, \infty) = \int_{\frac{1}{2i}}^{\infty} \frac{t^{-A_2q'}}{(1+t)^s} dt = \int_{0}^{\frac{2i}{2i+1}} u^{s+A_2q'-2} (1-u)^{-A_2q'} du$$
$$= B_{\frac{2i}{2i+1}}(s+A_2q'-1, 1-A_2q').$$

Similarly, the change of variable $t = \frac{u}{1-u}$ yields

$$k(2 - A_1 p' - s; 0, 2j) = \int_0^{2j} \frac{t^{s + A_1 p' - 2}}{(1 + t)^s} dt = \int_0^{\frac{2j}{2j + 1}} u^{s + A_1 p' - 2} (1 - u)^{-A_1 p'} du$$
$$= B_{\frac{2j}{2j + 1}}(s + A_1 p' - 1, 1 - A_1 p'),$$

that is, the result follows from inequality (5.11).

Note also that the intervals defining the parameters A_1 and A_2 are established due to the domain of the incomplete Beta function and the convexity of the functions $(1+t)^s t^{-A_1p'}$ and $(1+t)^s t^{-A_2q'}$.

Remark 5.7 Since the incomplete Beta function is bounded from above by the usual Beta function with the same arguments, the estimates

$$B_{\frac{2i}{2i+1}}(s+A_2q'-1,1-A_2q') \le B(s+A_2q'-1,1-A_2q')$$

and

$$B_{\frac{2j}{2j+1}}(s+A_1p'-1,1-A_1p') \le B(s+A_1p'-1,1-A_1p'),$$

hold for all $i, j \in \mathbb{N}$. Moreover, since the right-hand sides of the above estimates do not depend on *i* and *j*, it follows that the right-hand side of (5.13) does not exceed

$$L\left[\sum_{i=1}^{\infty} i^{\frac{p}{q'}(1-s)+p(A_1-A_2)} a_i^p\right]^{\frac{1}{p}} \left[\sum_{j=1}^{\infty} j^{\frac{q}{p'}(1-s)+q(A_2-A_1)} b_j^q\right]^{\frac{1}{q}},$$

where $L = B^{\frac{1}{q'}}(s + A_2q' - 1, 1 - A_2q')B^{\frac{1}{p'}}(s + A_1p' - 1, 1 - A_1p')$. Of course, we again obtain the interpolating sequence of inequalities and inequality (5.13) refines some known results related to the Beta function (see e.g. paper [16]).

Remark 5.8 According to the previous remark, when taking the values of $A_1 = A_2 = \frac{2-s}{\lambda p'q'}$, where $\frac{2-s}{\lambda p'}$, $\frac{2-s}{\lambda q'} \in (\max\{1-s,0\},1)$, Corollary 5.2, together with Remark 5.7, provides the following inequalities

$$\begin{split} \sum_{i=1}^{\infty} \sum_{j=1}^{\infty} \frac{a_i b_j}{(i+j)^{\lambda s}} &\leq \left[\sum_{i=1}^{\infty} i^{\frac{(1-s)p}{q'}} B_{\frac{2i}{2i+1}}^{\frac{p}{q'}} \left(\frac{s+\lambda q'-2}{\lambda q'}, \frac{s+\lambda p'-2}{\lambda p'} \right) a_i^p \right]^{\frac{1}{p}} \\ &\times \left[\sum_{j=1}^{\infty} j^{\frac{(1-s)q}{p'}} B_{\frac{2j}{2j+1}}^{\frac{q}{q'}} \left(\frac{s+\lambda p'-2}{\lambda p'}, \frac{s+\lambda q'-2}{\lambda q'} \right) b_j^q \right]^{\frac{1}{q}} \\ &\leq B^{\lambda} \left(\frac{s+\lambda p'-2}{\lambda p'}, \frac{s+\lambda q'-2}{\lambda q'} \right) \left[\sum_{i=1}^{\infty} i^{\frac{(1-s)p}{q'}} a_i^p \right]^{\frac{1}{p}} \left[\sum_{j=1}^{\infty} j^{\frac{(1-s)q}{p'}} b_j^q \right]^{\frac{1}{q}}, \end{split}$$

where we used the fact that the usual Beta function is symmetric in its arguments.

Moreover, considering the kernel of degree -1 in the conjugate case, that is, when s = 1, $\lambda = 1$, p = q', and q = p', the above inequalities reduce to

$$\begin{split} \sum_{i=1}^{\infty} \sum_{j=1}^{\infty} \frac{a_i b_j}{i+j} &\leq \left[\sum_{i=1}^{\infty} B_{\frac{2i}{2i+1}} \left(\frac{1}{q}, \frac{1}{p} \right) a_i^p \right]^{\frac{1}{p}} \left[\sum_{j=1}^{\infty} B_{\frac{2j}{2j+1}} \left(\frac{1}{p}, \frac{1}{q} \right) b_j^q \right]^{\frac{1}{q}} \\ &\leq \frac{\pi}{\sin \frac{\pi}{p}} \left[\sum_{i=1}^{\infty} a_i^p \right]^{\frac{1}{p}} \left[\sum_{j=1}^{\infty} b_j^q \right]^{\frac{1}{q}}, \end{split}$$

since $B(\frac{1}{p}, \frac{1}{q}) = \pi/\sin(\pi/p)$. Note that this relation interpolates Hilbert double series theorem (1.1) (see Chapter 1). In addition, if p = q = 2, then $B_{\frac{2i}{2i+1}}(\frac{1}{2}, \frac{1}{2}) = \arctan\sqrt{2i}$, since $B_r(\frac{1}{2}, \frac{1}{2}) = 2 \arctan\sqrt{\frac{r}{1-r}}, 0 \le r \le 1$.

To end the previous discussion of some particular choices of parameters A_1 and A_2 , we give yet another example in which we are able to find the explicit formulas for the incomplete Beta functions. It is a content of the following remark.

Remark 5.9 If 1 < s < 2 then the parameters $A_1 = \frac{2-s}{p'}$ and $A_2 = \frac{2-s}{q'}$ are well-defined in the sense of Corollary 5.2. Considering inequality (5.13) in this particular case, we see that all terms with the incomplete Beta function have the form $B_r(1, s - 1)$. Now, since

$$B_r(1,s-1) = \frac{1}{s-1} \left[1 - (1-r)^{s-1} \right], \quad 0 \le r \le 1,$$

inequality (5.13), together with Remark 5.7, implies the following inequalities:

$$\begin{split} \sum_{i=1}^{\infty} \sum_{j=1}^{\infty} \frac{a_i b_j}{(i+j)^{\lambda_s}} &\leq (s-1)^{-\lambda} \left[\sum_{i=1}^{\infty} i^{(3-s)(p-1)-\lambda_p} \left[1 - (2i+1)^{1-s} \right]^{\frac{p}{q'}} a_i^p \right]^{\frac{1}{p}} \\ &\times \left[\sum_{j=1}^{\infty} i^{(3-s)(q-1)-\lambda_q} \left[1 - (2j+1)^{1-s} \right]^{\frac{q}{p'}} b_j^q \right]^{\frac{1}{q}} \\ &\leq (s-1)^{-\lambda} \left[\sum_{i=1}^{\infty} i^{(3-s)(p-1)-\lambda_p} a_i^p \right]^{\frac{1}{p}} \left[\sum_{j=1}^{\infty} i^{(3-s)(q-1)-\lambda_q} b_j^q \right]^{\frac{1}{q}} \end{split}$$

We continue with some other examples of homogeneous kernels. The next result refers to the homogeneous kernel $K:\mathbb{R}_+\times\mathbb{R}_+\to\mathbb{R}$, defined by $K(x,y)=\min^{-r}\{x,y\}\max^{r-s}\{x,y\}$, $s > 0, r \in (\frac{s}{2}, \frac{s}{2} + 1)$, to which Corollary 5.1 also applies.

Corollary 5.3 Suppose that p, q, and λ satisfy (2.1) and (2.2), and let s, r > 0 be real parameters such that $r \in (\frac{s}{2}, \frac{s}{2} + 1)$. If A_1 and A_2 are real parameters such that $A_1 \in (\frac{r-s+1}{p'}, \frac{1-r}{p'})$ and $A_2 \in (\frac{r-s+1}{q'}, \frac{1-r}{q'})$, then the inequality

$$\sum_{i=1}^{\infty} \sum_{j=1}^{\infty} \frac{a_i b_j}{\min^{\lambda_r} \{i, j\} \max^{\lambda(s-r)} \{i, j\}} \\ \leq \left[\sum_{i=1}^{\infty} i^{\frac{p}{q'}(1-s)+p(A_1-A_2)} \left[k(A_2q') - \frac{(2i)^{A_2q'+r-1}}{1-A_2q'-r} \right]^{\frac{p}{q'}} a_i^p \right]^{\frac{1}{p}} \\ \times \left[\sum_{j=1}^{\infty} j^{\frac{q}{p'}(1-s)+q(A_2-A_1)} \left[k(A_1p') - \frac{(2j)^{A_1p'+r-1}}{1-A_1p'-r} \right]^{\frac{q}{p'}} b_j^q \right]^{\frac{1}{q}}$$
(5.15)

holds for all sequences $(a_i)_{i \in \mathbb{N}}$ and $(b_j)_{j \in \mathbb{N}}$ of non-negative real numbers, where

$$k(a) = \frac{s - 2r}{(1 - a - r)(a + s - r - 1)}.$$

Proof. We have to check that the kernel $K(x,y) = \min^{-r} \{x,y\} \max^{r-s} \{x,y\}$ fulfills conditions of Corollary 5.1. Due to the symmetry, it suffices to show that the function $f(t) = K(1,t)t^{-a}$ is convex on \mathbb{R}_+ for $a \in (r-s+1,1-r)$. Clearly, f is defined by

$$f(t) = \begin{cases} t^{-a-r}, & 0 < t \le 1\\ t^{r-a-s}, & t > 1. \end{cases}$$

Obviously, f is convex on intervals (0,1] and $(1,\infty)$. Moreover, we have $f'_{-}(1) = -a - r < r - a - s = f'_{+}(1)$, which means that f is convex on \mathbb{R}_+ .

Now, exploiting Corollary 5.1 we have

$$k(A_2q'; \frac{1}{2i}, \infty) = \int_{\frac{1}{2i}}^{1} t^{-A_2q'-r} dt + \int_{1}^{\infty} t^{r-A_2q'-s} dt = k(A_2q') - \frac{(2i)^{A_2q'+r-1}}{1 - A_2q'-r}$$

and

$$k(2-A_1p'-s;0,2j) = \int_0^1 t^{A_1p'+s-r-2}dt + \int_1^{2j} t^{A_1p'+r-2}dt$$
$$= k(A_1p') - \frac{(2j)^{A_1p'+r-1}}{1-A_1p'-r},$$

that is, we get inequality (5.15) from (5.11).

So far, we have considered homogeneous kernels with the negative degree of homogeneity. This restriction was adjusted to particular settings including the Beta and the incomplete Beta function. However, assuming the convergence, Theorem 5.2 and Corollary 5.1 are also meaningful for homogeneous kernels with the non-negative degree of homogeneity.

Our next example deals with a particular homogeneous kernel of zero-degree, that is, $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$, defined by $K(x,y) = \frac{x}{y}$. Let us emphasize some significant characteristics of this kernel. In contrast to $K(x,y) = (x+y)^{-s}$, the kernel $K(x,y) = \frac{x}{y}$ is not symmetric, and is not strictly decreasing in both of its arguments. On the other hand, it fulfills convexity conditions of Corollary 5.1, hence we have the following result:

Corollary 5.4 Suppose that p, q, and λ satisfy conditions (2.1) and (2.2). If A_1 and A_2 are real parameters such that $A_1 > \frac{2}{p'}$ and $A_2 > 0$, then the inequality

$$\sum_{i=1}^{\infty} \sum_{j=1}^{\infty} \left(\frac{i}{j}\right)^{\lambda} a_{i} b_{j}$$

$$\leq \frac{2^{A_{1}+A_{2}-\frac{2}{p'}}}{(A_{1}p'-2)^{\frac{1}{p'}} (A_{2}q')^{\frac{1}{q'}}} \left[\sum_{i=1}^{\infty} i^{\frac{p}{q'}+pA_{1}} a_{i}^{p}\right]^{\frac{1}{p}} \left[\sum_{j=1}^{\infty} j^{-\frac{q}{p'}+qA_{2}} b_{j}^{q}\right]^{\frac{1}{q}}$$
(5.16)

holds for all sequences $(a_i)_{i \in \mathbb{N}}$ and $(b_j)_{i \in \mathbb{N}}$ of non-negative real numbers.

Proof. In order to exploit Corollary 5.1 for the homogeneous kernel $K(x,y) = \frac{x}{y}$, we easily check that the functions $K(1,t)t^{-A_2q'} = t^{-1-A_2q'}$ and $K(t,1)t^{-A_1p'} = t^{1-A_1p'}$ are convex on \mathbb{R}_+ for the parameters A_1 and A_2 as in the statement of this corollary. Therefore we have

$$k(A_2q';\frac{1}{2i},\infty) = \int_{\frac{1}{2i}}^{\infty} t^{-1-A_2q'} dt = \frac{(2i)^{A_2q'}}{A_2q'}$$

and similarly,

$$k(2 - A_1 p'; 0, 2j) = \int_0^{2j} t^{A_1 p' - 3} dt = \frac{(2j)^{A_1 p' - 2}}{A_1 p' - 2},$$

so the result follows from (5.11).

5.4 A non-conjugate example

The method developed in this chapter is often useful when dealing with some non-homogeneous kernels. Here we are going to consider the kernel $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$, defined by $K(x,y) = (1 + xy)^{-s}$, s > 0. Clearly, we start here with a general result, that is, with Theorem 5.1, since the kernel is non-homogeneous. As a consequence, we obtain the following:

Corollary 5.5 Let p, q, and λ satisfy conditions (2.1) and (2.2), and let s > 0. If A_1 and A_2 are real parameters such that $A_1 \in \left(\max\{\frac{1-s}{p'}, 0\}, \frac{1}{p'}\right)$ and $A_2 \in \left(\max\{\frac{1-s}{q'}, 0\}, \frac{1}{q'}\right)$, then the inequality

$$\sum_{i=1}^{\infty} \sum_{j=1}^{\infty} \frac{a_i b_j}{(1+ij)^{\lambda_s}} \\ \leq \left[\sum_{i=1}^{\infty} i^{-\frac{p}{q'} + p(A_1 + A_2)} B_{\frac{2}{i+2}}^{\frac{p}{q'}} (s + A_2 q' - 1, 1 - A_2 q') a_i^p \right]^{\frac{1}{p}} \\ \times \left[\sum_{j=1}^{\infty} j^{-\frac{q}{p'} + q(A_1 + A_2)} B_{\frac{2}{j+2}}^{\frac{q}{q'}} (s + A_1 p' - 1, 1 - A_1 p') b_j^q \right]^{\frac{1}{q}}$$
(5.17)

holds for all sequences $(a_i)_{i \in \mathbb{N}}$ and $(b_j)_{i \in \mathbb{N}}$ of non-negative real numbers.

Proof. In order to be able to apply (5.7), we have to check convexity conditions (i) and (ii) from Theorem 5.1, for the kernel $K(x,y) = (1+xy)^{-s}$ and the weight functions $\varphi(x) = x^{A_1}$ and $\psi(y) = y^{A_2}$.

Due to the symmetry, it suffices to show that the functions $g_i(t) = (1+it)^{-s}t^{-a}$, $i \in \mathbb{N}$, are convex on \mathbb{R}_+ for a > 0. These functions can be rewritten as $g_i(t) = i^a f(it)$, where $f(t) = (1+t)^{-s}t^{-a}$. The second derivative is $g''_i(t) = i^{a+2}f''(it)$, $i \in \mathbb{N}$. In addition, the second derivative of f is given by (5.14), which proves the convexity of functions $g_i, i \in \mathbb{N}$, on \mathbb{R}_+ .

Therefore, the assumptions of Theorem 5.1 are fulfilled, so we use (5.7) with m = n = 1. Using the change of variable $u = \frac{1}{it+1}$, $i \in \mathbb{N}$, we conclude that

$$\int_{\frac{1}{2}}^{\infty} \frac{K(i,t)}{\psi^{q'}(t)} dt = \int_{\frac{1}{2}}^{\infty} (1+it)^{-s} t^{-A_2q'} dt = i^{q'A_2-1} \int_{0}^{\frac{2}{l+2}} u^{s+A_2q'-2} (1-u)^{-A_2q'} du$$
$$= i^{q'A_2-1} B_{\frac{2}{l+2}} (s+A_2q'-1, 1-A_2q').$$

Due to the symmetry, we also have

$$\int_{\frac{1}{2}}^{\infty} \frac{K(t,j)}{\varphi^{p'}(t)} dt = \int_{\frac{1}{2}}^{\infty} (1+jt)^{-s} t^{-A_1p'} dt = j^{p'A_1-1} B_{\frac{2}{j+2}}(s+A_1p'-1,1-A_1p'),$$

where $j \in \mathbb{N}$, so the result follows.

Remark 5.10 Since the incomplete Beta function is bounded from above by the usual Beta function with the same parameters, the right-hand side of inequality (5.17) does not exceed

$$L\left[\sum_{i=1}^{\infty} i^{-\frac{p}{q'}+p(A_1+A_2)} a_i^p\right]^{\frac{1}{p}} \left[\sum_{j=1}^{\infty} j^{-\frac{q}{p'}+q(A_1+A_2)} b_j^q\right]^{\frac{1}{q}},$$
(5.18)

where $L = B^{\frac{1}{q'}}(s + A_2q' - 1, 1 - A_2q')B^{\frac{1}{p'}}(s + A_1p' - 1, 1 - A_1p')$. This also yields interpolating sequence of inequalities that we have already discussed in the previous section. As an illustration, we consider a particular choice of parameters A_1 and A_2 as in Remark 5.8, that is, $A_1 = A_2 = \frac{2-s}{\lambda p'q'}$, where $\frac{2-s}{\lambda p'}, \frac{2-s}{\lambda q'} \in (\max\{1-s,0\},1)$. In this case, the above expression (5.18) reads

$$B^{\lambda}\left(\frac{s+\lambda p'-2}{\lambda p'},\frac{s+\lambda q'-2}{\lambda q'}\right)\left[\sum_{i=1}^{\infty}i^{\frac{p}{q'}\left[\frac{2(2-s)}{\lambda p'}-1\right]}a_{i}^{p}\right]^{\frac{1}{p}}\left[\sum_{j=1}^{\infty}j^{\frac{q}{p'}\left[\frac{2(2-s)}{\lambda q'}-1\right]}b_{j}^{q}\right]^{\frac{1}{q}}.$$
(5.19)

In the conjugate case, expression (5.19) represents the right-hand side of the appropriate inequality from [159]. Hence, relation (5.17) may be regarded as both a refinement and a generalization of the above mentioned result from [159].

Remark 5.11 The method of improving Hilbert-type inequalities via the Hermite-Hadamard inequality, presented in this chapter, has been recently developed in [66].

Chapter **6**

Hilbert-type inequalities and the Laplace transform

In this short chapter we study refinements of some particular Hilbert-type inequalities involving the Laplace transform. Let us recall that if $f : \mathbb{R}_+ \to \mathbb{R}$ is a Lebesgue measurable function, then the Laplace transform $\mathscr{L}f$ of f is defined by $(\mathscr{L}f)(x) = \int_0^\infty \exp(-xt)f(t)dt$, for each x such that the above integral converges.

Considering the Hilbert-type inequality with the kernel $(x+y)^{-s}$, s > 0, Peachey [108], obtained the following interpolating sequence of inequalities in the case of conjugate parameters.

Theorem 6.1 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $b > -\frac{1}{p}$, $c > -\frac{1}{q}$. If $f, g : \mathbb{R}_+ \to \mathbb{R}$, $f, g \neq 0$, are non-negative measurable functions with the respective Laplace transforms $\mathscr{L}f$ and $\mathscr{L}g$, then

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{f(x)g(y)}{(x+y)^{b+c+1}} dx dy \leq \frac{1}{\Gamma(b+c+1)} \|t^{b} (\mathscr{L}f)(t)\|_{L^{p}(\mathbb{R}_{+})} \|t^{c} (\mathscr{L}g)(t)\|_{L^{q}(\mathbb{R}_{+})} < B\left(b+\frac{1}{p},c+\frac{1}{q}\right) \|x^{1-b-\frac{2}{p}}f(x)\|_{L^{p}(\mathbb{R}_{+})} \|y^{1-c-\frac{2}{q}}g(y)\|_{L^{q}(\mathbb{R}_{+})}.$$
(6.1)

Note that the middle term in (6.1) that includes norms with the Laplace transforms of functions f and g, interpolates between the left-hand side and the right-hand side of the corresponding Hilbert-type inequality. Moreover, taking into account the form of the kernel, the above inequalities include the constant factors expressed in terms of the Beta and Gamma functions.

In the same paper Peachey also derived an analogue of the above interpolating sequence of inequalities, in the setting with non-conjugate parameters.

Theorem 6.2 Suppose that p > 1, q > 1, $\frac{1}{p} + \frac{1}{q} \ge 1$, $q \le r \le p'$, $b + \frac{1}{r'} > 0$, and $c + \frac{1}{r} > 0$. If $f, g : \mathbb{R}_+ \to \mathbb{R}$, $f, g \ne 0$, are non-negative measurable functions with the respective Laplace transforms $\mathcal{L}f$ and $\mathcal{L}g$, then

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{f(x)g(y)}{(x+y)^{b+c+1}} dx dy \leq \frac{1}{\Gamma(b+c+1)} \|t^{b} (\mathscr{L}f)(t)\|_{L^{r'}(\mathbb{R}_{+})} \|t^{c} (\mathscr{L}g)(t)\|_{L^{r}(\mathbb{R}_{+})} < C \|x^{\frac{1}{p'} - \frac{1}{r'} - b} f(x)\|_{L^{p}(\mathbb{R}_{+})} \|y^{\frac{1}{q'} - \frac{1}{r} - c} g(y)\|_{L^{q}(\mathbb{R}_{+})},$$
(6.2)

where $\beta = \frac{1}{p'} + \frac{1}{r'}$, $\gamma = \frac{1}{q'} + \frac{1}{r}$, and

$$C = \beta^{b + \frac{1}{r'}} \gamma^{c + \frac{1}{r}} \Gamma^{\beta} \left(\frac{b}{\beta} + \frac{1}{r'\beta} \right) \Gamma^{\gamma} \left(\frac{c}{\gamma} + \frac{1}{r\gamma} \right) \Gamma^{-1}(b + c + 1)$$

Our aim in this chapter is to derive multidimensional versions of Theorems 6.1 and 6.2 with a more general parameters. In other words, we shall extend inequalities (6.1) and (6.2) regarding multidimensional conjugate and non-conjugate parameters. In such a way we shall obtain refinements of some particular Hilbert-type inequalities from Chapters 1 and 2.

Basic results will be established by virtue of the general Hardy-Hilbert-type inequality in both conjugate and non-conjugate setting.

6.1 The case of conjugate parameters

To prove the main result we first establish the following two lemmas.

Lemma 6.1 Let a > -1. If $f_i : \mathbb{R}_+ \to \mathbb{R}$, i = 1, ..., n, are non-negative measurable functions with the respective Laplace transforms $\mathcal{L}f_i$, i = 1, ..., n, then

$$\int_{\mathbb{R}^{n}_{+}} \frac{\prod_{i=1}^{n} f_{i}(x_{i})}{(\sum_{j=1}^{n} x_{j})^{a+1}} dx_{1} \dots dx_{n} = \frac{1}{\Gamma(a+1)} \int_{0}^{\infty} t^{a} \prod_{i=1}^{n} (\mathscr{L}f_{i})(t) dt.$$
(6.3)

Proof. The proof is obtained using a simple application of the Fubini theorem:

$$\int_{0}^{\infty} t^{a} \prod_{i=1}^{n} (\mathscr{L}f_{i})(t) dt = \int_{0}^{\infty} t^{a} \left(\prod_{i=1}^{n} \int_{0}^{\infty} \exp(-tx_{i}) f_{i}(x_{i}) dx_{i} \right) dt$$

= $\int_{\mathbb{R}^{n}_{+}} \prod_{i=1}^{n} f_{i}(x_{i}) \left(\int_{0}^{\infty} \exp(-t(x_{1} + \ldots + x_{n})) t^{a} dt \right) dx_{1} \ldots dx_{n}$
= $\int_{\mathbb{R}^{n}_{+}} \frac{\Gamma(a+1) \prod_{i=1}^{n} f_{i}(x_{i})}{(\sum_{j=1}^{n} x_{j})^{a+1}} dx_{1} \ldots dx_{n}.$

The following lemma is a consequence of the general Hardy-Hilbert-type inequality (1.13) (see Chapter 1), observed in a suitable setting.

Lemma 6.2 Suppose $\frac{1}{p} + \frac{1}{q} = 1$, p > 1. If $f : \mathbb{R}_+ \to \mathbb{R}$, $f \neq 0$, is a non-negative measurable function with the Laplace transform $\mathcal{L}f$, then

$$\int_{0}^{\infty} y^{p-1-p(B+C)} (\mathscr{L}f)^{p} (y) dy$$

< $\Gamma(1-pC)\Gamma^{p-1}(1-qB) \int_{0}^{\infty} x^{p(B+C)-1} f^{p}(x) dx,$ (6.4)

where $B < \frac{1}{a}$ and $C < \frac{1}{p}$.

Proof. Inequality (6.4) is an immediate consequence of (1.13) (see Theorem 1.9, Chapter 1). Namely, using the notation from the mentioned theorem and setting $K(x,y) = \exp(-yx)$, $\varphi(x) = x^B$, and $\psi(y) = y^C$, we have

$$F(x) = \int_0^\infty \psi^{-p}(y) K(x, y) dy = \int_0^\infty y^{-pC} \exp(-xy) dy = x^{pC-1} \Gamma(1 - pC)$$

and similarly, $G(y) = \int_0^\infty \varphi^{-q}(x) K(x, y) dx = y^{qB-1} \Gamma(1-qB)$, so that inequality (1.13) yields

$$\Gamma^{1-p}(1-qB) \int_{0}^{\infty} y^{(qB-1)(1-p)-pC} \left(\int_{0}^{\infty} \exp(-xy)f(x)dx \right)^{p} dy$$

$$\leq \Gamma(1-pC) \int_{0}^{\infty} x^{p(B+C)-1} f^{p}(x)dx.$$
(6.5)

Observe that equality in (6.5) holds if and only if $f(x) = K_1 x^{-qB}$ for arbitrary non-negative constant K_1 (see Remark 1.3, Chapter 1). Clearly, this condition immediately gives that nontrivial case of equality in (6.5) leads to a divergent integral.

Finally, since (qB-1)(1-p) - pC = p - 1 - p(B+C), the above relation (6.5) yields (6.4).

Now, we are ready to state and prove the basic result of this section.

Theorem 6.3 Let s > 0, $\sum_{i=1}^{n} \frac{1}{p_i} = 1$, $p_i > 1$, $\frac{1}{p_i} + \frac{1}{q_i} = 1$, i = 1, ..., n, and $\sum_{i=1}^{n} \alpha_i = 0$. If $f_i : \mathbb{R}_+ \to \mathbb{R}$, $f_i \neq 0$, i = 1, ..., n, are non-negative measurable functions with the respective Laplace transforms $\mathscr{L} f_i$, then

$$\int_{\mathbb{R}^{n}_{+}} \frac{\prod_{i=1}^{n} f_{i}(x_{i})}{(\sum_{j=1}^{n} x_{j})^{s}} dx_{1} \dots dx_{n} \leq \frac{1}{\Gamma(s)} \prod_{i=1}^{n} \|t^{\frac{1}{q_{i}} + \frac{s-n}{p_{i}} - \alpha_{i}} (\mathscr{L}f_{i})(t)\|_{L^{p_{i}}(\mathbb{R}_{+})} \\
< \frac{1}{\Gamma(s)} \prod_{i=1}^{n} \left[\Gamma^{\frac{1}{q_{i}}} (1 - q_{i}B_{i}) \Gamma^{\frac{1}{p_{i}}} (1 - p_{i}C_{i}) \right] \\
\times \prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{n-1-s+p_{i}\alpha_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}},$$
(6.6)

where $B_i + C_i = \frac{n-s}{p_i} + \alpha_i$, $B_i < \frac{1}{q_i}$, and $C_i < \frac{1}{p_i}$, i = 1, ..., n.

Proof. Utilizing Lemma 6.1 and setting exponents β_i , i = 1, ..., n, such that $\sum_{i=1}^n \beta_i = s - 1$, we have

$$\begin{split} \int_{\mathbb{R}^n_+} \frac{\prod_{i=1}^n f_i(x_i)}{(\sum_{j=1}^n x_j)^s} dx_1 \dots dx_n \ = \ \frac{1}{\Gamma(s)} \int_0^\infty t^{s-1} \prod_{i=1}^n (\mathscr{L}f_i)(t) dt \\ = \ \frac{1}{\Gamma(s)} \int_0^\infty \prod_{i=1}^n t^{\beta_i} (\mathscr{L}f_i)(t) dt. \end{split}$$

Moreover, applying the Hölder inequality, we have that

$$\int_{\mathbb{R}^n_+} \frac{\prod_{i=1}^n f_i(x_i)}{(\sum_{j=1}^n x_j)^s} dx_1 \dots dx_n \le \frac{1}{\Gamma(s)} \prod_{i=1}^n \left[\int_0^\infty t^{p_i \beta_i} \left(\mathscr{L}f_i\right)^{p_i}(t) dt \right]^{\frac{1}{p_i}}.$$
(6.7)

On the other hand, setting

$$\beta_i = \frac{1}{q_i} - (B_i + C_i), \qquad i = 1, \dots, n,$$
(6.8)

where $B_i < \frac{1}{q_i}$ and $C_i < \frac{1}{p_i}$, and taking into account Lemma 6.2, we have

$$\left[\int_{0}^{\infty} t^{p_{i}\beta_{i}} \left(\mathscr{L}f_{i}\right)^{p_{i}}(t)dt\right]^{\frac{1}{p_{i}}} = \left[\int_{0}^{\infty} t^{p_{i}-1-p_{i}(B_{i}+C_{i})} \left(\mathscr{L}f_{i}\right)^{p_{i}}(t)dt\right]^{\frac{1}{p_{i}}} < \Gamma^{\frac{1}{q_{i}}}(1-q_{i}B_{i})\Gamma^{\frac{1}{p_{i}}}(1-p_{i}C_{i})\left[\int_{0}^{\infty} x_{i}^{p_{i}(B_{i}+C_{i})-1}f_{i}^{p_{i}}(x_{i})dx_{i}\right]^{\frac{1}{p_{i}}}.$$
(6.9)

Now, putting $B_i + C_i = \frac{n-s}{p_i} + \alpha_i$ in (6.8), it follows that $\sum_{i=1}^n \beta_i = s - 1$, that is, inequality (6.9) reads

$$\begin{bmatrix} \int_{0}^{\infty} t^{p_{i}\beta_{i}} (\mathscr{L}f_{i})^{p_{i}}(t)dt \end{bmatrix}^{\frac{1}{p_{i}}} = \begin{bmatrix} \int_{0}^{\infty} t^{\frac{p_{i}}{q_{i}}-n+s-p_{i}\alpha_{i}} (\mathscr{L}f_{i})^{p_{i}}(t)dt \end{bmatrix}^{\frac{1}{p_{i}}} < \Gamma^{\frac{1}{q_{i}}} (1-q_{i}B_{i})\Gamma^{\frac{1}{p_{i}}} (1-p_{i}C_{i}) \begin{bmatrix} \int_{0}^{\infty} x_{i}^{n-1-s+p_{i}\alpha_{i}} f_{i}^{p_{i}}(x_{i})dx_{i} \end{bmatrix}^{\frac{1}{p_{i}}}.$$
(6.10)

Finally, utilizing (6.7) and (6.10), we obtain (6.6).

Remark 6.1 Applying Lemma 6.2 to parameters $B = \frac{1}{q} \left(\frac{1}{q} - \beta \right)$ and $C = \frac{1}{p} \left(\frac{1}{q} - \beta \right)$, where $\frac{1}{p} + \beta > 0$, we have $\Gamma(1 - pC)\Gamma^{p-1}(1 - qB) = \Gamma^p(\frac{1}{p} + \beta)$, that is,

$$\left\|t^{\beta}\left(\mathscr{L}f\right)\left(t\right)\right\|_{L^{p}\left(\mathbb{R}_{+}\right)} < \Gamma\left(\beta + \frac{1}{p}\right)\left\|x^{1-\beta-\frac{2}{p}}f(x)\right\|_{L^{p}\left(\mathbb{R}_{+}\right)}.$$
(6.11)

The above inequality was also established in [108]. Moreover, utilizing (6.11) in the proof of Theorem 6.3, the constant $\frac{1}{\Gamma(s)}\prod_{i=1}^{n}[\Gamma^{\frac{1}{q_i}}(1-q_iB_i)\Gamma^{\frac{1}{p_i}}(1-p_iC_i)]$ becomes $\frac{1}{\Gamma(s)}\prod_{i=1}^{n}\Gamma(1-\frac{n-s}{p_i}-\alpha_i)$.

We restate Theorem 6.3 in the case of n = 2. This result is interesting in its own right, since it may be regarded as a generalization of the mentioned result from [108], as well as a refinement of inequality (1.23) with the kernel $K(x,y) = (x+y)^{-s}$ (see Corollary 1.1, Chapter 1).

Theorem 6.4 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and s > 0. If $f, g : \mathbb{R}_+ \to \mathbb{R}$, $f, g \neq 0$, are non-negative measurable functions with the respective Laplace transforms $\mathcal{L}f$ and $\mathcal{L}g$, then

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{f(x)g(y)}{(x+y)^{s}} dx dy
\leq \frac{1}{\Gamma(s)} \|t^{\frac{1}{q} + \frac{s-2}{p} + A_{2} - A_{1}} (\mathscr{L}f)(t)\|_{L^{p}(\mathbb{R}_{+})} \|t^{\frac{1}{p} + \frac{s-2}{q} + A_{1} - A_{2}} (\mathscr{L}g)(t)\|_{L^{q}(\mathbb{R}_{+})}
< L \left[\int_{0}^{\infty} x^{1-s+p(A_{1} - A_{2})} f^{p}(x) dx \right]^{\frac{1}{p}} \left[\int_{0}^{\infty} x^{1-s+q(A_{2} - A_{1})} g^{q}(x) dx \right]^{\frac{1}{q}}, \quad (6.12)$$

where $L = B^{\frac{1}{p}}(1 - pA_2, s - 1 + pA_2)B^{\frac{1}{q}}(1 - qA_1, s - 1 + qA_1)$, and $A_1 \in (\frac{1-s}{q}, \frac{1}{q})$, $A_2 \in (\frac{1-s}{p}, \frac{1}{p})$.

Proof. The proof follows directly from Theorem 6.3 defining $p_1 = q_2 = p$, $p_2 = q_1 = q$, $B_i = A_i$, $C_i = \frac{2-s}{q_i} - A_{i+1}$, $\alpha_i = A_i - A_{i+1}$ for i = 1, 2 (the indices are taken modulo 2). \Box

In order to obtain inequality (6.1), we consider (6.12) with the parameters $A_1 = \frac{1-b}{q} - \frac{1}{pq}$ and $A_2 = \frac{1-c}{p} - \frac{1}{pq}$. Obviously, since $A_1 \in \left(\frac{1-s}{q}, \frac{1}{q}\right)$ and $A_2 \in \left(\frac{1-s}{p}, \frac{1}{p}\right)$, it follows that $0 < b + \frac{1}{p} < s$ and $0 < c + \frac{1}{q} < s$.

Corollary 6.1 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and s > 0. If $f,g : \mathbb{R}_+ \to \mathbb{R}$, $f,g \neq 0$, are non-negative measurable functions with the respective Laplace transforms $\mathscr{L}f$ and $\mathscr{L}g$, then

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{f(x)g(y)}{(x+y)^{s}} dx dy \\
\leq \frac{1}{\Gamma(s)} \| t^{\frac{1}{p}(s-c-1)+\frac{b}{q}} (\mathscr{L}f)(t) \|_{L^{p}(\mathbb{R}_{+})} \| t^{\frac{1}{q}(s-b-1)+\frac{c}{p}} (\mathscr{L}g)(t) \|_{L^{q}(\mathbb{R}_{+})} \\
< C_{1} \left[\int_{0}^{\infty} x^{(p-1)(1-b)+c-s} f^{p}(x) dx \right]^{\frac{1}{p}} \left[\int_{0}^{\infty} x^{(q-1)(1-c)+b-s} g^{q}(x) dx \right]^{\frac{1}{q}}, \quad (6.13)$$

where $C_1 = B^{\frac{1}{p}} \left(c + \frac{1}{q}, s - c - \frac{1}{q} \right) B^{\frac{1}{q}} \left(b + \frac{1}{p}, s - b - \frac{1}{p} \right).$

Remark 6.2 If s = b + c + 1, the above inequality (6.13) reduces to (6.1), that is, Theorem 6.3 can be seen as a generalization of the Peachey result from [108].

6.2 The case of non-conjugate parameters

The following result is an extension of Lemma 6.2 to the setting of non-conjugate exponents.

Lemma 6.3 Let p, p', q, q', and λ satisfy (2.1) and (2.2). If $f : \mathbb{R}_+ \to \mathbb{R}$, $f \neq 0$, is a non-negative measurable function with the Laplace transform $\mathscr{L}f$, then

$$\left[\int_{0}^{\infty} y^{\frac{q'}{p'} - q'(B+C)} \left(\mathscr{L}f \right)^{q'}(y) dy \right]^{\frac{1}{q'}} < \lambda^{\lambda - (B+C)} \Gamma^{\frac{1}{p'}} (1 - p'B) \Gamma^{\frac{1}{q'}} (1 - q'C) \left[\int_{0}^{\infty} x^{p(B+C) - \frac{p}{q'}} f^{p}(x) dx \right]^{\frac{1}{p}},$$
 (6.14)

where $B < \frac{1}{p'}$ and $C < \frac{1}{q'}$.

Proof. We utilize general Hardy-Hilbert-type inequality (2.10) with non-conjugate exponents (Theorem 2.1, Chapter 2). Similarly to the procedure used in the proof of Lemma 6.2, setting $K(x,y) = \exp(-\frac{xy}{\lambda})$, $\varphi(x) = x^B$, and $\psi(y) = y^C$ in (2.10), we obtain inequality (6.14). It follows also that the equality in (6.14) is possible only in the trivial case (see Remark 2.3, Chapter 2).

In order to obtain a non-conjugate version of Theorem 6.3, we introduce real parameters r'_i such that $p_i \le r'_i$, i = 1, ..., n, and $\sum_{i=1}^n \frac{1}{r'_i} = 1$. For example, we can define $\frac{1}{r'_i} = \frac{1}{q_i} + \frac{1-\lambda}{n}$ or $r'_i = (n - n\lambda + \lambda)p_i$, i = 1, ..., n. Now, utilizing Lemma 6.3 we obtain the following general result:

Theorem 6.5 Let s > 0 and $p_1, \ldots, p_n, \lambda$ be as in (2.26) and (2.27). Let r'_1, \ldots, r'_n and $\alpha_1, \ldots, \alpha_n$ be such that $p_i \leq r'_i$, $i = 1, \ldots, n$, $\sum_{i=1}^n \frac{1}{r'_i} = 1$, and $\sum_{i=1}^n \alpha_i = 0$. If $f_i : \mathbb{R}_+ \to \mathbb{R}$, $f_i \neq 0$, $i = 1, \ldots, n$, are non-negative measurable functions with the respective Laplace transforms $\mathcal{L}f_i$, then

$$\begin{split} \int_{\mathbb{R}^{n}_{+}} \frac{\prod_{i=1}^{n} f_{i}(x_{i})}{(\sum_{j=1}^{n} x_{j})^{s}} dx_{1} \dots dx_{n} &\leq \frac{1}{\Gamma(s)} \prod_{i=1}^{n} \left\| t^{\frac{1}{p'_{i}} - (B_{i} + C_{i})} \left(\mathscr{L}f_{i} \right)(t) \right\|_{L^{r'_{i}}(\mathbb{R}_{+})} \\ &< \frac{1}{\Gamma(s)} \prod_{i=1}^{n} \left[\gamma_{i}^{\gamma_{i} - (B_{i} + C_{i})} \Gamma^{\frac{1}{p'_{i}}} (1 - p'_{i}B_{i}) \Gamma^{\frac{1}{r'_{i}}} (1 - r'_{i}C_{i}) \right] \\ &\qquad \times \prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{p_{i}(B_{i} + C_{i}) - \frac{p_{i}}{r'_{i}}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}}, \tag{6.15}$$

where $\gamma_i = \frac{1}{p'_i} + \frac{1}{r'_i}$, $B_i < \frac{1}{p'_i}$, and $C_i < \frac{1}{r'_i}$, i = 1, ..., n.
Proof. Let β_i , i = 1, ..., n, be parameters such that $\sum_{i=1}^{n} \beta_i = s - 1$. Similarly to the proof of Theorem 6.3, applying Lemma 6.1 and the Hölder inequality, we have

$$\int_{\mathbb{R}^n_+} \frac{\prod_{i=1}^n f_i(x_i)}{(\sum_{j=1}^n x_j)^s} dx_1 \dots dx_n \le \frac{1}{\Gamma(s)} \prod_{i=1}^n \left[\int_0^\infty t^{r'_i \beta_i} \left(\mathscr{L}f_i\right)^{r'_i}(t) dt \right]^{\frac{1}{r'_i}}, \tag{6.16}$$

where $\sum_{i=1}^{n} \frac{1}{r'_{i}} = 1$ and $p_{i} \leq r'_{i}, i = 1, ..., n$.

On the other hand, setting

$$\beta_i = \frac{1}{p'_i} - (B_i + C_i), \qquad i = 1, \dots, n, \qquad (6.17)$$

where $B_i < \frac{1}{p'_i}$ and $C_i < \frac{1}{r'_i}$, and taking into account Lemma 6.3, we obtain

$$\left[\int_{0}^{\infty} t^{r'_{i}\beta_{i}} \left(\mathscr{L}f_{i}\right)^{r'_{i}}(t)dt \right]^{\frac{1}{r'_{i}}} = \left[\int_{0}^{\infty} t^{\frac{p'_{i}}{p'_{i}} - r'_{i}(B_{i}+C_{i})} \left(\mathscr{L}f_{i}\right)^{r'_{i}}(t)dt \right]^{\frac{1}{r'_{i}}}$$

$$< \gamma_{i}^{\gamma_{i}-(B_{i}+C_{i})} \Gamma^{\frac{1}{p'_{i}}}(1-p'_{i}B_{i}) \Gamma^{\frac{1}{r'_{i}}}(1-r'_{i}C_{i}) \left[\int_{0}^{\infty} x_{i}^{p_{i}(B_{i}+C_{i}) - \frac{p_{i}}{r'_{i}}} f_{i}^{p_{i}}(x_{i})dx_{i} \right]^{\frac{1}{p_{i}}}, \qquad (6.18)$$

where $\gamma_i = \frac{1}{p'_i} + \frac{1}{r'_i}$. Finally, relations (6.16) and (6.18) yield (6.15).

Setting $B_i + C_i = \frac{n-s+1}{r'_i} - \frac{1}{p_i} + \alpha_i$ in (6.17), we have $\sum_{i=1}^n \beta_i = s - 1$, that is, inequality (6.18) reduces to

$$\|t^{\frac{s-n-1}{r_{i}'}+1-\alpha_{i}}(\mathscr{L}f_{i})(t)\|_{L^{r_{i}'}(\mathbb{R}_{+})} < \gamma_{i}^{\frac{s-n}{r_{i}'}+1-\alpha_{i}}\Gamma^{\frac{1}{p_{i}'}}(1-p_{i}'B_{i})\Gamma^{\frac{1}{r_{i}'}}(1-r_{i}'C_{i}) \\ \times \left[\int_{0}^{\infty} x_{i}^{\frac{p_{i}}{r_{i}'}(n-s)-1+p_{i}\alpha_{i}}f_{i}^{p_{i}}(x_{i})dx_{i}\right]^{\frac{1}{p_{i}'}}.$$
 (6.19)

Now, exploiting (6.19) we obtain the following result:

Corollary 6.2 Let s > 0 and $p_1, \ldots, p_n, \lambda$ be as in (2.26) and (2.27). Let r'_1, \ldots, r'_n , $\alpha_1, \ldots, \alpha_n$ satisfy conditions of Theorem 6.5. If $f_i : \mathbb{R}_+ \to \mathbb{R}$, $f_i \neq 0$, $i = 1, \ldots, n$, are non-negative measurable functions with the respective Laplace transforms $\mathscr{L}f_i$, then

$$\int_{\mathbb{R}^{n}_{+}} \frac{\prod_{i=1}^{n} f_{i}(x_{i})}{(\sum_{j=1}^{n} x_{j})^{s}} dx_{1} \dots dx_{n} \leq \frac{1}{\Gamma(s)} \prod_{i=1}^{n} \left\| t^{\frac{s-n-1}{r'_{i}}+1-\alpha_{i}} (\mathscr{L}f_{i})(t) \right\|_{L'_{i}(\mathbb{R}_{+})} \\
< \frac{1}{\Gamma(s)} \prod_{i=1}^{n} \left[\gamma_{i}^{\frac{s-n}{r'_{i}}+1-\alpha_{i}} \Gamma^{\frac{1}{p'_{i}}} (1-p'_{i}B_{i}) \Gamma^{\frac{1}{r'_{i}}} (1-r'_{i}C_{i}) \right] \\
\times \prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{\frac{p_{i}}{r'_{i}}(n-s)-1+p_{i}\alpha_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}}},$$
(6.20)

where $\gamma_i = \frac{1}{p'_i} + \frac{1}{r'_i}$, $B_i + C_i = \frac{n-s+1}{r'_i} - \frac{1}{p_i} + \alpha_i$, $B_i < \frac{1}{p'_i}$, and $C_i < \frac{1}{r'_i}$, $i = 1, \dots, n$.

Remark 6.3 Setting $\frac{1}{r'_i} = \frac{1}{q_i} + \frac{1-\lambda}{n}$, $\alpha_i = \frac{1}{p_i} - \frac{1}{r'_i}(n-s)$, $B_i = \frac{1}{r'_i + p'_i}$, and $C_i = \frac{p'_i}{r'_i(r'_i + p'_i)}$ in (6.20), we get the following inequalities:

$$\begin{split} &\int_{\mathbb{R}^{n}_{+}} \frac{\prod_{i=1}^{n} f_{i}(x_{i})}{(\sum_{j=1}^{n} x_{j})^{(n-1)\lambda}} dx_{1} \dots dx_{n} \\ &\leq \frac{1}{\Gamma((n-1)\lambda)} \prod_{i=1}^{n} \|t^{\frac{2}{p_{i}^{\prime}} - \frac{(n-1)\lambda+1}{n}} (\mathscr{L}f_{i})(t)\|_{L^{\frac{nq_{i}}{n+q_{i}(1-\lambda)}}(\mathbb{R}_{+})} \\ &< \frac{1}{\Gamma((n-1)\lambda)} \left[\lambda + \frac{1-\lambda}{n}\right]^{(n-1)\lambda} \left[\prod_{i=1}^{n} \Gamma\left(\frac{n}{(n-1)\lambda+1} \cdot \frac{1}{p_{i}^{\prime}}\right)\right]^{\lambda + \frac{1-\lambda}{n}} \prod_{i=1}^{n} \|f_{i}\|_{L^{p_{i}}(\mathbb{R}_{+})}. \end{split}$$

Obviously, according to the conditions from the statement of Corollary 6.2, we can use other choices of parameters r'_i , and consequently, take the infimum of the right-hand sides of the obtained inequalities over all such parameters r'_i .

We restate Corollary 6.2 for the case n = 2. This result is interesting in its own right, and it will be applied to obtain Theorem 6.2.

Corollary 6.3 Suppose p, p', q, q', and λ satisfy (2.1) and (2.2), and let s > 0. Further, let $p \le r' \le q'$, $\frac{1}{r'} + \frac{1}{r} = 1$, and $\alpha_1 + \alpha_2 = 0$. If $f, g : \mathbb{R}_+ \to \mathbb{R}$, $f, g \ne 0$, are non-negative measurable functions with the respective Laplace transforms $\mathscr{L}f$ and $\mathscr{L}g$, then

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{f(x)g(y)}{(x+y)^{s}} dx dy
\leq \frac{1}{\Gamma(s)} \|t^{\frac{s-3}{r'}+1-\alpha_{1}}(\mathscr{L}f)(t)\|_{L^{r'}(\mathbb{R}_{+})} \|t^{\frac{s-3}{r}+1-\alpha_{2}}(\mathscr{L}g)(t)\|_{L^{r}(\mathbb{R}_{+})}
< \frac{M}{\Gamma(s)} \|x^{\frac{2-s}{r'}-\frac{1}{p}+\alpha_{1}}f(x)\|_{L^{p}(\mathbb{R}_{+})} \|y^{\frac{2-s}{r}-\frac{1}{q}+\alpha_{2}}g(y)\|_{L^{q}(\mathbb{R}_{+})},$$
(6.21)

where

$$M = \gamma_1^{\frac{s-2}{r'}+1-\alpha_1} \gamma_2^{\frac{s-2}{r}+1-\alpha_2} \Gamma^{\frac{1}{p'}} (1-p'B_1) \Gamma^{\frac{1}{r'}} (1-r'C_1) \Gamma^{\frac{1}{q'}} (1-q'B_2) \Gamma^{\frac{1}{r}} (1-rC_2),$$

$$\gamma_1 = \frac{1}{p'} + \frac{1}{r'}, \ \gamma_2 = \frac{1}{q'} + \frac{1}{r}, \ B_1 + C_1 = \frac{3-s}{r'} - \frac{1}{p} + \alpha_1, \ B_2 + C_2 = \frac{3-s}{r} - \frac{1}{q} + \alpha_2, \ B_1 < \frac{1}{p'}, \ C_1 < \frac{1}{r'},$$

$$B_2 < \frac{1}{q'}, \ and \ C_2 < \frac{1}{r}.$$

Proof. The proof follows directly from Corollary 6.2 using substitutions $p_1 = p$, $p_2 = q$, $p'_1 = q_2 = p'$, $p'_2 = q_1 = q'$, $r'_1 = r'$, and $r'_2 = r$.

Remark 6.4 Setting s = b + c + 1, $\alpha_1 = \frac{c-2}{r'} + 1 - \frac{b}{r}$, $\alpha_2 = -\alpha_1$, $B_1 = \frac{1}{p'} - \frac{1}{p'\gamma_1}(b + \frac{1}{r'})$, $C_1 = \frac{1}{r'} - \frac{1}{r'\gamma_1}(b + \frac{1}{r'})$, $B_2 = \frac{1}{q'} - \frac{1}{q'\gamma_2}(c + \frac{1}{r})$, and $C_2 = \frac{1}{r} - \frac{1}{r\gamma_2}(c + \frac{1}{r})$ in Corollary 6.3, we obtain Theorem 6.2.

170

Remark 6.5 Set $s = \lambda$, $\alpha_1 = \frac{\lambda - 2}{r'} + \frac{1}{p}$, $\alpha_2 = -\alpha_1$, $B_1 = \frac{1}{r'p'}$, $C_1 = \frac{1}{r'p}$, $B_2 = \frac{1}{rq'}$, and $C_2 = \frac{1}{rq}$, $p \le r' \le q'$, in Corollary 6.3. Then inequality (6.21) becomes

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{f(x)g(y)}{(x+y)^{\lambda}} dx dy \leq \frac{1}{\Gamma(\lambda)} \| t^{\frac{1}{p'} - \frac{1}{r'}} \left(\mathscr{L}f \right)(t) \|_{L^{r'}(\mathbb{R}_{+})} \| t^{\frac{1}{q'} - \frac{1}{r}} \left(\mathscr{L}g \right)(t) \|_{L^{r}(\mathbb{R}_{+})} \\
< \frac{M}{\Gamma(\lambda)} \| f \|_{L^{p}(\mathbb{R}_{+})} \| g \|_{L^{q}(\mathbb{R}_{+})},$$
(6.22)

where $\gamma_1 = \frac{1}{p'} + \frac{1}{r'}$, $\gamma_2 = \frac{1}{q'} + \frac{1}{r}$, and $M = \gamma_1^{\frac{1}{p'}} \gamma_2^{\frac{1}{q'}} \Gamma^{\frac{1}{p'}} (\frac{1}{r}) \Gamma^{\frac{1}{r'}} (\frac{1}{p'}) \Gamma^{\frac{1}{r'}} (\frac{1}{r'}) \Gamma^{\frac{1}{r}} (\frac{1}{q'})$. It should be noticed here that in conjugate case, the constant $\frac{M}{\Gamma(\lambda)}$ reduces to the best constant $\pi/\sin(\pi/p)$.

Remark 6.6 Similarly to Corollary 6.2, if we put $B_i + C_i = \frac{1}{r'_i} + \frac{1}{q_i}(n-1-\frac{s}{\lambda}) + \alpha_i$, i = 1, ..., n, in (6.17), then $\sum_{i=1}^n \beta_i = s - 1$. Thus, inequality (6.15) from Theorem 6.5 becomes

$$\begin{split} \int_{\mathbb{R}^{n}_{+}} \frac{\prod_{i=1}^{n} f_{i}(x_{i})}{(\sum_{j=1}^{n} x_{j})^{s}} dx_{1} \dots dx_{n} &\leq \frac{1}{\Gamma(s)} \prod_{i=1}^{n} \| t^{\lambda - \frac{1}{r_{i}^{\prime}} - \frac{1}{q_{i}}(n - \frac{s}{\lambda}) - \alpha_{i}} (\mathscr{L}f_{i})(t) \|_{L^{\prime}(\mathbb{R}_{+})} \\ &< \frac{1}{\Gamma(s)} \prod_{i=1}^{n} \left[\gamma_{i}^{\lambda - \frac{1}{q_{i}}(n - \frac{s}{\lambda}) - \alpha_{i}} \Gamma^{\frac{1}{p_{i}^{\prime}}} (1 - p_{i}^{\prime}B_{i}) \Gamma^{\frac{1}{r_{i}^{\prime}}} (1 - r_{i}^{\prime}C_{i}) \right] \\ &\qquad \times \prod_{i=1}^{n} \left[\int_{0}^{\infty} x_{i}^{\frac{p_{i}}{q_{i}}(n - 1 - \frac{s}{\lambda}) + p_{i}\alpha_{i}} f_{i}^{p_{i}}(x_{i}) dx_{i} \right]^{\frac{1}{p_{i}^{\prime}}}, \end{split}$$
(6.23)

where γ_i , r'_i , α_i are as in Corollary 6.2. Observe that the integrals on the right-hand side of the second inequality in (6.23) coincide with integrals on the right-hand side of inequality (2.35) (see Chapter 2), when *s* is replaced by $\frac{s}{\lambda}$.

Remark 6.7 Multidimensional refinements of Hilbert-type inequalities via the Laplace transform, presented in this chapter, are derived in [110] by Pečarić et.al.

Chapter 7

A class of Hilbert-Pachpatte-type inequalities

In this chapter we investigate a particular class of the so-called Hilbert-Pachpatte-type inequalities which are closely connected to Hilbert-type inequalities. An interesting feature of this class is that it controls the size (in the sense of L^p or l^p spaces) of the modified Hilbert transform of a function or of a series with the size of its derivative or its backward differences, respectively.

We start this overview with the results of Lü [86], in a slightly altered form, in both continuous and discrete case. For a sequence $(a_n)_{n \in \mathbb{N}_0}$, the sequence $(\nabla a_n)_{n \in \mathbb{N}}$ is defined by $\nabla a_n = a_n - a_{n-1}$, while for a function $u : \mathbb{R}_+ \to \mathbb{R}$, u' denotes its usual derivative.

Theorem 7.1 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $s > 2 - \min\{p,q\}$. If $f,g : \mathbb{R}_+ \to \mathbb{R}$ are absolutely continuous functions such that f(0) = g(0) = 0, then

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{|f(x)||g(y)|}{(qx^{p-1} + py^{q-1})(x+y)^{s}} dx dy$$

$$\leq \frac{B\left(\frac{q+s-2}{q}, \frac{p+s-2}{p}\right)}{pq} \left[\int_{0}^{\infty} \int_{0}^{x} x^{1-s} |f'(\tau)|^{p} d\tau dx \right]^{\frac{1}{p}} \left[\int_{0}^{\infty} \int_{0}^{y} y^{1-s} |g'(\delta)|^{q} d\delta dy \right]^{\frac{1}{q}}.$$
(7.1)

Theorem 7.2 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $2 - \min\{p,q\} < s \le 2$. If $(a_m)_{m \in \mathbb{N}_0}$, $(b_n)_{n \in \mathbb{N}_0}$ are sequences of real numbers such that $a_0 = b_0 = 0$, then the following inequality hold:

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \frac{|a_m| |b_n|}{(qm^{p-1} + pn^{q-1})(m+n)^s} \le \frac{B\left(\frac{q+s-2}{q}, \frac{p+s-2}{p}\right)}{pq} \left[\sum_{m=1}^{\infty} \sum_{\tau=1}^{m} m^{1-s} |\nabla a_{\tau}|^p\right]^{\frac{1}{p}} \left[\sum_{n=1}^{\infty} \sum_{\delta=1}^{n} n^{1-s} |\nabla b_{\delta}|^q\right]^{\frac{1}{q}}.$$
 (7.2)

Due to the form of the kernel, the above Hilbert-Pachpatte-type inequalities include the constant factors expressed in terms of the usual Beta function. Our main task here is to obtain generalizations of the above inequalities which include arbitrary kernels and weight functions, with a special emphasis on homogeneous kernels. This can be done in a simpler way than in [86], by virtue of the general Hilbert-type and Hardy-Hilbert-type inequalities from Chapters 1 and 2. Hence, generalizations that follow will be given in both Hilbert and Hardy-Hilbert form. Moreover, the established results will also be considered in the setting of non-conjugate exponents.

7.1 Integral case

We start with the following general result.

Theorem 7.3 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$, $\varphi, \psi : \mathbb{R}_+ \to \mathbb{R}$ be nonnegative functions. If $f, g : \mathbb{R}_+ \to \mathbb{R}$ are absolutely continuous functions such that f(0) = g(0) = 0, and $F(x) = \int_0^\infty K(x,y)\psi^{-p}(y)dy$, $G(y) = \int_0^\infty K(x,y)\varphi^{-q}(x)dx$, then the following inequalities hold:

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{K(x,y)|f(x)||g(y)|}{qx^{p-1} + py^{q-1}} dx dy \\
\leq \int_{0}^{\infty} \int_{0}^{\infty} K(x,y)|f(x)||g(y)|d(x^{\frac{1}{p}})d(y^{\frac{1}{q}}) \\
\leq \frac{1}{pq} \left[\int_{0}^{\infty} \int_{0}^{x} \varphi^{p}(x)F(x)|f'(\tau)|^{p} d\tau dx \right]^{\frac{1}{p}} \left[\int_{0}^{\infty} \int_{0}^{y} \psi^{q}(y)G(y)|g'(\delta)|^{q} d\delta dy \right]^{\frac{1}{q}} (7.3)$$

and

$$\int_0^\infty G^{1-p}(y)\psi^{-p}(y)\left[\int_0^\infty K(x,y)|f(x)|d(x^{\frac{1}{p}})\right]^p dy$$

$$\leq \frac{1}{p^p}\int_0^\infty \int_0^x \varphi^p(x)F(x)|f'(\tau)|^p d\tau dx.$$
(7.4)

Proof. Utilizing the Hölder inequality (see also [86]), we have

$$|f(x)||g(y)| \le x^{\frac{1}{q}} y^{\frac{1}{p}} \left[\int_0^x |f'(\tau)|^p d\tau \right]^{\frac{1}{p}} \left[\int_0^y |g'(\delta)|^q d\delta \right]^{\frac{1}{q}}.$$
 (7.5)

Taking into account (7.5) and the well-known Young inequality

$$xy \le \frac{x^p}{p} + \frac{y^q}{q}, \ x \ge 0, \ y \ge 0, \ \frac{1}{p} + \frac{1}{q} = 1, \ p > 1,$$

we observe that

$$\frac{pq|f(x)||g(y)|}{qx^{p-1} + py^{q-1}} \le \frac{|f(x)||g(y)|}{x^{\frac{1}{q}}y^{\frac{1}{p}}} \le \left[\int_0^x |f'(\tau)|^p d\tau\right]^{\frac{1}{p}} \left[\int_0^y |g'(\delta)|^q d\delta\right]^{\frac{1}{q}}$$

and therefore,

$$pq \int_{0}^{\infty} \int_{0}^{\infty} \frac{K(x,y)|f(x)||g(y)|}{qx^{p-1} + py^{q-1}} dx dy$$

$$\leq \int_{0}^{\infty} \int_{0}^{\infty} \frac{K(x,y)}{x^{\frac{1}{q}}y^{\frac{1}{p}}} |f(x)||g(y)| dx dy$$

$$\leq \int_{0}^{\infty} \int_{0}^{\infty} K(x,y) \left[\int_{0}^{x} |f'(\tau)|^{p} d\tau \right]^{\frac{1}{p}} \left[\int_{0}^{y} |g'(\delta)|^{q} d\delta \right]^{\frac{1}{q}} dx dy.$$
(7.6)

Now, putting

$$f_1(x) = \left[\int_0^x |f'(\tau)|^p d\tau\right]^{\frac{1}{p}}, \qquad g_1(y) = \left[\int_0^y |g'(\delta)|^q d\delta\right]^{\frac{1}{q}}$$

in the general Hilbert-type inequality (1.12) in the conjugate case (Theorem 1.9, Chapter 1), we obtain

$$\int_0^\infty \int_0^\infty K(x,y) f_1(x) g_1(y) dx dy$$

$$\leq \left[\int_0^\infty \varphi^p(x) F(x) f_1^p(x) dx \right]^{\frac{1}{p}} \left[\int_0^\infty \psi^q(y) G(y) g_1^q(y) dy \right]^{\frac{1}{q}}$$

$$= \left[\int_0^\infty \int_0^x \varphi^p(x) F(x) |f'(\tau)|^p d\tau dx \right]^{\frac{1}{p}} \left[\int_0^\infty \int_0^y \psi^q(y) G(y) |g'(\delta)|^q d\delta dy \right]^{\frac{1}{q}}.$$
(7.7)

Finally, using (7.6) and (7.7) we obtain (7.3).

The second inequality (7.4) follows from the general Hardy-Hilbert-type inequality (1.13) in the conjugate case (Theorem 1.9, Chapter 1), and the inequality $|f(x)| \le x^{\frac{1}{q}} (\int_0^x |f'(t)|^p dt)^{\frac{1}{p}}$.

In the sequel we apply the above Theorem 7.3 to a homogeneous kernel. Similarly to the previous chapters we use the notation $k(\alpha) = \int_0^\infty K(1,u)u^{-\alpha}du$, where $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ is a non-negative homogeneous function of degree -s, s > 0, provided $k(\alpha) < \infty$ for $1-s < \alpha < 1$.

Corollary 7.1 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a non-negative symmetric homogeneous function of degree -s, s > 0. If $f, g : \mathbb{R}_+ \to \mathbb{R}$ are absolutely continuous functions such that f(0) = g(0) = 0, then

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{K(x,y)|f(x)||g(y)|}{qx^{p-1} + py^{q-1}} dx dy \leq \int_{0}^{\infty} \int_{0}^{\infty} K(x,y)|f(x)||g(y)|d(x^{\frac{1}{p}})d(y^{\frac{1}{q}})$$
$$\leq \frac{L}{pq} \left[\int_{0}^{\infty} \int_{0}^{x} x^{1-s+p(A_{1}-A_{2})} |f'(\tau)|^{p} d\tau dx \right]^{\frac{1}{p}}$$
$$\times \left[\int_{0}^{\infty} \int_{0}^{y} y^{1-s+q(A_{2}-A_{1})} |g'(\delta)|^{q} d\delta dy \right]^{\frac{1}{q}}$$
(7.8)

and

$$\int_{0}^{\infty} y^{(p-1)(s-1)+p(A_{1}-A_{2})} \left[\int_{0}^{\infty} K(x,y) |f(x)| d(x^{\frac{1}{p}}) \right]^{p} dy$$

$$\leq \left(\frac{L}{p}\right)^{p} \int_{0}^{\infty} \int_{0}^{x} x^{1-s+p(A_{1}-A_{2})} |f'(\tau)|^{p} d\tau dx,$$
(7.9)

where $A_1 \in (\frac{1-s}{q}, \frac{1}{q}), A_2 \in (\frac{1-s}{p}, \frac{1}{p})$, and $L = k^{\frac{1}{p}}(pA_2)k^{\frac{1}{q}}(qA_1)$.

Proof. We prove (7.8) only. Let F(x) and G(y) be as in the statement of Theorem 7.3. Setting $\varphi(x) = x^{A_1}$ and $\psi(y) = y^{A_2}$, we obtain

$$\int_{0}^{\infty} \int_{0}^{x} \varphi^{p}(x) F(x) |f'(\tau)|^{p} d\tau dx$$

= $\int_{0}^{\infty} \int_{0}^{x} |f'(\tau)|^{p} \left[\int_{0}^{\infty} K(x,y) \left(\frac{x}{y}\right)^{pA_{2}} dy \right] x^{p(A_{1}-A_{2})} d\tau dx$
= $k(pA_{2}) \int_{0}^{\infty} \int_{0}^{x} x^{1-s+p(A_{1}-A_{2})} |f'(\tau)|^{p} d\tau dx,$ (7.10)

and similarly,

$$\int_0^\infty \int_0^y \psi^q(y) G(y) |g'(\delta)|^q d\delta dy = k(qA_1) \int_0^\infty \int_0^y y^{1-s+q(A_2-A_1)} |g'(\delta)|^q d\delta dy.$$
(7.11)

Now, relations (7.3), (7.10), and (7.11) yield (7.8).

Our first application of Corollary 7.1 refers to the homogeneous kernel $K(x,y) = (x + y)^{-s}$, s > 0.

Corollary 7.2 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let s > 0. If $f, g : \mathbb{R}_+ \to \mathbb{R}$ are absolutely

continuous functions such that f(0) = g(0) = 0, then

$$\begin{split} \int_{0}^{\infty} \int_{0}^{\infty} \frac{|f(x)||g(y)|}{(qx^{p-1} + py^{q-1})(x+y)^{s}} dx dy &\leq \int_{0}^{\infty} \int_{0}^{\infty} \frac{|f(x)||g(y)|}{(x+y)^{s}} d(x^{\frac{1}{p}}) d(y^{\frac{1}{q}}) \\ &\leq \frac{L_{1}}{pq} \left[\int_{0}^{\infty} \int_{0}^{x} x^{1-s+p(A_{1}-A_{2})} |f'(\tau)|^{p} d\tau dx \right]^{\frac{1}{p}} \\ &\times \left[\int_{0}^{\infty} \int_{0}^{y} y^{1-s+q(A_{2}-A_{1})} |g'(\delta)|^{q} d\delta dy \right]^{\frac{1}{q}} \end{split}$$

and

$$\int_{0}^{\infty} y^{(p-1)(s-1)+p(A_{1}-A_{2})} \left[\int_{0}^{\infty} \frac{|f(x)|}{(x+y)^{s}} d(x^{\frac{1}{p}}) \right]^{p} dy$$

$$\leq \left(\frac{L_{1}}{p} \right)^{p} \int_{0}^{\infty} \int_{0}^{x} x^{1-s+p(A_{1}-A_{2})} |f'(\tau)|^{p} d\tau dx,$$

where $A_1 \in (\frac{1-s}{q}, \frac{1}{q}), A_2 \in (\frac{1-s}{p}, \frac{1}{p})$, and $L_1 = B^{\frac{1}{p}}(1 - pA_2, s - 1 + pA_2)B^{\frac{1}{q}}(1 - qA_1, s - 1 + qA_1)$.

Remark 7.1 Setting $A_1 = A_2 = \frac{2-s}{pq}$ in Corollary 7.2, provided $s > 2 - \min\{p,q\}$, we obtain Theorem 7.1.

Considering the homogeneous kernel $K(x,y) = \frac{\log \frac{y}{x}}{y-x}$ of degree -1, we obtain the following consequence of Corollary 7.1:

Corollary 7.3 Let $\frac{1}{p} + \frac{1}{q} = 1$ and p > 1. If $f, g : \mathbb{R}_+ \to \mathbb{R}$ are absolutely continuous functions such that f(0) = g(0) = 0, then

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{\log \frac{y}{x} |f(x)| |g(y)|}{(qx^{p-1} + py^{q-1})(y-x)} dx dy \leq \int_{0}^{\infty} \int_{0}^{\infty} \frac{\log \frac{y}{x} |f(x)| |g(y)|}{y-x} d(x^{\frac{1}{p}}) d(y^{\frac{1}{q}})$$
$$\leq \frac{L_{2}}{pq} \left[\int_{0}^{\infty} \int_{0}^{x} x^{p(A_{1}-A_{2})} |f'(\tau)|^{p} d\tau dx \right]^{\frac{1}{p}} \left[\int_{0}^{\infty} \int_{0}^{y} y^{q(A_{2}-A_{1})} |g'(\delta)|^{q} d\delta dy \right]^{\frac{1}{q}}$$

and

$$\int_0^\infty y^{p(A_1-A_2)} \left[\int_0^\infty \frac{|f(x)|\log\frac{y}{x}}{y-x} d(x^{\frac{1}{p}}) \right]^p dy$$

$$\leq \left(\frac{L_2}{p}\right)^p \int_0^\infty \int_0^x x^{p(A_1-A_2)} |f'(\tau)|^p d\tau dx,$$

where $A_1 \in (0, \frac{1}{q}), A_2 \in (0, \frac{1}{p})$, and $L_2 = \pi^2 (\sin p A_2 \pi)^{-\frac{2}{p}} (\sin q A_1 \pi)^{-\frac{2}{q}}$.

We conclude the above discussion regarding symmetric homogeneous kernels with the function $K(x,y) = \max\{x, y\}^{-s}$, s > 0.

Corollary 7.4 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let s > 0. If $f,g : \mathbb{R}_+ \to \mathbb{R}$ are absolutely continuous functions such that f(0) = g(0) = 0, then

$$\begin{split} \int_{0}^{\infty} \int_{0}^{\infty} \frac{|f(x)||g(y)|}{(qx^{p-1} + py^{q-1}) \max\{x, y\}^{s}} dx dy &\leq \int_{0}^{\infty} \int_{0}^{\infty} \frac{|f(x)||g(y)|}{\max\{x, y\}^{s}} d(x^{\frac{1}{p}}) d(y^{\frac{1}{q}}) \\ &\leq \frac{L_{3}}{pq} \left[\int_{0}^{\infty} \int_{0}^{x} x^{1-s+p(A_{1}-A_{2})} |f'(\tau)|^{p} d\tau dx \right]^{\frac{1}{p}} \\ &\times \left[\int_{0}^{\infty} \int_{0}^{y} y^{1-s+q(A_{2}-A_{1})} |g'(\delta)|^{q} d\delta dy \right]^{\frac{1}{q}} \end{split}$$

and

$$\int_{0}^{\infty} y^{(p-1)(s-1)+p(A_{1}-A_{2})} \left[\int_{0}^{\infty} \frac{|f(x)|}{\max\{x,y\}^{s}} d(x^{\frac{1}{p}}) \right]^{p} dy$$

$$\leq \left(\frac{L_{3}}{p}\right)^{p} \int_{0}^{\infty} \int_{0}^{x} x^{1-s+p(A_{1}-A_{2})} |f'(\tau)|^{p} d\tau dx,$$

where $A_1 \in (\frac{1-s}{q}, \frac{1}{q}), A_2 \in (\frac{1-s}{p}, \frac{1}{p})$, and

$$L_{3} = \frac{1}{(1 - pA_{2})^{\frac{1}{p}}(1 - qA_{1})^{\frac{1}{q}}(s + pA_{2} - 1)^{\frac{1}{p}}(s + qA_{1} - 1)^{\frac{1}{q}}}$$

By virtue of general multidimensional Hilbert-type inequalities in conjugate setting (Theorem 1.18, Chapter 1), Theorem 7.3 can also be extended to the case of n conjugate exponents. We provide here the multidimensional version of inequality (7.3).

Theorem 7.4 Let $\sum_{i=1}^{n} \frac{1}{p_i} = 1$, $p_i > 1$, and let $\alpha_i = \prod_{j=1, j \neq i}^{n} p_j$, i = 1, 2, ..., n. If $K : \mathbb{R}^n_+ \to \mathbb{R}$, $\phi_{ij} : \mathbb{R}_+ \to \mathbb{R}$, i, j = 1, ..., n, are non-negative functions such that $\prod_{i,j=1}^{n} \phi_{ij}(x_j) = 1$, and $f_i : \mathbb{R}_+ \to \mathbb{R}$, i = 1, ..., n, are absolutely continuous functions such that $f_i(0) = 0$, i = 1, ..., n, then

$$\begin{split} &\int_{\mathbb{R}^{n}_{+}} \frac{K(x_{1},\ldots,x_{n})\prod_{i=1}^{n}|f_{i}(x_{i})|}{\sum_{i=1}^{n}\alpha_{i}x_{i}^{p_{i}-1}}dx_{1}\ldots dx_{n} \\ &\leq \int_{\mathbb{R}^{n}_{+}} K(x_{1},\ldots,x_{n})\prod_{i=1}^{n}|f_{i}(x_{i})|d(x_{1}^{\frac{1}{p_{1}}})\ldots d(x_{n}^{\frac{1}{p_{n}}}) \\ &\leq \frac{1}{p_{1}\ldots p_{n}}\prod_{i=1}^{n}\left[\int_{0}^{\infty}\int_{0}^{x_{i}}\phi_{ii}^{p_{i}}(x_{i})F_{i}(x_{i})|f_{i}^{'}(\tau_{i})|^{p_{i}}d\tau_{i}dx_{i}\right]^{\frac{1}{p_{i}}}, \end{split}$$

where $F_i(x_i) = \int_{\mathbb{R}^{n-1}_+} K(x_1, \dots, x_n) \prod_{j=1, j \neq i}^n \phi_{ij}^{p_i}(x_j) dx_1 \dots dx_{i-1} dx_{i+1} \dots dx_n$ for $i = 1, \dots, n$.

7.2 Discrete case

As we have already seen, Hilbert-Pachpatte-type inequalities can also be considered in the discrete setting. We first obtain a general result covering homogeneous kernels. It is established by virtue of discrete Hilbert-type inequalities for homogeneous kernels, that is, Theorem 1.14 (Chapter 1).

Theorem 7.5 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $A, B, \alpha, \beta > 0$. If $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ is a nonnegative homogeneous function of degree -s, s > 0, strictly decreasing in both arguments and $(a_m)_{m \in \mathbb{N}_0}$, $(b_n)_{n \in \mathbb{N}_0}$ are sequences of real numbers such that $a_0 = b_0 = 0$, then

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \frac{K(Am^{\alpha}, Bn^{\beta}) |a_{m}| |b_{n}|}{qm^{p-1} + pn^{q-1}} \leq \frac{1}{pq} \sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \frac{K(Am^{\alpha}, Bn^{\beta}) |a_{m}| |b_{n}|}{m^{\frac{1}{q}} n^{\frac{1}{p}}}$$
$$\leq \frac{N}{pq} \left[\sum_{m=1}^{\infty} \sum_{\tau=1}^{m} m^{\alpha(1-s) + \alpha p(A_{1}-A_{2}) + (p-1)(1-\alpha)} |\nabla a_{\tau}|^{p} \right]^{\frac{1}{p}}$$
$$\times \left[\sum_{n=1}^{\infty} \sum_{\delta=1}^{n} n^{\beta(1-s) + \beta q(A_{2}-A_{1}) + (q-1)(1-\beta)} |\nabla b_{\delta}|^{q} \right]^{\frac{1}{q}}$$
(7.12)

and

$$\sum_{n=1}^{\infty} n^{\beta(s-1)(p-1)+p\beta(A_1-A_2)+\beta-1} \left[\sum_{m=1}^{\infty} K(Am^{\alpha}, Bn^{\beta}) \frac{|a_m|}{m^{\frac{1}{q}}} \right]^p \\ \leq N^p \sum_{m=1}^{\infty} \sum_{\tau=1}^m m^{\alpha(1-s)+\alpha p(A_1-A_2)+(p-1)(1-\alpha)} |\nabla a_{\tau}|^p,$$
(7.13)

where $A_1 \in (\max\{\frac{1-s}{q}, \frac{\alpha-1}{\alpha q}\}, \frac{1}{q}), A_2 \in (\max\{\frac{1-s}{p}, \frac{\beta-1}{\beta p}\}, \frac{1}{p}), and$ $N = \alpha^{-\frac{1}{q}} \beta^{-\frac{1}{p}} A^{\frac{2-s}{p}+A_1-A_2-1} B^{\frac{2-s}{q}+A_2-A_1-1} k^{\frac{1}{p}} (pA_2) k^{\frac{1}{q}} (2-s-qA_1).$

Proof. Similarly to the proof of Theorem 7.3, utilizing the Hölder and the Young inequalities, we have

$$\frac{pq|a_m||b_n|}{qm^{p-1} + pn^{q-1}} \le \frac{|a_m||b_n|}{m^{\frac{1}{q}}n^{\frac{1}{p}}} \le \left[\sum_{\tau=1}^m |\nabla a_\tau|^p\right]^{\frac{1}{p}} \left[\sum_{\delta=1}^n |\nabla b_\delta|^q\right]^{\frac{1}{q}},$$

and therefore,

$$pq\sum_{m=1}^{\infty}\sum_{n=1}^{\infty}\frac{K(Am^{\alpha},Bn^{\beta})|a_{m}||b_{n}|}{qm^{p-1}+pn^{q-1}} \leq \sum_{m=1}^{\infty}\sum_{n=1}^{\infty}\frac{K(Am^{\alpha},Bn^{\beta})|a_{m}||b_{n}|}{m^{\frac{1}{q}}n^{\frac{1}{p}}}$$
$$\leq \sum_{m=1}^{\infty}\sum_{n=1}^{\infty}K(Am^{\alpha},Bn^{\beta})\left[\sum_{\tau=1}^{m}|\nabla a_{\tau}|^{p}\right]^{\frac{1}{p}}\left[\sum_{\delta=1}^{n}|\nabla b_{\delta}|^{q}\right]^{\frac{1}{q}}.$$
(7.14)

Now, setting $\widetilde{a}_m = (\sum_{\tau=1}^m |\nabla a_\tau|^p)^{\frac{1}{p}}$, $\widetilde{b}_n = (\sum_{\delta=1}^n |\nabla b_\delta|^q)^{\frac{1}{q}}$, $u(m) = Am^{\alpha}$, and $v(n) = Bn^{\beta}$ in (1.42) (Theorem 1.14, Chapter 1), we have

$$\sum_{m=1}^{n} \sum_{n=1}^{\infty} K(Am^{\alpha}, Bn^{\beta}) \widetilde{a}_{m} \widetilde{b}_{n}$$

$$\leq N \left[\sum_{m=1}^{\infty} m^{\alpha(1-s) + \alpha p(A_{1}-A_{2}) + (p-1)(1-\alpha)} \widetilde{a}_{m}^{p} \right]^{\frac{1}{p}} \left[\sum_{n=1}^{\infty} n^{\beta(1-s) + \beta q(A_{2}-A_{1}) + (q-1)(1-\beta)} \widetilde{b}_{n}^{q} \right]^{\frac{1}{q}}$$

$$= N \left[\sum_{m=1}^{\infty} \sum_{\tau=1}^{m} m^{\alpha(1-s) + \alpha p(A_{1}-A_{2}) + (p-1)(1-\alpha)} |\nabla a(\tau)|^{p} \right]^{\frac{1}{p}}$$

$$\times \left[\sum_{n=1}^{\infty} \sum_{\delta=1}^{n} n^{\beta(1-s) + \beta q(A_{2}-A_{1}) + (q-1)(1-\beta)} |\nabla b(\delta)|^{q} \right]^{\frac{1}{q}}.$$
(7.15)

Finally, using (7.14) and (7.15), we obtain (7.12).

The second inequality (7.13) follows from (1.43) (Theorem 1.14, Chapter 1) and using $|a_m| \leq m^{\frac{1}{q}} (\sum_{\tau=1}^m |\nabla a_{\tau}|^p)^{\frac{1}{p}}$.

Remark 7.2 The above Theorem 7.5 can be regarded as a generalization of Theorem 7.2. Namely, applying Theorem 7.5 to $K(x,y) = (x+y)^{-s}$, s > 0, $A = B = \alpha = \beta = 1$, and $A_1 = A_2 = \frac{2-s}{pq}$, assuming $2 - \min\{p,q\} < s \le 2$, we obtain Theorem 7.2.

We conclude this section with an interesting extension of Theorem 7.2. Namely, by virtue of Corollary 4.7 (Chapter 4), we can obtain a larger interval of admissible values of the parameter *s*. Recall that this corollary was established by means of the Euler-Maclaurin summation formula.

Now, following the same lines as in the proof of Theorem 7.5 and utilizing the above mentioned Corollary 4.7 instead of Theorem 1.14, we obtain the following extension of Theorem 7.2.

Corollary 7.5 Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $2 - \min\{p,q\} < s \le 2 + \min\{p,q\}$. If $(a_m)_{m \in \mathbb{N}_0}$, $(b_n)_{n \in \mathbb{N}_0}$ are sequences of real numbers such that $a_0 = b_0 = 0$, then

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \frac{|a_m| |b_n|}{(qm^{p-1} + pn^{q-1})(m+n)^s} \le \frac{1}{pq} \sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \frac{|a_m| |b_n|}{m^{\frac{1}{q}} n^{\frac{1}{p}} (m+n)^s} \le \frac{N_1}{pq} \left[\sum_{m=1}^{\infty} \sum_{\tau=1}^m m^{1-s} |\nabla a_{\tau}|^p \right]^{\frac{1}{p}} \left[\sum_{n=1}^{\infty} \sum_{\delta=1}^n n^{1-s} |\nabla b_{\delta}|^q \right]^{\frac{1}{q}}$$
(7.16)

and

$$\sum_{k=1}^{\infty} n^{(s-1)(p-1)} \left[\sum_{m=1}^{\infty} \frac{|a_m|}{m^{\frac{1}{q}} (m+n)^s} \right]^p \le N_1^p \sum_{m=1}^{\infty} \sum_{\tau=1}^m m^{1-s} |\nabla a_\tau|^p,$$
(7.17)

where $N_1 = B(\frac{s+q-2}{q}, \frac{s+p-2}{p}).$

Non-conjugate exponents 7.3

Having in mind Theorem 2.1 (Chapter 2), Hilbert-Pachpatte-type inequalities considered in this chapter can also be extended to the case of non-conjugate exponents. Let p, p', q, q', and λ be as in (2.1) and (2.2). To obtain analogous results in the case of non-conjugate exponents, we introduce real parameters r', r such that $p \le r' \le q'$ and $\frac{1}{r'} + \frac{1}{r} = 1$. For example, we can define $\frac{1}{r'} = \frac{1}{q'} + \frac{1-\lambda}{2}$ or $r' = (2 - \lambda)p$. It is easy to see that inequalities

$$x^{\frac{1}{p'}}y^{\frac{1}{q'}} \le \frac{1}{rr'} \left(rx^{\frac{r'}{p'}} + r'y^{\frac{r}{q'}} \right), \ x \ge 0, \ y \ge 0$$
(7.18)

and

$$|f(x)||g(y)| \le x^{\frac{1}{p'}} y^{\frac{1}{q'}} \left[\int_0^x |f'(\tau)|^p d\tau \right]^{\frac{1}{p}} \left[\int_0^y |g'(\delta)|^q d\delta \right]^{\frac{1}{q}}$$
(7.19)

hold, provided $f, g: \mathbb{R}_+ \to \mathbb{R}$ are absolutely continuous functions.

Now, utilizing Theorem 2.1 and inequalities (7.18) and (7.19), we obtain the following general result for non-conjugate exponents, in the same way as in the proof of Theorem 7.3.

Theorem 7.6 Let p, p', q, q', and λ satisfy (2.1) and (2.2), and let r', r be real parameters such that $p \leq r' \leq q'$ and $\frac{1}{r'} + \frac{1}{r} = 1$. If $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$, $\varphi, \psi : \mathbb{R}_+ \to \mathbb{R}$ are non-negative functions and $f, g : \mathbb{R}_+ \to \mathbb{R}$ are absolutely continuous functions such that f(0) = g(0) = 0, then

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{K^{\lambda}(x,y)|f(x)||g(y)|}{rx^{\frac{p'}{p'}} + r'y^{\frac{r}{q'}}} dx dy$$

$$\leq \frac{pq}{rr'} \int_{0}^{\infty} \int_{0}^{\infty} K^{\lambda}(x,y)|f(x)||g(y)|d(x^{\frac{1}{p}})d(y^{\frac{1}{q}})$$

$$\leq \frac{1}{rr'} \left[\int_{0}^{\infty} \int_{0}^{x} (\varphi F)^{p}(x)|f'(\tau)|^{p} d\tau dx \right]^{\frac{1}{p}} \left[\int_{0}^{\infty} \int_{0}^{y} (\psi G)^{q}(y)|g'(\delta)|^{q} d\delta dy \right]^{\frac{1}{q}} (7.20)$$

and

$$\left[\int_0^\infty \left[\frac{1}{\psi G(y)} \int_0^\infty K^\lambda(x,y) |f(x)| d(x^{\frac{1}{p}})\right]^{q'} dy\right]^{\frac{1}{q'}}$$
$$\leq \frac{1}{p} \left[\int_0^\infty \int_0^x (\varphi F)^p(x) |f'(\tau)|^p d\tau dx\right]^{\frac{1}{p}},\tag{7.21}$$

where $F(x) = \left(\int_0^\infty K(x,y)\psi^{-q'}(y)dy\right)^{\frac{1}{q'}}$ and $G(y) = \left(\int_0^\infty K(x,y)\phi^{-p'}(x)dx\right)^{\frac{1}{p'}}$.

Obviously, Theorem 7.6 is a generalization of Theorem 7.3. Namely, setting $\lambda = 1$, r' = p, and r = q in Theorem 7.6, inequalities (7.20) and (7.21) reduce to (7.3) and (7.4) respectively.

The following consequence of Theorem 7.6 may be regarded as a non-conjugate extension of Corollary 7.1.

Corollary 7.6 Let p, p', q, q', and λ satisfy (2.1) and (2.2), and let $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ be a non-negative symmetric homogeneous function of degree -s, s > 0. If $f, g : \mathbb{R}_+ \to \mathbb{R}$ are absolutely continuous functions such that f(0) = g(0) = 0, then

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{K^{\lambda}(x,y)|f(x)||g(y)|}{qx^{(p-1)(2-\lambda)} + py^{(q-1)(2-\lambda)}} dxdy \\
\leq \frac{1}{2-\lambda} \int_{0}^{\infty} \int_{0}^{\infty} K^{\lambda}(x,y)|f(x)||g(y)|d(x^{\frac{1}{p}})d(y^{\frac{1}{q}}) \\
\leq \frac{M}{pq(2-\lambda)} \left[\int_{0}^{\infty} \int_{0}^{x} x^{\frac{p}{q'}(1-s)+p(A_{1}-A_{2})} |f'(\tau)|^{p} d\tau dx \right]^{\frac{1}{p}} \\
\times \left[\int_{0}^{\infty} \int_{0}^{y} y^{\frac{q}{p'}(1-s)+q(A_{2}-A_{1})} |g'(\delta)|^{q} d\delta dy \right]^{\frac{1}{q}}$$
(7.22)

and

$$\left[\int_{0}^{\infty} y^{\frac{q'}{p'}(s-1)+q'(A_{1}-A_{2})} \left[\int_{0}^{\infty} K^{\lambda}(x,y)|f(x)|d(x^{\frac{1}{p}})\right]^{q'} dy\right]^{\frac{1}{q'}} \leq \frac{M}{p} \left[\int_{0}^{\infty} \int_{0}^{x} x^{\frac{p}{q'}(1-s)+p(A_{1}-A_{2})} |f'(\tau)|^{p} d\tau dx\right]^{\frac{1}{p}},$$
(7.23)

where $A_1 \in (\frac{1-s}{p'}, \frac{1}{p'}), A_2 \in (\frac{1-s}{q'}, \frac{1}{q'})$, and $M = k^{\frac{1}{p'}}(p'A_1)k^{\frac{1}{q'}}(q'A_2)$.

Proof. The proof follows directly from Theorem 7.6, setting $r' = (2 - \lambda)p$, $r = (2 - \lambda)q$, $\varphi(x) = x^{A_1}$, and $\psi(y) = y^{A_2}$ in inequalities (7.20) and (7.21). Namely, if F(x) and G(y) are defined as in the statement of Theorem 7.6, it follows that

$$(\varphi F)^{p}(x) = x^{pA_{1}} \left(\int_{0}^{\infty} K(x, y) y^{-q'A_{2}} dy \right)^{\frac{p}{q'}}$$

= $x^{pA_{1}-pA_{2}} \left(\int_{0}^{\infty} K(x, y) \left(\frac{x}{y} \right)^{q'A_{2}} dy \right)^{\frac{p}{q'}}$
= $x^{\frac{p}{q'}(1-s)+p(A_{1}-A_{2})} k^{\frac{p}{q'}}(q'A_{2}),$ (7.24)

and similarly,

$$(\psi G)^{q}(y) = y^{\frac{q}{p'}(1-s)+q(A_2-A_1)} k^{\frac{q}{p'}}(p'A_1).$$
(7.25)

Now, utilizing (7.20), (7.21), (7.24), and (7.25) we obtain (7.22) and (7.23).

Of course, remaining results from previous two sections can also be extended to include non-conjugate parameters. Here, they are omitted.

Remark 7.3 The general Hilbert-Pachpatte-type inequalities in this chapter, as well as their consequences, are taken from [109]. For related results and some other forms of Hilbert-Pachpatte-type inequalities, the reader is referred to [28], [29], [35], [47], [105], [106], [107], and [171].



General Hardy-type inequalities with non-conjugate exponents

In 1925, G. H. Hardy stated and proved in [33] the following integral inequality:

$$\int_0^\infty \left(\frac{1}{x}\int_0^x f(y)\,dy\right)^p dx \le \left(\frac{p}{p-1}\right)^p \|f\|_{L^p(\mathbb{R}_+)},\tag{8.1}$$

where p > 1, and $f \in L^p(\mathbb{R}_+)$ is a non-negative function. This is the original form of the Hardy integral inequality, which later on has been extensively studied and used as a model example for investigations of more general integral inequalities.

During subsequent decades, the Hardy inequality was generalized in many different ways. Roughly speaking, the Hardy inequality was extended to what we call nowadays the general Hardy inequality, or the Hardy-type inequality,

$$\left[\int_{a}^{b} \left(\int_{a}^{x} f(y) \, dy\right)^{q'} u(x) \, dx\right]^{\frac{1}{q'}} \le C_{p,q'} \left(\int_{a}^{b} f^{p}(x) v(x) \, dx\right)^{\frac{1}{p}}, \ f \ge 0,$$
(8.2)

with parameters a, b, p, q', such that $-\infty \le a < b \le \infty$, $0 < q' \le \infty$, $1 \le p \le \infty$, and with u, v given weight functions. The main problem in connection with the Hardy inequality is to determine conditions on the parameters p, q' and on the weight functions u, v under which the inequality holds for some classes of functions.

The Hardy inequality plays important role in various parts of mathematics, especially in functional and spectral analysis, where one investigates properties of the Hardy operator, like continuity and compactness, and also its behavior in more general function spaces. For a more details about the Hardy inequality, its history and related results, the reader is referred to [33], [69], [71], and [103].

Although classical, the Hardy inequality is still a field of interest to numerous authors. In [14] and [15], A. Čizmešija and J. Pečarić investigated finite sections of the Hardy inequality, i.e. inequalities of the same type, where the integrals are taken over certain subsets of \mathbb{R}_+ . In such a way they obtained some generalizations and refinements of (8.1). For example, in [15], they proved that

$$\int_{0}^{b} x^{-k} \left(\int_{0}^{x} f(y) \, dy \right)^{p} dx \le \left(\frac{p}{k-1} \right)^{p} \int_{0}^{b} \left[1 - \left(\frac{x}{b} \right)^{\frac{k-1}{p}} \right] x^{p-k} f^{p}(x) \, dx, \tag{8.3}$$

where $0 < b < \infty$, $1 < p, k < \infty$, $f \ge 0$, and $x^{1-\frac{k}{p}} f \in L^{p}(0, b)$.

It is well known that the Hardy inequality is closely connected to the Hilbert inequality. That connection may be explained in a more general setting. Namely, Theorem 1.9 (Chapter 1) provides a unified treatment of Hilbert-type inequalities with conjugate exponents. In addition, as a consequence of the above mentioned theorem, M. Krnić and J. Pečarić [53], extended the Hardy integral inequality to cover the case when p and p' are conjugate exponents. More precisely, they obtained the following pair of equivalent inequalities:

$$\int_{a}^{b} \int_{a}^{y} (hg)(y)f(x) d\mu_{1}(x)d\mu_{2}(y)$$

$$\leq \left[\int_{a}^{b} \varphi^{p}(x) \left(\int_{x}^{b} H(y) d\mu_{2}(y)\right) f^{p}(x) d\mu_{1}(x)\right]^{\frac{1}{p}}$$

$$\times \left[\int_{a}^{b} (\psi^{p'}h)(y) \left(\int_{a}^{y} \varphi^{-p'}(x) d\mu_{1}(x)\right) g^{p'}(y) d\mu_{2}(y)\right]^{\frac{1}{p'}} (8.4)$$

and

$$\int_{a}^{b} H(y) \left(\int_{a}^{y} \varphi^{-p'}(x) d\mu_{1}(x) \right)^{1-p} \left(\int_{a}^{y} f(x) d\mu_{1}(x) \right)^{p} d\mu_{2}(y) \\ \leq \int_{a}^{b} (\psi^{p'}h)(y) \left(\int_{a}^{y} \varphi^{-p'}(x) d\mu_{1}(x) \right) g^{p'}(y) d\mu_{2}(y),$$
(8.5)

where p > 1, μ_1, μ_2 are positive σ -finite measures, h, f, g, φ, ψ are measurable, positive functions a.e. on (a,b), and $H = h\psi^{-p}$. Inequality (8.5) extends (8.1) and (8.3), as well as numerous results known from the literature. Therefore, the inequalities deduced from (8.5) will be referred to as the Hardy-type inequalities.

On the other hand, Theorem 2.1 (Chapter 2) covers the Hilbert-type inequalities with non-conjugate exponents. The main objective of this chapter is to extend the general Hardy-type inequality to the case of non-conjugate exponents. This will be done with the help of the above mentioned result regarding non-conjugate exponents.

186

This chapter is organized in the following way: In Section 8.1 we state and prove a pair of equivalent Hilbert and Hardy-type inequalities with non-conjugate exponents p and q, in the context of general measure spaces with positive σ -finite measures, and to the Hardy-type kernels. In Section 8.2 we discuss duality in Hardy-type inequalities, that is, we obtain dual analogues of the results in Section 8.1. Further, in Section 8.3 general results are applied to special Hardy-type kernels and power weight functions with integrals taken over intervals in \mathbb{R}_+ . In such a way, a numerous new inequalities with explicit constant factors on their right-hand sides are obtained. In Section 8.4 we estimate some factors included in the inequalities from the previous section, depending on non-conjugate parameters and the exponents of power weight functions. Section 8.5 is dedicated to some uniform bounds for constant factors in Hardy-type inequalities. We perform a detailed analysis of optimal constants, depending on non-conjugate parameters p and q. Finally, in the last section, we synthesize the methods developed in Sections 8.3, 8.4, and 8.5.

8.1 General inequalities of the Hardy-type

In this section we prove the main result that extends relations (8.4) and (8.5) to the case of non-conjugate exponents.

Let (a,b) be an interval in \mathbb{R} , $T = \{(x,y) \in \mathbb{R}^2 : a < x \le y < b\}$, and let μ_1 and μ_2 be positive σ -finite measures on (a,b). We define the Hardy-type kernel $K : (a,b) \times (a,b) \rightarrow \mathbb{R}$ by

$$K(x,y) = h(y)\chi_T(x,y) = \begin{cases} h(y) , x \le y, \\ 0 , x > y, \end{cases}$$
(8.6)

where *h* is a measurable, a.e. positive function on (a,b). Further, we define the functions $F:(a,b) \to \mathbb{R}$ and $G:(a,b) \to \mathbb{R}$ by

$$F(x) = \left[\int_{x}^{b} h(y) \psi^{-q'}(y) d\mu_{2}(y) \right]^{\frac{1}{q'}}, \ x \in (a,b),$$

$$G(y) = \left[h(y) \int_{a}^{y} \varphi^{-p'}(x) d\mu_{1}(x) \right]^{\frac{1}{p'}}, \ y \in (a,b),$$
(8.7)

where ψ and ϕ are measurable, a.e. positive functions on (a, b).

We also introduce the related Hardy-type operator by the formula

$$(Hf)(y) = \int_{a}^{y} f(x) d\mu_{1}(x), \qquad y \in (a,b).$$
(8.8)

Now, we are ready to state and prove the main result.

Theorem 8.1 Let p, q, and λ be real parameters satisfying (2.1) and (2.2), and let μ_1 and μ_2 be σ -finite measures on (a,b), $-\infty \le a < b \le \infty$. Let h, φ , ψ be measurable,

a.e. positive functions on (a,b), and let H be the operator defined by (8.8). If the functions F and G are defined by (8.7), then the inequalities

$$\int_{a}^{b} (h^{\lambda}g)(y) (Hf)(y) d\mu_{2}(y) \leq \|\varphi Ff\|_{L^{p}(\mu_{1})} \|\psi Gg\|_{L^{q}(\mu_{2})}$$
(8.9)

and

1

$$\left\{\int_{a}^{b}(h\psi^{-q'})(y)\left[\int_{a}^{y}\varphi^{-p'}(x)\,d\mu_{1}(x)\right]^{-\frac{q'}{p'}}(Hf)^{q'}(y)\,d\mu_{2}(y)\right\}^{\frac{1}{q'}} \leq \|\varphi Ff\|_{L^{p}(\mu_{1})}$$
(8.10)

hold for all non-negative functions f and g on (a,b), such that $\varphi F f \in L^p(\mu_1)$ and $\psi G g \in L^q(\mu_2)$, and they are equivalent.

Proof. We follow the same lines as in the proof of Theorem 2.1 (see Chapter 2). Namely, the left-hand side of (8.9) can be rewritten as

$$L = \int_{a}^{b} \int_{a}^{y} (h^{\lambda}g)(y)f(x) d\mu_{1}(x)d\mu_{2}(y)$$

=
$$\int_{a}^{b} \int_{a}^{y} \left[(h\psi^{-q'})(y)(\varphi^{p}F^{p-q'}f^{p})(x) \right]^{\frac{1}{q'}} \times \left[\varphi^{-p'}(x)(h\psi^{q}G^{q-p'}g^{q})(y) \right]^{\frac{1}{p'}} \times \left[(\varphi Ff)^{p}(x)(\psi Gg)^{q}(y) \right]^{1-\lambda} d\mu_{1}(x)d\mu_{2}(y),$$
(8.11)

since $\frac{1}{q'} + \frac{1}{p'} + (1 - \lambda) = 1$. Further, utilizing the Hölder inequality, either with the parameters $q', p', \frac{1}{1-\lambda} > 1$ in the case of non-conjugate exponents p and q, or with the parameters p and p' when q' = p, and then applying the Fubini theorem, we obtain that L does not exceed

$$R = \left\{ \int_{a}^{b} \left[\int_{x}^{b} (h\psi^{-q'})(y) \, d\mu_{2}(y) \right] (\varphi^{p}F^{p-q'}f^{p})(x) \, d\mu_{1}(x) \right\}^{\frac{1}{q'}} \\ \times \left\{ \int_{a}^{b} \left[h(y) \int_{a}^{y} \varphi^{-p'}(x) \, d\mu_{1}(x) \right] (\psi^{q}G^{q-p'}g^{q})(y) \, d\mu_{2}(y) \right\}^{\frac{1}{p'}} \\ \times \left\{ \int_{a}^{b} (\psi Gg)^{q}(y) \int_{a}^{y} (\varphi Ff)^{p}(x) \, d\mu_{1}(x) d\mu_{2}(y) \right\}^{1-\lambda}.$$

Now, exploiting definitions (8.7), the above expression can be rewritten as

$$R = \|\varphi F f\|_{L^{p}(\mu_{1})}^{\frac{p}{q'}} \|\psi G g\|_{L^{q}(\mu_{2})}^{q} \\ \times \left\{ \|\varphi F f\|_{L^{p}(\mu_{1})}^{p} \|\psi G g\|_{L^{q}(\mu_{2})}^{q} - \int_{a}^{b} \int_{y}^{b} (\varphi F f)^{p} (x) (\psi G g)^{q} (y) d\mu_{1}(x) d\mu_{2}(y) \right\}^{1-\lambda}.$$
(8.12)

Of course, relations (8.11) and (8.12) yield

$$L \le R. \tag{8.13}$$

Moreover, using (8.12), we easily obtain

$$R \le \|\varphi Ff\|_{L^{p}(\mu_{1})}^{\frac{p}{q'}+p(1-\lambda)} \|\psi Gg\|_{L^{q}(\mu_{2})}^{\frac{q}{p'}+q(1-\lambda)} = \|\varphi Ff\|_{L^{p}(\mu_{1})} \|\psi Gg\|_{L^{q}(\mu_{2})},$$

so that (8.9) is proved.

The further step is to prove that (8.9) implies that (8.10) holds for all non-negative measurable functions f on (a,b). In particular, for any such f and the function

$$g(y) = (\psi^{-q'} h^{1-\lambda})(y) \left[\int_{a}^{y} \varphi^{-p'}(x) \, d\mu_1(x) \right]^{-\frac{q'}{p'}} (Hf)^{q'-1}(y), \quad y \in (a,b), \tag{8.14}$$

applying the Fubini theorem, the left-hand side of (8.9) becomes

$$L_f = \int_a^b (h\psi^{-q'})(y) \left[\int_a^y \varphi^{-p'}(x) \, d\mu_1(x) \right]^{-\frac{q'}{p'}} (Hf)^{q'}(y) \, d\mu_2(y),$$

that is, we get the integral on the left-hand side of (8.10), while on the right-hand side of (8.9) we have

$$R_{f} = \|\varphi Ff\|_{L^{p}(\mu_{1})} \left\{ \int_{a}^{b} (h\psi^{-q'})(y) \left[\int_{a}^{y} \varphi^{-p'}(x) d\mu_{1}(x) \right]^{-\frac{q'}{p'}} \times (Hf)^{q'}(y) d\mu_{2}(y) \right\}^{\frac{1}{q}} = \|\varphi Ff\|_{L^{p}(\mu_{1})} L_{f}^{\frac{1}{q}}.$$

Hence, $L_f \leq \|\varphi F f\|_{L^p(\mu_1)} L_f^{\frac{1}{q}}$, which directly yields (8.10). Conversely, utilizing the Hölder inequality for conjugate exponents q and q', together with relation (8.10) and definitions (8.7), we have

$$\begin{split} &\int_{a}^{b} (h^{\lambda}g)(y)(Hf)(y) \, d\mu_{2}(y) \\ &= \int_{a}^{b} (\psi Gg)(y) \left[(\psi G)^{-1}(y) h^{\lambda}(y)(Hf)(y) \right] d\mu_{2}(y) \\ &\leq \|\psi Gg\|_{L^{q}(\mu_{2})} \left\{ \int_{a}^{b} (h\psi^{-q'})(y) \left[\int_{a}^{y} \varphi^{-p'}(x) \, d\mu_{1}(x) \right]^{-\frac{q'}{p'}} (Hf)^{q'}(y) \, d\mu_{2}(y) \right\}^{\frac{1}{q'}} \\ &\leq \|\varphi Ff\|_{L^{p}(\mu_{1})} \|\psi Gg\|_{L^{q}(\mu_{2})}. \end{split}$$

Thus, (8.10) implies (8.9), so these inequalities are equivalent.

Remark 8.1 At the first sight, the proof of Theorem 8.1 is redundant since inequalities (8.9) and (8.10) follow from inequalities (2.9) and (2.10) (Theorem 2.1, Chapter 2) equipped with the Hardy-type kernel. On the other hand, in the proof of Theorem 8.1, we have obtained inequality (8.13) which is a refinement of inequality (8.9). Let us write that inequality once again, i.e.

$$\int_{a}^{b} (h^{\lambda}g)(y) (Hf)(y) d\mu_{2}(y) \leq \|\varphi Ff\|_{L^{p}(\mu_{1})}^{\frac{p}{q'}} \|\psi Gg\|_{L^{q}(\mu_{2})}^{\frac{q}{p'}} \\ \times \left\{ \int_{a}^{b} \int_{a}^{y} (\varphi Ff)^{p}(x) (\psi Gg)^{q}(y) d\mu_{1}(x) d\mu_{2}(y) \right\}^{1-\lambda}.$$
(8.15)

Clearly, substituting the function g, defined by (8.14), in the above inequality, we obtain its equivalent Hardy-type form

$$\left\{ \int_{a}^{b} (h\psi^{-q'})(y) \left[\int_{a}^{y} \varphi^{-p'}(x) d\mu_{1}(x) \right]^{-\frac{q'}{p'}} (Hf)^{q'}(y) d\mu_{2}(y) \right\}^{\frac{1}{p}} \\
\leq \|\varphi Ff\|_{L^{p}(\mu_{1})}^{\frac{p}{q'}} \cdot \left\{ \int_{a}^{b} (h\psi^{-q'})(y) \left[\int_{a}^{y} \varphi^{-p'}(x) d\mu_{1}(x) \right]^{-\frac{q'}{p'}} \\
\times (Hf)^{q'}(y) \int_{a}^{y} (\varphi Ff)^{p} d\mu_{1}(x) d\mu_{2}(y) \right\}^{1-\lambda}.$$
(8.16)

Inequality (8.16) is also a slight refinement of (8.10). It should be noticed here that these refinements hold only in non-conjugate case.

Remark 8.2 Taking into account the form of Hardy-type kernel, it follows from Remark 2.3 (Chapter 2) that the equality in (8.9) and (8.10) holds only in the trivial case, that is, when f = 0 or g = 0 a.e. on (a, b). In addition, the reverse inequalities in (8.9) and (8.10) hold if the conditions (2.12) or (2.13) are fulfilled (see Remark 2.2, Chapter 2).

Remark 8.3 In the case of conjugate exponents, that is, when q = p' and $\lambda = 1$, inequalities (8.9) and (8.10) reduce to relations (8.4) and (8.5). Thus, Theorem 8.1 may be regarded as an extension of the corresponding results from [53] to the case of non-conjugate exponents. Clearly, reverse inequalities in (8.4) and (8.5) hold if $0 \neq p < 1$.

8.2 General inequalities with dual Hardy-type kernel

One of important properties of the Hardy inequality is duality. Here we obtain dual analogues of the relations from the previous section.

Let $(a,b) \subseteq \mathbb{R}$, let $\widetilde{T} = \{(x,y) \in \mathbb{R}^2 : a < y \le x < b\}$, and let μ_1 and μ_2 be positive σ -finite measures on (a,b). We define the dual Hardy-type kernel $\widetilde{K} : (a,b) \times (a,b) \to \mathbb{R}$ by

$$\widetilde{K}(x,y) = h(y)\chi_{\widetilde{T}}(x,y) = \begin{cases} h(y) , & x \ge y, \\ 0 , & x < y, \end{cases}$$
(8.17)

where *h* is a measurable function which is a.e. positive on (a,b). Moreover, we define the functions $\widetilde{F}: (a,b) \to \mathbb{R}$ and $\widetilde{G}: (a,b) \to \mathbb{R}$ by

$$\widetilde{F}(x) = \left[\int_{a}^{x} (h\psi^{-q'})(y) \, d\mu_{2}(y) \right]^{\frac{1}{q'}}, \ x \in (a,b),$$

$$\widetilde{G}(y) = \left[h(y) \int_{y}^{b} \varphi^{-p'}(x) \, d\mu_{1}(x) \right]^{\frac{1}{p'}}, \ y \in (a,b),$$
(8.18)

where ψ and φ are measurable functions that are a. e. positive on (a,b) with respect to the corresponding σ -finite measures.

Further, the dual Hardy-type operator with respect to the operator H in (8.8) is defined in the following way:

$$(\widetilde{H}f)(y) = \int_{y}^{b} f(x) d\mu_{1}(x), \qquad y \in (a,b).$$
 (8.19)

In this setting, we obtain a dual analogue of Theorem 8.1.

Theorem 8.2 Let p, q, and λ be real parameters satisfying (2.1) and (2.2), and let μ_1 and μ_2 be σ -finite measures on (a,b), $-\infty \le a < b \le \infty$. Let h, φ , ψ be measurable, a.e. positive functions on (a,b), and let \widetilde{H} be the operator defined by (8.19). If the functions \widetilde{F} and \widetilde{G} are defined by (8.18), then the inequalities

$$\int_{a}^{b} (h^{\lambda}g)(y)(\widetilde{H}f)(y) d\mu_{2}(y) \leq \|\varphi\widetilde{F}f\|_{L^{p}(\mu_{1})} \|\psi\widetilde{G}g\|_{L^{q}(\mu_{2})}$$

$$(8.20)$$

and

$$\left\{\int_{a}^{b}(h\psi^{-q'})(y)\left[\int_{y}^{b}\varphi^{-p'}(x)\,d\mu_{1}(x)\right]^{-\frac{q'}{p'}}(\widetilde{H}f)^{q'}(y)\,d\mu_{2}(y)\right\}^{\frac{1}{q'}} \leq \|\varphi\widetilde{F}f\|_{L^{p}(\mu_{1})}$$
(8.21)

hold for all non-negative functions f and g on (a,b), such that $\varphi \widetilde{F} f \in L^p(\mu_1)$ and $\psi \widetilde{G} g \in L^q(\mu_2)$, and are equivalent.

Proof. It follows the same lines as the proof of Theorem 8.1.

Remark 8.4 The equality in dual inequalities (8.20) and (8.21) holds only in the trivial case, that is, when f = 0 or g = 0 a.e. on (a, b). Moreover, the discussion about the reverse inequalities in (8.20) and (8.21) is the same as in Remark 8.2.

Remark 8.5 Similarly to Remark 8.1, one easily obtains a refinement of (8.20) in the non-conjugate case, that is,

$$\int_{a}^{b} (h^{\lambda}g)(y)(\widetilde{H}f)(y) d\mu_{2}(y) \leq \|\varphi\widetilde{F}f\|_{L^{p}(\mu_{1})}^{\frac{p}{q'}} \|\psi\widetilde{G}g\|_{L^{q}(\mu_{2})}^{\frac{q}{p'}} \\ \times \left\{\int_{a}^{b} \int_{y}^{b} (\varphi\widetilde{F}f)^{p}(x)(\psi\widetilde{G}g)^{q}(y) d\mu_{1}(x)d\mu_{2}(y)\right\}^{1-\lambda},$$

$$(8.22)$$

with sharp inequality for $f, g \neq 0$ a.e. on (a, b). Furthermore, inserting the function g defined by

$$g(y) = (\psi^{-q'}h^{1-\lambda})(y) \left[\int_{y}^{b} \varphi^{-p'}(x) \, d\mu_1(x) \right]^{-\frac{q'}{p'}} (\widetilde{H}f)^{q'-1}(y), \quad y \in (a,b),$$

in (8.22), we obtain the inequality

$$\left\{ \int_{a}^{b} (h\psi^{-q'})(y) \left[\int_{y}^{b} \varphi^{-p'}(x) d\mu_{1}(x) \right]^{-\frac{q'}{p'}} (\widetilde{H}f)^{q'}(y) d\mu_{2}(y) \right\}^{\frac{1}{p}} \\
\leq \|\varphi\widetilde{F}f\|_{L^{p}(\mu_{1})}^{\frac{p}{q'}} \left\{ \int_{a}^{b} (h\psi^{-q'})(y) \left[\int_{y}^{b} \varphi^{-p'}(x) d\mu_{1}(x) \right]^{-\frac{q'}{p'}} \\
\times (\widetilde{H}f)^{q'}(y) \int_{y}^{b} (\varphi\widetilde{F}f)^{p} d\mu_{1}(x) d\mu_{2}(y) \right\}^{1-\lambda},$$
(8.23)

which can be regarded as a refinement of (8.21).

The most interesting case in connection with dual inequalities appears when $(a,b) \subseteq \mathbb{R}_+$. Namely, we show that Theorems 8.1 and 8.2 are equivalent in the case of Lebesgue measures.

Theorem 8.3 Let $0 \le a < b \le \infty$ and let $d\mu_1(x) = dx$, $d\mu_2(y) = dy$. Then inequalities (8.9) and (8.20) are equivalent. Moreover, inequalities (8.10) and (8.21) are equivalent as well.

Proof. Suppose that inequality (8.9) holds for an arbitrary interval $(a,b) \subseteq \mathbb{R}_+$ and arbitrary non-negative measurable functions φ, ψ, h, f, g on (a,b). We define $\tilde{a} = \frac{1}{b}$ and $\tilde{b} = \frac{1}{a}$, with conventions $\tilde{a} = 0$ for $b = \infty$ and $\tilde{b} = \infty$ for a = 0. We also define the functions $\tilde{h}, \tilde{\varphi}, \tilde{\psi}, \tilde{f}$, and \tilde{g} on (\tilde{a}, \tilde{b}) by $\tilde{h}(t) = h\left(\frac{1}{t}\right), \tilde{\varphi}(t) = t^{\frac{2}{p'}}\varphi\left(\frac{1}{t}\right), \tilde{\psi}(t) = t^{\frac{2}{q'}}\psi\left(\frac{1}{t}\right), \tilde{f}(t) = t^{-2}f\left(\frac{1}{t}\right)$, and $\tilde{g}(t) = t^{-2}g\left(\frac{1}{t}\right)$.

Rewrite (8.9) with these new parameters. More precisely, using the change of variables $x = \frac{1}{y}$ and $y = \frac{1}{y}$, the left-hand side of (8.9) becomes

$$\int_{\widetilde{a}}^{\widetilde{b}} (\widetilde{h}^{\lambda} \widetilde{g})(y) t(H\widetilde{f})(y) \, dy = \int_{a}^{b} (h^{\lambda} g)(v) \int_{v}^{b} f(u) \, du \, dv,$$

that is, the left-hand side of inequality (8.20). Analogously, for the first factor on the right-hand side of (8.9) we have

$$\|\widetilde{\varphi}F\widetilde{f}\|_{L^{p}(\widetilde{a},\widetilde{b})}^{p} = \int_{a}^{b} (\varphi f)^{p}(u) \left[\int_{a}^{u} (h\psi^{-q'})(v)dv\right]^{\frac{p}{q'}} du$$

which obviously represents the first factor on the right-hand side of inequality (8.20). The same argument holds for the second factor on the right-hand side of (8.9). Thus, inequality (8.9) implies (8.20). In the same manner, one obtains the reverse implication, so inequalities (8.9) and (8.20) are equivalent.

Finally, pairs of inequalities (8.9) and (8.10) as well as (8.20) and (8.21) are equivalent (see Theorems 8.1 and 8.2), which implies the equivalence of (8.10) and (8.21). \Box

8.3 Some special Hardy-type kernels and weight functions

In this section, we consider the case of Lebesgue measure for some particular Hardy-type kernels and weight functions. Namely, let $0 \le a < b \le \infty$ and let $h, \varphi, \psi : (a, b) \to \mathbb{R}$ be defined by $h(y) = \frac{1}{y}, \varphi(x) = x^{A_1}, \psi(y) = y^{A_2}, A_1, A_2 \in \mathbb{R}$, respectively. As it was shown in the previous section, it is sufficient to consider only the Hardy-type inequalities in Theorem 8.1, since their dual inequalities are equivalent with them.

In particular, we have to distinguish the following cases:

$$0 < a < b < \infty, \tag{8.24}$$

20

$$0 = a < b < \infty, \tag{8.25}$$

$$0 < a < b = \infty, \tag{8.26}$$

$$0 = a < b = \infty, \tag{8.27}$$

since one obtains different integration formulas for the functions *F* and *G*, defined by (8.7). In particular, if $0 < a < b < \infty$, then

$$F(x) = \begin{cases} |q'A_2|^{-\frac{1}{q'}} x^{-A_2} \left| 1 - \left(\frac{x}{b}\right)^{q'A_2} \right|^{\frac{1}{q'}}, A_2 \neq 0, \\ \left(\log \frac{b}{x}\right)^{\frac{1}{q'}}, & A_2 = 0, \end{cases}$$
(8.28)

$$G(y) = \begin{cases} \left|1 - p'A_1\right|^{-\frac{1}{p'}} y^{-A_1} \left|1 - \left(\frac{a}{y}\right)^{1 - p'A_1}\right|^{\frac{1}{p'}}, A_1 \neq \frac{1}{p'}, \\ y^{-\frac{1}{p'}} \left(\log \frac{y}{a}\right)^{\frac{1}{p'}}, A_1 = \frac{1}{p'}. \end{cases}$$
(8.29)

It should be noticed here that we have included the cases $A_1 > \frac{1}{p'}$ and $A_2 < 0$, by means of the modulus function. In this setting, we obtain four corollaries arising from Theorem 8.1. If $A_1 \neq \frac{1}{p'}$ and $A_2 \neq 0$, then we have the following result:

Corollary 8.1 Let $-\infty < a < b < \infty$, assume that real parameters p, q, and λ satisfy (2.1) and (2.2), let A_1 , A_2 be real parameters such that $A_1 \neq \frac{1}{p'}$, $A_2 \neq 0$, and let H be the operator defined by (8.8). Then the inequalities

$$\int_{a}^{b} y^{-\lambda} g(y)(Hf)(y) \, dy$$

$$\leq \frac{|1 - p'A_{1}|^{-\frac{1}{p'}}}{|q'A_{2}|^{\frac{1}{q'}}} \left[\int_{a}^{b} x^{(A_{1} - A_{2})p} \left| 1 - \left(\frac{x}{b}\right)^{q'A_{2}} \right|^{\frac{p}{q'}} f^{p}(x) \, dx \right]^{\frac{1}{p}}$$

$$\times \left[\int_{a}^{b} y^{(A_{2} - A_{1})q} \left| 1 - \left(\frac{a}{y}\right)^{1 - p'A_{1}} \right|^{\frac{q}{p'}} g^{q}(y) \, dy \right]^{\frac{1}{q}}$$
(8.30)

and

$$\int_{a}^{b} y^{(A_{1}-A_{2}-\lambda)q'} \left| 1 - \left(\frac{a}{y}\right)^{1-p'A_{1}} \right|^{-\frac{q'}{p'}} (Hf)^{q'}(y) \, dy \right]^{\frac{1}{q'}} \\
\leq \frac{\left|1 - p'A_{1}\right|^{-\frac{1}{p'}}}{\left|q'A_{2}\right|^{\frac{1}{q'}}} \left[\int_{a}^{b} x^{(A_{1}-A_{2})p} \left| 1 - \left(\frac{x}{b}\right)^{q'A_{2}} \right|^{\frac{p}{q'}} f^{p}(x) \, dx \right]^{\frac{1}{p}} \tag{8.31}$$

hold for all non-negative measurable functions f and g on (a,b), and are equivalent.

The case of $A_1 \neq \frac{1}{p'}$, $A_2 = 0$ is described in the following result.

Corollary 8.2 Let $-\infty < a < b < \infty$, let p, q, and λ be as in (2.1) and (2.2), assume that A_1 is a real parameter such that $A_1 \neq \frac{1}{p'}$, and let H be the operator defined by (8.8). Then the inequalities

$$\int_{a}^{b} y^{-\lambda} g(y)(Hf)(y) \, dy \leq |1 - p'A_{1}|^{-\frac{1}{p'}} \left[\int_{a}^{b} x^{A_{1}p} \left(\log \frac{b}{x} \right)^{\frac{p}{q'}} f^{p}(x) \, dx \right]^{\frac{1}{p}} \\ \times \left[\int_{a}^{b} y^{-A_{1}q} \left| 1 - \left(\frac{a}{y} \right)^{1 - p'A_{1}} \right|^{\frac{q}{p'}} g^{q}(y) \, dy \right]^{\frac{1}{q}}$$
(8.32)

$$\left[\int_{a}^{b} y^{(A_{1}-\lambda)q'} \left|1 - \left(\frac{a}{y}\right)^{1-p'A_{1}}\right|^{-\frac{q'}{p'}} (Hf)^{q'}(y) \, dy\right]^{\frac{1}{q'}} \\ \leq \left|1 - p'A_{1}\right|^{-\frac{1}{p'}} \left[\int_{a}^{b} x^{A_{1}p} \left(\log\frac{b}{x}\right)^{\frac{p}{q'}} f^{p}(x) \, dx\right]^{\frac{1}{p}}$$
(8.33)

hold for all non-negative measurable functions f and g on (a,b), and are equivalent.

If $A_1 = \frac{1}{p'}$ and $A_2 \neq 0$, we obtain the following corollary.

Corollary 8.3 Let $-\infty < a < b < \infty$, let p, q, and λ be as in (2.1) and (2.2), assume that A_2 is a real parameter such that $A_2 \neq 0$, and let H be the operator defined by (8.8). Then the inequalities

$$\int_{a}^{b} y^{-\lambda} g(y)(Hf)(y) \, dy$$

$$\leq |q'A_{2}|^{-\frac{1}{q'}} \left[\int_{a}^{b} x^{(1-A_{2})p-1} \left| 1 - \left(\frac{x}{b}\right)^{q'A_{2}} \right|^{\frac{p}{q'}} f^{p}(x) \, dx \right]^{\frac{1}{p}}$$

$$\times \left[\int_{a}^{b} y^{(A_{2}-\frac{1}{p'})q} \left(\log \frac{y}{a} \right)^{\frac{q}{p'}} g^{q}(y) \, dy \right]^{\frac{1}{q}}$$
(8.34)

and

$$\left[\int_{a}^{b} y^{-q'A_{2}-1} \left(\log \frac{y}{a}\right)^{-\frac{q'}{p'}} (Hf)^{q'}(y) \, dy\right]^{\frac{1}{q'}} \\ \leq |q'A_{2}|^{-\frac{1}{q'}} \left[\int_{a}^{b} x^{(1-A_{2})p-1} \left|1-\left(\frac{x}{b}\right)^{q'A_{2}}\right|^{\frac{p}{q'}} f^{p}(x) \, dx\right]^{\frac{1}{p}}$$
(8.35)

hold for all non-negative measurable functions f and g on (a,b), and are equivalent.

Finally, if $A_1 = \frac{1}{p'}$ and $A_2 = 0$ we have:

Corollary 8.4 Let $-\infty < a < b < \infty$, let p, q, and λ be as in (2.1) and (2.2), and let H be the operator defined by (8.8). Then the inequalities

$$\int_{a}^{b} y^{-\lambda} g(y)(Hf)(y) \, dy \leq \left[\int_{a}^{b} x^{p-1} \left(\log \frac{b}{x} \right)^{\frac{p}{q'}} f^{p}(x) \, dx \right]^{\frac{1}{p}} \\ \times \left[\int_{a}^{b} y^{-\frac{q}{p'}} \left(\log \frac{y}{a} \right)^{\frac{q}{p'}} g^{q}(y) \, dy \right]^{\frac{1}{q}}$$
(8.36)

1

$$\left[\int_{a}^{b} y^{-1} \left(\log \frac{y}{a}\right)^{-\frac{q'}{p'}} (Hf)^{q'}(y) \, dy\right]^{\frac{1}{q'}} \le \left[\int_{a}^{b} x^{p-1} \left(\log \frac{b}{x}\right)^{\frac{p}{q'}} f^{p}(x) \, dx\right]^{\frac{1}{p}}$$
(8.37)

hold for all non-negative measurable functions f and g on (a,b), and are equivalent.

Now, we consider the second case, that is, (8.25), where a = 0. Then F is defined by (8.28), and

$$G(y) = (1 - p'A_1)^{-\frac{1}{p'}} y^{-A_1}, \qquad y \in (0, b),$$
(8.38)

where $1 - p'A_1 > 0$. In this case, we obtain the following two results, dependent on the value of the parameter A_2 ($A_2 \neq 0$ or $A_2 = 0$).

Corollary 8.5 Let p, q, and λ be as in (2.1) and (2.2), let $0 < b < \infty$, assume that A_1 , A_2 are real parameters such that $p'A_1 < 1$, $A_2 \neq 0$, and let H be the operator defined by (8.8). Then the inequalities

$$\int_{0}^{b} y^{-\lambda} g(y)(Hf)(y) \, dy$$

$$\leq \frac{(1-p'A_{1})^{-\frac{1}{p'}}}{|q'A_{2}|^{\frac{1}{q'}}} \left[\int_{0}^{b} x^{(A_{1}-A_{2})p} \left| 1 - \left(\frac{x}{b}\right)^{q'A_{2}} \right|^{\frac{p}{q'}} f^{p}(x) \, dx \right]^{\frac{1}{p}}$$

$$\times \left[\int_{0}^{b} y^{(A_{2}-A_{1})q} g^{q}(y) \, dy \right]^{\frac{1}{q}}$$
(8.39)

and

$$\left[\int_{0}^{b} y^{(A_{1}-A_{2}-\lambda)q'}(Hf)^{q'}(y) \, dy\right]^{\frac{1}{q'}} \leq \frac{(1-p'A_{1})^{-\frac{1}{p'}}}{|q'A_{2}|^{\frac{1}{q'}}} \left[\int_{0}^{b} x^{(A_{1}-A_{2})p} \left|1-\left(\frac{x}{b}\right)^{q'A_{2}}\right|^{\frac{p}{q'}} f^{p}(x) \, dx\right]^{\frac{1}{p}}$$
(8.40)

hold for all non-negative measurable functions f and g on (0,b), and are equivalent.

Corollary 8.6 Let p, q, and λ be as in (2.1) and (2.2), let $0 < b < \infty$, assume that A_1 is a real parameter such that $p'A_1 < 1$, and let H be the operator defined by (8.8). Then the inequalities

$$\int_{0}^{b} y^{-\lambda} g(y)(Hf)(y) \, dy \le (1 - p'A_1)^{-\frac{1}{p'}} \left[\int_{0}^{b} x^{A_1 p} \left(\log \frac{b}{x} \right)^{\frac{p}{q'}} f^p(x) \, dx \right]^{\frac{1}{p}} \\ \times \left[\int_{0}^{b} y^{-A_1 q} g^q(y) \, dy \right]^{\frac{1}{q}}$$
(8.41)

$$\left[\int_{0}^{b} y^{(A_{1}-\lambda)q'}(Hf)^{q'}(y) \, dy\right]^{\frac{1}{q'}} \leq \left(1-p'A_{1}\right)^{-\frac{1}{p'}} \left[\int_{0}^{b} x^{A_{1}p} \left(\log\frac{b}{x}\right)^{\frac{p}{q'}} f^{p}(x) \, dx\right]^{\frac{1}{p}}$$
(8.42)

hold for all non-negative measurable functions f and g on (0,b), and are equivalent.

The next case (8.26) includes $b = \infty$. Then G is defined by (8.29) and

$$F(x) = (q'A_2)^{-\frac{1}{q'}} x^{-A_2}, \qquad x \in (a, \infty),$$
(8.43)

where $q'A_2 > 0$. In this setting, we obtain the following two results, depending on value of the parameter A_1 ($A_1 \neq \frac{1}{p'}$ or $A_1 = \frac{1}{p'}$).

Corollary 8.7 Let $0 < a < \infty$, let p, q, and λ be as in (2.1) and (2.2), assume that A_1, A_2 are two real parameters such that $A_1 \neq \frac{1}{p'}$, $q'A_2 > 0$, and let H be the operator defined by (8.8). Then the inequalities

$$\int_{a}^{\infty} y^{-\lambda} g(y)(Hf)(y) \, dy$$

$$\leq \frac{|1 - p'A_{1}|^{-\frac{1}{p'}}}{(q'A_{2})^{\frac{1}{q'}}} \left[\int_{a}^{\infty} x^{(A_{1} - A_{2})p} f^{p}(x) \, dx \right]^{\frac{1}{p}}$$

$$\times \left[\int_{a}^{\infty} y^{(A_{2} - A_{1})q} \left| 1 - \left(\frac{a}{y}\right)^{1 - p'A_{1}} \right|^{\frac{q}{p'}} g^{q}(y) \, dy \right]^{\frac{1}{q}}$$
(8.44)

and

$$\left[\int_{a}^{\infty} y^{(A_{1}-A_{2}-\lambda)q'} \left|1 - \left(\frac{a}{y}\right)^{1-p'A_{1}}\right|^{-\frac{q'}{p'}} (Hf)^{q'}(y) \, dy\right]^{\frac{1}{q'}} \\ \leq \frac{\left|1 - p'A_{1}\right|^{-\frac{1}{p'}}}{(q'A_{2})^{\frac{1}{q'}}} \left[\int_{a}^{\infty} x^{(A_{1}-A_{2})p} f^{p}(x) \, dx\right]^{\frac{1}{p}}$$
(8.45)

hold for all non-negative measurable functions f and g on (a, ∞) , and are equivalent.

Corollary 8.8 Let p, q, and λ be as in (2.1) and (2.2), let $0 < a < \infty$, assume that A_2 is a real parameter such that $q'A_2 > 0$, and let H be the operator defined by (8.8). Then the inequalities

$$\int_{a}^{\infty} y^{-\lambda} g(y)(Hf)(y) \, dy \le \left(q'A_{2}\right)^{-\frac{1}{q'}} \left[\int_{a}^{\infty} x^{(1-A_{2})p-1} f^{p}(x) \, dx \right]^{\frac{1}{p}} \\ \times \left[\int_{a}^{\infty} y^{q(A_{2}-\frac{1}{p'})} \left(\log \frac{y}{a} \right)^{\frac{q}{p'}} g^{q}(y) \, dy \right]^{\frac{1}{q}}$$
(8.46)

$$\begin{bmatrix} \int_{a}^{\infty} y^{-q'A_{2}-1} \left(\log \frac{y}{a} \right)^{-\frac{q'}{p'}} (Hf)^{q'}(y) \, dy \end{bmatrix}^{\frac{1}{q'}} \\ \leq \left(q'A_{2} \right)^{-\frac{1}{q'}} \left[\int_{a}^{\infty} x^{(1-A_{2})p-1} f^{p}(x) \, dx \right]^{\frac{1}{p}}$$
(8.47)

hold for all non-negative measurable functions f and g on (a, ∞) , and are equivalent.

Finally, we consider the case of (8.27), that is, a = 0 and $b = \infty$. In that case, the functions *F* and *G* are defined by (8.43) and (8.38) respectively, where $1 - p'A_1 > 0$ and $q'A_2 > 0$. Hence, we have only one possibility described by the following corollary.

Corollary 8.9 Suppose p, q, and λ are as in (2.1) and (2.2), A_1 and A_2 are real parameters such that $p'A_1 < 1$, $q'A_2 > 0$, and H is the operator defined by (8.8). Then the inequalities

$$\int_{0}^{\infty} y^{-\lambda} g(y)(Hf)(y) \, dy \leq \frac{(1 - p'A_1)^{-\frac{1}{p'}}}{(q'A_2)^{\frac{1}{q'}}} \left[\int_{0}^{\infty} x^{(A_1 - A_2)p} f^p(x) \, dx \right]^{\frac{1}{p}} \\ \times \left[\int_{0}^{\infty} y^{(A_2 - A_1)q} g^q(y) \, dy \right]^{\frac{1}{q}}$$
(8.48)

and

$$\left[\int_{0}^{\infty} y^{(A_{1}-A_{2}-\lambda)q'}(Hf)^{q'}(y) \, dy\right]^{\frac{1}{q'}} \leq \frac{(1-p'A_{1})^{-\frac{1}{p'}}}{(q'A_{2})^{\frac{1}{q'}}} \left[\int_{0}^{\infty} x^{(A_{1}-A_{2})p} f^{p}(x) \, dx\right]^{\frac{1}{p}}$$
(8.49)

hold for all non-negative measurable functions f and g on \mathbb{R}_+ , and are equivalent.

Remark 8.6 Some results from this section, considered in conjugate setting, can be found in [53]. Hence, the above relations may be regarded as an extension to non-conjugate setting.

8.4 Further analysis of parameters

We proceed with estimates for some constant factors included in Hardy-type inequalities from the previous section, depending on non-conjugate exponents and on the real parameters A_1 and A_2 . Applying these estimates, we shall get closer to the classical Hardy inequality. More precisely, retaining the notation from the previous section, the estimates

$$\left|1 - \left(\frac{a}{y}\right)^{1 - p'A_1}\right| \le \left|1 - \left(\frac{a}{b}\right)^{1 - p'A_1}\right|, \ y \in (a, b), A_1 \neq \frac{1}{p'},$$
(8.50)

and

$$\left|1 - \left(\frac{x}{b}\right)^{q'A_2}\right| \le \left|1 - \left(\frac{a}{b}\right)^{q'A_2}\right|, \ x \in (a,b), A_2 \neq 0,$$

$$(8.51)$$

where $0 < a < b < \infty$, hold. In addition, assuming that $0 < a < b < \infty$, the estimates

$$\log \frac{b}{x} \le \log \frac{b}{a}, \ x \in (a, b), \tag{8.52}$$

and

$$\log \frac{y}{a} \le \log \frac{b}{a}, \ y \in (a, b), \tag{8.53}$$

are obviously valid for the logarithm function.

Our aim here is to apply the above estimates to the results obtained in Section 8.3. In such a way, we shall simplify these inequalities by obtaining the corresponding constant factors included in the right-hand sides of inequalities. These constant factors will be expressed in terms of the function $l : \mathbb{R} \to \mathbb{R}$, defined by

$$l(\alpha) = \begin{cases} \frac{1 - \left(\frac{a}{b}\right)^{\alpha}}{\alpha}, & \alpha \neq 0, \\ \log \frac{b}{a}, & \alpha = 0, \end{cases}$$
(8.54)

where $0 < a < b < \infty$. Obviously, *l* is a continuous function since $\lim_{\alpha \to 0} l(\alpha) = l(0)$.

Combining Corollary 8.1 and estimates (8.50), (8.51), we obtain the following pair of inequalities.

Corollary 8.10 Suppose p, q, and λ are as in (2.1) and (2.2), $-\infty < a < b < \infty$, A_1 and A_2 are real parameters such that $A_1 \neq \frac{1}{p'}$, $A_2 \neq 0$, and H is the operator defined by (8.8). Then the inequalities

$$\int_{a}^{b} y^{-\lambda} g(y)(Hf)(y) \, dy \le l^{\frac{1}{p'}} (1 - p'A_1) l^{\frac{1}{q'}} (q'A_2) \\ \times \left[\int_{a}^{b} x^{(A_1 - A_2)p} f^p(x) \, dx \right]^{\frac{1}{p}} \left[\int_{a}^{b} y^{(A_2 - A_1)q} g^q(y) \, dy \right]^{\frac{1}{q}}$$
(8.55)

$$\left[\int_{a}^{b} y^{(A_{1}-A_{2}-\lambda)q'}(Hf)^{q'}(y) \, dy\right]^{\frac{1}{q'}} \leq l^{\frac{1}{p'}}(1-p'A_{1})l^{\frac{1}{q'}}(q'A_{2}) \left[\int_{a}^{b} x^{(A_{1}-A_{2})p} f^{p}(x) \, dx\right]^{\frac{1}{p}}$$
(8.56)

hold for all non-negative measurable functions f and g on (a,b), and are equivalent.

Note that inequality (8.56) has a form of the classical Hardy inequality (8.1). Now, let us compare inequalities (8.31) and (8.56). The left-hand side of inequality (8.31) is not less than the corresponding side of (8.56), while the right-hand side of (8.31) is not greater than the corresponding side of (8.56). Thus, we can regard (8.31) as both generalization and refinement of the classical Hardy inequality. The same reasoning will be valid for the remaining results of the Hardy-type in Section 8.3.

Of course, in a similar way, we obtain the results that correspond to Corollaries 8.2, 8.3 and 8.4.

Corollary 8.11 Let $-\infty < a < b < \infty$, let p, q, and λ be as in (2.1) and (2.2), assume that A_1 is a real parameter such that $A_1 \neq \frac{1}{p}$, and let H be the operator defined by (8.8). Then the inequalities

$$\int_{a}^{b} y^{-\lambda} g(y)(Hf)(y) \, dy \leq l^{\frac{1}{p'}} (1 - p'A_1) l^{\frac{1}{q'}}(0) \left[\int_{a}^{b} x^{A_1 p} f^p(x) \, dx \right]^{\frac{1}{p}} \\ \times \left[\int_{a}^{b} y^{-A_1 q} g^q(y) \, dy \right]^{\frac{1}{q}}$$
(8.57)

and

$$\left[\int_{a}^{b} y^{(A_{1}-\lambda)q'}(Hf)^{q'}(y) \, dy\right]^{\frac{1}{q'}} \leq l^{\frac{1}{p'}}(1-p'A_{1})l^{\frac{1}{q'}}(0) \left[\int_{a}^{b} x^{A_{1}p}f^{p}(x) \, dx\right]^{\frac{1}{p}}$$
(8.58)

hold for all non-negative measurable functions f and g on (a,b), and are equivalent.

Corollary 8.12 Suppose p, q, and λ are as in (2.1) and (2.2), $0 < a < b < \infty$, A_2 is a real parameter such that $A_2 \neq 0$, and H is the operator defined by (8.8). Then the inequalities

$$\int_{a}^{b} y^{-\lambda} g(y)(Hf)(y) \, dy \leq l^{\frac{1}{p'}}(0) l^{\frac{1}{q'}}(q'A_2) \left[\int_{a}^{b} x^{(1-A_2)p-1} f^p(x) \, dx \right]^{\frac{1}{p}} \\ \times \left[\int_{a}^{b} y^{(A_2 - \frac{1}{p'})q} g^q(y) \, dy \right]^{\frac{1}{q}}$$
(8.59)

$$\left[\int_{a}^{b} y^{-q'A_{2}-1} (Hf)^{q'}(y) \, dy\right]^{\frac{1}{q'}} \leq l^{\frac{1}{p'}}(0) l^{\frac{1}{q'}}(q'A_{2}) \left[\int_{a}^{b} x^{(1-A_{2})p-1} f^{p}(x) \, dx\right]^{\frac{1}{p}}$$
(8.60)

hold for all non-negative measurable functions f and g on (a,b), and are equivalent.

Corollary 8.13 Let $0 < a < b < \infty$. Suppose p, q, and λ are as in (2.1) and (2.2) and H is the operator defined by (8.8). Then the inequalities

$$\int_{a}^{b} y^{-\lambda} g(y)(Hf)(y) \, dy$$

$$\leq l^{\lambda}(0) \left[\int_{a}^{b} x^{p-1} f^{p}(x) \, dx \right]^{\frac{1}{p}} \left[\int_{a}^{b} y^{-\frac{q}{p'}} g^{q}(y) \, dy \right]^{\frac{1}{q}}$$
(8.61)

and

$$\left[\int_{a}^{b} y^{-1} (Hf)^{q'}(y) \, dy\right]^{\frac{1}{q'}} \le l^{\lambda}(0) \left[\int_{a}^{b} x^{p-1} f^{p}(x) \, dx\right]^{\frac{1}{p}}$$
(8.62)

hold for all non-negative measurable functions f and g on (a,b), and are equivalent.

Remark 8.7 It is easy to see that reverse inequalities are not valid in Corollaries 8.10–8.13.

Finally, utilizing the established estimates, we can also obtain results that correspond to Corollaries 8.5 and 8.7. Since a = 0 or $b = \infty$, we do not need to express the constant factors in terms of the function *l*.

Corollary 8.14 Let $0 < b < \infty$, let p, q, and λ be as in (2.1) and (2.2), assume that A_1 , A_2 are two real parameters such that $p'A_1 < 1$, $q'A_2 > 0$, and let H be the operator defined by (8.8). Then the inequalities

$$\int_{0}^{b} y^{-\lambda} g(y)(Hf)(y) \, dy \leq \frac{(1 - p'A_1)^{-\frac{1}{p'}}}{(q'A_2)^{\frac{1}{q'}}} \left[\int_{0}^{b} x^{(A_1 - A_2)p} f^p(x) \, dx \right]^{\frac{1}{p}} \\ \times \left[\int_{0}^{b} y^{(A_2 - A_1)q} g^q(y) \, dy \right]^{\frac{1}{q}}$$
(8.63)

and

$$\left[\int_{0}^{b} y^{(A_{1}-A_{2}-\lambda)q'}(Hf)^{q'}(y) \, dy\right]^{\frac{1}{q'}} \leq \frac{(1-p'A_{1})^{-\frac{1}{p'}}}{(q'A_{2})^{\frac{1}{q'}}} \left[\int_{0}^{b} x^{(A_{1}-A_{2})p} f^{p}(x) \, dx\right]^{\frac{1}{p}}$$
(8.64)

hold for all non-negative measurable functions f and g on (0,b), and are equivalent.

Remark 8.8 If the parameters *p* and *q* satisfy conditions (2.12) (see Remark 2.2, Chapter 2), then the inequalities (8.63) and (8.64) are reversed.

Corollary 8.15 Suppose p, q, and λ are as in (2.1) and (2.2), $0 < a < \infty$, A_1 and A_2 are real parameters such that $p'A_1 < 1$, $q'A_2 > 0$, and H is the operator defined by (8.8). Then the inequalities

$$\int_{a}^{\infty} y^{-\lambda} g(y)(Hf)(y) \, dy \leq \frac{(1 - p'A_1)^{-\frac{1}{p'}}}{(q'A_2)^{\frac{1}{q'}}} \left[\int_{a}^{\infty} x^{(A_1 - A_2)p} f^p(x) \, dx \right]^{\frac{1}{p}} \\ \times \left[\int_{a}^{\infty} y^{(A_2 - A_1)q} g^q(y) \, dy \right]^{\frac{1}{q}}$$
(8.65)

and

$$\left[\int_{a}^{\infty} y^{(A_{1}-A_{2}-\lambda)q'}(Hf)^{q'}(y) \, dy\right]^{\frac{1}{q'}} \leq \frac{(1-p'A_{1})^{-\frac{1}{p'}}}{(q'A_{2})^{\frac{1}{q'}}} \left[\int_{a}^{\infty} x^{(A_{1}-A_{2})p} f^{p}(x) \, dx\right]^{\frac{1}{p}}$$
(8.66)

hold for all non-negative measurable functions f and g on (a, ∞) , and are equivalent.

Remark 8.9 If p and q are non-conjugate exponents which fulfill conditions (2.13) (see Remark 2.2, Chapter 2), then the inequalities (8.65) and (8.66) are reversed.

8.5 Uniform bounds of constant factors

We investigate here some further estimates for Hardy-type inequalities. First, recall that Corollary 8.10 was obtained from Corollary 8.1 by means of estimates (8.50) and (8.51). On the other hand, we may apply uniform upper bound $1 - t^x \le 1$, $t \in (0, 1)$, $x \ge 0$, to Corollary 8.1. The corresponding result, under some stronger conditions, is contained in the following corollary.

Corollary 8.16 Let $0 < a < b < \infty$, let p, q, and λ be as in (2.1) and (2.2), assume that A_1 , A_2 are two real parameters such that $p'A_1 < 1$, $q'A_2 > 0$, and let H be the operator defined by (8.8). Then the inequalities

$$\int_{a}^{b} y^{-\lambda} g(y)(Hf)(y) \, dy \leq \frac{(1 - p'A_1)^{-\frac{1}{p'}}}{(q'A_2)^{\frac{1}{q'}}} \left[\int_{a}^{b} x^{(A_1 - A_2)p} f^p(x) \, dx \right]^{\frac{1}{p}} \\ \times \left[\int_{a}^{b} y^{(A_2 - A_1)q} g^q(y) \, dy \right]^{\frac{1}{q}}$$
(8.67)

$$\left[\int_{a}^{b} y^{(A_{1}-A_{2}-\lambda)q'}(Hf)^{q'}(y) \, dy\right]^{\frac{1}{q'}} \leq \frac{(1-p'A_{1})^{-\frac{1}{p'}}}{(q'A_{2})^{\frac{1}{q'}}} \left[\int_{a}^{b} x^{(A_{1}-A_{2})p} f^{p}(x) \, dx\right]^{\frac{1}{p}}$$
(8.68)

hold for all non-negative measurable functions f and g on (a,b), and are equivalent.

Remark 8.10 Comparing Corollaries 8.9, 8.14, and 8.15 with Corollary 8.16, we easily conclude that Corollary 8.16 holds also if a = 0 or $b = \infty$.

The constant factor on the right-hand sides of inequalities (8.67) and (8.68) depends on the parameters A_1 and A_2 , while the corresponding integrals are dependent only on the parameter $A = A_1 - A_2$. Hence, it is interesting to consider such constant factor for a fixed value of A. Then $A_2 = A_1 - A$ and the constant factor can be regarded as a function

$$C(A_1) = (1 - p'A_1)^{-\frac{1}{p'}} (q'A_1 - q'A)^{-\frac{1}{q'}}.$$
(8.69)

It is interesting to find the optimal value for the constant factor (8.69). More precisely, depending on the inequality sign, we find maximal or minimal values for this factor. Having in mind Remark 8.2, we have to consider three cases:

1.
$$p, q > 1, \lambda \ge 1$$

In this case we have $A < \frac{1}{p'}$ and $A_1 \in (A, \frac{1}{p'})$, so, we have to find

$$\inf_{A < x < \frac{1}{p'}} C(x) = \inf_{A < x < \frac{1}{p'}} (1 - p'x)^{-\frac{1}{p'}} (q'x - q'A)^{-\frac{1}{q'}}.$$

One easily obtains that C'(x) = 0 if and only if

$$x_0 = \frac{1 + q'A}{p' + q'}.$$
(8.70)

Further, since $x_0 \in (A, \frac{1}{p'})$ and $C''(x_0) > 0$, it follows that C(x) attains its minimum value on the interval $(A, \frac{1}{p'})$ at the point x_0 . Hence, a straightforward computation gives the following value of the minimal constant factor:

$$\inf_{A < x < \frac{1}{p'}} C(x) = C(x_0) = \left(\frac{p'\lambda}{1 - p'A}\right)^{\lambda}.$$

2. $p < 0, q \in (0,1), \lambda \ge 1$

It is easy to see that $A_1 \in (-\infty, \min\{\frac{1}{p'}, A\})$, so we distinguish two cases, depending on the relationship between the parameters *A* and $\frac{1}{p'}$.

If $A < \frac{1}{p'}$, we conclude, by a similar reasoning as in the previous case, that the function C(x) attains its maximal value on the interval $(-\infty, A)$ at the point x_0 defined by (8.70). Finally, since C(A) = 0, we have

$$\sup_{x \le A} C(x) = C(x_0) = \left(\frac{p'\lambda}{1 - p'A}\right)^{\lambda}$$

If $A \ge \frac{1}{p'}$, then the stationary point (8.70) does not belong to the interval $\left(-\infty, \frac{1}{p'}\right)$ and C(x) is strictly increasing on that interval. Further, since $\lim_{x\to \frac{1}{p'}} C(x) = \infty$, there is no upper bound for the constant factor C(x) in this case.

3. $p \in (0,1), q < 0, \lambda \ge 1$

Here, we have to find the optimal value of C(x) on the interval $\left(\max\left\{\frac{1}{p'},A\right\},\infty\right)$. Similarly as above, we have to consider two cases. For $A < \frac{1}{p'}$, it follows that the function C(x) attains its maximal value on the interval $\left(\frac{1}{p'},\infty\right)$ at the point defined by (8.70). Moreover, since $C\left(\frac{1}{p'}\right) = 0$, we have

$$\sup_{x \ge \frac{1}{p'}} C(x) = C(x_0) = \left(\frac{p'\lambda}{1 - p'A}\right)^{\lambda}.$$

If $A \ge \frac{1}{p'}$, then the stationary point (8.70) is not contained in the interval (A, ∞) and C(x) is strictly decreasing on that interval. Since $\lim_{x\to A+} C(x) = \infty$, there is no upper bound for the constant factor C(x) in this case.

According to the previous analysis, we have just proved the following result.

Theorem 8.4 Let $0 < a < b < \infty$, let p, q, and λ be as in (2.1) and (2.2), assume that A is a real parameter such that $A < \frac{1}{p'}$, and let H be the operator defined by (8.8). Then the inequalities

$$\int_{a}^{b} y^{-\lambda} g(y)(Hf)(y) \, dy \leq \left(\frac{p'\lambda}{1-p'A}\right)^{\lambda} \left[\int_{a}^{b} x^{pA} f^{p}(x) \, dx\right]^{\frac{1}{p}} \\ \times \left[\int_{a}^{b} y^{-qA} g^{q}(y) \, dy\right]^{\frac{1}{q}}$$
(8.71)

and

$$\int_{a}^{b} y^{(A-\lambda)q'}(Hf)^{q'}(y) \, dy \bigg]^{\frac{1}{q'}} \le \left(\frac{p'\lambda}{1-p'A}\right)^{\lambda} \left[\int_{a}^{b} x^{pA} f^{p}(x) \, dx\right]^{\frac{1}{p}}$$
(8.72)

hold for all non-negative measurable functions f and g on (a,b), and are equivalent.

Remark 8.11 If a = 0 and p, q, λ are as in (2.12) (see Remark 2.2, Chapter 2), then the inequality signs in (8.71) and (8.72) are reversed. On the other hand, if $b = \infty$ and p, q, λ are as in (2.13) (see Remark 2.2, Chapter 2), the inequality signs in (8.71) and (8.72) are reversed as well.
8.6 Applications

Finally, in the last section, we consider some interesting special cases involving the optimal constant factor in the Hardy-type inequality established in the previous section. Namely, we shall synthesize the methods developed in Sections 8.3, 8.4 and 8.5 for such cases. By virtue of established estimates, we obtain the numerous interpolating inequalities which provide both generalizations and refinements of some recent results, known from the literature.

We can gather the previous discussion in the following two sets of inequalities:

$$\begin{split} \int_{a}^{b} y^{-\lambda} g(y)(Hf)(y) \, dy \\ &\leq \left(\frac{p'\lambda}{1-p'A}\right)^{\lambda} \left\{ \int_{a}^{b} x^{pA} \left[1 - \left(\frac{x}{b}\right)^{\frac{1-p'A}{\lambda p'}}\right]^{\frac{p}{q'}} f^{p}(x) \, dx \right\}^{\frac{1}{p}} \\ &\qquad \times \left\{ \int_{a}^{b} y^{-qA} \left[1 - \left(\frac{a}{y}\right)^{\frac{1-p'A}{\lambda p'}}\right]^{\frac{q}{p'}} g^{q}(y) \, dy \right\}^{\frac{1}{q}} \\ &\leq \left(\frac{p'\lambda}{1-p'A}\right)^{\lambda} \left[1 - \left(\frac{a}{b}\right)^{\frac{1-p'A}{\lambda p'}}\right]^{\lambda} \\ &\qquad \times \left[\int_{a}^{b} x^{pA} f^{p}(x) \, dx\right]^{\frac{1}{p}} \left[\int_{a}^{b} y^{-qA} g^{q}(y) \, dy\right]^{\frac{1}{q}} \\ &\leq \left(\frac{p'\lambda}{1-p'A}\right)^{\lambda} \left[\int_{a}^{b} x^{pA} f^{p}(x) \, dx\right]^{\frac{1}{p}} \left[\int_{a}^{b} y^{-qA} g^{q}(y) \, dy\right]^{\frac{1}{q}} \end{split}$$

$$(8.73)$$

and

$$\begin{split} \int_{a}^{b} y^{(A-\lambda)q'}(Hf)^{q'}(y) \, dy \Big]^{\frac{1}{q'}} \\ &\leq \left(\frac{p'\lambda}{1-p'A}\right)^{\lambda} \left[1 - \left(\frac{a}{b}\right)^{\frac{1-p'A}{\lambda p'}}\right]^{\frac{1}{p'}} \\ &\quad \times \left\{\int_{a}^{b} x^{pA} \left[1 - \left(\frac{x}{b}\right)^{\frac{1-p'A}{\lambda p'}}\right]^{\frac{p}{q'}} f^{p}(x) \, dx\right\}^{\frac{1}{p}} \\ &\leq \left(\frac{p'\lambda}{1-p'A}\right)^{\lambda} \left[1 - \left(\frac{a}{b}\right)^{\frac{1-p'A}{\lambda p'}}\right]^{\lambda} \left[\int_{a}^{b} x^{pA} f^{p}(x) \, dx\right]^{\frac{1}{p}} \end{split}$$

$$\leq \left(\frac{p'\lambda}{1-p'A}\right)^{\lambda} \left[\int_{a}^{b} x^{pA} f^{p}(x) \, dx\right]^{\frac{1}{p}},\tag{8.74}$$

which hold under assumptions of Theorem 8.4. Of course, these sets of inequalities are equivalent and reverse sets of inequalities hold as described in Theorem 8.4. For A = 0, the above sets of inequalities (8.73) and (8.74) reduce respectively to

$$\int_{a}^{b} y^{-\lambda} g(y)(Hf)(y) dy$$

$$\leq (p'\lambda)^{\lambda} \left\{ \int_{a}^{b} \left[1 - \left(\frac{x}{b}\right)^{\frac{1}{\lambda p'}} \right]^{\frac{p}{q'}} f^{p}(x) dx \right\}^{\frac{1}{p}}$$

$$\times \left\{ \int_{a}^{b} \left[1 - \left(\frac{a}{y}\right)^{\frac{1}{\lambda p'}} \right]^{\frac{q}{p'}} g^{q}(y) dy \right\}^{\frac{1}{q}}$$

$$\leq (p'\lambda)^{\lambda} \left[1 - \left(\frac{a}{b}\right)^{\frac{1}{\lambda p'}} \right]^{\lambda} ||f||_{L^{p}} ||g||_{L^{q}} \leq (p'\lambda)^{\lambda} ||f||_{L^{p}} ||g||_{L^{q}}$$
(8.75)

and

$$\begin{split} \left[\int_{a}^{b} y^{-\lambda q'} (Hf)^{q'}(y) \, dy \right]^{\frac{1}{q'}} \\ &\leq \left(p'\lambda \right)^{\lambda} \left[1 - \left(\frac{a}{b}\right)^{\frac{1}{\lambda p'}} \right]^{\frac{1}{p'}} \left\{ \int_{a}^{b} \left[1 - \left(\frac{x}{b}\right)^{\frac{1}{\lambda p'}} \right]^{\frac{p}{q'}} f^{p}(x) \, dx \right\}^{\frac{1}{p}} \\ &\leq \left(p'\lambda \right)^{\lambda} \left[1 - \left(\frac{a}{b}\right)^{\frac{1}{\lambda p'}} \right]^{\lambda} \|f\|_{L^{p}} \leq \left(p'\lambda \right)^{\lambda} \|f\|_{L^{p}}. \end{split}$$
(8.76)

Remark 8.12 It follows easily that inequalities (8.75) and (8.76), with A = 0, are equivalent to inequalities (8.73) and (8.74), with condition $A < \frac{1}{p'}$. Namely, setting

$$a^{1-p'A}, \quad b^{1-p'A}, \quad x^{\frac{1}{1-p'A}-1}f(x^{\frac{1}{1-p'A}}), \quad y^{(1-\lambda)\frac{p'A}{1-p'A}}g(y^{\frac{1}{1-p'A}})$$

in (8.75), respectively instead of a, b, f(x), g(y), and then, applying the variable substitution theorem, the set of inequalities (8.75) become (8.73). So, the case of $A < \frac{1}{p'}$ is equivalent to the case of A = 0. Thus, it is enough to observe the cases with A = 0, since all others follow by suitable substitutions.

Finally, to conclude the chapter, we compare the results obtained in this chapter with some previously known from the literature.

Remark 8.13 Setting a = 0 and $b = \infty$ in (8.76), and isolating the outer expressions, we obtain the inequality

$$\left[\int_0^\infty y^{-\lambda q'} (Hf)^{q'}(y) \, dy\right]^{\frac{1}{q'}} \le \left(p'\lambda\right)^\lambda \|f\|_{L^p},$$

which coincides with Opic's estimate (see [69]). Clearly, for $\lambda = 1$, we obtain the Hardy inequality (8.1) in its original form. Moreover, the above inequality in conjugate form can also be found in Kufner's paper [68]. So, the interpolating sets of inequalities established in this section may be regarded as both generalizations and refinements of the above mentioned results.

Remark 8.14 Considering the parameter $A = \lambda - \frac{k}{q'}$, k > 1, we have $A < \lambda - \frac{1}{q'} = \frac{1}{p'}$. Hence, setting $A = \lambda - \frac{k}{q'}$ in inequalities (8.73) and (8.74), the optimal constant factor established in Theorem 8.4 takes the following value:

$$C = \left(\frac{\lambda q'}{k-1}\right)^{\lambda}.$$
(8.77)

In this setting, inequalities (8.73) and (8.74) provide an extension to non-conjugate case of the corresponding results from already mentioned paper [53]. Moreover, relation (8.74) can be seen as both a refinement and an extension of the corresponding results from [14] and [15]. Namely, in the conjugate case ($p = q', \lambda = 1$) with *C* defined by (8.77), relation (8.74) provides related results from mentioned papers (for example, see [14], Theorem 2, [15], relation (13), and also relation (8.3) at the beginning of this chapter).

Remark 8.15 A unified approach to Hardy-type inequalities with non-conjugate exponents, presented in this chapter, is developed recently in [18].

Chapter 9

Hilbert-type inequalities in the weighted Orlicz spaces

In the previous chapter we studied general Hardy-type inequalities with non-conjugate exponents. However, all derived results were related with the Lebesgue spaces. Nowadays, the Hardy inequality is investigated in more general spaces, for example in Orlicz spaces, Lorenz spaces, rearrangement invariant spaces and their weighted versions, as well as in general Banach ideal spaces. For a comprehensive survey of recent results about the Hardy inequality in Banach function spaces, the reader is referred to [71].

On the other hand, much less attention has been given to Hilbert-type inequalities in such function spaces. Recently, K. Jichang and L. Debnath [43], obtained two-dimensional Hilbert-type inequality in a weighted Orlicz spaces, including a homogeneous kernel. That result will be cited in the next section, after we present basic definitions and properties of weighted Orlicz spaces.

Our main task in this chapter is to establish a multidimensional Hilbert-type inequality in a weighted Orlicz space. In other words, we shall derive the multidimensional inequality in a weighted Orlicz space that corresponds to the classical inequality (1.73) (Theorem 1.18, Chapter 1).

In the next section we present some basic properties of Orlicz spaces, as well as the above mentioned two-dimensional Hilbert-type inequality, obtained by Jichang and Debnath [43]. Further, in Section 9.2 we state and prove the general multidimensional Hilbert-type inequality in weighted Orlicz spaces. A special emphasis is placed on inequalities including a homogeneous kernel with the negative degree of homogeneity. As an application, in Section 9.3 we derive the Hardy-Hilbert-type inequality related to (1.74) (Theorem 1.18, Chapter 1) in some particular cases. Finally, in the last section the general method regarding Orlicz spaces is applied to the case of the weighted Lebesgue spaces.

9.1 Weighted Orlicz spaces and a two-dimensional Hilbert-type inequality

An Orlicz function $\Phi:[0,\infty) \to [0,\infty)$ is a continuous increasing unbounded function with $\Phi(0) = 0$. Convex Orlicz functions are called Young functions. A Young function Φ is called *N*-function if moreover

$$\lim_{x \to 0^+} \frac{\Phi(x)}{x} = 0, \quad \lim_{x \to \infty} \frac{\Phi(x)}{x} = \infty.$$

For an Orlicz function Φ , a σ -finite measure space (Ω, Σ, μ) , and a weight w on Ω , the weighted Orlicz space $L^{\Phi}_{w}(\mu)$ is defined as the space of all classes of μ -measurable functions $f: \Omega \to \mathbb{R}$ such that the modular

$$\rho_{\Phi,w}\left(\frac{f}{\lambda}\right) = \int_{\Omega} \Phi\left(\frac{|f(x)|}{\lambda}\right) w(x) d\mu(x)$$

is finite for some $\lambda > 0$. Moreover, if Φ is a Young function, then the weighted Orlicz space $L^{\Phi}_{w}(\mu)$ becomes Banach function space with the Luxemburg-Nakano norm

$$||f||_{\Phi,w} = \inf\left\{\lambda > 0 : \rho_{\Phi,w}\left(\frac{f}{\lambda}\right) \le 1\right\}.$$
(9.1)

Recall that if $\Phi(x) = x^p$, p > 1, the weighted Orlicz space $L^{\Phi}_w(\mu)$ becomes the weighted Lebesgue space, denoted by $L^p_w(\mu)$, with the norm

$$\|f\|_{L^p_w(\mu)} = \left[\int_{\Omega} w(x) f^p(x) d\mu(x)\right]^{\frac{1}{p}}$$

In addition, if $\Phi(x) = x^p [\log(e+x)]^{\alpha}$, $p \ge 1$, $\alpha > 0$, then, in non-weighted case we obtain the classical Zygmund space.

On the other hand, to Young function Φ one can associate its convex conjugate function Φ^* defined by

$$\Phi^*(y) = \sup_{x \ge 0} \{xy - \Phi(x)\}.$$
(9.2)

It is easy to see that the convex conjugate function Φ^* is also a Young function, as well as $(\Phi^*)^* = \Phi$. Moreover, definition of convex conjugate function provides the so called Young inequality

$$xy \le \Phi(x) + \Phi^*(y), \ x, y \ge 0.$$
 (9.3)

Besides, inverse functions Φ^{-1} and Φ^{*-1} fulfill the following inequalities:

$$x \le \Phi^{-1}(x)\Phi^{*-1}(x) \le 2x, \ x \ge 0.$$
(9.4)

The right inequality in (9.4) follows immediately from (9.3), and for the other inequality the reader is referred to [8] or [91].

Further, throughout this chapter the Young function Φ is assumed to be submultiplicative (see [90]), that is,

$$\Phi(xy) \le \Phi(x)\Phi(y), \ x, y \ge 0. \tag{9.5}$$

Condition (9.5) can be regarded as an extension of Orlicz Δ_2 condition for Young function Φ , that is, there exists a positive constant *C* such that $\Phi(2x) \leq C\Phi(x)$, $x \geq 0$. For more details about standard theory of Orlicz spaces the reader is referred to [114] and [184], while the weighted theory is developed in the monograph [51].

Now, we are ready to state the two-dimensional Hilbert-type inequality in the weighted Orlicz spaces, obtained by Jichang and Debnath [43].

Theorem 9.1 Suppose $\Phi, \Phi^* : \mathbb{R}_+ \to \mathbb{R}_+$ are submultiplicative conjugate Young functions and $K : \mathbb{R}^2_+ \to \mathbb{R}$ is a non-negative homogeneous function of degree -s, s > 0. Further, let $f, g : \mathbb{R}_+ \to \mathbb{R}$ be non-negative functions such that $||f||_{\Phi,w} > 0$ and $||g||_{\Phi^*,w} > 0$, where $w(x) = x^{1-s}$. Then,

$$\int_{\mathbb{R}^2_+} K(x,y) f(x) g(y) dx dy \le (c_1 + c_2) \|f\|_{\Phi,w} \|g\|_{\Phi^*,w},$$
(9.6)

where

$$c_1 = \int_{\mathbb{R}_+} K(1, u) \Phi^{*-1}\left(\frac{1}{u}\right) du < \infty,$$

$$c_2 = \int_{\mathbb{R}_+} K(u, 1) \Phi^*\left(\frac{1}{\Phi^{-1}\left(\Phi^{*-1}(u)\right)}\right) du < \infty.$$

9.2 Multidimensional Hilbert-type inequality

Concerning the statement of Theorem 9.1, we see that the authors deal with a pair of submultiplicative conjugate functions. Moreover, in the proof of Theorem 9.1 (see [43]), they used properties (9.3) and (9.4), which hold for a pair of conjugate Young functions.

Guided by that idea, we establish here a class of Young functions fulfilling the above mentioned conditions. In such a way we shall naturally extend Hilbert-type inequality (9.6) to a multidimensional case, and even more, formulate the corresponding result for an arbitrary measurable kernel.

More precisely, we assume that $\Phi_i : [0, \infty) \to [0, \infty)$, i = 1, 2, ..., n, are Young functions satisfying the following inequality:

$$\alpha \prod_{i=1}^{n} x_i \le \sum_{i=1}^{n} \Phi_i(x_i), \quad x_i \ge 0, \ i = 1, 2, \dots, n,$$
(9.7)

where $\alpha \ge 1$. In addition, the following inequality will be assumed to hold for inverses of the above Young functions:

$$x \le \prod_{i=1}^{n} \Phi_i^{-1}(x), \quad x \ge 0.$$
 (9.8)

Now we are ready to state and prove the general result, that is, the multidimensional Hilbert-type inequality for an arbitrary measurable kernel.

Theorem 9.2 Suppose $\Phi_i : \mathbb{R}_+ \to \mathbb{R}_+$, i = 1, 2, ..., n, are submultiplicative Young functions satisfying conditions (9.7) and (9.8). Let $(\Omega_i, \Sigma_i, \mu_i)$, i = 1, 2, ..., n, be σ -finite measure spaces, let $K : \prod_{i=1}^n \Omega_i \to \mathbb{R}$ be a non-negative measurable function, and let $\varphi_{ij} : \Omega_i \times \Omega_j \to \mathbb{R}$, $i \neq j$, be non-negative measurable functions satisfying the condition

$$\prod_{\substack{i,j=1\\i\neq j}}^{n} \varphi_{ij}\left(x_{i}, x_{j}\right) = 1.$$
(9.9)

Further, suppose $F_i : \Omega_i \to \mathbb{R}$ *are defined by*

$$F_i(x_i) = \int_{\prod_{\substack{j=1\\j\neq i}}^n \Omega_j} K(x_1, x_2, \dots, x_n) \prod_{\substack{j=1\\j\neq i}}^n \left[\Phi_i\left(\varphi_{ij}(x_i, x_j)\right) d\mu_j(x_j) \right].$$
(9.10)

If $f_i : \Omega_i \to \mathbb{R}$ are non-negative functions such that $f_i \in L_{F_i}^{\Phi_i}(\mu_i)$ and $||f_i||_{\Phi_i,F_i} > 0$, then the following inequality holds:

$$\int_{\prod_{i=1}^{n} \Omega_{i}} K(x_{1}, x_{2}, \dots, x_{n}) \left[\prod_{i=1}^{n} f_{i}(x_{i}) d\mu_{i}(x_{i}) \right] \leq \frac{n}{\alpha} \prod_{i=1}^{n} \|f_{i}\|_{\Phi_{i}, F_{i}}.$$
 (9.11)

Proof. Applying property (9.8) to the kernel $K : \prod_{i=1}^{n} \Omega_i \to \mathbb{R}$, we obtain that

$$K(x_1, x_2, \dots, x_n) \le \prod_{i=1}^n \Phi_i^{-1} [K(x_1, x_2, \dots, x_n)]$$

Consequently, considering functions $\varphi_{ij} : \Omega_i \times \Omega_j \to \mathbb{R}$ fulfilling relation (9.9), we obtain the following inequalities

$$\begin{split} &\int_{\prod_{i=1}^{n}\Omega_{i}}K(x_{1},x_{2},\ldots,x_{n})\left[\prod_{i=1}^{n}f_{i}(x_{i})d\mu_{i}(x_{i})\right] \\ &\leq \int_{\prod_{i=1}^{n}\Omega_{i}}\prod_{i=1}^{n}\left[\Phi_{i}^{-1}\left[K(x_{1},x_{2},\ldots,x_{n})\right]f_{i}(x_{i})\prod_{\substack{j=1\\j\neq i}}^{n}\varphi_{ij}(x_{i},x_{j})\right]\prod_{j=1}^{n}d\mu_{j}(x_{j}) \\ &\leq \frac{1}{\alpha}\sum_{i=1}^{n}\int_{\prod_{i=1}^{n}\Omega_{i}}\Phi_{i}\left[\Phi_{i}^{-1}\left[K(x_{1},x_{2},\ldots,x_{n})\right]f_{i}(x_{i})\prod_{\substack{j=1\\j\neq i}}^{n}\varphi_{ij}(x_{i},x_{j})\right]\prod_{j=1}^{n}d\mu_{j}(x_{j}) \end{split}$$

where the second inequality sign holds due to relation (9.7). Note that in the previous relation we write $\prod_{j=1}^{n} d\mu_j(x_j) = d\mu_1(x_1)d\mu_2(x_2)\dots d\mu_n(x_n)$.

Further, since the Young functions $\Phi_i : \mathbb{R}_+ \to \mathbb{R}_+$, i = 1, 2, ..., n, are submultiplicative, the Fubini theorem and definition (9.10) of functions $F_i : \Omega_i \to \mathbb{R}$ yield another inequality:

$$\begin{split} &\frac{1}{\alpha} \sum_{i=1}^{n} \int_{\prod_{i=1}^{n} \Omega_{i}} \Phi_{i} \left[\Phi_{i}^{-1} \left[K(x_{1}, x_{2}, \dots, x_{n}) \right] f_{i}(x_{i}) \prod_{\substack{j=1\\j \neq i}}^{n} \varphi_{ij}(x_{i}, x_{j}) \right] \prod_{j=1}^{n} d\mu_{j}(x_{j}) \\ &\leq \frac{1}{\alpha} \sum_{i=1}^{n} \int_{\prod_{i=1}^{n} \Omega_{i}} K(x_{1}, x_{2}, \dots, x_{n}) \Phi_{i}\left(f_{i}(x_{i})\right) \left[\prod_{\substack{j=1\\j \neq i}}^{n} \Phi_{i}\left(\varphi_{ij}(x_{i}, x_{j})\right) d\mu_{j}(x_{j}) \right] d\mu_{i}(x_{i}) \\ &= \frac{1}{\alpha} \sum_{i=1}^{n} \int_{\Omega_{i}} \Phi_{i}\left(f_{i}(x_{i})\right) \\ &\qquad \times \left[\int_{\prod_{\substack{j=1\\j \neq i}}^{n} \Omega_{j}} K(x_{1}, x_{2}, \dots, x_{n}) \prod_{\substack{j=1\\j \neq i}}^{n} \left[\Phi_{i}\left(\varphi_{ij}(x_{i}, x_{j})\right) d\mu_{j}(x_{j}) \right] \right] d\mu_{i}(x_{i}) \\ &= \frac{1}{\alpha} \sum_{i=1}^{n} \int_{\Omega_{i}} \Phi_{i}\left(f_{i}(x_{i})\right) F_{i}(x_{i}) d\mu_{i}(x_{i}). \end{split}$$

Hence, we obtain

$$\int_{\prod_{i=1}^{n}\Omega_{i}} K(x_{1}, x_{2}, \dots, x_{n}) \left[\prod_{i=1}^{n} f_{i}(x_{i}) d\mu_{i}(x_{i}) \right]$$

$$\leq \frac{1}{\alpha} \sum_{i=1}^{n} \int_{\Omega_{i}} \Phi_{i}\left(f_{i}(x_{i})\right) F_{i}(x_{i}) d\mu_{i}(x_{i}).$$
(9.12)

Now, replacing functions f_i in (9.12) respectively with $f_i/||f_i||_{\Phi_i,F_i}$, we have

$$\int_{\prod_{i=1}^{n}\Omega_{i}} K(x_{1},x_{2},\ldots,x_{n}) \left[\prod_{i=1}^{n} \frac{f_{i}(x_{i})}{\|f_{i}\| \Phi_{i},F_{i}} d\mu_{i}(x_{i})\right]$$
$$\leq \frac{1}{\alpha} \sum_{i=1}^{n} \int_{\Omega_{i}} \Phi_{i}\left(\frac{f_{i}(x_{i})}{\|f_{i}\| \Phi_{i},F_{i}}\right) F_{i}(x_{i}) d\mu_{i}(x_{i}).$$

On the other hand, utilizing the definition of the Luxemburg-Nakano norm, it follows that

$$\int_{\Omega_i} \Phi_i\left(\frac{f_i(x_i)}{\|f_i\|_{\Phi_i,F_i}}\right) F_i(x_i) d\mu_i(x_i) \le 1, \ i=1,2,\ldots,n,$$

which yields

$$\int_{\prod_{i=1}^n \Omega_i} K(x_1, x_2, \dots, x_n) \left[\prod_{i=1}^n \frac{f_i(x_i)}{\|f_i\|_{\Phi_i, F_i}} d\mu_i(x_i) \right] \leq \frac{n}{\alpha},$$

that is, (9.11), as required.

Our first application of Theorem 9.2 refers to a homogeneous kernel of the negative degree, together with the Lebesgue measures $d\mu_i(x_i) = dx_i$, i = 1, 2, ..., n, on \mathbb{R}_+ . In that case the weighted Orlicz space $L_w^{\Phi}(\mu)$ will simply be denoted by L_w^{Φ} . Moreover, we shall deal with a more suitable form of the non-negative functions $\varphi_{ij} : \Omega_i \times \Omega_j \to \mathbb{R}$, $i \neq j$, defined in the previous theorem. In described setting, we have the following result.

Corollary 9.1 Suppose $\Phi_i : \mathbb{R}_+ \to \mathbb{R}_+$, i = 1, 2, ..., n, are submultiplicative Young functions fulfilling conditions (9.7) and (9.8). Let $K : \mathbb{R}^n_+ \to \mathbb{R}$ be a non-negative homogeneous function of degree -s, s > 0, and let $h_{ij} : \mathbb{R}_+ \to \mathbb{R}$, $i \neq j$, be non-negative measurable functions satisfying condition

$$\prod_{\substack{i,j=1\\i\neq j}}^{n} h_{ij}\left(\frac{x_j}{x_i}\right) = 1.$$
(9.13)

Further, assume that C_i , i = 1, 2, ..., n, are real constants defined by

$$C_{i} = \int_{\mathbb{R}^{n-1}_{+}} K(u_{1}, \dots u_{i-1}, 1, u_{i+1}, \dots, u_{n}) \prod_{\substack{j=1\\j \neq i}}^{n} \left[\Phi_{i}(h_{ij}(u_{j})) du_{j} \right],$$
(9.14)

and $w : \mathbb{R}_+ \to \mathbb{R}$ is the weight function, defined by $w(x) = x^{n-s-1}$.

If $f_i : \mathbb{R}_+ \to \mathbb{R}$ are non-negative functions such that $f_i \in L_{C_iw}^{\Phi_i}$ and $||f_i||_{\Phi_i, C_iw} > 0$, i = 1, 2, ..., n, then the following inequality holds:

$$\int_{\mathbb{R}^{n}_{+}} K(x_{1}, x_{2}, \dots, x_{n}) \left[\prod_{i=1}^{n} f_{i}(x_{i}) d\mu_{i}(x_{i}) \right] \leq \frac{n}{\alpha} \prod_{i=1}^{n} \|f_{i}\|_{\Phi_{i}, C_{i} w}.$$
(9.15)

Proof. We utilize Theorem 9.2 with the functions $\varphi_{ij} : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ defined by

$$\varphi_{ij}(x_i, x_j) = h_{ij}\left(\frac{x_j}{x_i}\right), \quad i \neq j$$

In this setting, the functions $F_i : \mathbb{R}_+ \to \mathbb{R}$ can be rewritten in the following form:

$$F_i(x_i) = \int_{\mathbb{R}^{n-1}_+} K(x_1, \dots, x_n) \prod_{\substack{j=1\\j\neq i}}^n \left[\Phi_i\left(h_{ij}\left(\frac{x_j}{x_i}\right)\right) dx_j \right]$$

Now, taking into account substitutions $x_j = x_i u_j$, j = 1, ..., i - 1, i + 1, ..., n, and making use of the homogeneity of the kernel $K : \mathbb{R}^n_+ \to \mathbb{R}$, we have

$$\begin{split} F_{i}(x_{i}) &= \int_{\mathbb{R}^{n-1}_{+}} K(x_{i}u_{1}, \dots, x_{i}u_{i-1}, x_{i}, x_{i}u_{i+1}, \dots, x_{i}u_{n}) \prod_{\substack{j=1\\j\neq i}}^{n} \left[\Phi_{i}\left(h_{ij}(u_{j})\right) x_{i}^{n-1} du_{j} \right] \\ &= \int_{\mathbb{R}^{n-1}_{+}} x_{i}^{-s} K(u_{1}, \dots, u_{i-1}, 1, u_{i+1}, \dots, u_{n}) \prod_{\substack{j=1\\j\neq i}}^{n} \left[\Phi_{i}\left(h_{ij}(u_{j})\right) x_{i}^{n-1} du_{j} \right] \\ &= x_{i}^{n-s-1} \int_{\mathbb{R}^{n-1}_{+}} K(u_{1}, \dots, u_{i-1}, 1, u_{i+1}, \dots, u_{n}) \prod_{\substack{j=1\\j\neq i}}^{n} \left[\Phi_{i}\left(h_{ij}(u_{j})\right) du_{j} \right] \\ &= C_{i} w(x_{i}), \end{split}$$

which completes the proof.

Remark 9.1 It is not hard to find non-negative functions $h_{ij} : \mathbb{R}_+ \to \mathbb{R}, i \neq j, i, j \in \{1, 2, ..., n\}$, fulfilling relation (9.13). Namely, if $\beta_{ij}, i \neq j, i, j \in \{1, 2, ..., n\}$, are positive real numbers satisfying

$$\sum_{\substack{i=1\\i\neq k}}^{n} \beta_{ik} = \sum_{\substack{j=1\\j\neq k}}^{n} \beta_{kj}, \quad k = 1, 2, \dots, n,$$
(9.16)

then the functions $h_{ij}(t) = t^{\beta_{ij}}, i \neq j$, obviously fulfill (9.13), since

$$\prod_{\substack{i,j=1\\i\neq j}}^n \left(\frac{x_j}{x_i}\right)^{\beta_{ij}} = 1.$$

In the previous corollary referring to a homogeneous kernel of degree -s, s > 0, we obtained the Hilbert-type inequality for weighted Orlicz spaces where the weight functions were multiples of the particular weight $w(x) = x^{n-s-1}$. On the other hand, utilizing the same method as in the proof of Theorem 9.2, one can obtain the inequality which includes the weighted Orlicz spaces with the same weight function. This is the content of the following theorem.

Theorem 9.3 Under the assumptions of Corollary 9.1, inequality

$$\int_{\mathbb{R}^{n}_{+}} K(x_{1}, x_{2}, \dots, x_{n}) \left[\prod_{i=1}^{n} f_{i}(x_{i}) d\mu_{i}(x_{i}) \right] \leq \frac{\sum_{i=1}^{n} C_{i}}{\alpha} \prod_{i=1}^{n} \|f_{i}\|_{\Phi_{i}, w}$$
(9.17)

holds for all non-negative functions $f_i : \mathbb{R}_+ \to \mathbb{R}$ such that $f_i \in L_w^{\Phi_i}$ and $||f_i||_{\Phi_i,w} > 0$, i = 1, 2, ..., n.

Proof. We follow the same procedure as in the proof of Theorem 9.2 and take into account the specific form of functions $F_i : \mathbb{R}_+ \to \mathbb{R}$ deduced in Corollary 9.1, that is $F_i(x_i) = C_i w(x_i), i = 1, 2, ..., n$. In this setting, inequality (9.12) can be rewritten as

$$\int_{\mathbb{R}^n_+} K(x_1, x_2, \dots, x_n) \left[\prod_{i=1}^n f_i(x_i) d\mu_i(x_i) \right] \leq \frac{1}{\alpha} \sum_{i=1}^n C_i \int_{\mathbb{R}^n} \Phi_i\left(f_i(x_i)\right) w(x_i) dx_i$$

Now, replacing functions f_i in the above inequality respectively with $f_i/||f_i||_{\Phi_i,w}$, i = 1, 2, ..., n, we have

$$\int_{\mathbb{R}^n_+} K(x_1, x_2, \dots, x_n) \left[\prod_{i=1}^n \frac{f_i(x_i)}{\|f_i\|_{\Phi_i, w}} dx_i \right] \le \frac{1}{\alpha} \sum_{i=1}^n C_i \int_{\mathbb{R}^n} \Phi_i \left(\frac{f_i(x_i)}{\|f_i\|_{\Phi_i, w}} \right) w(x_i) dx_i$$

Finally, since

$$\int_{\mathbb{R}_+} \Phi_i\left(\frac{f_i(x_i)}{\|f_i\|_{\Phi_i,w}}\right) w(x_i) dx_i \leq 1,$$

definition of the Luxemburg-Nakano norm provides the inequality

$$\int_{\mathbb{R}^n_+} K(x_1, x_2, \dots, x_n) \left[\prod_{i=1}^n \frac{f_i(x_i)}{\|f_i\|_{\Phi_{i,w}}} dx_i \right] \leq \frac{\sum_{i=1}^n C_i}{\alpha},$$

and the proof is completed.

Remark 9.2 If n = 2, then the constants C_1 and C_2 included in the inequality (9.17) reduce to

$$C_1 = \int_{\mathbb{R}_+} K(1, u_2) \Phi_1(h_{12}(u_2)) du_2 \quad \text{and} \quad C_2 = \int_{\mathbb{R}_+} K(u_1, 1) \Phi_2(h_{21}(u_1)) du_1.$$

Now, considering a pair of conjugate functions, that is, $\Phi_1 = \Phi$ and $\Phi_2 = \Phi^*$, and defining $h_{12}(u_2) = \Phi^{-1}(\Phi^{*-1}(\frac{1}{u_2}))$, condition (9.13) can be rewritten in the form

$$\Phi^{-1}\left(\Phi^{*-1}\left(u_{1}\right)\right)h_{21}\left(u_{1}\right)=1, \quad u_{1}=\frac{x_{1}}{x_{2}}>0,$$

yielding an explicit formula for the function h_{21} :

$$h_{21}(u_1) = \frac{1}{\Phi^{-1}(\Phi^{*-1}(u_1))}.$$

Thus, if n = 2, $\alpha = 1$ and $h_{12}(u_2) = \Phi^{-1}(\Phi^{*-1}(\frac{1}{u_2}))$, inequality (9.17) coincides with inequality (9.6). Therefore, inequality (9.17) may be regarded as a multidimensional extension of (9.6).

9.3 A version of Hardy-Hilbert-type inequality

In this section we derive Hardy-Hilbert-type inequalities associated to Hilbert-type inequalities from the previous section, in a particular case. Namely, in the sequel we obtain Hardy-Hilbert versions of inequalities (9.11), (9.15), and (9.17) assuming that one of Young functions provides the weighted Lebesgue space.

A Hardy-Hilbert-type inequality that corresponds to (9.11) in the above described setting is a content of the following result.

Corollary 9.2 Let $\frac{1}{r} + \frac{1}{r'} = 1$, r > 1, and let $\Phi_i : \mathbb{R}_+ \to \mathbb{R}_+$, i = 1, 2, ..., n, be submultiplicative Young functions satisfying conditions (9.7) and (9.8), where $\Phi_n(x) = x^{r'}$. Further, suppose $(\Omega_i, \Sigma_i, \mu_i)$, i = 1, 2, ..., n, are σ -finite measure spaces, $K : \prod_{i=1}^n \Omega_i \to \mathbb{R}$ is a nonnegative measurable function, and $F_i : \Omega_i \to \mathbb{R}$, i = 1, 2, ..., n, are defined by (9.10).

If $f_i : \Omega_i \to \mathbb{R}$, i = 1, 2, ..., n-1, are non-negative functions such that $f_i \in L_{F_i}^{\Phi_i}(\mu_i)$ and $||f_i||_{\Phi_i, F_i} > 0$, then

$$\left[\int_{\Omega_n} F_n^{1-r}(x_n) \left[\int_{\prod_{i=1}^{n-1} \Omega_i} K(x_1, x_2, \dots, x_n) \prod_{i=1}^{n-1} f_i(x_i) d\mu_i(x_i)\right]^r d\mu_n(x_n)\right]^{\frac{1}{r}} \le \frac{n}{\alpha} \prod_{i=1}^{n-1} ||f_i||_{\Phi_i, F_i},$$
(9.18)

provided the integrals on the left-hand side of the inequality converge.

Proof. Let *I* denote the left-hand side of (9.18). If we define the function $f_n : \Omega_n \to \mathbb{R}$ by

$$f_n(x_n) = F_n^{1-r}(x_n) \left[\int_{\prod_{i=1}^{n-1} \Omega_i} K(x_1, x_2, \dots, x_n) \prod_{i=1}^{n-1} f_i(x_i) d\mu_i(x_i) \right]^{r-1},$$
(9.19)

then, utilizing the Fubini theorem, the *r*-th power of *I* can be rewritten as

$$I^{r} = \int_{\prod_{i=1}^{n} \Omega_{i}} K(x_{1}, x_{2}, \dots, x_{n}) \prod_{i=1}^{n} f_{i}(x_{i}) d\mu_{i}(x_{i}).$$

Now, the Hilbert-type inequality (9.11) yields

$$I^{r} \leq \frac{n}{\alpha} \prod_{i=1}^{n} \|f_{i}\|_{\Phi_{i}, F_{i}}.$$
(9.20)

On the other hand, the Young function $\Phi_n(x) = x^{r'}$ provides the corresponding weighted Lebesgue space $L_{F_n}^{r'}(\mu_n)$. Moreover, taking into account (9.19), we obtain

$$||f_n||_{\Phi_n,F_n} = \left[\int_{\Omega_n} f_n^{r'}(x_n)F_n(x_n)d\mu_n(x_n)\right]^{\frac{1}{r'}} = I^{\frac{r}{r'}}.$$
(9.21)

Finally, relations (9.20) and (9.21) yield inequality

$$I^{r-\frac{r}{r'}} \leq \frac{n}{\alpha} \prod_{i=1}^{n-1} \|f_i\|_{\Phi_i,F_i},$$

that is, (9.18), since $r - \frac{r}{r'} = 1$.

The following two corollaries refer to homogeneous kernels. Namely, as a special case, we obtain Hardy-Hilbert-type inequalities that correspond to Hilbert-type inequalities (9.15) and (9.17).

Corollary 9.3 Let $\frac{1}{r} + \frac{1}{r'} = 1$, r > 1, and let $\Phi_i : \mathbb{R}_+ \to \mathbb{R}_+$, i = 1, 2, ..., n, be submultiplicative Young functions fulfilling conditions (9.7) and (9.8), where $\Phi_n(x) = x^{r'}$. Let

 $K : \mathbb{R}^n_+ \to \mathbb{R}$ be a non-negative homogeneous function of degree -s, s > 0, and let C_i , i = 1, 2, ..., n, be real constants defined by (9.14).

If $f_i : \mathbb{R}_+ \to \mathbb{R}$, i = 1, 2, ..., n-1, are non-negative functions such that $f_i \in L_{C_iw}^{\Phi_i}$ and $||f_i||_{\Phi_i, C_iw} > 0$, where $w(x) = x^{n-s-1}$, then

$$\left[\int_{\mathbb{R}_{+}} x_{n}^{(n-1-s)(1-r)} \left[\int_{\mathbb{R}_{+}^{n-1}} K(x_{1}, x_{2}, \dots, x_{n}) \prod_{i=1}^{n-1} f_{i}(x_{i}) dx_{i} \right]^{r} dx_{n} \right]^{\frac{1}{r}}$$

$$\leq \frac{n C_{n}^{1/r'}}{\alpha} \prod_{i=1}^{n-1} ||f_{i}||_{\Phi_{i}, C_{i}w},$$
(9.22)

provided the integrals on the left-hand side of the inequality converge.

Corollary 9.4 Under the assumptions of Corollary 9.3, the inequality

$$\left[\int_{\mathbb{R}_{+}} x_{n}^{(n-1-s)(1-r)} \left[\int_{\mathbb{R}_{+}^{n-1}} K(x_{1}, x_{2}, \dots, x_{n}) \prod_{i=1}^{n-1} f_{i}(x_{i}) dx_{i} \right]^{r} dx_{n} \right]^{\frac{1}{r}}$$

$$\leq \frac{\sum_{i=1}^{n} C_{i}}{\alpha} \prod_{i=1}^{n-1} ||f_{i}||_{\Phi_{i}, w}$$

$$(9.23)$$

holds for all non-negative functions $f_i : \mathbb{R}_+ \to \mathbb{R}$ such that $f_i \in L_w^{\Phi_i}$ and $||f_i||_{\Phi_i,w} > 0$, i = 1, 2, ..., n-1.

9.4 Some examples in the weighted Lebesgue spaces

In order to conclude this chapter, we provide here some remarks about reduction to the case of the weighted Lebesgue spaces. More precisely, we consider the method developed in Section 9.2 in the case of the weighted Lebesgue spaces.

For that sake, we assume that Φ_i are Young functions defined by $\Phi_i(x_i) = x_i^{p_i}$, i = 1, 2, ..., n, where $p_i > 1, i = 1, 2, ..., n$, are conjugate exponents, that is, $\sum_{i=1}^{n} \frac{1}{p_i} = 1$.

The above power functions define the appropriate weighted Lebesgue spaces. Moreover, since $p_i > 1$, i = 1, 2, ..., n, are conjugate exponents, the classical Young inequality implies

$$\sum_{i=1}^{n} x_i^{p_i} \ge \prod_{i=1}^{n} p_i^{\frac{1}{p_i}} \prod_{i=1}^{n} x_i,$$

and hence, according to condition (9.7), we can take $\alpha = \prod_{i=1}^{n} p_i^{\frac{1}{p_i}}$. In addition, since the

inverses of Young functions $\Phi_i(x_i) = x_i^{p_i}$ are respectively $\Phi_i^{-1}(x_i) = x_i^{\frac{1}{p_i}}$, we have

$$\prod_{i=1}^{n} \Phi_{i}^{-1}(x) = \prod_{i=1}^{n} x^{\frac{1}{p_{i}}} = x^{\sum_{i=1}^{n} \frac{1}{p_{i}}} = x,$$

which means that condition (9.8) is fulfilled as well. Finally, the above power functions are multiplicative, so the assumptions as in Section 9.2 are also fulfilled.

Remark 9.3 Regarding the above setting, inequality (9.15) can be reduced to a form which includes Lebesgue spaces with the same weight function, as inequality (9.17). Namely, using the notation from Corollary 9.1 and taking into account the above power Young functions, we have

$$\|f_i\|_{\Phi_i, C_i w} = \|f_i\|_{L^{p_i}_{C_i w}} = \left[\int_{\mathbb{R}_+} f_i^{p_i}(x_i) C_i w(x_i) dx_i\right]^{\frac{1}{p_i}} = C_i^{\frac{1}{p_i}} \|f_i\|_{L^{p_i}_w},$$

for i = 1, 2, ..., n. In such a way we obtain an inequality related to (9.17), but with a different constant factor. Moreover, in described setting, inequalities (9.15) and (9.17) yield the inequality

$$\int_{\mathbb{R}^{n}_{+}} K(x_{1}, x_{2}, \dots, x_{n}) \prod_{i=1}^{n} f_{i}(x_{i}) dx_{i}$$

$$\leq \min\left\{\frac{n \prod_{i=1}^{n} C_{i}^{\frac{1}{p_{i}}}}{\prod_{i=1}^{n} p_{i}^{\frac{1}{p_{i}}}}, \frac{\sum_{i=1}^{n} C_{i}}{\prod_{i=1}^{n} p_{i}^{\frac{1}{p_{i}}}}\right\} \prod_{i=1}^{n} \|f_{i}\|_{L_{w}^{p_{i}}}.$$
(9.24)

Note that we cannot decide which constant factor is smaller.

Remark 9.4 We know from previous sections that the constant factors C_i involved in inequality (9.24) can be explicitly computed for some particular choices of the kernel K and the functions $h_{ij} : \mathbb{R}_+ \to \mathbb{R}, i \neq j, i, j \in \{1, 2, ..., n\}$, fulfilling relation (9.13).

For example, considering inequality (9.24) with the kernel $K(x_1, x_2, ..., x_n) = (\sum_{i=1}^n x_i)^{-s}$, s > 0, and the power functions $h_{ij}(t) = t^{\beta_{ij}}$, $i \neq j$, where the parameters β_{ij} fulfill relations (9.16), the above constant factors can be rewritten as

$$C_{i} = \int_{\mathbb{R}^{n-1}_{+}} \frac{\prod_{j=1}^{n} u_{j}^{p;p_{ij}} du_{j}}{\left(1 + \sum_{j\neq i}^{n} u_{j}\right)^{s}}.$$

Now, taking into account Lemma 1.3 (Section 1.6, Chapter 1), the above integral can be expressed in terms of the usual Gamma function, that is,

$$C_{i} = \frac{\Gamma\left(s - n + 1 - p_{i}\sum_{\substack{j=1\\j\neq i}}^{n}\beta_{ij}\right)\prod_{\substack{j=1\\j\neq i}}^{n}\Gamma(p_{i}\beta_{ij} + 1)}{\Gamma(s)},$$

provided that $s - n + 1 - p_i \sum_{\substack{j=1 \ j \neq i}}^n \beta_{ij} > 0$ and $p_i \beta_{ij} + 1 > 0, j \neq i$.

Remark 9.5 Multidimensional Hilbert-type inequalities in weighted Orlicz spaces, introduced in this chapter, are derived recently in [64].

Chapter 10

Some particular inequalities

In this book we have established a unified treatment of Hilbert-type inequalities in both conjugate and non-conjugate case. We have derived numerous inequalities involving diverse choices of function spaces, sets of integration, kernels and weight functions. We have also presented several methods for refinements of Hilbert-type inequalities.

Finally, in this last section we review some particular results, interesting on its own right, which are closely connected with the theory exposed in this book. More precisely, we give here some related Hilbert-type inequalities as well as some other refinements of Hilbert-type inequalities known from the literature. The following recent results are cited without proofs and are listed in the chronological order. For more details, the reader should consult the corresponding literature.

10.1. G. Mingzhe [95], 1997.

If $(a_n)_{n\in\mathbb{N}_0}$ and $(b_n)_{n\in\mathbb{N}_0}$ are non-negative sequences such that $\sum_{n=0}^{\infty} a_n^2 < \infty$ and $\sum_{n=0}^{\infty} b_n^2 < \infty$, then

$$\sum_{n=0}^{\infty} \sum_{n=0}^{\infty} \frac{a_m b_n}{m+n+1} \le \left[\sum_{n=0}^{\infty} \omega(n) a_n^2\right]^{\frac{1}{2}} \left[\sum_{n=0}^{\infty} \omega(n) b_n^2\right]^{\frac{1}{2}},$$
(10.1)

where the weight coefficient ω is defined by $\omega(n) = \pi - \frac{\theta(n)}{\sqrt{2n+1}}$ and

$$\theta(n) = 2\sqrt{2n+1}\arctan\left(\frac{1}{\sqrt{2n+1}}\right) - \frac{2n+1}{n+1} > 0$$

for $n \ge 0$. The equality sign in (10.1) holds if and only if $(a_n)_{n \in \mathbb{N}}$ or $(b_n)_{n \in \mathbb{N}}$ is a zero-sequence.

10.2. G. Mingzhe, B. Yang [97], 1998.

Let $q \ge p > 1$ be conjugate exponents. If $0 < \sum_{n=1}^{\infty} a_n^p < \infty$ and $0 < \sum_{n=1}^{\infty} b_n^q < \infty$, then

$$\sum_{m=1}^{\infty}\sum_{n=1}^{\infty}\frac{a_mb_n}{m+n} < \left[\sum_{n=1}^{\infty}\left(\frac{\pi}{\sin\frac{\pi}{p}} - \frac{\lambda}{n^{\frac{1}{p}}}\right)a_n^p\right]^{\frac{1}{p}} \left[\sum_{n=1}^{\infty}\left(\frac{\pi}{\sin\frac{\pi}{p}} - \frac{\lambda}{n^{\frac{1}{q}}}\right)b_n^q\right]^{\frac{1}{q}},\qquad(10.2)$$

where $\lambda = 1 - \gamma$ and $\gamma = 0.57721566...$ is the Euler constant. In addition, λ is the largest constant that keeps (10.2) valid and is independent of p and q.

10.3. B. Yang [145], 2000.

Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $(a_n)_{n \in \mathbb{N}_0}$ and $(b_n)_{n \in \mathbb{N}_0}$ be non-negative sequences such that $0 < \sum_{n=0}^{\infty} a_n^p < \infty$ and $0 < \sum_{n=0}^{\infty} b_n^q < \infty$. Then the following two inequalities hold and are equivalent:

$$\sum_{m=0}^{\infty} \sum_{n=0}^{\infty} \frac{a_m b_n}{m+n+1} < \left[\sum_{n=0}^{\infty} \left(\frac{\pi}{\sin \frac{\pi}{p}} - \frac{1}{13(n+1)(2n+1)^{\frac{1}{p}}} \right) a_n^p \right]^{\frac{1}{p}} \\ \times \left[\sum_{n=0}^{\infty} \left(\frac{\pi}{\sin \frac{\pi}{p}} - \frac{1}{13(n+1)(2n+1)^{\frac{1}{q}}} \right) b_n^q \right]^{\frac{1}{q}}$$
(10.3)

and

$$\sum_{m=0}^{\infty} \left(\sum_{n=0}^{\infty} \frac{a_m b_n}{m+n+1} \right)^p < \left(\frac{\pi}{\sin \frac{\pi}{p}} \right)^{p-1} \sum_{n=0}^{\infty} \left[\frac{\pi}{\sin \frac{\pi}{p}} - \frac{1}{13(n+1)(2n+1)^{\frac{1}{p}}} \right] a_n^p.$$
(10.4)

10.4. M. Bencze, C. J. Zhao [6], 2002.

Let $p \ge 1$, $q \ge 1$ be real parameters and let k, r, e be positive integers. If $(a_n)_{n \in \mathbb{N}}$, $(b_n)_{n \in \mathbb{N}}$ are non-negative sequences and $A_m = \sum_{s=1}^m a_s$, $B_n = \sum_{t=1}^n b_t$, then

$$\sum_{m=1}^{k} \sum_{n=1}^{r} \frac{A_{m}^{p} B_{n}^{q}(mn)^{\frac{2}{e}}}{(m \cdot n^{\frac{1}{e}})^{2} + (n \cdot m^{\frac{1}{e}})^{2}} \le \frac{1}{2} pq(kr)^{\frac{e-1}{e}} \left[\sum_{m=1}^{k} (k-m+1)(a_{m}A_{m}^{p-1})^{e} \right]^{\frac{1}{e}} \left[\sum_{n=1}^{r} (r-n+1)(b_{n}B_{n}^{q-1})^{e} \right]^{\frac{1}{e}}.$$
 (10.5)

10.5. B. He, Y. Li [35], 2006.

Let $m \ge 1$, $n \ge 1$, $p_i > 1$, $\frac{1}{p_i} + \frac{1}{q_i} = 1$ for i = 0, 1, 2, 3, 4, $p_0 = p$, $p_3 = k$, $p_4 = r q_0 = q$, $q_3 = l$, $q_4 = \omega$. Further, suppose $f, g : \mathbb{R}_+ \to \mathbb{R}$ are non-negative measurable functions and let $F(s) = \int_0^s f(\sigma) d\sigma$ and $G(t) = \int_0^t g(\tau) d\tau$. If

$$0 < \int_0^\infty s^{p(1-\frac{\lambda}{r})-1} F_f^p(s) ds < \infty, \quad 0 < \int_0^\infty t^{q(1-\frac{\lambda}{\omega})-1} G_g^q(t) dt < \infty,$$

then

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{F^{m}(s)G^{n}(t)}{(ls^{\frac{k}{p_{1}}} + kt^{\frac{1}{p_{2}}})(s^{\lambda} + t^{\lambda})} dsdt$$

$$\leq E_{1}(m, n, k, r, \lambda) \left[\int_{0}^{\infty} s^{p(1-\frac{\lambda}{r})-1} F_{f}^{p}(s) ds \right]^{\frac{1}{p}} \left[\int_{0}^{\infty} t^{q(1-\frac{\lambda}{\omega})-1} G_{g}^{q}(t) dt \right]^{\frac{1}{q}}, (10.6)$$

where $E_1(m, n, k, r, \lambda) = \frac{\pi m n}{\lambda k l \sin(\pi/r)}$,

$$F_f(s) = \left[\int_0^s (F^{m-1}(\sigma)f(\sigma))^{q_1} d\sigma\right]^{\frac{1}{q_1}}, \text{ and } G_g(t) = \left[\int_0^t (G^{m-1}(\tau)g(\tau))^{q_2} d\tau\right]^{\frac{1}{q_2}}.$$

10.6. B. He, Y. Li [34], 2006.

Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $f, g : \mathbb{R}_+ \to \mathbb{R}$ be non-negative measurable functions such that

$$0 < \int_0 (x+1)^{p-1} f^p(x) dx < \infty$$
 and $0 < \int_0 (x+1)^{q-1} g^q(x) dx < \infty$.

Then the inequalities

$$\int_0^\infty \int_0^\infty \frac{f(x)g(y)}{\log(x+1) + \log(y+1) + 1} dx dy$$

$$< \frac{\pi}{\sin\frac{\pi}{p}} \left[\int_0^\infty \omega(x,p) f^p(x) dx \right]^{\frac{1}{p}} \left[\int_0^\infty \omega(x,q) g^q(x) dx \right]^{\frac{1}{q}}$$
(10.7)

and

$$\int_{0}^{\infty} (\omega(y,q))^{1-p} \left[\int_{0}^{\infty} \frac{f(x)}{\log(x+1) + \log(y+1) + 1} dx \right]^{p} dy$$

$$< \left[\frac{\pi}{\sin \frac{\pi}{p}} \right]^{p} \int_{0}^{\infty} \omega(x,p) f^{p}(x) dx$$
(10.8)

hold and are equivalent, where $\omega(x,r) = \left[1 - \frac{1 - \frac{s\sin(\pi/r)}{\pi}}{(\log(x+1)+1)^{1/r}}\right](x+1)^{r-1}$, r = p,q, $s = \frac{r}{r-1}$. In addition, the constant factors included on the right-hand sides of inequalities (10.7) and (10.8) are the best possible.

10.7. J. Weijian, G. Mingzhe [130], 2006.

Let

$$\omega(r,x) = x^{(1+x)(1-r)} \left(\frac{1}{2} - \varphi(x)\right)^{r-1},$$

where r > 1 and $\varphi(x) = \frac{1-x+x\log x}{2(1+x+x\log x)}$, $x \in \mathbb{R}_+$. If $p \ge q > 1$ are conjugate parameters and $f, g: \mathbb{R}_+ \to \mathbb{R}$ are non-negative measurable functions such that

$$0 < \int_0^\infty \omega(p,x) f^p(x) dx < +\infty, \quad 0 < \int_0^\infty \omega(q,x) g^q(x) dx < +\infty,$$

then

$$\int_0^\infty \int_0^\infty \frac{f(x)g(y)}{ax^{1+x} + by^{1+y}} dx dy$$

$$\leq \frac{\mu\pi}{\sin\frac{\pi}{p}} \left[\int_0^\infty \omega(p,x) f^p(x) dx \right]^{\frac{1}{p}} \left[\int_0^\infty \omega(q,x) g^q(x) dx \right]^{\frac{1}{q}}, \tag{10.9}$$

where $\mu = (\frac{1}{a})^{\frac{1}{q}}(\frac{1}{b})^{\frac{1}{p}}$. Moreover, the constant factor $\frac{\mu\pi}{\sin\frac{\pi}{p}}$ is the best possible.

10.8. W. T. Sulaiman [122], 2006.

Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $\lambda > \max\{p,q\}$. Further, suppose $f, g : \mathbb{R}_+ \to \mathbb{R}$ are non-negative functions such that f(0) = g(0) = 0, $f(\infty) = g(\infty) = \infty$, $f'(s) \ge 0$, $g'(s) \ge 0$, $s \in \{x^p, y^q\}$, and let $\log f$, $\log g$ be convex functions. If

$$0 < \int_0^\infty \frac{t^{-\frac{p^2}{q^2}} [f(t^p)]^{2-\lambda+\frac{p}{q}}}{[f'(t)]^{\frac{p}{q}}} dt < \infty, \quad 0 < \int_0^\infty \frac{t^{-\frac{q^2}{p^2}} [g(t^q)]^{2-\lambda+\frac{q}{p}}}{[g'(t)]^{\frac{q}{p}}} dt < \infty,$$

then

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{f(xy)g(xy)}{(f(x^{p}) + g(y^{q}))^{\lambda}} dx dy \\
\leq \frac{1}{\sqrt[q]{p}\sqrt[q]{q}} B^{\frac{1}{p}}(p,\lambda-p) B^{\frac{1}{q}}(q,\lambda-q) \\
\times \left[\int_{0}^{\infty} \frac{t^{-\frac{p^{2}}{q^{2}}} [f(t^{p})]^{2-\lambda+\frac{p}{q}}}{[f'(t)]^{\frac{p}{q}}} dt \right]^{\frac{1}{p}} \left[\int_{0}^{\infty} \frac{t^{-\frac{q^{2}}{p^{2}}} [g(t^{q})]^{2-\lambda+\frac{q}{p}}}{[g'(t)]^{\frac{q}{p}}} dt \right]^{\frac{1}{q}}, \quad (10.10)$$

where B is the usual Beta function.

10.9. W. Wang, D. Xin [129], 2006.

Let p > 1, $\frac{1}{p} + \frac{1}{q} = 1$, $\alpha \ge e^{\frac{7}{6}}$, and $s, t \in \mathbb{R}$. If $(a_n)_{n \in \mathbb{N}}$ and $(b_n)_{n \in \mathbb{N}}$ are non-negative real sequences such that $0 < \sum_{n=1}^{\infty} (n^{\frac{1}{q}-s}a_n)^p < \infty$ and $0 < \sum_{n=1}^{\infty} (n^{\frac{1}{p}-t}b_n)^q < \infty$, then

$$\sum_{n=1}^{\infty} \sum_{m=1}^{\infty} \frac{a_m b_n}{m^s n^t \log \alpha m n} < \left[\sum_{n=1}^{\infty} \left(\frac{\pi}{\sin \frac{\pi}{p}} - \frac{3(p-1)}{8(2\log n+1)^{\frac{1}{p}}} \right) (n^{\frac{1}{q}-s} a_n)^p \right]^{\frac{1}{p}} \\ \times \left[\sum_{n=1}^{\infty} \left(\frac{\pi}{\sin \frac{\pi}{p}} - \frac{3(q-1)}{8(2\log n+1)^{\frac{1}{q}}} \right) (n^{\frac{1}{p}-s} b_n)^q \right]^{\frac{1}{q}}.$$
 (10.11)

10.10. Z. Lü, H. Xie [88], 2007.

Let $-c \le a < b < \infty$ and let $f, g : [a, b] \to \mathbb{R}$ be non-negative measurable functions such that $\int_a^b (x+c)f^2(x)dx < \infty$ and $\int_a^b (y+c)g^2(y)dy < \infty$. Then the following inequalities hold and are equivalent:

$$\int_{a}^{b} \int_{a}^{b} \frac{f(x)g(y)}{\log(x+c)(y+c)} dx dy$$

$$\leq \left(\pi - 4\arctan\sqrt[4]{\frac{\log(a+c)}{\log(b+c)}}\right) \left[\int_{a}^{b} (x+c)f^{2}(x) dx \int_{a}^{b} (y+c)g^{2}(y) dy\right]^{\frac{1}{2}} (10.12)$$

and

$$\int_{a}^{b} \left[\int_{a}^{b} \frac{f(x)}{\log(x+c)(y+c)} dx \right]^{2} dy$$

$$\leq \left(\pi - 4 \arctan \sqrt[4]{\frac{\log(a+c)}{\log(b+c)}} \right)^{2} \int_{a}^{b} (x+c) f^{2}(x) dx. \tag{10.13}$$

10.11. Z. Lü, G. Mingzhe, L. Debnath [87], 2007.

Suppose F is defined as

$$F(s,t) = \|\alpha\|^2 s^2 - 2\langle \alpha, \beta \rangle st + \|\beta\|^2 t^2,$$
(10.14)

where α , β , and γ belong to inner product space E, $\langle \alpha, \beta \rangle$ indicates the inner product of vectors α and β , γ is the unit-vector, α and β are not simultaneously orthogonal to γ , and $s = \langle \beta, \gamma \rangle$ and $t = \langle \alpha, \gamma \rangle$.

Further, suppose $f, g \in L^2(\mathbb{R}_+), \phi(x), \psi(x)$ are differentiable functions in \mathbb{R}_+ , and $\phi(0) = \psi(0) \ge 0, \phi(\infty) = \psi(\infty) = \infty, \phi'(x), \psi'(x) > 0$. If the functions $\phi'(x)$ and $\psi'(x)$

have positive infimums, then

$$\left[\int_{0}^{\infty} \int_{0}^{\infty} \frac{f(x)g(y)}{\phi(x) + \psi(y)} dx dy\right]^{2}$$

$$\leq \frac{1}{\inf\{\phi'(x)\}\inf\{\psi'(y)\}} \int_{0}^{\infty} \left[\pi - \arctan\sqrt{\frac{\psi(0)}{\phi(x)}}\right] f^{2}(x) dx$$

$$\times \int_{0}^{\infty} \left[\pi - \arctan\sqrt{\frac{\phi(0)}{\psi(y)}}\right] g^{2}(y) dy - F(s,t), \qquad (10.15)$$

where F(s,t) is defined by (10.14) and F(s,t) > 0.

10.12. H. Leping, G. Xuemei, G. Mingzhe [78], 2008.

If $(a_n)_{n\in\mathbb{N}}$ and $(b_n)_{n\in\mathbb{N}}$ are non-negative sequences such that $\sum_{n=1}^{\infty} a_n^2 < +\infty$ and $\sum_{n=1}^{\infty} b_n^2 < +\infty$, then

$$\left[\sum_{m=1}^{\infty}\sum_{n=1}^{\infty}\frac{a_{m}b_{n}}{m+n}\right]^{4} \leq \pi^{4}\left[\left(\sum_{n=1}^{\infty}a_{n}^{2}\right)^{2} - \left(\sum_{n=1}^{\infty}\omega(n)a_{n}^{2}\right)^{2}\right] \times \left[\left(\sum_{n=1}^{\infty}b_{n}^{2}\right)^{2} - \left(\sum_{n=1}^{\infty}\omega(n)b_{n}^{2}\right)^{2}\right],$$
(10.16)

where the weight function $\omega(n)$ is defined by

$$\omega(n) = \frac{\sqrt{n}}{n+1} \left(\frac{\sqrt{n}-1}{\sqrt{n}+1} - \frac{\log n}{\pi} \right).$$

10.13. H. Leping, G. Mingzhe, Z. Yu [77], 2008.

Suppose *f* and *g* are non-negative real functions such that $0 < \int_{\alpha}^{\infty} (x - \alpha)^{1-\lambda} f^2(x) dx < \infty$ and $0 < \int_{\alpha}^{\infty} (x - \alpha)^{1-\lambda} g^2(x) dx < \infty$ for $\lambda > \frac{1}{2}$. Then,

$$\begin{bmatrix} \int_{\alpha}^{\infty} \int_{\alpha}^{\infty} \frac{f(x)g(y)}{(x-\alpha)^{\lambda} + (y-\alpha)^{\lambda}} dx dy \end{bmatrix}^{4} \\ \leq \left(\frac{\pi}{\lambda \sin \frac{\pi}{2\lambda}}\right)^{4} \left[\left(\int_{\alpha}^{\infty} (x-\alpha)^{1-\lambda} f^{2}(x) dx \right)^{2} - \left(\int_{\alpha}^{\infty} \omega_{\lambda}(x) f^{2}(x) dx \right)^{2} \right] \\ \times \left[\left(\int_{\alpha}^{\infty} (x-\alpha)^{1-\lambda} g^{2}(x) dx \right)^{2} - \left(\int_{\alpha}^{\infty} \omega_{\lambda}(x) g^{2}(x) dx \right)^{2} \right], \quad (10.17)$$

where the weight function $\omega_{\lambda}(x)$ is defined by

$$\omega_{\lambda}(x) = \begin{cases} (x-\alpha)^{1-\lambda} \left[\frac{(x-\alpha)^{\lambda-1/2}-1}{(x-\alpha)^{\lambda}-1} - \frac{1}{1+(x-\alpha)^{\lambda}} \right], \ x-\alpha \neq 1, \\ \frac{1}{2} - \frac{1}{2\lambda}, \qquad \qquad x-\alpha = 1. \end{cases}$$

10.14. W. T. Sulaiman [124], 2008.

Suppose p, q > 1 are conjugate parameters. A non-negative mapping $f : \mathbb{R}_+ \to \mathbb{R}$ is called (p,q)-Hölder type on \mathbb{R}_+ if the inequality $f(xy) \le f^{\frac{1}{p}}(x^p)f^{\frac{1}{q}}(y^q)$ holds for all $x, y \in \mathbb{R}_+$.

Let *f* and *F* be non-negative functions of (p,q)-Hölder type on \mathbb{R}_+ such that f(0) = 0and $f(\infty) = \infty$. Further, suppose that *f'* exist and is strictly positive on \mathbb{R}_+ . If $\lambda > 0$, $\mu > 1$, $\max\{\frac{p-1}{q}, \frac{q-1}{p}\} < \frac{\mu-1}{\lambda} < \min\{\frac{p}{q}, \frac{q}{p}\}$, and

$$K = B^{\frac{1}{p}} (\lambda p - \mu q + q, \lambda - \lambda p + \mu q - q) B^{\frac{1}{q}} (\lambda q - \mu p + p, \lambda - \lambda q + \mu p - p),$$

where B is the Beta function, then

$$\int_{0}^{\infty} \int_{0}^{\infty} F(xy) \left(\frac{f(xy)}{1+f(xy)}\right)^{\lambda} dxdy$$

$$\leq K \left[\int_{0}^{\infty} \frac{(f(x^{p}))^{(pq-2)\mu-\lambda-\frac{p}{q}}F(x^{p})}{x^{\frac{p(p-1)}{q}}(f'(x^{p}))^{\frac{p}{q}}} dx \right]^{\frac{1}{p}} \left[\int_{0}^{\infty} \frac{(f(y^{q}))^{(pq-2)\mu-\lambda-\frac{q}{p}}F(y^{q})}{y^{\frac{q(q-1)}{p}}(f'(y^{q}))^{\frac{q}{p}}} dy \right]^{\frac{1}{q}},$$
(10.18)

provided that integrals on the right-hand side of the inequality converge.

10.15. Z. Yu, G. Xuemei, G. Mingzhe [175], 2009.

Let $\lambda > 0$ and let f, g be non-negative functions such that $0 < \int_0^\infty x^{1-\lambda} f^2(x) dx < \infty$ and $0 < \int_0^\infty x^{1-\lambda} g^2(x) dx < \infty$. If m is a positive integer, then

$$\int_{0}^{\infty} \int_{0}^{\infty} \frac{(\log x - \log y)^{2m-1} f(x)g(y)}{x^{\lambda} - y^{\lambda}} dx dy < C_{B} \left[\int_{0}^{\infty} x^{1-\lambda} f^{2}(x) dx \right]^{\frac{1}{2}} \left[\int_{0}^{\infty} x^{1-\lambda} g^{2}(x) dx \right]^{\frac{1}{2}},$$
(10.19)

where the constant factor C_B is defined by

$$C_B = \frac{2^{2m-1}(2^{2m}-1)}{m} \left(\frac{\pi}{\lambda}\right)^{2m} B_m.$$

Here, B_m are the Bernoulli numbers, namely $B_1 = \frac{1}{6}$, $B_2 = \frac{1}{30}$, $B_3 = \frac{1}{42}$, $B_4 = \frac{1}{30}$, $B_5 = \frac{5}{66}$, and so forth. Moreover, the constant factor C_B on the right-hand side of (10.19) is the best possible.

10.16. W. Yang [169], 2009.

Let $p \ge 1$, $q \ge 1$, $\alpha > 1$, $\gamma > 1$ be real parameters and let k, r be positive integers. If $(a_n)_{n \in \mathbb{N}}$, $(b_n)_{n \in \mathbb{N}}$ are non-negative sequences and $A_m = \sum_{s=1}^m a_s$, $B_n = \sum_{t=1}^n b_t$, then

$$\sum_{m=1}^{k} \sum_{n=1}^{r} \frac{A_{m}^{p} B_{n}^{q}}{\gamma m^{\frac{(\alpha-1)(\alpha+\gamma)}{\alpha\gamma}} + \alpha n^{\frac{(\gamma-1)(\alpha+\gamma)}{\alpha\gamma}}} \leq C(p,q,k,r;\alpha,\gamma) \\ \times \left[\sum_{m=1}^{k} (k-m+1)(A_{m}^{p-1}a_{m})^{\alpha}\right]^{\frac{1}{\alpha}} \left[\sum_{n=1}^{r} (r-n+1)(B_{n}^{q-1}b_{n})^{\gamma}\right]^{\frac{1}{\gamma}}, \quad (10.20)$$

where $C(p,q,k,r;\alpha,\gamma) = \frac{pq}{\alpha+\gamma}k^{\frac{\alpha-1}{\alpha}}r^{\frac{\gamma-1}{\gamma}}.$

10.17. W. Yang [170], 2009.

Let $q_i \ge 1$, $p_i > 1$, $p = \sum_{i=1}^n \frac{1}{p_i}$, and $\alpha_i = \prod_{j=1, j \neq i}^n p_j$, i = 1, 2, ..., n. Further, suppose a_{i,m_i} , i = 1, 2, ..., n, are non-negative sequences defined for $m_i = 1, 2, ..., k_i$, where k_i are positive integers and let $A_{i,m_i} = \sum_{s_i=1}^{m_i} a_{i,s_i}$, i = 1, 2, ..., n. Then,

$$\sum_{m_{1}=1}^{k_{1}} \cdots \sum_{m_{n}=1}^{k_{n}} \frac{\prod_{i=1}^{n} A_{i,m_{i}}^{q_{i}}}{\sum_{i=1}^{n} \alpha_{i} m_{i}^{(p_{i}-1)p}} \leq C(k_{1},\ldots,k_{n}) \prod_{i=1}^{n} \left[\sum_{m_{i}=1}^{k_{i}} (k_{i}-m_{i}+1) (A_{i,m_{i}}^{q_{i}-1}a_{i,m_{i}})^{p_{i}} \right]^{\frac{1}{p_{i}}}, \quad (10.21)$$

$$k_{r}) = \frac{\prod_{i=1}^{n} q_{i}}{\prod_{r=1}^{n} \prod_{r=1}^{n} k_{r}^{\frac{p_{i}-1}{p_{i}}}}$$

where $C(k_1,\ldots,k_n) = \frac{\prod_{i=1}^n q_i}{\sum_{i=1}^n \alpha_i} \prod_{i=1}^n k_i^{\frac{p_i-1}{p_i}}$.

10.18. W. Yang [171], 2009.

Let us define the operator ∇ by $\nabla u(t) = u(t) - u(t-1)$ for a function $u : \mathbb{N}_0 \to \mathbb{R}$. Further, define the operators $\nabla_1 v(s,t) = v(s,t) - v(s-1,t)$, $\nabla_2 v(s,t) = v(s,t) - v(s,t-1)$, and $\nabla_2 \nabla_1 v(s,t) = \nabla_2 (\nabla_1 v(s,t)) = \nabla_1 (\nabla_2 v(s,t))$ for a function $v : \mathbb{N}_0 \times \mathbb{N}_0 \to \mathbb{R}$.

 $\begin{aligned} &\nabla_2 \nabla_1 v(s,t) = \nabla_2 (\nabla_1 v(s,t)) = \nabla_1 (\nabla_2 v(s,t)) \quad \forall (s-1,t), \quad \forall_2 v(s,t) = v(s,t) \quad \forall (s,t-1), \text{ and } \\ &\nabla_2 \nabla_1 v(s,t) = \nabla_2 (\nabla_1 v(s,t)) = \nabla_1 (\nabla_2 v(s,t)) \text{ for a function } v : \mathbb{N}_0 \times \mathbb{N}_0 \to \mathbb{R}. \\ &\text{Suppose } p > 1, \frac{1}{p} + \frac{1}{q} = 1, \text{ and } a, b : \mathbb{N}_0 \times \mathbb{N}_0 \to \mathbb{R} \text{ are such that } a(0,t) = b(0,t) = \\ &a(s,0) = b(s,0) = 0. \quad \text{If } 0 < \sum_{m=1}^{\infty} \sum_{r=1}^{m} \sum_{k=1}^{m} \sum_{r=1}^{n} |\nabla_2 \nabla_1 a(\tau,\delta)|^p < \infty \text{ and } \\ &0 < \sum_{s=1}^{\infty} \sum_{r=1}^{s} \sum_{k=1}^{s} \sum_{r=1}^{t} |\nabla_2 \nabla_1 b(k,r)|^q < \infty, \text{ then} \end{aligned}$

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \sum_{s=1}^{\infty} \sum_{t=1}^{\infty} \frac{|a(m,n)| |b(s,t)|}{(q(mn)^{p-1} + p(st)^{q-1})(m+s)(n+t)} < \frac{\pi^2}{pq \sin^2 \frac{\pi}{p}} \left[\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \sum_{\tau=1}^{m} \sum_{\delta=1}^{n} |\nabla_2 \nabla_1 a(\tau,\delta)|^p \right]^{\frac{1}{p}} \times \left[\sum_{s=1}^{\infty} \sum_{t=1}^{\infty} \sum_{k=1}^{\infty} \sum_{r=1}^{s} \frac{t}{p} |\nabla_2 \nabla_1 b(k,r)|^q \right]^{\frac{1}{q}}.$$
(10.22)

10.19. C. J. Zhao, W. S. Cheung [179], 2009.

Let $p \ge 1$, $q \ge 1$, t > 0, and $\frac{1}{\alpha} + \frac{1}{\beta} = 1$, $\alpha > 1$. Further, suppose a_{m_1,\dots,m_n} and b_{n_1,\dots,n_n} are non-negative sequences defined for $m_i = 1, 2, \dots, k_i$, and $n_i = 1, 2, \dots, r_i$, where k_i and r_i , $i = 1, \dots, n$, are positive integers. If $A_{m_1,\dots,m_n} = \sum_{s_1=1}^{m_1} \cdots \sum_{s_n=1}^{m_n} a_{s_1,\dots,s_n}$ and $B_{n_1,\dots,n_n} = \sum_{t_1=1}^{n_1} \cdots \sum_{t_n=1}^{n_n} b_{t_1,\dots,t_n}$, then

$$\sum_{m_{1}=1}^{k_{1}} \cdots \sum_{m_{n}=1}^{k_{n}} \sum_{n_{1}=1}^{r_{1}} \cdots \sum_{n_{n}=1}^{r_{n}} \frac{\alpha \beta t^{\frac{1}{\beta}} A_{m_{1},\dots,m_{n}}^{p} B_{n_{1},\dots,n_{n}}^{q}}{m_{1} \cdots m_{n} \beta + n_{1} \cdots n_{n} \alpha t}$$

$$\leq L(k_{1},\dots,k_{n},r_{1},\dots,r_{n},p,q,\alpha,\beta)$$

$$\times \left[\sum_{m_{1}=1}^{k_{1}} \cdots \sum_{m_{n}=1}^{k_{n}} \prod_{j=1}^{n} (k_{j}-m_{j}+1) \left(a_{m_{1},\dots,m_{n}} A_{m_{1},\dots,m_{n}}^{p-1}\right)^{\beta}\right]^{\frac{1}{\beta}}$$

$$\times \left[\sum_{n_{1}=1}^{r_{1}} \cdots \sum_{n_{n}=1}^{r_{n}} \prod_{j=1}^{n} (r_{j}-n_{j}+1) \left(b_{n_{1},\dots,n_{n}} B_{n_{1},\dots,n_{n}}^{q-1}\right)^{\alpha}\right]^{\frac{1}{\alpha}}, \quad (10.23)$$

where

$$L(k_1,\ldots,k_n,r_1,\ldots,r_n,p,q,\alpha,\beta)=pq(k_1\cdots k_n)^{\frac{1}{\alpha}}(r_1\cdots r_n)^{\frac{1}{\beta}}.$$

10.20. Q. Huang [38], 2010.

Suppose $p_i, r_i > 1$, i = 1, ..., n, are real parameters such that $\sum_{i=1}^{n} \frac{1}{p_i} = \sum_{i=1}^{n} \frac{1}{r_i} = 1$ and $\frac{1}{q_n} = 1 - \frac{1}{p_n}$. Further, let α, β , and λ be real parameters such that $\lambda > 0$, $0 < \alpha < 2$, $\beta \ge -\frac{1}{2}$, and $\lambda \alpha \max\{\frac{1}{2-\alpha}, 1\} \le \min_{1 \le i \le n}\{r_i\}$. Then the following two equivalent inequalities

$$\sum_{m_n=1}^{\infty} \cdots \sum_{m_1=1}^{\infty} \frac{1}{[\sum_{i=1}^{n} (m_i + \beta)^{\alpha}]^{\lambda}} \prod_{i=1}^{n} a_{m_i}^{(i)}$$

$$< \frac{\alpha^{1-n}}{\Gamma(\lambda)} \prod_{i=1}^{n} \Gamma\left(\frac{\lambda}{r_i}\right) \left[\sum_{m_i=1}^{\infty} (m_i + \beta)^{p_i(1-\frac{\lambda\alpha}{r_i})-1} \left(a_{m_i}^{(i)}\right)^{p_i}\right]^{\frac{1}{p_i}}, \quad (10.24)$$

$$\left[\sum_{m_n=1}^{\infty} (m_n + \beta)^{\frac{\lambda\alpha q_n}{r_n}-1} \left(\sum_{m_{n-1}=1}^{\infty} \cdots \sum_{m_1=1}^{\infty} \frac{\prod_{i=1}^{n-1} a_{m_i}^{(i)}}{[\sum_{i=1}^{n} (m_i + \beta)^{\alpha}]^{\lambda}}\right)^{q_n}\right]^{\frac{1}{q_n}}$$

$$< \frac{\Gamma\left(\frac{\lambda}{r_n}\right)}{\alpha^{n-1}\Gamma(\lambda)} \prod_{i=1}^{n-1} \Gamma\left(\frac{\lambda}{r_i}\right) \left[\sum_{m_i=1}^{\infty} (m_i + \beta)^{p_i(1-\frac{\lambda\alpha}{r_i})-1} \left(a_{m_i}^{(i)}\right)^{p_i}\right]^{\frac{1}{p_i}} \quad (10.25)$$

hold for all non-negative sequences $(a_{m_i}^{(i)})_{m_i \in \mathbb{N}}$, provided that

$$0 < \sum_{m_i=1}^{\infty} (m_i + \beta)^{p_i(1 - \frac{\lambda \alpha}{r_i}) - 1} \left(a_{m_i}^{(i)} \right)^{p_i} < \infty \quad i = 1, \dots, n.$$

Here, Γ is the Gamma function.

10.21. N. Das, S. Sahoo [20], 2010.

Suppose $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let λ , r, s be real parameters such that $0 < \lambda \le 4$, 0 < r, $s < \min\{2,\lambda\}$, and $r + s = \lambda$. If $A_n = \sum_{k=1}^n a_k$ and $B_n = \sum_{k=1}^n b_k$, where $(a_n)_{n \in \mathbb{N}}$, $(b_n)_{n \in \mathbb{N}}$ are non-negative sequences fulfilling $0 < \sum_{n=1}^{\infty} a_n^p < \infty$ and $0 < \sum_{n=1}^{\infty} b_n^q < \infty$, then

$$\sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \frac{m^{r-\frac{1}{q}-1} n^{s-\frac{1}{p}-1}}{(m+n)^{\lambda}} A_m B_n < pqB(r,s) \left[\sum_{n=1}^{\infty} a_n^p\right]^{\frac{1}{p}} \left[\sum_{n=1}^{\infty} b_n^q\right]^{\frac{1}{q}}, \quad (10.26)$$

$$\sum_{n=1}^{\infty} \left(\sum_{m=1}^{\infty} \frac{m^{r-\frac{1}{q}-1} n^{s-\frac{1}{p}}}{(m+n)^{\lambda}} A_m \right)^p < (qB(r,s))^p \sum_{n=1}^{\infty} a_n^p,$$
(10.27)

where *B* is the Beta function. In addition, the constant factors pqB(r,s) and $(qB(r,s))^p$ are the best possible in the above equivalent inequalities.

10.22. X. Liu, B. Yang [85], 2010.

Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let λ , λ_1 , λ_2 be real parameters such that $\lambda_1 + \lambda_2 = \lambda < 2$. Further, suppose $k_{\lambda} : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ is a non-negative homogeneous function of degree $-\lambda$ so that $0 < k(\lambda_1) = \int_0^\infty k(u, 1)u^{\lambda_1 - 1}du < \infty$ holds for all $\lambda_1 \in (\lambda - 1, 1)$. If $f, g : \mathbb{R}_+ \to \mathbb{R}$ are non-negative functions and $\tilde{\varphi}(x) = x^{p(2-\lambda-\lambda_1)-1}$, $\psi(y) = y^{q(1-\lambda_2)-1}$,

$$\widetilde{F}_{\lambda}(x) = \int_{x}^{\infty} \frac{1}{t^{\lambda}} f(t) dt, \quad \widetilde{G}_{\lambda}(y) = \int_{y}^{\infty} \frac{1}{t^{\lambda}} g(t) dt,$$

so that $0 < \|f\|_{L^p_{\widetilde{\varphi}}} = [\int_0^{\infty} \widetilde{\varphi}(x) |f(x)|^p dx]^{\frac{1}{p}} < \infty$ and $0 < \|\widetilde{G}_{\lambda}\|_{L^q_{\Psi}} < \infty$, then the inequalities

$$\int_{0}^{\infty} \int_{0}^{\infty} k_{\lambda}(x, y) \widetilde{F}_{\lambda}(x) \widetilde{G}_{\lambda}(y) dx dy < \frac{k(\lambda_{1})}{1 - \lambda_{1}} \|f\|_{L^{p}_{\widetilde{\varphi}}} \|\widetilde{G}_{\lambda}\|_{L^{q}_{\Psi}}$$
(10.28)

and

$$\left\{\int_0^\infty \psi^{1-p}(y) \left[\int_0^\infty k_\lambda(x,y)\widetilde{F}_\lambda(x)dx\right]^p dy\right\}^{\frac{1}{p}} < \frac{k(\lambda_1)}{1-\lambda_1} \|f\|_{L^p_{\widetilde{\varphi}}}$$
(10.29)

hold and are equivalent.

10.23. S. K. Sunanda, C. Nahak, S. Nanda [126], 2010.

Let $(p_k)_{k \in \mathbb{N}}$ and $(q_k)_{k \in \mathbb{N}}$ be non-negative bounded sequences such that $\frac{1}{p_k} + \frac{1}{q_k} = 1$, where $p_k > 1$ for all $k \in \mathbb{N}$. Then the inequality

$$\sum_{n=2}^{\infty} \sum_{m=2}^{\infty} \frac{a_m b_n}{\log(mn)} < \alpha \sup_{k \ge 1} \left[\frac{\pi}{\sin \frac{\pi}{p_k}} \left(\sum_{n=2}^{\infty} n^{p_k - 1} a_n^{p_k} \right)^{\frac{1}{p_k}} \left(\sum_{n=2}^{\infty} n^{q_k - 1} b_n^{q_k} \right)^{\frac{1}{q_k}} \right]$$
(10.30)

holds for all non-negative sequences $(a_n)_{n\in\mathbb{N}}, (b_n)_{n\in\mathbb{N}}$, such that $0 < \sum_{n=2}^{\infty} n^{p_k-1} a_n^{p_k} < \infty$, $0 < \sum_{n=2}^{\infty} n^{q_k-1} b_n^{q_k} < \infty$. Here,

$$\alpha = \sup_{k \ge 1} \frac{1}{p_k} + \sup_{k \ge 1} \frac{1}{q_k}$$

10.24. Z. Zeng, Z. Xie [176], 2010.

Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let

$$k = \frac{4\pi \sin \frac{r(\beta - \alpha)}{2} \cos \frac{r(\pi - \alpha - \beta)}{2}}{r \cos \frac{r\pi}{2}},$$

where -1 < r < 0 and $0 < \alpha < \beta < \pi$. Then the inequalities

$$\int_{-\infty}^{\infty} \int_{-\infty}^{\infty} f(x)g(y) \left| \log \frac{x^2 + 2xy\cos\alpha + y^2}{x^2 + 2xy\cos\beta + y^2} \right| dxdy$$

$$< k \left[\int_{-\infty}^{\infty} |x|^{p(1+r)-1} f^p(x) dx \right]^{\frac{1}{p}} \left[\int_{-\infty}^{\infty} |x|^{q(1-r)-1} g^q(x) dx \right]^{\frac{1}{q}}$$
(10.31)

and

$$\int_{-\infty}^{\infty} |y|^{pr-1} \left[\int_{-\infty}^{\infty} f(x) \left| \log \frac{x^2 + 2xy \cos \alpha + y^2}{x^2 + 2xy \cos \beta + y^2} \right| dx \right]^p dy$$

$$< k^p \int_{-\infty}^{\infty} |x|^{p(1+r)-1} f^p(x) dx$$
(10.32)

hold for all non-negative measurable functions $f,g: \mathbb{R} \to \mathbb{R}$, provided that $0 < \int_{-\infty}^{\infty} |x|^{p(1+r)-1} f^p(x) dx < \infty$ and $0 < \int_{-\infty}^{\infty} |x|^{q(1-r)-1} g^q(x) dx < \infty$. Moreover, the above inequalities are equivalent and include the best possible constant factors on their right-hand sides.

10.25. J. Jin, L. Debnath [45], 2010.

For p > 0, $n_0 \in \mathbb{Z}$, $w(n) \ge 0$, $n \ge n_0$, $n \in \mathbb{Z}$, we define the set of sequences l_{w,n_0}^p by

$$l_{w,n_0}^p = \left\{ (a_n)_{n \ge n_0}; \|a\|_{p,w} = \left(\sum_{n=n_0}^{\infty} w(n) |a_n|^p\right)^{\frac{1}{p}} < \infty \right\}.$$

For conjugate parameters r, s > 1 denote by H(r, s) the set of all non-negative functions $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ fulfilling the following conditions:

(i) $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ is continuous and decreasing in each variable,

(ii) K is homogeneous of degree -1,

(iii) there exist a constant M > 0 such that $\lim_{t\to 0^+} K(1,t) = M$. For sufficiently small $\varepsilon \ge 0$ the integral $K_l(\varepsilon) = \int_0^\infty K(1,t)t^{-\frac{1+\varepsilon}{l}} dt$, l = r, s, exists, where $K_l(0) = C_r$ is a positive constant and $K_l(\varepsilon) = C_r + o(1)$ as $\varepsilon \to 0^+$.

Further, let $F_{n_0}(r)$, r > 1, $n_0 \in \mathbb{Z}$, denote the set of all real-valued functions $\phi(x)$ such that:

(i) $\phi(x)$ is continuously differentiable and strictly increasing on $(n_0 - 1, \infty)$,

(ii) $\phi((n_0-1)+) = 0, \ \phi(\infty) = \infty, \ \phi'(x)[\phi(x)]^{-\frac{1}{p}}$ is decreasing on (n_0-1,∞) . Let $p > 1, \ \frac{1}{p} + \frac{1}{q} = 1, \ r > 1, \ \frac{1}{r} + \frac{1}{s} = 1, \ m_0, n_0 \in \mathbb{Z}, \ K \in H(r,s), \ \phi \in F_{m_0}(r), \ \psi \in F_{m_0}(s)$. Setting $w_1(m) = [\phi(m)]^{\frac{p}{r}-1}[\phi'(m)]^{1-p}, \ w_2(n) = [\psi(n)]^{\frac{q}{s}-1}[\psi'(n)]^{1-q}, \ \widetilde{w}_1(n) = \psi'(n)[\psi(n)]^{\frac{p}{r}-1}, \ \widetilde{w}_2(m) = \phi'(m)[\phi(m)]^{\frac{q}{s}-1}, \ we define the sequence operator T as follows: for <math>a \in l^{p}_{w_1,m_0}$,

$$(Ta)(n) = \sum_{m=m_0}^{\infty} K(\phi(m), \psi(n))a_m, \ n \ge n_0, n \in \mathbb{Z}$$

or, for $b \in l^q_{w_2,n_0}$,

$$(Tb)(m) = \sum_{n=n_0}^{\infty} K(\phi(m), \psi(n))b_n, \ m \ge m_0, m \in \mathbb{Z}.$$

Then we have

$$||Ta||_{p,\widetilde{w}_1} \le C_r ||a||_{p,w_1}$$
 and $||Tb||_{q,\widetilde{w}_2} \le C_r ||b||_{q,w_2}$. (10.33)

Moreover,

$$||T||_{p} = ||T||_{q} = C_{r} = \int_{0}^{\infty} K(1,t)t^{-\frac{1}{r}}dt, \qquad (10.34)$$

where

$$||T||_{p} = \sup_{a \in l_{w_{1},m_{0}}^{p}} \frac{||Ta||_{p,\widetilde{w}_{1}}}{||a||_{p,w_{1}}} \text{ and } ||T||_{q} = \sup_{b \in l_{w_{2},m_{0}}^{q}} \frac{||Tb||_{q,\widetilde{w}_{2}}}{||b||_{q,w_{2}}}.$$

10.26. L. E. Azar [5], 2011.

Let $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let a, b, c, s, A_1, A_2 be real parameters such that a, c > 0, $b^2 < ac$, $s > 0, A_1 \in (\frac{1-2s}{q}, \frac{1}{q}), A_2 \in (\frac{1-2s}{p}, \frac{1}{p})$, and let

$$L_{1} = a^{\frac{1-pA_{2}}{2}-s}c^{\frac{pA_{2}-1}{2}}B(1-pA_{2},2s+pA_{2}-1)F(\frac{1-pA_{2}}{2},s-\frac{1-pA_{2}}{2},s+\frac{1}{2};1-\frac{b^{2}}{ac}),$$

$$L_{2} = a^{\frac{qA_{1}-1}{2}}c^{\frac{1-qA_{1}}{2}-s}B(1-qA_{1},2s+qA_{1}-1)F(\frac{1-qA_{1}}{2},s-\frac{1-qA_{1}}{2},s+\frac{1}{2};1-\frac{b^{2}}{ac}),$$

where B and F respectively denote the usual Beta and the hypergeometric function (see relation (2.53), Chapter 2).

Further, suppose $u, v: (a, b) \to \mathbb{R}$, $-\infty \le a < b \le \infty$, are non-negative differentiable strictly increasing functions fulfilling conditions $\lim_{t\to a^+} u(t) = \lim_{t\to a^+} v(t) = 0$ and $\lim_{t\to b^-} u(t) = \lim_{t\to b^-} v(t) = \infty$. Then the inequalities

$$\int_{a}^{b} \int_{a}^{b} \frac{f(x)g(y)}{(au^{2}(x) + 2bu(x)v(y) + cv^{2}(y))^{s}} dxdy$$

$$< L_{1}^{\frac{1}{p}} L_{2}^{\frac{1}{q}} \left[\int_{a}^{b} \frac{u(x)^{1-2s+p(A_{1}-A_{2})}}{u'(x)^{p-1}} f^{p}(x)dx \right]^{\frac{1}{p}} \left[\int_{a}^{b} \frac{v(y)^{1-2s+q(A_{2}-A_{1})}}{v'(y)^{q-1}} g^{q}(y)dy \right]^{\frac{1}{q}} (10.35)$$

$$\int_{a}^{b} v(y)^{(2s-1)(p-1)+p(A_{1}-A_{2})} v'(y) \left[\int_{a}^{b} \frac{f(x)}{(au^{2}(x)+2bu(x)v(y)+cv^{2}(y))^{s}} dx \right]^{p} dy$$

$$< L_{1}L_{2}^{\frac{p}{q}} \int_{a}^{b} \frac{u(x)^{1-2s+p(A_{1}-A_{2})}}{u'(x)^{p-1}} f^{p}(x) dx$$
(10.36)

hold for all non-negative measurable functions $f,g:(a,b) \to \mathbb{R}$, provided that $0 < \int_a^b \frac{u(x)^{1-2s+p(A_1-A_2)}}{u'(x)^{p-1}} f^p(x) dx < \infty$ and $0 < \int_a^b \frac{v(y)^{1-2s+q(A_2-A_1)}}{v'(y)^{q-1}} g^q(y) dy < \infty$. Moreover, inequalities (10.35) and (10.36) are equivalent.

10.27. C. T. Chang, J. W. Lan, K. Z. Wang [13], 2011.

For p > 1, denote by \mathscr{H}_p the set of all non-negative functions $K : \mathbb{R}_+ \times \mathbb{R}_+ \to \mathbb{R}$ having the following properties:

(i) K is homogeneous of degree -1,

(ii) $K(x,1)x^{-\frac{1}{p}}$ is strictly decreasing function of x and $K(1,y)y^{-\frac{1}{q}}$ is strictly decreasing function of y, where $\frac{1}{p} + \frac{1}{q} = 1$,

(iii) $k_p(K) = \int_0^\infty K(1, y) y^{-\frac{1}{q}} dy < \infty$. Let l_p be the Banach space of all complex sequences $x = (x_n)_{n \in \mathbb{N}}$ with the norm $||x||_p =$ $(\sum_{n=1}^{\infty} |x_n|^p)^{\frac{1}{p}} < \infty$ and let $\mathscr{F}(\alpha)$ be a class of all non-negative differentiable functions $f : \mathbb{R}_+ \to \mathbb{R}$ such that $\inf_{x>0} f'(x) \ge \alpha > 0$. If $\phi_i \in \mathscr{F}(\alpha_i), i = 1, 2$, and $K(x, y) f_1(x) f_2(y) \in \mathscr{H}_p$, then

$$\sum_{n,m\geq 1} K(\phi_1(n),\phi_2(m))|a_n||b_m| < \alpha_1^{-\frac{1}{q}} \alpha_2^{-\frac{1}{p}} k_p \left\| \frac{a}{f_1(\phi_1)} \right\|_p \left\| \frac{b}{f_2(\phi_2)} \right\|_q,$$
(10.37)

$$\left[\sum_{m=1}^{\infty} \left(\sum_{n=1}^{\infty} K(\phi_1(n), \phi_2(m)) f_2(\phi_2(m)) |a_n|\right)^p\right]^{\frac{1}{p}} < \alpha_1^{-\frac{1}{q}} \alpha_2^{-\frac{1}{p}} k_p \left\|\frac{a}{f_1(\phi_1)}\right\|_p, \quad (10.38)$$

$$\left[\sum_{n=1}^{\infty} \left(\sum_{m=1}^{\infty} K(\phi_1(n), \phi_2(m)) f_1(\phi_1(n)) | b_m|\right)^q\right]^{\frac{1}{q}} < \alpha_1^{-\frac{1}{q}} \alpha_2^{-\frac{1}{p}} k_p \left\|\frac{b}{f_2(\phi_2)}\right\|_q, \quad (10.39)$$

where $k_p = k_p(K(x,y)f_1(x)f_2(y))$ and $\frac{a}{f_1(\phi_1)} = (\frac{a}{f_1(\phi_1(n))})_{n \in \mathbb{N}}, \frac{b}{f_2(\phi_2)} = (\frac{b}{f_2(\phi_2(m))})_{m \in \mathbb{N}}$ are complex sequences such that $0 < \|\frac{a}{f_1(\phi_1)}\|_p, \|\frac{b}{f_2(\phi_2)}\|_q < \infty$.

In addition, if $\lim_{x\to\infty} \phi'_i(x) = \alpha_i$, i = 1, 2, then the constant $\alpha_1^{-\frac{1}{q}} \alpha_2^{-\frac{1}{p}} k_p$ is the best possible in the above inequalities.

10.28. D. Xin, B. Yang [135], 2011.

Suppose $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let

$$k(\lambda) = \frac{\pi}{\sin \lambda \pi} \left[\frac{\sin \lambda \alpha_1}{\sin \alpha_1} + \frac{\sin \lambda (\pi - \alpha_2)}{\sin \alpha_2} \right],$$

where $0 < |\lambda| < 1$ and $0 < \alpha_1 < \alpha_2 < \pi$. Then the inequalities

$$\int_{-\infty}^{\infty} \int_{-\infty}^{\infty} \min_{i \in \{1,2\}} \left\{ \frac{1}{x^2 + 2xy \cos \alpha_i + y^2} \right\} f(x)g(y)dxdy$$

$$< k(\lambda) \left[\int_{-\infty}^{\infty} |x|^{-p\lambda - 1} f^p(x)dx \right]^{\frac{1}{p}} \left[\int_{-\infty}^{\infty} |y|^{q\lambda - 1} g^q(y)dy \right]^{\frac{1}{q}}$$
(10.40)

and

$$\int_{-\infty}^{\infty} |y|^{p(1-\lambda)-1} \left[\int_{-\infty}^{\infty} \min_{i \in \{1,2\}} \left\{ \frac{1}{x^2 + 2xy \cos \alpha_i + y^2} \right\} f(x) dx \right]^p dy$$

$$< k^p(\lambda) \int_{-\infty}^{\infty} |x|^{-p\lambda-1} f^p(x) dx$$
(10.41)

hold for all non-negative functions $f, g : \mathbb{R} \to \mathbb{R}$ such that $0 < \int_{-\infty}^{\infty} |x|^{-p\lambda-1} f^p(x) dx < \infty$ and $0 < \int_{-\infty}^{\infty} |y|^{q\lambda-1} g^q(y) dy < \infty$. Moreover, inequalities (10.40) and (10.41) are equivalent and include the best possible constant factors on their right-hand sides.

10.29. B. Yang, M. Krnić [168], 2012.

Suppose $\frac{1}{p} + \frac{1}{q} = 1$, p > 1, and let $\alpha \in \mathbb{R}$. Further, let $h : \mathbb{R}_+ \to \mathbb{R}$ be a non-negative function such that $0 < k(\alpha) = \int_0^\infty h(t)t^{\alpha-1}dt < \infty$ and $x^{-\alpha}\sum_{n=1}^\infty h(\frac{n}{x})n^{\alpha-1} < k(\alpha), x \in \mathbb{R}_+$. Then the inequality

$$\int_{0}^{\infty} f(x) \sum_{n=1}^{\infty} h\left(\frac{n}{x}\right) a_{n} dx$$
$$< k(\alpha) \left[\int_{0}^{\infty} x^{p(1+\alpha)-1} f^{p}(x) dx \right]^{\frac{1}{p}} \left[\sum_{n=1}^{\infty} n^{q(1-\alpha)-1} a_{n}^{q} \right]^{\frac{1}{q}}$$
(10.42)

holds for any non-negative function $f : \mathbb{R}_+ \to \mathbb{R}$ and any sequence $(a_n)_{n \in \mathbb{N}}$, provided that $0 < \int_0^\infty x^{p(1+\alpha)-1} f^p(x) dx < \infty$ and $0 < \sum_{n=1}^\infty n^{q(1-\alpha)-1} a_n^q < \infty$. In addition, the constant factor $k(\alpha)$ is the best possible in (10.42).

10.30. M. Krnić [65], 2012.

Suppose p_i, p'_i, q_i , i = 1, 2, ..., n, and λ are real parameters fulfilling conditions (2.26), (2.27) and (2.30) (see Section 2.4, Chapter 2). Further, let $(\Omega_i, \Sigma_i, \mu_i)$ be σ -finite measure spaces, and let $K : \prod_{i=1}^n \Omega_i \to \mathbb{R}, \phi_{ij} : \Omega_j \to \mathbb{R}, f_i : \Omega_i \to \mathbb{R}, i, j = 1, 2, ..., n$, be non-negative

measurable functions. If $\prod_{i,j=1}^{n} \phi_{ij}(x_j) = 1$, then

$$\frac{\left\{\int_{\mathbf{\Omega}} \left[K^{n}(\mathbf{x})\prod_{i=1}^{n}\left(\phi_{ii}\omega_{i}f_{i}\right)^{2p_{i}}\left(x_{i}\right)\omega_{i}^{-q_{i}}\left(x_{i}\right)\prod_{i,j=1,j\neq i}^{n}\phi_{ij}^{q_{i}}\left(x_{j}\right)\right]^{1/(n+1)}d\mu(\mathbf{x})\right\}^{(n+1)M}}{\prod_{i=1}^{n}\|\phi_{ii}\omega_{i}f_{i}\|_{L^{p_{i}}(\mu_{i})}^{2Mp_{i}}} \leq \frac{\int_{\mathbf{\Omega}}K^{\lambda}(\mathbf{x})\prod_{i=1}^{n}f_{i}(x_{i})d\mu(\mathbf{x})}{\prod_{i=1}^{n}\|\phi_{ii}\omega_{i}f_{i}\|_{L^{p_{i}}(\mu_{i})}} \leq \frac{\left\{\int_{\mathbf{\Omega}}\left[K^{n}(\mathbf{x})\prod_{i=1}^{n}\left(\phi_{ii}\omega_{i}f_{i}\right)^{2p_{i}}\left(x_{i}\right)\omega_{i}^{-q_{i}}\left(x_{i}\right)\prod_{i,j=1,j\neq i}^{n}\phi_{ij}^{q_{i}}\left(x_{j}\right)\right]^{1/(n+1)}d\mu(\mathbf{x})\right\}^{(n+1)m}}{\prod_{i=1}^{n}\|\phi_{ii}\omega_{i}f_{i}\|_{L^{p_{i}}(\mu_{i})}^{2mp_{i}}}, (10.43)$$

and

$$(n+1)m\prod_{i=1}^{n} \|\phi_{ii}\omega_{i}f_{i}\|_{L^{p_{i}}(\mu_{i})}$$

$$\times \left\{ 1 - \frac{\int_{\Omega} \left[K^{n}(\mathbf{x}) \prod_{i=1}^{n} (\phi_{ii}\omega_{i}f_{i})^{2p_{i}}(x_{i}) \omega_{i}^{-q_{i}}(x_{i}) \prod_{i,j=1, j\neq i}^{n} \phi_{ij}^{q_{i}}(x_{j}) \right]^{1/(n+1)} d\mu(\mathbf{x}) }{\prod_{i=1}^{n} \|\phi_{ii}\omega_{i}f_{i}\|_{L^{p_{i}}(\mu_{i})}} \right]$$

$$\leq \prod_{i=1}^{n} \|\phi_{ii}\omega_{i}f_{i}\|_{L^{p_{i}}(\mu_{i})} - \int_{\Omega} K^{\lambda}(\mathbf{x}) \prod_{i=1}^{n} f_{i}(x_{i}) d\mu(\mathbf{x})$$

$$\leq (n+1)M\prod_{i=1}^{n} \|\phi_{ii}\omega_{i}f_{i}\|_{L^{p_{i}}(\mu_{i})}$$

$$\times \left\{ 1 - \frac{\int_{\Omega} \left[K^{n}(\mathbf{x}) \prod_{i=1}^{n} (\phi_{ii}\omega_{i}f_{i})^{2p_{i}}(x_{i}) \omega_{i}^{-q_{i}}(x_{i}) \prod_{i,j=1, j\neq i}^{n} \phi_{ij}^{q_{i}}(x_{j}) \right]^{1/(n+1)} d\mu(\mathbf{x}) }{\prod_{i=1}^{n} \|\phi_{ii}\omega_{i}f_{i}\|_{L^{p_{i}}(\mu_{i})}^{2p_{i}/(n+1)}} \right],$$

$$(10.44)$$

where $\mathbf{\Omega} = \prod_{i=1}^{n} \Omega_i$, $\mathbf{x} = (x_1, x_2, \dots, x_n)$, $d\mu(\mathbf{x}) = \prod_{i=1}^{n} d\mu_i(x_i)$,

$$m = \min\left\{\frac{1}{q_1}, \frac{1}{q_2}, \dots, \frac{1}{q_n}, 1 - \lambda\right\}, M = \max\left\{\frac{1}{q_1}, \frac{1}{q_2}, \dots, \frac{1}{q_n}, 1 - \lambda\right\},\$$

and $\omega_i: \Omega_i \to \mathbb{R}$ is defined by

$$\boldsymbol{\omega}_{i}(x_{i}) = \left[\int_{\mathbf{\hat{\Omega}}^{i}} K(\mathbf{x}) \prod_{j=1, j \neq i}^{n} \phi_{ij}^{q_{i}}(x_{j}) d\hat{\mu}^{i}(\mathbf{x}) \right]^{\frac{1}{q_{i}}},$$

where $\hat{\boldsymbol{\Omega}}^{i} = \prod_{j=1, j \neq i}^{n} \Omega_{j}$ and $d\hat{\mu}^{i}(\mathbf{x}) = \prod_{j=1, j \neq i}^{n} d\mu_{j}(x_{j})$.

Bibliography

- M. Abramowitz, I. A. Stegun, *Handbook of mathematical functions with formulas*, graphs, and mathematical tables, National Bureau of Standards, Applied Math. Series 55, 4th printing, Washington 1965.
- [2] K. F. Andersen, *Weighted inequalities for Stieltjes transformation and Hilbert's double series*, Proc. Roy. Soc. Edinburgh Sect. A **86** (1980), 75–84.
- [3] G. Andrews, *The Theory of Partitions*, Encyclopedia of Mathematics and its Applications, Addison-Wesley, Reading, Massachusetts, 1976.
- [4] L. E. Azar, On some extensions of Hardy-Hilbert's inequality and applications, J. Inequal. Appl. Art. ID 546829 (2008), 1–14.
- [5] L. E. Azar, Some extensions of Hilbert's integral inequality, J. Math. Ineqal. 15 (2011), 131–140.
- [6] M. Bencze, C. J. Zhao, *On a few new inequalities similar to Hilbert's inequality*, RGMIA **5**(3), 2002.
- [7] A. Bényi, C. D. Oh, *Best constants for certain multilinear integral operators*, J. Inequal. Appl. Art. ID 28582 (2006), 1–12.
- [8] S. Bloom, R. Kerman, *Weighted Orlicz space integral inequalities for the Hardy-Littlewood maximal operator*, Studia Math. **110** (1994), 149–167.
- [9] F. F. Bonsall, *Inequalities with non-conjugate parameters*, Quart. J. Math. Oxford 2 (1951), 135–150.
- [10] I. Brnetić, J. Pečarić, Generalization of Hilbert's integral inequality, Math. Inequal. Appl. 7 (2004), 199–205.
- [11] I. Brnetić, J. Pečarić, *Generalization of inequalities of Hardy-Hilbert's type*, Math. Inequal. Appl. **7** (2004), 217–225.
- [12] I. Brnetić, M. Krnić, J. Pečarić, Multiple Hilbert and Hardy-Hilbert inequalities with non-conjugate parameters, Bull. Austral. Math. Soc. 71 (2005), 447–457.
- [13] C. T. Chang, J. W. Lan, K. Z. Wang, A new generalization of Hardy-Hilbert's inequality with non-homogeneous kernel, Math. Inequal. Appl. **14** (2011), 1–11.
- [14] A. Čižmešija, J. Pečarić, Some new genaralizations of inequalities of Hardy and Levin-Cochran-Lee, Bull. Austral. Math. Soc. 63 (2001), 105–113.
- [15] A. Čižmešija, J. Pečarić, On Bicheng-Debnath's generalizations of Hardy's Integral inequality, Internat. J. Math. & Math. Sci. 27 (2001), 237–250.

- [16] A. Čižmešija, M. Krnić, J. Pečarić, *General Hilbert's inequality with non-conjugate parameters*, Math. Inequal. Appl. **11** (2008), 237–269.
- [17] A. Čižmesija, I. Perić, P. Vuković, *Inequalities of the Hilbert type in* \mathbb{R}^n with nonconjugate exponents, Proc. Edinb. Math. Soc. **51** (2008), 11–26.
- [18] A. Čižmešija, M. Krnić, J. Pečarić, General Hardy-type inequalities with nonconjugate parameters, Bull. Iranian Math. Soc. 37 (2011), 77–108.
- [19] N. Das, S. Sahoo, A generalization of multiple Hardy-Hilbert's integral inequality, J. Math. Inequal. 3 (2009), 139-154.
- [20] N. Das, S. Sahoo, New inequalities similar to Hardy-Hilbert's inequality, Turk. J. Math. 34 (2010), 153–165.
- [21] S. S. Dragomir, B. Mond, J. E. Pečarić, *A functional inequality and its applications*, J. Math. Anal. Appl. **187** (1994), 287–295.
- [22] G. M. Fihtengolz, A Course in Differential and Integral Calculus, Izdavateljstvo Nauka, Moskva, 1966.
- [23] I. Gavrea, *Some remarks on Hardy-Hilbert's integral inequality*, Math. Inequal. Appl. **5** (2002), 473–477.
- [24] E. K. Godunova, *Generalization of a two-parameter Hilbert inequality*, (russian) Izv. Vysš. Učebn. Zaved. Matematika **56** (1967), 35–39.
- [25] L. Grafakos, C. Morpurgo, A Selberg integral formula and applications, Pacific J. Math. 191 (1999), 85–94.
- [26] R. L. Graham, D. E. Knuth, D. Patashnik, Concrete mathematics A Foundation for Computer Science, Addison-Wesley Publishing Company, 1989.
- [27] G. D. Handley, J. J. Koliha, J. Pečarić, A Hilbert type inequality, Tamkang J. Math. 31 (2000), 311–315.
- [28] G. D. Handley, J. J. Koliha, J. Pečarić, New Hilbert-Pachpatte type integral inequalities, J. Math. Anal. Appl. 257 (2001), 238–250.
- [29] G. D. Handley, J. J. Koliha, J. Pečarić, *Hilbert-Pachpatte type multidimensional integral inequalities*, J. Inequal. Pure Appl. Math. 5 Art. 34 (2004), 1–8.
- [30] G. H. Hardy, Note on a theorem of Hilbert, Math. Zeitsch. 6 (1920), 314–317.
- [31] G. H. Hardy, *Note on a theorem of Hilbert concerning series of positive terms*, Proc. London Math. Soc. **23** (1925), Records of Proc. XLV–XLVI.
- [32] G. H. Hardy, Notes on some points in the integral calculus, LX. An inequality between integrals, Messenger Math. 54 (1925), 150–156.
- [33] G. H. Hardy, J. E. Littlewood, G. Pólya, *Inequalities*, second edition, Cambridge University Press, Cambridge, 1967.
- [34] B. He, Y. Li, A new Hilbert's inequality with logarithm, RGMIA 9(2), 2006.
- [35] B. He, Y. Li, *On several new inequalities close to Hilbert-Pachpatte's inequality*, J. Inequal. Pure Appl. Math. **7** Art. 154 (2006), 1–9.
- [36] L. Hua, H. Ke, Y. Li, *The generalizations of Hilbert's inequality*, Internat. J. Math. & Math. Sci. Art. ID 918534 (2008), 1–12.

- [37] Q. Huang, B. Yang, *On a multiple Hilbert-type integral operator and applications*, J. Inequal. Appl. Art. ID 192197 (2009), 1–13.
- [38] Q. Huang, On a multiple Hilbert's inequality with parameters, J. Inequal. Appl. Art. ID 309319 (2010), 1–12.
- [39] K. Jichang, Introduction to real analysis, Hunan Education Press, Chansha, 1996.
- [40] K. Jichang, On new extension of Hilbert's integral inequality, J. Math. Anal. Appl. 235 (1999), 608–614.
- [41] K. Jichang, L. Debnath, On new generalizations of Hilbert's inequality and their applications, J. Math. Anal. Appl. 245 (2000), 248–265.
- [42] K. Jichang, T. M. Rassias, *Hilbert integral operator inequalities*, Math. Inequal. Appl. 3 (2000), 497–510.
- [43] K. Jichang, L. Debnath, On Hilbert's type inequalities on the weighted Orlicz spaces, Pacific J. Appl. Math. 1 (2007), 95–103.
- [44] J. Jin, On Hilbert's type inequalities, J. Math. Anal. Appl. 340 (2008), 932–942.
- [45] J. Jin, L. Debnath, On a Hilbert-type linear series operator and its applications, J. Math. Anal. Appl. 371 (2010), 691–704.
- [46] Y. H. Kim, B. I. Kim, *An anlogue of Hilbert's inequality and its extensions*, Bull. Korean Math. Soc. **39** (2002), 377–388.
- [47] Y. H. Kim, Some new inverse-type Hilbert-Pachpatte integral inequalities, Acta Math. Sinica 20 (2004), 57–62.
- [48] V. Kiryakova, Poisson and Rodrigues type fractional differential formulas for the generalized hypergeometric function ${}_{p}F_{q}$, Atti. Sem. Mat. Fis. Univ. Modena, **39** (1994), 311–322.
- [49] V. Kiryakova, Generalized Fractional Calculus and Applications, Longman Sci. & Techn., Harlow and J. Wiley & Sons, N. York, (1994).
- [50] V. Kiryakova, All the special functions are fractional differintegrals of elementary functions, J. Phys. A: Math. Gen. **30** (1997), 5085–5103.
- [51] V. Kokilashvili, M. Krbec, *Weighted inequalities in Lorentz and Orlicz spaces*, World Scientific, 1991.
- [52] M. Krnić, J. Pečarić, *Hilbert's inequalities and their reverses*, Publ. Math. Debrecen 67 (2005), 315–331.
- [53] M. Krnić, J. Pečarić, General Hilbert's and Hardy's inequalities, Math. Inequal. Appl. 8 (2005), 29–52.
- [54] M. Krnić, G. Mingzhe, J. Pečarić, G. Xuemei, On the best constant in Hilbert's inequality, Math. Inequal. Appl. 8 (2005), 317–329.
- [55] M. Krnić, J. Pečarić, *Extension of Hilbert's inequality*, J. Math. Anal. Appl. 324 (2006), 150–160.
- [56] M. Krnić, J. Pečarić, On some general Hilbert's inequality with non-conjugate parameters, Rend. Circ. Mat. Palermo 56 (2007), 219-234.
- [57] M. Krnić, J. Pečarić, A Hilbert inequality and Euler-Maclaurin summation formula, ANZIAM J. **48** (2007), 419–431.

- [58] M. Krnić, J. Pečarić, Some improvements on general Hilbert inequality, Nonlinear Funct. Anal. Appl. 12 (2007), 21–40.
- [59] M. Krnić, J. Pečarić, P. Vuković, On some higher-dimensional Hilbert's and Hardy-Hilbert's type integral inequalities with parameters, Math. Inequal. Appl. 11 (2008), 701–716.
- [60] M. Krnić, Ż. Tomovski, J. Pečarić, *Hilbert inequalities related to generalized hypergeometric functions* $_{m}F_{n}$, Math. Balkanica, **22** (2008), 307–322.
- [61] M. Krnić, I. Perić, J. Pečarić, P. Vuković, *Hardy-Hilbert type inequalities with fractional kernel in* \mathbb{R}^n , J. Inequal. Pure Appl. Math. **10** Art. 115 (2009), 1–19.
- [62] M. Krnić, *Hilbert inequality and Gaussian hypergeometric functions*, J. Math. Inequal. **3** (2009), 645–656.
- [63] M. Krnić, J. Pečarić, P. Vuković, Discrete Hilbert-type inequalities with general homogeneous kernels, Rend. Circ. Mat. Palermo 60 (2011), 161–171.
- [64] M. Krnić, Multidimensional Hilbert-type inequality on the weighted Orlicz spaces, Mediterranean J. Math. 9 (2012), 883–895.
- [65] M. Krnić, On some refinements and converses of multidimensional Hilbert-type inequalities, Bull. Austral. Math. Soc. **85** (2012), 380–394.
- [66] M. Krnić, A refined discrete Hilbert inequality via the Hermite-Hadamard inequality, Comput. Math. Appl. **63** (2012), 1587–1596.
- [67] V. I. Krylov, *Approximate Calculation of Integrals*, The Macmillan Company, New York, 1962.
- [68] A. Kufner, Zur Hardischen Ungleichung, Seminar Analysis 1982/83, IMath, Berlin 1983, 1–18.
- [69] A. Kufner, B. Opic, *Hardy-type Inequalities*, Pitman Researchs Notes in Mathematics Series 219, Longman Scientifics & Tehnical, Harlow, 1990.
- [70] A. Kufner, L. E. Persson, *Weighted Inequalities of Hardy type*, World Scientific, Publishing Co. Singapore/New Jersey/London/Hong Kong, 2003.
- [71] A. Kufner, L. Maligranda, L. E. Persson, *The Hardy inequality About its history and some related results*, Vydavatelský servis, Pilsen, 2007.
- [72] S. Kurepa, *Konačnodimenzionalni vektorski prostori i primjene*, Sveučilišna naklada Liber, Zagreb, 1986.
- [73] E. Landau, *A note on a theorem concerning series of positive terms*, J. Lond. Math. Soc. **1** (1926), 38–39.
- [74] H. Leping, G. Mingzhe, S. Wei, A note on Hilbert's inequality, Math. Inequal. Appl. 6 (2003), 283–288.
- [75] H. Leping, G. Mingzhe, J. Weijan, *On Hardy-Hilbert's integral inequality with parameters*, J. Inequal. Pure Appl. Math. **4** Art. 94 (2003), 1–8.
- [76] H. Leping, J. Pečarić, Y. Yu, On a further improvement of the extended Hardy-Hilbert's inequality, RGMIA 7(1), 2004.
- [77] H. Leping, G. Mingzhe, Z. Yu, *On new extensions of Hilbert's integral inequality*, Internat. J. Math. & Math. Sci. Art. ID 297508 (2008), 1–8.
- [78] H. Leping, G. Xuemei, G. Mingzhe, On a new weighted Hilbert inequality, J. Inequal. Appl. Art. ID 637397 (2008), 1–10.
- [79] V. Levin, On the two-parameter extension and analogue of Hilbert's inequality, J. London Math. Soc. **11** (1936), 119–124.
- [80] Y. Li, B. He, Some extensions of Hilbert's type inequality and its applications, J. Ineqal. Pure Appl. Math. 7 Art. 61 (2006), 1–7.
- [81] Y. Li, J. Wu, B. He, *A new Hilbert-type integral inequality and the equivalent form*, Internat. J. Math. & Math. Sci. Art. ID 45378 (2006), 1–6.
- [82] Y. Li, Z. Wang, B. He, Hilbert's type linear operator and some extensions of Hilbert's inequality, J. Ineqal. Appl. Art. ID 82138 (2007), 1–10.
- [83] E. H. Lieb, *Sharp constants in the Hardy-Littlewood-Sobolev and related inequalities*, Ann. of Math. **118** (1983), 349–374.
- [84] E. H. Lieb, M. Loss, *Analysis*, second edition, Graduate Studies in Mathematics 14, American Mathematical Society, Providence, RI, 2001.
- [85] X. Liu, B. Yang, On a new Hilbert-Hardy type integral operator and applications, J. Inequal. Appl. Art. ID 812636 (2010), 1–10.
- [86] Z. Lü, Some new inequalities similar to Hilbert-Pachpatte type inequalities, J. Ineqal. Pure Appl. Math. 4 Art. 33 (2003), 1–8.
- [87] Z. Lü, G. Mingzhe, L. Debnath, On new generalizations of the Hilbert integral inequality, J. Math. Anal. Appl. 326 (2007), 1452-1457.
- [88] Z. Lü, H. Xie, *New inequalities similar to Hardy-Hilbert inequality and their applications*, J. Inequal. Appl. Art. ID 90641 (2007), 1–12.
- [89] P. A. Macmahon, *Combinatory Analysis I, II*, Chelsea, New York, 1960.
- [90] L. Maligranda, *Generalized Hardy inequalities in rearrangement invariant spaces*, J. Math. Pures Appl. **59** (1980), 405–415.
- [91] L. Maligranda, *Orlicz spaces and interpolation*, Seminars in Math. 5, Univ. of Campinas, Campinas SP, Brazil 1989.
- [92] M. L. Mehta, *Random Matrices*, second edition, Academic Press, Boston, MA, 1990.
- [93] G. Mingzhe, A note on Hilbert double series theorem, Hunan Ann. Math. **12** (1992), 142–147.
- [94] G. Minzghe, *An improvement of Hardy-Riesz's extension of the Hilbert inequality*, J. Math. Res. Exposition **14** (1994), 255–259.
- [95] G. Mingzhe, On Hilbert's inequality and its applications, J. Math. Anal. Appl. 212 (1997), 316–323.
- [96] G. Mingzhe, On Hilbert's inequality, Math. Anal. Appl. 11, (1998), 32-35.
- [97] G. Mingzhe, B. Yang, On the extended Hilbert's inequality, Proc. Amer. Math. Soc. 126 (1998), 751–759.
- [98] G. Mingzhe, T. Li, L. Debnath, Some improvements on Hilbert's integral inequality, J. Math. Anal. Appl. 229 (1999), 682–689.

- [99] G. Mingzhe, W. Shongrong, H. Leping, *On the Hilbert inequality with weights*, Z. Anal. Anwendungen **21** (2002), 257–263.
- [100] G. Mingzhe, G. Xuemei, On the generalized Hardy-Hilbert's inequality and its applications, Math. Inequal. Appl. 7 (2004), 19–26.
- [101] G. Mingzhe, L. C. Hsu, A survey of various refinements and generalizations of Hilbert's inequalities, J. Math. Res. Exposition 25 (2005), 227–243.
- [102] D. S. Mitrinović, J. E. Pečarić, A. M. Fink, *Inequalities involving functions and their integrals and derivatives*, Kluwer Academic, Boston, 1991.
- [103] D. S. Mitrinović, J. E. Pečarić, A. M. Fink, *Classical and new inequalities in anal-ysis*, Kluwer Academic Publishers, Dordrecht/Boston/London, 1993.
- [104] Ž. Mitrović, On a generalization of Fan-Todd's inequality, Univ. Beograd Publ. Elektrotehn. Fak. Ser. Mat. Fiz. 412-460 (1973), 151–154.
- [105] B. G. Pachpatte, On some new inequalities similar to Hilbert's inequality, J. Math. Anal. Appl. 226 (1998), 166–179.
- [106] B. G. Pachpatte, Inequalities similar to the integral analogue of Hilbert's inequality, Tamkang J. Math. 30 (1999), 139–146.
- [107] B. G. Pachpatte, Inequalities similar to certain extensions of Hilbert's inequality, J. Math. Anal. Appl. 243 (2000), 217–227.
- [108] T. C. Peachey, Some integral inequalities related to Hilbert's, J. Ineqal. Pure Appl. Math. 4 Art. 19 (2003), 1–8.
- [109] J. E. Pečarić, I. Perić, P. Vuković, *Hilbert-Pachpatte type inequalities from Bonsall's form of Hilbert's inequality*, J. Ineqal. Pure Appl. Math. 9 Art. 9 (2008), 1–13.
- [110] J. Pečarić, I. Perić, P. Vuković, *Refinements of Hilbert's inequality involving the Laplace transform*, Tamsui Oxf. J. Math. Sci. 25 (2009), 71–85.
- [111] I. Perić, P. Vuković, Hardy-Hilbert's inequalities with a general homogeneous kernel, Math. Inequal. Appl. 12 (2009), 525–536.
- [112] I. Perić, P. Vuković, *Hilbert-type inequalities with a product-type homogeneous ker*nel and Schur polynomials, J. Math. Anal. Appl. **359** (2009), 786–793.
- [113] I. Perić, P. Vuković, *Multiple Hilbert's type inequalities with a homogeneous kernel*, Banach J. Math. Anal. **5** (2011), 33–43.
- [114] M. M. Rao, Z. D. Ren, *Theory of Orlicz spaces*, Marcel Dekker, New York, 1991.
- [115] W. Rudin, Real and Complex Analysis, McGraw-Hill, 1970.
- [116] I. Schur, Bemerkungen zür Teorie der beschränkten Bilinearformen mit unendlich vielen Veränderlichen, Journal F. Math. **140** (1911), 1–28.
- [117] G. J. Sinnamon, Operators on Lebesgue Spaces with General Measures, Thesis, McMaster Univ, 1987.
- [118] E. M. Stein, G. Weiss, Fractional integrals in n-dimensional Euclidean space, J. Math. Mech. 7 (1958), 503–514.
- [119] E. M. Stein, Singular integrals and differentiability properties af functions, Princeton University Press, Princeton, New Jersey, 1970.

- [120] E. M. Stein, G. Weiss, Introduction to Fourier Analysis on Euclidean Spaces, Princeton University Press, Princeton, New Jersey, 1971.
- [121] E. M. Stein, Harmonic Analysis: Real-Variable Methods, Orthogonality and Oscillatory Integrals, Princeton University Press, Princeton, New Jersey, 1993.
- [122] W. T. Sulaiman, Four inequalities similar to Hardy-Hilbert's integral inequality, J. Ineqal. Pure Appl. Math. 7 Art. 76 (2006), 1–8.
- [123] W. T. Sulaiman, On three inequalities similar to Hardy-Hilbert's integral inequality, Acta Math. Univ. Comenianae **76** (2007), 273–278.
- [124] W. T. Sulaiman, *Hardy-Hilbert's type inequalities for* (p,q)- $H\ddot{o}(0,\infty)$ *functions*, J. Math. Inequal. **2** (2008), 323–334.
- [125] B. Sun, A multiple Hilbert-type integral inequality with the best constant factor, J. Inequal. Appl. Art. ID 71049 (2007), 1–14.
- [126] S. K. Sunanda, C. Nahak, S. Nanda, Some new inequalities similar to Hardy-Hilbert's inequality, Math. Inequal. Appl. 13 (2010), 601–611.
- [127] P. Vuković, Multidimensional Hilbert-type inequalities with a homogeneous kernel, J. Inequal. Appl. Art. ID 130958 (2009), 1–12.
- [128] P. Vuković, Note on Hilbert-type inequalities, Turk. J. Math. 36 (2012), 253–262.
- [129] W. Wang, D. Xin, On a new strengthened version of a Hardy-Hilbert type inequality and applications, J. Ineqal. Pure Appl. Math. 7 Art. 180 (2006), 1–7.
- [130] J. Weijian, G. Mingzhe, An extended Hardy-Hilbert inequality and its applications, J. Ineqal. Pure Appl. Math. 7 Art. 30 (2006), 1–9.
- [131] G. Xi, A reverse Hardy-Hilbert-type inequality, J. Inequal. Appl. Art. ID 79758 (2007), 1–7.
- [132] G. Xi, A reverse Hardy-Hilbert-type integral inequality, J. Ineqal. Pure Appl. Math. 9 Art. 49 (2008), 1–9.
- [133] Z. Xie, Z. Zheng, A Hilbert-type integral inequality whose kernel is a homogeneous form of degree -3, J. Math. Anal. Appl. 339 (2008), 324-331.
- [134] D. Xin, A Hilbert-type integral inequality with a homogeneous kernel of zero degree, Math. Theory Appl. 30 (2010), 70–74.
- [135] D. Xin, B. Yang, A Hilbert-type integral inequality in the whole plane with the homogeneous kernel of degree -2, J. Ineqal. Appl. Art. ID 401428 (2011), 1–11.
- [136] B. Yang, G. Mingzhe, On a best value of Hardy-Hilbert's inequality, Adv. Math. 26 (1997), 159–164.
- [137] B. Yang, Z. Zhuohua, L. Debnath, On new generalizations of Hardy's integral inequality, J. Math. Anal. Appl. 217 (1998), 321–327.
- [138] B. Yang, On Hilbert's integral inequality, J. Math. Anal. Appl. 220 (1998), 778-785.
- [139] B. Yang, L. Debnath, On new strengthened Hardy-Hilbert's inequality, Internat. J. Math. & Math. Sci. 21 (1998), 403–408.
- [140] B. Yang, On a strengthened version of the more accurate Hardy-Hilbert's inequality, Acta. Math. Sinica **42** (1999), 1103–1110.

- [141] B. Yang, L. Debnath, Generalizations of Hardy's Integral Inequalities, Internat. J. Math. & Math. Sci. 22 (1999), 535–542.
- [142] B. Yang, L. Debnath, On a new generalization of Hardy-Hilbert's inequality and its applications, J. Math. Anal. Appl. 233 (1999), 484–497.
- [143] B. Yang, L. Debnath, Some inequalities involving π and an application to Hilbert's inequality, Appl. Math. Lett. **129** (1999), 101–105.
- [144] B. Yang, On a general Hardy-Hilbert's inequality with a best value, Chinese Ann. Math. Ser. A 21 (2000), 401–408.
- [145] B. Yang, On a strengthened Hardy-Hilbert inequality, J. Ineqal. Pure Appl. Math. 1 Art. 22 (2000), 1–6.
- [146] B. Yang, On new generalizations of Hilbert's inequality, J. Math. Anal. Appl. 248 (2000), 29–40.
- [147] B. Yang, A generalization of Hilbert double series theorem, J. Nanjing Univ. Math. Biquart. 18 (2001), 145–152.
- [148] B. Yang, On an extension of Hardy-Hilberts inequality, Chinese Ann. Math. Ser. A 23 (2002), 247–254.
- [149] B. Yang, A new inequality similar to Hilbert's inequality, J. Ineqal. Pure Appl. Math. 3, Art. 75 (2002), 1–6.
- [150] B. Yang, L. Debnath, On the extended Hardy-Hilbert's inequality, J. Math. Anal. Appl. 272 (2002), 187–199.
- [151] B. Yang, On new inequality similar to Hardy-Hilbert's inequality, Math. Ineqal. Appl. 6 (2003), 37–44.
- [152] B. Yang, T. M. Rassias, On the way of weight coefficients and research for the Hilbert-type inequalities, Math. Inequal. Appl. 6 (2003), 625–658.
- [153] B. Yang, On generalizations of Hardy-Hilberts inequality and its equivalent form, J. Math. 17 (2004), 24–30.
- [154] B. Yang, On a new multiple extension of Hilbert's integral inequality, J. Ineqal. Pure Appl. Math. 6 Art. 39 (2005), 1–8.
- [155] B. Yang, A relation to Hardy-Hilbert's integral inequality and Mulholland's inequality, J. Ineqal. Pure Appl. Math. 6 Art. 112 (2005), 1–12.
- [156] B. Yang, I. Brnetić, M. Krnić, J. Pečarić, *Generalization of Hilbert and Hardy-Hilbert integral inequalities*, Math. Inequal. Appl. 8 (2005), 259–272.
- [157] B. Yang, On Hilbert's integral inequality, Acta. Math. Sinica 49 (2006), 1121–1126.
- [158] B. Yang, On the norm of an integral operator and applications, J. Math. Anal. Appl. 321 (2006), 182–192.
- [159] B. Yang, A new Hilbert-type inequality, Bull. Belg. Math. Soc. 13 (2006), 479–487.
- [160] B. Yang, A Hilbert-type integral inequality, J. Zhejiang Univ. Sci. Ed. 24 (2007), 121–124.
- [161] B. Yang, On the norm of a Hilbert's type linear operator and applications, J. Math. Anal. Appl. 325 (2007), 529-541.

- [162] B. Yang, An extension on the Hilbert-type inequality and its reverse, J. Math. Inequal. 2 (2008), 139–149.
- [163] B. Yang, A relation to Hilbert's integral inequality and some basic Hilbert-type inequalities, J. Inequal. Pure Appl. Math. 9 (2008), 1–8.
- [164] B. Yang, On a Hilbert-type operator with a class of homogeneous kernels, J. Ineqal. Appl. Art. ID 572176 (2009), 1–9.
- [165] B. Yang, The norm of operator and Hilbert-type inequalities, Science Press, Beijing, 2009.
- [166] B. Yang, M. Krnić, Hilbert-type inequalities and related operators with homogeneous kernel of degree 0, Math. Inequal. Appl. 13 (2010), 817–829.
- [167] B. Yang, M. Krnić, On the norm of multidimensional Hilbert-type operator, Sarajevo J. Math. 7 (2011), 223–243.
- [168] B. Yang, M. Krnić, A half-discrete Hilbert-type inequality with a general homogeneous kernel of degree 0, J. Math. Ineqal., in press.
- [169] W. Yang, Some new Hilbert-Pachpatte inequalities, J. Ineqal. Pure Appl. Math. 10 Art. 26 (2009), 1–14.
- [170] W. Yang, Some new Hilbert-Pachpatte's inequalities for a finite number of nonnegative sequences, RGMIA 12(2), 2009.
- [171] W. Yang, Some new inequalities similar to Hilbert-Pachpatte's inequalities with two independent variable, RGMIA 12(2), 2009.
- [172] H. Yong, All-sided generalization about Hardy-Hilberts integral inequalities, Acta Math. Sinica 44 (2001), 691–626.
- [173] H. Yong, A multiple Hardy-Hilbert integral inequality with the best constant factor, J. Inequal. Pure Appl. Math. 7 Art. 139 (2006), 1–10.
- [174] H. Yong, On multiple Hardy-Hilbert's integral inequalities with some parameters, J. Ineqal. Appl. Art. ID 94960 (2006), 1–11.
- [175] Z. Yu, G. Xuemei, G. Mingzhe, An extension of the Hilbert's integral inequality, J. Ineqal. Appl. Art. ID 158690 (2009), 1–12.
- [176] Z. Zeng, Z. Xie, On a new Hilbert-type integral inequality with the integral in whole plane, J. Ineqal. Appl. Art. ID 256796 (2010), 1–8.
- [177] K. Zhang, A bilinear inequality, J. Math. Anal. Appl. 271 (2002), 288–296.
- [178] C. Zhao, L. Debnath, Some new inverse type Hilbert integal inequalities, J. Math. Anal. Appl. 262 (2001), 411–418.
- [179] C. J. Zhao, W. S. Cheung, Some new Hilbert's type inequalities, J. Ineqal. Appl. Art. ID 851360 (2009), 1–10.
- [180] W. Zhong, B. Yang, On a multiple Hilbert-type integral inequality with the symmetric kernel, J. Ineqal. Appl. Art. ID 27962 (2007), 1–17.
- [181] W. Zhong, The Hilbert-type integral inequalities with a homogeneous kernel of -λdegree, J. Ineqal. Appl. Art. ID 917392 (2008), 1–12.
- [182] W. Zhong, A Hilbert-type linear operator with the norm and its applications, J. Ineqal. Appl. Art. ID 494257 (2009), 1–18.

- [183] W. Zhong, A new Hilbert-type linear operator with a composite kernel and its applications, J. Ineqal. Appl. Art. ID 393025 (2010), 1–18.
- [184] A. Zygmund, *Trigonometric series*, second edition, Cambridge Univ. Press, Cambridge, 1959.

Zagreb, March 2013

246